ADVANCES IN NUMBER THEORY AND APPLIED ANALYSIS

Editors

Pradip Debnath Hari Mohan Srivastava Kalyan Chakraborty Poom Kumam



ADVANCES IN NUMBER THEORY AND APPLIED ANALYSIS

This page intentionally left blank

ADVANCES IN NUMBER THEORY AND APPLIED ANALYSIS

Editors

Pradip Debnath Assam University, India

Hari Mohan Srivastava University of Victoria, Canada

Kalyan Chakraborty

Kerala School of Mathematics, India

Poom Kumam King Mongkut's University of Technology Thonburi (KMUTT), Thailand



Published by
World Scientific Publishing Co. Pte. Ltd.
5 Toh Tuck Link, Singapore 596224
USA office: 27 Warren Street, Suite 401-402, Hackensack, NJ 07601
UK office: 57 Shelton Street, Covent Garden, London WC2H 9HE

Library of Congress Control Number: 2023006252

British Library Cataloguing-in-Publication Data

A catalogue record for this book is available from the British Library.

ADVANCES IN NUMBER THEORY AND APPLIED ANALYSIS

Copyright © 2023 by World Scientific Publishing Co. Pte. Ltd.

All rights reserved. This book, or parts thereof, may not be reproduced in any form or by any means, electronic or mechanical, including photocopying, recording or any information storage and retrieval system now known or to be invented, without written permission from the publisher.

For photocopying of material in this volume, please pay a copying fee through the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923, USA. In this case permission to photocopy is not required from the publisher.

ISBN 978-981-127-259-2 (hardcover) ISBN 978-981-127-260-8 (ebook for institutions) ISBN 978-981-127-261-5 (ebook for individuals)

For any available supplementary material, please visit https://www.worldscientific.com/worldscibooks/10.1142/13314#t=suppl

Desk Editors: Soundararajan Raghuraman/Rok Ting Tan

Typeset by Stallion Press Email: enquiries@stallionpress.com

Printed in Singapore

Preface

Historically, number theory has been the oldest branch of mathematics which was initially studied for pure beauty of the subject rather than its applications. But with the rapid advancement of science and technology and encouragement toward interdisciplinary research, the demarcation between different branches of mathematics became more and more blurred. Number theory and analysis have been very closely linked with each other. Analytic number theory, theory of curved spaces, and complex analysis are just a few topics to be named where applications of both these branches are heavily exploited.

Presently, the exploration of the applications of different techniques and tools of number theory and mathematical analysis are extensively prevalent in various areas of engineering, mathematical, physical, biological and statistical sciences. This book presents the most recent developments in these two fields through contributions from eminent scientists and mathematicians worldwide.

The book focuses on the current state-of-the-art development in these two areas through original new contributions and surveys. As such, readers will find several useful tools and techniques to develop their skills and expertise in number theory and applied analysis. New research directions are also indicated in each of the chapters. This book is meant for graduate students, faculty and researchers willing to expand their knowledge in number theory and mathematical analysis. The readers of this book will require the minimum prerequisites of analysis, topology, number theory and functional analysis.

Our book is an effort toward presenting the two very important topics in modern mathematics and their applications in diverse areas of science. This book consists of 18 chapters. The first half (Chapters 1–8) of the book deals with applications of number theory, whereas the second half (Chapters 9–18) will elaborate recent applications of mathematical analysis. The first two chapters present surveys on open problems in number theory, properties of partition function and their potential future research directions. The third chapter presents a discussion on the two old standing number theory problems. The fourth chapter is about zeta-functions and allied theta-functions. Sylvester sums on the Frobenius set in arithmetic progression with initial gaps are discussed in Chapter 5. Chapter 6 presents arithmetic properties of minimal excludants of partitions of integers. In Chapter 7, a survey on ℓ -regular partitions is given with recent developments. Chapters 8 elucidates some applications of an algorithm in number theory. From Chapter 9 onwards, the studies in the applications of mathematical analysis have been listed. Chapter 9 presents the component exponential function in scator hypercomplex space. Chapters 10–12 explores the applications of analysis in fixed point results, while Chapter 13 describes sufficient conditions for Mittag-Leffler functions associated with conic regions. Chapters 14–17 respectively investigate controlled optimization problems, integrable function spaces, Borel distribution series and conjugate method for solving nonlinear system of equations. Finally, Chapter 18 presents a closed form of integral transforms in terms of Lauricella function and their numerical simulations.

About the Editors



Pradip Debnath is an Assistant Professor in Mathematics at the Department of Applied Science and Humanities, Assam University, Silchar (a central university). He received his PhD in Mathematics from the National Institute of Technology Silchar, India. His research interests include fixed point theory, nonlinear functional analysis, soft computing and mathematical statistics. He has published more than 60 papers in various journals of international

repute and is an active reviewer for more than 40 international journals. He is also a reviewer for *Mathematical Reviews* published by the American Mathematical Society. He is the Lead Editor of the books *Metric Fixed Point Theory: Applications in Science, Engineering* and Behavioural Sciences (2021, Springer Nature), Soft Computing Techniques in Engineering, Health, Mathematical and Social Sciences (2021, CRC Press), Fixed Point Theory and Fractional Calculus: Recent Advances and Applications (2022, Springer Nature) and Soft Computing: Recent Advances and Applications in Engineering and Mathematical Sciences (2023, CRC Press). He is a topical advisory panel member of the journal Axioms and a Guest Editor of the special issue entitled "Nonlinear Functional Analysis in Natural Sciences" in the same journal. He has successfully guided PhD students in the areas of nonlinear analysis, soft computing and fixed point theory. He has recently completed a major Basic Science Research Project in fixed point theory funded by the UGC, the Government of India. Having been an academic gold medalist during his post graduation studies from Assam University, Silchar, Dr. Debnath has qualified several national-level examinations in mathematics in India.



Hari M. Srivastava is Professor Emeritus in the Department of Mathematics and Statistics at the University of Victoria in Canada since 01 July 2006. He has been Clarivate Analytics [Thomson-Reuters] (Web of Science) Highly Cited Researcher for the years 2015, 2017, 2018, 2019, 2020, 2021 and 2022. He has also been listed and ranked in the sixth place in General Mathematics among the Top 2% scientists in

the world. In fact, it is not quite possible to list out his tremendous academic achievements in a few sentences. The details of the same may be found in his regularly-updated website www.math.uvic.ca/ ~harimsri/.



Kalyan Chakraborty is at present the Director of the Kerala School of Mathematics, India. Previously, he was working as a Professor at Harish-Chandra Research Institute (HRI), Allahabad, India, where he also obtained his PhD in Mathematics. Professor Chakraborty was a postdoctoral fellow at IMSc, Chennai, and at Queen's University, Canada, and a visiting scholar at the University of Paris VI, VII, France: Tokyo Metropolitan University, Japan;

Universitá Roma Tre, Italy; the University of Hong Kong, Hong Kong; Northwest University and Shandong University, China; Mahidol University, Thailand; Mandalay University, Myanmar; and many more. His broad area of research is number theory, particularly class groups, Diophantine equations, automorphic forms, arithmetic functions, elliptic curves, and special functions. He has published more than 60 research articles in respected journals and two books on number theory and has been on the editorial boards of various leading journals. Professor Chakraborty is Vice President of the Society for Special Functions and their Applications.



Poom Kumam received the BS, MSc, and PhD degrees in mathematics from Burapha University (BUU), Chiang Mai University (CMU), and Naresuan University (NU), respectively. In 2008, he received a grant from Franco-Thai Cooperation for a short-term visit to the Laboratoire de Mathematiques, Universite de Bretagne Occidentale, France. He was also a Visiting Professor for a short-term research with Professor Anthony To-Ming Lau at the

University of Alberta, AB, Canada. He is currently a Full Professor with the Department of Mathematics. King Mongkut's University of Technology Thonburi (KMUTT), where he is also the Head of the KMUTT Fixed Point Theory and Applications Research Group since 2007 and also leading the Theoretical and Computational Science Center (TaCS-Center) in 2014 (now TaCS-Center of Excellence in 2021). He has successfully advised five Master's, and 44 PhD graduates. He had won some of the most important awards for mathematicians. The first one is the TRF-CHE-Scopus Young Researcher Award in 2010 which is the award given by the corporation from three organizations: Thailand Research Fund (TRF), the Commission of Higher Education (CHE), and Elsevier Publisher (Scopus). The second award was in 2012 when he received the TWAS Prize for Young Scientist in Thailand, which is given by the Academy of Sciences for the Developing World TWAS (UNESCO) together with the National Research Council of Thailand. In 2014, the third award is the Fellowship Award for Outstanding Contribution to Mathematics from International Academy of Physical Science, Allahabad, India. In 2015, Dr. Poom Kumam has been awarded Thailand Frontier Author Award 2015, an award for outstanding researcher who has published works and has often been used as a reference or evaluation criteria of the database Web of Science. Moreover, in 2016, Dr. Poom Kumam has been awarded 2016 Thailand Frontier Researcher Awards on Innovation Forum: Discovery, Protection, Commercialization By Intellectual Property & Science, and Thomson Reuters. Dr. Poom Kumam has been a highly cited researcher (HCR 2015, 2016, 2017). Moreover, he has received KMUTT-HALL OF FAME 2017, in honour of the recipients of academic awards, KMUTT Young Researcher Awards, Excellence in Teaching Awards for 2016.

In 2019, he received 2019 CMMSE Prize Winner: The CMMSE prize is given to computational researchers for important contributions in the developments of numerical methods for physics, chemistry, engineering and economics from CMMSE Conference June 30–July 6, 2019, Rota, Cadiz, Spain. He has also been listed and ranked in the 197th Place in General Mathematics among the Top 2% Scientists in the World 2021 (Published by Stanford University in USA).

He served on the editorial boards of various international journals and has also published more than 800 papers in Scopus and Web of Science (WoS) databases the also delivers many invited talks on different international conferences every year all around the world. Furthermore, his research interest focuses on fixed point theory, fractional differential equations and optimization related with optimization problems in both pure science and applied science.

Acknowledgments

The editors are immensely thankful to all the reviewers from different countries for investing their valuable time and reviewing the chapters of this book, which resulted in a drastic improvement in the quality of the manuscript.

The editors express their special gratitude to Ms. Tan Rok Ting (Senior Editor, World Scientific Publishing) for her wonderful support during the entire publication process. We also highly appreciate the efforts of the production team of World Scientific Publishing Co. This page intentionally left blank

© 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_fmatter

Contents

Preface		v
About the Editors		vii
Acknowledg	ments	xi
Chapter 1.	Open Problems in Number Theory Pradip Debnath	1
Chapter 2.	Arithmetic and Congruence Properties of Partition Function Nabanita Konwar	13
Chapter 3.	Collatz Hypothesis and Kurepa's Conjecture Nicola Fabiano, Nikola Mirkov, Zoran D. Mitrović, and Stojan Radenović	31
Chapter 4.	On Zeta Functions and Allied Theta Functions Hongyu Li, Takako Kuzumaki, and Shigeru Kanemitsu	51
Chapter 5.	Sylvester Sums on the Frobenius Set in Arithmetic Progression with Initial Gaps <i>Takao Komatsu</i>	99

Chapter 6.	Arithmetic Properties of Minimal Excludants of Partitions of Integers Nipen Saikia		
Chapter 7.	On Certain $\ell\text{-regular}$ Partitions: A Brief Survey Chiranjit Ray		
Chapter 8.	Some Applications of an Algorithm in Number Theory <i>Nihal Özgür</i>		
Chapter 9.	The Components Exponential Function in Scator Hypercomplex Space: Planetary Elliptical Motion and Three-Body Choreographies <i>M. Fernandez-Guasti</i>	195	
Chapter 10.	Caristi-Type Nonunique Fixed-Point Results and Fixed-Circle Problem on $b_v(s)$ -Metric Spaces Nihal Taş and Ozgur Ege	231	
Chapter 11.	Extended Interpolative Hardy–Rogers–Geraghty–Wardowski Contractions and an Application Samira Hadi Bonab, Vahid Parvaneh, Zohreh Bagheri, and Roghayeh Jalal Shahkoohi	261	
Chapter 12.	(η, ψ) -Rational <i>F</i> -Contractions and Weak-Wardowski Contractions in a Triple-Controlled Modular-Type Metric Space Hemant Kumar Nashine, Samira Hadi Bonab, and Vahid Parvaneh	279	

	Contents	XV
Chapter 13.	Sufficient Conditions for Mittag-Leffler Function Associated with Conic Regions Amit Soni and Deepak Bansal	309
Chapter 14.	Results on the Existence of Solutions for Some Controlled Optimization Problems Savin Treanță	327
Chapter 15.	The Theory of Nonabsolute Integrable Function Spaces over \mathbf{R}^{∞} and Its Various Applications Hemanta Kalita and Bipan Hazarika	343
Chapter 16.	Subclass of Analytic Functions Involving Mittag-Leffler Type Borel Distribution Series P. Thirupathi Reddy and B. Venkateswarlu	365
Chapter 17.	Hybrid Accelerated Conjugate Gradient Method for Solving Nonlinear System of Equations Abubakar Sani Halilu, Arunava Majumder, Mohammed Yusuf Waziri, and Idris Ahmed	385
Chapter 18.	A Closed Form of Integral Transforms in terms of Lauricella Function $F_A^{(n)}$ and Their Numerical Simulations Abdelmajid Belafhal, Halima Benzehoua, and Talha Usman	405

Index

441

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0001

Chapter 1

Open Problems in Number Theory

Pradip Debnath

Department of Applied Science and Humanities, Assam University, Silchar, Cachar 788011, Assam, India debnath.pradip@yahoo.com pradip.debnath@aus.ac.in

In spite of being one of the oldest branches of mathematics, number theory has an abundance of unsolved and open problems. It will require multiple volumes of a book to list out even a portion of those open problems. Our aim in this introductory chapter is to provide brief commentary about the most celebrated open problems and the progress toward their solutions. New research directions are also indicated accordingly.

1. Introduction

Number theory primarily deals with attributes of the integers and more specifically with the positive integers (i.e. natural numbers). This nomenclature is considered as a misnomer because by the word *number*, the early Greeks meant only the positive integers.

A number greater than 1 is prime if its only positive divisors are 1 and itself. It is also well known that every positive integer greater than 1 can be represented as a product of primes and the representation is unique apart from the order in which the prime factors occur. The set of positive integers can be partitioned into three classes:

- The unit 1
- The prime numbers $2, 3, 5, 7, 11, 13, 17, \ldots$
- The composite numbers $4, 6, 8, 9, 10, 12, 14, \ldots$

The primes have fascinated mathematicians since the time of Euclid who proved that there is an infinitude of them.

Theorem 1 (Euclid, 350 BC). There are infinitely many primes.

Proof. Suppose that there are finite number of primes and $p_1, p_2, p_3, \ldots, p_n$ is the finite list of primes. Consider the product $P = p_1 p_2 p_3 \ldots p_n$ and let K = P + 1. Since K > 1, it must have a prime divisor p. Hence, p must be one of $p_1, p_2, p_3, \ldots, p_n$. Now, since p divides K and the product P, we must have p divides 1, a contradiction. Therefore, there are infinitely many primes. \Box

Dirichlet's theorem further extends Euclid's result.

Theorem 2 (Dirichlet, 1837). Let $a, d \in \mathbb{Z}$ and gcd(a, d) = 1. Then there are infinitely many primes in the sequence $a, a + d, a + 2d, \ldots, a + nd \ldots$ for $n \in \mathbb{N}$.

Alternately, Dirichlet's theorem states that there are infinitely many primes congruent to a modulo d. Further, the numbers of the form a + nd generate an arithmetic progression and this theorem asserts that this sequence contains infinitely many primes.

Dirichlet's theorem also gives assurance to the existence of primes of particular form. For instance, it ensures that there are infinite number of primes ending with 777 such as 1777, 1000777,... because these numbers belong to the arithmetic progression a + nd, where gcd(777, 1000) = 1.

The most celebrated result concerning the distribution of primes is however the famously known prime number theorem, which was independently proved by Hadamard and Poussin.

Theorem 3 (Hadamard and Poussin, 1896). Let $\pi(x)$ be the number of primes less than or equal to x. Then $\frac{x}{\log x}$ is a good approximation to $\pi(x)$ in the sense that

$$\lim_{x \to \infty} \frac{\pi(x)}{\left[\frac{x}{\log x}\right]} = 1.$$

Using asymptotic notation, in 1792, Gauss stated the prime number theorem as

$$\pi(x) \approx \frac{x}{\log x}.$$

Later, Gauss refined his estimate to $\pi(x) \approx Li(x)$, where $Li(x) = \int_2^n \frac{dx}{\ln x}$ is the logarithmic integral.

In spite of being one of the ancient branches of mathematics, number theory has an abundance of unsolved and open problems. It will require multiple volumes of a book to list out even a portion of those open problems. Our aim in this chapter is to provide a brief commentary about the most celebrated open problems and the progress toward their solutions.

From the following section onwards, we present our discussion on some famous unsolved and open problems in number theory.

2. Twin Prime Conjecture

A pair of primes are said to be twin primes if their difference is 2, i.e. the pair is of the form (p, p + 2). The first few twin prime pairs are $(3, 5), (5, 7), (11, 13), (17, 19), (29, 31), (41, 43), (59, 61), (71, 73), (101, 103), (107, 109), (137, 139), \ldots$

The question of whether or not there are infinitely many twin primes has remained unsolved till date. The **twin prime conjecture** states that there are infinitely many primes p such that p + 2 is also a prime.

In 1849, a more general conjecture was made by de Polignac that for every natural number k, there are infinitely many primes p such that p+2k is a prime. Obviously, when k = 1, de Polignac's conjecture reduces to the twin prime conjecture.

The following important results related to twin primes are consequences of Wilson's theorem $[(p-1)! \equiv -1 \mod p, p \text{ is a prime}]$:

- Clements, 1949: m, m + 2 are twin primes if and only if $4[(m-1)!+1] + m \equiv 0 \mod m(m+2)$.
- Sergusov, 1971: m, m + 2 are twin primes if and only if $\phi(k)\sigma(k) = (k-3)(k+1)$, where k = m(m+2) and ϕ is Euler's totient function (i.e. $\phi(k)$ counts the number of positive integers up to k that are relatively prime to k) and $\sigma(k)$ is the sum of positive divisors of k (including 1 and k).

It is a well known result due to Euler that the infinite series $\sum \frac{1}{p}$ over all primes diverges.

However, in 1919, Norwegian mathematician Viggo Brun proved a remarkable theorem which states that the sum of reciprocals of all the twin primes converges to a finite value known as Brun's constant which is approximately equal to 1.90216054. Formally, this theorem may be stated as follows.

Theorem 4 (Brun's theorem, 1919). If \mathbb{P} denotes the set of twin primes, then the summation

$$\sum_{p,p+2\in\mathbb{P}} \left(\frac{1}{p} + \frac{1}{p+2}\right) = \frac{1}{3} + \frac{1}{5} + \frac{1}{5} + \frac{1}{7} + \frac{1}{11} + \frac{1}{13} + \frac{1}{17} + \frac{1}{19} \cdots$$

either has finitely or infinitely many terms but converges to Brun's constant.

Chinese mathematician Jing-Run Chen, in 1966 (published in 1978) proved the following which is quite close to proving that there are infinite number of twin primes:

Theorem 5 (Chen, 1978). There are infinitely many primes p such that (p+2) is either prime or the product of two primes.

In 2013, Chinese mathematician Yitan Zhang came out with a breakthrough result which is also known as the weak conjecture of twin primes. His theorem roughly states that there are infinitely many pairs of primes that differ by a positive integer which is less than 70 million.

Theorem 6 (Zhang, 2013). There exists an even integer $M \ge 2$ with the property that there are infinitely many primes of the form (p, p + M). In fact, there exists an M with $M \le 7 \times 10^7$.

It was only a matter of time that the upper bound of 70 million in Zhang's theorem got reduced significantly. By July 2013, Australian mathematician Terence Tao, via his Polymath8 project, had a remarkable contribution in reducing this upper bound from 70 million to merely 4680.

In November 2013, British mathematician James Maynard presented an independent proof that pushed down the gap in Zhang's theorem to 600. In 2014, Indo-Canadian mathematician M. R. Pedaprolu Murty at Queen's University also contributed to major developments in this direction. As of April 2014, the gap has been reduced to 246 using the methods of Maynard and Tao. The reduction of the gap up to 2 is still awaited.

3. The Goldbach Conjecture

In 1742, German mathematician Christian Goldbach came up with the conjecture that every positive even integer greater than 2 is the sum of two prime numbers. Despite considerable efforts, this conjecture remains unproven till date.

With the help of computer programming, as of 2013, the conjecture has been verified completely to hold true for all integers up to 4×10^{18} .

The weak Goldbach conjecture states that every odd positive integer greater than 5 can be written as the sum of three primes. Vinogradov's three primes theorem (1930) is worth mentioning here which states that any sufficiently large odd integer can be represented as the sum of three primes. After enormous efforts, in 2002, the weak Goldbach conjecture was verified for odd numbers greater than 2×10^{1346} . In 2013, the Peruvian mathematician Harald Helfgott came up with his magnificent proof that the weak Goldbach conjecture is true.

4. The Riemann Hypothesis

By $\Re(s)$ and $\Im(s)$, we denote the real and imaginary parts of the complex variable *s*, respectively. In 1859, Riemann published a groundbreaking article in which he obtained an analytic formula for the number of primes up to a given limit by introducing the Riemann zeta function as a function of the complex variable *s*, defined in the complex half-plane $\Re(s) > 1$ in terms of the absolutely convergent infinite series

$$\zeta(s) = \sum_{n=1}^{\infty} \frac{1}{n^s} = \frac{1}{1^s} + \frac{1}{2^s} + \frac{1}{3^s} + \cdots$$

P. Debnath

Riemann showed that $\zeta(s)$ can be extended to \mathbb{C} as a meromorphic function having a simple pole at s = 1 with residue 1 and can be derived from the functional equation

$$\pi^{-s/2}\Gamma\left(\frac{s}{2}\right)\zeta(s) = \pi^{-(1-s)/2}\Gamma\left(\frac{1-s}{2}\right)\zeta(1-s).$$

The zeros of the zeta function, i.e. the solutions $\omega \in \mathbb{C}$ of the equation $\zeta(\omega) = 0$, play a pivotal role in the representation of this formula. In this article, Riemann established the complex valued function ξ defined in the following manner:

$$\xi(\tau) = \frac{1}{2}s(s-1)\pi^{-s/2}\Gamma\left(\frac{s}{2}\right)\zeta(s),$$

with $s = \frac{1}{2} + i\tau$. He further established that $\xi(\tau)$ is an even entire function of τ and the imaginary parts of those zeros lie between -i/2and i/2. He also speculated that in the range between 0 and T, the function $\xi(\tau)$ has approximately $(\frac{T}{2\pi})\log(\frac{T}{2\pi}) - \frac{T}{2\pi}$ zeros. Riemann stated that within that range, all zeros were likely to be real.

It has been proven that the Riemann zeta function has zeros at the negative even integers, i.e. $\zeta(s) = 0$ for $s = -2, -4, -6, \ldots$ and these zeros are known as trivial zeros of the zeta function. The other roots of the zeta function are the complex numbers $\frac{1}{2} + i\alpha$ where α is a root of $\xi(\tau)$. The Riemann hypothesis is all about the locations of the nontrivial zeros.

Riemann hypothesis. The nontrivial zeros of the Riemann zeta function $\zeta(s)$ have real part equal to $\frac{1}{2}$.

The line $\frac{1}{2} + it, t \in \mathbb{R}$ is called the critical line and the domain $S = \{\rho + it \in \mathbb{C} : \rho \in [0, 1], t \in \mathbb{R}\}$ is called the critical strip. It is a proven fact that there are no nontrivial zeros outside the critical strip.

In the year 2000, Clay Mathematics Institute selected seven wellknown problems known as The Millennium Prize Problems and pledged US dollar 1 million prize for the correct solution of any of them. Riemann hypothesis is one of those seven problems.

Zeros of the Riemann zeta function are closely linked with the distribution of primes. However, the uses and applications of the Riemann hypothesis is vastly widespread in different branches of science. There has been a colossal research toward proof of the hypothesis. We list here some of the major developments in this direction:

- Hardy, 1914: There are infinitely many zeros of the Riemann zeta function on the critical line.
- Selberg, 1942: If $N(\zeta)$ denotes the number of zeros of $\zeta(s)$ on the critical line $\frac{1}{2} + i\tau$ such that $\tau \in (0, T)$, then $N(\zeta) > C\frac{T}{2\pi} \log(\frac{T}{2\pi})$ for some constant $C \in (0, 1)$.
- $\zeta(s)$ has no nontrivial zeros outside the critical strip.
- Levinson, 1974: On the critical line, $\zeta(s)$ has at least one-third of its total nontrivial zeros. The bound was further improved to two-fifth by Conrey in 1989.
- Van de Lune, Riele and Winter in 1986 computed the first 1,500,000,001 nontrivial zeros of $\zeta(s)$ and speculated that they all were simple zeros.
- Ramanujan in 1914 gave an astonishing formula (without proof) for $\zeta(2n+1)$ which states that if $\alpha, \beta > 0, \alpha\beta = \pi^2$, then

$$\begin{aligned} \alpha^{-n} \left(\frac{1}{2} \zeta(2n+1) + \sum_{k=1}^{\infty} \frac{k^{-2n-1}}{e^{2\alpha k} - 1} \right) \\ &= \beta^{-n} \left(\frac{1}{2} \zeta(2n+1) + \sum_{k=1}^{\infty} \frac{k^{-2n-1}}{e^{2\beta k} - 1} \right) \\ &- 2^{2n} \sum_{k=0}^{n+1} (-1)^k \frac{B_{2k} B_{2n+2-2k}}{(2k)! (2n+2-2k)!} \alpha^{n+1-k} \beta^k, \end{aligned}$$

where B_{2k} denotes the 2kth Bernoulli number. Berndt gave a complete proof of this formula in 1977.

• Finding the exact order of the error term in the prime number theorem is one of the most famous and important problems in number theory. In 1901, Koch showed that if the Riemann hypothesis were true, then

$$\pi(x) = Li(x) + O(\sqrt{x}\ln x),$$

where O(x) is the asymptotic notation Big-O.

• There have been numerous claims for the proof of Riemann hypothesis, but none of them happened to be correct. In 2018,

British mathematician Michael Atiyah at the Heidelberg Laureate Forum gave a lecture in which he claimed to have proved the hypothesis. But unfortunately, his proof also turned out to be another failed attempt.

5. The Mersenne Primes

Named after the 17th century French mathematician Martin Mersenne, a Mersenne prime is a prime number which is one less than a power of two. Hence, these are prime numbers of the form $M_n = 2^n - 1$ for some positive integer n. When the primality condition is dropped, a number of this form is called a Mersenne number.

In his book Cogitata Physica-Mathematica (1644), Mersenne (incorrectly) conjectured that M_p is prime for p = 2, 3, 5, 7, 13, 17, 19, 31, 67, 127, 257 and composite for all other primes less than 257. More than 300 years later, by 1947, Mersenne's list was corrected to

$$p = 2, 3, 5, 7, 13, 17, 19, 31, 61, 89, 107, 127.$$

Thus, Mersenne erroneously conjectured that M_{67} and M_{257} are primes and missed M_{61} , M_{89} and M_{107} from his predicted list. Since 1997, all the newly discovered Mersenne primes have been obtained by the *Great Internet Mersenne Prime Search*, which is a distributed computing project. As of 2020, 51 Mersenne primes have been discovered and the 51st is $2^{82,589,933} - 1$.

There was another conjecture that if M_n is a prime, then so is M_{M_n} . If this conjecture were true, it would mean that there are infinitely many Mersenne primes. However, in 1953, a computer search proved that $M_{M_{13}}$ is composite.

Hence, there remain several open problems about Mersenne primes:

- Is the number of Mersenne primes infinite?
- Is it that every Mersenne number happens to be square free?
- Are there infinitely many composite Mersenne numbers?

A positive integer n is said to be perfect if n is equal to the sum of all its positive divisors, excluding n itself.

The numbers $6, 28, 496, 8128, \ldots$ are perfect. However, all the perfect numbers discovered till date are even. A famous open question is as follows:

• Are there odd perfect numbers?

6. The Fermat Numbers

Another class of numbers that provide us with a rich source of open problems and conjectures is the Fermat numbers.

A Fermat number is a number of the form $F_n = 2^{2^n} + 1$ for $n \ge 0$. If F_n is prime, it is called a Fermat prime.

Fermat observed that $F_0 = 3, F_1 = 5, F_2 = 17, F_3 = 257, F_4 = 65537$ are all primes and believed that F_n is a prime for each $n \ge 0$. However, this belief was resolved in the negative by Euler in 1732 as he observed that $F_5 = 429, 496, 7297$ is divisible by 641.

Similar to Mersenne numbers, we have the following open questions:

- Is the number of Fermat primes infinite?
- Is it true that every Fermat number is square free?
- Is the number of composite Fermat numbers infinite?

7. Some Other Famous Open Problems

It is quite clear from the present discussion that one could make a conjecture that there are infinitely many open problems in number theory. In this section, we list out some other famous open problems:

• The Fibonacci numbers may be defined by the following recurrence relation:

$$F_0 = 1, F_1 = 1$$
 and $F_n = F_{n-1} + F_{n+2}$

for n > 1.

Thus, Fibonacci numbers form a sequence in which each number is the sum of the two preceding ones. A prime number is said to be Fibonacci prime if it appears in the Fibonacci sequence. An important open question is as follows: are there infinitely many Fibonacci primes?

- If an odd prime number p does not divide the numerator of the Bernoulli number B_k , for all even $k \leq p-3$, then it is called regular. An irregular prime is an odd prime that is not regular. The odd numbers $3, 5, 7, \ldots, 31$ are all regular. 37 is the first irregular prime. The infinitude of irregular primes is a well-known fact. However, the following question is open: is the number of regular primes infinite?
- The Euler constant γ is defined as

$$\gamma = \lim_{n \to \infty} \left(\sum_{k=1}^{n} \frac{1}{k} - \ln n \right).$$

It is an open problem whether the Euler constant γ is irrational or rational. The question about γ being transcendental is also an open problem.

• The set of Gaussian integers $\mathbb{Z}[i]$ is a unique factorization domain (UFD). However, there exist $m \in \mathbb{Z}$ such that $\mathbb{Z}[m]$ is not a UFD. Exactly nine negative integers have been discovered, known as Heegner numbers, for which $\mathbb{Z}[m]$ is a UFD. These numbers are

$$-1, -2, -3, -7, -11, -19, -43, -67, -163.$$

It is still not established whether there are infinitely many positive integers m so that $\mathbb{Z}[m]$ is a UFD.

- For the Riemann zeta function $\zeta(s)$, the value of $\zeta(3)$ is well known as the Apéry constant and it is an irrational number. The question about Apéry constant being transcendental is an open problem. Further, the values of $\zeta(2k + 1)$, for all $k \in \mathbb{N}$, being irrational or transcendental also remains unsolved.
- Lothar Collatz in 1937 introduced an idea which is popularly known as Collatz conjecture or 3n + 1 problem. Paul Erdös, who offered USD 500 for its correct solution, stated about this conjecture as follows: "Mathematics may not be ready for such problems."

To state the problem formally, choose a positive integer a_0 and construct the sequence $\{a_n\}$ as

$$a_{n+1} = \begin{cases} \frac{a_n}{2}, & \text{if } a_n \text{ is even,} \\ 3a_n + 1, & \text{if } a_n \text{ is odd.} \end{cases}$$

For example, if we choose $a_0 = 1$, we obtain the sequence $1, 4, 2, 1, \ldots$, which repeats in the triplet 1, 4, 2. The Collatz conjecture states that if we initiate with any positive number a_0 , the sequence will eventually take the value 1. The conjecture remains unproved till date.

• Lagrange's four-square theorem of 1770 states that every natural number can be expressed as the sum of four integer squares. In the same year, Waring raised the question: how about the minimum number of cubes to do the same? More generally, he wanted to obtain the minimum number of kth powers necessary to express all positive integers. This open problem is known as Waring's problem.

Acknowledgments

For this chapter, much information has been collected from Refs. $1{-}8$ and the references therein.

The author acknowledges the constructive comments and suggestions received from Prof. Bruce C. Berndt (University of Illinois, USA), Prof. Thomas C. Brown (Simon Fraser University, Canada) and Prof. Shigeru Kanemitsu (Kindai University, Japan) towards improving this manuscript.

References

- T. M. Apostol, Introduction to Analytic Number Theory. Springer-Verlag, New York (1976).
- B. C. Berndt, Modular transformations and generalizations of several formulae of Ramanujan, *Rocky Mountain J. Math.* 7, 147–189 (1977).
- D. M. Burton, *Elementary Number Theory*, 7th edn. McGraw Hill Education, New York, US (2017).
- 4. R. Crandall and C. Pomerance, *Prime Numbers: A Computational Perspective*, 2nd edn. Springer, New York, US (2005).
- H. M. Edwardsl, *Riemann's Zeta Function*. Academic Press, New York - London (1974).
- R. K. Guy, Unsolved Problems in Number Theory, 3rd edn. Springer, New York (2004).
- P. Ribenboim, Prime Numbers, Friends Who Give Problems: A Trialogue With Papa Paulo. World Scientific, Singapore (2017).
- D. Shanks, Solved and Unsolved Problems in Number Theory, 4th edn. American Mathematical Society, Providence, RI (2003).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0002

Chapter 2

Arithmetic and Congruence Properties of Partition Function

Nabanita Konwar

Department of Mathematics, Birjhora Mahavidyalaya, Bongaigaon 783380, Assam, India nabnitakonwar@qmail.com

The purpose of this chapter is to present the systematic development and evolution of the partition function. We particularly discuss the arithmetic and congruence properties. Various types of identities are discussed as well. Finally, we recapitulate different types of Ramanujan congruences and also put forward some new generalizations of Ramanujan congruences on p(n) and $p_j(n)$.

1. Introduction

In mathematics, number theory always occupied a unique position due to its historical importance. It is one of the oldest branches of mathematics. Starting from the Babylonians and then from Pythagoras to Ramanujan, the subject has undergone a huge evolution. The Euclidean Algorithm, developed by Euclid, is an efficient tool toward the application of the Division Algorithm.

In this chapter, we present the systematic development and evolution of the partition function, and discuss the arithmetic and congruence properties and also various types of identities. Finally, different types of Ramanujan congruences are recapitulated and some new generalizations of Ramanujan's congruences on p(n) and $p_j(n)$ are also recommended.

2. Partitions

Partition¹ of a positive integer n is a way by which the number can be expressed as a sum of one or more positive integers.

For example, 3 can be expressed as 3; 1 + 2; 1 + 1 + 1. Thus, the number of partitions of 3 is 3.

We can denote the number of the partition of n positive number as p(n).

In the following sections, we discuss various types of partitions.

2.1. Unrestricted partitions

When in a partition, there is no restriction of any kind on the number or size of the parts, such a partition is called unrestricted partitions.

For example, the positive integer 5 can be expressed as a sum in the following seven ways:

5; 1+4; 2+3; 1+2+2; 1+1+3; 1+1+1+2; and 1+1+1+1+1.

These are all the partitions of 5 and these partitions are unrestricted partitions of 5.

The number of unrestricted partitions of any given positive integer n is denoted by p(n), and for convenience, we consider p(0) = 1.

2.2. Restricted partitions

If some restrictions are imposed in a_i (like all a'_i s are positive), then such types of partitions are known as restricted partitions. Restrictions may be classified as follows:

(1) repetitions of the integers being allowed or not,

- (2) partitions into distinct parts,
- (3) partitions into odd parts,
- (4) partitions into even parts,
- (5) partitions into a specified number of parts,
- (6) partitions into a specified number of distinct parts.

Besides these, the restrictions can be placed on the size or on the number of parts or both.

Consider an example, where the number g(n, m, h, k) of partitions of n into exactly k summands, each $\leq m$, any h of the positive integers $\leq m$ being used as summands in any partition. Then we have

$$\sum_{n=1}^{\infty} g(n,m,h,k) = \binom{k-1}{h-1} \binom{m}{h}.$$

We denote p(n, k) as the number of partitions of n into exactly k summands, q(n) as the number of partitions of n into distinct parts and q(n, k) as the number of partitions of n into exactly k distinct parts.

2.3. Perfect partitions

A partition of n is said to be perfect when it contains just one partition of every number up to n. Hence, one perfect partition of 7 is 1+2+2+2 because every number up to 7 can be expressed uniquely as a sum by using the summands: 1, 2, 2, 2. Other such partitions of 7 are as follows:

1 + 1 + 1 + 1 + 1 + 1 + 1; 1 + 1 + 1 + 4; 1 + 2 + 4.

2.4. Plane and solid partitions

For a partition of n if the summands are arranged in the form of a matrix so that the elements in each row and in each column are in a descending order, then such a type of arrangement is called a plane partition of n.

Thus,

3	3	2	1
2	2	2	1
1	1		

is a plane partition of 18.

A plane partition is also known as a rowed partition. A k-rowed partition of n includes all partitions of n with k or fewer number of rows.

The number of k-rowed partitions of n is denoted by $t_k(n)$.

In a solid partition, the summands are arranged in a threedimensional space as they are arranged in a two-dimensional space in the case of a plane partition. The elements are arranged in descending order of magnitude in each of the three principal directions.

2.5. l-regular partitions²

Let l be a prime that satisfies $p(ln + b) \equiv 0 \pmod{l}$ for 0 < l < b. Then a partition of n is said to be l-regular if none of its parts are multiples of l. For example, 3-regular partitions of 7 are as follows:

 $7;5+2;5+1+1;4+2+1;4+1+1+1;2+2+2+1;2+2+\\1+1+1;2+1+1+1+1+1;1+1+1+1+1+1,$

l-regular partitions of n by $b_l(n)$.

2.6. $Overpartitions^3$

An overpartition of the positive integer n is an ordinary partition of n where the first occurrence of parts of each size may be overlined. For example, the overpartitions of the integer 3 are as follows:

$$3; \overline{3}; 2+1; \overline{2}+1; 2+\overline{1}; \overline{2}+\overline{1}; 1+1+1 \text{ and } \overline{1}1+1+1$$

The number of overpartitions of n is denoted by $\overline{p}(n)$, and $\overline{p_0}(n)$

is the number of overpartitions of n in which only odd parts are used. Hence, $\overline{x}(2) = 8$ and $\overline{x}(2) = 4$

Hence, $\overline{p}(3) = 8$ and $\overline{p_0}(3) = 4$.

3. Identities

In 1742, Euler developed an expression for the study of partitions which is eagerly known as Euler's identity:

$$\prod_{r=1}^{\infty} (1 - x^r) = 1 - x - x^2 + x^5 + x^7 - x^{12} - x^{15} + \dots + (-1)^k \left\{ x^{\frac{k(3k-1)}{2}} + x^{\frac{k(3k+1)}{2}} \right\} + \dots = \sum_{k=-\infty}^{\infty} (-1)^k x^{\frac{k(3k+1)}{2}}$$

In 1830, Legendre noted that Euler's identity implies that every number which is not pentagonal can be partitioned into an even number of distinct parts as often as it can be partitioned into an odd number of distinct integers, while the pentagonal number k(3k+1)/2 can be partitioned into an even number of distinct parts once oftener or once fewer times than into an odd number of distinct parts based on whether k is even or odd.

In 1843, Cauchy provided an expression as follows:

$$\prod_{r=0}^{n-1} (1+zx^r) = 1 + \frac{1-x^n}{1-x}z - \frac{(1-x^n)(x-x^n)}{(1-x)(1-x^2)}z^2 + \cdots + \frac{(1-x^n)(x-x^n)(x^2-x^n)\dots(x^{n-1}-x^n)}{(1-x)(1-x^2)(1-x^3)\dots(1-x^n)}z^n$$

and

$$\prod_{r=0}^{n-1} (1-zx^r)^{-1} = 1 + \frac{1-x^n}{1-x}z - \frac{(1-x^n)(1-x^{n+1})}{(1-x)(1-x^2)}z^2 + \frac{(1-x^n)(1-x^{n+1})(1-x^{n+2})}{(1-x)(1-x^2)(1-x^3)}z^3 + \cdots$$

This expression was known as Cauchy's identity and one can obtain the Euler's identity from this identity.

Simultaneously, in 1929, Jacobi put forward the applications of elliptic functions to the study of the theory of partitions:

$$\prod_{r=1}^{\infty} (1 - x^{2r})(1 + zx^{2r-1})(1 + z^{-1}x^{2r-1})$$

= 1 + x(z + z^{-1}) + x⁴(z² + z^{-2}) + x⁹(z³ + z^{-3}) + \cdots

This new expression was called as Jacobi's identity and it is verified that Euler's identity is a particular case of Jacobi's identity.

In 1966, Sudler⁴ provided two enumerative proofs of Jacobi's identity.

In the following section, we discuss some interesting identities known as Ramanujan's Identities.

3.1. Ramanujan's identities

Ramanujan established two remarkable identities without proof:

$$p(4) + p(9)x + p(l4)x^{2} + \dots = \frac{5\{f(x)\}^{6}}{\{f(x)^{5}\}^{5}},$$

$$p(5) + p(12)x + p(l9)x^{2} + \dots = \frac{7\{f(x)\}^{4}}{\{f(x)^{7}\}^{3}} + \frac{49x\{f(x)\}^{8}}{\{f(x)^{7}\}^{7}}.$$

Further, these two identities imply the congruences of Ramanujan for the moduli 5, 5^2 , 7, 7^2 . In 1950, Kruyswijk⁵ proved the proof of these two Ramanujan identities using power series.

3.2. Rogers-Ramanujan identities

In 1913, Ramanujan rediscovered Rogers's identities:

$$\prod_{r=0}^{\infty} \{(1-x^{5r+1})(1-x^{5r+4})\}^{-1} = 1 + \prod_{r=1}^{\infty} \frac{x^{r^2}}{(1-x)(1-x^2)\dots(1-x^r)}$$

and

$$\prod_{r=0}^{\infty} \{(1-x^{5r+2})(1-x^{5r+3})\}^{-1} = 1 + \prod_{r=1}^{\infty} \frac{x^{r(r+1)}}{(1-x)(1-x^2)\dots(1-x^r)}$$

These two identities are first found by Rogers^6 in 1894, and later on, these identities are known as Rogers–Ramanujan identities. Many mathematicians like Dobbie⁷ in 1962 and Alder⁸ in 1954 generalized these identities. Simultaneously, in 1961, Gordon⁹ and in 1966 Andrews^{10, 11} generalized these identities in different directions.

4. Generating Function

The function

$$f(x) = \prod_{r=1}^{\infty} (1 - x^r)^{-1}$$
(1)

is called the generating function for p(n).

The generating function for q(n) is

$$g(x) = \prod_{r=1}^{\infty} (1 + x^r).$$

When a set A consists of all the positive integers, the generating function for p(n, k) is

$$x^k \prod_{r=1}^k (1-x^r)^{-1}.$$

Similarly, the generating function for q(n, k) is

$$x^{\frac{k(k+1)}{2}}\prod_{r=1}^{k}(1-x^{r})^{-1}.$$

4.1. Generating function for unrestricted partitions

Euler established the generating function for the number of unrestricted partitions of a number n, denoted by p(n), as

$$P(q) = \sum_{n \ge 0} p(n)q^n = \frac{1}{(q;q)_{\infty}},$$

where $(q;q)_{\infty} = \prod_{n=1}^{\infty} (1-q^n).$

4.2. Generating function for overpartitions

The generating function for the overpartition $\overline{p}(n)$ is defined as

$$\sum_{n=0}^{\infty} \overline{p}(n)q^n = \frac{(-q;q)_{\infty}}{(q;q)_{\infty}},$$

where $(a;q)_{\infty} = \lim_{n \to \infty} (1-a)(1-aq)(1-aq^2)\dots(1-aq^{n-1}).$

4.3. Generalizations of p(n)

In 1951, Gupta¹² generalized the generating function

$$J(x) = \prod_{r=1}^{\infty} (1 - x^r)^{-r^{j-1}},$$

which is known as the generalization of the generating function p(n).
Another generalization of p(n) is

$$J(x,m) = \prod_{r=1}^{m} (1-x^r)^{-r^{j-1}}.$$

In 1960, Kolberg¹³ established another generalization:

$$P(x,j) = \{f(x)\}^{-j} = 1 + \prod_{n=1}^{\infty} P_j(n)x^n.$$

For positive values of j, $p_j(n)$ is a polynomial in j of degree n. Kolberg¹⁴ established another generalization function for p(n):

$$P(x^2,5)P(-x,-2) = \sum_{m=-\infty}^{\infty} (3m+1)x^{m(3m+2)}$$

and

$$P(x,5)P(x^2,-2) = \sum_{m=-\infty}^{\infty} (6m+1)x^{\frac{m(3m+2)}{2}}$$

In 1953, Carlitz¹⁵ provided one generalization form for p(n) which is equivalent to Newman's formula:

$$P(x,6)P(x^5,-1) = \sum_{m=0}^{\infty} P_5(5m)x^m.$$

5. Congruences

The theory of congruences was first introduced by the German mathematician Carl Friedrich Gauss in his *Disquisitiones Arithmeticae* in 1801. Gauss also introduced the notation of congruence which makes the theory of congruence a powerful technique. According to Gauss, "If a number n measures the difference between two numbers a and b, then a and b are said to be congruent with respect to n; if not incongruent."

Definition 1. Let n be a fixed positive integer. Two integers a and b are said to be congruent modulo n,

$$a \equiv b \pmod{n}$$

if n divides the difference a - b, i.e. a - b = kn for some integer k.

For example, consider n = 5. Then $29 \equiv 4 \pmod{5}$, we say that 29 is congruent to 4 modulo 5.

5.1. Ramanujan's Conjecture

In 1919, Ramanujan established the following congruences:

$$p(5m + 4) = 0 \pmod{5}, \ p(25m + 24) = 0 \pmod{25},$$

$$p(7m + 5) = 0 \pmod{7}, \ p(49m + 47) = 0 \pmod{49},$$

$$p(llm + 6) = 0 \pmod{11}$$

for any nonnegative m.

These results are known as the Ramanujan conjecture which is stated as

If
$$p = 5, 7$$
 or 11 and $(24n - 1) \equiv 0 \pmod{p^a}, a \ge 1$, then
 $p(n) \equiv 0 \pmod{p^a}.$

But after some calculations, Chowla¹⁶ found that the Ramanujan conjecture failed for some n = 243. According to the Ramanujan conjecture for the value of n = 243,

$$(24n - 1) = 5831 \equiv 0 \pmod{7^3}$$

while

$$p(243) \equiv 0 \pmod{7^2}$$
$$\not\equiv 0 \pmod{7^3}.$$

Next, it was seen that the conjecture also failed for n = 586.

Finally, Lehmer¹⁷ established a new form of p(n) for the large values of n using the Hardy–Ramanujan and Rademacher series. He provided the following:

$$p(599) \equiv 0 \pmod{54}, \ p(721) \equiv 0 \pmod{113},$$

 $p(1224) \equiv 0 \pmod{54}, \ p(2052) \equiv 0 \pmod{113},$
 $p(2474) \equiv 0 \pmod{55}, \ p(l4031) \equiv 0 \pmod{114}.$

In 1948, Lahiri¹⁸ introduced some new types of congruences:

$$p(49m + r) \equiv 0 \pmod{49}$$
 for $r = 19, 33, 40,$
 $p(125m + r) \equiv 0 \pmod{125}$ for $r = 74, 124.$

In 1952, Rushforth¹⁹ proved another form of congruence:

$$p(121m + 116) \equiv 0 \pmod{121},$$

 $p(49m + r) \equiv 0 \pmod{49}$ for $r = 19, 33, 40, 47$

In 1960, Newman²⁰ provided the following for (m, 30) = 1:

$$p\left(\frac{167m^2+1}{24}\right) \equiv 0 \pmod{5}.$$

All the above results were the generalized form of the Ramanujan congruences.

Along with the generalization, Watson^{21,22} had proved Ramanujan's conjecture completely for powers of 5 and also provided the modification of the conjecture for powers of 7:

If

$$(24n-1) \equiv 0 \pmod{7^b},$$

then

$$p(n) \equiv 0 \pmod{7^d},$$

where $d = \left[\frac{(b+2)}{2}\right]$. Finally, in 1943 and 1950, Atkin²³ proved the Ramanujan conjecture for powers of 11, 11^2 and 11^3 . Accordingly, Atkin and O'Brien²⁴ studied the Ramanujan conjecture for powers of 13.

The Parity of p(n)5.2.

In 1920, MacMahon²⁵ introduced a table for p(n) such that

$$p(n) \equiv \sum_{t} p(t) \pmod{2},$$

where t is the positive integral value of $8t = 2n - j(j+1), j \ge 0$.

Garvan,²⁶ Kolberg,²⁷ Hirschhorn,^{28,29} Stanton²⁶ and Subbarao³⁰ have generalized it for every arithmetic progression with modulus $t \in \{1, 2, 3, 4, 5, 6, 8, 10, 12, 16, 20, 40\}$.

5.3. Congruences for $p_j(n)$

In 1956, Newman³¹ gave extensive tables of values of $p_j(n)$ for $j \le 16$. For j = 2, 4, 6, 8, 10, 4, 16 and a prime p > 3,

$$j(p+l) \equiv 0 \pmod{24}.$$

 $Newman^{32}$ established that

$$p_j(np+k) = (-p)^{\frac{(j-2)}{2}} p_j\left(\frac{n}{p}\right),$$

where $k = \frac{j(p^2-1)}{24}$ and $p_j(n) = 0$ if n is not a nonnegative integer. Newman also provided that

$$p_6(3n+2) = 9p_6\left(\frac{n}{3}\right).$$

After Newman, Ramanathan³³ provided that

if

$$24n + j \equiv 0 \pmod{5^a}, j \equiv 16, 21, 26 \pmod{30},$$

then

$$p_j(n) \equiv 0 \pmod{5^b},$$

where $b = [\frac{(a+1)}{2}].$

6. Generalization of the Congruences Modulo for All $p_j(n)$

In 2004, Hammond and Lewis³⁴ introduced that

$$p_{-2}(5n+l) \equiv 0 \pmod{5},$$

where $l \in 2, 3, 4$.

In 2014, Chen *et al.*³⁵ established that

$$p_{-2}(25n+23) \equiv 0 \pmod{25}.$$

Recently, in 2018, Tang³⁶ proved some congruences modulo powers of 5 for $p_j(n)$ with $j \in \{2, 6, 7\}$ such that

$$p_{-2}\left(5^{2d-1}n + \frac{7 \times 5^{2d-1} + 1}{12}\right) \equiv p_{-6}\left(5^{2d}n + \frac{3 \times 5^{2d} + 1}{4}\right)$$
$$\equiv p_{-7}\left(5^{2d-1}n + \frac{13 \times 5^{2d-1} + 7}{24}\right)$$
$$\equiv 0 \pmod{5^d}.$$

6.1. New infinite families of congruences modulo 11 for $p_j(n)$

Some new infinite families of congruences modulo 11 for $p_j(n)$ have been established by using q-identities for any positive integer λ .³⁷ These are as follows:

If t = 3, 6, 8, 9, 10, then

$$p_{11\lambda+1}(11n+t) \equiv 0 \pmod{11}.$$

Next, if t = 2, 4, 5, 7, 8, 9, then

$$p_{11\lambda+3}(11n+t) \equiv 0 \pmod{11}.$$

One specific case is that

$$p_{11\lambda+6}(11n+8) \equiv 0 \pmod{11}.$$

Next, if $1 \le t \le 10$, then

$$p_{121\lambda+1}(121n+11t+5) \equiv 0 \pmod{11}$$

and if $1 \le t \le 10$, then

$$p_{121\lambda+2}(121n+11t+10) \equiv 0 \pmod{11}.$$

6.2. Ramanujan-type congruences modulo 8 for the overpartition function $\overline{p_0}(n)$

Recently, many congruences for the number of overpartitions into odd parts have been discovered. The first Ramanujan-type congruences modulo power of 2 for $\overline{p_0}(n)$ was established in 2006 by Hirschhorn and Sellers.³ Simultaneously, they also established a number of arithmetic results for Ramanujan-like congruences for the overpartition function $\overline{p_0}(n)$.

 $\overline{p_0}(n) = \begin{cases} 2(\mod 4) \text{ if } n \text{ is a square or } n \text{ is twice a square,} \\ 0(\mod 4) \text{ otherwise.} \end{cases}$

The prime factorization of n > 0 is given by

$$n = 2^{\alpha} \prod p_i^{\alpha_i} q_i^{\beta_i} r_i^{\gamma_i} s_i^{\delta_i},$$

where

$$p_i \equiv 1 \pmod{8}, q_i \equiv 3 \pmod{8}, r_i \equiv 5 \pmod{8}, s_i \equiv 7 \pmod{8}.$$

Hirschhorn and Sellers established some more relations on congruence modulo 8 as follows:

(1) $\overline{p_0}(n) \equiv 0 \pmod{8}$ if and only if one of the following holds:

- at least one δ_i is odd,
- all δ_i are even and at least one γ_i is odd,
- all δ_i are even, all γ_i are even, at least one β_i is odd, and $\prod (\alpha_i + 1)(\beta_i + 1) \equiv 0 \pmod{4}$,
- all δ_i are even, all γ_i are even, all β_i are even, and $\prod (\alpha_i + 1) \equiv 0 \pmod{4}$.

(2) $\overline{p_0}(n) \equiv 4 \pmod{8}$ if and only if one of the following holds:

- all δ_i are even, all γ_i are even, any β_i is odd, and $\prod (\alpha_i + 1)$ $(\beta_i + 1) \equiv 2 \pmod{4}$,
- all δ_i are even, all γ_i are even, all β_i are even, and $\prod (\alpha_i + 1) \equiv 2 \pmod{4}$.

(3) $\overline{p_0}(n) \equiv 2 \pmod{8}$ if and only if one of the following holds:

• *n* is an odd square or twice an odd square and $\prod (\alpha_i + \beta_i) \equiv 0 \pmod{4}$,

N. Konwar

- *n* is an even square or twice an even square and $\prod (\alpha_i + \beta_i) \equiv$ $2 \pmod{4}$.
- (4) $\overline{p_0}(n) \equiv 6 \pmod{8}$ if and only if one of the following holds:
 - *n* is an odd square or twice an odd square and $\prod (\alpha_i + \beta_i) \equiv$ $2 \pmod{4}$,
 - *n* is an even square or twice an even square and $\prod (\alpha_i + \beta_i) \equiv$ $0 \pmod{4}$.

In 2021, Mircea Merca³⁸ established Ramanujan-type congruences modulo 8 for the overpartition function $\overline{p_0}(n)$ considering the divisor function $T_{(odd)}(n)$ that counts the odd positive divisors of n as follows:

$$\overline{p_0}(2^{\alpha}(8n+l)) \equiv r \pmod{16},$$

where $\alpha > 0$ and $l \in \{1, 3, 5, 7\}$.

The partition of the set \mathbb{N} is defined as

$$A_l = \bigcup_{\alpha=0}^{\infty} \{2^{\infty}(8n+l) | n \in \mathbb{N}_{\not\vdash}\}.$$

Theorem 1. For $n, \alpha \geq 0$:

- (1) If $l \in \{5,7\}$, then $\overline{p_0}(2^{\infty}(8n+l)) \equiv 0 \pmod{8}$. (2) If $l \in \{1, 3\}$ and 8n + l is not a square, then

$$\overline{p_0}(2^{\infty}(8n+l)) \equiv \begin{cases} 4 \pmod{8}, & if \ \frac{T_{odd}(8n+l)}{2} \ is \ odd, \\ 0 \pmod{8}, & if \ \frac{T_{odd}(8n+l)}{2} \ is \ even \end{cases}$$

(3) If $l \in \{1,3\}$ and 8n+l is a square, then 8n+l is one of the form $(8k \pm l)^2$ or $(8k \pm 3)^2$. For $l \in 1, 3$ and $\alpha \in 0, 1$:

$$\overline{p_0}(2^{\infty}(8n\pm l)^2) \equiv \begin{cases} 6 \pmod{8}, & if \ \frac{T_{odd}((8n\pm l)^2) - 1}{2} \ is \ odd, \\\\ 2 \pmod{8}, & if \ \frac{T_{odd}((8n\pm l)^2) - 1}{2} \ is \ even. \end{cases}$$

For $l \in 1, 3$ and $\alpha > 1$:

$$\overline{p_0}(2^{\infty}(8n\pm l)^2) \equiv \begin{cases} 2 \pmod{8}, & \text{if } \frac{T_{odd}((8n\pm l)^2)-1}{2} \text{ is odd,} \\ \\ 6 \pmod{8}, & \text{if } \frac{T_{odd}((8n\pm l)^2)-1}{2} \text{ is even} \end{cases}$$

6.3. Some new identities for p(n) modulo 11, 17, 19 and 23

By using the theory of modular functions, Paule and Radu³⁹ obtained a witness identity which is different from Atkin and Lehner in 2017. After that, in 2019, the witness identity in modulo 11 was generalized for the Riemann surface.⁴⁰ In 2021, Goswami *et al.*⁴¹ generalized the Ramanujan congruence $p(11n + 6) \equiv 0 \pmod{11}$ and also established some new identities for the generating functions for p(17n + 5), p(19n + 7) and p(23n + 1). Some more works regarding Ramanujan congruence on the Riemann surface can also be found in Ref. 42. In future research, work on this area will be gradually developed and it will help mathematicians to establish more new concepts.

7. Conclusion

In this chapter, we discussed various types of partitions and their arithmetic and congruence properties. We explained congruences and focused on the Ramanujan congruences. Finally, we discussed some generalizations of the Ramanujan congruences. This chapter attempts to relate between the partition function and the Ramanujan congruences along with its generalized forms.

Acknowledgments

The author is thankful to the reviewers for their constructive comments, which have immensely contributed to the improvement of the manuscript.

References

- B. C. Berndt, Number Theory in the Spirit of Ramanujan. Vol. 34, Student Mathematical Library. American Mathematical Society, Providence, RI (2006). https://doi.org/10.1090/stml/034.
- J. Lovejoy and D. Penniston, 3-regular Partitions and a Modular K3 Surface. Vol. 291, Contemp. Math. American Mathematical Society, Providence, RI (2001). https://doi.org/10.1090/conm/291/04901.
- M. D. Hirschhorn and J. A. Sellers, Arithmetic properties of overpartitions into odd parts, Ann. Comb. 10(3), 353–367 (2006).
- C. Sudler, Two enumerative proofs of an identity of Jacobi, Proc. Edinburgh. 15(1), 67–71.
- 5. D. Kruyswijk, On some well-known properties of the partition function p(n) and Euler's infinite product, *Nieuw Arch. Wisk.* **23**, 97–107 (1950).
- J. Rogers, On the expansion of some infinite products, Proc. London Math. Soc. 24, 337–352 (1893).
- M. Dobbie, A simple proof of the Rogers–Ramanujan identities, *Quart. J. Math. Oxford Ser.* (2) 13, 31–34 (1962).
- H. L. Alder, Generalizations of the Rogers–Ramanujan identities, *Pacific. Math.* 4, 161–168 (1954).
- B. Gordon, A combinatorial generalization of the Rogers-Ramanujan identities, Am. J. Math. 83(2), 393–399 (1961).
- G. E. Andrews, An analytic proof of the Rogers-Ramanujan-Gordon identities, Am. J. Math. 88(4), 844–846 (1966).
- G. E. Andrews, Partition theorems related to the Rogers-Ramanujan identities, J. Comb. Theory 2(4), 422–430 (1967).
- H. Gupta, A generalization of the partition function, Proc. Nat. Inst. Sci. India 17, 231–238 (1951).
- O. Kolberg, Congruences Involving the Partition Function for the Moduli 17, 19 and 23. John Griegs Boktrykkeri, Norwegian Publisher (1960).
- O. Kolberg, Congruences involving the partition function for the moduli 17, 19 and 23, Bergen Arb. Mat. Nat. 15, 34–45 (1959).
- L. Carlitz, Note on some partition identities, Proc. Am. Math. Soc. 4(4), 530-534 (1953).
- S. Chowla, Congruence properties of partitions, J. London Math. Soc. 1(4), 247–247 (1934).
- D. H. Lehmer, On the Hardy-Ramanujan series for the partition function, J. London Math. Soc. 1(3), 171–176 (1937).
- D. B. Lahjri, Some non-Ramanujan congruence properties of the partition function, Proc. Nat. Inst. Sci. India 14, 337–338 (1948).

- 19. J. M. Rushforth, Congruence properties of the partition function and associated functions, *Proc. Cambridge Phil. Soc.* 48, 402–413 (1952).
- 20. M. Newman, Periodicity modulo m and divisibility properties of the partition function, *Trans. Am. Math. Soc.* **97**(2), 225–236 (1960).
- G. N. Watson, A new proof of the Rogers-Ramanujan identities, J. London Math. Soc. 1(1), 4–9 (1929).
- G. N. Watson, Ramanujans Vermutung über Zerfällungszahlen. Walter de Gruyter, Berlin/New York (1938).
- A. Atkin and S. Hussain, Some properties of partitions-II, Trans. Am. Math. Soc. 89(1), 184–200 (1958).
- 24. A. O. L. Atkin and J. N. O'brien, Some properties of p(n) and c(n) modulo powers of 13, Trans. Am. Math. Soc. **126**(3), 442–459 (1967).
- P. A. MacMahon, Note on the Parity of the Number Which Enumerates the Partitions of a Number. Vol. 20, Cambridge Phil. Soc., Cambridge (1921).
- F. Garvan and D. Stanton, Sieved partition functions and q-binomial coefficients, Math. Comput. 55(191), 299–311 (1990).
- O. Kolberg, Note on the parity of the partition functions, Math. Scand. 7, 377–378 (1959).
- 28. M. Hirschhorn, On the residue mod 2 and mod 4 of p(n), Acta Arith. **2**(38), 105–109 (1980).
- 29. M. D. Hirschhorn, On the parity of p(n)-ii, J. Comb. Theory Ser. A **62**(1), 128–138 (1993).
- 30. M. Hirschhorn and M. Subbarao, On the parity of p(n), Acta Arith. **50**(4), 355–356 (1988).
- M. Newman, Congruences for the coefficients of modular forms and some new congruences for the partition function, *Can. J. Math.* 9, 549–552 (1957).
- 32. M. Newman, Some theorems about $p_r(n)$, Can. J. Math. 9, 68–70 (1957).
- K. G. Ramanathan, Identities and congruences of the Ramanujan type, Can. J. Math. 2, 168–178 (1950).
- P. Hammond and R. Lewis, Congruences in ordered pairs of partitions, Int. J. Math. Sci. 45–48, 2509–2512 (2004).
- W. Y. C. Chen, D. K. Du, Q. H. Hou, and L. H. Sun, Congruences of multipartition functions modulo powers of primes, *Ramanujan J.* 35(1), 1–19 (2014).
- D. Tang, Congruences modulo powers of 5 for k-colored partitions, J. Number Theory 187, 198–214 (2018).
- 37. S. K. B. Radhakrishna, R. Narendra, and K. R. Rajanna, Arithmetic properties of Ramanujan's general partition function for modulo 11, *Kuwait J. Sci.* 48(1) (2021).

- 38. M. Merca, On the Ramanujan-type congruences modulo 8 for the overpartitions into odd parts, *Quaest. Math.* 1–8, 1567–1574 (2021).
- 39. P. Paule and C. S. Radu, A New Witness Identity for 11 p(11n+6). Springer, pp. 625–639 (2016).
- 40. P. Paule and C. S. Radu, A proof of the Weierstrass gap theorem not using the Riemann–Roch formula, Ann. Comb. 23(3), 963–1007 (2019).
- 41. A. Goswami, A. K. Jha, and A. K. Singh, Some identities for the partition function, J. Math. Anal. Appl. 508(1), 125–164 (2022).
- 42. N. A. Smoot, On the computation of identities relating partition numbers in arithmetic progressions with eta quotients: An implementation of Radu's algorithm, *J. Symb. Comput.* **104**, 276–311 (2021).

© 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0003

Chapter 3

Collatz Hypothesis and Kurepa's Conjecture

Nicola Fabiano^{*,§}, Nikola Mirkov^{*,¶}, Zoran D. Mitrović^{†,||}, and Stojan Radenović^{‡,**}

*"Vinča" Institute of Nuclear Sciences — National Institute of the Republic of Serbia, University of Belgrade, Mike Petrovića Alasa 12–14, 11351 Belgrade, Serbia
[†]Faculty of Electrical Engineering, University of Banja Luka Patre 5, 78 000 Banja Luka, Bosnia and Herzegovina
[‡]Faculty of Mechanical Engineering, University of Belgrade, Kraljice Marije 16, 11 120 Beograd 35, Serbia
[§]nicola.fabiano@gmail.com
¶nmirkov@vin.bg.ac.rs
"zoran.mitrovic@etf.unibl.org
**radens@beotel.net

We discuss and give some insights on the Collatz conjecture, known as 3N + 1, and Kurepa's hypothesis on the left factorial. First, the Collatz conjecture is considered and the density of values is compared to Planck's black body radiation in physics, showing a remarkable agreement between the two. We also briefly discuss a generalization of Collatz conjecture for a generic sequence qN + 1 by means of numerical analysis. Then, we give a brief historical excursus and prove in a simple way some properties of Kurepa's function, also called the left factorial. We introduce Kurepa's hypothesis, propose a new description, and the relation to Bezout's parameters and the Diophantine equation. A numerical analysis supports Kurepa's hypothesis and the conjecture about distribution for Kurepa's function.

1. Introduction

In this chapter, two different problems of number theory will be discussed. The first one concerns the Collatz function and the aspect of the total stopping time of iterations starting from different starting points, which distribution mimics Planck's black body radiation density for photons.

The second part considers a lesser known problem of Kurepa's function, or left factorial, and its factorizability. A Diophantine equation for this problem has been written, and some numerical results are shown to support that a particular distribution for Kurepa's function approaches a uniform distribution for large numbers.

2. The Collatz Conjecture

A very simple function is the basis of the Collatz conjecture. For $N \in \mathbb{N}$ define the function C(N) as

$$C(N) = \begin{cases} \frac{N}{2} & \text{if } N \text{ even,} \\ 3N+1 & \text{if } N \text{ odd.} \end{cases}$$
(1)

A recursive application of this function gives a sequence. Starting from a positive integer N and applying recursively the function C(N)one ends up with a sequence $\{a_i\}_{i\in\mathbb{N}}$ whose generic term a_i could be written as follows:

$$a_{i} = \begin{cases} N & \text{for } i = 0\\ C(a_{i-1}) & \text{for } i > 0, \end{cases}$$
(2)

so that $a_i = [C(N)]^i$. A couple of examples: from N = 7, the resulting sequence is

7, 22, 11, 34, 17, 52, 26, 13, 40, 20, 10, 5, 16, 8, 4, 2, 1,

while starting from N = 1236 the sequence is

 $1236, 618, 309, 928, 464, 232, 116, 58, 29, 88, 44, 22, 11, 34, 17, 52, 26, \\13, 40, 20, 10, 5, 16, 8, 4, 2, 1.$

Any sequence is concluded once it reaches the number 1. From the previous examples, the sequence is concluded after 17 steps in the case of N = 7, while 27 steps are needed for the case N = 1236. It is apparent that for a power of two, $N = 2^k$, $k \in \mathbb{N}$, the value of 1 is reached after k steps. The total stopping time is defined as the number of steps necessary for reaching the value of 1.

The conjecture stated by Collatz^1 in 1937 is the following: using the function (1) and starting from any natural number, the sequence (2) has a finite stopping time. To the present day (2022) his conjecture has neither being proved nor disproved.

There is plenty of literature on the subject, see for instance Refs. 2-6.

We are here not interested in proposing a possible solution to this conjecture. Rather, we will discuss the problem of total stopping times and its distribution. In Figures 1(a)-(d) we have shown total stopping times for different starting values of N.

A different kind of plot for total stopping times is presented in Figure 2(a) and (b). For fixed N, total stopping times are shown with respect to frequencies. The latter results suggest a close parallel to a well-known problem in quantum physics.

3. Planck's Radiation

Assume to have a photon gas (i.e. electromagnetic radiation) contained at equilibrium inside a cavity of volume V at a temperature T. Such a system is known as "blackbody cavity." It is wellknown that this free electromagnetic field can be described with a sum of harmonic oscillators, each with a fixed frequency ν . Quantum mechanics allows to each photon of frequency ν only discrete energy levels $(n' + \frac{1}{2})2\pi\hbar\nu$, for $n' \in \mathbb{N}$.



Fig. 1. Total stopping times for different starting N.



Fig. 2. Histograms for $N = 10^4$ of total stopping times with respect to frequencies for different bin widths.

The partition function of such a system at the temperature T, $\beta = 1/k_B T$, can be explicitly calculated as

$$Z = \sum_{n'=0}^{+\infty} \exp(-2\pi\beta\hbar\nu n') = \frac{1}{1 - \exp(-2\pi\beta\hbar\nu)}.$$
 (3)

The average energy for a photon of frequency ν is given by

$$\langle E \rangle = -\frac{\partial (\ln Z)}{\partial \beta} = \frac{2\pi\hbar\nu \exp(-2\pi\beta\hbar\nu)}{1 - \exp(-2\pi\beta\hbar\nu)}$$
$$= \frac{2\pi\hbar\nu}{\exp(2\pi\beta\hbar\nu) - 1} = \frac{2\pi\hbar\nu}{\exp[2\pi\hbar\nu/(k_B T)] - 1}.$$
(4)

Each photon of frequency ν has a definite momentum $\vec{p} = \hbar \vec{k}$, with wave number $k = |\vec{k}| = (2\pi/c)\nu$. Inside a volume V the number of photon momenta between k and k + dk is given by

$$\frac{V}{(2\pi)^3} 8\pi k^2 dk = 8\pi \frac{V}{c^3} \nu^2 d\nu.$$

The internal energy U of the system is therefore obtained integrating the average energy of each photon present inside the cavity

$$U = 8\pi \frac{V}{c^3} \int_0^{+\infty} \langle E \rangle \nu^2 d\nu = 16\pi^2 \frac{V}{c^3} \int_0^{+\infty} \frac{\hbar \nu^3}{\exp[2\pi \hbar \nu/(k_B T)] - 1} d\nu,$$

so that the internal energy per unit volume is obtained by

$$\frac{U}{V} = \int_0^{+\infty} u(\nu, T) d\nu, \qquad (5)$$

where

$$u(\nu,T) = 16\pi^2 \left(\frac{\hbar}{c^3}\right) \frac{\nu^3}{\exp[2\pi\hbar\nu/(k_B T)] - 1},\tag{6}$$

this is the famous Planck radiation law of photon energy density⁷ at frequency ν and fixed temperature T. Calculating the integral (5) one could observe that U/V increases with temperature as T^4 , thus obtaining the Stefan–Boltzmann law for the power radiated from a black body. The Stefan–Boltzmann law was discovered even before the advent of quantum mechanics by thermodynamics arguments alone.

Planck's radiation density function (6) has, remarkably, the same qualitative behavior as the total stopping times with respect to frequencies shown in Figures 2(a) and (b).

Table 1. Translation table Collatz–Planck.

Collatz	\longleftrightarrow	Planck
Frequency Total stopping time	$\underset{\longleftrightarrow}{\longleftrightarrow}$	Photon frequency Black body radiation density



Fig. 3. Histograms: Total stopping times with respect to frequencies. Functions: Planck's black body radiation density with respect to photon frequency.

In Table 1 we have proposed an equivalence language between Collatz distribution of total stopping times and Planck's radiation density.

The results of the comparison of total stopping times and Planck's radiation density are shown in Figures 3(a)-(d) for different values of starting N.

It could be observed that the agreement between the two functions increases with increasing N, and that Planck's radiation law tail overestimates a little the decrease of stopping times with respect to N.



Fig. 4. (a) Scaling of total stopping times with respect to starting value of n, $n = \log(N)$. (b) Scaling of temperatures with respect to starting value of n, $n = \log(N)$.

Using the results of Ref. 8, Planck's radiation law (6), and the Rosetta Stele of Table 1 we could infer that for large values of frequency ν of photon the total stopping time goes to zero at least as an exponential function

Total stopping time
$$\sim \nu^3 \exp(-\nu)$$
. (7)

In Figure 4(a) and (b) we show, respectively, the scaling of total stopping times, and the scaling of temperature T of Planck's radiation density with respect to n, the logarithm of starting N, $n = \log(N) (\approx 2.3 \ln(N))$. The temperature T is related to the longest stopping time measured, that is the peak of histogram graphs in Figure 3(a)–(d). Figure 4(a) shows a fairly linear scaling of total stopping times with respect to n, the logarithm of starting N,

Total stopping times
$$\sim \log(N)$$
. (8)

Figure 4(b) shows an almost perfect linear scaling for temperatures T, or maximum values of total stopping times, with the logarithm of starting N:

Temperatures
$$\sim \log(N)$$
. (9)

From Stefan–Boltzmann's law discussed before we could also estimate that the sum of total stopping times scales with the fourth power of $\log(N)$:

$$\sum (\text{Total stopping times}) \sim (\log(N))^4.$$
(10)

Although they are no rigorous proofs, but rather hints, all these collected results show that for a finite N total stopping times are finite, and their sum is finite as well, thus supporting the Collatz conjecture.

4. Collatz Generalization

Define the generalization of Collatz function (1) C(N), C(q, N), in the following manner:

$$C(q, N) = \begin{cases} \frac{N}{2} & \text{if } N \text{ even,} \\ qN+1 & \text{if } N \text{ odd,} \end{cases}$$
(11)

it has fixed points x = 0 and x = -1/(q-1), respectively. As even N behaves as the usual Collatz function, we have to focus only on odd N. Let q be even, q = 2j, $j \in \mathbb{N}$. For odd N = 2m + 1, $m \in \mathbb{N}$, one has

$$2j(2m+1) + 1 = 2[j(2m+1)] + 1$$

which is again an odd number and grows without bounds when applying recursively the function C(2j, 2m+1). Therefore, q has to be odd, q = 2j + 1.

For N = 2m + 1, we obtain

$$(2j+1)(2m+1) + 1 = 2[(2j+1)m + (j+1)]$$
(12)

which is even. Using the notation " \rightarrow " as "transforms to" we obtain the results presented in Table 2.

Actually, the original Collatz function allegedly loops on the final sequence 4, 2, 1 for any N, so one could say that the conjecture is verified when any starting number $N \in \mathbb{N}$ enters the aforementioned loop.

j	q	$(2m+1) \rightarrow$
1 2 3 4	3 5 7 9	2(3m+2) 2(5m+3) 2(7m+4) 2(9m+5)

Table 2.Transformation table forCollatz generalization.

For q > 3, the situation is different, probably due to the fact that, considering Table 2, and formula (12), one evinces that the ratio

$$\frac{2j+1}{j+1}$$

is larger than 3/2, so the sequence reaches fewer odd numbers than even numbers.

In the case q = 5, we obtain various different loops than the usual sequence 4, 2, 1. Here are some examples of finite iterations. For N = 5:

 $5, 26, 13, 66, 33, 166, 83, 416, 208, 104, 52, 26, 13, \ldots$

For N = 13:

 $13, 66, 33, 166, 83, 416, 208, 104, 52, 26, 13, 66, \ldots$

For N = 15:

 $15, 76, 38, 19, 96, 48, 24, 12, 6, 3, 16, 8, 4, 2, 1, 6, 3, 16, 8, 4, \ldots$

For N = 17:

 $17, 86, 43, 216, 108, 54, 27, 136, 68, 34, 17, 86, \ldots$

While, for other values of N, C(5, N) diverges, like for $N = 7, 9, 11, 14, 18, \ldots$

When q > 5, the situation drastically worsens, and C(q, N) diverges already for starting N = 5, C(q, 5), for the values of $q = 9, 11, 13, 15, 17, 19, 21, \ldots, 211, \ldots$

Considering all those numerical results, one could conjecture that only the Collatz function case, the one with q = 3, has no divergences and ends on a unique loop, 4, 2, 1, while for q > 3 there exist more final loops, all separated by some divergent points.

5. Kurepa's Hypothesis

In 1971, following his seminal works,^{9–11} Kurepa introduced¹² the left factorial function, with the symbol !n, where $n \in \mathbb{N}$, also known as Kurepa's function,

$$K(n) = !n = \sum_{i=0}^{n-1} i! = \sum_{i=0}^{n-1} \Gamma(i+1),$$
(13)

and later^{13} extended its definition to arguments z on the complex plane $\Re(z)>0$

$$K(z) = \int_0^{+\infty} e^{-t} \frac{t^z - 1}{t - 1} dt,$$

$$\Gamma(x + 1) = \int_0^{+\infty} e^{-t} t^x dt.$$
(14)

For $n \in \mathbb{N}$, $\Gamma(n+1) = n!$. The recurrence relation holds true:

$$\Gamma(x+1) = x\Gamma(x), \tag{15}$$

and its asymptotic behavior (known as Stirling's formula) for $x \to +\infty$ is given by

$$\Gamma(x+1) \sim \left(\frac{x}{e}\right)^x \sqrt{2\pi x} \left[1 + \frac{1}{12x} + \mathcal{O}\left(\frac{1}{x^2}\right)\right].$$
 (16)

Substituting (14) in (13), we obtain Kurepa's function on the complex plane as follows:

$$K(n) = \int_0^{+\infty} e^{-t} \sum_{i=0}^{n-1} t^i dt = \int_0^{+\infty} e^{-t} \frac{t^n - 1}{t - 1} dt$$

that is

$$K(x) = \int_0^{+\infty} e^{-t} \frac{t^x - 1}{t - 1} dt.$$
 (17)

The following theorems for the function K(x) and its relation with function $\Gamma(x)$ were first established in Ref. 13. Here we will show novel simple proofs for them.

Theorem 1. The following relation holds true for $z \in \mathbb{C}$:

$$K(z) = K(z+1) - \Gamma(z+1)$$
(18)

Proof. Using the expressions of (14) and (17) in (18), we obtain

$$\int_{0}^{+\infty} e^{-t} \frac{t^{z} - 1}{t - 1} dt = \int_{0}^{+\infty} e^{-t} \left[\frac{t^{z+1} - 1}{t - 1} - t^{z} \right] dt$$
$$= \int_{0}^{+\infty} e^{-t} \left[\frac{t^{z} - 1}{t - 1} \right] dt.$$
(19)

Theorem 2. Kurepa's function has the following limit for $x \to +\infty$:

$$\lim_{x \to +\infty} \frac{K(x)}{\Gamma(x)} = 1.$$
 (20)

Proof. From the definition (13) of K(x), for $n \in \mathbb{N}$ we have

$$K(n) = \sum_{i=0}^{n-1} i! \sim (n-1)! + (n-2)! + \dots + 1! + 0!.$$
(21)

Inserting (21) in (20), we obtain

$$\lim_{n \to +\infty} \frac{K(n)}{\Gamma(n)} = \lim_{n \to +\infty} \frac{(n-1)! + (n-2)! + \cdots}{(n-1)!}$$
$$= \lim_{n \to +\infty} \left[1 + \mathcal{O}\left(\frac{1}{n-1}\right) \right] = 1,$$

this result obtained so far is valid for the case $n \in \mathbb{N}$ only. The integrand of (19) is well-defined for any $t \in \mathbb{R}^+$, so $K(z), z \in \mathbb{C}, \Re(z) > 0$ could be rewritten as

$$K(z) = \int_{0}^{+\infty} e^{-t} t^{z-1} \left[\sum_{n=0}^{+\infty} \left(\frac{1}{t^n} - \frac{1}{t^{n+z}} \right) \right] dt$$
$$\sim \Gamma(z) + \Gamma(z-1) + \Gamma(z-2) + \dots,$$
(22)

the asymptotic behavior is given for $z \to \infty$, and using (15) one has

$$\lim_{z \to \infty} \frac{K(z)}{\Gamma(z)} = \lim_{z \to \infty} \frac{\Gamma(z) + \frac{\Gamma(z)}{z-1} + \mathcal{O}\left[\frac{\Gamma(z)}{(z-1)(z-2)}\right]}{\Gamma(z)} = 1.$$

Theorem 3. Kurepa's function has the following limit for $x \to +\infty$:

$$\lim_{x \to +\infty} \frac{K(x)}{\Gamma(x+1)} = 0.$$

Proof. Divide (18) by $\Gamma(x+1)$ obtaining

$$\lim_{x \to +\infty} \frac{K(x)}{\Gamma(x+1)} = \lim_{x \to +\infty} \frac{K(x+1)}{\Gamma(x+1)} - 1 = 1 - 1 = 0$$

by virtue of (20).

6. Diophantine Equation

In this section, we present some equivalent statements for Kurepa's hypothesis, proposing some other new forms, and a suggestion for a new research direction by means of a linear Diophantine equation.

In the following section we present new numerical results supporting the conjecture of uniform distribution for K(p).

Hypothesis 4. Kurepa's hypothesis is that the following, equivalent statements hold true:

$$gcd(K(n), n!) = 2, \qquad n \ge 2,$$

$$mod(K(n), n) \not\equiv 0 \qquad n > 2,$$

$$mod(K(p), p) \not\equiv 0, \qquad p \ge 3, \ p \text{ prime number.}$$
(23)

Let be $n \in \mathbb{N}$, $n \ge 2$, then, using the Euclidean algorithm for finding the gcd(!n, n!), the first formula of (23) could be rewritten as

$$x \cdot ! n + y \cdot n! = 2, \tag{24}$$

 \Box

if such $x, y \in \mathbb{Z}$ exist. This is a Diophantine equation, and (x, y) are Bezout's parameters.¹⁴ From Kurepa's hypothesis, gcd(!n, n!) = 2, therefore, Eq. (24) has an infinite number of solutions, that is Bezout parameters. If (x_0, y_0) is a pair that solves (24), then it follows that

$$x = x_0 + \frac{!n}{2}k, \ y = y_0 - \frac{n!}{2}k$$

with $k \in \mathbb{Z}$ all being Bezout's parameters that solve (24).

For instance, for n = 4 we have

$$x \cdot 10 + y \cdot 24 = 2$$

and $(x_0, y_0) = (5, -2)$. Then x = 5 - 12k and y = -2 + 5k are Bezout's parameters of this equation.

Dividing (24) by 2 one could also observe that !n/2 and n!/2 are coprime numbers (in particular !n/2 is odd while n!/2 is even). Therefore, !n/2 has no factors smaller than n, thus recovering the second formula of (23).

First formula of (23) is also equivalent to

$$gcd(!!(2n), (2n)!!) = 2,$$
$$gcd(!!(2n+1), (2n+1)!!) = 2,$$

where we have defined $!!(2n) = \sum_{i=0}^{n-1} (2i)!!$ with 0!! = 0, $!!(2n+1) = \sum_{i=0}^{n-1} (2i+1)!!$, and have used the relation $gcd(a, b) \times lcm(a, b) = a \times b$.

There are many equivalent forms for Kurepa's hypothesis (see Ref. 15 for more details), first one of (23) was stated in his original paper.¹²

In 2004, Ref. 16 presented a proof of Kurepa's hypothesis that was shown in 2011 to be wrong by the same authors,¹⁷ so they retracted the original paper. Until the present time (2022) there does not exist a real proof or a counterexample for (23).

7. Numerical Analysis for Kurepa's Hypothesis

We now present a numerical analysis and some new results of Kurepa's Hypothesis, starting from the last formula of (23). We have analyzed the first 11,000 prime numbers p, from 3 to the 11,000th prime number being 116,447, !116,447 \approx 116,446! \approx 1.045 \times 10^{539,361} (the first approximation is valid because of (20)). According to the hypothesis found in Ref. 18, K(p) modulo p is a random number with random uniform distribution. In order to check this hypothesis, we have shown the distribution of mod(K(p), p) and have compared it to random numbers, random (p), with uniform distribution in [0, p). In Figure 5, the value of mod(K(p), p) is shown as a function of the prime p for the first 11,000 prime numbers. There is no value of p for which mod(K(p), p) = 0, thus Kurepa's hypothesis has been



Fig. 5. mod (!p, p) versus p for the first 11,000 prime numbers.



Fig. 6. mod (!p, p)/p versus p for the first 11,000 prime numbers.

verified up to p = 116,447. Figure 6 shows the normalized plot of mod (K(p), p)/p, providing us with the more familiar rectangular uniform distribution of random numbers in [0, p).



Fig. 7. random (p) versus p for the first 11,000 prime numbers.



Fig. 8. random (p)/p versus p for the first 11,000 prime numbers.

Figures 7 and 8 show, respectively, the same kind of plots as before, this time for a uniform distribution of random numbers random (p) generated in the range [0, p).



Fig. 9. Difference of mod(!p,p)/p from a uniform distribution, in percentage, with respect to number of primes.

The comparison of the two groups of results does not show substantial differences in distributions between the results of mod(K(p), p) and the random number generator random (p). There is also no visible pattern that could possibly spoil the conjectured randomness of mod(K(p), p) in Figures 5 and 6.

In Figure 9 we compare the results of $\operatorname{mod}(!p, p)/p$, shown in Figure 6, with respect to a uniform distribution between [0, 1], as a function of number of primes taken into account. As expected, with increasing number of samples considered, that is with a larger number of primes, the distribution of $\operatorname{mod}(!p, p)/p$ is getting closer to a uniform distribution, with an error smaller than 0.5%, which further decreases when the number of primes is larger than 7000.

This numerical analysis concludes that Kurepa's hypothesis

$$\mod(K(p),p)\not\equiv 0$$

is true for all primes p verified so far, and also that this value is randomly uniformly distributed in the range [0, p).

In Ref. 18, the search for a counterexample of the conjecture was performed, without success, for $p < 2^{34} \approx 1.718 \times 10^{10}$ by means of GPU computing. A fairly exhaustive list of all numerical attempts to solve this problem is also given.

8. Conclusions

Almost 90 years ago, Collatz introduced his simple, often also called 3N + 1, yet unresolved problem. To use the words of Erdös, "Mathematics may not be ready for such problems." In fact, there is no general method suitable for tackling those kinds of problems. We have shown here heuristically that the stopping of time distributions closely resembles Planck's black body radiation density known in Physics. We have also shown that variations of the Collatz problem, qN + 1, lead to different loops separated by divergent points.

About 50 years have passed since the formulation of Kurepa's hypothesis.¹² Many attempts have been made in order to provide a solution to this problem, either by means of a rigorous theorem, with an approximate solution, or with a numerical approach.^{18–46} Various tentative solutions increase the importance of Kurepa's hypothesis, and considering its simple formulation it deserves to stay in the realm of other famous unsolved problems in number theory like Collatz conjecture, Goldback's conjecture, de Polignac's conjecture, Legendre's conjecture, Landau's problem, to name a few.

We have brought simplifications to some proofs concerning Kurepa's function. There are some other equivalent statements of Kurepa's hypothesis, and a new study route using a linear Diophantine equation. Quantitative computer simulations support all the aforementioned hypotheses, and show also how the distribution of mod(!p,p)/p approaches a uniform distribution — a fact not originally envisaged by Kurepa — with increasing prime p.

Acknowledgment

The author is thankful to the reviewers for their constructive comments.

References

- 1. Lothar Collatz biography. https://mathshistory.st-andrews.ac.uk/ Biographies/Collatz.
- 2. D. Applegate and J. C. Lagarias, Density bounds for the 3x + 1 problem. I. Tree-search method, *Math. Comput.* **64**(209), 411–426 (1995). doi: 10.2307/2153345.

- 3. D. Applegate and J. C. Lagarias, Density bounds for the 3x + 1 problem. II. Krasikov inequalities, *Math. Comput.* **64**(209), 427–438 (1995). doi: 10.2307/2153346.
- 4. E. W. Weisstein, Collatz problem. From MathWorld A Wolfram Web Resource. https://mathworld.wolfram.com/CollatzProblem.htm.
- R. K. Guy, Unsolved Problems in Number Theory, 2nd edn. Problem Books in Mathematics, Unsolved Problems in Intuitive Mathematics, I, Springer-Verlag, New York (1994).
- S. A. Kurtz and J. Simon, The undecidability of the generalized Collatz problem. In: J. Y. Cai, S. B. Cooper, and H. Zhu (eds.), *Theory and Applications of Models of Computation*, TAMC 2007. Lecture Notes in Computer Science, vol 4484, pp. 542–553, Springer-Verlag, Berlin, Heidelberg (2007). https://doi.org/10.1007/978-3-540-72504-6_49.
- M. Planck, Über das Gesetz der Energieverteilung im Normalspektrum, Annal. Phys. 4(3), 553–563 (1901).
- N. Fabiano, N. Mirkov, and S. Radenović, Collatz hypothesis and Planck's black body radiation, J. Siber. Federal Univ. Math. Phys. 2(3), 1–5 (2021).
- Đ. Kurepa, O faktorijelima konačnih i beskonačnih brojeva, Rad. Jugosl. Akad. Znan. Umjet. 296, 105–112 (1953).
- D. Kurepa, Über die Faktoriellen endlicher und unendlicher Zahlen, Bull. Inter. Acad. Yougoslave. Cl. Sci. Math. Phys. Tech. 12, 51–64 (1954).
- D. Kurepa, Factorials of cardinal numbers and trees, *Glasnik Mat.-Fiz.* Astronom. Društvo Mat. Fiz. Hrvatske Ser. II. 19, 7–21 (1964).
- D. Kurepa, On the left factorial function !n, Math. Balkan. 1, 147–153 (1971).
- D. Kurepa, Left factorial function in complex domain, Math. Balkan. 3, 297–307 (1973).
- N. Fabiano, N. Mirkov, Z. D. Mitrović, and S. Radenović, On some new observations on Kurepa's left factorial, *Math. Anal. Contemp. Appl.* 4(3), 1–8 (2022). https://doi.org/10.30495/maca.2022.1950392.1047.
- A. Ivić and Ž. Mijajlović, On Kurepa's problems in number theory, arXiv preprint math/0312202 (2003).
- D. Barsky and B. Benzaghou, Nombres de Bell et somme de factorielles, J. Théor. Nomb. Bordx. 16(1), 1–17 (2004).
- D. Barsky and B. Benzaghou, Erratum à l'article Nombres de Bell et somme de factorielles, J. Théor. Nomb. Bordx. 23(2), 527 (2011).
- V. Andrejić and M. Tatarević, Searching for a counterexample to Kurepa's conjecture, *Math. Comput.* 85(302), 3061–3068 (2016). doi: 10.1090/mcom/3098.

- V. Andrejić, A. Bostan, and M. Tatarević, Improved algorithms for left factorial residues, *Inform. Process. Lett.* 167, Paper No. 106078, 3 (2021). doi: 10.1016/j.ipl.2020.106078.
- 20. C. Ashbacher, An Introduction to the Smarandache Function. Erhus University Press, Vail, AZ (1995).
- L. Carlitz, A note on the left factorial function, Math. Balkan. 5, 37–42 (1975).
- I. Ilijaševic, Verification of Kurepa's left factorial conjecture for primes up to 2 31, IPSI Trans. Int. Res. 11(1), 7–12 (2015).
- B. C. Kellner, Some remarks on Kurepa's left factorial, arXiv preprint math/0410477 (2004).
- D. Kurepa, On some new left factorial propositions, *Math. Balkan.* 4, 383–386 (1974).
- 25. E. W. Weisstein, Smarandache-Kurepa function. From MathWorld A Wolfram Web Resource. https://mathworld.wolfram.com/ Smarandache-KurepaFunction.html.
- Đ. R. Kurepa, Right and left factorials, Boll. Un. Mat. Ital (4). 9 (suppl. fasc. 2), 171–189 (1974). Conferenze tenute in occasione del cinquantenario dell'Unione Matematica Italiana (1972).
- Đ. Kurepa and Ž. Mijajlović, Selected Papers of Uro Kurepa. Matematički institut SANU, Belgrade, Serbia (1996).
- B. J. Malešević, Some considerations in connection with Kurepa's function, *Publik. Elektroteh. Fakult. Ser. Mat.* 14, 26–36 (2003).
- B. J. Malešević, Some inequalities for Kurepa's functions, J. Inequal. Pure Appl. Math. 5(4), Paper No. 104–03 (2004).
- B. J. Malešević, Some inequalities for alternating Kurepa's function, Publik. Elektroteh. Fakult. Ser. Mat. 16, 70–76 (2005).
- B. J. Malešević, Some considerations in connection with alternating Kurepa's function, *Integr. Transf. Spec. Funct.* 19(10), 747–756 (2008).
- R. Meštrović, The Kurepa-Vandermonde matrices arising from Kurepa's left factorial hypothesis, *Filomat* 29(10), 2207–2215 (2015).
- Ž. Mijajlović, On some formulas involving !n and the verification of the !n-hypothesis by use of computers, Publ. Inst. Math. 47(61), 24–32 (1990).
- Ž. Mijajlović, Fifty years of Kurepa's !n hypothesis, Bull. T. CLIV l'Acad. Serb. Sci. Arts 46, 169–181 (2021).
- G. V. Milovanović, Expansions of the Kurepa function, Publ. Inst. Math. 57(71), 81–90 (1995).
- G. V. Milovanović, A sequence of Kurepa's functions, *Sci. Rev.* 19–20, 137–146 (1996).
- A. Petojević, On Kurepa's hypothesis for the left factorial, *Filomat* 12(Part 1), 29–37 (1998).

- A. Petojević and M. Żižović, Trees and the Kurepa hypothesis for left factorial, *Filomat* 13(13), 31–40 (1999).
- R. Rajkumar. Searching for a counterexample to Kurepa's conjecture in average polynomial time. PhD thesis, UNSW Sydney (2019).
- Z. Šami, On the *M*-hypothesis of D. Kurepa, *Math. Balkan.* 3, 530–532 (1973).
- 41. Z. Sami, A sequence $u_{n,m}$ and Kurepa's hypothesis on left factorial. Sci. Rev. Ser. Sci. Eng. **19–20**, 105–113 (1996).
- D. V. Slavić, On the left factorial function of the complex argument, Math. Balkan. 3, 472–477 (1973).
- J. Stanković, Über einige Relationen zwischen Fakultäten und den linken Fakultäten, Math. Balkan. 3, 488–495 (1973).
- V. S. Vladimirov, Left factorials, Bernoulli numbers, and the Kurepa conjecture. Publ. Inst. Math. 72(86), 11–22 (2002).
- 45. M. Živković, The number of primes $\sum_{i=1}^{n} (-1)^{n-i} i!$ is finite, *Math. Comput.* **68**(225), 403–409 (1999).
- M. Živković, Massive computation as a problem solving tool, *Intervals* 23(26), 8–64 (2001).

© 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0004

Chapter 4

On Zeta Functions and Allied Theta Functions^{*}

Hongyu Li^{‡,||}, Takako Kuzumaki^{†,§,**}, and Shigeru Kanemitsu^{¶,††}

[‡]Sanmenxia Suda Transportation Energy Saving Technology Co., Ltd., No. 1 Taiyan Rd, Henan 472000, People's Republic of China [§]Faculty of Engineering, Gifu University, Gifu 501-1193, Japan ¶Faculty of Engineering, Kyushu Inst. Tech., Kitakyushu 804-0855, Japan [∥]hongyu@sdjtjt.com **kuzumaki@gifu-u.ac.jp ^{††}omnikanemitsu@yahoo.com

In this chapter, we completely clarify the meaning of a ramified functional equation of Schnee type as the union of the Riemann–Hecke– Bochner (RHB) correspondence and the parity, thus enabling one to describe even and odd parts of an entity whenever it is possible to introduce parity.

We shall uplift the Hurwitz formula between the Hurwitz and Lerch zeta functions to the *ramified Riemann–Hecke–Bochner correspondence*, i.e. to a ramified functional equation whose even, resp. odd, part is in RHB correspondence with the theta function, resp. the conjugate theta function. This is shown not to be coincidental and will be extended to

^{*}Dedicated to Professor Dr. Antanas Laurinčikas with deep respect and friendship.

 $^{^{\}dagger}\!\mathrm{Corresponding}$ author.

H. Li et al.

a wider class of periodic Dirichlet series. It turns out that the Riemann zeta function $\zeta(s)$ is the even part, however the expression for merely the even part does not lead to $\zeta(s)$ but describes the Clausen function. It is the union of even and odd parts that gives rise to the functional equation for $\zeta(s)$ and entails almost all known formulas in the theory of gamma and zeta functions. The odd part itself vanishes in the functional equation for $\zeta(s)$, but it contains the partial fraction expansion for the cotangent function, an equivalent to it, thereby we establish the following principle: W = E + O, W = Whole, E = Even and O = Odd.

While in the theory of modular relations,¹³ equivalent expressions to the functional equation are pursued, we examine here the counterpart of the zeta functions: the theta-function-like objects in the ramified RHB correspondence which govern the intrinsic nature and properties of the corresponding zeta functions. Riemann's famous proof of the functional equation for $\zeta(s)$ by the even part of the theta function ϑ_3 is a typical one and is an example of the Edwald expansion.

Motivated by coincidence of the above ramified functional equation with the one established by Schnee in 1930 as modified by Ishibashi and Kanemitsu in 1999, we turn to the study of Dirichlet series with periodic coefficients in Section 1.1. We complete the evaluation of the Laurent coefficients and give material for Section 2, where we deduce the Chowla– Selberg integral formula as an example of the Fourier–Bessel expansion. It can be proved by the Hurwitz formula, i.e. by RHB correspondence under the beta transform in contrast with the Hecke gamma transform. Then we establish (genesis of) the Lerch–Chowla–Selberg formula as an example of the RHB correspondence as linked by the decomposition theorem.

Thus, we follow the natural flow of ideas: theta function \rightarrow Hurwitz (and Lerch) zeta-function \rightarrow Fourier-Bessel expansion with parity.

1. Introduction and Ramified RHB Correspondence

Decidedly one of the most influential and creative source of ideas lies in the Riemann–Hecke–Bochner correspondence, whose naming is in Ref. 1, meaning the correspondence between the transformation formula (of modular form-like objects) and the functional equation (of the corresponding zeta functions). The origin of this correspondence was found by Refs. 2–5. In Ref. 13, equivalent expressions for the functional equation have been developed under the name of modular relations — as a memorial to Bochner.

In this chapter, we clarify the meaning of a ramified functional equation of Schnee type to its root as the union of the Riemann– Hecke–Bochner (RHB) correspondence and the parity, thus enabling one to describe even and odd parts of an entity whenever it is possible to introduce parity. In the specific case of the complex exponential function basis, we may express the ramified functional equation in the form of the generalized Euler identity and as with the Euler identity, we may deduce almost all results from one formula.

As a precursor, we shall describe the RHB correspondence between the theta-like functions and the Hurwitz and the Lerch zeta functions as a natural generalization of the Riemann zeta function. Ehrenpreis (1987) contains another advanced treatment through Appel transform (p. 73). The odd part has been treated in Ref. 6 to some extent and here we make clear the source of the transformation formula and the even and odd parts of the Hurwitz and the Lerch zeta functions. The main results in this section are Theorem 1 and Theorem 2. The odd part alone in the former theorem entails the partial fraction expansion for the cotangent function, an equivalent to the functional equation for the Riemann zeta function while the even part gives slightly restricted information and the union of even and odd parts in the latter theorem describes the whole structure for an even function and includes the functional equation.

We speak of the RHB correspondence or the modular relation to mean the genesis of the functional equation *under the Mellin transform pair* (1) as well as the equivalent expressions to the functional equations.

The ramified functional equation in Definition 1 (18), as well as the Eisenstein formula (41), strongly motivates to shift to the study of Dirichlet series with periodic coefficients in Section 1.1. Already, Schnee,⁷ as formulated in Ref. 8, established the same functional equation as the Hurwitz and Lerch zeta functions. Supplementing the results in Ref. 6 etc., we describe the Laurent coefficients of the relevant Dirichlet series including the Dirichlet *L*-functions, which in particular gives one hand of the decomposition of the Dedekind zeta-function of an imaginary quadratic field.

Then as another direction of the modular relation, we establish the Chowla–Selberg integral formula for the Epstein zeta function with a positive definite quadratic form (the odd case as opposed to indefinite form) as the Fourier–Bessel expansion $H_{1,1}^{1,1} \leftrightarrow H_{0,2}^{2,0}$. This hinges on the beta transform as well as the RHB correspondence between the theta function and the lattice zeta function. This gives the evaluation of the Laurent constant of the Dedekind zeta function, the other part of the decomposition. Equating them, we deduce (genesis of) the Lerch–Chowla–Selberg formula in Section 2.

We write

$$s = \sigma + it, \quad \sigma = \operatorname{Re} s$$

once and for all.

The Mellin transform pair $(\operatorname{Re} x > 0)$

$$\Gamma(s) = \int_0^\infty t^s e^{-t} \frac{\mathrm{d}t}{t}, \sigma > 0, \quad e^{-x} = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \Gamma(s) x^{-s} \,\mathrm{d}s, \quad c > 0$$
(1)

is the genesis of the RHB correspondence, cf. (48).

The Riemann zeta function is defined by

$$\zeta(s) = \sum_{n=1}^{\infty} \frac{1}{n^s} \tag{2}$$

for $\sigma > 1$, where the series is absolutely convergent. This is continued analytically to a meromorphic function over the whole plane with a unique simple pole at s = 1 with residue 1. The analytic continuation is done by some expressions in $0 < \sigma < 1$ and by the functional equation

$$\pi^{-\frac{s}{2}}\Gamma\left(\frac{s}{2}\right)\zeta(s) = \pi^{-\frac{1-s}{2}}\Gamma\left(\frac{1-s}{2}\right)\zeta(1-s) \tag{3}$$

for $\sigma < 0$. It turns out that for deeper theory on distribution of prime numbers, one needs the Euler product and Riemann himself defines by the Euler product first, cf. (114).

Reference 9 gives a rather thorough table of functional equations which arise from the theta-transformation formula, Cf. also Kanemitsu and Tsukada (2014), Table 2.1, p. 34. There, after the Riemann zeta function, there comes the Dedekind zeta function and Epstein zeta functions associated to quadratic forms, etc. It is well known that Riemann gave two proofs of the functional equation (3), one depending on a variant of the theta functions ϑ_3 and the other on the contour integral expression. For Riemann's theta-transformation proof, cf. Ref. 10, pp. 61–63, where the Dirichlet *L*-function case is also treated by the same technique, cf. Remarks 2 and 3. Let $\ell_s(x)$ be the boundary Lerch zeta function defined by

$$\ell_s(x) = \sum_{n=1}^{\infty} e^{2\pi i x n} n^{-s}, \quad \sigma > 1 \quad \text{or} \quad \sigma > 0, x \notin \mathbb{Z},$$
(4)

which has its counterpart, the Hurwitz zeta function

$$\zeta(s, x) = \sum_{n=0}^{\infty} \frac{1}{(n+x)^s}, \quad \sigma > 1, \ 0 < x \le 1.$$
(5)

This is continued meromorphically over the whole plane with a simple pole at s = 1. Both of them reduce to the Riemann zeta function

$$\zeta(s,1) = \ell_s(1) = \zeta(s).$$

These are connected by the *Hurwitz formula* (i.e. the functional equation for the Hurwitz zeta function): for $\sigma > 1, 0 < x \leq 1$,

$$\zeta(1-s,x) = \frac{\Gamma(s)}{(2\pi)^s} \left(e^{-\frac{\pi i s}{2}} \ell_s(x) + e^{\frac{\pi i s}{2}} \ell_s(1-x) \right), \tag{6}$$

while its reciprocal is

$$\ell_{1-s}(x) = \frac{\Gamma(s)}{(2\pi)^s} \left(e^{\frac{\pi i s}{2}} \zeta(s, x) + e^{-\frac{\pi i s}{2}} \zeta(s, 1-x) \right), \quad 0 < x < 1.$$
(7)

The limiting case with $x \to 1$ of (6) reads

$$\zeta(1-s) = \frac{2\Gamma(s)}{(2\pi)^s} \cos\left(\frac{\pi s}{2}\right) \zeta(s),\tag{8}$$

which is seen to be equivalent to (3).

By (7), the boundary Lerch zeta function $\ell_{1-s}(x)$ is continued meromorphically over the whole plane with s = 0 a plausible singular point. However, it is a removable singularity if $x \in \mathbb{R} - \mathbb{Z}$ in view of (36). Indeed, comparison of both sides as $s \to 1$ leads to the relation

$$\psi(x) - \psi(1 - x) = 2\pi i \ell_0(x) + \pi i, \tag{9}$$

which together with (35) entails (30), cf. the second half of the proof of Corollary 1.
We assemble the identities for $\ell_1(s)$, cf. e.g. Refs. 6 and 11,

$$\sum_{n=1}^{\infty} \frac{\cos(2\pi nx)}{n} + i \sum_{n=1}^{\infty} \frac{\sin(2\pi nx)}{n} = \ell_1(x)$$
(10)

$$= -\log(1 - e^{2\pi ix}) = \sum_{n=1}^{\infty} \frac{e^{2\pi inx}}{n} = A_1(x) - \pi i \overline{B}_1(x),$$

0 < x < 1, where

$$A_1(x) = -\log 2|\sin \pi x| = \sum_{n=1}^{\infty} \frac{\cos(2\pi nx)}{n}$$

is its real part (even part), the first Clausen function (or the logsine function) and the imaginary part (odd part) is (27) with $\varkappa = 1$, which reads

$$x - [x] - \frac{1}{2} = \bar{B}_1(x) = -\frac{1}{\pi} \sum_{n=1}^{\infty} \frac{\sin(2\pi nx)}{n}$$
(11)

for $x \notin \mathbb{Z}$, where [x] indicates the greatest integer function.

In a totally forgotten paper,¹² Hutchinson considers the zeta function

$$Z_b(s) = Z(a, b, s) = \sum_{\nu=0}^{\infty} (a\nu + b)^{-s}, \quad 0 < b \le a,$$

which is $a^{-s}\zeta(s, x)$, where $x = \frac{b}{a}$, so that $0 < x \leq 1$. Hence, this is a slight generalization of the Hurwitz zeta function. He considers the conjugate functions Z_b and Z_{a-b} and states the functional equation satisfied by the even and odd parts, (12), (20), (21):

$$Z_{b} + Z_{a-b} = \frac{2}{\pi} \left(\frac{2\pi}{a}\right)^{s} \sin\left(\frac{\pi s}{2}\right) \Gamma(1-s) \sum_{n=1}^{\infty} \frac{\cos\left(2\pi \frac{b}{a}n\right)}{n^{1-s}}$$
$$= \frac{1}{\pi} \left(\frac{2\pi}{a}\right)^{s} \sin\left(\frac{\pi s}{2}\right) \Gamma(1-s) (\ell_{1-s}(x) + \ell_{1-s}(1-x))$$
(12)

and

$$Z_{b} - Z_{a-b} = \frac{2}{\pi} \left(\frac{2\pi}{a}\right)^{s} \cos\left(\frac{\pi s}{2}\right) \Gamma(1-s) \sum_{n=1}^{\infty} \frac{\sin\left(2\pi \frac{b}{a}n\right)}{n^{1-s}} = -\frac{i}{\pi} \left(\frac{2\pi}{a}\right)^{s} \cos\left(\frac{\pi s}{2}\right) \Gamma(1-s)(\ell_{1-s}(x) - \ell_{1-s}(1-x)),$$
(13)

where $x = \frac{b}{a}$. He also proves abundance principle¹ for the functions that satisfy the functional equations (12), resp. (13), cf. Ref. 12, Theorems 2 and 3.

In what follows, we use the notation

$$f^{e}(x) = \frac{1}{2}(f(x) + f(-x)), \quad f^{o}(x) = \frac{1}{2}(f(x) - f(-x))$$

or

$$f^{e}(x) = \frac{1}{2}(f(x) + f(1-x)), \quad f^{o}(x) = \frac{1}{2}(f(x) - f(1-x)) \quad (14)$$

as the case may be, so that

$$f = f^e + f^o.$$

We state the special case of (12) and (13) as a theorem.

Theorem 1. The even part

$$\frac{1}{2}(\zeta(s,x) + \zeta(s,1-x)) = \zeta^{e}(s,x)
= \frac{1}{\pi}(2\pi)^{s} \sin\left(\frac{\pi s}{2}\right) \Gamma(1-s)\ell_{1-s}^{e}(x)
= (2\pi)^{s-1} \sin\left(\frac{\pi s}{2}\right) \Gamma(1-s)(\ell_{1-s}(x) + \ell_{1-s}(1-x))$$
(15)

is in RHB correspondence with the theta-transformation formula (43). But it does not lead to the functional equation since it does not hold for x = 1.

On the other hand, the odd part

$$\frac{1}{2}(\zeta(s,x) - \zeta(s,1-x)) = \zeta^{o}(s,x)
= -\frac{i}{\pi}(2\pi)^{s} \cos\left(\frac{\pi s}{2}\right) \Gamma(1-s)\ell_{1-s}^{o}(x)
= -i(2\pi)^{s-1} \cos\left(\frac{\pi s}{2}\right) \Gamma(1-s)(\ell_{1-s}(x) - \ell_{1-s}(1-x)) \quad (16)$$

is in RHB correspondence (in the form of (56)) with the transformation formula (57) for the conjugate theta function (55). In the limit as $s \to 1$, (16) in the long run entails the partial fraction expansion for the cotangent function: for nonintegral values of z,

$$\cot \pi z = \frac{1}{\pi z} + \frac{1}{\pi} \sum_{n=1}^{\infty} \left(\frac{1}{n+z} - \frac{1}{n-z} \right)$$
$$= \frac{1}{\pi z} + \frac{2z}{\pi} \sum_{n=1}^{\infty} \frac{1}{z^2 - n^2},$$
(17)

where the last series is absolutely convergent. It is known that (17) is equivalent to the functional equation (3).

Proof of the even part is given in Lemma 1 below while the odd part will be settled in Corollary 1 as well as toward the end of this section.

Definition 1. Specifying the general case of the ramified functional equation (see Ref. 13, Chapter 7), we say that the Dirichlet series $\varphi(s)$ and $\psi_1(s)$, $\psi_2(s)$ satisfy the ramified functional equation

$$\Gamma\left(\frac{s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\varphi(s)$$

$$=\Gamma\left(\frac{1-s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\psi_1(1-s)+\Gamma\left(\frac{s}{2}\right)\Gamma\left(1-\frac{s}{2}\right)\psi_2(1-s).$$
(18)

The following is the reciprocal of what is enunciated in Ref. 14.

Theorem 2. The genesis of the functional equation (3) for the Riemann zeta function lies in the union of the even part (15) and

the odd part (16):

$$\zeta(s,x) = \frac{1}{\pi} (2\pi)^s \Gamma(1-s) \left(\sin\left(\frac{\pi s}{2}\right) \ell_{1-s}^e(x) - i \cos\left(\frac{\pi s}{2}\right) \ell_{1-s}^o(x) \right)$$
$$= (2\pi)^{s-1} \Gamma(1-s) \left(\sin\left(\frac{\pi s}{2}\right) \left(\ell_{1-s}(x) + \ell_{1-s}(1-x) \right) - i \cos\left(\frac{\pi s}{2}\right) \left(\ell_{1-s}(x) - \ell_{1-s}(1-x) \right) \right), \tag{19}$$

which are miniscence of Euler's identity, where the even, resp. odd, part is in RHB correspondence with the theta functions, resp. the conjugate theta function. This leads to the functional equation (8) in view of vanishing of the odd part, i.e. the odd part is discarded in the functional equation. This amounts to (18) with

$$\varphi(s) = \pi^{-\frac{s}{2}} \zeta(s, x),$$

$$\psi_1(s) = \pi^{-\frac{s}{2}} \ell_s^e(x) = \frac{1}{2} (\pi^{-\frac{s}{2}} \ell_s(x) + \pi^{-\frac{s}{2}} \ell_s(1-x)),$$

$$\psi_2(s) = -i\pi^{-\frac{s}{2}} \ell_s^o(x) = -\frac{i}{2} (\pi^{-\frac{s}{2}} \ell_s(x) - \pi^{-\frac{s}{2}} \ell_s(1-x)).$$

(20)

We may say that the zeta functions φ, ψ_1, ψ_2 are in **ramified RHB** correspondence with the theta function $\vartheta, \tilde{\vartheta}$ in the sense that φ , resp. ψ_i , is in RHB correspondence with ϑ , resp. $\tilde{\vartheta}$, with respective gamma factors.

Proof. The equation (19) follows from Theorem 1. Equation (19) with x = 1 leads to (8). To prove that (20) satisfies (20), we use the well-known results: the duplication formula

$$\Gamma(s) = \sqrt{\pi}^{-1} 2^{s-1} \Gamma\left(\frac{s}{2}\right) \Gamma\left(\frac{s+1}{2}\right)$$

and the reciprocal relation

$$\Gamma(s)\Gamma(1-s) = \frac{\pi}{\sin \pi s}.$$

We substitute

$$\Gamma(1-s) = \sqrt{\pi}^{-1} 2^{-s} \Gamma\left(\frac{1-s}{2}\right) \Gamma\left(1-\frac{s}{2}\right),$$

$$\sin\left(\frac{\pi s}{2}\right) = \frac{\pi}{\Gamma\left(\frac{s}{2}\right) \Gamma\left(1-\frac{s}{2}\right)},$$

$$\cos\left(\frac{\pi s}{2}\right) = \sin\left(\frac{\pi(1-s)}{2}\right) = \frac{\pi}{\Gamma\left(\frac{1-s}{2}\right) \Gamma\left(\frac{1+s}{2}\right)}$$

in the right-hand side of (19) and multiplying by $\Gamma\left(\frac{s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)$, we deduce that

$$\Gamma\left(\frac{s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\zeta(s,x) = \frac{1}{\sqrt{\pi}}\pi^s$$
$$\times \left(\Gamma\left(\frac{1-s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\ell_{1-s}^e(x) - i\Gamma\left(\frac{s}{2}\right)\Gamma\left(1-\frac{s}{2}\right)\ell_{1-s}^o(x)\right).$$

Finally, multiplying by $\pi^{-\frac{s}{2}}$, we conclude (18).

The modification (20) is common and has the effect of taking the basic sequence as $\{\sqrt{\pi n}\}$ in place of $\{n\}$.

Corollary 1. The derivative of (19) at s = 0 amounts to

$$\zeta'(0,x) = -\left(\log 2\pi + \gamma\right)\bar{B}_1(x) + \frac{1}{2}A_1(x) + \frac{i}{\pi}\left(\ell_1'\right)^o(x), \qquad (21)$$

where

$$\left(\ell_1'\right)^o(x) = \frac{1}{2}(\ell_1'(x) - \ell_1'(1-x)) = -i\sum_{n=1}^\infty \frac{\log n}{n}\sin 2\pi nx.$$
 (22)

Under the assumption of the Lerch formula

$$\zeta'(0,x) = \log \frac{\Gamma(x)}{\sqrt{2\pi}},\tag{23}$$

(21) leads to Kummer's Fourier series

$$\log \frac{\Gamma(x)}{\sqrt{2\pi}} = \sum_{n=1}^{\infty} \left(\frac{1}{2n} \cos 2\pi nx + \frac{\gamma + \log 2\pi n}{\pi n} \sin 2\pi nx \right).$$
(24)

For $0 \leq \varkappa \in \mathbb{Z}$ and 0 < x < 1, we have

$$\ell_{\varkappa}(x) = \frac{(2\pi i)^{\varkappa - 1}}{\varkappa!} \left(A_{\varkappa}(x) - \pi i \bar{B}_{\varkappa}(x) \right), \tag{25}$$

where the even part $A_{\varkappa}(x)$ is the Clausen function of order \varkappa

$$A_{\varkappa}(x) = \frac{\varkappa!}{2(2\pi i)^{\varkappa - 1}} \sum_{n = -\infty}^{\infty} \frac{\operatorname{sgn}(n)e^{2\pi i n x}}{n^{\varkappa}}$$
$$= \frac{\varkappa!}{2(2\pi i)^{\varkappa - 1}} \left(\ell_{\varkappa}(x) + (-1)^{\varkappa - 1} \ell_{\varkappa}(1 - x) \right), \qquad (26)$$

while the odd part $\bar{B}_{\varkappa}(x)$ is the periodic Bernoulli polynomial of order \varkappa :

$$\bar{B}_{\varkappa}(x) = -\frac{\varkappa!}{(2\pi i)^{\varkappa}} \sum_{n=-\infty}^{\infty'} \frac{e^{2\pi i n x}}{n^{\varkappa}}$$
$$= -\frac{\varkappa!}{(2\pi i)^{\varkappa}} \left(\ell_{\varkappa}(x) - (-1)^{\varkappa-1} \ell_{\varkappa}(1-x) \right)$$
(27)

whereby the formula

$$\zeta(1-\varkappa, \{x\}) = -\frac{1}{\varkappa} \bar{B}_{\varkappa}(x). \tag{28}$$

These are complementary to each other and satisfy the parity relation

$$A_{\varkappa}(x) = (-1)^{\varkappa - 1} A_{\varkappa}(1 - x), \quad \bar{B}_{\varkappa}(x) = (-1)^{\varkappa} \bar{B}_{\varkappa}(1 - x).$$
(29)

Regarding the odd part in Theorem 1, (16) in the limit as $s \to 1$ entails the well-known formula

$$\psi(x) - \psi(1 - x) = -\pi \cot \pi x$$
 (30)

valid for nonintegral values of x, where

$$\psi(x) = \frac{\Gamma'}{\Gamma}(x) \tag{31}$$

is the Euler digamma function and γ indicates the Euler constant

$$\gamma = -\psi(1). \tag{32}$$

Proof. The derivative of (19) at s = 0 amounts to

$$2\pi i \zeta'(0, x) = \left(\log 2\pi - \Gamma'(1)\right) \left(\ell_1(x) - \ell_1(1-x)\right) + \frac{\pi}{2} i \left(\ell_1(x) + \ell_1(1-x)\right) - \left(\ell'_1(x) - \ell'_1(1-x)\right)$$
(33)

on correcting (see Ref. 6, (4.38)). Substituting (27), (26) and

$$\ell_1'(x) = -\sum_{n=1}^{\infty} \frac{\log n}{n} e^{2\pi i n x}$$
(34)

in (33) and invoking (23), we deduce (21) and (22). Adding (33) with x replaced by 1-x to (33), we find the Fourier series for the Clausen function $A_1(x)$ in (26) with $\varkappa = 1$.

(19) with s = 0 leads to the $\varkappa = 1$ case of (27) in view of (28). The even part (15) leads only to the parity relation $B_1(x) + B_1(1-x) = 0$. These establish (10), the $\varkappa = 1$ case. Higher order case (25) follows by integration, implying (26) and (27).

On the other hand, regarding the limiting case as $s \to 1$ of (16), we compare the well-known formulas

$$\ell_0(x) = \frac{e^{2\pi i x}}{1 - e^{2\pi i x}} = \frac{1}{2} \left(-1 + i \cot \pi x \right) \tag{35}$$

for $x \in \mathbb{R} - \mathbb{Z}$ and the Laurent expansion

$$\zeta(s,x) = \frac{1}{s-1} - \psi(x) + O(s-1) \quad \text{as} \quad s \to 1$$
 (36)

to conclude (30).

Higher order case (26) follows by integration.

The Gaussian representation

$$\psi(x) = -\gamma - \frac{1}{x} + \sum_{n=1}^{\infty} \frac{x}{n(n+x)}$$
 (37)

and (31) give rise to (17), thereby completing the proof of the odd part assertion in Theorem 1. By (32), the Laurent expansion (36) reduces to

$$\zeta(s) = \frac{1}{s-1} + \gamma + O(s-1) \quad \text{as} \quad s \to 1. \tag{38}$$

Remark 1. Milnor (see Ref. 15, Appendix 1) incorporates the parity relation in (6) and Theorem 1 is its ultimate form.

(19) is valid for all x in the form

$$\zeta(s, 1 - \{x\}) = \frac{1}{\pi} (2\pi)^s \Gamma(1 - s)$$
$$\times \left(\sin\left(\frac{\pi s}{2}\right) \ell_{1-s}^e(x) - i \cos\left(\frac{\pi s}{2}\right) \ell_{1-s}^o(x) \right), \quad (39)$$

where

$$\{x\} = x - [x] = \bar{B}_1(x) + \frac{1}{2}$$
(40)

is the fractional part of x, [x] being the integral part of x defined below (11), cf. the passage after (61).

We note that we need the full force (19) or (5) to derive Kummer's Fourier series. With (the derivative of) the even part (15) alone, we can obtain only the Fourier series for the Clausen function $A_1(x)$ in (26), not Kummer's Fourier series. It can be shown that Kummer's Fourier series is, in the long run, equivalent to the functional equation for the Riemann zeta function (see Refs. 16, p. 108 and 17, pp. 168–175).

Under (37), all equations (9), (17), (30), (35) are equivalent:

$$\psi(x) - \psi(1 - x) = 2\pi i \ell_0(x) + \pi i = 2\pi i \frac{e^{2\pi i x}}{1 - e^{2\pi i x}} + \pi i$$
$$= -\pi \cot \pi x = -\frac{1}{x} - 2x \sum_{n=1}^{\infty} \frac{1}{x^2 - n^2}, \quad 0 < x < 1.$$

Equation (27), resp. (30), being the case s = 0 resp. (the limiting case as) $s \to 1$ of (19) are connected by the generalized (inverse) Eisenstein formula

$$\sum_{a=1}^{M-1} e^{-2\pi i \frac{ab}{M}} \bar{B}_1\left(\frac{a}{M}\right) = -\frac{1}{2i} \cot \frac{\pi b}{M},\tag{41}$$

cf. Corollary 2.

We briefly discuss Ogg's result,¹⁸ stated in Lemma 1, which refers to the RHB correspondence with the theta-transformation formula and establishes the case of the even part of Theorem 1. We return to Ogg's main result, RHB proof of the Jacobi triple product in Section 3.

Let $\vartheta(\tau, x)$ denote $\vartheta_3(\tau, x)$, one of 4 Jacobi theta functions, cf., e.g. Refs. 19, 20:

$$\vartheta(\tau, x) = \sum_{n=-\infty}^{\infty} e^{2\pi i n x + \pi i \tau n^2} = \sum_{n=-\infty}^{\infty} z^n q^{n^2}$$
$$= 1 + \sum_{n=1}^{\infty} 2\cos(2\pi n x) q^{n^2}, \qquad (42)$$

where $z = e^{2\pi i x}$, $q = e^{\pi i \tau}$ and $\tau \in \mathcal{H}$, \mathcal{H} indicating the upper half-plane. $\vartheta(it, 0) = \sum_{n=-\infty}^{\infty} e^{-\pi t n^2}$ is the theta function used by Riemann to deduce the functional equation for $\zeta(s)$, cf. Remark 2. It is clear that $\vartheta(\tau, x)$ is a doubly periodic function in τ and x with periods 2 and 1, respectively.

Its characteristic property is the theta-transformation formula

$$e^{\pi i \tau x^2} \vartheta(\tau, \tau x) = \sqrt{\frac{i}{\tau}} \vartheta\left(-\frac{1}{\tau}, x\right), \tag{43}$$

which we may express as

$$f_j(\tau, x) = \sqrt{\frac{i}{\tau}} f_k\left(-\frac{1}{\tau}, x\right), \quad \{j, k\} = \{1, 2\},$$

on putting

$$f_1(\tau, x) = e^{\pi i \tau x^2} \vartheta(\tau, \tau x), \quad f_2(\tau, x) = \vartheta(\tau, x).$$

We use the case of $\tau = it, t > 0$

$$f_j(it,x) = \sqrt{\frac{1}{t}} f_k\left(\frac{i}{t},x\right), \quad \{j,k\} = \{1,2\}.$$
 (44)

Let

$$f_j(\infty) = \lim_{\mathrm{Im}\,\tau\to\infty} f_j(\tau, x), \quad j = 1, 2.$$

Then

$$f_1(\infty) = \begin{cases} 1 & x \in \mathbb{Z}, \\ 0 & x \notin \mathbb{Z}, \end{cases} \quad f_2(\infty) = 1.$$

For $x \in \mathbb{R}$, consider the Mellin transform

$$\Phi_j(2s,x) = \int_0^\infty t^s (f_j(it,x) - f_j(\infty)) \,\frac{\mathrm{d}t}{t}, \quad j = 1, 2, \qquad (45)$$

where $\sigma > 0$.

Lemma 1 (Ogg). The functional equation

$$\Phi_j(s,x) = \Phi_k(1-s,x), \quad \{j,k\} = \{1,2\}$$
(46)

is in RHB correspondence with (43) and constitutes the even part in Theorem 1.

 $\Phi_j(s,x)$ with correction terms is EBV (entire and bounded in every vertical strip, (see Ref. 21, p. xiii)).

Proof. Note that

$$f_1(it, x) = e^{-\pi t x^2} \vartheta(it, itx) = \sum_{n = -\infty}^{\infty} e^{-\pi t (n+x)^2},$$

$$f_2(it, x) = \vartheta(it, x) = \sum_{n = -\infty}^{\infty} e^{2\pi i n x - \pi t n^2},$$
(47)

As a memorial to Hecke, we call the special case of (1) the *Hecke* gamma transform (for x > 0 and $\sigma > 0$):

$$\pi^{-s}\Gamma(s)y^{-s} = \int_0^\infty t^s e^{-\pi yt} \frac{\mathrm{d}t}{t},\tag{48}$$

which is used with $y = (n + x)^2$, resp. $y = n^2$. By this, we find that

$$\Phi_j(2s,x) = \pi^{-s} \Gamma(s) \phi_j(2s,x), \quad j = 1, 2,$$
(49)

where

$$\phi_1(s,x) = \sum_{n=-\infty, n+x\neq 0}^{\infty} |n+x|^{-s}, \quad |n+x|^{-s} = e^{-s\log|n+x|},$$

so that

$$\phi_1(s,x) = \zeta(s,\{x\}) + \zeta(s,1-\{x\}) = 2\zeta^e(s,\{x\})$$
(50)

and

$$\phi_2(s,x) = \sum_{\substack{n=-\infty, n\neq 0}}^{\infty} e^{2\pi i n x} |n|^{-s} = \ell_s(x) + \ell_s(1-x)$$
$$= \sum_{n=1}^{\infty} 2\cos(2\pi n x) n^{-s} = 2\ell_s^e(x).$$
(51)

Equation (49) leads to (46) since the functional equation in RHB correspondence with (44) is

$$\pi^{-\frac{s}{2}}\Gamma\left(\frac{s}{2}\right)\phi_j(s,x) = \pi^{-\frac{1-s}{2}}\Gamma\left(\frac{1-s}{2}\right)\phi_k(1-s,x).$$
 (52)

By the same argument as in the proof of Theorem 2, we may transform (52) into the even part of Theorem 1. \Box

Remark 2. $\phi_1(s, x)$ in (50) appears again as the lattice function in (96). This being a perturbed Dirichlet series, which are associated with the Fourier–Bessel expansion (see Ref. 13, Chapter 4), we will see two examples of the Fourier–Bessel expansion in subsequent sections.

In the above argument, the variable 2s is used instead of s which mean that one considers the sequence $\{n^2\}$ rather than $\{n\}$ to the effect that one considers the Epstein zeta function in a unary quadratic form.

We analyze Riemann's way of using the theta function (see Ref. 10, pp. 61–62). Let

$$\omega(t) = \frac{1}{2}\vartheta(it,0) - \frac{1}{2} = \sum_{n=1}^{\infty} e^{-\pi n^2 t}, \quad \text{Re}\,t > 0.$$

Then (43) reads for $\omega(t)$

$$\omega(t^{-1}) = -\frac{1}{2} + \frac{1}{2}\sqrt{t} + \sqrt{t}\omega(t).$$
(53)

Implicit in (45) and (51) with 2s replaced by s is

$$\pi^{-\frac{s}{2}}\Gamma\left(\frac{s}{2}\right)\zeta(s) = \int_0^\infty t^{\frac{s}{2}}\omega(t)\,\frac{\mathrm{d}t}{t}.\tag{54}$$

Riemann divides the integral in (54) into two parts, \int_0^1 and \int_1^∞ , makes the change of variable $t \leftrightarrow \frac{1}{t}$ and applies (53) to obtain a symmetric form. Thus, Riemann's method gives rise to the Ewald expansion — expansion with incomplete gamma function coefficients. It is developed by Lavrik and later by Terras, cf. Ref. 13, Chapter 5. The use of the full series amounts to Theorem 2.

Proof of the case of the odd part is given in Theorem 1.

It is natural to introduce the *conjugate* theta function as the theta function-like (modular) function corresponding to (13). This was known to de la Vallée Poussin, cf. Remark 3.

$$\tilde{\vartheta}(\tau, x) = \sum_{n=-\infty}^{\infty} n e^{2\pi i n x + \pi i \tau n^2} = \sum_{n=-\infty}^{\infty} n z^n q^{n^2} = z \frac{\partial}{\partial z} \vartheta(\tau, x)$$
$$= \sum_{n=1}^{\infty} 2i n \sin(2\pi n x) q^{n^2}.$$
(55)

(18) necessitates the validity of

$$\Gamma\left(\frac{1+s}{2}\right)\varphi(s) = \Gamma\left(1-\frac{s}{2}\right)\psi_2(s).$$
(56)

It follows from RHB correspondence (i.e. from the Mellin inversion (1)) that for (56) to hold, it is necessary and sufficient that the transformation formula is to be of the form

$$e^{\pi i \tau x^2} \tilde{\vartheta}(\tau, \tau x) = \sqrt{\frac{i}{\tau^3}} \tilde{\vartheta}\left(-\frac{1}{\tau}, x\right).$$
(57)

Summarizing, we have

Theorem 3. The conjugate theta function (55) satisfies the transformation formula (57) which is in RHB correspondence with the functional equation (56). It is odd w.r.t. x and $\tilde{\vartheta}(\tau, 0) = 0$. We quote results of Ref. 22 which are basis of the fact for the odd part in Theorem 1.

Lemma 2.

- (i) The partial fraction expansion for the cotangent function and the functional equation (3) for the Riemann zeta function are equivalent.
- (ii) The infinite product expansion for the sine function and the partial fraction expansion for the cotangent function are equivalent.

1.1. Periodic Dirichlet series

In connection with Remark 1, we summarize the results on periodic Dirichlet series as developed by Refs. 7, 8, 11, 23, etc.

The theory of DFT for arithmetic functions has been developed in Ref. 24 in the case of periodic functions. Cf. also Refs. 16 (Section 8.1) and 25 (Sections 4.1 and 4.3). Let C(M) be the vector space of all periodic arithmetic functions f with period M. As in (14), we let

$$f^{e} = \frac{1}{2} (f(n \mod M) + f(-n \mod M)),$$

$$f^{o} = \frac{1}{2} (f(n \mod M) - f(-n \mod M))$$

be the even, resp. odd, part of $f: f = f_{\text{even}} + f_{\text{odd}}$. Let

$$\varepsilon_j(a) = e^{2\pi i j a/M}, \quad 1 \le j \le M,$$

where a is an integer variable. Then the set $\{\varepsilon_j | 1 \leq j \leq M\}$ forms a basis of C(M). We define the DFT \hat{f} (or the bth Fourier coefficient) of $f \in C(M)$ by

$$\hat{f}(b) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M} \varepsilon_b(-a) f(a).$$

The parity inherits to the DFT and the Fourier inversion or Fourier expansion formula holds true:

$$f(a) = \frac{1}{\sqrt{M}} \sum_{b=1}^{M} \hat{f}(b)\varepsilon_b(a) = \hat{f}(-a).$$
(58)

Note that (58) is the expression of f with respect to the basis $\{\varepsilon_j\}$.

$$\hat{f}^{e}(b) = \frac{1}{2}(\hat{f}(b) + \hat{f}(-b)) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M} \cos\left(2\pi \frac{b}{M}a\right) f(a),$$
$$\hat{f}^{o}(b) = \frac{1}{2}(\hat{f}(b) - \hat{f}(-b)) = -i\frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \sin\left(2\pi \frac{b}{M}a\right) f(a).$$

To find another natural basis, let χ_a be the characteristic function $\chi_a \mod M$ (see Ref. 25, p. 73).

$$\chi_a(n) = \begin{cases} 1 & n \equiv a \mod M, \\ 0 & n \not\equiv a \mod M. \end{cases}$$

Then $\{\chi_a | 1 \le a \le M\}$ is a basis of C(M) and

$$\sqrt{M}\hat{\chi}_a(n) = \sum_{j=1}^M \varepsilon_n(-j)\chi_a(j) = \varepsilon_a(-n).$$

For $f \in C(M)$, let

$$D(s,f) = \sum_{n=1}^{\infty} \frac{f(n)}{n^s}.$$
 (59)

Since

$$\sum_{n=1}^{\infty} \frac{|f(n)|}{n^{\sigma}} \ll \zeta(\sigma),$$

the series in (59) is absolutely convergent for $\sigma > 1$. Let D(M) denote the set of all Dirichlet series of the form (59). Then it forms a vector space of dimension M canonically isomorphic to C(M). One of the bases of D(M) is $\{\ell_s\left(\frac{a}{M}\right) | 1 \le a \le M\}$, where $\ell_s(x)$ is the Lerch zeta function in (4). Hence, we have

$$D(s,f) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M} \hat{f}(a)\ell_s\left(\frac{a}{M}\right)$$
$$= \frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{f}(a)\ell_s\left(\frac{a}{M}\right) + \frac{\hat{f}(M)}{\sqrt{M}}\zeta(s).$$
(60)

It follows that D(s, f) can be continued meromorphically over the whole plane and that it is entire if and only if $\hat{f}(M) = 0$ which is

defined by

$$\hat{f}(M) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M} f(a).$$

Another basis of D(M) is $\{D(s, \chi_a) | 1 \le a \le M\}$, where

$$D(s,\chi_a) = \sum_{n=1}^{\infty} \frac{\chi_a(n)}{n^s} = \sum_{\substack{n=1\\n \equiv a \bmod M}}^{\infty} \frac{1}{n^s} = \zeta(s,a,M) = M^{-s}\zeta\left(s,\frac{a}{M}\right),$$
(61)

where $\zeta(s, a, M)$ indicates the partial zeta function. Note that it is $\zeta(s, 1 - \{\frac{a}{M}\})$ that belongs to D(M) rather than $\zeta(s, \{\frac{a}{M}\})$, cf. Remark 1. Hence, parallel to (60), we have another expression

$$D(s,f) = \sum_{a=1}^{M} f(a)\zeta(s,a,M) = \frac{1}{M^s} \sum_{a=1}^{M} f(a)\zeta\left(s,\frac{a}{M}\right).$$

Proposition 1.

$$\frac{1}{M^s} \sum_{a=1}^M f(a)\zeta\left(s, \frac{a}{M}\right) = \sum_{a=1}^M f(a)\zeta\left(s, a, M\right) = D(s, f)$$
$$= \frac{1}{\sqrt{M}} \sum_{a=1}^M \hat{f}(a)\ell_s\left(\frac{a}{M}\right)$$
$$= \frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{f}(a)\ell_s\left(\frac{a}{M}\right) + \frac{\hat{f}(M)}{\sqrt{M}}\zeta(s), \quad (62)$$

which entails

$$-\frac{1}{M}\sum_{k=1}^{M}f(k)\psi\left(\frac{k}{M}\right) = D(1,f) = \frac{1}{\sqrt{M}}\sum_{k=1}^{M-1}\hat{f}(k)\ell_1\left(\frac{k}{M}\right) + \frac{\hat{f}(M)}{\sqrt{M}}\gamma$$
(63)

as well as

$$D(s, f^{o}) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{f}^{o}(a) \ell_{s}\left(\frac{a}{M}\right).$$
(64)

70

Proof. Equation (63) is the equality between the Laurent constants of both sides of (62), where (36) and (38) are used. (64) follows since $\hat{f}^o(M) = 0$.

Corollary 2. (i) The generalized (inverse) Eisenstein formula (41) is the odd part of (63) with $f(k) = \chi_b(k)$. (ii) We have

$$D'(1, f^{o}) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{f}(a) \left(\ell_{1}'\right)^{o} \left(\frac{a}{M}\right)$$
$$= -\frac{i}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{f}(a) \sum_{n=1}^{\infty} \frac{\log n}{n} \sin\left(2\pi \frac{a}{M}n\right)$$
(65)

which leads, under the assumption of the Lerch formula (23) and the functional equation (6), to

$$D'(1, f^{o}) = -\frac{i}{\sqrt{M}} \frac{\pi}{2} \sum_{a=1}^{M-1} \hat{f}(a) \left(\log \Gamma\left(\frac{a}{M}\right) - \log \Gamma\left(1 - \frac{a}{M}\right) \right)$$
$$-\frac{i}{\sqrt{M}} \pi(\gamma + \log 2\pi) \sum_{a=1}^{M-1} \hat{f}(a) \bar{B}_{1}\left(\frac{a}{M}\right). \tag{66}$$

Proof. The odd part of (63) reads

$$-\frac{1}{M}\sum_{k=1}^{M}f(k)\left(\psi\left(\frac{k}{M}\right)-\psi\left(1-\frac{k}{M}\right)\right)$$
$$=\frac{1}{\sqrt{M}}\sum_{k=1}^{M-1}\hat{f}(k)\left(\ell_{1}\left(\frac{k}{M}\right)-\ell_{1}\left(-\frac{k}{M}\right)\right).$$

By the property of $\{\kappa_a\}$,

$$-\frac{1}{M}\left(\psi\left(\frac{b}{M}\right)-\psi\left(1-\frac{b}{M}\right)\right)$$
$$=\frac{1}{M}\sum_{k=1}^{M-1}\varepsilon_b(-k)\left(\ell_1\left(\frac{k}{M}\right)-\ell_1\left(-\frac{k}{M}\right)\right).$$

Substituting (27) with $\varkappa = 1$ and (30), we conclude (41).

H. Li et al.

The equation (65) follows by differentiating (62), whereby we appeal to (34). For (66), the Hurwitz formula (6) and the Kummer Fourier series (24) are used. \Box

Corollary 3. Let χ be a nonprincipal Dirichlet character mod M. Then

$$\frac{1}{M^s} \sum_{a=1}^{M-1} \chi(a) \zeta\left(s, \frac{a}{M}\right) = L(s, \chi) = \frac{1}{\sqrt{M}} \sum_{a=1}^{M-1} \hat{\chi}(a) \ell_s\left(\frac{a}{M}\right).$$

Corresponding to Theorem 2, Schnee's theorem as modified by Ishibashi and Kanemitsu reads as follows.

Theorem 4 (7, 8). Suppose f is a periodic arithmetic function with period M. Then

$$D(1-s,f) = \left(\frac{\pi}{M}\right)^{\frac{1}{2}-s} \left(\frac{\Gamma\left(\frac{s}{2}\right)}{\Gamma\left(\frac{1-s}{2}\right)} D(s,\hat{f}^e) + i\frac{\Gamma\left(\frac{1+s}{2}\right)}{\Gamma\left(1-\frac{s}{2}\right)} D(s,\hat{f}^o)\right)$$
(67)

or

$$\Gamma\left(\frac{s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\left(\frac{\pi}{M}\right)^{-\frac{s}{2}}D(s,f)$$

$$=\Gamma\left(\frac{1-s}{2}\right)\Gamma\left(\frac{1+s}{2}\right)\left(\frac{\pi}{M}\right)^{-\frac{1-s}{2}}D(1-s,\hat{f}^{e})$$

$$+i\Gamma\left(\frac{s}{2}\right)\Gamma\left(1-\frac{s}{2}\right)\left(\frac{\pi}{M}\right)^{-\frac{1-s}{2}}D(1-s,\hat{f}^{o}).$$
(68)

This is equivalent to the generalized Euler identity

$$\begin{split} \tilde{D}(1-s,f) &= \sqrt{\pi}^{-1} 2^{1-s} \Gamma(s) \\ &\times \left(\cos\left(\frac{\pi s}{2}\right) \tilde{D}(s,\hat{f}^e) + i \sin\left(\frac{\pi s}{2}\right) \tilde{D}(s,\hat{f}^o) \right), \end{split}$$

where

$$\tilde{D}(s,f) = \left(\frac{\pi}{M}\right)^{-\frac{s}{2}} D(s,f).$$

Proof. Equation (67) amounts to (68), i.e. the ramified functional equation (18), on clearing the denominators and multiplying by $\left(\frac{\pi}{M}\right)^{\frac{s-1}{2}}$.

Definition 2. Define the general Gauss sum

$$\tau(\chi,n) = \sum_{a=1}^{M} \chi(a) e^{\frac{2\pi i n a}{M}} = \sqrt{M} \hat{\chi}(-n)$$
(69)

and the (normalized) Gauss sum

$$\tau(\chi) = \tau(\chi, 1) = \sum_{a=1}^{M} \chi(a) e^{\frac{2\pi i a}{M}},$$

recall Ref. 26, Theorem 8.19, p. 171:

Lemma 3. (i) Suppose χ is a nonprincipal character mod M. Then the general Gauss sum (69) is separable, i.e.

$$\tau(\chi, n) = \bar{\chi}(n)\tau(\chi) \tag{70}$$

if and only if χ is a primitive character mod M.
(ii) For a Kronecker character, which is real primitive,

$$\tau(\chi_d) = \begin{cases} i\sqrt{|d|} & d < 0, \\ \sqrt{d} & d > 0. \end{cases}$$
(71)

It follows from (69) and (70) that for a primitive χ ,

$$\hat{\chi}(n) = \frac{\tau(\chi)}{\sqrt{M}} \chi(-1)\bar{\chi}(n)$$
(72)

and

$$\chi(-1)\tau(\chi)\tau(\bar{\chi}) = |\tau(\chi)|^2 = M.$$

Equation (65) reads for a primitive odd character χ :

$$L'(1,\chi) = -\frac{\tau(\chi)}{M} \sum_{a=1}^{M-1} \bar{\chi}(a) \left(\ell_1'\right)^o \left(\frac{a}{M}\right)$$

and under the Lerch formula (23),

$$L'(1,\chi) = \pi i \frac{\tau(\chi)}{M} \left(\sum_{a=1}^{M-1} \bar{\chi}(a) \log \Gamma\left(\frac{a}{M}\right) + (\gamma + \log 2\pi) B_{1,\bar{\chi}} \right),$$
(73)

where

$$B_{1,\bar{\chi}} = \sum_{a=1}^{M-1} \bar{\chi}(a)\bar{B}_1\left(\frac{a}{M}\right)$$

is the first generalized Bernoulli number.

Corollary 3 reads for a primitive odd character χ :

$$L(1,\chi) = \pi i \frac{\tau(\chi)}{M} B_{1,\bar{\chi}}$$
(74)

by (27).

Substituting (74) in (73), we deduce the following.

Lemma 4. For a primitive odd character χ , we have

$$\gamma L(1,\chi) + L'(1,\chi) = -\frac{\tau(\chi)}{M} \sum_{a=1}^{M-1} \bar{\chi}(a) (\ell_1')^o \left(\frac{a}{M}\right) + \gamma L(1,\chi),$$

which amounts to

$$\gamma L(1,\chi_d) + L'(1,\chi_d) = -\frac{i}{\sqrt{|d|}} \sum_{a=1}^{|d|-1} \bar{\chi}(a) \left(\ell_1'\right)^o \left(\frac{a}{|d|}\right) + \gamma L(1,\chi_d)$$
(75)

for the Kronecker character with d < 0 in view of (71). Under the Lerch formula (23), these read

$$\gamma L(1,\chi) + L'(1,\chi) = \pi i \frac{\tau(\chi)}{M} \sum_{a=1}^{M-1} \bar{\chi}(a) \log \Gamma\left(\frac{a}{M}\right) + (2\gamma + \log 2\pi)L(1,\chi)$$

and

$$\gamma L(1, \chi_d) + L'(1, \chi_d) = -\frac{\pi}{\sqrt{|d|}} \sum_{a=1}^{|d|-1} \bar{\chi}(a) \log \Gamma\left(\frac{a}{|d|}\right) + (2\gamma + \log 2\pi) L(1, \chi_d).$$
(76)

74

Corresponding to Theorem 2, Theorem 4 in the special case reads as follows.

Theorem 5. For a primitive character χ , the functional equation (67) reads

$$\left(\frac{\pi}{M}\right)^{-\frac{1-s}{2}} L(1-s,\chi) = \frac{\tau(\chi)}{\sqrt{M}} \left(\frac{\pi}{M}\right)^{-\frac{s}{2}}$$
$$\left(\frac{\Gamma\left(\frac{s}{2}\right)}{\Gamma\left(\frac{1-s}{2}\right)} L^e(s,\bar{\chi}) - \frac{\Gamma\left(\frac{1+s}{2}\right)}{\Gamma\left(1-\frac{s}{2}\right)} i L^o(s,\bar{\chi}) \right)$$

or

$$\left(\frac{\pi}{M}\right)^{-\frac{1-s}{2}}L(1-s,\chi) = \frac{\tau(\chi)}{\sqrt{M}} \left(\frac{\pi}{M}\right)^{-\frac{s}{2}} \left(\frac{\Gamma\left(\frac{s}{2}\right)}{\Gamma\left(\frac{1-s}{2}\right)} \frac{1+\chi(-1)}{2}L(s,\bar{\chi}) - \frac{\Gamma\left(\frac{1+s}{2}\right)}{\Gamma\left(1-\frac{s}{2}\right)} \frac{1-\chi(-1)}{2}iL(s,\bar{\chi})\right).$$
(77)

In the form of a generalized Euler identity,

$$L(s,\chi) = \frac{\tau(\chi)}{\pi} \left(\frac{2\pi}{M}\right)^s \Gamma(1-s)$$

$$\times \left(\sin\left(\frac{\pi s}{2}\right) L^e(1-s,\bar{\chi}) - i\cos\left(\frac{\pi s}{2}\right) L^o(1-s,\bar{\chi})\right)$$

$$= \frac{\tau(\chi)}{\pi} \left(\frac{2\pi}{M}\right)^s \Gamma(1-s)$$

$$\times \left(\sin\left(\frac{\pi s}{2}\right) \frac{1+\chi(-1)}{2} L(1-s,\bar{\chi}) - i\cos\left(\frac{\pi s}{2}\right) \frac{1-\chi(-1)}{2} L(1-s,\bar{\chi})\right)$$

$$= \frac{\tau(\chi)}{\pi} \left(\frac{2\pi}{M}\right)^s \Gamma(1-s)$$

$$\times \left(\sin\left(\frac{\pi s}{2}\right) (1-\mathfrak{a}(\chi))L(1-s,\bar{\chi}) - i\cos\left(\frac{\pi s}{2}\right) \mathfrak{a}(\chi)L(1-s,\bar{\chi})\right),$$
(78)

where $\mathfrak{a}(\chi)$ is the parity symbol

$$\mathfrak{a} = \mathfrak{a}(\chi) = \frac{1 - \chi(-1)}{2} = \begin{cases} 0 & \chi(-1) = 1, \\ 1 & \chi(-1) = -1. \end{cases}$$

This is the genesis of the functional equation for the Dirichlet Lfunction, as described by de La Vallée Poussin (as in Davenport).

Proof. The equation (77) is a direct application of Theorem 4 together with (72), the expression of the DFT of χ in view of separability of the general Gauss sum and (78) follows from (77) as in the proof of Theorem 2.

Then the common expression for the functional equation for the Dirichlet L-function

$$\xi(s,\chi) := \left(\frac{\pi}{M}\right)^{-\frac{s}{2}} \Gamma\left(\frac{s+\mathfrak{a}}{2}\right) L(s,\chi) = -\frac{i^{\mathfrak{a}}\tau(\chi)}{\sqrt{M}} \xi(1-s,\bar{\chi})$$

can be read out from (77).

It remains to specify the theta-like functions and their transformation formulas which are in RHB correspondence with $D(s, f^e)$, resp. $D(s, f^o)$, as in Theorem 1:

$$\psi(t, f^e) = \vartheta(it, 0, f^e) = \sum_{n = -\infty}^{\infty} f^e(n) e^{-\frac{\pi}{M} t n^2},$$

$$\tilde{\psi}(t, f^o) = \tilde{\vartheta}(it, 0, f^o) = \sum_{n = -\infty}^{\infty} n f^o(n) e^{-\frac{\pi}{M} t n^2},$$
(79)

which satisfies the transformation formula

$$\sqrt{\frac{1}{t}}\psi\left(\frac{1}{t},f^e\right) = \psi(t,f^e), \quad \sqrt{\frac{1}{t^3}}\psi\left(\frac{1}{t},f^o\right) = \tilde{\psi}(t,f^o)$$

These follow from the Mellin inversion (1), i.e. RHB correspondence. $\hfill \Box$

Remark 3. Theorem 5 is a rather special case of Theorem 4 for a primitive χ . It is possible to state more general results for χ not necessarily primitive, cf. Refs. 27 and 28. In the former, separability of Gauss sum in Lemma 3 is deduced from the functional equation. This general case will be studied elsewhere.

In the case of Dirichlet *L*-functions, (79) has been used by de la Vallée Poussin (see Ref. 29, p. 302) cf., e.g. Ref. 10, pp. 68–71, and similarly, the conjugate theta function (55).

Since our main concern lies in the odd part, only the Lerch zeta expression of Proposition 1 is used and Corollary 3 is to contain the unused part, i.e. (63) is to read for a nonprincipal character χ

$$-\frac{1}{M}\sum_{k=1}^{M}\chi(k)\psi\left(\frac{k}{M}\right) = L(1,\chi),$$

which is in Ref. 16, (8.28), p. 173, and is a basis of the main result of Ref. 30, (also see Ref. 16, Theorem 8.2, pp. 174–175) to the effect that Gauss first formula for ψ is equivalent to finite expressions for $L(1, \chi)$. We return to this elsewhere.

2. Genesis of the Lerch–Chowla–Selberg Formula

In this section, we shall pursue the genesis of the Lerch–Chowla–Selberg formula from the point of view of RHB correspondence. This is a relation between the theta function and the gamma function at rational argument. The latter is already prepared in the preceding sections and the main part is the expression of the constant term of the Laurent expansion of the Dedekind zeta function at s = 1 of a quadratic field. It is well known that the Dedekind zeta function may be studied through that of the Epstein zeta function or the Eisenstein series.

We consider $\vartheta(\tau, x) = \vartheta(\tau, z)$ of (42) with z = 1. Let $q = e^{\pi i \tau}$, $\tau \in \mathcal{H}$:

$$\theta(\tau) = \vartheta(\tau, 0) = \sum_{n = -\infty}^{\infty} q^{n^2} = 1 + \sum_{n = 1}^{\infty} 2q^{n^2}$$
$$= \prod_{n = 1}^{\infty} (1 - q^{2n}) (1 + q^{2n-1})^2,$$
(80)

the last equality because of the Jacobi triple product (115). The Dedekind eta function $\eta(\tau)$ is essentially the factor of theta

defined by

$$\eta(\tau) = q^{\frac{1}{12}} \prod_{n=1}^{\infty} (1 - q^{2n}) = q^{\frac{1}{12}} \theta^e(\tau)$$

or

$$\log \eta(\tau) = \frac{1}{12} \log q + \sum_{n=1}^{\infty} \log(1 - q^{2n}) = \frac{1}{12} \pi i \tau - \sum_{n=1}^{\infty} \sigma_{-1}(n) q^{2n}.$$

Reference 13 (pp. 12–15) streamlines the historical facts about the Lerch–Chowla–Selberg formula partially following Ref. 31 with reference to Refs. 32 and 43. The main interest lies in the fact that there is an equality of two irrelevant-looking special functions, gamma and theta, at rational arguments. Here, we shall show a plausible genesis of the formula from the viewpoint of RHB correspondence.

It is a combination of two remarkable formulas: Kronecker limit formula and Lerch formula. For the former, there is enormous amount of literature see Refs. 33–37, 53, and references therein. For the latter, there is a detailed account in Ref. 38, especially the references on pp. 67–68 are rather thorough. However, there is no mention of the first statement of the Lerch–Chowla–Selberg formula (see Ref. 39, (26), p. 303) nor earlier works by Berger *et al.*⁴⁰ Both Refs. 31 and 38 stress Lerch's formula (23) without resorting to its defining property as the principal solution to the difference equation

$$f(x+1) - f(x) = \log^{\alpha} x, \quad x > 0$$
 (81)

for each $\alpha \in \mathbb{N}$, cf. Refs. 41 and 42, 61 etc. To obtain the Chowla-Selberg formula for a real quadratic field, the evaluation of the cosine Fourier series is needed (see Ref. 43, (165), p. 180):

$$\Phi(x) = \sum_{n=1}^{\infty} \frac{\log n}{n} \cos 2\pi x n \tag{82}$$

as opposed to the Kummer Fourier series. It had been studied by Lerch⁴⁴ before Landau. It was taken up by Gut^{45} to derive the Kronecker limit formula for cyclotomic fields. It turns out that the function which Gut introduced is -(R(x)+R(1-x)), where $R(x) = R_2(s)$ is the Deninger *R*-function as the principal solution to (82) with

Quadr. field	Imaginary	Real
Fourier ser.	$\log \Gamma(x)$	$-\Phi(x)$
Hurwitz zeta	$\zeta'(0,x)$	R(x) + R(1-x)
Lerch zeta	$-\frac{i}{2}(\ell_1'(x) - \ell_1'(1-x))$	$-\frac{1}{2}(\ell_1'(x) + \ell_1'(1-x))$

Table 1. Fourier sine, cosine series and their parity.

 $\alpha = 2$. Deninger's standpoint is that the *R*-function occupies its space as a proper special function in the same way as $\log \Gamma(x)$ is the principal solution to (82) with $\alpha = 1$ (Bohr–Mollerup theorem). Then from the uniqueness of the principal solution, it follows that

$$R(x) = -\zeta''(0, x),$$

where the prime means the derivative w.r.t. s. As Table 1 shows, the parity of imaginary and real quadratic field and the associated Fourier series are consistent.

Here, $\log \Gamma(x)$ for the Fourier sine series is up to some extra terms. Higher order principal solutions $R_{\alpha}(x)$ to (81) stimulated the study on higher derivatives of *L*-functions.¹⁷

2.1. Chowla–Selberg integral formula

In this section, we deduce the Dirichlet class number formula and the Kronecker limit formula for an imaginary quadratic field $K = \mathbb{Q}(\sqrt{D})$ with discriminant d < 0. Let

$$\zeta_K(s) = \sum_{0 \neq \mathfrak{a} \in A} (N\mathfrak{a})^{-s}, \ \sigma > 1,$$

be the Dedekind zeta function of K, where $N\mathfrak{a} = [\mathfrak{O} : \mathfrak{a}]$ is the norm of the ideal \mathfrak{a} and where $\mathfrak{O} = \mathfrak{O}_K$ is the ring of algebraic integers in K. Then it amounts to finding the residue and the constant term of the Laurent expansion of $\zeta_K(s)$ at s = 1. The method is classical and is due to Dirichlet, resp. Kronecker, cf., e.g. Ref. 10, resp. Ref. 35.

Let I/P denote the absolute ideal class group, where I and P signify the group of all fractional ideals and the principal fractional ideals of K, respectively. The order |I/P| is finite and is called the

class number of K, denoted h. For an ideal class $A \in I/P$, let

$$\zeta(s,A) = \zeta_K(s,A) = \sum_{0 \neq \mathfrak{a} \in A} (N\mathfrak{a})^{-s}, \ \sigma > 1$$
(83)

be the class zeta function. Then we have the decomposition

$$\zeta_K(s) = \sum_{A \in I/P} \zeta(s, A).$$

Therefore, it suffices to consider the class zeta function and the approach for studying it (for a real or an imaginary quadratic field) remains the same, as described, e.g. in Ref. 37. Let \mathfrak{O}_K^{\times} denote the unit group of K, which is the finite group of roots of unity of order w for an imaginary quadratic field.

Now, fix $\mathfrak{b} \in A^{-1}$. Then for $\mathfrak{a} \in A$ in (83), $\mathfrak{a}\mathfrak{b} = (\lambda) \in P$, where $\lambda \in \mathfrak{b}$. Hence, $(N\mathfrak{a})^{-1} = N\mathfrak{b}(N\mathfrak{a}\mathfrak{b})^{-1} = N\mathfrak{b}|N(\lambda)|^{-1}$. Hence,

$$\zeta(s,A) = \sum_{\lambda \in \mathfrak{b}/\mathfrak{O}_K^{\times}} \frac{1}{\left(\frac{|N(\lambda)|}{N\mathfrak{b}}\right)^s},\tag{84}$$

where the prime on the summation sign means that the value 0 is to be omitted.

Then (84) reads

$$\zeta(s,A) = \frac{1}{w} \sum_{\lambda \in \mathfrak{b}}' \frac{1}{\left(\frac{N(\lambda)}{N\mathfrak{b}}\right)^s} = \frac{1}{w} \zeta_Q(s), \tag{85}$$

say, the absolute value sign being suppressed in view of $N\lambda = N(\lambda) = |\lambda|^2 > 0$. Here, $Q = Q(m, n) = \frac{|N(\lambda)|}{N\mathfrak{b}}$ in (84) is a positive definite quadratic form with discriminant d < 0, i.e. through (85), we view the class zeta function as the Epstein zeta function with a positive definite quadratic form of discriminant d < 0 in the setting as in Refs. 46, 47, 53, etc. to state the general Chowla–Selberg integral formula, Theorem 6:

$$Q(m,n) = am^{2} + bmn + cn^{2} = a|m + n\tau|^{2},$$

$$\tau = x + iy = \frac{b + i\sqrt{|d|}}{2a},$$
(86)

where

$$a = \frac{\sqrt{|d|}}{2\operatorname{Im}\tau},\tag{87}$$

so that

$$c = a|\tau|^2, \quad b = 2a \operatorname{Re} \tau, \quad x = \frac{b}{2a}, \quad y = \frac{\sqrt{|d|}}{2a}.$$
 (88)

Then in conformity with (86), τ is the solution of the quadratic equation with imaginary part positive:

$$a\tau^2 - b\tau + c = 0. \tag{89}$$

Hence, we express (85) as

$$\zeta(s,A) = \frac{1}{w}Z(s), \quad Z(s) = Z(s,Q) = \sum_{m,n}' \frac{1}{Q(m,n)^s} = a^{-s}\zeta_{\mathbb{Z}^2}(s),$$

$$\zeta_{\mathbb{Z}^2}(s) = \zeta_{\mathbb{Z}^2}(s,\tau) = \sum_{m,n}' \frac{1}{|m+n\tau|^{2s}},$$
(90)

where $\zeta_{\mathbb{Z}^2}(x)$ is the holomorphic Eisenstein series. It has several variations. Equation (90) will be assumed in what follows.

This setting corresponds to

$$\mathfrak{b} = \mathbb{Z} \oplus \mathbb{Z}\tau, \quad \tau \in \mathcal{H}.$$
(91)

The choice is possible since (85) is unchanged if we replace \mathfrak{b} by $\alpha \mathfrak{b}$, $K \ni \alpha \neq 0$.

The following lemma is a basis of the beta-transform, which seems to be first used by Hardy⁴⁸ (in previous literature, it used to be called as the Mellin–Barnes integral, but this refers to a much wider class of functions), cf. Ref. 49 and Remark 4.

Lemma 5. The beta-transform reads

$$(1+x)^{-s} = \frac{1}{2\pi i} \int_{(c)} \frac{\Gamma(z+s)\,\Gamma(-z)}{\Gamma(s)} \, x^z \mathrm{d}z = \frac{1}{\Gamma(s)} G_{1,1}^{1,1} \left(x^{-1} \, \middle| \, \stackrel{1}{s} \right)$$

for x > 0, $-\operatorname{Re} s < c < 0$, where (c) signifies the vertical Bromwich path s = c + it, $-\infty < t < \infty$.

Theorem 6. (i) Under the notation (86)–(90), in particular, $\tau = x + iy$, $y = \frac{\sqrt{|d|}}{2a}$, we have the Chowla–Selberg integral formula

$$\Gamma(s)Z(s) = 2\Gamma(s)\zeta(2s)a^{-s} + \frac{2^{2s}a^{s-1}\sqrt{\pi}}{|d|^{s-\frac{1}{2}}}\Gamma\left(s-\frac{1}{2}\right)\zeta(2s-1) + Q(s),$$
(92)

where

$$Q(s) = \frac{4\pi^{s} 2^{s+\frac{1}{2}}}{\sqrt{a}|d|^{\frac{s}{2}-\frac{1}{4}}} \sum_{n=1}^{\infty} n^{s-\frac{1}{2}} \sigma_{1-2s}(n) \cos 2\pi n x K_{s-\frac{1}{2}}(2\pi y n), \quad (93)$$

and where

$$\sigma_a(n) = \sum_{d|n} d^a$$

is the sum-of-divisors function.

The Equation (92) is equivalent to the functional equation

$$\left(\frac{2\pi}{\sqrt{|d|}}\right)^{-s} \Gamma(s)Z(s) = \left(\frac{2\pi}{\sqrt{|d|}}\right)^{-(1-s)} \Gamma(1-s)Z(1-s).$$

(ii) The Laurent expansion at s = 1 holds:

$$Z(s) = \frac{\frac{2\pi}{\sqrt{|d|}}}{s-1} + \frac{2\pi}{\sqrt{|d|}} \left(2\gamma + \log\frac{a}{|d|}\right) - \frac{4\pi}{\sqrt{|d|}} \log|\eta(\tau)|^2 + O(s-1).$$
(94)

Proof. (i) (92) (with (93)) has been proved in many places and elucidated as the Fourier–Bessel expansion (see Ref. 13, Chapter 4), i.e. as the correspondence $G_{1,1}^{1,1} \leftrightarrow G_{0,2}^{2,0}$. However, we reproduce a slightly modified proof of Ref. 46 to exhibit the use of the beta-transform as opposed to the Hecke gamma transform, elucidating the computations in Ref. 53.

We shall prove (92) and (93) in the form

$$Z(s,Q) = a^{-s} \zeta_{\mathbb{Z}^2}(s,\tau) = 2\zeta(2s)a^{-s} + \frac{2\sqrt{\pi}\Gamma(s-\frac{1}{2})}{\Gamma(s)}\zeta(2s-1)y^{-2s+1}a^{-s} + \frac{4a^{-s}}{\Gamma(s)}\sum_{m,n=1}^{\infty} (ny)^{-2s} \frac{\cos(2\pi mnx)}{m} I_{m,n}(s,\tau), \qquad (95)$$

where

$$I_{m,n}(s,\tau) = \frac{1}{\sqrt{\pi}} G_{0,2}^{2,0} \left((\pi mny)^2 \Big|_{s,\frac{1}{2}}^{-} \right)$$
$$= 2\pi^s (mny)^{s+\frac{1}{2}} K_{s-\frac{1}{2}} (2\pi mny).$$

We define the lattice zeta function by (50), i.e.

$$\phi_1(s, \{x\}) = \zeta_{\mathbb{Z}}(s, x) = \sum_{n = -\infty}^{\infty} |n + x|^{-s}$$
(96)

for $\{x\} > 0$, where $\{x\}$ is the fractional part of x, (40). Then ϕ_1 is the even part

$$\phi_1(s,x) = 2\zeta^e(s,\{x\}) = \zeta(s,\{x\}) + \zeta(s,1-\{x\}).$$
(97)

In (97), we use the functional equation for the Hurwitz zeta function (6): for $\sigma > 1$, $\{nx\} > 0$, deduce that

$$\phi_1(-2z, \{nx\}) = \zeta(-2z, 1 - \{nx\}) + \zeta(-2z, \{nx\})$$
$$= \frac{2\Gamma(1+2z)}{(2\pi)^{1+2z}} \sin(-\pi z) \left\{ \ell_{1+2z}(nx) + \ell_{1+2z}(1-nx) \right\},$$

i.e. the even part of $\ell_{1+2s}(nx)$, valid for $\{nx\} = 0$, also by interpreting the left-hand side by the right.

Collecting the terms with n = 0 in the definition of $\zeta_{\mathbb{Z}^2}(s,\tau)$, (90) gives $2\sum_{m=1}^{\infty} m^{-2s} = 2\zeta(2s)$. In the remaining sum $2\sum_{n=1}^{\infty}\sum_{m=-\infty}^{\infty} \{(m+nx)^2 + n^2y^2\}^{-s}$, we transform the summand

into $(ny)^{-2s} \left(1 + \left(\frac{m+nx}{ny}\right)^2\right)^{-s}$ and apply the beta transform in Lemma 5. We choose c such that $-\sigma < c < -\frac{1}{2}$ and obtain

$$\zeta_{\mathbb{Z}^2}(s,\tau) = 2\zeta(2s) + \frac{2}{\Gamma(s)} \sum_{n=1}^{\infty} (ny)^{-2s} \\ \times \frac{1}{2\pi i} \int_{(c)} \Gamma(z+s) \Gamma(-z) \phi_1(-2z, \{nx\}) (ny)^{-2z} \, \mathrm{d}z.$$
(98)

By argument similar to the one in the proof of Theorem 2, we may transform (98) into

$$\zeta_{\mathbb{Z}^2}(s,\tau) = 2\zeta(2s) + \frac{2}{\Gamma(s)} \sum_{n=1}^{\infty} (ny)^{-2s} J_n(c),$$

where

$$J_{n}(c) = J_{n}(c,s)$$

$$= \frac{1}{2\pi i} \int_{(c)} \frac{\Gamma(z+s)\Gamma(z+\frac{1}{2})}{\pi^{2z+\frac{1}{2}}}$$

$$\times \{\ell_{2z+1}(nx) + \ell_{2z+1}(1-nx)\} (ny)^{-2z} dz.$$
(99)

Moving the line of integration on the right of (99) to the right up to Re $z = c_1$, where $-\frac{1}{2} < c_1 < 0$, thereby noting the pole at $z = -\frac{1}{2}$, we deduce that

$$\zeta_{\mathbb{Z}^2}(s,\tau) = 2\zeta(2s) - \frac{2}{\Gamma(s)} \sum_{n=1}^{\infty} (ny)^{-2s+1} \frac{\Gamma(s-\frac{1}{2})}{\pi^{-\frac{1}{2}}} \{\ell_0(nx) + \ell_0(1-nx)\} + \frac{2}{\Gamma(s)} \sum_{n=1}^{\infty} (ny)^{-2s} J_n(c_1) = 2\zeta(2s) + \frac{2\sqrt{\pi}\Gamma(s-\frac{1}{2})}{\Gamma(s)} \zeta(2s-1)y^{-2s+1} + \frac{2}{\Gamma(s)} \sum_{n=1}^{\infty} (ny)^{-2s} J_n(c_1)$$
(100)

on using $\ell_0(nx) + \ell_0(1 - nx) = -1$ which follows from (35).

Now,

 $J_n(c_1)$

$$= \frac{1}{2\pi i \sqrt{\pi}} \sum_{m=1}^{\infty} \frac{2\cos(2\pi mnx)}{m} \int_{(-c_1)} \Gamma(s-z) \Gamma\left(\frac{1}{2}-z\right) (\pi mny)^{2z} dz$$
$$= \sum_{m=1}^{\infty} \frac{2\cos(2\pi mnx)}{m} I_{m,n}(s,\tau),$$
(101)

where

$$I_{m,n}(s,\tau) = \frac{1}{\sqrt{\pi}} G_{0,2}^{2,0} \left((\pi mny)^2 \Big|_{s,\frac{1}{2}}^{-} \right).$$
(102)

Hence, (100), (101) and (102) after multiplying by a^{-s} lead to (95). (ii) To deduce (94), we use (92) in the form

$$Z(s) = 2\zeta(2s)a^{-s} + \frac{2^{2s}a^{s-1}\sqrt{\pi}}{|d|^{s-\frac{1}{2}}} \frac{\Gamma\left(s-\frac{1}{2}\right)\zeta(2s-1)}{\Gamma(s)} + \frac{1}{\Gamma(s)}Q(s).$$

Near s = 1, the second term on the right of (2.1) has the expansion

$$\frac{\frac{2\pi}{\sqrt{|d|}}}{s-1} + \frac{2\pi}{\sqrt{|d|}} \left(2\gamma + \log\frac{a}{|d|}\right)$$

while the first term becomes $2\zeta(2)a^{-1} = \frac{\pi^2}{3a}$, whence it follows that

$$Z(s) = \frac{\frac{2\pi}{\sqrt{|d|}}}{s-1} + \frac{2\pi}{\sqrt{|d|}} \left(2\gamma + \log\frac{a}{|d|}\right) + \frac{\pi^2}{3a} + Q(1) + O(s-1),$$

where

$$Q(1) = \frac{8\sqrt{2}\pi}{\sqrt{a}|d|^{\frac{1}{4}}} \sum_{n=1}^{\infty} n^{\frac{1}{2}} \sigma_{-1}(n) \cos 2\pi x n \sqrt{\frac{1}{4yn}} e^{-2\pi y n}$$
$$= \frac{8\pi}{\sqrt{|d|}} \sum_{n=1}^{\infty} \sigma_{-1}(n) (\cos 2\pi x n) e^{-2\pi y n}$$

in view of

$$K_{\pm\frac{1}{2}}(z) = \sqrt{\frac{\pi}{2z}}e^{-z}.$$
(103)

Recalling $q = e^{\pi i \tau}$, we find that

$$q^{2n} + \bar{q}^{2n} = 2e^{-2\pi yn} \cos 2\pi xn,$$

$$Q(1) = \frac{4\pi}{\sqrt{|d|}} \sum_{n=1}^{\infty} \sigma_{-1}(n)(q^{2n} + \bar{q}^{2n})$$

= $-\frac{4\pi}{\sqrt{|d|}} \left(\log \eta(\tau) + \log \eta(-\bar{\tau})\right) + \frac{4\pi}{\sqrt{|d|}} \left(\frac{\pi i}{12}(\tau - \bar{\tau})\right).$

Here, the last term is

$$\frac{4\pi}{\sqrt{|d|}} \frac{\pi i}{12} 2i \operatorname{Im} \tau = -\frac{\frac{2}{3}\pi^2}{\sqrt{|d|}} y,$$

which cancels $\frac{\pi^2}{3a}$, amounting to (94).

Remark 4. (i) The Chowla–Selberg integral formula is expounded in [Kanemitsu and Tsukada (2014), pp. 136–138]. The case $\boldsymbol{g} = \boldsymbol{h} =$ $\boldsymbol{z}, n = m = 1$ is due to Chowla and Selberg,^{47,50} (cf. also Bateman and Grosswald [Bateman and Grosswald (1964)]), the case m = 1is due to Berndt⁵¹ and the general case with $\boldsymbol{g} = \boldsymbol{h} = \boldsymbol{z}$ is due to Terras (see Ref. 52, Example 4, p. 208).

In Ref. 13, the Epstein zeta function is treated as a quadratic form variation of the Fourier–Bessel expansion, $H_{1,1}^{1,1} \leftrightarrow H_{0,2}^{2,0}$, to the effect that the beta transform (or the perturbed Dirichlet series) corresponds to the K-Bessel series (or Fourier series). The K-Bessel function in (93) reduces to the exponential function for s = 1, whence the Bessel series is nothing but the Lambert series for the Dedekind eta function and so the Kronecker limit formula arises most naturally in which the theta function or the Dedekind eta function shows up. Note that with $\phi_1(s, x)$ in RHB correspondence is $\vartheta(it, itx)$ (47).

In Ref. 53, this natural deduction of eta function expression is presented suggestive of the modular relation. Substituting 2ay for $\sqrt{|\Delta|}$, we see that the coefficient of the second term in (92), resp. in (93), is

86

 \Box

 $2\sqrt{\pi}a^{-s}y^{1-2s}$, resp. $8\pi^sa^{-s}y^{\frac{1}{2}-s}$, in conformity with Ref. 53, (2.11) and (2.12). (ii) Note that

$$\Delta = \Delta(q) = (2\pi)^{12} q^2 \prod_{n=1}^{\infty} (1 - q^{2n})^{24} = \left((2\pi)\eta(\tau)^2\right)^{12}$$

is a cusp form of weight 12. This leads to the normalization

$$\mathbb{D}(A) = \mathbb{D}(\mathfrak{b}) = \sqrt{|\delta(\mathfrak{b})|} (2\pi) |\eta(\tau)|^2 = \frac{|d|^{\frac{1}{4}}}{\sqrt{a}} (2\pi) |\eta(\tau)|^2, \qquad (104)$$

depending only on A, where $\delta(\mathfrak{b})$ is defined by

$$|\delta(\mathfrak{b})| = \left| \det \begin{pmatrix} 1 & 1 \\ \tau & \bar{\tau} \end{pmatrix} \right| = |N\mathfrak{b}|\sqrt{|d|} \quad \text{or} \quad N\mathfrak{b} = \frac{\tau - \bar{\tau}}{i\sqrt{|d|}} = \frac{2\operatorname{Im}\tau}{\sqrt{|d|}},$$
(105)

with the basis $\{1, \tau\}$ in (91).

Under this, (94) leads to the class invariant form (107) of Meyer.

Theorem 7. (i) Under the notation (86)–(90), we have the Chowla–Selberg integral formula

$$\Gamma(s)\zeta(s,A) = \frac{1}{w} \left(2\Gamma(s)\zeta(2s)a^{-s} + \frac{2^{2s}a^{s-1}\sqrt{\pi}}{|d|^{s-\frac{1}{2}}}\Gamma\left(s-\frac{1}{2}\right)\zeta(2s-1) + Q_1(s)\right),$$
(106)

where

$$Q_1(s) = \frac{4\pi^s 2^{s+\frac{1}{2}}}{\sqrt{a}|d|^{\frac{s}{2}-\frac{1}{4}}} \sum_{n=1}^{\infty} n^{s-\frac{1}{2}} \sigma_{1-2s}(n) \cos 2\pi n x K_{s-\frac{1}{2}}(2\pi y n),$$

which is equivalent to the functional equation

$$\left(\frac{2\pi}{\sqrt{|d|}}\right)^{-s} \Gamma(s)\zeta(s,A) = \left(\frac{2\pi}{\sqrt{|d|}}\right)^{-(1-s)} \Gamma(1-s)\zeta(1-s,A).$$

(ii) The Laurent expansion at s = 1 in Meyer's form reads

$$\zeta(s,A) = \frac{\frac{2\pi}{w\sqrt{|d|}}}{s-1} + \frac{2\pi}{w\sqrt{|d|}} \left(2\gamma + \log\frac{(2\pi)^2}{\sqrt{|d|}}\right) - \frac{4\pi}{w\sqrt{|d|}}\log\mathbb{D}(A) + O(s-1),$$
(107)

where $\mathbb{D}(A)$ is the class invariant defined by (104).

Proof. Only (107) needs explanation. Substitution of (90) in (106) or a direct application of (106) gives

$$\zeta(s,A) = \frac{\frac{2\pi}{w\sqrt{|d|}}}{s-1} + \frac{2\pi}{w\sqrt{|d|}} \left(2\gamma + \log\frac{a}{|d|}\right) - \frac{4\pi}{w\sqrt{|d|}} \log|\eta(\tau)|^2 + O(s-1).$$
(108)

where by (105),

$$\mathbb{D}(\mathfrak{b}) = \frac{|d|^{\frac{1}{4}}}{\sqrt{a}} (2\pi) |\eta(\tau)|^2.$$

Hence, (108) amounts to (107).

Corollary 4. For an imaginary quadratic field with discriminant d and class number h, we have the Laurent expansion

$$\zeta_K(s) = \frac{\frac{2\pi\hbar}{w\sqrt{|d|}}}{s-1} + \frac{2\pi\hbar}{w\sqrt{|d|}} \left(2\gamma + \log\frac{(2\pi)^2}{\sqrt{|d|}}\right) - \frac{4\pi}{w\sqrt{|d|}} \sum_{A \in I/P} \log \mathbb{D}(A) + O(s-1).$$
(109)

Lemma 6 (Decomposition theorem). For a quadratic field $K = \mathbb{Q}(\sqrt{D})$ of discriminant d, the decomposition formula holds:

$$\zeta_K(s) = \zeta(s)L(s,\chi),$$

where $L(s, \chi)$ is the Dirichlet L-function associated to the Kronecker (Dirichlet) character $\chi = \chi_d$ corresponding to K. The Laurent expansion reads

$$\zeta_K(s) = \frac{L(1,\chi)}{s-1} + \gamma L(1,\chi) + L'(1,\chi) + O(s-1).$$

88

Comparing Corollary 4 and Lemma 6, we state and prove the Dirichlet class number formula and the Lerch–Chowla–Selberg formula for imaginary quadratic fields.

Theorem 8. Let h(d) = |I/P| denote the class number of the quadratic field $\mathbb{Q}(\sqrt{D})$ of discriminant d. Suppose d < 0.

(i) The Dirichlet class number formula holds:

$$h(d) = \frac{w\sqrt{|d|}}{2\pi}L(1,\chi_d).$$
 (110)

 (ii) The Lerch-Chowla-Selberg formulas hold with class invariants (see Refs. 32, (4.5) and 33, (**), p. 569):

$$\left(\prod_{A} \mathbb{D}(A)\right)^{2} = \left(\frac{2\pi}{\sqrt{|d|}}\right)^{h} \prod_{a=1}^{|d|-1} \Gamma\left(\frac{a}{|d|}\right)^{\frac{w}{2}\chi(a)}.$$
 (111)

In terms of the Lerch zeta function,

$$2\sum_{A\in I/P}\log\mathbb{D}(A) = h\left(\gamma + \log\frac{(2\pi)^2}{\sqrt{|d|}}\right) - \frac{iw}{2\pi}\sum_{a=1}^{|d|-1}\bar{\chi}(a)\left(\ell_1'\right)^o\left(\frac{a}{|d|}\right).$$
(112)

Proof. By Corollary 4, the residue of $\zeta_K(s)$ at s = 1 is $\frac{2\pi}{w\sqrt{|d|}}h$. Hence, by the decomposition formula, Lemma 6, (110) follows.

By (110), (76) reads

$$\gamma L(1,\chi_d) + L'(1,\chi_d) = -\frac{\pi}{\sqrt{|d|}} \sum_{a=1}^{|d|-1} \bar{\chi}(a) \log \Gamma\left(\frac{a}{|d|}\right) + \frac{2\pi h}{w\sqrt{|d|}} (2\gamma + \log 2\pi).$$

Equating this with the Laurent constant in (109) and dividing both sides by $\frac{2\pi}{w\sqrt{|d|}}$, we conclude that

$$2\sum_{A\in I/P}\log\mathbb{D}(A) = h\log\frac{2\pi}{\sqrt{|d|}} + \frac{w}{2}\sum_{a=1}^{|d|-1}\bar{\chi}(a)\log\Gamma\left(\frac{a}{|d|}\right)$$
(113)

(111) is the exponentiated form of this.

Equating (109) and (75), we derive that

$$\frac{2\pi h}{w\sqrt{|d|}} \left(\gamma + \log\frac{(2\pi)^2}{\sqrt{|d|}}\right) - \frac{4\pi}{w\sqrt{|d|}} \sum_{A \in I/P} \log \mathbb{D}(A)$$
$$= -\frac{i}{\sqrt{|d|}} \sum_{a=1}^{|d|-1} \bar{\chi}(a) \left(\ell_1'\right)^o \left(\frac{a}{|d|}\right),$$

 \square

which leads to (112).

2.2. Genesis of the formula

In the light of Tables 2 and 3, Theorem 8 may be described roughly as follows. The class invariant on the left of (113) is the processed (even part of) the theta function with one-time use of the functional equation while the right-hand side of (112) is in terms of the Lerch zeta function which reflects the property of the almost unprocessed conjugate theta function.

In terms of the gamma function, the right side of (113) is processed once by the functional equation and in more orderly form than (112).

Thus, in a sense, this is a relation between the processed theta functions and their counterpart under RHB correspondence linked by the decomposition theorem, Lemma 6, which is a consequence of the Artin reciprocity law (see Ref. 54, pp. 196–199).

Table 2. Lerch–Chowla–Selberg ingredients: Eisenstein ser.

Name	RHB	β trans.	f.e.	Product f.
Eisenstein ser.	$\vartheta\leftrightarrow \zeta^e$	$\zeta_{\mathbb{Z}^2} \to \zeta^e$	$\zeta^e \to \ell^e$	$\eta=\theta^e$

Ta	ble	3	. 1	Lerc	h–C	Chow.	la–	Sel	lberg	ingree	lients:	F	ourier	sine	ser
----	-----	---	-----	------	-----	-------	-----	-----	-------	--------	---------	---	--------	------	----------------------

Name	RHB	Diff. w.r.t. s	Termwise diff.
Der. of Hurwitz Der. of Lerch	$\begin{array}{c} \vartheta, \tilde{\vartheta} \leftrightarrow \zeta(s, x) \\ \tilde{\vartheta} \leftrightarrow \ell_s^o \end{array}$	$\begin{aligned} \zeta'(0,x) &\to \left(\ell_1^e\right)' \text{ (f.e.)}\\ \left(\ell_1^e\right)' \end{aligned}$	Fourier sine ser. Fourier sine ser.

3. Product Expansion of Theta-Allied Functions

In previous sections, we worked either in $\sigma > 1$ or $\sigma < 0$ save for the Lerch zeta function which can go into the critical strip $0 < \sigma < 1$. For the Riemann zeta function to go into the critical strip, it needs the Euler product and Riemann defines $\zeta(s)$ by (114) first and then by the additive expression (2). Euler products in general have been one of the most important objects of research from the time of Euler and the most famous Euler product for $\zeta(s)$ reads

$$\zeta(s) = \prod_{p} \left(1 - p^{-s} \right)^{-1} \tag{114}$$

for $\sigma > 1$ and where p runs through all the primes. As is known, this expresses the UFD property of the basic sequence $\{n\}$. Thus, wherever there is the UFD, there is the Euler product.

Here, we speak of the product expressions (80) for theta functions.

Corollary 5 (Infinite product for theta). For |q| < 1 and $z \neq 0$,

$$\vartheta(\tau, x) = \prod_{n=1}^{\infty} (1 - q^{2n})(1 + zq^{2n-1})(1 + z^{-1}q^{2n-1}), \qquad (115)$$

where $z = e^{2\pi i x}$.

Ogg's¹⁸ proof of Corollary 5 amounts to proving the inverse Mellin transform of the logarithm of the infinite product $\theta_0(\tau, x)$, say, satisfies the functional equation from which the same transformation formula as (43) is deduced. It therefore turns out that the quotient $\frac{\vartheta}{\vartheta_0}$, where ϑ is the additive expression in (42), is an elliptic function in x with periods 1, τ with at most one singularity, so that it is a constant, cf. e.g. Ref. 19, pp. 53–54. Use is made of the second equality in (1) in the form

$$e^{-\pi \frac{\tau}{i}u} = \frac{1}{2\pi i} \int_{(c)} \left(\frac{\tau}{i}\right)^{-s} \Gamma(s)(\pi u)^{-s} \,\mathrm{d}s, \quad c > 0, \quad u > 0.$$

Since the zeta function is $\zeta(s)\zeta(s+1)$ (e.g. Ref. 55) which is in RHB correspondence with $\log \eta$, it may be the case that if we employ an additive expression for the relevant theta-like function, we may argue
in the same way as Ogg to derive the product expression. Hitherto only the product expression is used as the definition. This direction will be pursued elsewhere.

We shall elucidate the genesis of Siegel's proof of Hamburger's theorem.⁵⁶ Reference 1 (pp. 201–211) contains a rather transparent description of the Siegel–Hecke proof of the Hamburger theorem (especially the translation of introduction of Ref. 57), cf. Refs. 3, 4, 56–58, etc. However, in neither Ref. 1 nor Ref. 59, there is genesis given of Siegel's proof.

We appeal to the following which is (an extract of) Ref. 13, Example 4.2, pp. 120–121.

Lemma 7. Suppose the Dirichlet series φ and ψ satisfy the Riemann-type functional equation

$$\Gamma\left(\frac{1}{2}s\right)\varphi(s) = \Gamma\left(\frac{1}{2}(r-s)\right)\psi(r-s)$$

with a simple pole at s = r > 0 with residue ρ . Then the Fourier-Bessel expansion $H_{1,1}^{1,1} \leftrightarrow H_{0,2}^{2,0}$ entails

$$2\Gamma\left(\frac{1}{2}s\right)\sum_{n=1}^{\infty}\frac{\alpha_{n}}{(\lambda_{n}^{2}+z^{2})^{\frac{1}{2}s}}$$

= $4z^{\frac{1}{2}(r-s)}\sum_{n=1}^{\infty}\frac{\beta_{n}}{\mu_{n}^{\frac{1}{2}(r-s)}}K_{\frac{1}{2}(r-s)}(2\mu_{n}z) + 2\Gamma\left(\frac{1}{2}s\right)\varphi(0)z^{-s}$
+ $\rho\Gamma\left(\frac{1}{2}(s-r)\right)z^{r-s},$

which is a generalization of Watson's formula.

We specialize Lemma 7 to the case

$$\lambda_n = \mu_n = \sqrt{\pi}n, \quad r = 1, \quad z = \sqrt{\pi}t, \quad s = 1$$

and simplify the resulting expression to derive

$$\sum_{n=1}^{\infty} \alpha_n \left(\frac{1}{t+in} + \frac{1}{t-in} \right) - \pi t H(t) = 2\pi \sum_{n=1}^{\infty} \beta_n e^{-2\pi nt},$$

which is in Ref. 56, (12). After this, Siegel's argument applies. Indeed, both Refs. 56 (5) and 59 (formula before (6)) are the reduction of the K-Bessel function to the exponential function, (103).

Thus, we have seen two occurrences of the Fourier–Bessel expansion. It may be proper to ask if the Hurwitz formula may be another occurrence. But it seems that direct application leads to the K-Bessel function series (see Ref. 13, Example 4.3, pp. 121–122) which suggests a reason why Siegel's proof fails in the case of Hecke (see Ref. 1, p. 207). As is elucidated in Ref. 14, it seems the Ewald expansion leads most naturally to the Hurwitz formula.

Acknowledgments

The third author would like to thank Professor Heng Huat Chan of NUS for his inspirational book,¹⁹ which sparked his interest in the theta functions and elliptic functions. The authors thank the referees for the helpful comments on the earlier version of paper.

References

- M. I. Knopp, Hamburger's theorem on ζ(s) and the abundance principle for Dirichlet series with functional equations. In *Number Theory*, R. P. Bambah, V. C. Dumir, and R. J. Hans-Gill (eds.). Trends Math., pp. 201–216. Birkhäuser, Basel (2000).
- B. Riemann, Über die anzahl der primzahlen unter einer gegebenen grösse, Monatsber. Berlin. Akad. ed. by H. Weber, 2nd ed. pp. 671– 680, Dover, New York 1953 (1859).
- E. Hecke, Über die Bestimmung Dirichletscher Reihen durch ihre Funktionalgleichung, Math. Ann. 112(1), 664–699 (1936). https://doi.org/ 10.1007/BF01565437.
- E. Hecke, Lectures on Dirichlet Series, Modular Functions and Quadratic Forms. Vandenhoeck & Ruprecht, Göttingen (1983). ISBN 3-525-40727-0. Edited by Bruno Schoeneberg, with the collaboration of Wilhelm Maak.
- S. Bochner, Some properties of modular relations, Ann. of Math. 53(2), 332–363 (1951). https://doi.org/10.2307/1969546.
- A. K. Wang, N. L., and S. Kanemitsu, Limiting values and functional and difference equations, *Math. Diff. Diff. Equations* 8, 407 (2020). https://doi.org/10.3390/math8030407.

- W. Schnee, Die funktionalgleichung der Zetafunktion und der Dirichletschen Reihen mit periodischen Koeffizienten, Math. Z. 31(1), 378–390 (1930). https://doi.org/10.1007/BF01246420.
- M. Ishibashi and S. Kanemitsu, Dirichlet series with periodic coefficients, *Results Math.* 35(1–2), 70–88 (1999). https://doi.org/10.1007/BF03322023.
- 9. F. Sato, Searching for the origin of prehomogeneous vector spaces, Keynote Address at Annual Meeting of the Math. Soc. Japan (1992).
- H. Davenport, *Multiplicative Number Theory*. Lectures in Advanced Mathematics, No. 1, Markham Publishing Co., Chicago (1967). Lectures given at the University of Michigan, Winter Term, 1966.
- Y. Yamamoto, Dirichlet series with periodic coefficients. In Algebraic number theory (Kyoto Internat. Sympos., Res. Inst. Math. Sci., Univ. Kyoto, Kyoto, 1976), pp. 275–289. Japan Soc. Promotion Sci., Tokyo (1977).
- 12. J. I. Hutchinson, Properties of functions represented by the Dirichlet series $\sum (a\nu+b)^{-s}$, or by linear combinations of such series, *Trans. Am. Math. Soc.* **31**(2), 322–344 (1929). https://doi.org/10.2307/1989387.
- S. Kanemitsu and H. Tsukada, Contributions to the Theory of Zeta-Functions. Vol. 10, Series on Number Theory and its Applications. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ (2015). https://doi.org/10.1142/8711. The modular relation supremacy.
- W. Li, H. Li, and J. Mehta, Around the Lipschitz summation formula, Math. Probl. Eng. pp. Art. ID 5762823, 16 (2020). https://doi.org/10. 1155/2020/5762823.
- J. Milnor, On polylogarithms, Hurwitz zeta functions, and the Kubert identities, *Enseign. Math. (2)* 29(3–4), 281–322 (1983). ISSN 0013-8584.
- S. Kanemitsu and H. Tsukada, Vistas of Special Functions. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ (2007). https://doi. org/10.1142/9789812708830.
- K. Chakraborty, S. Kanemitsu, and T. Kuzumaki, Finite expressions for higher derivatives of the Dirichlet L-function and the Deninger *R*-function, *Hardy-Ramanujan J.* **32**, 38–53 (2009). ISSN 2804-7370.
- A. P. Ogg, On product expansions of theta-functions, *Proc. Am. Math. Soc.* 21, 365–368 (1969). https://doi.org/10.2307/2037005.
- H. H. Chan, Theta Functions, Elliptic Functions and π. De Gruyter Textbook, De Gruyter, Berlin (2020). https://doi.org/10. 1515/9783110541915. With a foreword by Bruce Berndt.
- 20. E. T. Whittaker and G. N. Watson, A Course of Modern Analysis. Cambridge Mathematical Library, Cambridge University Press, Cambridge (1996). https://doi.org/10.1017/CBO9780511608759. An introduction to the general theory of infinite processes and of analytic

functions; with an account of the principal transcendental functions, Reprint of the fourth (1927) edition.

- A. Ogg, Modular Forms and Dirichlet Series. W. A. Benjamin, Inc., New York and Amsterdam (1969).
- J. Mehta and P. Y. Zhu, Proof of the functional equation for the Riemann zeta-function, *Hardy-Ramanujan J.* 44, 143–151 (2021). ISSN 2804-7370.
- T. Funakura, On Kronecker's limit formula for Dirichlet series with periodic coefficients, Acta Arith. 55(1), 59–73 (1990). https://doi.org/ 10.4064/aa-55-1-59-73.
- F. Li, N. Wang, and S. Kanemitsu, Number Theory and Its Applications. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ (2013). https://doi.org/10.1142/8601.
- K. Chakraborty, S. Kanemitsu, and H. Tsukada, Vistas of Special Functions II. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ (2010). ISBN 978-981-4273-97-8; 981-4273-97-X.
- 26. T. M. Apostol, *Introduction to Analytic Number Theory*. Undergraduate Texts in Mathematics. Springer-Verlag, New York-Heidelberg (1976).
- H. Joris, On the evaluation of Gaussian sums for non-primitive Dirichlet characters, *Enseign. Math. (2)* 23(1–2), 13–18 (1977). ISSN 0013-8584.
- S. Kanemitsu, H. Li, and N. Wang, Weighted short-interval character sums, *Proc. Am. Math. Soc.* 139(5), 1521–1532 (2011). https://doi. org/10.1090/S0002-9939-2010-10572-5.
- C. J. de la Vallée Poussin, Recherches analytiques de la théorie des nombres premiers, Ann. Soc. Sci. Bruxelles. 20 B, 183–256, 281–352, 363–397 (1896).
- 30. M. Hashimoto, S. Kanemitsu, and M. Toda, On Gauss' formula for ψ and finite expressions for the *L*-series at 1, *J. Math. Soc. Jpn.* **60**(1), 219–236 (2008). http://projecteuclid.org/euclid.jmsj/1206367961.
- N. Kurokawa, One hundred years of the zeta regularized products. In The 39th Algebra Sympos., pp. 153–166 (1994).
- 32. C. Deninger, On the analogue of the formula of Chowla and Selberg for real quadratic fields, J. Reine Angew. Math. 351, 171–191 (1984). https://doi.org/10.1515/crll.1984.351.171.
- C. Meyer, *l*-funktionen quadratischer zahlkörper, *Vorlesung, Köln.* (1981).
- Y. Motohashi, A new proof of the limit formula of Kronecker, Proc. Japan Acad. 44, 614–616 (1968). http://projecteuclid.org/euclid.pja/ 1195521077.
- 35. C. L. Siegel, *Lectures on Advanced Analytic Number Theory*. Tata Institute of Fundamental Research Lectures on Mathematics, No. 23,

Tata Institute of Fundamental Research, Bombay (1961). Notes by S. Raghavan.

- A. P. Novikov, The Kronecker limit formula in a real quadratic field, Izv. Akad. Nauk SSSR Ser. Mat. 44(4), 886–917, 974 (1980). ISSN 0373-2436.
- D. Zagier, A Kronecker limit formula for real quadratic fields, Math. Ann. 213, 153–184 (1975). https://doi.org/10.1007/BF01343950.
- A. Weil, Elliptic Functions According to Eisenstein and Kronecker. Ergebnisse der Mathematik und ihrer Grenzgebiete, Band 88, Springer-Verlag, Berlin-New York (1976). ISBN 3-540-07422-8.
- M. Lerch, Sur quelques formules du nombre des classes, Bull. Sci. Math. 21, 290–304 (1897).
- A. Berger, Sur une sommation de quelques séries, Nova Acta Reg. Soc. Sci. Upsal. 12(3), 29–31 (1883).
- 41. R. Campbell, Les intégrales eulériennes et leurs applications. Étude approfondie de la fonction gamma. Collection Universitaire de Mathématiques, XX, Dunod, Paris (1966).
- 42. J. Dufresnoy and C. Pisot, Sur la relation fonctionnelle $f(x+1)-f(x) = \varphi(x)$, Bull. Soc. Math. Belg. 15, 259–270 (1963). ISSN 0373-2053.
- E. Landau, Ueber die zu einem algebraischen Zahlkörper gehörige Zetafunction und die Ausdehnung der Tschebyschefschen Primzahlentheorie auf das Problem der Vertheilung der Primideale, J. Reine Angew. Math. 125, 64–188 (1903). https://doi.org/10.1515/crll.1903.125.64.
- M. Lerch, Bemerkungen über trigonometrischen reihen mit positive coeffizienten, Sitzungsber. König. Böhm. Ges. Wiss. Math. Naturwiss. Kl. 23, 1–19 (1898).
- M. Gut, Die Zetafunktion, die Klassenzahl und die Kronecker'sche Grenzformel eines beliebigen Kreiskörpers, Comment. Math. Helv. 1(1), 160–226 (1929). https://doi.org/10.1007/BF01208364.
- T. Kuzumaki, Asymptotic expansions for a class of zeta-functions, *Ramanujan J.* 24(3), 331–343 (2011). https://doi.org/10.1007/ s11139-010-9250-5.
- A. Selberg and S. Chowla, On Epstein's zeta-function, J. Reine Angew. Math. 227, 86–110 (1967). https://doi.org/10.1515/crll.1967.227.86.
- G. H. Hardy, Some multiple integrals, *Quart. J. Math.* 5(2), 357–375 (1908).
- D. Kaminski and R. B. Paris, Asymptotics via iterated Mellin-Barnes integrals: application to the generalised Faxén integral, *Methods Appl. Anal.* 4(3), 311–325 (1997). https://doi.org/10.4310/MAA.1997. v4.n3.a5.
- S. Chowla and A. Selberg, On Epstein's zeta function. I, *Proc. Nat. Acad. Sci. U.S.A.* **35**, 371–374 (1949). https://doi.org/10.1073/pnas. 35.7.371.

- B. C. Berndt, Identities involving the coefficients of a class of Dirichlet series. V, VI, Trans. Am. Math. Soc. 160, 139–156 (1971); *ibid.* 160, 157–167 (1971). https://doi.org/10.1090/s0002-9947-1971-0280693-9.
- A. Terras, Harmonic Analysis on Symmetric Spaces and Applications. I. Springer-Verlag, New York (1985). https://doi.org/10.1007/ 978-1-4612-5128-6.
- 53. N. Y. Zhang and K. S. Williams, On the Epstein zeta function, *Tamkang J. Math.* **26**(2), 165–176 (1995). ISSN 0049-2930.
- T. Takagi, Daisûteki Seisûron. Gaisetsu oyobi Ruitairon. Iwanami Shoten, Tokyo (1948).
- 55. A. Weil, Sur une formule classique, J. Math. Soc. Jpn. 20, 400–402 (1968). https://doi.org/10.2969/jmsj/02010400.
- 56. C. Siegel, Bemerkung zu einem Satz von Hamburger über die Funktionalgleichung der Riemannschen Zetafunktion, Math. Ann. 86(3–4), 276–279 (1922). https://doi.org/10.1007/BF01457989.
- E. Hecke, Herleitung des Euler-Produktes der Zetafunktion und einiger L-Reihen aus ihrer Funktionalgleichung, Math. Ann. 119, 266–287 (1944). https://doi.org/10.1007/BF01563745.
- H. Hamburger, Über die Riemannsche Funktionalgleichung der ξ-Funktion, Math. Z. 10(3–4), 240–254 (1921). https://doi.org/10.1007/ BF01211612.
- W. Culp-Ressler, K. Flood, A. Heath, and W. D. A. Pribitkin, On solutions to Riemann's functional equation, *Ramanujan J.* 4(1), 5–9 (2000). https://doi.org/10.1023/A:1009836519389.
- Ehrenpreis, L. Fourier analysis, partial differential equations and automorphic forms, *Proc. Sympos. Pure Math.*, 49, Part 2, *Amer. Math. Soc.*, Providence, RI, 45–100 (1987).
- 61. Nörlund, N. E. Vorlesungen über Differenzenrechnung, Chelsea, New York (1954).
- Bateman, P. T. and Grosswald, E. On Epstein's zeta function, Acta Arith. 9, 365–373, (1964).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0005

Chapter 5

Sylvester Sums on the Frobenius Set in Arithmetic Progression with Initial Gaps

Takao Komatsu

Department of Mathematical Sciences, School of Science, Zhejiang Sci-Tech University, Hangzhou 310018, China komatsu@zstu.edu.cn

Let a_1, a_2, \ldots, a_k be positive integers with $gcd(a_1, a_2, \ldots, a_k) = 1$. Frobenius number is the largest positive integer that is NOT representable in terms of a_1, a_2, \ldots, a_k . When $k \ge 3$, there is no explicit formula in general, but some formulae may exist for special sequences a_1, a_2, \ldots, a_k , including those forming arithmetic progressions and their modifications. In this chapter, we give explicit formulae for the sum of nonrepresentable positive integers (Sylvester sum) as well as Frobenius numbers and the number of nonrepresentable positive integers (Sylverster number) for a_1, a_2, \ldots, a_k forming arithmetic progressions with initial gaps.

1. Introduction

Let a_1, \ldots, a_k be positive integers with $gcd(a_1, \ldots, a_k) = 1$. It is well known that all sufficiently large integers can be represented as a nonnegative integer combination of a_1, \ldots, a_k . Then it is important to determine the largest positive integer that is not representable as a nonnegative integer combination of given positive integers that are coprime. Such a problem is known as the *Frobenius Problem* and this largest positive integer is denoted by $g(a_1, \ldots, a_k)$ and called the *Frobenius number* (see Ref. 1 for general references).¹ This problem has also been known as the Coin Exchange Problem, Postage Stamp Problem or Chicken McNugget Problem, and so has a long history. Together with the Frobenius numbers, the number of positive integers with no nonnegative integer representation by a_1, \ldots, a_k has also been studied for a long time. This number is sometimes called the *Sylvester number* (or *genus* in numerical semigroup) and denoted by $n(a_1, \ldots, a_k)$.

According to Sylvester, for positive integers a and b with gcd(a,b) = 1,

$$g(a,b) = (a-1)(b-1) - 1 \quad [4],$$

$$n(a,b) = \frac{1}{2}(a-1)(b-1) \quad [5]$$

There are many kinds of problems related to the Frobenius problem. The problems for the number of solutions (e.g. Ref. 6) and the sum of integer powers of values of the gaps in numerical semigroups (e.g. Refs. 7–9) are popular. In Ref. 10, the various results within the cyclotomic polynomial and numerical semigroup communities are unified. One of the other famous problems is about the so-called *Sylvester sums*:

$$s(a_1,\ldots,a_k) := \sum_{n \in \mathrm{NR}(a_1,\ldots,a_k)} n$$

(see, e.g. Ref. 1, Section 5.5, 11 and references therein), where $NR(a_1, \ldots, a_k)$ denotes the set of positive integers without nonnegative integer representation by a_1, \ldots, a_k . This is exactly the set of gaps in numerical semigroup. It is harder to obtain the Sylvester number than the Frobenius number and even harder to obtain the Sylvester sum. Finally, long time after Sylvester, Brown and Shiue⁷

 $^{^1\}mathrm{Some}$ other symbols have also been used by different backgrounds and authors. The symbols used in this paper are mainly based on the literature, such as Refs. 1–3.

found the exact value for positive integers a and b with gcd(a, b) = 1,

$$s(a,b) = \frac{1}{12}(a-1)(b-1)(2ab-a-b-1).$$
 (1)

Rødseth¹² generalized Brown and Shiue's result by giving a closed form for $s_{\mu}(a, b) := \sum_{n \in NR(a,b)} n^{\mu}$, where μ is a positive integer. When k = 2, there exist beautiful closed forms for Frobenius num-

When k = 2, there exist beautiful closed forms for Frobenius numbers, Sylvester numbers and Sylvester sums, but when $k \ge 3$, exact determination of these numbers is extremely difficult. The Frobenius number cannot be given by closed formulas of a certain type,¹³ the problem to determine $g(a_1, \ldots, a_k)$ is NP-hard under Turing reduction (see, e.g. Ref. 1). Nevertheless, one convenient formula is found by Johnson.¹⁴ One analytic approach to the Frobenius number can be seen in Refs. 15 and 16.

Though closed forms for the general case are hopeless for $k \ge 3$, several formulae for Frobenius numbers, Sylvester numbers and Sylvester sums have been considered under special cases. For example, one of the best expositions for the Frobenius number in three variables can be seen in Ref. 17. For general $k \ge 3$, the Frobenius number and the Sylvester number for some special cases are calculated, including arithmetic sequences and geometric-like sequences (e.g. Refs. 2, 18–21).

In fact, by introducing the Apéry set, it is possible to determine the functions g(A), n(A) and s(A) for the set of positive integers $A := \{a_1, a_2, \ldots, a_k\}$ with $gcd(a_1, a_2, \ldots, a_k) = 1$. For $a_1 = min(A)$, we denote by

$$Ape(A) = Ape(A, a_1) = \{m_0, m_1, \dots, m_{a_1-1}\}\$$

the Apéry set of A, where m_i is the least positive integer that can be represented by a nonnegative integral linear combination of a_2, \ldots, a_k , satisfying $m_i \equiv i \pmod{a_1} (1 \leq i \leq a_1 - 1)$. Note that m_0 is defined to be 0. The element 0 is often excluded because it does not affect the calculation.

Lemma 1. We have

$$g(a_1, a_2, \dots, a_k) = \left(\max_{1 \le i \le a_1 - 1} m_i\right) - a_1, \quad [22]$$

$$n(a_1, a_2, \dots, a_k) = \frac{1}{a_1} \sum_{i=1}^{a_1-1} m_i - \frac{a_1 - 1}{2}, \quad [2]$$
$$s(a_1, a_2, \dots, a_k) = \frac{1}{2a_1} \sum_{i=1}^{a_1-1} m_i^2 - \frac{1}{2} \sum_{i=1}^{a_1-1} m_i + \frac{a_1^2 - 1}{12}. \quad [3]$$

The third formula appeared with a typo in Ref. 3, and it has been corrected in Refs. 23 and 24. Recently, we study the weighted sums and weighted power sums. When k = 2, a general formula can be expressed in terms of the Apostol–Bernoulli numbers.²⁵ For general k, we can have a formula by using Eulerian numbers.²⁶ In Ref. 27, a more general formula including n(A) and s(A) is given by using Bernoulli numbers.

As a more general case than the arithmetic sequence, the sequence $a_1 = a, a_2 = a + d, \dots, a_k = a + (k - 1)d, a_{k+1} = a + Kd$ with K > k has been studied.² This is one typical case of the so-called almost arithmetic sequence. In this case, there is an additional term after some gap. As special cases, the Frobenius numbers and the Sylvester number are given for the sequences a, a + 1, a + 2, a + 4; a, a + 1, a + 2, a + 5; a, a + 1, a + 2, a + 6 and so on^{2, 28} i.e. a gap appears in the last. In Ref. 29, the Frobenius numbers of various cases are expressed in which the a_i 's lie in an arithmetic progression, but the results are incomplete. In Ref. 30, when the a_i form an almost arithmetic sequence, by considering the Apéry set, algorithms to determine the Sylvester number and sum are given. But, in this paper the computations rely on Ape(A, a + (K+1)d) instead of Ape(A, a). In Ref. 31, the authors gave an alternative description of the Apéry set of the first element in the arithmetic sequence. Their aim was different: they wanted to describe the minimal presentation of the semigroup. The approaches in Refs. 30 and 31 may apply to any almost arithmetic sequence, but they both have an extra burden: they require the pre-computation of a couple of constants depending on the sequence. Some other applications for the case of almost arithmetic sequences can be found in Section 4 of Ref. 32.

In Ref. 21, we give explicit expressions of the Sylvester sum and the weighed sum, where a_1, a_2, \ldots, a_k forms arithmetic sequences. In this chapter, we not only study only the Frobenius numbers but also the Sylvester number and the Sylvester sum, where $a_1 = a$, $a_2 = a + (K+1)d$, $a_3 = a + (K+2)d$,..., $a_{k-K+1} = a + kd$ with d > 0, gcd(a, d) = 1 and $k \ge K+1 \ge 2$, i.e. a gap appears in the first. As special cases, we yield these numbers and sums explicitly for the sequences a, a+2, a+3, a+4; a, a+3, a+4, a+5; a, a+4, a+5, a+6; a, a+5, a+6, a+7 and so on.

2. Main Result

For positive integers a and d with $a \ge 2$ and gcd(a, d) = 1, consider the sequence $a, a + (K+1)d, a + (K+2)d, \ldots, a + kd$. Note that the last (k - K) terms form an arithmetic sequence, and there is a gap in the first part. Determine nonnegative integers q and r as

$$a + K = qk + r, \quad 0 \le r < k, \tag{2}$$

i.e. $q = \lfloor (a+K)/k \rfloor$ and $r = a + K - \lfloor (a+K)/k \rfloor k$.

In this section, we assume that $1 \leq K \leq (k-1)/2$, i.e. the gaps between the first term and the second term are not so big. The case when K > (k-1)/2 is discussed in the following section. In addition to this condition, we shall discuss the two cases separately: r > K or $r \leq K$.

Case 1: When r > K, all the elements of Apéry set excluding 0 (mod *a*) can be determined in the following table:

			a + (K+1)d	 a + kd
2a + (k+1)d		2a + (k + K)d	2a + (k + K + 1)d	 2a + 2kd
3a + (2k+1)d		3a + (2k + K)d	3a + (2k + K + 1)d	 3a + 3kd
qa + ((q-1)k + 1)d		qa + ((q-1)k + K)d	qa + ((q-1)k + K + 1)d	 qa + qkd
(q+1)a + (qk+1)d		(q+1)a + (qk+r)d	
	(q + 1)a + ad		

The last line consists of r-1 terms excluding (q+1)a + ad = (q+1)a + (qk + r - K)d because it is equal to 0 (mod *a*). Note that by gcd(a, d) = 1, the set of all the elements in this table forms a complete residue system modulo *a* excluding 0 (mod *a*):

$$\{(K+1)d, (K+2)d, \dots, (a-1)d, (a+1)d, \dots, (a+K)d\} = \{1, 2, \dots, a-1\} \pmod{a}.$$

T. Komatsu

Since $K \leq (k-1)/2$, 2a + (k+1)d can be represented by using the elements of the last (K-k) terms of the given sequence, which appear in the first line, all terms in this table after 2a + (k+1)d can be also represented. In addition, none of the elements in the table can be represented by subtracting *a*. Therefore, the elements in this table form the Apéry set except 0 (mod *a*). Hence,

$$\sum_{i=1}^{a-1} m_i = \left(1 \cdot (k-K) + (2+3+\dots+q)k + (q+1)(r-1)\right)a + \left((K+1) + (K+2) + \dots + (a-1) + (a+1) + \dots + (a+K)\right)d = \left(\frac{q(q+1)}{2}k - K + (q+1)(r-1)\right)a + \left(\frac{a-1}{2} + K\right)ad$$

$$\begin{split} \sum_{i=1}^{a-1} m_i^2 &= \left(1^2 \cdot (k-K) + (2^2 + 3^2 + \dots + q^2)k + (q+1)^2(r-1)\right)a^2 \\ &+ \left((K+1)^2 + (K+2)^2 + \dots + (a-1)^2 \\ &+ (a+1)^2 + \dots + (a+K)^2\right)d^2 \\ &+ 2ad\left(1\left((K+1) + \dots + k\right) + 2\left((k+1) + \dots + (2k)\right) \\ &+ 3\left((2k+1) + \dots + (3k)\right) + \dots + q\left(((q-1)k+1) \\ &+ \dots + (qk)\right) + (q+1)\left((qk+1) + \dots + (qk+r) - a\right)\right) \\ &= \left(\frac{q(q+1)(2q+1)}{6}k - K + (q+1)^2(r-1)\right)a^2 \\ &+ \left(\frac{(a+K)(a+K+1)(2a+2K+1)}{6} \\ &- \frac{K(K+1)(2K+1)}{6} - a^2\right)d^2 + 2ad\left(\left(\frac{k(k+1)}{2} \\ &- \frac{K(K+1)}{2}\right) + 2\left(\frac{2k(2k+1)}{2} - \frac{k(k+1)}{2}\right) \end{split}$$

$$\begin{aligned} &+ 3\left(\frac{3k(3k+1)}{2} - \frac{2k(2k+1)}{2}\right) + \cdots \\ &+ q\left(\frac{qk(qk+1)}{2} - \frac{(q-1)k((q-1)k+1)}{2}\right) \\ &+ (q+1)\left(\frac{(qk+r)(qk+r+1)}{2} - \frac{qk(qk+1)}{2} - a\right)\right) \\ &= \left(\frac{q(q+1)(2q+1)}{6}k - K + (q+1)^2(r-1)\right)a^2 \\ &+ \frac{\left(6K(a+K+1) + (a-1)(2a-1)\right)}{6}ad^2 \\ &+ 2ad\left(-\frac{K(K+1)}{2} - \frac{qk(q+1)(2qk+k+3)}{12} \\ &+ \frac{(q+1)(a+K)(a+K+1)}{2} - a(q+1)\right). \end{aligned}$$

Therefore, by the third formula in Lemma 1 together with q = (a + K - r)/k, we have

$$\begin{split} s(a, a + (K + 1)d, \dots, a + kd) \\ &= \frac{1}{2a} \sum_{i=1}^{a-1} m_i^2 - \frac{1}{2} \sum_{i=1}^{a-1} m_i + \frac{a^2 - 1}{12} \\ &= \frac{1}{12k^2} \Big(2a^4 + 6(K - 1)a^3 + (6K(K - 2) - k(k + 6) \\ &- 6r^2 + 6(k + 2)r)a^2 + (2K^2(K - 3) - 2Kk(k + 3) \\ &+ 4r^3 - 6(K + k + 1)r^2 + (6K(k + 2) + 2k(k + 3))r)a \\ &- k^2 + (2a^2 + 3(2K - 1)a + 6K^2 + 6K + 1)k^2d^2 \\ &+ (4a^3 + 3(4K - 3)a^2 \\ &+ (6K(2K - 1) - k(k + 6) - 6r^2 + 6(k + 2)r)a \\ &+ 4K^3 - 3K^2(k - 1) - Kk(k + 3) \\ &+ 2r^3 - 3(2K + k + 1)r^2 + k(6K + k + 3)r)kd \Big). \end{split}$$

T. Komatsu

Case 2: Let $0 \le r \le K$. Since the term $qa + (qk + r - K)d = qa + ad \equiv 0 \pmod{a}$ is excluded, we have

$$\sum_{i=1}^{a-1} m_i = \left(1 \cdot (k-K) + (2+3+\dots+q)k - q + (q+1)r\right)a + \left(\frac{(a-1)a}{2} + aK\right)d$$
$$= \left(\frac{q(q+1)}{2}k - K - q + (q+1)r\right)a + \left(\frac{a-1}{2} + K\right)ad$$

$$\begin{split} \sum_{i=1}^{a-1} m_i^2 &= \left(1^2 \cdot (k-K) + (2^2 + 3^2 + \dots + q^2)k - q^2 + (q+1)^2 r\right)a^2 \\ &+ \left((K+1)^2 + (K+2)^2 + \dots + (a-1)^2 \\ &+ (a+1)^2 + \dots + (a+K)^2\right)d^2 \\ &+ 2ad\left(1\left((K+1) + \dots + k\right) + 2\left((k+1) + \dots + (2k)\right) \\ &+ 3\left((2k+1) + \dots + (3k)\right) + \dots + q\left(((q-1)k+1) \\ &+ \dots + (qk)\right) - qa\right) + (q+1)\left((qk+1) + \dots + (qk+r)\right)\right) \\ &= \left(\frac{q(q+1)(2q+1)}{6}k - K - q^2 + (q+1)^2 r\right)a^2 \\ &+ \frac{(6K(a+K+1) + (a-1)(2a-1))}{6}ad^2 \\ &+ 2ad\left(-\frac{K(K+1)}{2} - \frac{qk(q+1)(2qk+k+3)}{12} \\ &+ \frac{(q+1)(a+K)(a+K+1)}{2} - aq\right). \end{split}$$

Therefore, by the third formula in Lemma 1 together with q = (a + K - r)/k, we have

$$\begin{split} s(a, a + (K+1)d, \dots, a + kd) \\ &= \frac{1}{12k^2}(2a^4 + 6(K-1)a^3 \\ &+ \left(6K(K-2) - k(k-6) - 6r^2 + 6(k+2)r\right)a^2 \\ &+ \left(2K^2(K-3) - 2Kk(k-3) \\ &+ 4r^3 - 6(K+k+1)r^2 + (6K(k+2) - 2k(k-3))r\right)a \\ &- k^2 + \left(2a^2 + 3(2K-1)a + 6K^2 + 6K + 1\right)k^2d^2 \\ &+ \left(4a^3 + 3(4K-3)a^2 \\ &+ \left(6K(2K-1) - k(k-6) - 6r^2 + 6(k+2)r\right)a \\ &+ 4K^3 - 3K^2(k-1) - Kk(k+3) \\ &+ 2r^3 - 3(2K+k+1)r^2 + k(6K+k+3)r\right)kd). \end{split}$$

Theorem 1. Let a, d, K, k be positive integers with $a \ge 2$, gcd(a, d) = 1 and $K \le (k - 1)/2$. Let $r = a + K - \lfloor (a + K)/k \rfloor k$. If r > K, then

$$\begin{split} s(a, a + (K + 1)d, \dots, a + kd) \\ &= \frac{1}{12k^2}(2a^4 + 6(K - 1)a^3 + (6K(K - 2) - k(k + 6) \\ &- 6r^2 + 6(k + 2)r)a^2 + (2K^2(K - 3) - 2Kk(k + 3) \\ &+ 4r^3 - 6(K + k + 1)r^2 + (6K(k + 2) + 2k(k + 3))r)a \\ &- k^2 + (2a^2 + 3(2K - 1)a + 6K^2 + 6K + 1)k^2d^2 \\ &+ (4a^3 + 3(4K - 3)a^2 \\ &+ (6K(2K - 1) - k(k + 6) - 6r^2 + 6(k + 2)r)a \\ &+ 4K^3 - 3K^2(k - 1) - Kk(k + 3) \\ &+ 2r^3 - 3(2K + k + 1)r^2 + k(6K + k + 3)r)kd). \end{split}$$

$$\begin{split} If \ 0 &\leq r \leq K, \ then \\ s(a, a + (K+1)d, \dots, a + kd) \\ &= \frac{1}{12k^2}(2a^4 + 6(K-1)a^3 \\ &\quad + \left(6K(K-2) - k(k-6) - 6r^2 + 6(k+2)r\right)a^2 \\ &\quad + \left(2K^2(K-3) - 2Kk(k-3) \\ &\quad + 4r^3 - 6(K+k+1)r^2 + (6K(k+2) - 2k(k-3))r\right)a \\ &\quad - k^2 + \left(2a^2 + 3(2K-1)a + 6K^2 + 6K + 1\right)k^2d^2 \\ &\quad + \left(4a^3 + 3(4K-3)a^2 \\ &\quad + \left(6K(2K-1) - k(k-6) - 6r^2 + 6(k+2)r\right)a \\ &\quad + 4K^3 - 3K^2(k-1) - Kk(k+3) \\ &\quad + 2r^3 - 3(2K+k+1)r^2 + k(6K+k+3)r\right)kd). \end{split}$$

By applying the first formula in Lemma 1, we can obtain the Frobenius number of the almost arithmetic sequence with initial gaps. Here, integers a, d, K, k, r are determined as in Theorem 1.

Theorem 2. Under the same conditions as in Theorem 1, we have

$$g(a, a + (K+1)d, \dots, a + kd) = \begin{cases} \frac{a(a+K-r)}{k} + (a+K)d & \text{if } r > 0\\ \frac{a(a+K-k)}{k} + (a+K)d & \text{if } r = 0\\ \end{cases}$$
$$= \left(\left\lceil \frac{a+K}{k} \right\rceil - 1 \right)a + (a+K)d.$$

Proof. If r > 0, (a + K)/k is not an integer. Then by q = (a + K - r)/k,

$$g(a, a + (K+1)d, \dots, a + kd) = (q+1)a + (qk+r)d - a$$

= $\frac{a(a+K-r)}{k} + (a+K)d$

$$= \left\lfloor \frac{a+K}{k} \right\rfloor a + (a+K)d$$
$$= \left(\left\lceil \frac{a+K}{k} \right\rceil - 1 \right) a + (a+K)d.$$

If r = 0, (a + K)/k = q is an integer. Then

$$g(a, a + (K+1)d, \dots, a + kd) = qa + (qk+r)d - a$$
$$= \frac{a(a+K-k)}{k} + (a+K)d.$$

By applying the second formula in Lemma 1, we have the Sylvester number of the almost arithmetic sequence with initial gaps.

Theorem 3. Under the same conditions as in Theorem 1, if r > K, then

$$n(a, a + (K+1)d, \dots, a + kd) = \frac{a^2 + 2(K-1)a + (a+2K-1)kd + K(K-k-2) - k - r(r-k-2)}{2k}$$

and if
$$0 \le r \le K$$
, then

$$n(a, a + (K+1)d, \dots, a + kd)$$

$$= \frac{a^2 + 2(K-1)a + (a + 2K - 1)kd + K(K - k - 2) + k - r(r - k - 2)}{2k}$$

Proof. If r > K, by the second formula in Lemma 1 together with q = (a + K - r)/k, we have

$$n(a, a + (K+1)d, \dots, a + kd)$$

$$= \frac{q(q+1)}{2}k - K + (q+1)(r-1) + \left(\frac{a-1}{2} + K\right)d - \frac{a-1}{2}$$

$$= \frac{a^2 + 2(K-1)a + (a+2K-1)kd + K(K-k-2) - k - r(r-k-2)}{2k}.$$

Other cases are proved similarly and omitted.

 \Box

2.1. Examples

Consider the sequence 11, 23, 27, 31. Then, a = 11, d = 4, K = 2, k = 5, q = 2 and r = 3. By Theorem 1, we have s(11, 23, 27, 31) = 1149. Indeed,

$$s(11, 23, 27, 31) = 1 + 2 + 3 + 4 + 5 + 6 + 7 + 8 + 9 + 10$$

+ 12 + 13 + 14 + 15 + 16 + 17
+ 18 + 19 + 20 + 21 + 24 + 25 + 26 + 28
+ 29 + 30 + 32 + 35 + 36
+ 37 + 39 + 40 + 41 + 43 + 47 + 48 + 51
+ 52 + 59 + 63 + 70 + 74
= 1149.

Consider the sequence 13, 22, 25, 28. Then, a = 13, d = 3, K = 2, k = 5 and q = 3. By Theorem 1, we have

$$\sum_{i=1}^{a-1} m_i = 22 + 25 + 28 + 44 + 47 + 50 + 53 + 56 + 72 + 75 + 81 + 84$$

= 637,
$$\sum_{i=1}^{a-1} m_i^2 = 22^2 + 25^2 + 28^2 + 44^2 + 47^2 + 50^2 + 53^2 + 56^2 + 72^2 + 75^2$$

+ 81² + 84²
= 38,909

and s(13, 22, 25, 28) = 1192.

Consider the sequence 10, 22, 25, 28, 31, 34, 37, 40. Then, a = 10, d = 3, K = 3, k = 10, q = 1 and r = 3. By Theorem 1, we have

$$\sum_{i=1}^{a-1} m_i = 22 + 25 + 28 + 31 + 34 + 37 + 53 + 56 + 59$$

= 345,

$$\sum_{i=1}^{a-1} m_i^2 = 22^2 + 25^2 + 28^2 + 31^2 + 34^2 + 37^2 + 53^2 + 56^2 + 59^2$$

= 14,805

and s(10, 22, 25, 28, 31, 34, 37, 40) = 576.

2.2. Special patterns

For an integer $a \ge 2$, let us consider the sequence a, a+2, a+3, a+4. So, K = 1, k = 4 and d = 1. Nonnegative integers q and r are determined as

$$a+1 = 4q + r, \quad 0 \le r < 4$$

When r = 2, 3, i.e. $a \equiv 1, 2 \pmod{4}$, by

$$\sum_{i=1}^{a-1} m_i = \frac{a(a-3+4q^2+2r+2q(r+1))}{2}$$
$$= \frac{a(a(a+8)-(r-3)^2)}{8}$$

and

$$\sum_{i=1}^{a-1} m_i^2 = \frac{1}{6} a \left(2a^2 + (8q^3 + 6(r+1)q^2 + 4(3r-5)q + 6r - 21)a + 64q^3 + 12(4r+5)q^2 + (6r^2 + 54r - 4)q + 6r^2 + 6r + 1 \right)$$
$$= \frac{1}{48} a \left(a^4 + 14a^3 - 3(r^2 - 6r - 13)a^2 + 2(r^3 - 18r^2 + 77r - 64)a + (4r^3 - 42r^2 + 104r + 38) \right),$$

we have

$$s(a, a + 2, a + 3, a + 4) = \frac{(a - r + 5)(a^3 + (r + 3)a^2 - 2(r^2 - 8r + 8)a - 4r^2 + 22r + 6)}{96}.$$

When r = 0, i.e. $a \equiv 3 \pmod{4}$, by

$$\sum_{i=1}^{a-1} m_i = \frac{a(a-1+4q^2+2q)}{2}$$
$$= \frac{a(a^2+8a-1)}{8}$$

and

$$\sum_{i=1}^{a-1} m_i^2 = \frac{1}{6} a \left(2a^2 + (8q^3 + 6q^2 - 8q - 3)a + 64q^3 + 60q^2 - 4q + 1 \right)$$
$$= \frac{a(a^4 + 14a^3 + 63a^2 + 40a + 38)}{48},$$

we have

$$s(a, a+2, a+3, a+4) = \frac{(a+1)(a+5)(a^2+2a+6)}{96}.$$

When r = 1, i.e. $a \equiv 0 \pmod{4}$, by

$$\sum_{i=1}^{a-1} m_i = \frac{a(a+(2q+1)^2)}{2}$$
$$= \frac{a(a^2+8a+4)}{8}$$

and

$$\sum_{i=1}^{a-1} m_i^2 = \frac{1}{6} a \left(2a^2 + (8q^3 + 12q^2 + 4q + 3)a + 64q^3 + 108q^2 + 56q + 13 \right) \\ = \frac{a(a^4 + 14a^3 + 78a^2 + 136a + 108)}{48},$$

we have

$$s(a, a+2, a+3, a+4) = \frac{(a+4)(a^3+4a^2+22a+24)}{96}.$$

112

In conclusion, we have the following.

Corollary 1. For $a \ge 2$, we have

$$s(a, a + 2, a + 3, a + 4)$$

$$= \begin{cases} \frac{(a+4)(a^3 + 4a^2 + 22a + 24)}{96} & \text{if } a \equiv 0 \pmod{4}, \\ \frac{(a+3)(a^3 + 5a^2 + 8a + 34)}{96} & \text{if } a \equiv 1 \pmod{4}, \\ \frac{(a+2)(a^3 + 6a^2 + 14a + 36)}{96} & \text{if } a \equiv 2 \pmod{4}, \\ \frac{(a+5)(a^3 + 3a^2 + 8a + 6)}{96} & \text{if } a \equiv 3 \pmod{4}. \end{cases}$$

For example,

$$s(12, 14, 15, 16) = 432,$$

 $s(13, 15, 16, 17) = 530,$
 $s(14, 16, 17, 18) = 692,$
 $s(15, 17, 18, 19) = 870.$

From Theorem 2, we have

$$g(a, a+2, a+3, a+4) = \begin{cases} \frac{a^2 + 4a + 4}{4} & \text{if } a \equiv 0 \pmod{4}, \\ \frac{a^2 + 3a + 4}{4} & \text{if } a \equiv 1 \pmod{4}, \\ \frac{a^2 + 2a + 4}{4} & \text{if } a \equiv 2 \pmod{4}, \\ \frac{a^2 + a + 4}{4} & \text{if } a \equiv 3 \pmod{4}. \end{cases}$$

Note that r = 0, 1, 2, 3 implies that $a \equiv 3, 0, 1, 2 \pmod{4}$, respectively. By using the floor function, we can rewritten as follows.

Corollary 2. For $a \ge 2$, we have

$$g(a, a+2, a+3, a+4) = \left(1 + \left\lfloor \frac{a}{4} \right\rfloor\right)a + 1.$$

From Theorem 3, we have the following.

Corollary 3. For $a \ge 2$, we have

$$n(a, a+2, a+3, a+4) = \begin{cases} \frac{a^2 + 4a + 8}{8} & \text{if } a \equiv 0 \pmod{4}, \\ \frac{a^2 + 4a + 3}{8} & \text{if } a \equiv 1 \pmod{4}, \\ \frac{a^2 + 4a + 4}{8} & \text{if } a \equiv 2 \pmod{4}, \\ \frac{a^2 + 4a + 3}{8} & \text{if } a \equiv 3 \pmod{4}. \end{cases}$$

The sequence a, a + 3, a + 4, a + 5 also satisfies the condition $1 \le K \le (k-1)/2$ as K = 2, k = 5 and d = 1.

Corollary 4. For $a \geq 3$, we have

$$s(a, a + 3, a + 4, a + 5)$$

$$= \begin{cases} \frac{(a+5)(a^3 + 8a^2 + 55a + 90)}{150} & \text{if } a \equiv 0 \pmod{5}, \\ \frac{(a+4)(a^3 + 9a^2 + 35a + 105)}{150} & \text{if } a \equiv 1 \pmod{5}, \\ \frac{(a+3)(a^3 + 10a^2 + 41a + 110)}{150} & \text{if } a \equiv 2 \pmod{5}, \\ \frac{(a+2)(a^3 + 11a^2 + 43a + 105)}{150} & \text{if } a \equiv 3 \pmod{5}, \\ \frac{(a+6)(a^3 + 7a^2 + 41a + 65)}{150} & \text{if } a \equiv 4 \pmod{5}. \end{cases}$$

Corollary 5. For $a \geq 3$, we have

$$g(a, a + 3, a + 4, a + 5) = \left(1 + \left\lfloor\frac{a+1}{5}\right\rfloor\right)a + 2.$$

Corollary 6. For $a \geq 3$, we have

$$n(a, a+3, a+4, a+5) = \begin{cases} \frac{a^2 + 7a + 20}{10} & \text{if } a \equiv 0 \pmod{5}, \\ \frac{a^2 + 7a + 12}{10} & \text{if } a \equiv 1 \pmod{5}, \\ \frac{a^2 + 7a + 12}{10} & \text{if } a \equiv 2 \pmod{5}, \\ \frac{a^2 + 7a + 10}{10} & \text{if } a \equiv 3 \pmod{5}, \\ \frac{a^2 + 7a + 16}{10} & \text{if } a \equiv 4 \pmod{5}. \end{cases}$$

However, the sequence a, a + 4, a + 5, a + 6 does not satisfy the condition $1 \le K \le (k - 1)/2$. In fact, if the gap K becomes bigger compared to k, the situation becomes more complicated. This case is discussed in the following section.

3. Bigger Gaps

When the gap K is bigger, the situation becomes more complicated. Consider the same almost arithmetic sequence $a, a + (K + 1)d, a + (K + 2)d, \ldots, a + kd$ with $a \ge 2$ and gcd(a, d) = 1. In this section, we treat with the case when K > (k - 1)/2. Nevertheless, this case cannot be treated in a unified manner. Cases still need to be divided.

3.1. General case

Assume that $(k-1)/2 < K \le (2k-2)/3$. Nonnegative integers q and r are determined as in (2). We also assume that $q = \lfloor (a+K)/k \rfloor \ge 2$. In addition to these conditions, we shall discuss four cases separately: $r = 0, 1 \le r \le k - K - 1, k - K \le r \le K$ or $K + 1 \le r < k$.

Case 1: When $r = a + K - \lfloor (a + K)/k \rfloor k = 0$, all the elements of Apéry set of $a, a + (K + 1)d, a + (K + 2)d, \ldots, a + kd$ excluding 0 (mod a) can be determined in the following table:

		a + (K+1)d	 a + kd
	2a + (2K + 2)d	 2a + (K+k+1)d	 2a + 2kd
3a + (2k+1)d	 3a + (2K + k + 2)d	 3a + (K + 2k + 1)d	 3a + 3kd
qa + ((q-1)k + 1)d		 [qa + (qk - K)d]	 qa + qkd
	(q+1)a + (qk+k-K+1)d	 (q+1)a + (qk + K + 1)d	

Since any of the terms $2a + (k+1)d, \ldots, 2a + (2K+1)d$ cannot be expressed in terms of $a, a + (K+1)d, a + (K+2)d, \ldots, a + kd$, they cannot be elements of the Apéry set, and hence do not exist in this table. In addition, the term $qa + (qk - K)d = qa + ad \equiv 0 \pmod{a}$ cannot exist in this table, and the terms $(q+1)a + (qk+1)d, \ldots,$ (q+1)a + (qk + k - K)d are also out. Note that by gcd(a, d) = 1,

$$\{(K+1)d, (K+2)d, \dots, kd, (2K+2)d, \dots, (a-1)d, (a+1)d, \dots, (a+K)d, (a+k+1)d, \dots, (a+2K+1)d\} = \{1, 2, \dots, a-1\} \pmod{a}.$$

It is not difficult to see that all the elements in this table can be represented by a + (K+1)d, a + (K+2)d, ..., a + kd and that none of the elements can be represented by subtracting a. Hence,

$$\begin{split} \sum_{i=1}^{a-1} m_i &= \left((1+2+\dots+q)k - K - 2(2K+1-k) - q \right. \\ &+ (q+1)(2K-k+1) \right) a + \left((k+1) + \dots + k + (2K+2) \right. \\ &+ \dots + (a-1) + (a+1) + \dots + (a+K) \\ &+ (a+k+1) + \dots + (a+2K+1) \right) d \\ &= \left(\frac{q(q+1)}{2}k - 5K - 2 + 2k - q + (q+1)(2K-k+1) \right) a \\ &+ \left(\frac{a(a-1)}{2} + (3K-k+1)a \right) d \\ &= \frac{a(a^2 + (6K-k)a + K(5K-7k) + 2k(k-1))}{2k} \\ &+ \left(\frac{a+1}{2} + 3K - k \right) ad \end{split}$$

and

$$\begin{split} \sum_{i=1}^{a-1} m_i^2 &= \left((1^2 + 2^2 + \dots + q^2)k - K - 2^2(2K + 1 - k) \right. \\ &\quad - q^2 + (q+1)^2(2K - k+1))a^2 + ((k+1)^2 + \dots + k^2 \\ &\quad + (2K+2)^2 + \dots + (a-1)^2 + (a+1)^2 + \dots + (a+K)^2 \\ &\quad + (a+k+1)^2 + \dots + (a+2K+1)^2)d^2 \\ &\quad + 2ad \left(\left(\frac{k(k+1)}{2} - \frac{K(K+1)}{2} \right) + 2 \left(\frac{2k(2k+1)}{2} \right) \\ &\quad - \frac{k(k+1)}{2} \right) - 2 \left(\frac{(2K+1)(2K+2)}{2} - \frac{k(k+1)}{2} \right) \\ &\quad + 3 \left(\frac{3k(3k+1)}{2} - \frac{2k(2k+1)}{2} \right) \\ &\quad + \dots + q \left(\frac{qk(qk+1)}{2} - \frac{(qk-q)(qk-q+1)}{2} \right) \\ &\quad - qa + (q+1) \left(\frac{(a+2K+1)(a+2K+2)}{2} \right. \\ &\quad - \frac{(a+k)(a+k+1)}{2} \right) \right) = \frac{a^2}{6k^2} (2a^3 + 3(6K-k)a^2 \\ &\quad + (6K(5K+3k))a + 14K^3 + (21K-53k+12)Kk \\ &\quad + 18k^2(k-1))a^2 + \frac{a}{6} (2a^2 + 3(6K-2k+1)a \\ &\quad + 6K(5K+7k) - 6k^2 - 6k + 13)d^2 \\ &\quad + \frac{ad}{6k} (4a^3 + 3(12K - 3k+1)a^2 + (6K(10K+3k+7)) \\ &\quad - 19k^2 + 9k + 12)a + K(28K^2 + 39K + 12) \\ &\quad - 3(9K^2 + 15K + 4)k - (7K - 6k - 6)k^2), \end{split}$$

by the third formula in Lemma 1, we have

$$s(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{1}{12k^2} (2a^4 + 6(3K - k)a^3 + (30K^2 - k(7k - 12))a^2)$

$$+ (14K^{3} + 12k^{3} - 4(8K + 3)k^{2} + 6K(K + 2)k)a - k^{2} + (2a^{2} + 3(6K - 2k + 1)a + 30K^{2} + 42K - 6k^{2} - 6k + 13)k^{2}d^{2} + (4a^{3} + 3(12K - 4k + 1)a^{2} + (60K^{2} + 42K - 13k^{2} + 6k + 12)a + 6k^{3} - (7K - 6)k^{2} - 3(9K^{2} + 15K + 4)k + K(28K^{2} + 39K + 12))kd).$$

In addition, by the first and second formulae in Lemma 1, we have

$$g(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{a(a+K)}{k} + (a+2K+1)d$

and

$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{a^2 + 2(3K - k)a + K(5K - 7k) + k(2k - 1)}{2k}$
+ $\left(\frac{a+1}{2} + 3K - k\right)d,$

respectively.

Case 2: Assume that $1 \le r \le k - K - 1$. Since k - K - 1 < K, the *q*-th and the (q + 1)th lines are replaced by

$$\underbrace{\frac{qa + ((q-1)k + 1)d \dots qa + (a-1)d}{k-K+r-1}}_{qa + (a+1)d \dots qa + qkd}\underbrace{[gap]}_{K-r}$$

$$\underbrace{(q+1)a + (qk+1)d\dots(q+1)a + (qk+r)d}_{r} [\underbrace{gap}_{k-K}]$$

$$\underbrace{(q+1)a + (a+k+1)d\dots(q+1)a + (a+2K+1)d}_{2K-k+1} [\underbrace{gap}_{k-K-r-1}]$$

from Case 1, respectively. Then, by

$$\begin{split} \sum_{i=1}^{a-1} m_i &= \left(\frac{q(q+1)}{2}k - K - 2(2K+1-k) - q \right. \\ &+ (q+1)(2K-k+1+r)\right)a + \left(\frac{a(a+1)}{2} + (3K-k)a\right)d \\ &= \frac{a(a^2 + (6K-k)a + K(5K-7k) + 2k(k-1) - r(r+4K-3k))}{2k} \\ &+ \left(\frac{a+1}{2} + 3K - k\right)ad \end{split}$$

and

$$\begin{split} \sum_{i=1}^{a-1} m_i^2 &= \bigg(\frac{q(q+1)(2q+1)}{6} k - 9K - 4 + 4k - q^2 \\ &+ (q+1)^2 (2K - k + 1 + r) \bigg) a^2 + \bigg(\frac{(a+1)(2a+1)}{6} \\ &+ (3K - k)a + (K+1)(5K+2) - k(k+1) \bigg) a d^2 \\ &+ 2ad \bigg(- \frac{(K+1)(9K+4)}{2} + k(k+1) \\ &+ \frac{qk(q+1)\big((4q-1)k+3\big)}{12} - qa + (q+1) \\ &\times \bigg(\frac{(a+2K+1)(a+2K+2)}{2} - \frac{(a+k)(a+k+1)}{2} \\ &+ \bigg(qk + \frac{r+1}{2} \bigg) r \bigg) \bigg), \end{split}$$

we have

$$s(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{1}{12k^2}(2a^4 + 6(3K - k)a^3 + (30K^2 - 7k^2 + 12k) - 6(r + 4K - 3k)r)a^2 + (K^2(14K + 6k))$

$$\begin{aligned} &-4Kk(8k-3)+12k^2(k-1)+4r^3+6(K-2k)r^2\\ &-(24K^2-6Kk-8k^2+12k)r\big)a-k^2\\ &+(2a^2+3(6K-2k+1)a+6K(5K+7)\\ &-(6k^2+6k-13)\big)k^2d^2+(4a^3+3(12K-4k+1)a^2\\ &+(6K(10K+7k)-(13k^2-6k-12)-6r(r+4K-3k))a\\ &+28K^3-K^2(27k-39)-K(7k^2+45k-12)\\ &+6(k-1)k(k+2)\big)kd).\end{aligned}$$

This also holds for r = 0.

We also have

$$g(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{a(a+K-r)}{k} + (a+2K+1)d$

and

$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

$$= \frac{a^2 + 2(3K - k)a + K(5K - 7k) + k(2k - 1) - r(r + 4K - 3k)}{2k}$$

$$+ \left(\frac{a+1}{2} + 3K - k\right)d.$$

Case 3: Assume that $k - K \le r \le K$. Then, the qth, the (q + 1)th and (q + 2)th lines are replaced by

$$\underbrace{\begin{array}{c} \underbrace{qa + ((q-1)k+1)d \cdots qa + (a-1)d}_{k-K+r-1} \underbrace{[gap]}_{1} \\ \underbrace{qa + (a+1)d \cdots qa + qkd}_{K-r} \\ \underbrace{(q+1)a + (qk+1)d \cdots (q+1)a + (qk+r)d}_{r} \underbrace{[gap]}_{k-K} \\ \underbrace{(q+1)a + (a+k+1)d \cdots (q+1)a + (a+k+K-r)d}_{K-r} \\ \underbrace{(q+1)a + (a+k+1)d \cdots (q+1)a + (a+k+K-r)d}_{K-r} \\ \end{array}}$$

and

$$\underbrace{(q+2)a + ((q+1)k+1)d\cdots(q+2)a + (a+2K+1)d}_{K-k+r+1} \underbrace{[gap]}_{2k-K-r-1},$$

respectively. Then, by

$$\begin{split} \sum_{i=1}^{a-1} m_i &= \left(\frac{q(q+1)}{2}k - K - 2(2K+1-k) - q + (q+1)K\right. \\ &+ (q+2)(K-k+r+1)\right)a + \left(\frac{a(a+1)}{2} + (3K-k)a\right)d\\ &= \frac{a\left(a^2 + (6K-k)a + (K-r)(5K-5k+r)\right)}{2k} \\ &+ \left(\frac{a+1}{2} + 3K - k\right)ad \end{split}$$

$$\begin{split} \sum_{i=1}^{a-1} m_i^2 &= \bigg(\frac{q(q+1)(2q+1)}{6} k - 9K - 4 + 4k - q^2 \\ &+ (q+1)^2 K + (q+2)^2 (K-k+r+1) \bigg) a^2 \\ &+ \bigg(\frac{(a+1)(2a+1)}{6} + (3K-k)a + (K+1)(5K+2) \\ &- k(k+1) \bigg) ad^2 + 2ad \bigg(-\frac{(K+1)(9K+4)}{2} + k(k+1) \\ &+ \frac{qk(q+1)\big((4q-1)k+3\big)}{12} - qa + (q+1) \\ &+ \frac{qk(q+1)\big((4q-1)k+3\big)}{2} - qa + (q+1) \\ &\times \bigg(\frac{(a+K+k-r)(a+K+k-r+1)}{2} \\ &- \frac{(a+k)(a+k+1)}{2} + \bigg(qk + \frac{r+1}{2} \bigg) r \bigg) \\ &+ (q+2)\left((q+1)k(K-k+r+1) \\ &+ \frac{(K-k+r+1)(K-k+r+1)}{2} \bigg) \bigg), \end{split}$$

we have

$$\begin{split} s \big(a, a + (K+1)d, a + (K+2)d, \dots, a + kd\big) \\ &= \frac{1}{12k^2} \bigg(2a^4 + 6(3K-k)a^3 + \big(30K^2 + 12Kk \\ &- 19k^2 + 24k - 6(r+4K-5k)r\big)a^2 \\ &+ \big(K^2(14K+18k) - 8Kk(4k-3) + 4r^3 + 6(K-4k)r^2 \\ &- (24K^2 - 6Kk - 32k^2 + 24k)r\big)a - k^2 \\ &+ \big(2a^2 + 3(6K-2k+1)a + 6K(5K+7) \\ &- 6k^2 - 6k + 13\big)k^2d^2 + \big(4a^3 + 3(12K-4k+1)a^2 \\ &+ \big(6K(10K+2k+7) - (25k^2 - 18k - 12) \\ &- 6r(r+4K-5k)\big)a + 28K^3 - K^2(9k-39) \\ &- K(19k^2 + 15k - 12) + 2r^3 - 3r^2(2K+3k+1) \\ &+ r\big(-K(24K-18k+36) + (19k^2 + 15k - 12)\big)\big)kd\bigg). \end{split}$$

We also have

$$g(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$
$$= \frac{a(a+K-r+k)}{k} + (a+2K+1)d$$

$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{a^2 + 2(3K - k)a + (K - r)(5K - 5k + r) + k}{2k}$
+ $\left(\frac{a+1}{2} + 3K - k\right)d.$

Case 4: Finally, assume that $K + 1 \leq r < k$. Then, the *q*th line consists of no gaps. The (q+1)th and (q+2)th lines are replaced by

$$\underbrace{(q+1)a + (qk+1)d\dots(q+1)a + (qk+r-K-1)d}_{r-K-1}\underbrace{[\operatorname{gap}]}_{1}$$

$$\underbrace{(q+1)a + (a+1)d\dots(q+1)a + (a+K)d}_{K}\underbrace{[\operatorname{gap}]}_{k-r}$$

and

$$\underbrace{[\text{gap}]}_{r-K} \underbrace{(q+2)a + (a+k+1)d\dots(q+2)a + (a+2K+1)d}_{2K-k+1}}_{2K-k+1}$$

respectively. Then, by

$$\sum_{i=1}^{a-1} m_i = \left(\frac{q(q+1)}{2}k - K - 2(2K+1-k) + (q+1)(r-1) + (q+2)(2K-k+1)\right)a + \left(\frac{a(a+1)}{2} + (3K-k)a\right)d$$
$$= \frac{a(a^2 + (6K-k)a + K(5K-3k) - 2k - r(r+4K-3k))}{2k}$$
$$+ \left(\frac{a+1}{2} + 3K - k\right)ad$$

$$\sum_{i=1}^{a-1} m_i^2 = \left(\frac{q(q+1)(2q+1)}{6}k - 9K - 4 + 4k + (q+1)^2(r-1) + (q+2)^2(2K-k+1)\right)a^2 + \left(\frac{(a+1)(2a+1)}{6} + (3K-k)a + (K+1)(5K+2) - k(k+1)\right)ad^2$$

T. Komatsu

$$\begin{split} &+ 2ad \bigg(-\frac{(K+1)(9K+4)}{2} + k(k+1) \\ &+ \frac{qk(q+1)\big((4q-1)k+3\big)}{12} + (q+1) \\ &\times \bigg((r-K-1)\left(qk + \frac{r-K}{2}\right) + K\left(a + \frac{K+1}{2}\right) \bigg) \\ &+ (q+2)\left(\frac{(a+2K+1)(a+2K+2)}{2} \\ &- \frac{(a+k)(a+k+1)}{2}\right) \bigg), \end{split}$$

we have

$$\begin{split} s \left(a, a + (K+1)d, a + (K+2)d, \dots, a + kd\right) \\ &= \frac{1}{12k^2} (2a^4 + 6(3K-k)a^3 + (30K^2 + 24Kk) \\ &- 19k^2 + 12k - 6(r + 4K - 3k)r)a^2 \\ &+ (K^2(14K + 30k) - 4Kk(5k - 3) \\ &+ 12k^2(k - 1) + 4r^3 + 6(K - 2k)r^2 \\ &- (24K^2 + 18Kk - 20k^2 + 12k)r)a - k^2 \\ &+ (2a^2 + 3(6K - 2k + 1)a + 6K(5K + 7) \\ &- 6k^2 - 6k + 13)k^2d^2 + (4a^3 + 3(12K - 4k + 1)a^2 \\ &+ (6K(10K + 4k + 7) - (25k^2 - 6k - 12) - 6r(r + 4K - 3k))a \\ &+ 28K^3 - 3K^2(k - 13) - K(7k^2 + 9k - 12) + 2r^3 \\ &- 3r^2(2K + k + 1) + r(-6K(4K - k + 6) \\ &+ 7k^2 + 9k - 12))kd). \end{split}$$

We also have

$$g(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$
$$= \frac{a(a+K-r+k)}{k} + (a+2K+1)d$$

and

$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

= $\frac{a^2 + 2(3K - k)a + K(5K - 3k) - k - r(r + 4K - 3k)}{2k}$
+ $\left(\frac{a+1}{2} + 3K - k\right)d.$

In conclusion, we have the Sylvester sums.

Theorem 4. Let a and d be positive integers with $a \ge 2$ and gcd(a,d) = 1 K and k be positive integers with $(k-1)/2 < K \le (2k-2)/3$, and let $r = a + K - \lfloor (a+K)/k \rfloor k$. Assume that $\lfloor (a+K)/k \rfloor \ge 2$. If $0 \le r \le k - K - 1$, then

$$\begin{split} s \big(a, a + (K+1)d, a + (K+2)d, \dots, a + kd \big) \\ &= \frac{1}{12k^2} \big(2a^4 + 6(3K-k)a^3 + \big(30K^2 - 7k^2 + 12k \\ &- 6(r+4K-3k)r \big)a^2 + \big(K^2(14K+6k) \\ &- 4Kk(8k-3) + 12k^2(k-1) + 4r^3 + 6(K-2k)r^2 \\ &- (24K^2 - 6Kk - 8k^2 + 12k)r \big)a - k^2 \\ &+ \big(2a^2 + 3(6K-2k+1)a + 6K(5K+7) \\ &- (6k^2 + 6k - 13) \big)k^2d^2 + \big(4a^3 + 3(12K - 4k + 1)a^2 \\ &+ \big(6K(10K+7k) - (13k^2 - 6k - 12) - 6r(r+4K-3k) \big)a \\ &+ 28K^3 - K^2(27k - 39) - K(7k^2 + 45k - 12) \\ &+ 6(k-1)k(k+2) \big)kd \big). \end{split}$$

$$\begin{split} &If \ k - K \leq r \leq K, \ then \\ &s \big(a, a + (K+1)d, a + (K+2)d, \dots, a + kd \big) \\ &= \frac{1}{12k^2} \big(2a^4 + 6(3K-k)a^3 + \big(30K^2 + 12Kk - 19k^2 \\ &+ 24k - 6(r + 4K - 5k)r \big)a^2 + \big(K^2(14K + 18k) \\ &- 8Kk(4k-3) + 4r^3 + 6(K - 4k)r^2 \\ &- (24K^2 - 6Kk - 32k^2 + 24k)r \big)a - k^2 \\ &+ \big(2a^2 + 3(6K - 2k + 1)a + 6K(5K + 7) - 6k^2 - 6k + 13 \big)k^2d^2 \\ &+ \big(4a^3 + 3(12K - 4k + 1)a^2 + \big(6K(10K + 2k + 7) \\ &- (25k^2 - 18k - 12) - 6r(r + 4K - 5k) \big)a \\ &+ 28K^3 - K^2(9k - 39) - K(19k^2 + 15k - 12) \\ &+ 2r^3 - 3r^2(2K + 3k + 1) + r \big(-K(24K \\ &- 18k + 36) + \big(19k^2 + 15k - 12 \big) \big) \big)kd \big). \end{split}$$

If $K + 1 \leq r < k$, then

$$\begin{split} s \big(a, a + (K+1)d, a + (K+2)d, \dots, a + kd \big) \\ &= \frac{1}{12k^2} \big(2a^4 + 6(3K-k)a^3 + \big(30K^2 + 24Kk \\ &- 19k^2 + 12k - 6(r+4K-3k)r \big)a^2 \\ &+ \big(K^2(14K+30k) - 4Kk(5k-3) + 12k^2(k-1) + 4r^3 \\ &+ 6(K-2k)r^2 - (24K^2 + 18Kk - 20k^2 + 12k)r \big)a - k^2 \\ &+ \big(2a^2 + 3(6K-2k+1)a + 6K(5K+7) - 6k^2 \\ &- 6k + 13) \big)k^2d^2 + \big(4a^3 + 3(12K-4k+1)a^2 \\ &+ \big(6K(10K+4k+7) - (25k^2 - 6k - 12) \big) \end{split}$$

$$- 6r(r + 4K - 3k))a + 28K^3 - 3K^2(k - 13)$$

- $K(7k^2 + 9k - 12) + 2r^3 - 3r^2(2K + k + 1)$
+ $r(-6K(4K - k + 6) + 7k^2 + 9k - 12))kd).$

Concerning Frobenius and Sylvester numbers, we have the following. a, d, K, k, r are determined as in Theorem 4.

Theorem 5. Under the same conditions as in Theorem 4, we have $g(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$

$$= \begin{cases} \frac{a(a+K-r)}{k} + (a+2K+1)d & \text{if } 0 \le r \le k-K-1\\ \frac{a(a+K-r+k)}{k} + (a+2K+1)d & \text{if } k-K \le r < k. \end{cases}$$

Theorem 6. Under the same conditions as in Theorem 4, if $0 \le r \le k - K - 1$, then

$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

$$= \frac{a^2 + 2(3K - k)a + K(5K - 7k) + k(2k - 1) - r(r + 4K - 3k)}{2k}$$

$$+ \left(\frac{1}{2} + 3K - k\right) d.$$
If $k - K \le r \le K$, then
$$n(a, a + (K+1)d, a + (K+2)d, \dots, a + kd)$$

$$= \frac{a^2 + 2(3K - k)a + (K - r)(5K - 5k + r) + k}{2k}$$

$$+ \left(\frac{a+1}{2} + 3K - k\right) d.$$
If $K + 1 \le -k$ but

If
$$K + 1 \le r < k$$
, then
 $n(a, a + (K + 1)d, a + (K + 2)d, \dots, a + kd)$
 $= \frac{a^2 + 2(3K - k)a + K(5K - 3k) - k - r(r + 4K - 3k)}{2k}$
 $+ \left(\frac{a+1}{2} + 3K - k\right)d.$
3.2. Special patterns

For an integer $a \ge 2$, let us consider the sequence a, a + 4, a + 5, a+6. Then, we apply the Theorem 4 as K = 3, k = 6 and d = 1, and nonnegative integers q and r are determined by a + 3 = 6q + r with $0 \le r \le 5$. $q \ge 2$ implies that $a \ge 9$. When r = 0, 1, 2, i.e. $a \equiv 3, 4, 5 \pmod{6}$, we have

$$s(a, a + 4, a + 5, a + 6) = \frac{1}{216} (a^4 + 21a^3 - 3(r^2 - 6r - 66)a^2 + (2r^3 - 45r^2 + 162r + 927)a + 3(2r^3 - 39r^2 + 78r + 495)).$$

When r = 3, i.e. $a \equiv 0 \pmod{6}$, we have

$$s(a, a + 4, a + 5, a + 6) = \frac{1}{216} (a^4 + 21a^3 - 3(r^2 - 18r - 42)a^2 + (2r^3 - 81r^2 + 774r - 153)a + 3(2r^3 - 75r^2 + 762r - 793)).$$

When r = 4, 5, i.e. $a \equiv 1, 2 \pmod{6}$, we have

$$s(a, a + 4, a + 5, a + 6) = \frac{1}{216} (a^4 + 21a^3 - 3(r^2 - 6r - 66)a^2 + (2r^3 - 45r^2 + 162r + 1143)a + 3(2r^3 - 39r^2 + 78r + 999)).$$

The final result also holds for q = 1, i.e. a = 8.

Corollary 7. For $a \ge 8$, we have

$$s(a, a + 4, a + 5, a + 6)$$

$$= \begin{cases} \frac{a^4 + 21a^3 + 261a^2 + 1494a + 2808}{216} & \text{if } a \equiv 0 \pmod{6}, \\ \frac{a^4 + 21a^3 + 222a^2 + 1199a + 2445}{216} & \text{if } a \equiv 1 \pmod{6}, \\ \frac{a^4 + 21a^3 + 213a^2 + 1078a + 1992}{216} & \text{if } a \equiv 2 \pmod{6}, \\ \frac{(a + 3)(a^3 + 18a^2 + 144a + 495)}{216} & \text{if } a \equiv 3 \pmod{6}, \\ \frac{a^4 + 21a^3 + 213a^2 + 1046a + 1608}{216} & \text{if } a \equiv 4 \pmod{6}, \\ \frac{a^4 + 21a^3 + 222a^2 + 1087a + 1533}{216} & \text{if } a \equiv 5 \pmod{6}. \end{cases}$$

By applying Theorem 5 as K = 3, k = 6 and d = 1, for r = 0, 1, 2, i.e. $a \equiv 3, 4, 5 \pmod{6}$, we have

$$g(a, a+4, a+5, a+6) = \frac{a^2 + (9-r)a + 42}{6},$$

and for r = 3, 4, 5, i.e. $a \equiv 0, 1, 2 \pmod{6}$, we have

$$g(a, a + 4, a + 5, a + 6) = \frac{a^2 + (15 - r)a + 42}{6}$$

The case for a = 8 is also valid. We can conclude that

$$g(a, a+4, a+5, a+6) = \begin{cases} \frac{a^2 + 12a + 42}{6} & \text{if } a \equiv 0 \pmod{6}, \\ \frac{a^2 + 11a + 42}{6} & \text{if } a \equiv 1 \pmod{6}, \\ \frac{a^2 + 10a + 42}{6} & \text{if } a \equiv 2 \pmod{6}, \\ \frac{a^2 + 9a + 42}{6} & \text{if } a \equiv 3 \pmod{6}, \\ \frac{a^2 + 8a + 42}{6} & \text{if } a \equiv 4 \pmod{6}, \\ \frac{a^2 + 7a + 42}{6} & \text{if } a \equiv 5 \pmod{6}. \end{cases}$$

The coefficient of a can be unified by using the floor function.

Corollary 8. For $a \ge 8$, we have

$$g(a, a+4, a+5, a+6) = \left(2 + \left\lfloor \frac{a}{6} \right\rfloor\right)a + 7.$$

By applying Theorem 6, for r = 0, 1, 2, i.e. $a \equiv 3, 4, 5 \pmod{6}$, we have

$$n(a, a+4, a+5, a+6) = \frac{a^2 + 12a - (r+3)(r-9)}{12},$$

for r = 3, i.e. $a \equiv 0 \pmod{6}$, we have

$$n(a, a+4, a+5, a+6) = \frac{a^2 + 12a - r^2 + 18r + 3}{12},$$

for r = 4, 5, i.e. $a \equiv 1, 2 \pmod{6}$, we have

$$n(a, a+4, a+5, a+6) = \frac{a^2 + 12a - r^2 + 6r + 27}{12}$$

Finally, we can check manually that the result is also valid for a = 4, 5, 6.

Corollary 9. For a = 4, 5, 6 and $a \ge 8$, we have

$$n(a, a+4, a+5, a+6) = \begin{cases} \frac{a^2 + 12a + 48}{12} & \text{if } a \equiv 0 \pmod{6}, \\ \frac{a^2 + 12a + 35}{12} & \text{if } a \equiv 1 \pmod{6}, \\ \frac{a^2 + 12a + 32}{12} & \text{if } a \equiv 2 \pmod{6}, \\ \frac{a^2 + 12a + 27}{12} & \text{if } a \equiv 3 \pmod{6}, \\ \frac{a^2 + 12a + 32}{12} & \text{if } a \equiv 4 \pmod{6}, \\ \frac{a^2 + 12a + 35}{12} & \text{if } a \equiv 5 \pmod{6}. \end{cases}$$

Remark. The coefficient of the constant term can be unified by using the floor function.

$$\begin{split} n(a, a+4, a+5, a+6) \\ &= \frac{a^2}{12} + a + \frac{1}{12 \cdot 15} \bigg(720 + 209a - 14 \left\lfloor \frac{a}{6} \right\rfloor - 164 \left\lfloor \frac{a+1}{6} \right\rfloor \\ &- 134 \left\lfloor \frac{a+2}{6} \right\rfloor - 284 \left\lfloor \frac{a+3}{6} \right\rfloor - 254 \left\lfloor \frac{a+4}{6} \right\rfloor - 404 \left\lfloor \frac{a+5}{6} \right\rfloor \bigg) \\ &= \frac{a^2}{12} + \frac{389}{180}a + 4 - \frac{1}{90} \bigg(7 \left\lfloor \frac{a}{6} \right\rfloor + 82 \left\lfloor \frac{a+1}{6} \right\rfloor + 67 \left\lfloor \frac{a+2}{6} \right\rfloor \\ &+ 142 \left\lfloor \frac{a+3}{6} \right\rfloor + 127 \left\lfloor \frac{a+4}{6} \right\rfloor + 202 \left\lfloor \frac{a+5}{6} \right\rfloor \bigg). \end{split}$$

For an integer $a \ge 2$, let us consider the sequence a, a + 5, a + 6, a + 7. Then, K = 4, k = 7 and d = 1 in Theorem 4. Nonnegative integers q and r are determined by a + 4 = 7q + r with $0 \le r \le 6$. $q \ge 2$ implies that $a \ge 10$. When r = 0, 1, 2, i.e. $a \equiv 3, 4, 5$ (mod 7), we have

$$s(a, a + 5, a + 6, a + 7) = \frac{1}{294} (a^4 + 29a^3 - (3r^2 - 15r - 380)a^2 + (2r^3 - 51r^2 + 151r + 2296)r + 7(r^3 - 24r^2 + 17r + 672)).$$

When r = 3, 4, i.e. $a \equiv 6, 0 \pmod{7}$, we have

$$s(a, a + 5, a + 6, a + 7) = \frac{1}{294} \left(a^4 + 29a^3 - (3r^2 - 57r - 296)a^2 + (2r^3 - 93r^2 + 991r + 784)r + 7(r^3 - 45r^2 + 500r - 210) \right).$$

When r = 5, 6, i.e. $a \equiv 1, 2 \pmod{7}$, we have

$$s(a, a + 5, a + 6, a + 7) = \frac{1}{294} (a^4 + 29a^3 - (3r^2 - 15r - 422)a^2 + (2r^3 - 51r^2 + 109r + 3346)a + 7(r^3 - 24r^2 + 17r + 1386)).$$

We can check manually that the result is also valid for a = 5, 6, 7. Corollary 10. For a = 5, 6, 7 and $a \ge 10$, we have

$$s(a, a + 5, a + 6, a + 7)$$

$$= \begin{cases} \frac{a^4 + 29a^3 + 476a^2 + 3388a + 7938}{294} & \text{if } a \equiv 0 \pmod{7}, \\ \frac{a^4 + 29a^3 + 422a^2 + 2866a + 6972}{294} & \text{if } a \equiv 1 \pmod{7}, \\ \frac{a^4 + 29a^3 + 404a^2 + 2596a + 5880}{294} & \text{if } a \equiv 2 \pmod{7}, \\ \frac{a^4 + 29a^3 + 380a^2 + 2296a + 4704}{294} & \text{if } a \equiv 3 \pmod{7}, \\ \frac{a^4 + 29a^3 + 392a^2 + 2398a + 4662}{294} & \text{if } a \equiv 4 \pmod{7}, \\ \frac{a^4 + 29a^3 + 398a^2 + 2410a + 4326}{294} & \text{if } a \equiv 5 \pmod{7}, \\ \frac{a^4 + 29a^3 + 440a^2 + 2974a + 6384}{294} & \text{if } a \equiv 6 \pmod{7}. \end{cases}$$

By applying Theorem 5 as K = 4, k = 7 and d = 1, for r = 0, 1, 2, i.e. $a \equiv 3, 4, 5 \pmod{7}$, we have

$$g(a, a + 5, a + 6, a + 7) = \frac{a^2 + (11 - r)a + 63}{7},$$

and for r = 3, 4, 5, 6, i.e. $a \equiv 6, 0, 1, 2 \pmod{7}$, we have

$$g(a, a+5, a+6, a+7) = \frac{a^2 + (18-r)a + 63}{7}$$

The cases for a = 5, 6, 7, 8 are also valid. The coefficient of a can be unified by using the floor function.

Corollary 11. For a = 5, 6, 7, 8 and $a \ge 10$, we have

$$g(a, a+5, a+6, a+7) = \left(2 + \left\lfloor \frac{a+1}{7} \right\rfloor\right)a + 9.$$

By applying Theorem 6, for r = 0, 1, 2, i.e. $a \equiv 3, 4, 5 \pmod{7}$, we have

$$n(a, a+5, a+6, a+7) = \frac{a^2 + 17a - r^2 + 5r + 52}{14},$$

for r = 3, 4, i.e. $a \equiv 6, 0 \pmod{7}$, we have

$$n(a, a+5, a+6, a+7) = \frac{a^2 + 17a - r^2 + 19r + 24}{14},$$

for r = 5, 6, i.e. $a \equiv 1, 2 \pmod{7}$, we have

$$n(a, a+5, a+6, a+7) = \frac{a^2 + 17a - r^2 + 5r + 66}{14}$$

Finally, we can check manually that the result is also valid for a = 5, 6, 7.

Corollary 12. For a = 5, 6, 7 and $a \ge 10$, we have

$$n(a, a+4, a+5, a+6) = \begin{cases} \frac{a^2 + 17a + 84}{14} & \text{if } a \equiv 0 \pmod{7}, \\ \frac{a^2 + 17a + 66}{14} & \text{if } a \equiv 1 \pmod{7}, \\ \frac{a^2 + 17a + 60}{14} & \text{if } a \equiv 2 \pmod{7}, \\ \frac{a^2 + 17a + 52}{14} & \text{if } a \equiv 3 \pmod{7}, \\ \frac{a^2 + 17a + 56}{14} & \text{if } a \equiv 4 \pmod{7}, \\ \frac{a^2 + 17a + 56}{14} & \text{if } a \equiv 5 \pmod{7}, \\ \frac{a^2 + 17a + 58}{14} & \text{if } a \equiv 5 \pmod{7}, \\ \frac{a^2 + 17a + 72}{14} & \text{if } a \equiv 6 \pmod{7}. \end{cases}$$

4. Comments

Similarly, we can consider the sequence $a, a + (K + 1)d, a + (K + 2)d, \ldots, a + kd$ when $(2k - 2)/3 < K \leq (3k - 3)/4$. Then, as a special case, we can get Frobenius number, Sylvester number and sum for a, a + 6, a + 7, a + 8 and so on. After that, we may continue to consider the cases $(3k - 3)/4 < K \leq (4k - 4)/5, (4k - 4)/5 < K \leq (5k - 5)/6, \ldots$ However, the situation becomes more and more complicated. Is there any more convenient method to find their Sylvester sums?

The approaches in Refs. 30 and 31 may apply to any almost arithmetic sequence, but they both have an extra burden: they require the pre-computation of a couple of constants depending on the sequence. Particularizing their results to ours would be of some interest.

Acknowledgments

The author thanks the anonymous referees for their valuable comments which helped improve the manuscript.

References

- J. L. Ramírez Alfonsín, *The Diophantine Frobenius Problem*. Vol. 30, Oxford Lecture Series in Mathematics and its Applications. Oxford University Press, Oxford (2005).
- E. S. Selmer, On the linear Diophantine problem of Frobenius, J. Reine Angew. Math. 293/294, 1–17 (1977).
- A. Tripathi, On sums of positive integers that are not of the form ax + by, Am. Math. Monthly. 115(4), 363-364 (2008).
- J. J. Sylvester, Mathematical questions with their solutions, *Educa*tional Times. 41(21) (1884).
- J. J. Sylvester, On subvariants, i.e. semi-invariants to binary quantics of an unlimited order, Am. J. Math. 579–136 (1882).
- 6. A. Tripathi, The number of solutions to ax + by = n, Fibonacci Quart. **38**(4), 290–293 (2000).
- T. C. Brown and P. J.-S. Shiue, A remark related to the Frobenius problem, *Fibonacci Quart.* 31(1), 32–36 (1993).
- 8. L. G. Fel and B. Rubinstein, Power sums related to semigroups $s(d_1, d_2, d_3)$, Semigroup Forum. 74, 93–98 (2007).

- L. G. Fel, T. Komatsu, and A. I. Suriajaya, A sum of negative degrees of the gaps values in 2 and 3-generated numerical semigroups, Ann. Math. Inform. 52, 85–95 (2020).
- 10. P. Moree, Numerical semigroups, cyclotomic polynomials, and Bernoulli numbers, *Am. Math. Monthly.* **121**(10), 890–902 (2014).
- H. J. H. Tuenter, The Frobenius problem, sums of powers of integers, and recurrences for the Bernoulli numbers, J. Number Theory. 117(2), 376–386 (2006).
- 12. Ø. J. Rødseth, A note on Brown and Shiue's paper on a remark related to the Frobenius problem, *Fibonacci Quart.* **32**(5), 407–408 (1994).
- F. Curtis, On formulas for the Frobenius number of a numerical semigroup, Math. Scand. 67(2), 190–192 (1990).
- S. M. Johnson, A linear diophantine problem, Canadian J. Math. 12, 390–398 (1960).
- M. Beck, I. M. Gessel, and T. Komatsu, The polynomial part of a restricted partition function related to the Frobenius problem, *Elec*tron. J. Combin. 8(1), Note 7, 5 p. (2001).
- 16. T. Komatsu, On the number of solutions of the Diophantine equation of Frobenius—General case, *Math. Commun.* 8(2), 195–206 (2003).
- A. Tripathi, Formulae for the Frobenius number in three variables, J. Number Theory. 170, 368–389 (2017).
- A. Brauer, On a problem of partitions, Am. J. Math. 64, 299–312 (1942).
- D. C. Ong and V. Ponomarenko, The Frobenius number of geometric sequences, *Integers.* 8, A33, 3 p. (2008).
- J. B. Roberts, Note on linear forms, Proc. Am. Math. Soc. 7, 465–469 (1956).
- T. Komatsu, Sylvester sums on the Frobenius set in arithmetic progression. In F. Yilmaz et al. (eds.), Mathematical Methods for Engineering Applications, Springer Proceedings in Mathematics & Statistics, Vol. 384, Springer, Cham. (May 2022), pp. 1–23.
- A. Brauer and J. E. Shockley, On a problem of Frobenius, J. Reine Angew. Math. 211, 215–220 (1962).
- P. Punyani and A. Tripathi, On changes in the Frobenius and Sylvester numbers, *Integers.* 18B, Paper No. A8, 12 p. (2018).
- A. Tripathi, On a special case of the Frobenius problem, J. Integer Seq. 20(7), Art. 17.7.2, 12 p. (2017).
- T. Komatsu and Y. Zhang, Weighted Sylvester sums on the Frobenius set, *Irish Math. Soc. Bull.* (87), 35–43 (2021).
- 26. T. Komatsu and Y. Zhang, Weighted Sylvester sums on the Frobenius set in more variables, *Kyushu J. Math.* **76**(1), 163–175 (2022).

- 27. T. Komatsu, Sylvester power and weighted sums on the Frobenius set in arithmetic progression, *Discrete Appl. Math.* **315**, 110–126 (2022).
- A. L. Dulmage and N. S. Mendelsohn, Gaps in the exponent set of primitive matrices, *Illinois J. Math.* 8, 642–656 (1964).
- E. Siering, Über lineare Formen und ein Problem von Frobenius. I, J. Reine Angew. Math. 271, 177–202 (1974).
- Ø. J. Rødseth, On a linear Diophantine problem of Frobenius. II, J. Reine Angew. Math. 307/308, 431–440 (1979).
- D. P. Patil and B. Singh, Generators for the derivation modules and the relation ideals of certain curves, *Manuscripta Math.* 68(3), 327–335 (1990).
- 32. J. L. Ramírez Alfonsín and Ø. J. Rødseth, Numerical semigroups: Apéry sets and Hilbert series, *Semigroup Forum*. **79**(2), 323–340 (2009).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0006

Chapter 6

Arithmetic Properties of Minimal Excludants of Partitions of Integers

Nipen Saikia

Department of Mathematics, Rajiv Gandhi University, Rono Hills, Doimukh, Arunachal Pradesh, India nipennak@yahoo.com

Andrews and Newman⁵ defined the minimal excludant (resp. minimal odd exludant) of the partition ξ of an integer $n \ge 1$ as the smallest positive integer (resp. smallest odd positive integer) which is not a part of the partition ξ and is denoted by $Mex(\xi)$ (resp. and $Moex(\xi)$). Some congruence and other divisibility properties of the following two arithmetic functions are discussed:

$$\sigma_{\mathsf{Mex}}(n) = \sum_{\xi \in \mathcal{S}} \mathsf{Mex}(\xi) \quad \text{and} \quad \sigma_{\mathsf{Moex}}(n) = \sum_{\xi \in \mathcal{S}} \mathsf{Moex}(\xi).$$

1. Introduction

Throughout the chapter, define the q-series

$$(\beta;q)_{\infty} := \prod_{j \ge 0} (1 - \beta q^j), \quad |q| < 1,$$
 (1)

where β and q are complex numbers. For brevity one often writes

$$(\beta_1; q)_{\infty}(\beta_2; q)_{\infty}(\beta_3; q)_{\infty} \cdots (\beta_m; q)_{\infty} = (\beta_1, \beta_2, \beta_3, \dots, \beta_m; q)_{\infty}.$$
(2)

A partition of a positive integer n is a sequence of integers $\xi_1 \geq \xi_2 \geq \cdots \geq \xi_j \geq 1$ such that $\sum_{j=1}^k \xi_j = n$. The integers ξ_j are called parts or summands of the partition ξ . If $\lambda(n)$ denotes the number of partitions of the positive integer n, then the generating function of $\lambda(n)$ (due to Euler) is given by

$$\sum_{n\geq 0} \lambda(n)q^n = \frac{1}{(q;q)_{\infty}} = 1 + q + 2q^2 + 3q^3 + 5q^4 + 7q^5 + \cdots$$
 (3)

Thus, $\lambda(4) = 5$ with the partitions of 5 given by

$$4, \quad 3+1, \quad 2+2, \quad 2+1+1, \quad 1+1+1+1.$$

To satisfy the generating function, it is assumed that $\lambda(0) = 1$. Euler also considered two restricted partition functions $\lambda_d(n)$ (resp. $\lambda_o(n)$) which counts the number of partitions of n into distinct (resp. odd) parts, and proved the famous theorem, $\lambda_d(n) = \lambda_o(n)$. In terms of generating functions or q-series, Euler's theorem is stated as

$$\sum_{n \ge 0} \lambda_d(n) q^n = (-q; q)_{\infty} = \frac{1}{(q; q^2)_{\infty}} = \sum_{n \ge 0} \lambda_o(n) q^n.$$
(4)

Euler also gave the following recursion formula for the partition function $\lambda(n)$: If $\lambda(0) = 1$ and $\lambda(n) = 0$ if n < 1, then for $n \ge 1$,

$$\lambda(n) = \lambda(n-1) + \lambda(n-2) - \lambda(n-5) - \lambda(n-7) \cdots .$$
 (5)

Using Euler's recursion formula (5), MacMahon calculated the values of $\lambda(n)$ upto n = 200. By examining the values of $\lambda(n)$ calculated by MacMahon, Ramanujan¹ offered the following congruences for $\lambda(n)$:

$$\lambda(5n+4) \equiv 0 \pmod{5},$$

$$\lambda(7n+5) \equiv 0 \pmod{7},$$

and

$$\lambda(11n+6) \equiv 0 \pmod{11}.$$

Ramanujan published two more papers^{2,3} on $\lambda(n)$ to establish his congruences. Ramanujan's congruences of $\lambda(n)$ motivated mathematicians to investigate different restricted partition functions or

functions related to $\lambda(n)$ in search of analogous results. One such interesting function is the minimal excludant or the mex function. Fraenkel and Peled⁴ defined minimal excludant as the smallest positive integer in a set S of positive integers that is not an element of S. The minimal excludant or the mex function was originally applied to study combinatorial game theory.

The concept of minimal excludants in the partition theory of integer was first introduced by Andrews and Newman⁵ in 2019. In partition theory of integer, the minimal excludant of the partition ξ of a positive integer n is defined as the smallest positive integer that is not a part of ξ and is denoted by $Mex(\xi)$. They also defined minimal odd exludant which is the smallest odd positive integer that is not a part of ξ . The minimal odd exludant of the partition ξ is denoted by $Moex(\xi)$. For example, for the partition 3 + 1 + 1 of 5, minimal excludant is Mex(3 + 1 + 1) = 2 and minimal odd excludant is Moex(3 + 1 + 1) = 5.

In Ref. 5, Andrews and Newman considered the following two arithmetic functions connected with the functions $Mex(\xi)$ and $Moex(\xi)$: Let S denote the set of all partition of positive integer n. Then

$$\sigma_{\mathsf{Mex}}(n) = \sum_{\xi \in \mathsf{S}} \mathsf{Mex}(\xi) \tag{6}$$

and

$$\sigma_{\mathsf{Moex}}(n) = \sum_{\xi \in \mathsf{S}} \mathsf{Moex}(\xi).$$
⁽⁷⁾

Andrews and Newman [5, p. 252, Theorem 4.1] also established that

$$\sum_{n\geq 0} \sigma_{\mathsf{Moex}}(n)q^n = (-q;q)_{\infty}(-q;q^2)_{\infty}^2.$$
(8)

A generalization of the function $\mathsf{Mex}(\xi)$ is also considered by Andrews and Neuman⁶ and gave applications in partition theory of numbers. Andrews and Neuman⁶ defined $\rho_{m,\alpha}(n)$ and $\overline{\rho}_{m,\alpha}(n)$ which counts the number of partitions of ξ of n, where $\mathsf{Mex}_{m,\alpha}(\xi) \equiv \alpha \pmod{2m}$ and $\mathsf{Mex}_{m,a}(\xi) \equiv m + a \pmod{2m}$, respectively. They showed that $\lambda(n) = \rho_{m,\alpha}(\xi) + \overline{\rho}_{m,\alpha}(\xi)$. They also gave partition-theoretic interpretation of $\rho_{m,\alpha}(\xi)$ for $(m,\alpha) = (1,1), (3,3), (2,1), (4,2)$ and (6,3). The parity of the function $\rho_{r,r}(n)$ was considered by Silva and Sellers,⁷ who discussed the characterization of $\rho_{1,1}(n)$ and $\rho_{3,3}(n)$ completely. Recently, Barman and Singh⁸ studied parity of $\rho_{2^t,2^t}(n)$ and $\rho_{3\cdot2^t,3\cdot2^t}(n)$ for $t \geq 1$. They also found infinite families of congruences modulo 2 using the theory of modular form.

This chapter is devoted to discuss some divisibility properties of the arithmetic functions $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$. Specifically, congruences moduli 2 and 4 for $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$ will be proved and also identities that connect $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$ with some other partition functions will be established by applying q-operations and some q-series identities. The concept of color partitions of positive integer is also important in proving the results. A part or summand in a partition ξ is said to have ℓ different colors if each part of ξ is considered in ℓ different copies and all of them are taken as distinct. For example, if each part in the partitions of 2 has three colors, say red(r), blue(b) and green(g), then the number of three color partitions of 2 is 9, namely

$$2_r, 2_b, 2_g, 1_r + 1_r, 1_b + 1_b, 1_g + 1_g, 1_g + 1_r, 1_g + 1_b, 1_r + 1_b$$

Let $\lambda_{\ell}(n)$ denote the number of partitions of n with each part having ℓ different colors. Then the generating function of $\lambda_{\ell}(n)$ is given by

$$\sum_{n\geq 0} \lambda_{\ell}(n)q^n = \frac{1}{(q;q)_{\infty}^{\ell}}.$$
(9)

The partition function $\lambda_{\ell}(n)$ was first introduced by Ramanujan in one of his letters to Hardy.⁹ The case $\ell = 1$, that is, $\lambda_1(n)$ is the partition function $\lambda(n)$ given in (3). The case $\ell = -1$ of $\lambda_{\ell}(n)$ defined in (9) is the celebrated Pentagonal Number Theorem given by

$$\sum_{n\geq 0} \lambda_{-1}(n)q^n = (q;q)_{\infty} = \begin{cases} (-1)^j & \text{for} \quad n = j(3j \pm 1)/2, \\ 0 & \text{otherwise.} \end{cases}$$
(10)

In fact,

$$\lambda_{-1}(n) = \lambda_{-1}^{e}(n) - \lambda_{-1}^{o}(n), \qquad (11)$$

where $\lambda_{-1}^{e}(n)$ (resp. $\lambda_{-1}^{o}(n)$) counts the number of partitions of integer n with even (resp. odd) number of distinct parts. For example, if n = 5, then $\lambda_{-1}^{e}(5) = 2$ with relevant partitions 4 + 1 and 3 + 2 and $\lambda_{-1}^{o}(5) = 1$ with the relevant partition 5. Thus, $\lambda_{-1}(5) = 2 - 1 = 1$. Generalizing the idea, for any integer $\ell \geq 1$, one has

$$\lambda_{-\ell}(n) = \lambda_{-\ell}^e(n) - \lambda_{-\ell}^o(n), \qquad (12)$$

where $\lambda_{-r}^{e}(n)$ (resp. $\lambda_{-\ell}^{o}(n)$) is the number of partitions of n having even (resp. odd) number of distinct parts such that each part has ℓ distinct colors. One can see Refs. 10 and 11 and references therein for congruences of the $\lambda_{-\ell}(n)$ and $\lambda_{\ell}(n)$.

It is useful to note here that if $k,\,m$ and ℓ are positive integers, then

$$\frac{1}{(q^k;q^m)_{\infty}^{\ell}} \tag{13}$$

is the generating function of the number of partitions of a positive integer with parts congruent to k modulo m (that is, $\equiv k \pmod{m}$) and each part has ℓ different colors. Similarly,

$$\frac{1}{(q^{k_1};q^m)_{\infty}^{\ell} \ (q^{k_2};q^m)_{\infty}^{\ell}} \tag{14}$$

is the generating function of the number of partitions with parts $\equiv k_1$ or $k_2 \pmod{m}$ and each part has ℓ different colors.

It is also noteworthy that using color partition of positive integer, Andrews and Newman [5, p. 250, Theorem 1.1] proved that

$$\sigma_{\mathsf{Mex}}(n) = D_2(n),\tag{15}$$

where $D_2(n)$ counts the number of partitions of n into distinct parts such that each part has two colors. The equality (15) can be stated as

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(n) q^n = (-q; q)_{\infty}^2 = \sum_{n \ge 0} D_2(n) q^n.$$
(16)

For combinatorial proof of (15), refer paper by Ballantine and Merca.¹²

2. Congruences for $\sigma_{Mex}(n)$

Theorem 1. One has

$$\sigma_{\mathsf{Mex}}(2n+1) \equiv 0 \pmod{2}.$$

Proof. From (16), note that

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}}(n)q^n = (-q;q)_{\infty}^2.$$
(17)

Simplify (17) by employing elementary q-operations to write as

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(n) q^n = \frac{(q^2; q^2)_{\infty}^2}{(q; q)_{\infty}^2}.$$
 (18)

Using binomial theorem in (1), one can easily show that

$$(q^2; q^2)_{\infty} \equiv (q; q)_{\infty}^2 \pmod{2}.$$
(19)

Employ (19) in (18) to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(n) q^n \equiv (q^2; q^2)_{\infty} \pmod{2}.$$
(20)

It is easily seen that the right-hand side of (20) contains no terms involving odd powers of q. So, extracting the terms involving q^{2n+1} from (20), one can arrive at the desired result.

To verify Theorem 1, take n = 1. Then for the partitions 3, 2 + 1and 1 + 1 + 1 of $2n + 1 = 2 \cdot 1 + 1 = 3$, Mex(3) = 1, Mex(2 + 1) = 3and Mex(1 + 1 + 1) = 2, respectively. So, $\sigma_{Mex}(3) = 1 + 3 + 2 = 6 \equiv 0$ (mod 2).

Theorem 2. Let $\delta \geq 0$ be any integer and $\omega \geq 5$ be any prime. Then

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}} \left(2\omega^{2\delta} n + \frac{\omega^{2\delta} - 1}{12} \right) q^n \equiv (q;q)_{\infty} \pmod{2}.$$
(21)

Proof. The method of mathematical induction with δ will be employed to prove the result. Extract the terms containing q^{2n} from (20) and then replace q^2 by q to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(2n) q^n \equiv (q; q)_{\infty} \pmod{2}.$$
(22)

This proves $\delta = 0$ case of (21). Assume that (21) is true for some $\delta \geq 0$. Now, for any prime $\omega \geq 5$, from [13, Theorem 2.2], one has

$$(q;q)_{\infty} = \sum_{\substack{j=-(\omega-1)/2\\j\neq(\pm\omega-1)/6}}^{(\omega-1)/2} (-1)^{j} q^{(3j^{2}+j)/2} f$$

$$\times \left(-q^{(3\omega^{2}+(6j+1)\omega)/2}, -q^{(3\omega^{2}-(6j+1)\omega)/2}\right),$$

$$+ (-1)^{(\pm\omega-1)/6} q^{(\omega^{2}-1)/24} (q^{\omega^{2}};q^{\omega^{2}})_{\infty}$$
(23)

where $\frac{\pm \omega - 1}{6} := \begin{cases} (\omega - 1)/6, & \text{if } \omega \equiv 1 \pmod{6}, \\ (-\omega - 1)/6, & \text{if } \omega \equiv -1 \pmod{6}, \end{cases}$ and f(u, v) is the Ramanujan's general theta function defined by

$$f(u,v) = \sum_{t=-\infty}^{\infty} u^{t(t+1)/2} v^{t(t-1)/2}, \qquad |uv| < 1.$$

Furthermore, if $-\frac{\omega-1}{2} \leq j \leq \frac{\omega-1}{2}$ and $j \neq \frac{\pm \omega - 1}{2}$, then $\frac{3j^2+j}{2} \not\equiv \frac{\omega^2-1}{24} \pmod{\omega}$.

Employ (23) in (21) to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}} \left(2\omega^{2\delta} n + \frac{\omega^{2\delta} - 1}{12} \right)$$

$$\equiv \sum_{\substack{j=-(\omega-1)/2\\ j\neq (\pm\omega-1)/6}}^{(\omega-1)/2} (-1)^j q^{(3j^2+j)/2} f\left(-q^{(3\omega^2 + (6j+1)\omega)/2}, -q^{(3\omega^2 - (6j+1)\omega)/2} \right)$$

$$+ (-1)^{(\pm\omega-1)/6} q^{(\omega^2-1)/24} (q^{\omega^2}; q^{\omega^2})_{\infty} \pmod{2}.$$
(mod 2). (24)

Extract the terms that contain $q^{\omega n+(\omega^2-1)/24}$ from (24) and then divide by $q^{(\omega^2-1)/24}$ and replace q^{ω} by q to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}} \left(2\omega^{2\delta+1}n + \frac{\omega^{2\delta+2} - 1}{12} \right) q^n \equiv (q^\omega; q^\omega)_\infty \pmod{2}. \tag{25}$$

Again, extract the terms involving $q^{\omega n}$ and replace q^{ω} by q to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}} \left(2\omega^{2(\delta+1)} n + \frac{\omega^{2(\delta+1)} - 1}{12} \right) q^n \equiv (q;q)_{\infty} \pmod{2}.$$
(26)

Equation (26) is the $\delta + 1$ case of (21), and this completes the proof.

Corollary 1. For any integer $\delta \geq 0$ and any prime $\omega \geq 5$, one has

$$\sigma_{\mathsf{Mex}}\left(2\omega^{2\delta+2}n + \frac{\omega^{2\delta+1}(24j+\omega) - 1}{12}\right) \equiv 0 \pmod{2},$$

where $j = 1, 2, 3, \dots, \omega - 1$.

Proof. Extract the terms containing $q^{\omega n+j}$ for $j = 1, 2, 3, \ldots, \omega - 1$ from (25) to obtain the desired result.

Corollary 2. One has

$$\sigma_{\mathsf{Mex}}(2n) \equiv \left(\lambda_{-1}^e(n) - \lambda_{-1}^o(n)\right) \pmod{2},$$

where $\lambda_{-1}^{e}(n)$ (resp. $\lambda_{-1}^{o}(n)$) counts the number of partitions of n with even (resp. odd) number of distinct parts.

Proof. From (10) and (11), it is easily seen that

$$\sum_{n \ge 0} \left(\lambda_{-1}^e(n) - \lambda_{-1}^o(n) \right) q^n = (q; q)_{\infty}.$$
 (27)

Employ (22) in (27) to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}}(2n)q^n \equiv \sum_{n\geq 0} \left(\lambda_{-1}^e(n) - \lambda_{-1}^o(n)\right) q^n \pmod{2}.$$
(28)

The required result now follows directly from (28).

To verify Corollary 2, set n = 2. Then, for the partitions 4, 3 + 1, 2 + 2, 2 + 1 + 1 and 1 + 1 + 1 + 1 of 4, Mex(4) = 1, Mex(3 + 1) = 2, Mex(2 + 2) = 1, Mex(2 + 1 + 1) = 3, and Mex(1 + 1 + 1 + 1) = 2, respectively. Therefore, $\sigma_{Mex}(4) = 1 + 2 + 1 + 3 + 2 = 9$. Again, $\lambda_{-1}^{e}(2) = 0$ and $\lambda_{-1}^{o}(2) = 1$ with the relevant partition 2. Thus, Corollary 2 is true as $9 \equiv (0 - 1) = -1 \pmod{2}$.

Theorem 3. Let $\lambda_{-2}^e(n)$ (resp. $\lambda_{-2}^o(n)$) be the number of partitions of n with even (resp. odd) number of distinct parts and each part has two colors. Then

$$\sigma_{\mathsf{Mex}}(n) \equiv \left(\lambda_{-2}^{e}(n) - \lambda_{-2}^{o}(n)\right) \pmod{4}.$$

Proof. Employ (35) in (18) to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(n) q^n = \frac{(q^2; q^2)_\infty^2}{(q; q)_\infty^2} \equiv (q; q)_\infty^2 \pmod{4}.$$
 (29)

From Pentagonal Number Theorem, one can see that

$$\sum_{n \ge 0} \left(\lambda_{-2}^e(n) - \lambda_{-2}^o(n) \right) q^n = (q; q)_{\infty}^2.$$
(30)

Employ (30) in (29) to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}}(n)q^n \equiv \sum_{n\geq 0} \left(\lambda_{-2}^e(n) - \lambda_{-2}^o(n)\right) q^n \pmod{4}.$$
(31)

Now, extract the coefficients of q^n from both sides of (31) to complete the proof.

Set n = 2 in Theorem 3. One can see that $\sigma_{Mex}(2) = Mex(2) + Mex(1+1) = 1+2 = 3$. Next, consider two colors green(g) and blue(b), then $\lambda_{-2}^e(n) = 1$ with relevant partition $1_b + 1_g$ and $\lambda_{-2}^o(n) = 2$ with relevant partitions 2_b and 2_g . Thus, Theorem 3 is easily verified as $3 \equiv (1-2) = -1 \pmod{4}$.

Theorem 4. Let C(n) denote the number of partitions of n such that parts $\equiv 0 \pmod{4}$ have two colors and parts $\equiv 1, 2, \text{ or } 3 \pmod{4}$ have three colors with the generating function defined by

$$\sum_{n \ge 0} \mathcal{C}(n)q^n = \frac{1}{(q;q)_{\infty}^2(q,q^2,q^3;q^4)_{\infty}} = \frac{1}{(q^4;q^4)_{\infty}^2(q,q^2,q^3;q^4)_{\infty}^3}$$

Then

$$\sigma_{\mathsf{Mex}}(2n) \equiv \mathcal{C}(n) \pmod{4}.$$

Proof. From [14, Lemma 2.1], one has

$$\frac{1}{(q;q)_{\infty}^2} = \frac{(q^8;q^8)_{\infty}^5}{(q^2;q^2)_{\infty}^5 (q^{16};q^{16})_{\infty}^2} + 2q \ \frac{(q^4;q^4)_{\infty}^2 (q^{16};q^{16})_{\infty}^2}{(q^2;q^2)_{\infty}^5 (q^8;q^8)_{\infty}}.$$
 (32)

Employ (32) in (18) and simplify to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}}(n)q^n = \frac{(q^8; q^8)_\infty^5}{(q^2; q^2)_\infty^3 (q^{16}; q^{16})_\infty^2} + 2q \, \frac{(q^4; q^4)_\infty^2 (q^{16}; q^{16})_\infty^2}{(q^2; q^2)_\infty^3 (q^8; q^8)_\infty}.$$
(33)

Extract the terms containing q^{2n} from (33) and then replace q^2 by q to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(2n) q^n = \frac{(q^4; q^4)_\infty^5}{(q; q)_\infty^3 \ (q^8; q^8)_\infty^2}.$$
(34)

From (1) and binomial theorem, one has

$$(q^2; q^2)^2_{\infty} \equiv (q; q)^4_{\infty} \pmod{4}.$$
 (35)

Simplify (34) by employing (35) to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Mex}}(2n) q^n \equiv \frac{(q^4; q^4)_\infty}{(q; q)^3_\infty} \pmod{4}.$$
 (36)

Again, simplify (36) by using elementary q-operations to write as

$$\sum_{n\geq 0} \sigma_{\mathsf{Mex}}(2n)q^n \equiv \frac{1}{(q;q)^2_{\infty}(q,q^2,q^3;q^4)_{\infty}} = \sum_{n\geq 0} \mathcal{C}(n)q^n \pmod{4}.$$
(37)

Now, equate the coefficients of q^n on both sides of (37) to complete the proof. \Box

Setting n = 2 in Theorem 4, we see that for the partitions of 4, $\sigma_{\mathsf{Mex}}(4) = 1 + 2 + 1 + 3 + 2 = 9$, as $\mathsf{Mex}(4) = 1$, $\mathsf{Mex}(3+1) = 2$, $\mathsf{Mex}(2+2) = 1$, $\mathsf{Mex}(2+1+1) = 3$, and $\mathsf{Mex}(1+1+1+1) = 2$. Again, if there are three different colors, say blue(b), green(g) and red(r), then $\mathcal{C}(2) = 9$ with relevant partitions 2_r , 2_g , 2_b , $1_r + 1_r$, $1_r + 1_g$, $1_r + 1_b$, $1_g + 1_g$, $1_g + 1_b$, and $1_b + 1_b$. Thus, Theorem 4 is true as $9 \equiv 9$ (mod 4).

3. Congruence for $\sigma_{Moex}(n)$

Theorem 5. One has

$$\sigma_{\mathsf{Moex}}(n) \equiv p(n) \pmod{2}.$$

Proof. Use elementary q-operations in (8) to obtain

$$\sum_{n \ge 0} \sigma_{\mathsf{Moex}}(n) q^n = \frac{(q^2; q^2)_\infty^5}{(q; q)_\infty^3 (q^4; q^4)_\infty^2}.$$
(38)

Simplify (38) using (19) and then employ (3) to obtain

$$\sum_{n\geq 0} \sigma_{\mathsf{Moex}}(n)q^n \equiv \frac{1}{(q;q)_{\infty}} = \sum_{n\geq 0} p(n)q^n \pmod{2}. \tag{39}$$

Now, extract the coefficients of q^n in (39) to arrive at the desired result. \Box

Taking n = 3 in Theorem 5, we find that $\sigma_{Moex}(3) = 1 + 3 + 3 = 7$ for Moex(3) = 1, Moex(2 + 1) = 3, and Moex(1 + 1 + 1) = 3. Again, p(3) = 3 with the relevant partitions 3, 2 + 1, and 1 + 1 + 1. Thus, Theorem 5 is readily verified as $7 \equiv 3 \pmod{2}$.

Theorem 6. Let $\mathcal{D}(n)$ be the number of partitions of a positive integer n such that even parts have two colors and odd parts have three colors with the generating function given by

$$\sum_{n \ge 0} \mathcal{D}(n)q^n = \frac{1}{(q;q)^2_{\infty}(q;q^2)_{\infty}} = \frac{1}{(q^2;q^2)^2_{\infty}(q;q^2)^3_{\infty}}$$

Then

$$\sigma_{\mathsf{Moex}}(n) \equiv \mathcal{D}(n) \pmod{4}.$$

Proof. Simplify (38) by elementary *q*-operations and then employ (35) to obtain

$$\sum_{n \ge 0} \sigma_{\text{Moex}}(n) q^n \equiv \frac{(q^2; q^2)_{\infty}}{(q; q)_{\infty}^3} = \frac{1}{(q; q)_{\infty}^2 (q; q^2)_{\infty}}$$
$$= \sum_{n \ge 0} \mathcal{D}(n) q^n \pmod{4}.$$
(40)

The desired result now easily follows by equating the coefficients of q^n on both sides of (40).

Setting n = 2 in Theorem 6, we see that $\sigma_{moex}(2) = 1 + 3 = 4$ for Moex(2) = 1 and Moex(1+1) = 3. Again, if there are three different colors, say red(r), blue(b) and green(g) for odd parts and two colors blue(b) and green(g) for even parts, then $\mathcal{D}(2) = 8$ with relevant partitions 2_b , 2_g , $1_r + 1_r$, $1_r + 1_g$, $1 + 1_r + 1_b$, $1_g + 1_g$, $1_b + 1_g$, and $1_b + 1_b$. Thus, Theorem 6 is easily verified as $4 \equiv 8 \pmod{4}$ and holds true.

4. Conclusion

In this chapter, some new congruences modulo 2 and 4 for the functions $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$ are presented. Some identities connecting $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$ with the color partition functions are also established. The two functions $\sigma_{Mex}(n)$ and $\sigma_{Moex}(n)$ can be further investigated for congruences modulo higher primes. One can also note that the proofs of many results related to mex functions in the literature relied on the theory of modular forms, so elementary proofs using *q*-series are always desired.

Acknowledgments

The author is extremely grateful to the anonymous referees who read the manuscript and offered helpful comments and suggestions.

References

- 1. S. Ramanujan, Some properties of p(n), the number of partition of n, *Proceedings of the Cambridge Philosophical Society* **19**, 207–210 (1919).
- 2. S. Ramanujan, Congruence properties of partitions, *Proceedings of the London Mathematical Society* 18, xix (1920).
- S. Ramanujan, Congruence properties of partitions, Mathematische Zeitschrift 9, 147–153 (1921).
- A. S. Fraenkel and U. Peled, Harnessing the unwieldy MEX function. In Games of No Chance 4, vol. 63, Math. Sci. Res. Inst. Publ., pp. 77–94. Cambridge Univ. Press, New York (2015).
- G. E. Andrews and D. Newman, Partitions and the minimal excludant, Annals of Combinatorics 23(2), 249–254 (2019).

- 6. G. E. Andrews and D. Newman, The minimal excludant in integer partitions, *Journal of Integer Sequences* **23**(2), Art. 20.2.3, 11 (2020).
- R. da Silva and J. A. Sellers, Parity considerations for the mexrelated partition functions of Andrews and Newman, *Journal of Integer* Sequences 23(5), Art. 20.5.7, 9 (2020).
- R. Barman and A. Singh, On mex-related partition functions of Andrews and Newman, *Research in Number Theory* 7(3), Paper No. 53, 11 (2021).
- B. C. Berndt and R. A. Rankin, *Ramanujan*. Vol. 9, *History of Mathematics*. American Mathematical Society, Providence, RI; London Mathematical Society, London (1995). Letters and commentary.
- N. Saikia and J. Chetry, Infinite families of congruences modulo 7 for Ramanujan's general partition function, Annales Mathematiques du Quebec 42(1), 127–132 (2018).
- 11. N. Saikia and C. Boruah, General congruences modulo 5 and 7 for colour partitions, *Journal of Analysis* **29**(3), 917–926 (2021).
- 12. C. Ballantine and M. Merca, The minimal excludant and colored partitions, *Seminaire Lotharingien Combinatorie* 84B, Art. 23, 12 (2020).
- S.-P. Cui and N. S. S. Gu, Arithmetic properties of l-regular partitions, Advances in Applied Mathematics 51(4), 507–523 (2013).
- E. X. W. Xia, Congruences modulo 9 and 27 for overpartitions, Ramanujan Journal 42(2), 301–323 (2017).

This page intentionally left blank

© 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0007

Chapter 7

On Certain ℓ -regular Partitions: A Brief Survey

Chiranjit Ray

Indian Statistical Institute Delhi, Theoretical Stat-Math Unit, New Delhi 110016, India chiranjitray.m@isid.ac.in

In this chapter, we give an introduction to ℓ -regular partitions and ℓ -regular overpartitions. We also give a brief review of the literature on the works done so far on these partition functions.

1. Introduction

A partition of a positive integer n is a representation of n as a sum of positive integers $\lambda_1, \lambda_2, \ldots, \lambda_k$ such that the order of these positive integers occurs being immaterial. We thus do not distinguish between $n = \lambda_1 + \lambda_2 + \cdots + \lambda_k$ and $n = \lambda_{i_1} + \lambda_{i_2} + \cdots + \lambda_{i_k}$, where $\sigma = i_1 i_2 \cdots i_k$ is a permutation of the set $\{1, 2, \ldots, k\}$. The positive integers $\lambda_1, \lambda_2, \ldots, \lambda_k$ occur in a partition called *parts*. If there is no restriction of any kind on the number or size of parts, we call a partition as *unrestricted* partition.

One can restrict a partition in many ways. For example, imposing restrictions on parts has famous partition functions such as partitions into distinct parts and partitions into odd parts. Euler's partition theorem states that the number of partitions of an integer n into odd parts is equal to the number of partitions of an integer n into distinct parts (see Ref. 1).

The number of unrestricted partitions of any given positive integer n is denoted by p(n). Conventionally, we set p(0) = 1. For example, 7 can be expressed as a sum in the following 15 ways:

7;
$$6 + 1$$
; $5 + 2$; $5 + 1 + 1$; $4 + 3$; $4 + 2 + 1$; $4 + 1 + 1 + 1$;
 $3 + 3 + 1$; $3 + 2 + 2$; $3 + 2 + 1 + 1$;
 $3 + 1 + 1 + 1 + 1$; $2 + 2 + 2 + 1$; $2 + 2 + 1 + 1 + 1$;
 $2 + 1 + 1 + 1 + 1 + 1$; $1 + 1 + 1 + 1 + 1 + 1 + 1$.

These are all the unrestricted partitions of 7. For complex numbers a and q, the symbol $(a; q)_{\infty}$ stands for the *q*-shifted factorial defined as

$$(a;q)_{\infty} = \prod_{n=1}^{\infty} (1 - aq^{n-1}), \quad |q| < 1.$$

The generating function for p(n) is given by

$$\sum_{n=0}^{\infty} p(n)q^n = \frac{1}{(q;q)_{\infty}}.$$

Ramanujan had found three simple congruences satisfied by p(n):

$$p(5n+4) \equiv 0 \pmod{5},$$

$$p(7n+5) \equiv 0 \pmod{7},$$

and

$$p(11n+6) \equiv 0 \pmod{11}.$$

He gave proofs of the first two congruences in Ref. 2 and remarked that there are no simple congruences for p(n) whose moduli involve primes other than 5, 7 and 11. Ahlgren and Boylan³ justifies Ramanujan's claim by proving that if ℓ is a prime and $0 \leq \beta < \ell$ is an integer for which

$$p(\ell n + \beta) \equiv 0 \pmod{\ell}$$

for every nonnegative integer n, then $(\ell, \beta) \in \{(5, 4), (7, 5), (11, 6)\}$. Ono⁴ made a remarkable breakthrough by proving that for every prime $m \geq 5$, there exist infinitely many arithmetic progressions an+b such that for every nonnegative integer n, we have $p(an+b) \equiv 0 \pmod{m}$. Later, Ahlgren⁵ has extended this result to include every modulus m coprime to 6. Looking at these congruences, one may wonder exactly how often p(n) is divisible by m > 1. The answer to this question is not known for any such m. The well-known parity conjecture of Parkin and Shanks⁶ predicts that the values of p(n) are equally distributed modulo 2.

Conjecture 1. If $r \in \{0, 1\}$, then

$$\lim_{X \to +\infty} \frac{\#\{0 \le n < X : p(n) \equiv r \pmod{2}\}}{X} = \frac{1}{2}.$$

This conjecture is still open. In fact, very little is known regarding the above conjecture.

For certain ℓ -regular partition functions, the state of knowledge about similar questions as above is better.

A partition of n is ℓ -regular if none of its parts are multiples of ℓ . For example, the nine 3-regular partition of 7 are

7; 5+2; 5+1+1; 4+2+1; 4+1+1+1; 2+2+2+1; 2+2+1+1+1; 2+1+1+1+1+1; 1+1+1+1+1+1+1.

We denote the number of ℓ -regular partitions of n by $b_{\ell}(n)$ and the generating function⁷ is given by

$$\sum_{n=0}^{\infty} b_{\ell}(n)q^n = \frac{(q^{\ell}; q^{\ell})_{\infty}}{(q; q)_{\infty}}.$$

In the classical representation theory, the number of irreducible p-modular representations of the symmetric group S_n is the same as $b_p(n)$ when p is prime (see Ref. 8). Recently, Gordon and Ono⁹ proved that the number of p-modular irreducible representations of almost every symmetric group S_n is a multiple of p^j . Many good results are proved for ℓ -regular partitions using q-series identities, modular form, and combinatorics.

In Section 2, we briefly discuss the results for ℓ -regular partitions that are currently available in the literature.

Corteel and Lovejoy¹⁰ introduced overpartitions. An *overpartition* of n is an unrestricted partition of n in which the first occurrence of a number may be overlined. Suppose $\overline{p}(n)$ counts the total number of overpartitions of n. For example, the overpartitions of 4 are

4;
$$\overline{4}$$
; $3 + 1$; $\overline{3} + 1$; $3 + \overline{1}$; $\overline{3} + \overline{1}$; $2 + 2$; $\overline{2} + 2$; $2 + 1 + 1$;
 $\overline{2} + 1 + 1$; $2 + \overline{1} + 1$; $\overline{2} + \overline{1} + 1$; $1 + 1 + 1 + 1$; $\overline{1} + 1 + 1 + 1$

From the above example, $\overline{p}(4) = 14$. The generating function for $\overline{p}(n)$ is given by

$$\sum_{n=0}^{\infty} \overline{p}(n)q^n = \frac{(q^2; q^2)_{\infty}}{(q; q)_{\infty}^2}.$$

In Ref. 11, Lovejoy investigated the ℓ -regular overpartition $\overline{A}_{\ell}(n)$, which counts the number of overpartitions of n with no parts divisible by ℓ . From the above example, it is easy to see that $\overline{A}_3(4) = 10$. The generating function for $\overline{A}_{\ell}(n)$ is

$$\sum_{n=0}^{\infty} \overline{A}_{\ell}(n) q^n = \frac{(-q;q)_{\infty}(q^{\ell};q^{\ell})_{\infty}}{(q;q)_{\infty}(-q^{\ell};q^{\ell})_{\infty}}.$$

We give a brief review of the literature on the works done so far for ℓ -regular overpartitions in Section 3.

2. *l*-Regular Partitions

In Ref. 9, Gordon and Ono consider $\ell = p_1^{a_1} p_2^{a_2} \dots p_m^{a_m}$ as a product of primes and $S_k(X; M)$ is the number of positive integers $n \leq X$ for which $b_k(n) \equiv 0 \pmod{M}$. They prove the following theorem.

Theorem 1. Let $\ell = p_1^{a_1} p_2^{a_2} \dots p_m^{a_m}$ be the prime factorization of a positive integer ℓ . If $p_i^{2a_i} \ge \ell$, then for every positive integer j,

$$\lim_{X \to \infty} \frac{S_k(X; p_i^j)}{X} = 1.$$

Although the set of nonnegative integers n for which $b_{\ell}(n) \equiv 0 \pmod{p_i^j}$ has density one when $p_i^{2a_i} \geq \ell$, the authors also show that

the set of those m for which $b_{\ell}(m) \neq 0 \pmod{p_i^j}$ is indeed infinite in Corollary 2.

Lovejoy and Penniston¹¹ consider the 3-regular partitions, and they find a relation between $b_3(n)$ and the zeta function of a modular K3 surface. Particularly, they proved the following theorem.

Theorem 2. Let X be the K3 surface defined by

 $X: s^{2} = x(x+1)y(y+1)(x+8y).$

If $p \equiv 1 \pmod{12}$ is a prime, then

$$b_3\left(\frac{p-1}{12}\right) = \#X(\mathbb{F}_p) - (p+1)^2 \pmod{9}.$$

Using the Hecke theory and the arithmetic of the Gaussian integers, they also proved the following result.

Theorem 3. Let n be a positive integer such that $12n + 1 = N^2M$ with M square-free, and for a prime divisor p of 12n + 1, define $k_p := ord_p(12n + 1)$. If $p \equiv 1 \pmod{12}$, let d_p and e_p be integers such that $3 \mid d_p$ and $p = d_p^2 + e_p^2$.

• If there is a prime $p \equiv 5, 7$ or 11 (mod 12) such that $p \mid M$, then

$$b_3(n) \equiv 0 \pmod{9}.$$

If every prime divisor of M satisfies that there is a prime p ≡ 1 (mod 12), then

$$b_{3}(n) \equiv (3n+1) \prod_{\substack{p \mid (12n+1)\\ p \equiv 1 \pmod{12}}} (-1)^{k_{p}d_{p}}(k_{p}+1)$$
$$\times \prod_{\substack{p \mid (12n+1)\\ p \equiv 5 \pmod{12}}} (-1)^{k_{p/2}} \pmod{9}.$$

Remember that $b_2(n)$ also denotes the number of partitions of an integer n into odd parts (which is equal to counting the number of partitions of an integer n into distinct parts). In Refs. 12 and 13, Ono and Penniston address the more general question of the distribution of $b_2(n)$ compared to the previous result by Gordon and Ono.⁹ They proved the following results.

Theorem 4. If j is a positive integer and $1 \le i \le 2^j - 1$, then

 $\#\{0 \le n \le X : b_2(n) \equiv i \pmod{2^j}\} \gg_{i,j} \sqrt{X}/\log X.$

It would be difficult to substantially improve Theorem 4 for odd *i*. However, one can often improve the lower bound in Theorem 4 to $X/\log X$, in case *i* is even. In the series of articles,^{14–16} Penniston studied ℓ -regular partitions intensely and proved many significant results. In Ref. 14, Penniston extended Theorem 4 from $\ell = 2$ to an arbitrary prime power $\ell = p^a$ and proved the following two theorems.

Theorem 5. Let a be a positive integer, p an odd prime, and let $\ell = p^a$. If j is a positive integer and $1 \le i \le p^j - 1$, then

$$#\{0 \le n \le X : b_{\ell}(n) \equiv i \pmod{p^{j}} \} \gg_{\ell,i,j} X/\log X.$$

Theorem 6. Let a be a positive integer, and let $\ell = p^a$. If j is a positive integer and $1 \le i \le 2^j - 1$, then

$$\#\{0 \le n \le X : b_{\ell}(n) \equiv i \pmod{2^j}\} \gg_{\ell,i,j} \begin{cases} X/\log X & \text{if } i \text{ is even,} \\ \sqrt{X}/\log X & \text{if } i \text{ is odd.} \end{cases}$$

Now, we discuss the article 15 which is a follow-up of the above article by Penniston.

Theorem 7. Suppose $3 \le \ell \le 23$ and $p \ge 5$ are distinct primes, and s is a positive integer. Then

$$\lim_{X \to \infty} \inf \frac{\#\{1 \le n \le X : b_{\ell}(n) \equiv 0 \pmod{p^s}\}}{X}$$
$$\geq \begin{cases} \frac{p+1}{2p} & \text{if } p \nmid \ell - 1, \\ \frac{p-1}{p} & \text{if } p \mid \ell - 1. \end{cases}$$

Theorem 8. Suppose $3 \le \ell \le 23$ is prime and m > 1 is odd. Assume $3 \nmid m$ if $\ell \ne 3$. Further suppose there exists an integer $k' \equiv \ell - 1$ (mod 24) such that if we set $n' = \frac{k'-\ell+1}{24}$, then $n' \in \bigcap_{p|m} S_{\ell,p}$ and

 $b_{\ell}(n')$ is coprime to m. Then for any $1 \leq a \leq m-1$,

$$\#\{0 \le n \le X : b_{\ell}(n) \equiv a \pmod{m}\} \gg X/\log X.$$

In Ref. 15, Penniston showed that for any $j \ge 1$, $b_{11}(n)$ is divisible by 5^j for at least 80% of positive integers n and in Ref. 16, the author gave a complete description of 5-divisibility of $b_{11}(n)$ if $5 \nmid n$. Consider the binary quadratic forms

$$F_{a,b,c}(X,Y) = aX^2 + bXY + cY^2,$$

and let p be an odd prime with $\left(\frac{-33}{p}\right) = 1$. One can find $x, y \in \mathbb{Z}$ such that

$$p = F_{a,b,c}(x,y)$$

for $(a, b, c) \in \{(1, 0, 33), (2, 2, 17), (6, 6, 7), (3, 0, 11)\}$. Now, fix such x and y, and define

$$\delta_p := \begin{cases} 4px^2 & \text{if } p \equiv 1 \pmod{12}, \\ py^2 & \text{if } p \equiv 5 \pmod{12}, \\ p(y-3x)^2 & \text{if } p \equiv 7 \pmod{12}, \\ 4py^2 & \text{if } p \equiv 11 \pmod{12}. \end{cases}$$

Theorem 9. Suppose n is a positive integer such that $5 \nmid n$. Then $b_{11}(n)$ is divisible by 5 if and only if there exists a prime p such that (i) $\left(\frac{-33}{p}\right) = -1$ and $ord_p(12n + 5) \equiv 1 \pmod{2}$, or (ii) $\left(\frac{-33}{p}\right) = 1$ and one of the following conditions holds.

(1) $ord_p(12n+5) \equiv 1 \pmod{2}$ and $\delta_p \equiv 0 \pmod{2}$,

(2) $ord_p(12n+5) \equiv 2 \pmod{2}$ and $\delta_p \equiv 1 \pmod{2}$,

- (3) $ord_p(12n+5) \equiv 4 \pmod{2}$ and $\delta_p \equiv 4 \pmod{2}$,
- (4) $ord_p(12n+5) \equiv 5 \pmod{2}$ and $\delta_p \equiv 2 \pmod{2}$,

where $ord_p(n)$ denotes the highest power of p that divides n.

Dandurand and Penniston¹⁷ studied the divisibility property for $b_5(n)$, $b_7(n)$, and $b_{11}(n)$. They proved the following results for $b_5(n)$. Consider the prime factorization of 6n + 1 by

$$6n + 1 = \prod_{i=1}^{s} p_i^{e_i},$$

where p_1, \ldots, p_s are distinct primes and $e_i > 0$ for all $1 \le i \le s$. For each *i* with $p_i \equiv 1 \pmod{3}$, let us write $p_i = x_i^2 + 3y_i^2$ with $x_i, y_i \in \mathbb{Z}$.

Theorem 10. Let n be a nonnegative integer. Then $b_5(n)$ is divisible by 5 if and only if for some $1 \le i \le s$, one of the following holds:

(1) $p_i = 5$,

(2)
$$p_i \equiv 2 \pmod{3}, p_i \neq 5 \text{ and } e_i \text{ is odd}$$

- (3) $p_i \equiv 1 \pmod{3}$, $5 \mid x_i \text{ and } e_i \text{ is odd}$,
- (4) $p_i \equiv 1 \pmod{3}, 5 \mid y_i \text{ and } e_i \equiv 4 \pmod{5},$

(5) $p_i \equiv 1 \pmod{3}, 5 \mid (x_i^2 - y_i^2) \text{ and } e_i \equiv 2 \pmod{3},$

(6) $p_i \equiv 1 \pmod{3}, 5 \nmid x_i y_i (x_i^2 - y_i^2) \text{ and } e_i \equiv 5 \pmod{6}.$

Hirschhorn and Sellers¹⁸ used Jacobi triple product identity and various 2-dissection formulas to prove the following results.

Theorem 11. For all nonnegative integers n, $b_5(2n)$ is odd if and only if 12n + 1 is a perfect square.

Theorem 12. For all nonnegative integers n, $b_5(4n + 1)$ is even unless $24n + 7 = 2x^2 + 5y^2$ for some integers x and y.

Theorem 13. For all nonnegative integers n, we have

$$b_5(20n+5) \equiv 0 \pmod{2}$$

and

$$b_5(20n+13) \equiv 0 \pmod{2}.$$

Theorem 14. Suppose that p is any prime greater than 3 such that -10 is a quadratic nonresidue modulo p, u is the reciprocal of 24 modulo p^2 , and $r \not\equiv 0 \pmod{p}$. Then, for all m, we have

$$b_5(4p^2m + 4upr - 28u + 1) \equiv 0 \pmod{2}.$$

Lastly, they proved the following density result.

Theorem 15. $b_5(n)$ is even for at least 75% of the positive integers n.

Here, we like to mention that Theorems 11 and 13 were also proved by Calkin *et al.*¹⁹ using a different method. In a very recent work, Dai^{20} studied the parity results for 25-regular partitions.

Theorem 16. For all nonnegative integers n, we have

 $b_{25}(10n+9) \equiv 0 \pmod{2}$.

Theorem 17. Let n be a nonnegative integer and p > 5 be a prime. Then for all $\alpha > 0$ and $1 \le r < 4p$ with gcd(r,p) = 1 and $rp \equiv 1 \pmod{4}$, we have

$$b_{25}(20p^{2\alpha}n + 5rp^{2\alpha-1} - 1) \equiv 0 \pmod{2}.$$

Xia²¹ derived the congruence properties for b_{13} , b_{17} and b_{19} modulo 13, 17 and 19, respectively, by using some theta-function identities of Ramanujan.

Theorem 18. For all nonnegative integers n and k, we have

$$b_{13}\left(3^{6k}n + \frac{3^{6k}-1}{2}\right) \equiv a(6k)b_{13}(n) \pmod{13},$$

where $a(k) = \left(\frac{1}{2} - \frac{\sqrt{17}}{34}\right) \left(\frac{1}{2} + \frac{\sqrt{17}}{2}\right)^k + \left(\frac{1}{2} + \frac{\sqrt{17}}{34}\right) \left(\frac{1}{2} - \frac{\sqrt{17}}{2}\right)^k$.

Theorem 19. For all nonnegative integers n and k, we have

$$b_{17}\left(4^{9k}n + \frac{2(4^{9k}-1)}{3}\right) \equiv c(9k)b_{17}(n) \pmod{17}$$

where $c(k) = \left(\frac{1}{2} - \frac{\sqrt{5}}{10}\right) \left(1 + \sqrt{5}\right)^k + \left(\frac{1}{2} + \frac{\sqrt{5}}{10}\right) \left(1 - \sqrt{5}\right)^k$.

Theorem 20. For all nonnegative integers n and k, we have

$$b_{19}\left(5^{10k}n + \frac{3(5^{10k} - 1)}{4}\right) \equiv d(10k)b_{19}(n) \pmod{19},$$

where $d(k) = \left(\frac{1}{2} - \frac{\sqrt{69}}{46}\right) \left(\frac{3}{2} + \frac{\sqrt{69}}{2}\right)^k + \left(\frac{1}{2} + \frac{\sqrt{69}}{46}\right) \left(\frac{3}{2} - \frac{\sqrt{69}}{2}\right)^k$.

Later using Theorems 18–20, the author have the following congruences, respectively:

$$b_{13}\left(\frac{151 \times 3^{6k} - 1}{2}\right) \equiv 0 \pmod{13},$$
$$b_{17}\left(\frac{89 \times 4^{9k} - 1}{3}\right) \equiv 0 \pmod{17},$$

and

$$b_{19}\left(\frac{123 \times 5^{10k} - 1}{4}\right) \equiv 0 \pmod{19}.$$

Ramanujan's general theta-function f(a, b) is defined as

$$f(a,b) = \sum_{n=-\infty}^{\infty} a^{n(n+1)/2} b^{n(n-1)/2}.$$

Then, the Jacobi triple product identity [22, Entry 19, p. 36] takes the shape

$$f(a,b) = (-a;ab)_{\infty}(-b;ab)_{\infty}(ab;ab)_{\infty}.$$

The most important two special cases of f(a, b) are

$$\psi(q) := f(q, q^3) = \sum_{n=0}^{\infty} q^{n(n+1)/2},$$
$$f(-q) := f(-q, -q^2) = \sum_{n=-\infty}^{\infty} (-1)^n q^{n(3n-1)/2}$$

Recently, Cui and Gu²³ developed the following two *p*-dissection identities for $\psi(q)$ and f(-q).

Theorem 21. For any odd prime p,

$$\psi(q) = \sum_{k=0}^{\frac{p-3}{2}} q^{\frac{k^2+k}{2}} f\left(\frac{p^2 + (2k+1)p}{2}, \frac{p^2 - (2k+1)p}{2}\right) + q^{\frac{p^2-1}{8}} \psi(q^{p^2}).$$

Furthermore, $\frac{k^2+k}{2} \not\equiv \frac{p^2-1}{8} \pmod{p}$, for $0 \le k \le \frac{p-3}{2}$.

Theorem 22. For any prime $p \ge 5$,

$$f(-q) = \sum_{\substack{k=-\frac{p-1}{2}\\k\neq\frac{\pm p-1}{6}}}^{\frac{p-1}{2}} (-1)^k q^{\frac{3k^2+k}{2}} f\left(-q^{\frac{3p^2+(6k+1)p}{2}}, -q^{\frac{3p^2-(6k+1)p}{2}}\right) + (-1)^{\frac{\pm p-1}{6}} q^{\frac{p^2-1}{24}} f(-q^{p^2}),$$

where

$$\frac{\pm p - 1}{6} := \begin{cases} \frac{p - 1}{6} & \text{if } p \equiv 1 \pmod{6};\\ \frac{-p - 1}{6} & \text{if } p \equiv -1 \pmod{6}. \end{cases}$$

Furthermore, for $\frac{-(p-1)}{2} \le k \le \frac{p-1}{2}$ and $k \ne \frac{\pm p-1}{6}$,

$$\frac{3k^2 + k}{2} \not\equiv \frac{p^2 - 1}{24} \pmod{p}.$$

In a series of articles, Cui and Cu^{23–25} studied the arithmetic properties of ℓ -regular partitions for several values of ℓ , using *p*-dissection formulas for Ramanujan's theta functions and several modular equations.

Theorem 23. For any prime $p \ge 5$, $\alpha \ge 0$, and $n \ge 0$, we have

$$b_2\left(p^{2\alpha+1}n + \frac{(24j+1)p^{2\alpha}-1}{24}\right) \equiv 0 \pmod{2},$$

where $0 \le j \le p - 1$ and $\left(\frac{24j+1}{p}\right) = -1$.

Theorem 24. For any odd prime $p, \alpha \ge 0$, and $n \ge 0$, we have

$$b_4\left(p^{2\alpha+1}n + \frac{(8j+1)p^{2\alpha}-1}{8}\right) \equiv 0 \pmod{2},$$

where $0 \le j \le p-1$ and $\left(\frac{8j+1}{p}\right) = -1$.

Theorem 25. For any prime $p \ge 5$, $\left(\frac{-10}{p}\right) = -1$, $\alpha \ge 0$, and $n \ge 0$, we have

$$b_5\left(4p^{2\alpha}n + \frac{(24i+7p)p^{2\alpha-1}-1}{6}\right) \equiv 0 \pmod{2},$$

where $1 \le i \le p - 1$.

Theorem 26. For any prime $p \not\equiv -1 \pmod{6}$, $\alpha \ge 0$, and $n \ge 0$, we have

$$b_9\left(2p^{2\alpha}n + \frac{(6i+p)p^{2\alpha-1}-1}{3}\right) \equiv 0 \pmod{3},$$

where $1 \leq i \leq p-1$.

Theorem 27. For $k = 0, 2, 3, 4, \alpha \ge 1$, and $n \ge 0$,

$$b_9\left(5^{2\alpha}n + \frac{(3k+2)5^{2\alpha-1}-1}{3}\right) \equiv 0 \pmod{3}.$$

Theorem 27 was conjectured by ${\rm Keith^{26}}$ and was first proven by Xia and Yao. 27

3. *l*-Regular Overpartitions

Andrews²⁸ defined the singular overpartition function $\overline{C}_{k,i}(n)$, which counts the number of overpartitions of n in which no part is divisible by k and only parts $\equiv \pm i \pmod{k}$ may be overlined. For $k \geq 3$ and $1 \leq i \leq \lfloor \frac{k}{2} \rfloor$, the generating function for $\overline{C}_{k,i}(n)$ is given by

$$\sum_{n=0}^{\infty} \overline{C}_{k,i}(n) q^n = \frac{(q^k; q^k)_{\infty}(-q^i; q^k)_{\infty}(-q^{k-i}; q^k)_{\infty}}{(q; q)_{\infty}}.$$

Note that the number of 3-regular overpartition of n is equal to the number of (3, 1)-singular overparition of n, i.e. for $n \ge 0$,

$$\overline{A}_3(n) = \overline{C}_{3,1}(n).$$

For example,

$$\overline{A}_3(4) = \overline{C}_{3,1}(4) = 10.$$

In Ref. 28, Andrews established the following congruences.

Theorem 28. For a nonnegative integer n, we have

$$\overline{C}_{3,1}(9n+3) \equiv \overline{C}_{3,1}(9n+6) \equiv 0 \pmod{3}.$$

Theorem 29. Let n be nonnegative integer and $r_2(n)$ be the number of representations of n as the sum of two squares. Then

$$\overline{C}_{3,1}(9n) \equiv (-1)^n r_2(n) \pmod{3}.$$

Chen *et al.*²⁹ studied the parity of $\overline{C}_{3,1}(n)$ and showed that $\overline{C}_{3,1}(n)$ is always even. They also prove the following results.

Theorem 30. Let N be a positive integer which is not expressible as a square or the sum of two positive squares. Then $\overline{C}_{3,1}(N) \equiv 0 \pmod{3}$.

Theorem 31. We have

$$\sum_{n=0}^{\infty} (-1)^n \overline{C}_{3,1}(n) q^n \equiv 1 + 6 \sum_{k \ge 1} q^{k^2} + 4 \sum_{k \ge 1} q^{2k^2} + 2 \sum_{k \ge 1} q^{3k^2} + 4 \sum_{k,\ell \ge 1} q^{k^2 + 3\ell^2} \pmod{8}.$$

Ahmed and Baruah³⁰ found several congruences for $\overline{C}_{3,1}(n)$ modulo 4, 18 and 36. For example, they proved the following.

Theorem 32. If $p \ge 5$ is a prime and $1 \le j \le p-1$, then for any nonnegative integers α and n, we have

$$\overline{C}_{3,1}\left(24p^{2\alpha+1}(pn+j)+p^{2\alpha+2}\right) \equiv 0 \pmod{4}.$$

In Ref. 31, Naika and Gireesh proved some infinite families of congruences for $\overline{C}_{3,1}(n)$ modulo 12, 18, 48, and 72.

Theorem 33. If $p \ge 5$ is a prime, $\left(\frac{-2}{p}\right) = -1$, $\alpha \ge 1$ and n > 0, we have

$$\overline{C}_{3,1}\left(16p^{2\alpha-1}(pn+j)+6p^{2\alpha}\right) \equiv 0 \pmod{48}.$$

They also conjectured the following.

Conjecture 2. For each nonnegative integer n,

$$\overline{C}_{3,1}(12n+11) \equiv 0 \pmod{144}.$$
The above conjecture was confirmed by Barman and the author.³² Isnaini and Toh³³ extended some of the results mentioned above and proved many new congruences modulo 108, 192, 288 and 432 for $\overline{C}_{3,1}(n)$.

In a recent work, Barman and Ray³⁴ studied the arithmetic density of $\overline{C}_{3,1}(n)$ and proved the following.

Theorem 34. Let k be a fixed positive integer and $p \in \{2,3\}$. Then $\overline{C}_{3,1}(n)$ is almost always divisible by p^k , namely,

$$\lim_{X \to \infty} \frac{\# \left\{ 0 < n \le X : \overline{C}_{3,1}(n) \equiv 0 \pmod{p^k} \right\}}{X} = 1.$$

In the same article, the authors also showed that $\overline{C}_{3,1}(n)$ is almost always divisible by $2 \cdot 3^k$ for any positive integer k. In Refs. 32 and 34, Barman and the author studied the divisibility properties for $\overline{A}_9(n)$, $\overline{A}_{2k}(n)$, and $\overline{A}_{4k}(n)$ for any positive integer k. Some of the sample results are the following.

Theorem 35. If $p \ge 5$ is a prime such that $\left(\frac{-2}{p}\right) = -1$. Then for all positive integers n, k and α , we have

$$\overline{A}_{4k}\left(4p^{2\alpha}n + (4j+3p)p^{2\alpha-1}\right) \equiv 0 \pmod{16}.$$

Theorem 36. For any positive integers k and n, we have

$$\overline{A}_9(n) \equiv \overline{A}_9(3^k n) \pmod{3}.$$

Particularly, if n is odd, then

$$\overline{A}_9(n) \equiv \overline{A}_9(3^k n) \pmod{6}.$$

Theorem 37. For i = 1, 2, ..., 6, we have

$$\overline{A}_8(28n+4i) \equiv 0 \pmod{7}.$$

In Ref. 35, Chakraborty and the author studied the arithmetic and density properties for $\overline{A}_{p_1^{a_1}p_2^{a_2}\dots p_m^{a_m}}$, where $p_i \geq 5$'s are primes. More particularly, they proved the following results.

Theorem 38. Let $\ell = p_1^{a_1} p_2^{a_2} \dots p_m^{a_m}$ where p_i 's are primes > 3. If $p_i^{2a_i} \ge \ell$, then for every positive integer j,

$$\lim_{X \to \infty} \frac{\#\left\{ 0 < n \le X : \overline{A}_{\ell}(n) \equiv 0 \pmod{p_i^j} \right\}}{X} = 1.$$

Theorem 39. Let k be a positive integer and p be a prime number such that $p \equiv 3 \pmod{4}$. Let r be a nonnegative integer such that p divides 4r + 3, then

$$\overline{A}_5\left(4p^{k+1}n + p(4r+3)\right) \equiv f(p)\overline{A}_5\left(4p^{k-1}n + \frac{4r+3}{p}\right) \pmod{8},$$

where f(p) is defined by

$$f(p) = \begin{cases} -1 & \text{if } p \equiv 3,7 \pmod{20}, \\ 1 & \text{if } p \equiv 11,19 \pmod{20}. \end{cases}$$

As a continuation of the article,³⁵ for any positive integer ℓ and k, the author³⁶ proved that the distribution for $\overline{A}_{2\ell}$ modulo 2^k is exactly one. In the same article, the author also studied the arithmetic properties of Fourier coefficients of certain integer weight modular form by using Hecke eigenforms and find infinitely many arithmetic progressions where congruences for $\overline{A}_{2\ell}(n)$ hold. One of the sample results is the following.

Theorem 40. Let k, n be nonnegative integers. For each i with $1 \le i \le k+1$, if $p_i \ge 5$ is a prime such that $p_i \equiv 3 \pmod{4}$, then for any positive integer ℓ and $j \not\equiv 0 \pmod{p_{k+1}}$, we have

$$\overline{A}_{2\ell} \left(4p_1^2 \dots p_{k+1}^2 n + (4j+p_{k+1})p_1^2 \dots p_k^2 p_{k+1} \right) \equiv 0 \pmod{4}.$$

Suppose $p \ge 5$ be a prime and $p_1 = p_2 = \cdots = p_{k+1} = p$. Then from the above theorem, we have

$$\overline{A}_{2\ell}\left(4p^{2(k+1)}n + 4p^{2k+1}j + p^{2(k+1)}\right) \equiv 0 \pmod{4},$$

where $j \not\equiv 0 \pmod{p}$. In particular, for all $n \geq 0$ and $j \not\equiv 0 \pmod{11}$, we have

$$\overline{A}_{2\ell} (484n + 44j + 121) \equiv 0 \pmod{4}.$$

Acknowledgments

The author would like to thank the National Board for Higher Mathematics (Department of Atomic Energy, India) for the postdoctoral fellowship and the Indian Statistical Institute Delhi for providing the research facilities. The author would also like to thank the anonymous referees for the careful review and comments.

References

- B. C. Berndt, Number Theory in the Spirit of Ramanujan. Vol. 34, Student Mathematical Library. American Mathematical Society, Providence, RI (2006). https://doi.org/10.1090/stml/034.
- S. Ramanujan, Some properties of p(n), the number of partitions of n [Proc. Cambridge Philos. Soc. 19 (1919), 207–210]. In Collected Papers of Srinivasa Ramanujan, pp. 210–213, AMS Chelsea Publ., Providence, RI (2000). https://doi.org/10.1016/s0164-1212(00)00033-9.
- S. Ahlgren and M. Boylan, Arithmetic properties of the partition function, *Invent. Math.* 153(3), 487–502 (2003). https://doi.org/10.1007/ s00222-003-0295-6.
- K. Ono, Distribution of the partition function modulo m, Ann. of Math. (2) 151(1), 293–307 (2000). https://doi.org/10.2307/121118.
- S. Ahlgren, Distribution of the partition function modulo composite integers M, Math. Ann. 318(4), 795–803 (2000). https://doi.org/10. 1007/s002080000142.
- T. R. Parkin and D. Shanks, On the distribution of parity in the partition function, *Math. Comp.* 21, 466–480 (1967). https://doi.org/10. 2307/2003251.
- G. E. Andrews, *The Theory of Partitions*. Cambridge Mathematical Library, Cambridge University Press, Cambridge (1998). ISBN 0-521-63766-X. Reprint of the 1976 original.
- 8. G. James and A. Kerber, *The Representation Theory of the Symmetric Group*. Addison-Wesley, Reading (1979).
- B. Gordon and K. Ono, Divisibility of certain partition functions by powers of primes, *Ramanujan J.* 1(1), 25–34 (1997). https://doi.org/ 10.1023/A:1009711020492.
- S. Corteel and J. Lovejoy, Overpartitions, Trans. Amer. Math. Soc. 356(4), 1623–1635 (2004). https://doi.org/10.1090/S0002-9947-03-03328-2.

- J. Lovejoy and D. Penniston. 3-regular partitions and a modular K3 surface. In q-Series with Applications to Combinatorics, Number Theory, and Physics (Urbana, IL, 2000), vol. 291, Contemp. Math., pp. 177–182. Amer. Math. Soc., Providence, RI (2001). https://doi.org/ 10.1090/conm/291/04901.
- K. Ono and D. Penniston, The 2-adic behavior of the number of partitions into distinct parts, J. Combin. Theory Ser. A 92(2), 138–157 (2000). https://doi.org/10.1006/jcta.2000.3057.
- K. Ono and D. Penniston, Corrigendum: "The 2-adic behavior of the number of partitions into distinct parts", J. Combin. Theory Ser. A 95(1), 196 (2001). https://doi.org/10.1006/jcta.2001.3193.
- D. Penniston, The p^a-regular partition function modulo p^j, J. Number Theory 94(2), 320–325 (2002). https://doi.org/10.1006/jnth.2001. 2736.
- D. Penniston, Arithmetic of *l*-regular partition functions, *Int. J. Number Theory* 4(2), 295–302 (2008). https://doi.org/10.1142/S1793042108001341.
- D. Penniston, 11-regular partitions and a Hecke eigenform, Int. J. Number Theory 15(6), 1251–1259 (2019). https://doi.org/10.1142/ S1793042119500696.
- B. Dandurand and D. Penniston, *l*-divisibility of *l*-regular partition functions, *Ramanujan J.* 19(1), 63–70 (2009). https://doi.org/10.1007/ s11139-007-9042-8.
- M. D. Hirschhorn and J. A. Sellers, Elementary proofs of parity results for 5-regular partitions, *Bull. Aust. Math. Soc.* 81(1), 58–63 (2010). https://doi.org/10.1017/S0004972709000525.
- N. Calkin, N. Drake, K. James, S. Law, P. Lee, D. Penniston, and J. Radder, Divisibility properties of the 5-regular and 13-regular partition functions, *Integers* 8, A60, 10 (2008). ISSN 1553-1732.
- H. Dai, Parity results for 25-regular partitions, J. Comb. Number Theory 7(3), 171–176 (2015). ISSN 1942-5600.
- E. X. W. Xia, Congruences for some *l*-regular partitions modulo *l*, J. Number Theory 152, 105–117 (2015). https://doi.org/10.1016/j.jnt. 2014.12.026.
- B. C. Berndt, Ramanujan's Notebooks. Part III. Springer-Verlag, New York (1991). https://doi.org/10.1007/978-1-4612-0965-2.
- S.-P. Cui and N. S. S. Gu, Arithmetic properties of *l*-regular partitions, *Adv. in Appl. Math.* 51(4), 507–523 (2013). https://doi.org/10.1016/j. aam.2013.06.002.
- S.-P. Cui and N. S. S. Gu, Congruences for 9-regular partitions modulo 3, *Ramanujan J.* 38(3), 503–512 (2015). https://doi.org/10.1007/ s11139-014-9586-3.

- S.-P. Cui and N. S. S. Gu, Congruences for 13-regular partitions, Util. Math. 107, 103–114 (2018). ISSN 0315-3681.
- 26. W. J. Keith, Congruences for 9-regular partitions modulo 3, *Ramanujan J.* **35**(1), 157–164 (2014). https://doi.org/10.1007/s11139-013-9522-y.
- 27. E. X. W. Xia and O. X. M. Yao, A proof of Keith's conjecture for 9-regular partitions modulo 3, Int. J. Number Theory 10(3), 669–674 (2014). https://doi.org/10.1142/S1793042113501169.
- G. E. Andrews, Singular overpartitions, Int. J. Number Theory 11(5), 1523–1533 (2015). https://doi.org/10.1142/S1793042115400059.
- S.-C. Chen, M. D. Hirschhorn, and J. A. Sellers, Arithmetic properties of Andrews' singular overpartitions, *Int. J. Number Theory* 11(5), 1463–1476 (2015). https://doi.org/10.1142/S1793042115400011.
- Z. Ahmed and N. D. Baruah, New congruences for Andrews' singular overpartitions, *Int. J. Number Theory* 11(7), 2247–2264 (2015). https://doi.org/10.1142/S1793042115501018.
- M. S. Mahadeva Naika and D. S. Gireesh, Congruences for Andrews' singular overpartitions, J. Number Theory 165, 109–130 (2016). https://doi.org/10.1016/j.jnt.2016.01.015.
- 32. R. Barman and C. Ray, Congruences for ℓ-regular overpartitions and Andrews' singular overpartitions, *Ramanujan J.* 45(2), 497–515 (2018). https://doi.org/10.1007/s11139-016-9860-7.
- U. Isnaini and P. C. Toh, On some congruences for Andrews' singular overpartitions, *Integers* 18, Paper No. A5, 17 (2018).
- 34. R. Barman and C. Ray, Divisibility of Andrews' singular overpartitions by powers of 2 and 3, *Res. Number Theory* 5(3), Paper No. 22, 7 (2019). https://doi.org/10.1007/s40993-019-0161-2.
- 35. C. Ray and K. Chakraborty, Certain eta-quotients and ℓ -regular overpartitions, *Ramanujan J.* **57**(2), 453–470 (2022). https://doi.org/10. 1007/s11139-020-00322-6.
- C. Ray, Divisibility and distribution of 2l-regular overpartitions, J. Ramanujan Math. Soc. 36(4), 331–337 (2021). ISSN 0970-1249.

© 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0008

Chapter 8

Some Applications of an Algorithm in Number Theory

Nihal Özgür

Izmir Democracy University, Department of Mathematics, 35140 Karabağlar, Izmir, Türkiye nihal.ozgur@idu.edu.tr nyozgur@gmail.com

In this chapter, we mainly deal with applications of a known algorithm (called Algorithm \mathcal{M}) in number theory obtained by the elegant theory of Möbius transformations. An extended version of Fermat's well-known "two-square theorem" can be stated as follows: "A positive integer nis the sum of two squares if -1 is a quadratic residue modulo n. Conversely, if $n = x^2 + y^2$ with (x, y) = 1 then -1 is a quadratic residue modulo n." The Algorithm \mathcal{M} is related to this extended version of Fermat's two-square theorem and computes the integers x and y satis fying the equation $n = x^2 + y^2$ for a given positive integer n such that -1 is a quadratic residue modulo n. First, we consider the positive integers n which can be written in the form $n = x^2 + Ny^2$ with (x, y) = 1 (N is a fixed positive integer and n is a given positive integer relatively prime to N). Using the group structure of the Modular group $\Gamma = PSL(2,\mathbb{Z})$, we prove that the Algorithm \mathcal{M} can be used to compute the integers x and y satisfying the equation $n = x^2 + Ny^2$ with (x,y) = 1 for $N \in \{4, 9, 16, 25\}$. Finally, we consider an open question raised in a recent paper on the existence of a quick way to compute uand v in the form $4A_p = Du^2 + pv^2$ where $p \equiv 3 \pmod{4}$ is a prime number and A_n is the well-known generalized Fibonacci sequence defined by $A_0 = 0, A_1 = 1$ and $A_n = rA_{n-1} + sA_{n-2}$ $(n \ge 2)$ (r and s are fixed integers). We discuss the usage of the *Algorithm* \mathcal{M} for the computation of u and v.

1. Introduction and Preliminaries

Let a, b, c, d be complex constants satisfying $ad - bc \neq 0$. A Möbius transformation (also called fractional linear transformation or bilinear transformation) is a rational function of the form

$$T(z) = \frac{az+b}{cz+d}.$$
(1)

The condition $ad - bc \neq 0$ is necessary to ensure a Möbius transformation is a one-to-one mapping since we have

$$T(z_1) - T(z_2) = (ad - bc) \frac{z_1 - z_2}{(cz_1 + d)(cz_2 + d)},$$

for any complex numbers z_1 and z_2 . We may assume that ad - bc = 1 by multiplying the numerator and denominator of (1) by a suitable constant. Möbius transformations are directly conformal homeomorphisms of the extended complex plane $\overline{\mathbb{C}} = \mathbb{C} \cup \{\infty\}$ (see Refs. 1–5 for the basic knowledge about Möbius transformations).

The set of all Möbius transformations is a group under composition; the identity is the function I(z) = z and the inverse of a Möbius transformation T is

$$T^{-1}(z) = \frac{-dz+b}{cz-a}.$$

The group of Möbius transformations is isomorphic to $PSL(2, \mathbb{C})$, the projective special linear group of order 2 over \mathbb{C} (see Ref. 4). The Modular group $\Gamma = PSL(2, \mathbb{Z})$ consists of all Möbius transformations with integer coefficients

$$z \to \frac{az+b}{cz+d}$$
, $a, b, c, d \in \mathbb{Z}$ and $ad-bc=1$.

All elements of Γ can also be considered as projective matrices $\pm \begin{pmatrix} a & b \\ c & d \end{pmatrix}$ with a, b, c, d rational integers and ad - bc = 1. It is

well known that Γ is generated by the transformations

$$R(z) = -\frac{1}{z}$$
 and $S(z) = -\frac{1}{z+1}$

and can be presented as

$$\Gamma = \left\langle R, S; R^2 = S^3 = 1 \right\rangle.$$
(2)

Group theoretically, Γ has the structure of a free product of the cyclic group of order 2 generated by R and the cyclic group of order 3 generated by S, that is, we have $\Gamma \cong C_2 * C_3$ (see Ref. 6). Another set of generators for the Modular group Γ consists of the transformations

$$R(z) = -\frac{1}{z}$$
 and $T(z) = z + 1$, (3)

where $T(z) = (R \circ S)(z)$.

Besides the applications in mathematics, the theory of Möbius transformations (especially the theory of Modular group) has had wide applications in a range of scientific disciplines, such as physics, engineering, and music. For example, recently Möbius transformations have been used as activation functions in complex valued neural networks. There are many studies on complex valued neural networks using some theoretical properties of Möbius transformations such as fixed points, and cross-ratio (see Refs. 7–9 and the references therein). In Ref. 10, a transformational approach to musical intervals was presented using the theory of principal congruence subgroups of the Modular group Γ .

In this paper, we deal with applications of a known algorithm (called *Algorithm* \mathcal{M}) obtained by the group theoretic properties of the Modular group. This algorithm is related to the well-known Fermat's two-square theorem in number theory. Fermat's two-square theorem states that a prime p is expressible as the sum of two squares if and only if -1 is a quadratic residue modulo p (see, for instance, Refs. 11–20 and the references therein). There are many proofs of Fermat's two-square theorem. For example, in Ref. 21, Fine presented a new proof of this theorem using the group structure of the Modular group Γ . Fine proved the following theorem.

Theorem 1.²¹ A positive integer n is the sum of two squares if -1 is a quadratic residue mod n. Conversely, if $n = x^2 + y^2$ with (x, y) = 1, then -1 is a quadratic residue mod n. Fine's result extends the two-square theorem for an arbitrary positive integer n. Such proofs only deal with the existence of the representation of a prime p (or a positive integer n) without any algorithm to find constructions for the numbers x and y in terms of p (or n). On the other hand, there are many constructions in the literature (see, for example, Refs. 19 and 22).

In Ref. 22, given a positive integer n such that -1 is a quadratic residue modulo n, an algorithm, called *Algorithm* \mathcal{M} , that computes the integers x and y in Theorem 1 was presented. To do this, some facts about the group structure of the Modular group $\Gamma = \text{PSL}(2,\mathbb{Z})$ together with the generator set (3) were used. The proof of this algorithm is based on the standard algorithm to express any projective matrix $A \in \Gamma$ in terms of the generators R and T (see Ref. 6). The *Algorithm* \mathcal{M} is briefly described as follows (see Ref. 22 for more details).

Algorithm \mathcal{M} :

1. Find two coprime integers u and q such that $u^2 = -1 + qn$.

2. Define the following two functions:

$$\begin{split} f &: (a, b, c, d) \to (d, -c, -b, a), \\ g &: (a, b, c, d) \to (a - c, 2a + b - c, c, c + d). \end{split}$$

3. Start with the quadruple (-u, n, -q, u) and apply f if the first coordinate is positive and apply g if not.

4. Proceed likewise until the quadruple (0, 1, -1, 0) is obtained.

5. For f, write $R = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}$, and for r_i times g, write $T^{r_i} = \begin{pmatrix} 1 & r_i \end{pmatrix}$

 $\begin{pmatrix} 1 & r_i \\ 0 & 1 \end{pmatrix}$. Then compute the matrix $B = T^{r_0}RT^{r_1}R\cdots RT^{r_m}$ where only r_0 and r_m may be zero.

If $B = \begin{pmatrix} x & y \\ z & t \end{pmatrix}$, then we have $n = x^2 + y^2$ with (x, y) = 1. This is an efficient algorithm for the computation of the integers x and y in Theorem 1 since it works easily even for large values of n due to matrix multiplication.

Let N be a fixed positive integer and x, y integers. There are several papers on the solution of the equation $n = x^2 + Ny^2$. Especially, primes written in this form have been studied extensively (see, for example, Refs. 16, 23–30 and the references therein). Recently, Fibonacci numbers written in this form have been investigated (see Refs. 31–34). For the case N = 1, besides the Algorithm \mathcal{M} , there are some other algorithms that compute the integers x and y satisfying the equation $n = x^2 + y^2$ for a given positive integer n such that -1 is a quadratic residue modulo n (see Refs. 16, 18, 19, 24, 35, 36). For N > 1, besides the Smith–Cornacchia algorithm (see Ref. 37), there is another algorithm (called Algorithm \mathcal{H}) that computes the integers x and y satisfying the equation $n = x^2 + Ny^2$ (for more details, see Ref. 30). For N = 2 and N = 3, the Algorithm \mathcal{H} can be used for all n which can be written as $n = x^2 + Ny^2$. For $N \ge 5$, the Algorithm \mathcal{H} was obtained by some facts about the group structure of the Hecke groups $H(\sqrt{N})$ generated by two linear fractional transformations

$$R(z) = -\frac{1}{z}$$
 and $T_{\sqrt{N}}(z) = z + \sqrt{N}$.

But in these last cases, the Algorithm \mathcal{H} can be used as long as some conditions related to the transformations of the corresponding Hecke group $H(\sqrt{N})$ hold (for more details, see Ref. 30 and the references therein). Then a natural problem arises:

Can we use the Algorithm \mathcal{M} in the cases where the Algorithm \mathcal{H} cannot be applied?

Now, we recall the *Algorithm* \mathcal{H} briefly.

Algorithm \mathcal{H} :

1. Find two coprime integers u and q such that $Nu^2 = -1 + qn$.

2. Define the following functions:

$$\begin{split} f:(a,b,c,d) &\to (d,-c,-b,a),\\ g_{\sqrt{N}}:(a,b,c,d) &\to (a-c,2Na+b-Nc,c,c+d). \end{split}$$

3. Start with the quadruple (-u, n, -q, u) and apply f if the first coordinate is positive and apply $g_{\sqrt{N}}$ if not.

4. Proceed likewise until the quadruple (0, 1, -1, 0) is obtained.

5. For f, write $R = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}$, and for r_i times $g_{\sqrt{N}}$, write $T_{\sqrt{N}}^{r_i} = \begin{pmatrix} 1 & r_i \sqrt{N} \\ 0 & 1 \end{pmatrix}$. Then compute the matrix $B = T_{\sqrt{N}}^{r_0} R T_{\sqrt{N}}^{r_1} R \cdots R T_{\sqrt{N}}^{r_m}$ where only r_0 and r_m may be zero.

N. Özgür

If $B = \begin{pmatrix} x & y\sqrt{N} \\ z\sqrt{N} & t \end{pmatrix}$, then we have $n = x^2 + Ny^2$ with (x, y) = 1; if $B = \begin{pmatrix} x\sqrt{N} & y \\ z & t\sqrt{N} \end{pmatrix}$, then we have $n = Nx^2 + y^2$ with (x, y) = 1. It is easy to compute the matrix B since we have

$$T_{\sqrt{N}}^{r} = \begin{pmatrix} 1 & r\sqrt{N} \\ 0 & 1 \end{pmatrix}, \quad T_{\sqrt{N}}^{r}R = \begin{pmatrix} -r\sqrt{N} & 1 \\ -1 & 0 \end{pmatrix}$$

and $RT_{\sqrt{N}}^{r} = \begin{pmatrix} 0 & 1 \\ -1 & -r\sqrt{N} \end{pmatrix}$

for any integer r. This is an efficient algorithm for the computation of the integers x and y for a positive integer n that can be written as $n = x^2 + Ny^2$. We note that Modular group Γ is one of the Hecke groups.

Let us consider the generalized Fibonacci sequence A_n defined by

$$A_0 = 0, A_1 = 1$$
 and $A_n = rA_{n-1} + sA_{n-2} (n \ge 2),$

where r and s are fixed integers. Let $p \equiv 3 \pmod{4}$ be a prime number. In Ref. 38, it was proved that $4A_p$ is represented by the quadratic form

$$Du^2 + pv^2. (4)$$

Here, $D = r^2 + 4s$ is the discriminant of the sequence. Then an open question was left on the existence of a quick way to compute u and v in representation (4) (see Remark on page 138 in Ref. 38).

In this paper, our aim is to explore some applications of the Algorithm \mathcal{M} . In Section 2, we prove that the Algorithm \mathcal{M} can be used to find the integers x and y satisfying the equation $n = x^2 + Ny^2$ with (x, y) = 1 for $N \in \{9, 16, 25\}$. Our arguments work also in the case of N = 4. Although there are many simple proofs of the case N = 4 in the literature, we include this case in our main theorem. Some remarks and necessary examples are also given. We see that the Algorithm \mathcal{M} is an efficient tool to provide a simple solution to the problem when a natural number n (relatively prime to N) can be represented in the form $n = x^2 + Ny^2$ with the advantage of calculating the integers x and y. In Section 3, we discuss the usage of the Algorithm \mathcal{M} for the open question on the existence

174

of a quick way to compute u and v in (4). We refer the reader to Refs. 13, 14 and 39 for the basic information on the quadratic reciprocity.

2. An Application of the Algorithm \mathcal{M} to the Computation of the Integers x and y Satisfying the Equation $n = x^2 + Ny^2$

First, we begin with an example that shows us how the Algorithm ${\mathcal M}$ works.

Example 1. Let us consider the prime number p = 1289. Using the Euler criterion, it is easy to check that -1 is a quadratic residue mod p (see, for example, Refs. 13 and 19 for more details). By Fermat's two-square theorem (also Theorem 1), we know that the prime number p = 1289 can be written as the sum of two squares, that is, we can write $1289 = x^2 + y^2$ for some integers x and y. Now, we want to compute the integers x and y using the *Algorithm* \mathcal{M} . We can find the integers 810,509 such that $(810)^2 = -1 + 1289 \cdot 509$. Then, the matrix

$$A = \begin{pmatrix} -810 & 1289\\ -509 & 810 \end{pmatrix}$$

is in the Modular group Γ and we can apply the *Algorithm* \mathcal{M} . We have

$$\begin{array}{c} (-810, 1289, -509, 810) \underbrace{g} (-301, 178, -509, 301) \\ \underbrace{g} (208, 85, -509, -208) \\ \underbrace{f} (-208, 509, -85, 208) \underbrace{g} (-123, 178, -85, 123) \underbrace{g} (-38, 17, -85, 38) \\ \underbrace{g} (47, 26, -85, -47) \underbrace{f} (-47, 85, -26, 47) \underbrace{g} (-21, 17, -26, 21) \\ \underbrace{g} (5, 1, -26, -5) \underbrace{f} (-5, 26, -1, 5) \underbrace{g} (-4, 17, -1, 4) \underbrace{g} (-3, 10, -1, 3) \\ \underbrace{g} (-2, 5, -1, 2) \underbrace{g} (-1, 2, -1, 1) \underbrace{g} (0, 1, -1, 0). \end{array}$$

Then, we obtain $B = T^2 R T^3 R T^2 R T^5$. If we compute the matrix B, we get

$$B = \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -3 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & 5 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} -8 & -35 \\ -5 & -22 \end{pmatrix}.$$

Hence, we find $1289 = (8)^2 + (35)^2 = 64 + 1225$.

Let N be a fixed positive integer. Now, we investigate whether the Algorithm \mathcal{M} can be used to calculate the integers x and y satisfying the equation $n = x^2 + Ny^2$ with (x, y) = 1 for a given natural number n relatively prime to N. We see that this is possible for the cases N = 4, 9, 16 or 25.

From now on, we will assume that N = 4, 9, 16 or 25, $\alpha = \sqrt{N}$, $n > 0, n \in \mathbb{N}$ and (n, N) = 1.

Theorem 2. Let $N \in \{4, 9, 16, 25\}$. Suppose that n is a natural number relatively prime to N.

- (i) Let N = 4. If -4 is a quadratic residue mod n, then the Algorithm M can be used to find an integer solution x, y of the equation n = x² + 4y² with (x, y) = 1.
- (ii) Let $N \in \{9, 25\}$. If -N is a quadratic residue mod n and n is a quadratic residue mod α , then the Algorithm \mathcal{M} can be used to find an integer solution x, y of the equation $n = x^2 + Ny^2$ with (x, y) = 1.
- (iii) Let N = 16. If -N is a quadratic residue mod n, n is a quadratic residue mod α and $n \equiv 1, -7 \pmod{16}$, then the Algorithm \mathcal{M} can be used to find an integer solution x, y of the equation $n = x^2 + Ny^2$ with (x, y) = 1.

Proof. Assume that -N is a quadratic residue mod n. We can find some integers k, l such that kN - ln = 1 since (n, N) = 1. This implies $kN \equiv 1 \mod n$, and so -k is a quadratic residue mod n. Then $u^2 \equiv -k \mod n$ for some integer u. We get $u^2N \equiv -kN \mod n$, $u^2N \equiv -1 \mod n$, and so we have

$$u^2 N = -1 + qn \tag{5}$$

for some integer q. Define the matrix

$$A = \begin{pmatrix} -u\alpha & n \\ -q & u\alpha \end{pmatrix}.$$
 (6)

The determinant of the matrix A is $-u^2N + qn = 1$, where $\alpha = \sqrt{N}$. Clearly, $A \in \Gamma$ and A has order 2 since trA = 0. Now, each element of order 2 in Γ is conjugate to the generator R because of the group structure $\Gamma \cong C_2 * C_3$. That is, we have $A = BRB^{-1}$ for some $B \in \Gamma$. Let $B = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$; $a, b, c, d \in \mathbb{Z}, ad - bc = 1$. Then we obtain

$$A = \begin{pmatrix} -(ac+bd) & a^2+b^2 \\ -(c^2+d^2) & (ac+bd) \end{pmatrix}.$$
 (7)

Comparing the entries in (6) and (7), we find

$$n = a^2 + b^2 \tag{8}$$

for some integers a, b. By the determinant condition ad - bc = 1, we see that (a, b) = 1. Also we have

$$q = c^2 + d^2 \text{ and } u\alpha = ac + bd \tag{9}$$

with (c, d) = 1.

Now, if we show that $\alpha \mid a \text{ or } \alpha \mid b$, then the proof is completed. Indeed, rearranging (8), we have

$$n = (\alpha r)^2 + (b)^2 = Nr^2 + b^2$$

or

$$n = a^2 + (\alpha s)^2 = a^2 + Ns^2,$$

according to $a = \alpha r$ or $b = \alpha s$ $(r, s \in \mathbb{Z})$, respectively. It is evident that (b, r) = 1 or (a, s) = 1.

For $\alpha = 3, 4, 5$, we use the extra hypothesis that *n* is a quadratic residue mod α to show that $\alpha \mid a$ or $\alpha \mid b$.

Case 1. Let N = 4. By Eq. (5), we have $nq \equiv 1 \pmod{2}$. So, we find $n \equiv 1 \pmod{2}$ and $q \equiv 1 \pmod{2}$. $n = a^2 + b^2 \equiv 1 \pmod{2}$ implies $a \equiv 0 \pmod{2}$ or $b \equiv 0 \pmod{2}$. Similarly, $q = c^2 + d^2 \equiv 1 \pmod{2}$ implies $c \equiv 0 \pmod{2}$ or $d \equiv 0 \pmod{2}$. By the determinant

condition ad - bc = 1, we find $a \equiv 0 \pmod{2}$ and $d \equiv 0 \pmod{2}$ or $b \equiv 0 \pmod{2}$ and $c \equiv 0 \pmod{2}$. That is, $2 \mid a \text{ or } 2 \mid b$.

Case 2. Let N = 9. Using the condition n is a quadratic residue modulo 3, we shall prove that $3 \mid a$ or $3 \mid b$. By Eq. (5), we have $nq \equiv 1 \pmod{3}$ and (3,q) = 1. If n is a quadratic residue modulo 3, we obtain $n \equiv 1 \pmod{3}$ by Euler's criterion. The conditions $n \equiv 1 \pmod{3}$ and $nq \equiv 1 \pmod{3}$ imply $q \equiv 1 \pmod{3}$. Then $n = a^2 + b^2 \equiv 1 \pmod{3}$ implies $a \equiv 0 \pmod{3}$ or $b \equiv 0 \pmod{3}$. Similarly, $q = c^2 + d^2 \equiv 1 \pmod{3}$ implies $c \equiv 0 \pmod{3}$ or $d \equiv 0 \pmod{3}$. By the determinant condition ad - bc = 1, we find $a \equiv 0 \pmod{3}$ and $d \equiv 0 \pmod{3}$ or $b \equiv 0 \pmod{3}$ and $c \equiv 0 \pmod{3}$. This implies $3 \mid a$ or $3 \mid b$.

Case 3. Let N = 16. By Eq. (5), we have $nq \equiv 1 \pmod{4}$ and (4,q) = 1. Using the condition n is a quadratic residue modulo 4, we shall prove that $4 \mid a$ or $4 \mid b$. We know that the only quadratic residues modulo 4 are 0 and 1. Hence, we have $n \equiv 1 \pmod{4}$ and $q \equiv 1 \pmod{4}$. Considering the condition (n, N) = 1, it is easy to check that n should be congruent to 1 or -7 modulo 16. Then, taking into account the determinant condition ad - bc = 1, it is easy to check that $n = a^2 + b^2 \equiv 1 \pmod{4}$ implies $a \equiv 0 \pmod{4}$ or $b \equiv 0 \pmod{4}$. Similarly, $q = c^2 + d^2 \equiv 1 \pmod{4}$ implies $c \equiv 0 \pmod{4}$ or $b \equiv 0 \pmod{4}$. The determinant condition ad - bc = 1 also implies that $a \equiv 0 \pmod{4}$ and $d \equiv 0 \pmod{4}$ or $b \equiv 0 \pmod{4}$ and $c \equiv 0 \pmod{4}$.

Case 4. Let N = 25. Using the condition n is a quadratic residue modulo 5, we prove that $5 \mid a$ or $5 \mid b$. By Eq. (5), we have $nq \equiv 1 \pmod{5}$ and (5, q) = 1. If n is a quadratic residue modulo 5, we obtain $n^2 \equiv 1 \pmod{5}$ by Euler's criterion. $nq \equiv 1 \pmod{5}$ implies $(nq)^2 \equiv 1 \pmod{5}$ and $q^2 \equiv 1 \pmod{5}$. Then the possible cases are $n \equiv 1 \pmod{5}$, $q \equiv 1 \pmod{5}$ or $n \equiv 4 \pmod{5}$, $q \equiv 4 \pmod{5}$.

 $n = a^2 + b^2 \equiv 1 \pmod{5}$ implies $a \equiv 0 \pmod{5}$ or $b \equiv 0 \pmod{5}$ and $q = c^2 + d^2 \equiv 1 \pmod{5}$ implies $c \equiv 0 \pmod{5}$ or $d \equiv 0 \pmod{5}$. By the determinant condition ad - bc = 1, we find $a \equiv 0 \pmod{5}$ and $d \equiv 0 \pmod{5}$ or $b \equiv 0 \pmod{5}$ and $c \equiv 0 \pmod{5}$. Similarly, $n = a^2 + b^2 \equiv 4 \pmod{5}$ implies $a \equiv 0 \pmod{5}$ or $b \equiv 0 \pmod{5}$ and $q = c^2 + d^2 \equiv 4 \pmod{5}$ implies $c \equiv 0 \pmod{5}$ or $d \equiv 0 \pmod{5}$. Again, we find $a \equiv 0 \pmod{5}$ and $d \equiv 0 \pmod{5}$ or $b \equiv 0 \pmod{5}$ and $c \equiv 0 \pmod{5}$ by the determinant condition. Consequently, the Algorithm \mathcal{M} can be used to compute the integers x and y satisfying the equation $n = x^2 + Ny^2$ with $x, y \in \mathbb{Z}$, (x, y) = 1 for N = 4, 9, 16, 25 under the hypothesis of the theorem.

Corollary 1. Let N be as defined in Theorem 2 and n = p be a prime number which does not divide N. Then, the Algorithm \mathcal{M} can be used to find an integer solution (x, y) of the equation $p = x^2 + Ny^2$ with (x, y) = 1 if p satisfies the following condition:

N	p
N = 4	$p \equiv 1 \pmod{4}$
N = 9	$p \equiv 1 \pmod{12}$
N = 16	$p \equiv -7,1 \pmod{16}$
N = 25	$p \equiv -11, 1 \pmod{20}$

Note that the Algorithm \mathcal{M} enables us to find the matrix B in the proof of Theorem 2. Now, we adapt this algorithm to our cases.

Application of the Algorithm: Let $N \in \{4, 9, 16, 25\}$ be fixed. Suppose that all of the assumptions of Theorem 2 hold for N and a given positive integer n. Let u and q(>N) be the integers satisfying the equation $Nu^2 = -1 + qn$. If we start with the quadruple $(-u\alpha, n, -q, u\alpha)$ and apply the Algorithm \mathcal{M} , we obtain the matrix $B = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$. Then we find $n = a^2 + b^2$ with $a, b \in \mathbb{Z}$, $(a, b) = 1, \alpha \mid a$ or $\alpha \mid b$ for $\alpha = 2, 3, 4, 5$. By rearranging the equality $n = a^2 + b^2$, we obtain the integers x and y satisfying the equation $n = x^2 + Ny^2$.

Remark 1. (1) For any integer r, as we have

$$T^r = \begin{pmatrix} 1 & r \\ 0 & 1 \end{pmatrix}, T^r R = \begin{pmatrix} -r & 1 \\ -1 & 0 \end{pmatrix}$$
 and $RT^r = \begin{pmatrix} 0 & 1 \\ -1 & -r \end{pmatrix}$,

the matrix B can be computed easily using these equalities.

(2) In Example 1, we have easily found $1289 = (8)^2 + (35)^2$ via the *Algorithm* \mathcal{M} . Rearranging this equality, we get $1289 = (8)^2 + 25(7)^2$. That is, this last equality is also the solution of the problem that the calculation of the integers x, y satisfying the equation $n = x^2 + 25y^2$ with (x, y) = 1 for the given prime number 1289. Note that -25 is

a quadratic residue mod 1289 and 1289 is a quadratic residue mod $\alpha = 5$, then Theorem 2 guarantees that the Algorithm \mathcal{M} can be used to calculate the integers x, y in the equation $n = x^2 + 25y^2$. Note that the equality $1289 = (8)^2 + (35)^2$ can also be rearranged as $1289 = 4(4)^2 + (35)^2$.

(3) Let us consider the matrix A defined in (6). The Algorithm \mathcal{H} can also be used to compute the integers x and y satisfying the equation $n = x^2 + Ny^2$ with (x, y) = 1 for a given positive integer n. But, in the cases $N \geq 5$, the Algorithm \mathcal{H} can be used as long as the following conditions hold:

- (i) -N is a quadratic residue mod n,
- (ii) n is a quadratic residue mod N,
- (iii) $A \in H(\sqrt{N})$ hold (see Ref. 30 for more details). From Ref. 40 we know that $A \in H(\sqrt{N})$ if and only if $\frac{u\alpha}{q}$ is a finite \sqrt{N} -fraction. Since we are unable to give the explicit conditions which determine whether $A \in H(\sqrt{N})$ or not, it is not practical to use the *Algorithm* \mathcal{H} for $N \geq 5$. However, in the cases where the *Algorithm* \mathcal{H} can be applied, we see that the *Algorithm* \mathcal{H} is more efficient than the *Algorithm* \mathcal{M} (see Example 3).

In the following example, we see an example of a case that we cannot use the Algorithm \mathcal{H} but can apply the Algorithm \mathcal{M} .

Example 2. Let N = 9 and consider the prime number $p = 17257 \equiv 1 \pmod{12}$. We find the integers 342, 61 such that $9(342)^2 = -1 + 17257 \cdot 61$. Using the *Algorithm* \mathcal{M} , we have the following:

$$\begin{array}{c} (-1026, 17257, -61, 1026) \underbrace{g} (-965, 15266, -61, 965) \\ \underbrace{g} (-904, 13397, -61, 904) \underbrace{g} (-843, 11650, -61, 843) \\ \underbrace{g} (-782, 10025, -61, 782) \underbrace{g} (-721, 8522, -61, 721) \\ \underbrace{g} (-660, 7141, -61, 660) \underbrace{g} (-599, 5882, -61, 599) \\ \underbrace{g} (-538, 4745, -61, 538) \underbrace{g} (-477, 3730, -61, 477) \end{array}$$

$$\begin{array}{c} \underline{g} \left(-416, 2837, -61, 416\right) \underline{g} \left(-355, 2066, -61, 355\right) \\ \underline{g} \left(-294, 1417, -61, 294\right) \underline{g} \left(-233, 890, -61, 233\right) \\ \underline{g} \left(-172, 485, -61, 172\right) \underline{g} \left(-111, 202, -61, 111\right) \\ \underline{g} \left(-50, 41, -61, 50\right) \underline{g} \left(11, 2, -61, -11\right) \\ \underline{f} \left(-11, 61, -2, 11\right) \underline{g} \left(-9, 41, -2, 9\right) \underline{g} \left(-7, 25, -2, 7\right) \\ \underline{g} \left(-5, 13, -2, 5\right) \underline{g} \left(-3, 5, -2, 3\right) \underline{g} \left(-1, 1, -2, 1\right) \\ \underline{g} \left(1, 1, -2, -1\right) \underline{f} \left(-1, 2, -1, 1\right) \underline{g} \left(0, 1, -1, 0\right). \end{array}$$

Then $B = T^{17}RT^6RT$. If we compute the matrix B, we obtain

$$B = \begin{pmatrix} -17 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -6 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 101 & 84 \\ 6 & 5 \end{pmatrix}.$$

Then we find $17257 = (101)^2 + (84)^2 = (101)^2 + 9(28)^2$. Note that the matrix $A = \begin{pmatrix} -1026 \ 17257 \\ -61 \ 1026 \end{pmatrix} = \begin{pmatrix} -342\sqrt{9} \ 17257 \\ -61 \ 342\sqrt{9} \end{pmatrix}$ is in Γ , but we have $A \notin H(\sqrt{9})$ and so we cannot use the *Algorithm* \mathcal{H} . Indeed, using the nearest integer algorithm, we find

$$\frac{342\sqrt{9}}{61} = 6\sqrt{9} - \frac{1}{\frac{61\sqrt{9}}{216}}$$

Hence, $\frac{1026}{61} = \frac{342\sqrt{9}}{61}$ cannot be expanded into a finite $\sqrt{9}$ -fraction.

Example 3. Let N = 4 and $p = 1109 \equiv 1 \pmod{4}$. We can find the integers 177, 113 such that $4(177)^2 = -1 + 113 \cdot 1109$. Then, the matrix

$$A = \begin{pmatrix} -354 & 1109\\ -113 & 354 \end{pmatrix} = \begin{pmatrix} -177\sqrt{4} & 1109\\ -113 & 177\sqrt{4} \end{pmatrix}$$

N. Özgür

is in the Hecke group $H(\sqrt{4})$. Hence, by Theorem 2, we can use both the *Algorithm* \mathcal{M} and the *Algorithm* \mathcal{H} to calculate the integers x, ysatisfying the equation $n = x^2 + 4y^2$. First, we use the *Algorithm* \mathcal{M} :

$$\begin{array}{c} (-354,1109,-113,354) \underbrace{g} (-241,514,-113,241) \\ \underbrace{g} (-128,145,-113,128) \underbrace{g} (-15,2,-113,15) \\ \underbrace{g} (98,85,-113,-98) \underbrace{f} (-98,113,-85,98) \\ \underbrace{g} (-13,2,-85,13) \underbrace{g} (72,61,-85,-72) \underbrace{f} (-72,85,-61,72) \\ \underbrace{g} (-11,2,-61,11) \underbrace{g} (50,41,-61,-50) \underbrace{f} (-50,61,-41,50) \\ \underbrace{g} (-9,2,-41,9) \underbrace{g} (32,25,-41,-32) \underbrace{f} (-32,41,-25,32) \\ \underbrace{g} (-7,2,-25,7) \underbrace{g} (18,13,-25,-18) \underbrace{f} (-18,25,-13,18) \\ \underbrace{g} (-5,2,-13,5) \underbrace{g} (8,5,-13,-8) \underbrace{f} (-8,13,-5,8) \\ \underbrace{g} (-1,2,-1,1) \underbrace{g} (0,1,-1,0). \end{array}$$

Then $B = T^4 R (T^2 R)^6 T^2$. If we compute the matrix B, we obtain

$$B = \begin{pmatrix} -4 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix}^{6} \begin{pmatrix} 1 & 2 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} -22 & -25 \\ -7 & -8 \end{pmatrix}.$$

So, we find $1109 = (22)^2 + (25)^2 = 4(11)^2 + (25)^2$, that is, we have x = 25 and y = 11.

Now, we use the Algorithm \mathcal{H} to make a comparison. Using the Algorithm \mathcal{H} , we obtain

$$\begin{array}{c} (-177, 1109, -113, 177) \xrightarrow{g_{\sqrt{4}}} (-64, 145, -113, 64) \\ \xrightarrow{g_{\sqrt{4}}} (49, 85, -113, -49) \xrightarrow{f} (-49, 113, -85, 49) \\ \xrightarrow{g_{\sqrt{4}}} (36, 61, -85, -36) \xrightarrow{f} (-36, 85, -61, 36) \end{array}$$

$$\begin{split} & \underbrace{g_{\sqrt{4}}\left(25,41,-61,-25\right)}_{\xrightarrow{f}} \left(-25,61,-41,25\right) \underbrace{g_{\sqrt{4}}\left(16,25,-41,-16\right)}_{\xrightarrow{f}} \\ & \underbrace{f_{\rightarrow}\left(-16,41,-25,16\right)}_{\xrightarrow{f}} \underbrace{g_{\sqrt{4}}\left(9,13,-25,-9\right)}_{\xrightarrow{f}} \underbrace{f_{\rightarrow}\left(-9,25,-13,9\right)}_{\xrightarrow{f}} \\ & \underbrace{g_{\sqrt{4}}\left(4,5,-13,-4\right)}_{\xrightarrow{f}} \underbrace{f_{\rightarrow}\left(-4,13,-5,4\right)}_{\xrightarrow{f}} \underbrace{g_{\sqrt{4}}\left(1,1,-5,-1\right)}_{\xrightarrow{f}} \\ & \underbrace{f_{\rightarrow}\left(-1,5,-1,1\right)}_{\xrightarrow{f}} \underbrace{g_{\sqrt{4}}\left(0,1,-1,0\right)}_{\xrightarrow{f}} \end{split}$$

Then, we find

$$B = T_{\sqrt{4}}^2 R(T_{\sqrt{4}}R)^6 T_{\sqrt{4}} = \begin{pmatrix} -2\sqrt{4} & 1\\ -1 & 0 \end{pmatrix} \begin{pmatrix} -\sqrt{4} & 1\\ -1 & 0 \end{pmatrix}^6 \begin{pmatrix} 1 & \sqrt{4}\\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} -11\sqrt{4} & -25\\ -7 & -4\sqrt{4} \end{pmatrix},$$

and hence, $1109 = 4(11)^2 + (25)^2$.

Remark 2. We note that our arguments do not work for the general cases $N = \alpha^2$, $\alpha \ge 6$, because of the reason that the conditions -N is a quadratic residue mod n and n is a quadratic residue mod α are not sufficient to show that $\alpha \mid a$ or $\alpha \mid b$. For example, let N = 36 and n = 1237. -36 is a quadratic residue mod 1237 and 1237 is a quadratic residue mod 6. We find the integers 91,241 such that $36(91)^2 = -1 + 1237 \cdot 241$. Using the *Algorithm* \mathcal{M} , we have

$$\begin{array}{c} (-546, 1237, -241, 546) \underbrace{g} (-305, 386, -241, 305) \\ \underbrace{g} (-64, 17, -241, 64) \underbrace{g} (177, 130, -241, -177) \\ \underbrace{f} (-177, 241, -130, 177) \underbrace{g} (-47, 17, -130, 47) \\ \underbrace{g} (83, 53, -130, -83) \underbrace{f} (-83, 130, -53, 83) \underbrace{g} (-30, 17, -53, 30) \\ \underbrace{g} (23, 10, -53, -23) \underbrace{f} (-23, 53, -10, 23) \underbrace{g} (-13, 17, -10, 13) \\ \underbrace{g} (-3, 1, -10, 3) \underbrace{g} (7, 5, -10, -7) \underbrace{f} (-7, 10, -5, 7) \underbrace{g} (-2, 1, -5, 2) \\ \underbrace{g} (3, 2, -5, -3) \underbrace{f} (-3, 5, -2, 3) \underbrace{g} (-1, 1, -2, 1) \underbrace{g} (1, 1, -2, -1) \\ \underbrace{f} (-1, 2, -1, 1) \underbrace{g} (0, 1, -1, 0). \end{array}$$

Then, we get $B = T^3 R (T^2 R)^2 T^3 R (T^2 R)^2 T$. If we compute the matrix B, we obtain

$$B = \begin{pmatrix} -3 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix}^2 \begin{pmatrix} -3 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix}^2 \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 34 & 9 \\ 15 & 4 \end{pmatrix}.$$

Consequently, we find $1237 = (34)^2 + (9)^2$. But, neither 34 nor 9 is divisible by 6. Hence, 1237 cannot be written in the form $x^2 + 36y^2$. If we find the sufficient conditions, our arguments used in the proof of Theorem 2 are valid, and hence, we can use the *Algorithm* \mathcal{M} for all $N = \alpha^2$, $\alpha \ge 6$.

Now, before testing the efficiency of the Algorithm \mathcal{M} for some large values of n, we give a theorem for the reduction of the steps of the Algorithm \mathcal{M} .

Theorem 3. Suppose all of the assumptions of Theorem 2 hold. Consider the integers $u\alpha$ and q obtained from (5). If we have $u\alpha = mq - \delta$ where $0 \leq \delta < q$, then applying the Algorithm \mathcal{M} to the matrix $T^{-m}AT^m$ reduces the steps of the Algorithm \mathcal{M} by m times.

Proof. The proof is based on the fact that the matrix A, defined in (6), must be conjugate to R in Γ . The purpose of the Algorithm \mathcal{M} is to find the matrix B such that $A = BRB^{-1}$, and so, $B^{-1}AB = R$. It is well known that every element of Γ can be expressed as a word in R and T. So, we can write $B = T^{r_0}RT^{r_1}R\cdots RT^{r_n}$ where the r_i , (0 < i < n) are integers and only r_0 and r_n may be zero. Then we have

$$R = B^{-1}AB = (T^{-r_n}R\cdots RT^{-r_1}RT^{-r_0})A(T^{r_0}RT^{r_1}R\cdots RT^{r_n})$$

= $T^{-r_n}R\cdots RT^{-r_1}R(T^{-r_0}AT^{r_0})RT^{r_1}R\cdots RT^{r_n}.$

Since f represents the coefficients of the matrix RXR and g represents the coefficients of the matrix $T^{-1}XT$ for any matrix $X = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma$, then the proof follows easily. \Box

Example 4. Let N = 16 and consider the prime number $p = 426161 \equiv 1 \pmod{16}$. We find the integers 2996, 337 such that

 $16(2996)^2 = -1 + 426161 \cdot 337$. Since we have $11984 = 36 \cdot 337 - 148$, we shall apply the *Algorithm* \mathcal{M} to the matrix

$$T^{-36}AT^{36} = \begin{pmatrix} 1 & -36\\ 0 & 1 \end{pmatrix} \begin{pmatrix} -11984 & 426161\\ -337 & 11984 \end{pmatrix} \begin{pmatrix} 1 & 36\\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 148 & 65\\ -337 & -148 \end{pmatrix}.$$

We obtain

$$\begin{array}{c} (148, 65, -337, -148) \underbrace{f}_{\rightarrow} (-148, 337, -65, 148) \underbrace{g}_{\rightarrow} (-83, 106, -65, 83) \\ \underbrace{g}_{\rightarrow} (-18, 5, -65, 18) \underbrace{g}_{\rightarrow} (47, 34, -65, -47) \underbrace{f}_{\rightarrow} (-47, 65, -34, 47) \\ \underbrace{g}_{\rightarrow} (-13, 5, -34, 13) \underbrace{g}_{\rightarrow} (21, 13, -34, -21) \underbrace{f}_{\rightarrow} (-21, 34, -13, 21) \\ \underbrace{g}_{\rightarrow} (-8, 5, -13, 8) \underbrace{g}_{\rightarrow} (5, 2, -13, -5) \underbrace{f}_{\rightarrow} (-5, 13, -2, 5) \underbrace{g}_{\rightarrow} (-3, 5, -2, 3) \\ \underbrace{g}_{\rightarrow} (-1, 1, -2, 1) \underbrace{g}_{\rightarrow} (1, 1, -2, -1) \underbrace{f}_{\rightarrow} (-1, 2, -1, 1) \underbrace{g}_{\rightarrow} (0, 1, -1, 0). \end{array}$$

Then, we get $B = T^{36} R T^3 R (T^2 R)^2 T^3 R T$. If we compute the matrix B, we obtain

$$B = \begin{pmatrix} -36 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -3 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix}^2 \begin{pmatrix} -3 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} -569 & -320 \\ -16 & -9 \end{pmatrix}.$$

Consequently, we find $426161 = (569)^2 + (320)^2 = (569)^2 + 16(80)^2$, and so, we have x = 569 and y = 80.

Example 5. Let N = 25. We consider the prime number $p = 813089 \equiv -11 \pmod{100}$. We shall use the reduced version of the Algorithm \mathcal{M} to find the numbers x, y satisfying the equation $n = x^2 + 25y^2$. We find the integers 4802,709 such that $25(4802)^2 = -1 + 813089 \cdot 709$. Since we have $24010 = 34 \cdot 709 - 96$,

N. Özgür

we shall apply the *Algorithm* \mathcal{M} to the matrix

$$T^{-34}AT^{34} = \begin{pmatrix} 1 & -34 \\ 0 & 1 \end{pmatrix} \begin{pmatrix} -24010 & 813089 \\ -709 & 24010 \end{pmatrix} \begin{pmatrix} 1 & 34 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 96 & 13 \\ -709 & -96 \end{pmatrix}.$$

We find

$$\begin{array}{c} (96, 13, -709, -96) \underbrace{f} (-96, 709, -13, 96) \underbrace{g} (-83, 530, -13, 83) \\ \underbrace{g} (-70, 377, -13, 70) \underbrace{g} (-57, 250, -13, 57) \underbrace{g} (-44, 149, -13, 44) \\ \underbrace{g} (-31, 74, -13, 31) \underbrace{g} (-18, 25, -13, 18) \underbrace{g} (-5, 2, -13, 5) \\ \underbrace{g} (8, 5, -13, -8) \underbrace{f} (-8, 13, -5, 8) \underbrace{g} (-3, 2, -5, 3) \\ \underbrace{g} (2, 1, -5, -2) \underbrace{f} (-2, 5, -1, 2) \underbrace{g} (-1, 2, -1, 1) \underbrace{g} (0, 1, -1, 0). \end{array}$$

Then, we get $B = T^{34}RT^8RT^2RT^2$. If we compute the matrix B, we obtain

$$B = \begin{pmatrix} -34 & 1\\ -1 & 0 \end{pmatrix} \begin{pmatrix} -8 & 1\\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1\\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & 2\\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} -508 & -745\\ -15 & -22 \end{pmatrix}.$$

Consequently, we find $813089 = (508)^2 + (745)^2 = (508)^2 + 25(149)^2$, and so, we have x = 508 and y = 149.

3. Application of the Algorithm \mathcal{M} to an Open Question

Let $p \equiv 3 \pmod{4}$ be a prime number. In this section, we consider the open question on the computation of the numbers u and v in the representation $4A_p = Du^2 + pv^2$. This is a hard problem for general values of r and s. It seems that it is difficult to find a fast algorithm for solving it.

The general expression of the first 15 generalized Fibonacci numbers is presented in Table 1.

n	A_n
0	0
1	1
2	r
3	$r^2 + s$
4	$r^3 + 2rs$
5	$r^4 + 3r^2s + s^2$
6	$r^{5} + 4r^{3}s + 3rs^{2}$
7	$r^6 + 5r^4s + 6r^2s^2 + s^3$
8	$r^7 + 6r^5s + 10r^3s^2 + 4rs^3$
9	$r^8 + 7r^6s + 15r^4s^2 + 10r^2s^3 + s^4$
10	$r^9 + 8r^7s + 21r^5s^2 + 20r^3s^3 + 5rs^4$
11	$r^{10} + 9r^8s + 28r^6s^2 + 35r^4s^3 + 15r^2s^4 + s^5$
12	$r^{11} + 10r^9s + 36r^7s^2 + 56r^5s^3 + 35r^3s^4 + 6rs^5$
13	$r^{12} + 11r^{10}s + 45r^8s^2 + 84r^6s^3 + 70r^4s^4 + 21r^2s^5 + s^6$
14	$r^{13} + 12r^{11}s + 55r^9s^2 + 120r^7s^3 + 126r^5s^4 + 56r^3s^5 + 7rs^6$
15	$r^{14} + 13r^{12}s + 66r^{10}s^2 + 165r^8s^3 + 210r^6s^4 + 126r^4s^5 + 28r^2s^6 + s^7$

Table 1. The list of the first 15 generalized Fibonacci numbers.

The following formula is well known:

$$A_{2n+1} = A_{n+1}^2 + sA_n^2. (10)$$

For the basic facts about the generalized Fibonacci sequence A_n , one can consult Refs. 41 and 42. Replacing n with 2m + 1, we get

$$A_{4m+3} = A_{2m+2}^2 + sA_{2m+1}^2.$$
(11)

For p = 3, from Ref. 38, we know that

$$4A_3 = D(1)^2 + 3(r)^2, (12)$$

that is, we have u = 1 and v = r. Since $A_0 = 0$ and $A_1 = 1$, this can also be obtained from (11) as follows:

$$4A_3 = 4A_2^2 + 4sA_1^2 = 4[rA_1 + sA_0]^2 + 4sA_1^2$$
$$= (r^2 + 4s)A_1^2 + 3r^2A_1^2.$$

N. Özgür

For p = 7, by (11), observe that we have

$$A_7 = A_4^2 + sA_3^2, (13)$$

and so,

$$4A_7 = 4A_4^2 + 4sA_3^2.$$

It is difficult to rearrange this last equation using the properties of generalized Fibonacci numbers for general r and s.

Now, we discuss the possible applications of the Algorithm \mathcal{M} to find the integers u and v in the representation $4A_p = Du^2 + pv^2$ for a given prime number $p \equiv 3 \pmod{4}$. If -1 is a quadratic residue modulo $2A_p$, then we can find the integers a, b with $a^2 = -1 + 2bA_p$. Then the matrix

$$A = \begin{pmatrix} -a & 2A_p \\ -b & a \end{pmatrix}$$

is in Γ and has order 2. Then, we can apply the *Algorithm* \mathcal{M} to find the integers α and β such that $2A_p = \alpha^2 + \beta^2$ (see Ref. 22 for more details). Rearranging this last equation, we get the integers u and v such that $4A_p = Du^2 + pv^2$.

Now, we give some illustrative examples.

Example 6. Let p = 7, r = 2 and s = 1. Then D = 8. By (11), we have $4A_7 = 4A_4^2 + 4A_3^2$. Observe that 7 (resp. 8) does not divide A_4 or A_3 . For the integer $2A_7 = 338$, we obtain the equation $29.338 - 99^2 = 1$. Then, the matrix

$$A = \begin{pmatrix} -99 & 338\\ -29 & 99 \end{pmatrix}$$

is in Γ and has order 2. Applying the *Algorithm* \mathcal{M} to the matrix

$$T^{-4}AT^{4} = \begin{pmatrix} 1 & -4 \\ 0 & 1 \end{pmatrix} \begin{pmatrix} -99 & 338 \\ -29 & 99 \end{pmatrix} \begin{pmatrix} 1 & 4 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 17 & 10 \\ -29 & -17 \end{pmatrix},$$

188

we obtain

$$\begin{array}{c} (17, 10, -29, -17) \underbrace{f\left(-17, 29, -10, 17\right)}_{\rightarrow} \underbrace{g\left(-7, 5, -10, 7\right)}_{\rightarrow} \\ \underbrace{g\left(3, 1, -10, -3\right)}_{\rightarrow} \underbrace{f\left(-3, 10, -1, 3\right)}_{\rightarrow} \underbrace{g\left(-2, 5, -1, 2\right)}_{g\left(-1, 2, -1, 1\right)} \underbrace{g\left(0, 1, -1, 0\right)}_{\rightarrow}. \end{array}$$

We find

$$B = T^4 R T^2 R T^3 = \begin{pmatrix} -4 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} 1 & 3 \\ 0 & 1 \end{pmatrix} = \begin{pmatrix} 7 & 17 \\ 2 & 5 \end{pmatrix},$$

and so, $2A_7 = (7)^2 + (17)^2$. We have

$$4A_7 = 2(7)^2 + 2(17)^2.$$

Rearranging this equation, we get

$$4A_7 = 7.14 + 578 = 7(2)^2 + 70 + 578$$
$$= 7(2)^2 + 648 = 7(2)^2 + 8(9)^2,$$

that is, u = 9 and v = 2.

Example 7. Let p = 11, r = 2 and s = 1. We have $2A_{11} = 11482$ and $985.11482 - 3363^2 = 1$. The matrix

$$A = \begin{pmatrix} -3363 & 11482\\ -985 & 3363 \end{pmatrix}$$

is in Γ and has order 2. Applying the *Algorithm* \mathcal{M} to the matrix

$$T^{-4}AT^{4} = \begin{pmatrix} 1 & -4 \\ 0 & 1 \end{pmatrix} \begin{pmatrix} -3363 & 11482 \\ -985 & 3363 \end{pmatrix} \begin{pmatrix} 1 & 4 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 577 & 338 \\ -985 & -577 \end{pmatrix},$$

we obtain

$$\begin{array}{c} (577, 338, -985, -577) & \underline{f} (-577, 985, -338, 577) \\ \underline{g} (-239, 169, -338, 239) & \underline{g} (99, 29, -338, -99) \\ \underline{f} (-99, 338, -29, 99) & \underline{g} (-70, 169, -29, 70) \\ \underline{g} (-41, 58, -29, 41) & \underline{g} (-12, 5, -29, 12) & \underline{g} (17, 10, -29, -17) \\ \underline{f} (-17, 29, -10, 17) & \underline{g} (-7, 5, -10, 7) & \underline{g} (3, 1, -10, -3) \\ \underline{f} (-3, 10, -1, 3) & \underline{g} (-2, 5, -1, 2) & \underline{g} (-1, 2, -1, 1) & \underline{g} (0, 1, -1, 0). \end{array}$$

Then, we get $B = (T^4 R T^2 R)^2 T^3$. If we compute the matrix B, we obtain

$$B = \left(\begin{pmatrix} -4 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} -2 & 1 \\ -1 & 0 \end{pmatrix} \right)^2 \begin{pmatrix} 1 & 3 \\ 0 & 1 \end{pmatrix}$$
$$= \begin{pmatrix} 41 & 99 \\ 12 & 29 \end{pmatrix}.$$

We find $2A_{11} = (41)^2 + (99)^2$, and so, we have

$$4A_{11} = 2(99)^2 + 2(41)^2 = 2(11)^2(9)^2 + 3362$$

= 11(1782) + 3362 = 11(14)^2 + 8(51)^2.

Consequently, we obtain u = 51 and v = 14.

4. Conclusion and Future Works

In Theorem 2, we have proved that the Algorithm \mathcal{M} can be used to compute the integers x and y satisfying the equation $n = x^2 + Ny^2$ with (x, y) = 1 for $N \in \{4, 9, 16, 25\}$. This also means that n can be written in the form $n = x^2 + Ny^2$ with (x, y) = 1 under the hypothesis stated in Theorem 2. So, using the group structure of the Modular group Γ , we have also provided a solution to the problem when a natural number n (relatively prime to N) can be represented in the form $n = x^2 + Ny^2$. As a future work, sufficient conditions can be investigated for further values of N to prove similar results with the help of the modular group Γ . Of course, this method also provides the usage of the *Algorithm* \mathcal{M} in the computation of the integers x and y. It is well known that Cox answered the problem "which primes p can be expressed in the form $p = x^2 + Ny^2$ " completely by using the class field theory.²⁷

Finally, we note that some Hecke groups and discontinuous groups have been used for this purpose in Refs. 28 and 30 for some special cases of the number N. The interested reader can refer to Refs. 28 and 30 for more details. In Ref. 28, the special cases for N = 2, 3, 4, 5, 6, 7, 8, 9, 10, 12, 13, 16, 18, 22, 25, 28, 37, 58 are considered. As stated there, the importance of these results is that they yield the desired results easily and directly for general numbers n, not only for prime numbers. Furthermore, we do not need the restriction that N is a prime number.

Acknowledgments

The author is thankful to the anonymous referees for their careful reading and constructive comments.

References

- A. F. Beardon, *The Geometry of Discrete Groups*, vol. 91. Springer Science & Business Media, New York (2012).
- 2. D. E. Blair, *Inversion Theory and Conformal Mapping*, vol. 9. American Mathematical Soc., Providence, RI (2000).
- L. R. Ford, Automorphic Functions. Chelsea Publishing Co., New York (1951).
- 4. G. A. Jones and D. Singerman, *Complex Functions: An Algebraic and Geometric Viewpoint*. Cambridge University Press, Cambridge (1987).
- T. Needham, Visual Complex Analysis. Oxford University Press, New York (1997).
- M. Newman, *Integral Matrices*. Academic Press, New York-London (1972).
- V. Chutchavong, T. Dokyam, K. Janchitrapongvej, and C. Benjangkaprasert, On approximation of linear gain equalizer by q-Bernstein-Stancu operators and Möbius transformation. In 2018 15th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology (ECTI-CON), pp. 493–496, IEEE (2018).

- 8. D. P. Mandic, The use of Möbius transformations in neural networks and signal processing. In *Neural Networks for Signal Processing X. Proceedings of the 2000 IEEE Signal Processing Society Workshop* (*Cat. No. 00TH8501*), vol. 1, pp. 185–194, IEEE (2000).
- D. Mandic and J. Chambers, Recurrent Neural Networks for Prediction: Learning Algorithms, Architectures and Stability. Wiley, Hoboken, NJ, USA (2001).
- T. Noll, Musical intervals and special linear transformations, J. Math Music 1(2), 121–137 (2007).
- 11. P. Bussotti and R. Pisano, Historical and foundational details on the method of infinite descent: Every prime number of the form 4n + 1 is the sum of two squares, *Found. of Sci.* **25**(3), 671–702 (2020).
- 12. A. D. Christopher, A partition-theoretic proof of Fermat's two squares theorem, *Discrete Math.* **339**(4), 1410–1411 (2016).
- B. Fine, Number Theory: An Introduction via the Density of Primes, (2nd edn.). Springer, Cham (2016).
- G. H. Hardy and E. M. Wright, An Introduction to the Theory of Numbers. Oxford University Press (Clarendon), London (1960).
- D. Heath-Brown, Fermat's two squares theorem, *Invariant* 11, 3–5 (1984).
- A. Malter, D. Schleicher, and D. Zagier, New looks at old number theory, Amer. Math. Monthly 120(3), 243–264 (2013).
- N. Sam, A short proof of Fermat's two-square theorem, Amer. Math. Monthly 127(7), 638–638 (2020).
- P. Shiu, Involutions associated with sums of two squares, *Publ. Inst.* Math. (Beograd) (NS) 59(73), 18–30 (1996).
- 19. P. Shiu, Fermat's two squares theorem for a prime $p \equiv 1 \pmod{4}$, LMS Newsletter **494**, 26–31 (2021).
- 20. D. Zagier, A one-sentence proof that every prime $p \equiv 1 \pmod{4}$ is a sum of two squares, *Amer. Math. Monthly* **97**(2), 144 (1990).
- B. Fine, A note on the two-square theorem, Can. Math. Bull. 20(1), 93-94 (1977).
- N. Y. Özgür, On the two-square theorem and the modular group, Ars Combin. 95, 251–255 (2010).
- 23. J. M. Basilla, On the solution of $x^2 + dy^2 = m$, Proceedings of the Japan Academy, Series A, Mathematical Sciences **80**(5), 40–41 (2004).
- J. Brillhart, Note on representing a prime as a sum of two squares, Mathematics of Computation 26(120), 1011–1013 (1972).
- 25. B. Cho, Primes of the form $x^2 + ny^2$ with conditions $x \equiv 1 \mod N$, $y \equiv 0 \mod N$, Journal of Number Theory **130**(4), 852–861 (2010).
- 26. B. Cho, Integers of the form $x^2 + ny^2$, Monatshefte für Mathematik 174(2), 195–204 (2014).

- D. A. Cox, Primes of the Form x² + ny²: Fermat, Class Field Theory, and Complex Multiplication, 2nd ed. John Wiley & Sons, Inc., Hoboken, NJ. 2014, 2012 (2013).
- 28. G. Kern-Isberner and G. Rosenberger, A note on numbers of the form $n = x^2 + Ny^2$, Archiv der Mathematik. **43**(2), 148–156 (1984).
- 29. M. Kreuzer and G. Rosenberger. On the numbers of the form x²+11y². In *Elementary Theory of Groups and Group Rings, and Related Topics*, pp. 177–202. De Gruyter (2020).
- 30. N. Y. Özgür, On numbers of the form $n = x^2 + Ny^2$ and the Hecke groups $H(\sqrt{N})$, Journal of Number Theory **130**(6), 1274–1281 (2010).
- 31. J. A. González and F. Luca, On the positive integers n satisfying the equation $F_n = x^2 + ny^2$, Contemporary Math. 587, 95–109 (2013).
- 32. J. J. Alba González, P. Berrizbeitia, and F. Luca, On the formula $F_p = u^2 + pv^2$, International Journal of Number Theory **11**(01), 185–191 (2015).
- 33. C. Ballot and F. Luca, On the equation $x^2 + dy^2 = F_n$, Acta Arithmetica **127**(2), 145–155 (2007).
- D. Savin, Fibonacci primes of special forms, Notes on Number Theory and Discrete Math. 20(2), 10–19 (2014).
- I. Beck, Partial orders and the modular group, *Linear Algebra and Its Applications*. **120**, 139–147 (1989).
- 36. H. L. Chan, Windmills of the minds: An algorithm for Fermat's two squares theorem. In Proceedings of the 11th ACM SIGPLAN International Conference on Certified Programs and Proofs, pp. 251–264, Association for Computing Machinery, New York (2022).
- J. Buhler and S. Wagon, Basic algorithms in number theory, Algorithmic Number Theory. Lattices, Number Fields, Curves and Cryptography. pp. 25–68 (2008).
- P. Berrizbeitia, R. Chapman, F. Luca, and A. Mendoza, Quadratic forms representing *p*th terms of Lucas sequences, *Journal of Number Theory* 175, 134–139 (2017).
- 39. A. Dujella, Number Theory. Školska knjiga Zagreb, Zagreb (2021).
- D. Rosen, A class of continued fractions associated with certain properly discontinuous groups, *Duke Mathematical Journal* 21(3), 549–563 (1954).
- A. T. Benjamin and J. J. Quinn, The Fibonacci numbers-exposed more discretely, *Mathematics Magazine* 76(3), 182–192 (2003).
- D. Kalman and R. Mena, The Fibonacci numbers-exposed, Mathematics Magazine 76(3), 167–181 (2003).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0009

Chapter 9

The Components Exponential Function in Scator Hypercomplex Space: Planetary Elliptical Motion and Three-Body Choreographies

M. Fernandez-Guasti

Lab. de Óptica Cuántica, Depto. de Física, Universidad Autónoma Metropolitana — Iztapalapa, C.P. 09340 Ciudad de México, Ap. postal. 55-534, Mexico mfg@xanum.uam.mx

Scator algebra is a nondistributive extension of complex algebra and perplex numbers. The rectangular and polar representations are extended to higher dimensions in the additive (rectangular) and multiplicative (polar) representations of components. The complex exponential function is generalized to scator hypercomplex space by the components exponential function. This novel mathematical formalism permits a three-dimensional mapping that can be used to describe elementary curves, such as the conics, as well as other more complex curves. An asset of this approach is that trajectories are naturally parameterized in terms of angle variables. The usual parametrization of the two-body problem is readily obtained using this technique. Three-body choreographies in the trifolium rose, the trisectrix limaçon and a Gerono-type lemniscata are discussed. These two-dimensional projections are ruled out as three-body choreographies. However, these plane curves are encompassed in a scator curve embedded in a three-dimensional space where collisions do not take place. From the constant angular momentum condition, the time dependence of the parameter is readily obtained. Much insight is gained from the examination of these curves. Scator algebra provides a promising route toward the analytical study of two- and threedimensional choreographies with n bodies in given curves.

1. Introduction

Scator algebras are 1 + n dimensional algebras endowed with two operations and a main involution that serves as an order parameter. Scator elements are compound numbers that have one scalar component and an arbitrary number n of director components. Two branches of these algebras, analogous to complex and double (or perplex) numbers, have been acknowledged: (i) Elliptic scators that have been employed in polynomial solutions,¹ powers and roots² of scators, and (ii) Hyperbolic scators that have been applied to describe deformed Lorentz spaces^{3,4} as well as superluminal motion.⁵ A different product rule constitutes the departure between these two ramifications. Here, we shall be mainly concerned with elliptic scators.

In Section 2, the essentials of elliptic scator algebra in 1 + n dimensions are presented. The reader may skip this section on a first approach if in haste, since thereafter the dissertation is framed in one scalar and two director components. The scator representations in 1+2 dimensions and their geometrical interpretation are undertaken in Section 3. The elliptical trajectory and two-body motion in the scator formalism is described in Section 4. The components exponential (cexp) mapping of inclined lines with constant scalar component is introduced in Section 5. The next three sections consider the projections of the cexp mapping with m = 3, namely, the trifolium rose in Section 6, the trisectrix limaçon in Section 7 and the Gerono-type lemniscatae in Section 8.2. The dynamics of the lemniscata are discussed in Section 9. Conclusions are drawn in the last section.

2. Scator Algebra Preamble

Scator elements can be written in two ways, an additive representation that generalizes the rectangular complex representation and a multiplicative representation that generalizes the polar complex representation. In the *additive representation*, scator elements are described by a sum of components,

$$\overset{o}{\varphi} = f_0 + \sum_{j=1}^n f_j \check{\mathbf{e}}_j,\tag{1}$$

where the scator additive coefficients, usually written in lowercase Latin letters, are $f_0, f_j \in \mathbb{R}$, the director components are labeled with subindices 1 to $n \in \mathbb{N}$ and $\check{\mathbf{e}}_j \notin \mathbb{R}$. The *multiplicative* or polar *representation* of a scator is

$$\overset{o}{\varphi} = \varphi_0 \prod_{j=1}^n e^{\varphi_j \check{\mathbf{e}}_j} = \varphi_0 \prod_{j=1}^n \exp\left(\varphi_j \check{\mathbf{e}}_j\right).$$
(2)

where the scator multiplicative variables, labeled with Greek letters, are $\varphi_0 \in \mathbb{R}^+, \varphi_j \in \mathbb{R}$, and $\check{\mathbf{e}}_j \notin \mathbb{R}$; \mathbb{R}^+ represents the interval $[0, \infty)$. The zero subindex component is the scalar component in either representation. For all elements in the \mathbb{S}^{1+n} scator set, the additive scalar component must be different from zero if two or more director components are not zero,

$$\mathbb{S}^{1+n} = \left\{ \stackrel{o}{\varphi} = f_0 + \sum_{j=1}^n f_j \check{\mathbf{e}}_j : f_0 \neq 0 \text{ if } \exists f_j f_l \neq 0, \\ \text{for } j \neq l \text{ from 1 to } n \right\}.$$
(3)

The components exponential function⁶ in scator hypercomplex space is a scator function of scator variable $\stackrel{o}{\operatorname{cexp}}$: $\mathbb{R}^{1+n} \to \mathbb{S}^{1+n}$ defined for a scator argument $\stackrel{o}{\zeta} = z_0 + \sum_{j=1}^n z_j \check{\mathbf{e}}_j$, by

$$\overset{o}{\operatorname{cexp}} \begin{pmatrix} \overset{o}{\zeta} \end{pmatrix} \equiv \overset{o}{\operatorname{cexp}} \left(z_0 + \sum_{j=1}^n z_j \check{\mathbf{e}}_j \right) \equiv \operatorname{exp} \left(z_0 \right) \prod_{j=1}^n \operatorname{exp} \left(z_j \check{\mathbf{e}}_j \right), \quad (4a)$$

where $\exp(z_j \check{\mathbf{e}}_j) = \cos(z_j) + \sin(z_j) \check{\mathbf{e}}_j$. An Euler-type formula is thus satisfied for each hypercomplex component $\check{\mathbf{e}}_j$. In the additive

representation, the *components exponential function* is defined by

$$\stackrel{o}{\operatorname{cexp}}\left(z_{0} + \sum_{j=1}^{n} z_{j} \check{\mathbf{e}}_{j}\right) \equiv e^{z_{0}} \prod_{k=1}^{n} \cos\left(z_{k}\right)$$
$$+ e^{z_{0}} \sum_{j=1}^{n} \prod_{k\neq j}^{n} \cos\left(z_{k}\right) \sin\left(z_{j}\right) \check{\mathbf{e}}_{j}. \quad (4b)$$

The components exponential function given by (4b) may be viewed as a higher dimensional extension of Euler's formula. The multiplicative scator representation in terms of the cexp function is

$$\overset{o}{\varphi} = \varphi_0 \overset{o}{\operatorname{cexp}} \left(\sum_{j=1}^n \varphi_j \check{\mathbf{e}}_j \right).$$
 (5)

The mapping of the multiplicative to additive representations is given by the function $\mathfrak{f}_{mar}: (\mathbb{R}^+; \mathbb{R}^n) \to \mathbb{S}^{1+n}$,

$$\mathfrak{f}_{mar}:\varphi_0\prod_{j=1}^n e^{\varphi_j\check{\mathbf{e}}_j}\longmapsto\underbrace{\varphi_0\prod_{k=1}^n\cos(\varphi_k)}_{f_0}+\underbrace{\varphi_0\sum_{j=1}^n\prod_{k\neq j}^n\cos(\varphi_k)\sin(\varphi_j)\check{\mathbf{e}}_j}_{f_j}.$$
(6)

Scator algebra is endowed with two fundamental operations. The sum of scators $\overset{o}{\alpha} = a_0 + \sum_{j=1}^n a_j \check{\mathbf{e}}_j \in \mathbb{R}^{1+n}$ and $\overset{o}{\beta} = b_0 + \sum_{j=1}^n b_j \check{\mathbf{e}}_j \in \mathbb{R}^{1+n}$ is

$$\overset{o}{\alpha} + \overset{o}{\beta} \equiv (a_0 + b_0) + \sum_{j=1}^n (a_j + b_j) \check{\mathbf{e}}_j.$$
 (7)

Scator addition satisfies Abelian group properties. The product of scators $\overset{o}{\alpha} = \alpha_0 \prod_{j=1}^n \exp(\alpha_j \check{\mathbf{e}}_j)$ and $\overset{o}{\beta} = \beta_0 \prod_{j=1}^n \exp(\beta_j \check{\mathbf{e}}_j)$ is

$$\overset{o}{\alpha}\overset{o}{\beta} \equiv \alpha_0 \beta_0 \prod_{j=1}^n \exp\left[\left(\alpha_j + \beta_j\right) \check{\mathbf{e}}_j\right].$$
(8)

In the multiplicative representation, the scator product, provided that zero is excluded $\alpha_0, \beta_0 \neq 0$, satisfies Abelian group properties.

However, in the additive representation this is no longer true. The product $\overset{o}{\alpha}, \overset{o}{\beta} \in \mathbb{S}^{1+n}$ in the additive representation is: if $a_0 b_0 \neq 0$,

$$\overset{o}{\alpha} \overset{o}{\beta} = a_0 b_0 \prod_{k=1}^n \left(1 - \frac{a_k b_k}{a_0 b_0} \right)$$

$$+ a_0 b_0 \sum_{j=1}^n \left[\prod_{k \neq j}^n \left(1 - \frac{a_k b_k}{a_0 b_0} \right) \left(\frac{a_j}{a_0} + \frac{b_j}{b_0} \right) \right] \check{\mathbf{e}}_j;$$
(9a)

If $a_0 = 0$ and $b_0 \neq 0$, $\overset{o}{\alpha} = a_l \check{\mathbf{e}}_l$ has a single nonvanishing director component,

$$\overset{o}{\alpha}\overset{o}{\beta} = -a_l b_l + a_l b_0 \check{\mathbf{e}}_l - \sum_{j \neq l}^n \left(a_l \frac{b_l b_j}{b_0} \right) \check{\mathbf{e}}_j; \tag{9b}$$

if $a_0 = b_0 = 0$, $\overset{o}{\alpha} = a_l \check{\mathbf{e}}_l$ and $\overset{o}{\beta} = b_j \check{\mathbf{e}}_j$, $\overset{o}{\alpha}\overset{o}{\beta} = -a_l b_j \delta_{lj}$, where δ_{lj} is a Kroneker delta. The scator product in the additive representation does not satisfy group properties because it has zero divisors, i.e. $\overset{o}{\alpha}\overset{o}{\beta} = 0$ for nonzero factors, and is not associative if the scalar component of a product is zero. However, the scator product is always commutative. The main second-order involution is *Conjugation*, defined by the negative of the director components in either representation. The conjugate scator of $\overset{o}{\varphi}$, given by (2), is $\overset{o}{\varphi}^* = \varphi_0 \prod_{j=1}^n e^{-\varphi_j \check{\mathbf{e}}_j}$, whereas the conjugate of (1) is $\overset{o}{\varphi}^* \equiv f_0 - \sum_{j=1}^n f_j \check{\mathbf{e}}_j$. The magnitude is given by the positive square root of the scator

The magnitude is given by the positive square root of the scator times its conjugate $\| \overset{o}{\varphi} \| = \sqrt{\overset{o o^*}{\varphi \varphi}} = \varphi_0$. In the additive representation, for $f_0 \neq 0$,

$$\| \overset{o}{\varphi} \| = \sqrt{\overset{o}{\varphi} \overset{o}{\varphi}^{*}} = |f_0| \prod_{k=1}^n \sqrt{1 + \frac{f_k^2}{f_0^2}}, \tag{10}$$

and if the additive scalar component is zero, there is only one director component, say the *l*th component, $\| \overset{o}{\varphi} \| = |f_l|$. The multiplicative inverse of a scator $\overset{o}{\varphi}$ is $\overset{o}{\varphi}^{-1} = \overset{o}{\varphi}^* / \| \overset{o}{\varphi} \|^2$.

In the rest of this chapter, we shall restrict to \mathbb{S}^{1+2} . A description having two hypercomplex director components will already reveal many of the interesting features and possibilities of scator algebra.
3. Scators in \mathbb{S}^{1+2}

In $\mathbb{S}^{1+2},$ the scator $\stackrel{o}{\varphi}$ in the additive and multiplicative representations is

$$\overset{o}{\varphi} = s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y = \varphi_0 \exp\left(\varphi_x \check{\mathbf{e}}_x\right) \exp\left(\varphi_y \check{\mathbf{e}}_y\right).$$
 (11a)

The additive representation with multiplicative variables is

$$\ddot{\varphi} = \varphi_0 \cos \varphi_x \cos \varphi_y + \varphi_0 \sin \varphi_x \cos \varphi_y \,\check{\mathbf{e}}_x + \varphi_0 \cos \varphi_x \sin \varphi_y \,\check{\mathbf{e}}_y.$$
 (11b)

The relationship between additive and multiplicative variables is

$$s = \varphi_0 \cos \varphi_x \cos \varphi_y, \quad x = \varphi_0 \sin \varphi_x \cos \varphi_y, \quad y = \varphi_0 \cos \varphi_x \sin \varphi_y.$$
(12)

This scator can be represented geometrically in our familiar threedimensional space with three orthogonal axes, letting one axis to represent the scalar axis, labeled with the letter "s" and the remaining two axes representing the $\check{\mathbf{e}}_x$ and $\check{\mathbf{e}}_y$ directions, respectively. An example of such a scator is depicted in Figure 1.

This is, so to speak, an Argand diagram in \mathbb{S}^{1+2} -dimensional space. The angle in the $(s, \check{\mathbf{e}}_x)$ projection is $\tan \varphi_x = \frac{x}{s}$ and in the $(s, \check{\mathbf{e}}_y)$ plane $\tan \varphi_y = \frac{y}{s}$.

The angle in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane is

$$\frac{y}{x} = \tan \theta = \frac{\varphi_0 \cos \varphi_x \sin \varphi_y}{\varphi_0 \sin \varphi_x \cos \varphi_y} = \frac{\tan \varphi_y}{\tan \varphi_x}.$$
 (13)

The scator magnitude squared of $\overset{o}{\varphi} = s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y$ is

$$\| \overset{o}{\varphi} \|^2 = s^2 \left(1 + \frac{x^2}{s^2} \right) \left(1 + \frac{y^2}{s^2} \right).$$
 (14)

The scator $\overset{o}{\varphi}(\varphi_0; \varphi_x, \varphi_y)$ refers to $\overset{o}{\varphi}$ written in terms of its multiplicative (or polar) variables. The magnitude φ_0 in the argument is separated by a semicolon from the director angles φ_x, φ_y . The scator $\overset{o}{\varphi}(s; x, y)$ refers to $\overset{o}{\varphi}$ written in terms of its additive (or rectangular) variables. The additive scalar component is again separated by a semicolon from the director components. However, it does not represent the scator magnitude that in terms of (s; x, y) is given by (14).



Fig. 1. Geometric representation of a unit magnitude scator $\overset{\circ}{\varphi}$ (solid green). The tip of the scator lies on the surface of the cusphere (shown in semitransparent pearl). The tips of the projections of $\overset{\circ}{\varphi}$ (dotted green) in the $(s, \check{\mathbf{e}}_x)$ and $(s, \check{\mathbf{e}}_y)$ planes also lie in the cusphere surface. The scator in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane (dotted orange) does not belong to the \mathbb{S}^{1+2} set, its magnitude is not defined.

For a unit magnitude scator $\| \varphi^{o} \|^{2} = 1$, the equation of the isometric two-dimensional surface embedded in three-dimensional space is

$$s^{2} = (s^{2} + x^{2}) (s^{2} + y^{2}).$$
(15)

This cusphere surface is rendered in Figure 1. The scalar and the two director components are represented in three mutually orthogonal axes. The projections of the cusphere surface onto the $s, \check{\mathbf{e}}_x$ and $s, \check{\mathbf{e}}_y$ planes are filled circles while its projection in the $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ plane is a filled square (see Ref. 6, Section 6.3); thus, the portmanteau of cube and sphere. The components exponential function produces this mapping for a constant scalar component, $\overset{o}{\operatorname{cexp}}(\overset{o}{\zeta}) = \overset{o}{\operatorname{cexp}}(c_0 + z_x \check{\mathbf{e}}_x + z_y \check{\mathbf{e}}_y) \rightarrow \text{cusphere.}$

3.1. Scator projections

The unit magnitude scator $\overset{o}{\varphi} = \cos \varphi_x \cos \varphi_y + \sin \varphi_x \cos \varphi_y \check{\mathbf{e}}_x + \cos \varphi_x \sin \varphi_y \check{\mathbf{e}}_y \in \mathbb{S}^{1+2}$ can be written in terms of unit magnitude projections in the $(s, \check{\mathbf{e}}_x)$ and $(s, \check{\mathbf{e}}_y)$ planes,

$$\begin{split} \stackrel{o}{\varphi} (1; \varphi_x, \varphi_y) &= \exp\left(\varphi_x \check{\mathbf{e}}_x\right) \exp\left(\varphi_y \check{\mathbf{e}}_y\right) \\ &= \left(\cos\varphi_x + \sin\varphi_x \check{\mathbf{e}}_x\right) \left(\cos\varphi_y + \sin\varphi_y \check{\mathbf{e}}_y\right). \end{split}$$

Since

$$\stackrel{o}{\varphi}(1;\varphi_x,0) = \exp\left(\varphi_x\check{\mathbf{e}}_x\right) = (\cos\varphi_x + \sin\varphi_x\check{\mathbf{e}}_x).$$
$$\stackrel{o}{\varphi}(1;0,\varphi_y) = \exp\left(\varphi_y\check{\mathbf{e}}_y\right) = (\cos\varphi_y + \sin\varphi_y\check{\mathbf{e}}_y).$$

Thus, $\overset{o}{\varphi}(1; \varphi_x, \varphi_y) = \overset{o}{\varphi}(1; \varphi_x, 0) \overset{o}{\varphi}(1; 0, \varphi_y).$

Important 1. Elements as product of its projections. Scator elements, in contrast with other objects like vectors, can be decomposed as the product of its projections (see Ref. 7, Section 4.)

In the multiplicative representation, to drop out a component (say $\exp(\varphi_y \check{\mathbf{e}}_y)$) is equal to making the corresponding multiplicative coefficient zero i.e. $\varphi_y = 0$.

In general, the scator product does not distribute over addition, as seen from (9a). However, if the director components of the factors are different, the product distributes over addition. In particular, for $\overset{o}{\alpha} = a_0 + a_x \check{\mathbf{e}}_x$ and $\overset{o}{\beta} = b_0 + b_y \check{\mathbf{e}}_y$

$$\overset{o}{\alpha}\overset{o}{\beta} = (a_0 + a_x \check{\mathbf{e}}_x) (b_0 + b_y \check{\mathbf{e}}_y) = a_0 b_0 + b_0 a_x \check{\mathbf{e}}_x + a_0 b_y \check{\mathbf{e}}_y,$$

where $\check{\mathbf{e}}_x \check{\mathbf{e}}_y = 0$. Let $\overset{o}{\varphi}(1; \varphi_x, 0) = \overset{o}{\varphi}(s_1; x_1, 0)$ and $\overset{o}{\varphi}(1; 0, \varphi_y) = \overset{o}{\varphi}(s_2; 0, y_2)$. From the identifications in (12) and $s_1 = \cos \varphi_x$, $x_1 = \sin \varphi_x$, $s_2 = \cos \varphi_y$, $y_2 = \sin \varphi_y$, the unit magnitude factors in terms of additive coefficients are

$$\begin{aligned} \stackrel{o}{\varphi}(s_1; x_1, 0) &= s_1 + x_1 \,\check{\mathbf{e}}_x = \left(\frac{s}{\sqrt{s^2 + x^2}} + \frac{x}{\sqrt{s^2 + x^2}} \,\check{\mathbf{e}}_x\right) \\ &= (\cos\varphi_x + \sin\varphi_x \,\check{\mathbf{e}}_x), \end{aligned}$$
(16a)

$$\overset{o}{\varphi}(s_2; 0, y_2) = s_2 + y_2 \,\check{\mathbf{e}}_y = \left(\frac{s}{\sqrt{s^2 + y^2}} + \frac{y}{\sqrt{s^2 + y^2}}\,\check{\mathbf{e}}_y\right)$$
$$= (\cos\varphi_y + \sin\varphi_y \,\check{\mathbf{e}}_y).$$
(16b)

The scator $\overset{o}{\varphi}(s; x, y) = s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y$ in terms of its unit magnitude factors in additive variables is then

$$s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y$$
$$= \left(\frac{s}{\sqrt{s^2 + x^2}} + \frac{x}{\sqrt{s^2 + x^2}} \,\check{\mathbf{e}}_x\right) \left(\frac{s}{\sqrt{s^2 + y^2}} + \frac{y}{\sqrt{s^2 + y^2}} \,\check{\mathbf{e}}_y\right).$$

Evaluation of the product

$$s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y = \frac{s^2}{\sqrt{s^2 + x^2}\sqrt{s^2 + y^2}} + \frac{sx}{\sqrt{s^2 + x^2}\sqrt{s^2 + y^2}} \,\check{\mathbf{e}}_x \\ + \frac{sy}{\sqrt{s^2 + x^2}\sqrt{s^2 + y^2}} \,\check{\mathbf{e}}_y,$$

recalling from (15), that for unit magnitude scators in \mathbb{S}^{1+2} , $\sqrt{s^2 + x^2}\sqrt{s^2 + y^2} = s$ confirms the identity. It is also possible to evaluate the projection of $\overset{o}{\varphi}(s; x, y) = s + x \,\check{\mathbf{e}}_x + y \,\check{\mathbf{e}}_y$ by dropping out one of the director components in the additive representation

$$\overset{o}{\varphi}(s;x) = s + x \,\check{\mathbf{e}}_x = \cos\varphi_x \cos\varphi_y + \sin\varphi_x \cos\varphi_y \,\check{\mathbf{e}}_x,$$
 (17)

$$\overset{o}{\varphi}(s;y) = s + y\,\check{\mathbf{e}}_y = \cos\varphi_x\cos\varphi_y + \cos\varphi_x\sin\varphi_y\,\check{\mathbf{e}}_y,\tag{18}$$

where the rightmost equalities follow from (12). Therefore, dropping out a component or making it zero in the additive representation gives different results, as can be seen from a comparison of (16a)– (17) and (16b)–(18). If the component is dropped out, say y in (17), it is allowed to acquire all its permitted values, $y = \varphi_0 \cos \varphi_x \sin \varphi_y$. Whereas in the unit magnitude projection, it is restricted to zero, y = 0. Hereafter, we refer simply to "projections" when the component is dropped out and explicitly to "unit magnitude projections."

In either projection, the scator projections point in the same direction, since the quotients of components $\frac{x_1}{s_1} = \frac{x}{s} = \tan \varphi_x$ and $\frac{y_1}{s_1} = \frac{y}{s} = \tan \varphi_y$ are the same. However, their magnitudes differ, and this will be an important difference when curves are parameterized. In the $\overset{o}{\varphi}(s;x)$ (dropout) projection, the magnitude is $\|\overset{o}{\varphi}(s;x)\| = \sqrt{s^2 + x^2}$ and for $\overset{o}{\varphi}(s;y)$, the magnitude is $\|\overset{o}{\varphi}(s,y)\| = \sqrt{s^2 + y^2}$. However, if the scalar component is dropped out, the scator magnitude (14) diverges. In the limit when $s \to 0$ with $xy \neq 0$, the scator element $s + x \check{\mathbf{e}}_x + y \check{\mathbf{e}}_y$ is not in the \mathbb{S}^{1+2} set. The scator magnitude is then $\lim_{s\to 0} ||\overset{o}{\varphi}|| = \infty$. Thus, in the limit $s \to 0$, the Euclidean metric is not recovered. It is nonetheless possible to obtain the Euclidean limit for $s^2 \gg x^2, y^2$, following an analogous procedure to the classical limit of a hyperbolic scator in a deformed Lorentz metric.⁴ The scalar magnitude of a scator is $\|\overset{o}{\varphi}\|_0 \equiv \|\frac{1}{2}(\overset{o}{\varphi} + \overset{o}{\varphi}^*)\| = |s|$. The director magnitude is defined in an analogous fashion as it was done for hyperbolic scators

$$\| \overset{o}{\varphi} \|_{\check{\mathbf{e}}} \equiv \sqrt{\| \overset{o}{\varphi} \|^2 - \| \overset{o}{\varphi} \|_0^2} = \left(s^2 \left(1 + \frac{x^2}{s^2} \right) \left(1 + \frac{y^2}{s^2} \right) - s^2 \right)^{\frac{1}{2}}.$$
 (19)

The director magnitude limit when the additive scalar component s becomes much larger than the director components is

$$\lim_{\substack{xy \\ s} \to 0} \left\| \stackrel{o}{\varphi} \right\|_{\check{\mathbf{e}}} = \lim_{\substack{xy \\ s} \to 0} \left(x^2 + y^2 + \frac{x^2 y^2}{s^2} \right)^{\frac{1}{2}} = \sqrt{x^2 + y^2}.$$

In this limit, the two-dimensional Euclidean metric is recovered. This $\sqrt{x^2 + y^2}$ circle is depicted on the cusphere surface in Figure 1, for s close to one and small x, y.

There are three variables in the \mathbb{S}^{1+2} scator set but only two in \mathbb{C} . So, there are many possibilities in order to embed a curve or trajectory (one-dimensional) in 1 + 2-(three)-dimensional space.

4. Elliptical Trajectory-Curve-Motion

The components exponential function $\stackrel{o}{\operatorname{cexp}}$ in \mathbb{S}^{1+2} with argument $\stackrel{o}{\zeta} = z_0 + z_x \check{\mathbf{e}}_x + z_y \check{\mathbf{e}}_y$ is

 $\overset{o}{\operatorname{cexp}} \begin{pmatrix} \overset{o}{\zeta} \end{pmatrix} = e^{z_0} \cos z_x \cos z_y + e^{z_0} \sin z_x \cos z_y \,\check{\mathbf{e}}_1 + e^{z_0} \cos z_x \sin z_y \,\check{\mathbf{e}}_2.$ Consider the scalar and the $\check{\mathbf{e}}_x$ director components to be constant, $z_0 = c_0$ and $z_x = c_1$. The scator $\overset{o}{\zeta} = c_0 + c_1 \,\check{\mathbf{e}}_x + z_y \,\check{\mathbf{e}}_y$ is then a

line parallel to the $\check{\mathbf{e}}_y$ axis. The components exponential function maps line segments parallel to one of the director axes into ellipses,⁶ as portrayed in Figure 2. Comparison with the additive representation using multiplicative variables (11b) gives the following identifications, the scator magnitude is $\varphi_0 = e^{z_0}$ and $z_x = c_1 = \varphi_x$.

Allow for the constants a, b to be $a = \varphi_0 \sin \varphi_x$ and $b = \varphi_0 \cos \varphi_x$. The scator $\overset{o}{\varphi}_{\text{elli}}$, from (11b), is then

$$\overset{o}{\varphi}_{\text{elli}} = \underbrace{b\cos\varphi_y}_{s} + \underbrace{a\cos\varphi_y}_{x} \check{\mathbf{e}}_x + \underbrace{b\sin\varphi_y}_{y} \check{\mathbf{e}}_y.$$
(20)

The plane with $\varphi_x = \arctan \frac{a}{b} = \text{constant}$ is shown in semitransparent yellow in Figure 3. A semiplane with a given φ_y is drawn in semitransparent green in the same figure. The $\overset{o}{\varphi}_{elli}$ scator lies on the intersection of these two planes. As the angle φ_y is swept from 0 to 2π , the tip of the $\overset{o}{\varphi}_{elli}$ scator describes an ellipse in the $\varphi_x = \text{constant}$ plane (shown in maroon in Figure 3). To confirm this assertion, perform a clockwise Euclidean rotation by $\chi = \frac{\pi}{2} - \varphi_x$ of the coordinates in the $(s, \check{\mathbf{e}}_x)$ plane. $\overset{o'}{\varphi}_{elli} = \sqrt{a^2 + b^2} \cos \varphi_y \check{\mathbf{e}}_x + b \sin \varphi_y \check{\mathbf{e}}_y$ is indeed an ellipse with semiaxes $\sqrt{a^2 + b^2}$ and b in the $\varphi_x = \text{constant}$ plane. Euclidean rotations do not preserve the scator magnitude so they should be treated with great care. Furthermore, in this case, $\overset{o'}{\varphi}_{elli} \notin \mathbb{S}^{1+2}$ is not in the scator set, the sole purpose of the transformation was to show that an ellipse is obtained in the $\varphi_x = \text{constant}$ plane.

The projections of the $\overset{o}{\varphi}_{\text{elli}}$ scator are as follows:

- (i) a circle radius *b* is obtained in the $(s, \check{\mathbf{e}}_y)$ plane, with parametric representation $\overset{o}{\varphi}_{\text{elli}}\Big|_{(s,\check{\mathbf{e}}_y)} = \overset{o}{\varphi}_{\text{circle}}(s,\check{\mathbf{e}}_y) = b\cos\varphi_y + b\sin\varphi_y\,\check{\mathbf{e}}_y,$
- (ii) a line segment with length $2\sqrt{a^2 + b^2}$ and slope $\frac{a}{b}$ is obtained in the $(s, \check{\mathbf{e}}_x)$ plane, with parametric representation $\overset{o}{\varphi}_{\text{elli}}\Big|_{(s,\check{\mathbf{e}}_x)} = \overset{o}{\varphi}_{\text{line}}(s,\check{\mathbf{e}}_x) = b\cos\varphi_y + a\cos\varphi_y\check{\mathbf{e}}_x,$
- (iii) an ellipse is obtained in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane with semiaxes a and b, respectively, with parametric representation $\overset{o}{\varphi}_{\text{elli}}\Big|_{(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)} = a \cos \varphi_y \check{\mathbf{e}}_x + b \sin \varphi_y \check{\mathbf{e}}_y$.



(Scator ellipses lie on the cusphere surface)

Fig. 2. Cusphere unit magnitude surfaces shown in semitransparency. Ellipses with semiaxes $a = \varphi_0 \sin \varphi_x$ and $b = \varphi_0 \cos \varphi_x$, for $\varphi_x = 0.96 \frac{\pi}{2}$ (blue), $\varphi_x = 0.84 \frac{\pi}{2}$ (green), $\varphi_x = 0.7 \frac{\pi}{2}$ (red), $\varphi_x = 0.5 \frac{\pi}{2}$ (orange) and $\varphi_x = 0.3 \frac{\pi}{2}$ (yellow).



Fig. 3. A scator $\overset{\circ}{\varphi}$ depicted as an orange arrow in \mathbb{S}^{1+2} -dimensional space. The ellipse in maroon lies in the φ_x constant plane (yellow). The φ_y plane (green) intersects the φ_x constant plane at a line (orange). The magnitude is chosen $\varphi_0 = \sqrt{a^2 + b^2}$, where a, b are the semiaxes of the magenta ellipse in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane. The angle φ_y is swept from 0 to 2π . The curve thus obtained is an ellipse in the φ_x constant plane (yellow). In scator space the magnitude is constant although the shape of the curve is an ellipse. It lies in the isometric cusphere surface. The projection of this ellipse in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane is also an ellipse (magenta). The φ_y angle corresponds to the eccentric anomaly angle E defined in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane (see Figure).

These projections are depicted in Figure 4. It is interesting to point out that fixing the scator magnitude and the angle φ_x does not make any of the additive scator coefficients (s; x, y) constant. In contrast, in complex algebra, a constant magnitude and phase define a point in the complex plane. The extra degree of freedom in \mathbb{S}^{1+2} now defines a curve, that is, a one-dimensional object embedded in a three-dimensional space. The scator magnitude squared is

$$\begin{aligned} \left\| \stackrel{o}{\varphi} \right\|^2 &= b^2 \cos^2 \varphi_y + a^2 \cos^2 \varphi_y + b^2 \sin^2 \varphi_y \\ &+ \frac{a^2 \cos^2 \varphi_y b^2 \sin^2 \varphi_y}{b^2 \cos^2 \varphi_y} = a^2 + b^2. \end{aligned}$$



(Projections in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane are ellipses)



(Projections in the $(s, \check{\mathbf{e}}_y)$ plane are circles)



(Projections in the $(s, \check{\mathbf{e}}_x)$ plane are line segments)

Fig. 4. Orthographic projections of scator ellipses.

Thus, $\varphi_0 = \sqrt{a^2 + b^2}$ and the constant angle is $c_1 = \varphi_x = \arctan \frac{a}{b}$. The representation of an ellipse in \mathbb{S}^{1+2} scator space in the multiplicative representation is

$$\overset{o}{\varphi}_{\text{elli}} = \sqrt{a^2 + b^2} \exp\left(\arctan\frac{a}{b}\check{\mathbf{e}}_x\right) e^{\varphi_y \check{\mathbf{e}}_y}.$$
(21)

The multiplicative representations of the projections in the $(s, \check{\mathbf{e}}_y)$ and $(s, \check{\mathbf{e}}_x)$ planes are $\overset{o}{\varphi}_{\text{circle}} = be^{\varphi_y \check{\mathbf{e}}_y}$ and $\overset{o}{\varphi}_{\text{line}} = \sqrt{a^2 + b^2} \cos \varphi_y$ $\exp\left(\pm \arctan \frac{a}{b} \check{\mathbf{e}}_x\right)$. However, there is no multiplicative representation of the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ projection because $\overset{o}{\varphi}_{\text{elli}}|_{(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)} \notin \mathbb{S}^{1+2}$. It is therefore preferable to work in scator algebra retaining always the scalar component keeping in mind that the director part represents the trajectory in a two-dimensional plane.

4.1. Eccentric anomaly

Consider a shift in the $\check{\mathbf{e}}_x$ direction to a focus as is usually done in celestial mechanics, $f_x \check{\mathbf{e}}_x \mapsto (f_x - ae) \check{\mathbf{e}}_x$, where the eccentricity is $e = \sqrt{1 - \frac{b^2}{a^2}}$. The $\overset{o}{\varphi}_{\text{elli}}$ scator is shifted to $\overset{o}{\varphi}_{\text{elli-ecc}} = \overset{o}{\varphi}_{\text{elli}} - ae \check{\mathbf{e}}_x = b \cos \varphi_y + a (\cos \varphi_y - e) \check{\mathbf{e}}_x + b \sin \varphi_y \check{\mathbf{e}}_y.$ (22)

Recall that the parametric vector representation of position of a twobody problem with one-body sitting at one focus is⁸

$$\mathbf{r} = a \left(\cos E - e\right) \hat{\mathbf{e}}_x + b \sin E \, \hat{\mathbf{e}}_y,\tag{23}$$

where E is the eccentricity angle. Comparison of (22) and (23) evinces that φ_y is equal to the eccentricity angle E. This is quite an unexpected result since φ_y is the projection angle of a scator in the $(s, \check{\mathbf{e}}_y)$ plane. The assertion $\varphi_y = E$ implies that φ_y is equal to the eccentricity angle in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane, as shown in Figure 5.

4.2. True anomaly

The φ_y angle can be written in terms of the true anomaly ϕ following the known relationships between eccentricity and true angles. However, it is readily obtained in the scator formalism if the origin is shifted to a focus in the 1 + 2-dimensional space. Note that the triangle in the $s, \check{\mathbf{e}}_x$ plane in Figure 3 tan $\varphi_x = \frac{a}{b} = \frac{ae}{be}$. Thus, the scalar component has to be shifted by $s \mapsto (s - be)$ and the $\check{\mathbf{e}}_x$ component by $x \check{\mathbf{e}}_x \mapsto (x - ae) \check{\mathbf{e}}_x$

$$\overset{o'}{\varphi} = b\left(\cos\varphi_y - e\right) + a\left(\cos\varphi_y - e\right)\check{\mathbf{e}}_x + b\sin\varphi_y\check{\mathbf{e}}_y.$$
 (24)

From (13), the angle in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane is

$$\tan \phi = \frac{\tan \varphi'_y}{\tan \varphi'_x} = \frac{b}{a} \frac{\sin \varphi_y}{(\cos \varphi_y - e)} = \sqrt{1 - e^2} \frac{\sin \varphi_y}{(\cos \varphi_y - e)}, \quad (25)$$



Fig. 5. Ellipse (magenta) in the $(\check{\mathbf{e}}_x,\check{\mathbf{e}}_y)$ plane. The ellipse is circumscribed by the green circle and the orange circle is inscribed in the ellipse. A point in the ellipse is given by (r,θ) , where r is clearly not constant but given by the magnitude of (23). Taking a vertical line from the point until it intersects the circumscribed circle or a horizontal line until it intersects the inscribed circle that defines the angle φ_y . The ellipse can then be parametrically represented by $a \cos \varphi_y \check{\mathbf{e}}_x + b \sin \varphi_y \check{\mathbf{e}}_y$, where a, b are the ellipse semiaxes in $\check{\mathbf{e}}_x$ and $\check{\mathbf{e}}_y$, respectively.

where $\tan \varphi'_y = \frac{b \sin \varphi_y}{b(\cos \varphi_y - e)}$ and $\tan \varphi'_x = \frac{a(\cos \varphi_y - e)}{b(\cos \varphi_y - e)}$. Equation (25) reproduces the usual relationship between eccentric and true anomalies.

4.3. Dynamics

To obtain the dynamics, consider the derivative of (23) with respect to time,

$$\mathbf{v} = -a\sin E\,\partial_t E\,\hat{\mathbf{e}}_x + b\cos E\,\partial_t E\,\hat{\mathbf{e}}_y.$$

The angular momentum is then $\mathbf{L} = m\mathbf{r} \times \mathbf{v} = m ab (1 - e \cos E)$ $\partial_t E \hat{\mathbf{e}}_z$. Since the system is isolated, the angular momentum should be constant $L = m ab (1 - e \cos E) \partial_t E$, and upon integration Kepler's equation is obtained

$$E - e\sin E = \frac{L}{m \, ab}t.$$

The usual approximation methods are required to solve this transcendental equation for the eccentricity as a function of time E(t).⁹

4.4. Hyperbolae

Since the cusphere surface is bounded, only the circle and the ellipse can be obtained within this surface. Hyperbolic or real scators in \mathbb{S}^{1+2} are of the form¹⁰

$$\begin{aligned} \ddot{\varphi} &= s + x \, \hat{\mathbf{e}}_x + y \, \hat{\mathbf{e}}_y = \varphi_0 \exp\left(\varphi_x \hat{\mathbf{e}}_x\right) \exp\left(\varphi_y \hat{\mathbf{e}}_y\right) \\ &= \varphi_0 \cosh\varphi_x \cosh\varphi_y + \varphi_0 \sinh\varphi_x \cosh\varphi_y \hat{\mathbf{e}}_x \\ &+ \varphi_0 \cosh\varphi_x \sinh\varphi_y \hat{\mathbf{e}}_y. \end{aligned}$$
(26)

Hyperbolic trigonometric functions are now involved in the additive representation with multiplicative variables. The product of real unit directors is $\hat{\mathbf{e}}_j \hat{\mathbf{e}}_j = \delta_{ij}$ (in contrast to $\check{\mathbf{e}}_x \check{\mathbf{e}}_y = -\delta_{ij}$), hats instead of checks are used to distinguish them. The hyperbolic scator magnitude squared of $\hat{\varphi} = s + x \, \hat{\mathbf{e}}_x + y \, \hat{\mathbf{e}}_y$ is

$$\| \overset{o}{\varphi} \|^2 = s^2 \left(1 - \frac{x^2}{s^2} \right) \left(1 - \frac{y^2}{s^2} \right).$$
 (27)

Consider the constants a, b to be $a = \varphi_0 \sinh \varphi_x$ and $b = \varphi_0 \cosh \varphi_x$, so that the angle φ_x is constant and equal to $\operatorname{arctanh}(\frac{a}{b})$. The constant a is positive or negative depending on the sign of φ_x , while $b \ge 1$ and b > a. The scator $\overset{o}{\varphi}_{hyp}$, from (26), is then

$$\overset{o}{\varphi}_{\text{hyp}} = \underbrace{b \cosh \varphi_y}_{s} + \underbrace{a \cosh \varphi_y}_{x} \hat{\mathbf{e}}_x + \underbrace{b \sinh \varphi_y}_{y} \hat{\mathbf{e}}_y.$$
(28)

A parametric equation of the x > 0 branch of a rectangular hyperbola is obtained in the $(s, \hat{\mathbf{e}}_y)$ plane and the positive x branch of a hyperbola with canonical form $\frac{x^2}{a^2} - \frac{y^2}{b^2} = \cosh^2 \varphi - \sinh^2 \varphi = 1$



Fig. 6. Families of hyperbolae in the (s; x, y) scator space.

in the $(\hat{\mathbf{e}}_x, \hat{\mathbf{e}}_y)$ plane. The multiplicative or polar representation of hyperbolae in scator algebra is

$$\overset{o}{\varphi}_{\text{hyp}} = \sqrt{b^2 - a^2} \exp\left(\operatorname{arctanh}\left(\frac{a}{b}\right) \hat{\mathbf{e}}_x\right) \exp\left(\varphi_y \hat{\mathbf{e}}_y\right).$$

Hyperbolae with major axis in the $\hat{\mathbf{e}}_y$ direction are obtained if the φ_y director component is considered constant. Hyperbolae in \mathbb{S}^{1+2} are illustrated in Figure 6.

5. Three Bodies in the C_{13} Curve

Consider straight lines in the $z_0 = c_0$ plane with slope m in the $\hat{\mathbf{e}}_x, \hat{\mathbf{e}}_y$ plane. z_y is then equal to $z_y = m z_x + b$, where m, b are real constants. The scator $\zeta = c_0 + z_x \, \check{\mathbf{e}}_x + (m \, z_x + b) \, \check{\mathbf{e}}_y$ then describes line segments inclined in the $\hat{\mathbf{e}}_x, \hat{\mathbf{e}}_y$ plane. The components exponential mapping of $\stackrel{o}{\zeta}$ in \mathbb{S}^{1+2} is then

$$\overset{o}{\operatorname{cexp}} \begin{pmatrix} \overset{o}{\zeta} \end{pmatrix} = e^{c_0} \left(\cos \varphi \cos \left(m\varphi \right) + \sin \varphi \cos \left(m\varphi + b \right) \check{\mathbf{e}}_x \right. \\ \left. + \cos \varphi \sin \left(m\varphi + b \right) \check{\mathbf{e}}_y \right),$$

where the continuous free parameter is $\varphi = z_x$. If *m* is rational, the curves are closed. For integer *m*, odd *m* coefficients exhibit a π periodicity.¹ For $c_0 = 0$ and m = 3,

$$\overset{o}{\varphi} = \cos\varphi_x \cos\left(3\varphi_x\right) + \cos\left(3\varphi_x\right) \sin\varphi_x \check{\mathbf{e}}_x + \cos\varphi_x \sin\left(3\varphi_x\right) \check{\mathbf{e}}_y,$$
(29)

This parametric representation of the curve C_{13} is depicted in Figure 7. C_{13} is a one-dimensional object embedded in \mathbb{S}^{1+2} -(three)-dimensional space, the subindices stand for the φ_x and φ_y coefficients. The components exponential function is one of the few scator holomorphic functions, according to the Cauchy differential quotient criterion.^{11,12} It thus insures that a continuous argument will produce a continuous mapping. The additive and multiplicative variables are related by

$$\tan\varphi_x = \frac{x}{s}, \quad \tan 3\varphi_x = \frac{y}{s}.$$
 (30)

From the quotient of these two equations, expanding the triple angle,

$$\frac{\tan(3\varphi_x)}{\tan\varphi_x} = \frac{y}{x} = \frac{3 - \tan^2\varphi_x}{1 - 3\tan^2\varphi_x} \frac{3 - \frac{x^2}{s^2}}{1 - 3\frac{x^2}{s^2}}.$$
 (31)

Substitution of (30) gives y in terms of the other two variables

$$y = x \frac{3s^2 - x^2}{s^2 - 3x^2}.$$
(32a)

$$\stackrel{o}{\operatorname{cexp}} (c_0 + (z_1 + \pi) \,\check{\mathbf{e}}_1 + m \, (z_1 + \pi) \,\check{\mathbf{e}}_2) = -e^{c_0} \cos \left(m \, \pi\right) \cos \left(z_1\right) \cos \left(m \, z_1\right) - e^{c_0} \cos \left(m \, \pi\right) \sin \left(z_1\right) \cos \left(m \, z_1\right) \,\check{\mathbf{e}}_1 - e^{c_0} \cos \left(m \, \pi\right) \cos \left(z_1\right) \sin \left(m \, z_1\right) \,\check{\mathbf{e}}_2.$$

¹There is an errata in Ref. 6, First Eq. p. 1031. The $\check{\mathbf{e}}_2$ argument on the left of the equation should be $m(z_1 + \pi)$. The right side should be multiplied by e^{c_0} . The equation should be



Fig. 7. Curve C_{13} given by (29) embedded in three-dimensional space. Notes: Powers of the scator with $\varphi_y = 3\varphi_x$, for $\mathring{\varphi}(1; \frac{\pi}{32}, \frac{3\pi}{32})$ in large dots $\mathring{\varphi}(1; \frac{\pi}{32}, \frac{3\pi}{32})$ in smaller dots. [Reproduced from Ref. 7]; Curve C_{13} is shown superimposed on a semitransparent cusphere. Three bodies are positioned at the vertices of an isosceles triangle. A yellow dot at the origin represents the center of mass.

If we write $\frac{\tan(3\varphi_x)}{\tan\varphi_x} = \frac{y}{x}$ in (31) and solve for $\tan^2 \varphi_x = \frac{x^2}{s^2}$ from (31), s is obtained

$$s^{2} = \frac{x^{2} \left(x - 3y\right)}{3x - y}.$$
 (32b)

In terms of algebraic geometry, the curve C_{13} in \mathbb{S}^{1+2} is the set of all points (s; x, y) which satisfy the polynomials

$$(s^{2} + x^{2})(s^{2} + y^{2}) - s^{2} = 0, \qquad \text{[unit magnitude scator]}$$
$$y(s^{2} - 3x^{2}) - x(3s^{2} - x^{2}) = 0. \qquad [\varphi_{y} = 3\varphi_{x} \text{ condition]}$$

5.1. Three bodies in the C_{13} curve

Consider three bodies located symmetrically with respect to the parameter φ_x , $\overset{o}{\varphi}_1 = \overset{o}{\varphi}(\varphi)$, $\overset{o}{\varphi}_2 = \overset{o}{\varphi}(\varphi + \frac{2\pi}{3})$ and $\overset{o}{\varphi}_3 = \overset{o}{\varphi}(\varphi + \frac{4\pi}{3})$, where the subindex in φ_x has been dropped. The parametric representation of the three bodies is

$$\begin{aligned} \overset{o}{\varphi}_{1} &= \cos\varphi\cos\left(3\varphi\right) + \cos\left(3\varphi\right)\sin\varphi\check{\mathbf{e}}_{x} + \cos\varphi\sin\left(3\varphi\right)\check{\mathbf{e}}_{y}, \quad (33a) \\ \overset{o}{\varphi}_{2} &= \cos\left(\varphi + \frac{2\pi}{3}\right)\cos\left(3\varphi\right) + \cos\left(3\varphi\right)\sin\left(\varphi + \frac{2\pi}{3}\right)\check{\mathbf{e}}_{x} \\ &+ \cos\left(\varphi + \frac{2\pi}{3}\right)\sin\left(3\varphi\right)\check{\mathbf{e}}_{y}, \quad (33b) \\ \overset{o}{\varphi}_{3} &= \cos\left(\varphi - \frac{2\pi}{3}\right)\cos\left(3\varphi\right) + \cos\left(3\varphi\right)\sin\left(\varphi - \frac{2\pi}{3}\right)\check{\mathbf{e}}_{x} \end{aligned}$$

$$+\cos\left(\varphi-\frac{2\pi}{3}\right)\sin\left(3\varphi\right)\check{\mathbf{e}}_{y}.$$
 (33c)

5.2. Curve projection in two-dimensional planes

The projections of the three-dimensional curve (29) onto the planes produced by each pair of axes produce the trifolium rhodonea, Pascal's trisectrix limaçon and Gerono's lemniscata are depicted in Figure 8. Let us appraise the possibilities of these curves regarding three-body choreographies. Three elementary conditions imposed are on the motion for an isolated system:

- (1) The center of mass should be constant in an adequate inertial frame of reference. No external force acts on the system.
- (2) No collisions should take place.



Fig. 8. The projection of the $\varphi_0 = 1, \varphi_y = 3\varphi_x$ scator curve generates: the trifolium rose (left) in the $s, \check{\mathbf{e}}_x$ plane, Pascal's trisectrix (middle) in the $s, \check{\mathbf{e}}_y$ plane and a scaled and rotated Gerono's lemniscata (right) in the $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ plane.

(3) The angular momentum should be constant. The direction of the force between bodies lies on the line joining them or is a central force.

6. Trifolium Rose

Substitution of y from (32a) in the unit scator square magnitude (15) gives upon rearrangement,

$$s^{2}(s^{2}-3x^{2})^{2} = (s^{2}+x^{2})^{4}.$$

From the square root of this expression, the trifolium rose implicit equation is obtained,

$$s(s^2 - 3x^2) = (s^2 + x^2)^2.$$

If the equation for the square magnitude is plotted, the trifolium rose and its mirror image are obtained, i.e. an hexa-folium rose. The parametric curve is given by the $s, \check{\mathbf{e}}_x$ components of the 3D curve

$$\left. \stackrel{o}{\varphi} \right|_{s, \check{\mathbf{e}}_x} = \cos\left(\varphi\right) \cos\left(3\varphi\right) + \cos\left(3\varphi\right) \sin\left(\varphi\right) \check{\mathbf{e}}_x,$$

If three bodies are located symmetrically with respect to the angle parameter φ , the center of mass remains at the origin throughout the motion as φ evolves from 0 to 2π . The three bodies lie in the vertices of an equilateral triangle, as shown in Figure 9. As the parameter evolves, the triangle rotates and its length changes. However, the



Fig. 9. Three-body choreography in the trifolium rose.

three bodies collide at the origin when $\varphi = \frac{\pi}{2}$, as can be seen from 33(a) to 33(c). So, this trajectory is not a candidate for a three-body choreography. However, it should be noted that four bodies located symmetrically with respect to the angle parameter, $\overset{o}{\varphi}_1 = \overset{o}{\varphi}(\varphi), \overset{o}{\varphi}_2 = \overset{o}{\varphi}(\varphi + \frac{\pi}{4}), \overset{o}{\varphi}_3 = \overset{o}{\varphi}(\varphi + \frac{\pi}{2})$ and $\overset{o}{\varphi}_4 = \overset{o}{\varphi}(\varphi + \frac{3\pi}{4})$, still retain the center of mass at the origin but no longer exhibit collisions.

7. Trisectrix Limaçon

From the unit magnitude squared (15), x is equal to

$$x^{2} = s^{2} \frac{1 - (s^{2} + y^{2})}{(s^{2} + y^{2})}.$$

If x is substituted in (32a), upon rearrangement, the implicit curve for the trisectrix limaçon of Pascal and its mirror image are obtained. The curve $(2s + 1)^2 + 4y^2 = ((2s + 1)^2 + 4y^2 - 2(2s + 1))^2$, is displaced by 1 in the abscissas and variables scaled by 2 from the usual $(s^2 + y^2) = (s^2 + y^2 - 2s)^2$ limaçon. The parametric curve is given by the $s, \check{\mathbf{e}}_y$ components,

$$\left. \stackrel{o}{\varphi} \right|_{s,\check{\mathbf{e}}_{y}} = \cos\left(\varphi\right) \cos\left(3\varphi\right) + \cos\left(\varphi\right) \sin\left(3\varphi\right) \check{\mathbf{e}}_{y}.$$

If three bodies are located symmetrically with respect to the parameter, $\overset{o}{\varphi}_1 = \overset{o}{\varphi}(\varphi), \overset{o}{\varphi}_2 = \overset{o}{\varphi}(\varphi + \frac{2\pi}{3})$ and $\overset{o}{\varphi}_3 = \overset{o}{\varphi}(\varphi + \frac{4\pi}{3})$, the center of mass remains at the origin throughout the motion as φ evolves from 0 to 2π . It is interesting to note that every position of the three bodies throughout the trajectory is an Euler configuration, that is, the three bodies always lie in a straight line, as shown for $\varphi = 0.252$ in Figure 10. However, two of the bodies, in turn, collide at the crossing point when $\varphi = \frac{\pi}{3} \mod \frac{\pi}{3}$. If the initial positions are modified (not symmetrical with respect to the parameter) in order to avoid collisions, the center of mass is no longer fixed. So, this trajectory should also be excluded. Nonetheless, it should be noted that in the full three-dimensional curve, there are no collisions, since there are no crossings of the curve C_{13} in \mathbb{S}^{1+2} .



Fig. 10. Three-body configuration in Pascal's limaçon. The center of mass is located at the origin and the three bodies always lie in a line. However, two bodies at the crossing inevitably collide.

8. Lemniscatae

Lemniscatae can be obtained in two ways from the isometric cusphere surface.

8.1. Lemniscata with constant additive director component

One possibility is to consider a constant x or y director component, say $y = \cos \varphi_x \sin \varphi_y = c_1$. The magnitude squared is then

$$(s^{2} + x^{2})(s^{2} + c_{1}^{2}) = s^{2}.$$
(34)

Plots of this curve are shown in Figure 11. For $c_1 = 0$, the curve becomes a circle, for $c_1 \ll 1$, the curves look like wings of butterflies, Papalotl in Nahuatl language. As c_1 approaches one, lemniscataelike curves are obtained. The scator 11(b), written in terms of the φ_x parameter, is

$$\overset{o}{\varphi} = \cos\varphi_x \sqrt{1 - c_1^2 \sec^2\varphi_x} + \sin\varphi_x \sqrt{1 - c_1^2 \sec^2\varphi_x} \,\check{\mathbf{e}}_x + c_1 \,\check{\mathbf{e}}_y. \tag{35}$$



Fig. 11. Papálotl–lemniscata curves for different values of $0 \le c_1 \le 1$.

The implicit equation of the curve is $(x^2 + y^2)(x^2 + c_1^2) = x^2$. The curves can be prevented from becoming ever smaller as $c_1 \to 1$ by an adequate normalization so that its width remains constant, $(x^2 + y^2)(x^2(1 - c_1^2) + c_1^2) = x^2$. From (35), the coefficients are real for $0 \leq |\varphi_x| \leq |\arccos c_1|$. As far as generating the lemniscata curve is concerned, this poses no problem. However, if the parameter is used to describe the trajectory of a body, no position is obtained in the intervals where $|\varphi_x| > |\arccos c_1| \mod 2\pi$.

8.2. Lemniscata from $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ projection with $\varphi_y = 3\varphi_x$

The other approach comes from the projection of (7) in the $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ plane. Substitution of s from (32b) in the scator magnitude (15) gives

$$1 = \frac{x^2 \left(x - 3y\right)}{3x - y} \left(1 + \frac{3x - y}{x - 3y}\right) \left(1 + \frac{y^2 \left(3x - y\right)}{x^2 \left(x - 3y\right)}\right),$$

that can be rearranged to obtain the implicit scator lemniscata equation

$$(3x - y) (x - 3y) = 4 (x - y)^4.$$
(36)

The parametric curve is given by the $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ components,

$$\left. \overset{o}{\varphi} \right|_{\check{\mathbf{e}}_{x},\check{\mathbf{e}}_{y}} = \cos\left(\varphi\right)\cos\left(3\varphi\right) + \cos\left(3\varphi\right)\sin\left(\varphi\right)\check{\mathbf{e}}_{x} + \cos\left(\varphi\right)\sin\left(3\varphi\right)\check{\mathbf{e}}_{y}.$$

This lemniscata is rotated $\frac{\pi}{4}$ with respect to the coordinate axes (see Figure 8). A Euclidean rotation in the $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ plane, $x = \frac{1}{\sqrt{2}}(x'+y')$, $y = \frac{1}{\sqrt{2}}(y'-x')$, involves the substitutions $x-y \to \sqrt{2x'}, 3x-y \to \sqrt{2}(2x'+y')$ and $x-3y \to \sqrt{2}(2x'-y')$. The implicit equation is then $y'^2 = 4x'^2(1-2x'^2)$. Scaling by $y' \to \sqrt{2y}, x' \to \frac{x}{\sqrt{2}}$, the polynomial becomes $y^2 = x^2(1-x^2)$, which is the canonical form of the Gerono/Huygens lemniscata. These curves are shown in Figure 12. The Euclidean $\frac{\pi}{4}$ rotation in the parametric variable is

$$x' = \frac{1}{\sqrt{2}} \left(\cos 3\varphi \sin \varphi - \cos \varphi \sin 3\varphi \right) = -\frac{1}{\sqrt{2}} \sin 2\varphi.$$
$$y' = \frac{1}{\sqrt{2}} \left(\cos \varphi \sin 3\varphi + \cos 3\varphi \sin \varphi \right) = \frac{1}{\sqrt{2}} \sin 4\varphi.$$



Fig. 12. Lemniscatae in $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ plane, scator lemniscata given by Eq. (36) (thick black), $\frac{\pi}{4}$ rotated (green), and scaled by $y' \to \sqrt{2}y$, $x' \to \frac{x}{\sqrt{2}}$, reproducing the Gerono lemniscata (magenta).

The rotated scator is

$$\overset{o'}{\varphi} = \cos\varphi\cos3\varphi \mp \frac{1}{\sqrt{2}}\sin\left(2\varphi\right)\,\check{\mathbf{e}}_x + \frac{1}{\sqrt{2}}\sin\left(4\varphi\right)\,\check{\mathbf{e}}_y.$$

The minus or plus sign is introduced to allow for the parameter increment to draw the figure counterclockwise or clockwise, respectively. This rotation has to be handled with great care in scator space because it is not scator magnitude invariant. The unrotated scator magnitude equation (15) with rotated variables is

$$s^{2} = s^{4} + s^{2} \left(x^{\prime 2} + y^{\prime 2} \right) + \frac{1}{4} \left(x^{\prime 2} - y^{\prime 2} \right)^{2}.$$
 (37)

It is clear that $\dot{\varphi}' = s + x' \check{\mathbf{e}}_x + y' \check{\mathbf{e}}_y$ no longer satisfies an expression of the form $s^2 = (s^2 + x'^2) (s^2 + y'^2)$ but (37). The $\check{\mathbf{e}}_x, \check{\mathbf{e}}_y$ components

of the rotated scator are

$$\left. \begin{array}{c} {}^{o} \varphi' \right|_{\check{\mathbf{e}}_{x},\check{\mathbf{e}}_{y}} = \mp \frac{1}{\sqrt{2}} \sin\left(2\varphi\right) \check{\mathbf{e}}_{x} + \frac{1}{\sqrt{2}} \sin\left(4\varphi\right) \check{\mathbf{e}}_{y}. \end{array} \right.$$

The π periodicity for m = 3, mentioned in Section 5, is explicit in the even arguments of this expression. The parameter is scaled to $\phi = 2\varphi$ to recover the usual 2π periodicity. Recall that the scator products of elements with vanishing scalar component (necessarily with a single director component in order to belong to the \mathbb{S}^{1+2} where the product is defined) is $\check{\mathbf{e}}_x \check{\mathbf{e}}_x = \check{\mathbf{e}}_y \check{\mathbf{e}}_y = -1$ and $\check{\mathbf{e}}_x \check{\mathbf{e}}_y = 0$. In contrast, vectors do not have a scalar component and three different products are defined: product of a scalar with a vector, and inner and outer products between vector elements. The vector set is not a subset of the scator set. However, the real $\mathbb{R} = \mathbb{S}^{1+0}$ and complex $\mathbb{C} = \mathbb{S}^{1+1}$ sets are subsets of the scator set $\mathbb{S}^{1+0} \subset \mathbb{S}^{1+1} \subset \mathbb{S}^{1+2}$. Nonetheless, it is possible to map the scator coefficients onto a vector set, $s + x \check{\mathbf{e}}_x + y \check{\mathbf{e}}_y \mapsto x \hat{\mathbf{e}}_x + y \hat{\mathbf{e}}_y + s \hat{\mathbf{e}}_z$.

9. Dynamics of the Gerono-Type Lemniscata

The vector representing the parametric representation of the Gerono type Lemniscata is

$$\mathbf{r} = -\sin\phi\,\hat{\mathbf{e}}_x + \sin\left(2\phi\right)\,\hat{\mathbf{e}}_y,$$

where the usual unit vectors (with hats) instead of checks are used. The angle θ , measured counterclockwise from the abscissas, in terms of the parameter ϕ is

$$\tan \theta = \frac{\sin\left(2\phi\right)}{-\sin\phi}.$$

The positions of the three bodies are

$$\mathbf{r}_1 = -\sin\phi \,\hat{\mathbf{e}}_x + \sin\left(2\phi\right) \,\hat{\mathbf{e}}_y,\tag{38a}$$

$$\mathbf{r}_{2,3} = \left(\frac{1}{2}\sin\phi \mp \frac{\sqrt{3}}{2}\cos\phi\right)\,\hat{\mathbf{e}}_x + \left(-\frac{1}{2}\sin\left(2\phi\right) \mp \frac{\sqrt{3}}{2}\cos\left(2\phi\right)\right)\,\hat{\mathbf{e}}_y,\tag{38b}$$

where the upper sign corresponds to \mathbf{r}_2 and the lower sign to \mathbf{r}_3 . Evaluation of the sum of (38a) and (38b) shows that the center of mass is located at the origin for all times, $\mathbf{r}_{cm} = m \sum_{i}^{3} \mathbf{r}_{i} = 0$. The moment of inertia is

$$I = m \sum_{i}^{3} \mathbf{r}_{i} \cdot \mathbf{r}_{i} = 3 m.$$
(39)

Since the center of mass is zero, the sum of distances between bodies is equal to $\frac{3I}{m}$,

$$\sum_{i< j} |\mathbf{r}_i - \mathbf{r}_j|^2 = \frac{1}{m} 3m \sum \mathbf{r}_i \cdot \mathbf{r}_i = 9.$$
(40)

For $\phi = 0$, $\mathbf{r}_1 = 0$ and $\mathbf{r}_{2,3} = \mp \left(\frac{\sqrt{3}}{2}\hat{\mathbf{e}}_x + \frac{\sqrt{3}}{2}\hat{\mathbf{e}}_y\right)$, $\tan \theta_{2,3} = \mp 1$. The three bodies lie in an Euler configuration, co-linear in a line with $\frac{\pi}{4}$ slope, as shown in Figure 13. The Euler configuration is attained again for $\phi = \frac{\pi}{3}$, but the slope is then $-\frac{\pi}{4}$ and the body 2 sits at the origin. The pattern is repeated after $\frac{\pi}{3}$, with alternating slopes as well as the body located at the origin. For $\phi = \frac{\pi}{2}$, the bodies lie in the vertices of an equilateral triangle, $\mathbf{r}_1 = -1$ and $\mathbf{r}_{2,3} = \frac{1}{2}\hat{\mathbf{e}}_x \mp \frac{\sqrt{3}}{2}\hat{\mathbf{e}}_y$, $\tan \theta_{2,3} = \mp \frac{\pi}{3}$.

From the three curves, trifolium, limaçon and lemniscata, only the lemniscata is invariant under inversion. The following theorem, due to Ozaki,¹³ can then be used to find the position of the three points given one of them:

Theorem (Construction of three points). If a curve γ in \mathbb{R}^d with $d = 2, 3, 4, \ldots$ is invariant under the inversion $q \mapsto -q$, then the set $\{\{q_1, q_2\} | q_1, q_2 \in \gamma, q_1 + q_2 + q_3 = 0\}$ for a given $q_3 \in \gamma$ is equal to the set $\{\{q, q^*\} | q \in \gamma \cap \gamma_{\parallel}\}$ where γ_{\parallel} is the parallel translation $q \mapsto q - q_3$ of the curve γ and $q^* = -q - q_3$.

Consider the point $\mathbf{r}_1 = -1$ (q_3 in Theorem (Construction of three points)), shifting this point to the origin, the intersection of the curves, γ and γ_{\parallel} , give the other two positions, as illustrated in Figure 14.



Fig. 13. Three bodies in a Gerono-type lemniscata. (a) Euler configuration, total angular momentum is zero (b) Equilateral triangle configuration with $\sqrt{3}$ per side.



Fig. 14. Construction of three points using Ozaki's theorem.¹³ Body 1 (leftmost in red) is displaced to the origin with a parallel translation of the lemniscata curve. The intersections between the two curves give the positions of the other two bodies (equal to configuration in Figure 13(b)).

The velocities of the three bodies, obtained by differentiation of (38a),(38b), are

$$\mathbf{v}_1 = -\cos\phi \,\dot{\phi} \,\hat{\mathbf{e}}_x + 2\cos\left(2\phi\right) \,\dot{\phi} \,\hat{\mathbf{e}}_y,\tag{41}$$

and

$$\mathbf{v}_{2,3} = \left(\frac{1}{2}\cos\phi \pm \frac{\sqrt{3}}{2}\sin\phi\right)\dot{\phi}\,\hat{\mathbf{e}}_x + \left(-\cos\left(2\phi\right)\pm\sqrt{3}\sin\left(2\phi\right)\right)\dot{\phi}\,\hat{\mathbf{e}}_y,\tag{42}$$

The kinetic energy of the system is

$$\sum_{i}^{3} \mathbf{v}_{ix} \cdot \mathbf{v}_{ix} = \frac{3}{2} \dot{\phi}^{2}, \quad \sum_{i}^{3} \mathbf{v}_{iy} \cdot \mathbf{v}_{iy} = 6 \dot{\phi}^{2}, \quad \sum_{i}^{3} \mathbf{v}_{i} \cdot \mathbf{v}_{i} = \frac{15}{2} \dot{\phi}^{2}.$$
(43)

If the parameter ϕ is linear in time, the kinetic energy is conserved, as well as the kinetic energies in each direction. Recall that the sum of velocities in each direction is also constant since the center of mass is constant.

9.1. Angular momentum

For the three-body problem, the shape of the orbit, conservation of the center of mass and a constant of motion (the angular momentum or the total energy) determine the motion of the three bodies.¹³ Let us determine the angular momentum of the three bodies to describe the motion,

$$\left(\mathbf{r}_{1} \times \mathbf{v}_{1}\right)_{z} = \left(-2\sin\left(\phi\right)\cos\left(2\phi\right) + \cos\left(\phi\right)\sin\left(2\phi\right)\right)\dot{\phi}$$

and

$$(\mathbf{r}_{2,3} \times \mathbf{v}_{2,3})_z = \left(\frac{1}{4}\sin\phi\cos\left(2\phi\right) - \frac{5}{4}\cos\phi\sin\left(2\phi\right)\right)\dot{\phi}$$
$$\pm \frac{3\sqrt{3}}{4}\left(\cos\phi\cos\left(2\phi\right) + \sin\phi\sin\left(2\phi\right)\right)\dot{\phi}$$

The total angular momentum L of the system is then

$$L = \sum_{i}^{3} \left(\mathbf{r}_{i} \times \mathbf{v}_{i} \right)_{z} = -\frac{3}{2} \sin \left(3\phi \right) \dot{\phi}.$$

$$\tag{44}$$

Therefore, the angular momentum for a linear parameter ϕ is not constant. The same issue is present in the two-body problem (Section 4.3). If L is requested to be constant, upon integration

$$\phi = \frac{1}{3}\arccos(2L\,t)\,, \quad \dot{\phi} = -\frac{1}{3}\frac{2L}{\sqrt{1-4L^2t^2}}$$

In contrast with the two-body solution, no periodic motion can be achieved with constant angular momentum. This impossibility is due to the fact that $\sin(3\phi)$ is zero for $\frac{\pi}{3} \mod \pi$. At those points, the body at the origin obviously has zero momentum, and the other two have opposite positions but the same velocity

$$\mathbf{r}_{2} = -\mathbf{r}_{3} = -\frac{\sqrt{3}}{2}\,\hat{\mathbf{e}}_{x} - \frac{\sqrt{3}}{2}\,\hat{\mathbf{e}}_{y}, \qquad \mathbf{v}_{2} = \mathbf{v}_{3} = \left(\frac{1}{2}\hat{\mathbf{e}}_{x} - \hat{\mathbf{e}}_{y}\right)\dot{\phi}.$$

Their momenta are thus opposite $(\mathbf{r}_{2,3} \times \mathbf{v}_{2,3})_z = \pm \frac{3\sqrt{3}}{4}\dot{\phi}$ and their sum is zero, as graphically seen in Figure 13(a). A constant L, (Eq. (44)), then requires a nonphysical infinite parameter velocity $\dot{\phi}$. On the other hand, it is not possible to impose a zero angular momentum condition, L = 0, because then ϕ must be constant.

10. Conclusions

Scator algebra provides an adequate framework to describe geometrical objects and trajectories in 1 + n-dimensional spaces. This algebra is endowed with sum and product operations as well as a main second-order involution. It has two representations, additive and multiplicative that correspond to the rectangular and polar representations in complex algebra. These representations provide a natural parametrization of the trajectories when the additive components (s; x, y) are written in terms of the multiplicative angle variables (φ_x, φ_y) . Scator elements in \mathbb{S}^{1+2} are represented geometrically in three-dimensional Euclidean space, letting one axis represent the scalar component and the other two axes represent the director components.

Important 2. Cusphere: The cusphere is the scator isometric surface in \mathbb{S}^{1+2} where various curves have been described.

Orthographic projections of the curves lying on the cusphere recreate ellipses, circles and lines when one of the two director angles is considered constant. The projections reproduce the familiar trifolium rose, Pascal's limaçon and Gerono's lemniscata plane curves, when the director angles are in a relationship of 1:3.

Important 3. cexp: The components exponential function is a scator function $\stackrel{o}{\operatorname{cexp}} \in \mathbb{S}^{1+n}$ of scator variable $\stackrel{o}{\zeta} \in \mathbb{R}^{1+n}$ that generalizes the complex exponential to higher dimensions.

It maps a constant scalar component argument into a constant magnitude surface, that is, a cusphere. Lines in $\zeta \in \mathbb{S}^{1+2}$ parallel to one director axis are mapped into ellipses. In \mathbb{S}^{1+2} geometry, the parameter φ_y represents the angle between the scalar and the $\check{\mathbf{e}}_y$ axes. However, in elliptical trajectories, it also represents the eccentric anomaly angle in the $(\check{\mathbf{e}}_x, \check{\mathbf{e}}_y)$ plane. In contrast with the real and complex formalisms, the ellipse in \mathbb{S}^{1+2} has constant scator magnitude along the trajectory, the two points when the curve crosses the s = 0 plane being exceptional points. Being a unitary continuous transformation, the scator formalism provides a simple and straightforward description of the elliptical planetary orbit. The cexp functions map lines with slope three onto the C_{13} line that produce the trifolium rose, trisectrix limaçon and Gerono-type lemniscatae in the mutually orthogonal axes planes. Symmetrical placement of three bodies with respect to the φ parameter insures that the center of mass is located at the origin for all times. Threebody choreographies in the trifolium rose and the trisectrix limaçon trajectories have a constant center of mass but exhibit collisions that make them unsuitable in two-dimensional plane trajectories. The Gerono-type lemniscata does not present collisions and has a constant sum of square distances. However, a constant angular momentum is not physically realizable.

The three-body choreography in the Bernoulli lemniscata has been successfully parameterized with elliptic functions by Fujiwara *et al.*¹⁴ A comparison of relevant parameters between these two lemniscatae is shown in Table 1. The main departure comes from the angular momentum, it is null for the Bernoulli lemniscata parameterized with elliptic functions, whereas it is not constant for the Gerono lemniscata parameterized with trigonometric functions. A variable angular momentum is usually compensated by an appropriate dependence of the parameter with respect to time. However, in contrast with the elliptic trajectory, the angular momentum is zero for the Euler configuration and finite for other configurations. Thus, a function without singularities is not realizable in this case. Nonetheless, it should be possible to propose other scator curves where the angular momentum expression is never zero (or always zero) throughout the trajectory. In that case, the time dependence of the parameter

	Bernoulli lemniscata 14	Gerono-type lemniscata
I	$\sqrt{3}$	3
cm	0	0
$\sum_{i < j} \left \mathbf{r}_i - \mathbf{r}_j \right ^2$	$3\sqrt{3}$	9
$\sum_{i}^{3} \mathbf{v}_{i} \cdot \mathbf{v}_{i}$	$\frac{3}{4}$	$\frac{15}{2}\dot{\phi}^2$
$\sum_{i}^{3} \left(\mathbf{r}_{i} \times \mathbf{v}_{i} ight)_{z}$	0	$-\frac{3}{2}\sin(3\phi)\dot{\phi}$
Parameterize	Jacobian elliptic functions	Trigonometric products

Table 1.Comparison of some relevant quantities in the Bernoulli andGerono Lemniscatae.

as a function of time could be physically attainable. Rational relationships between the φ_x and φ_y angle parameters produce closed curves. A constant φ_x with two bodies and a 3:1 relation with three bodies has been explored here. A wealth of other trajectories, projections and number of bodies is clearly possible. Some of them look remarkably similar to the numerical trajectories obtained by Simó.¹⁵

Acknowledgments

The author acknowledges the effort of the reviewers leading to the improvement of the manuscript.

References

- M. Fernandez-Guasti, Roots of second order polynomials with real coefficients in elliptic scator algebra, J. New Theory 36, 39–48 (2021).
- M. Fernandez-Guasti, Roots of elliptic scator numbers, Axioms 10(4), 321 (2021).
- J. L. Cieśliński and A. Kobus, On the product rule for the hyperbolic scator algebra, Axioms 9(55) (2020).
- M. Fernandez-Guasti, Composition of velocities in a scator deformed Lorentz metric, *Eur. Phys. J. Plus* 135(542) (2020).
- M. Fernández-Guasti and F. Zaldívar, Hyperbolic superluminal scator algebra, Adv. Appl. Clifford Algebr. 25(2), 321–335 (2015).
- M. Fernandez-Guasti, Components exponential scator holomorphic function, Math. Methods Appl. Sci. 43(3), 1017–1034 (2020).
- M. Fernandez-Guasti, Powers of elliptic scator numbers, Axioms 10(4), 250 (2021).
- 8. M. Valtonen and H. Karttunen, *The Three-Body Problem*. Cambridge University Press, Cambridge (2006).
- M. Valtonen, J. Anosova, K. Kholshevnikov, A. Mylläri, V. Orlov, and K. Tanikawa, *The Three-Body Problem from Pythagoras to Hawking*. Springer, Cham (2016).
- M. Fernández-Guasti and F. Zaldívar, An elliptic non distributive algebra, Adv. Appl. Clifford Algebr. 23(4), 825–835 (2013).
- M. Fernández-Guasti, Differential quotients in elliptic scator algebra, Math. Methods Appl. Sci. 41(12), 4827–4840 (2018).
- J. L. Cieśliński and D. Zhalukevich, Explicit formulas for all scator holomorphic functions in the (1+2)-dimensional case, Symmetry 12(9), 1550 (2020).

- 13. H. Ozaki, H. Fukuda, and T. Fujiwara, Three-body choreographies in given curves, J. Phys. A 42(39), 395205, 16 (2009).
- 14. T. Fujiwara, H. Fukuda, and H. Ozaki, Choreographic three bodies on the lemniscate, J. Phys. A **36**(11), 2791–2800 (2003).
- R. Moeckel, Central configurations. In Central Configurations, Periodic Orbits, and Hamiltonian Systems, Adv. Courses Math. CRM Barcelona, pp. 105–167. Birkhäuser/Springer, Basel (2015).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0010

Chapter 10

Caristi-Type Nonunique Fixed-Point Results and Fixed-Circle Problem on $b_v(s)$ -Metric Spaces

Nihal Taş^{*,†,§} and Ozgur $\operatorname{Ege}^{\ddagger,\P}$

[†]Department of Mathematics, Arts and Science Faculty, Balıkesir University, 10145, Çağış Campus, Balıkesir, Turkey [‡]Department of Mathematics, Faculty of Science, Ege University, Bornova, 35100, Izmir, Turkey [§]nihaltas@balikesir.edu.tr ¶ozqur.eqe@eqe.edu.tr

Fixed-point theory has been comprehensively studied with several methods. One of these methods is to generalize the used metric space such as $b_v(s)$ -metric spaces. Another method is to analyze the geometric features of the fixed-point set. In the light of these methods, in this chapter, we prove Caristi's fixed-point theorem and new fixed-figure theorems in $b_v(s)$ -metric spaces. We present some examples to emphasize the significance of geometrical results. To further strengthen the obtained theoretical results, we establish an application to S-Shaped Rectified Linear Unit (SReLU) activation functions.

^{*}Corresponding author.

1. Introduction and Preliminaries

Fixed-point theory has been a very popular theory in mathematics and sciences. It has a great variety of applications to various kind of problems arise in diverse fields. Existence and uniqueness problems of fixed points are practical in many areas.¹⁻¹²

Since the notion of metric space is a very useful tool for many areas of mathematics, researchers generalize this and then try to find corresponding findings in generalized cases. The notion of b-metric space is the most eminent generalized metric space which was defined in 1989.

Definition 1 (Ref. 13). Let \mathfrak{E} be a nonempty set and $s \geq 1$ a real number. A function $\mathfrak{b} : \mathfrak{E} \times \mathfrak{E} \to [0, \infty)$ is a *b*-metric if, for all $\mathfrak{e}, \mathfrak{a}, \mathfrak{z} \in \mathfrak{E}$, the following hold:

 $\begin{array}{l} (b_1) \ \mathfrak{b}(\mathfrak{e},\mathfrak{a}) = 0 \ \text{iff} \ \mathfrak{e} = \mathfrak{a}, \\ (b_2) \ \mathfrak{b}(\mathfrak{e},\mathfrak{a}) = \mathfrak{b}(\mathfrak{a},\mathfrak{e}), \\ (b_3) \ \mathfrak{b}(\mathfrak{e},\mathfrak{z}) \leq s[\mathfrak{b}(\mathfrak{e},\mathfrak{a}) + \mathfrak{b}(\mathfrak{a},\mathfrak{z})]. \end{array}$

The pair $(\mathfrak{E}, \mathfrak{b})$ is said to be a *b*-metric space.

A *b*-metric space is a generalization of a metric space. With this new concept, mathematicians have obtained numerous fixed-point results on various mappings in this space (see Refs. 14-19).

Definition 2 (Ref. 20). A function $f : \mathfrak{E} \to \mathbb{R}$ is called lower semicontinuous mapping if for any $\{\mathfrak{e}_n\} \subset \mathfrak{E}$ and $\mathfrak{e} \in \mathfrak{E}$

$$\mathfrak{e}_n \to \mathfrak{e} \quad \Rightarrow \quad f(\mathfrak{e}) \leq \lim_{n \to \infty} \inf(f(\mathfrak{e}_n)).$$

After *b*-metric spaces, some generalizations of this space such as extended *b*-metric space, partial *b*-metric space, dislocated *b*-metric space, and rectangular *b*-metric space²¹ were investigated. The most recent type generalized *b*-metric space was given in 2017. This metric space which is called $b_v(s)$ -metric space is very important because it is the most complicated and generalized version of metric space.

Definition 3 (Ref. 22). Let \mathfrak{E} be a nonempty set, $\mathfrak{b}_{\mathfrak{v}} : \mathfrak{E} \times \mathfrak{E} \to [0,\infty)$ a function and $v \in \mathbb{N}$. *d* is called a $b_v(s)$ -metric space if there is a real number $s \geq 1$ such that for all $\mathfrak{e}, \mathfrak{z} \in \mathfrak{E}$ and for all distinct

points $\mathfrak{a}_1, \mathfrak{a}_2, \ldots, \mathfrak{a}_v \in \mathfrak{E}$, each of them different from \mathfrak{e} and \mathfrak{z} the following hold:

$$\begin{array}{l} (b_{v_1}) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{a}) = 0 \ \text{iff} \ \mathfrak{e} = \mathfrak{a}, \\ (b_{v_2}) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{a}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a},\mathfrak{e}), \\ (b_{v_3}) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{z}) \leq s[\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{a}_1) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a}_1,\mathfrak{a}_2) + \ldots + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a}_v,\mathfrak{z})]. \end{array}$$

The pair $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is said to be a $b_v(s)$ -metric space.

Definition 4 (Ref. 22). Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_v(s)$ -metric space, $\{\mathfrak{e}_n\}$ a sequence in \mathfrak{E} and $\mathfrak{e} \in \mathfrak{E}$.

- (i) $\{\mathbf{e}_n\}$ is convergent in $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ and converges to \mathbf{e} , if for every $\epsilon > 0$ there exists $n_0 \in \mathbb{N}$ such that $\mathfrak{b}_{\mathfrak{v}}(\mathbf{e}_n, \mathbf{e}) < \epsilon$ for all $n > n_0$ and this fact is represented by $\lim_{n\to\infty} \mathbf{e}_n = \mathbf{e}$ or $\mathbf{e}_n \to \mathbf{e}$ as $n \to \infty$.
- (ii) $\{\mathbf{e}_n\}$ is a Cauchy sequence in $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ if for every $\epsilon > 0$ there exists $n_0 \in \mathbb{N}$ such that $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+p}) < \epsilon$ for all $n > n_0, p > 0$ or if $\lim_{n\to\infty} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+p}) = 0$ for all p > 0.
- (iii) $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is called a complete $b_v(s)$ -metric space if every Cauchy sequence in \mathfrak{E} converges to some $\mathfrak{e} \in \mathfrak{E}$.

Example 1 (Ref. 23). Define a mapping $\mathfrak{b}_{\mathfrak{v}} : \mathfrak{E} \times \mathfrak{E} \to [0, \infty)$ by

$$\mathfrak{b}_{\mathfrak{v}}\left(\frac{1}{k},\frac{1}{m}\right) = \begin{cases} |k-m|, & |k-m| \neq 1\\ \frac{1}{2}, & |k-m| = 1 \end{cases}$$

on the set $\mathfrak{E} = \left\{ \frac{1}{n} \mid n \in \mathbb{N}, n \geq 2 \right\}$. $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a $b_3(3)$ -metric space.

 $b_v(s)$ -metric space generalizes *v*-generalized metric space, *b*-metric space, rectangular metric space and rectangular *b*-metric space. Banach and Reich fixed-point theorems are proved in Ref. 22. Abdullahi and Kumam²⁴ introduced partial $b_v(s)$ -metric spaces. They provided topological features and proved some fixed-point results in this space. Khan and Dass–Gupta type fixed-point theorems are proved.²⁵ Aydi *et al.*²⁶ obtained some common fixed-point theorems in partial $b_v(s)$ -metric spaces. For more details, see Refs. 27–32.

Next, we recall Caristi's fixed-point theorem.

Theorem 1 (Ref. 20). Let (\mathfrak{E}, d) be a complete metric space. Suppose that $\xi : \mathfrak{E} \to \mathbb{R}^+$ is lower semicontinuous and the mapping $\Gamma: \mathfrak{E} \to \mathfrak{E} \text{ satisfies}$

$$d(\mathfrak{e}, \Gamma \mathfrak{e}) \leq \xi(\mathfrak{e}) - \xi(\Gamma \mathfrak{e})$$

for every $\mathfrak{e} \in \mathfrak{E}$. Then there is $\mathfrak{e}_0 \in \mathfrak{E}$ such that $\Gamma \mathfrak{e}_0 = \mathfrak{e}_0$.

Let (\mathbb{R}, d) be a usual metric space with the metric $d : \mathbb{R} \times \mathbb{R} \to [0, \infty)$ defined by

$$d(\mathfrak{e},\mathfrak{a}) = |\mathfrak{e} - \mathfrak{a}|,$$

for all $\mathfrak{e}, \mathfrak{a} \in \mathbb{R}$. Consider the following mapping $\Gamma_i : \mathbb{R} \to \mathbb{R}$ $(i \in \{1, 2, 3, 4\})$ as

$$\Gamma_1 \mathfrak{e} = 2\mathfrak{e} + \frac{3}{2},$$

$$\Gamma_2 \mathfrak{e} = \mathfrak{e}^3 - 5\mathfrak{e}^2 + 5\mathfrak{e}^3,$$

$$\Gamma_3 \mathfrak{e} = \mathfrak{e} + 100$$

and

 $\Gamma_4 \mathfrak{e} = \mathfrak{e},$

for all $\mathfrak{e} \in \mathbb{R}$. Then we have the fixed-point sets of Γ_i , $Fix(\Gamma_i) = {\mathfrak{e} \in \mathbb{R} : \mathfrak{e} = \Gamma_i \mathfrak{e}}$ such that

$$Fix(\Gamma_1) = \left\{\frac{3}{2}\right\}, Fix(\Gamma_2) = \{-1, 1, 5\},$$

$$Fix(\Gamma_3) = \emptyset \text{ and } Fix(\Gamma_4) = \mathbb{R}.$$

From the above examples, we say that the number of fixed points of a self-mapping can change. Just in this case, fixed-circle problem has been shown up as a geometric generalization of fixed-point theory.³³ The fundamental point in this problem is to increase the number of fixed points. There are some advantages working this problem. We list a few of them:

- A geometric meaning can be attributed the fixed-point set $Fix(\Gamma)$ when the number of fixed points of Γ is more than one.
- The obtained results for this problem can be applied to some reallife problems.

• Under favor of this problem, some eminent fixed-point results can be generalized.

This problem has been studied from different perspectives. For example, in Ref. 34, some solutions of this problem were given using the Khan-type contractions.³⁵ In Ref. 36, techniques of Ćirić, Hardy– Rogers, Reich and Chatterjea were used to investigate some solutions on metric spaces (for more of these techniques, see Refs. 37–41). In Ref. 42, Mlaiki *et al.* used Wardowski's approach for fixed-circle problem on *S*-metric spaces. In Ref. 43, Özgür presented some fixeddisc results via simulation functions. Recently, this problem has been labored on metric and some generalized metric spaces (for more examples, see Refs. 44–47 and the references therein). Also, this problem has been a generalized fixed-figure problem.⁴⁸ In this context, fixed-ellipse, fixed-hyperbola, fixed-Cassini curve, fixed-Apollonious circle and fixed-*k*-ellipse theorems were obtained with diverse methods (see Refs. 48–52).

In this chapter, we get some fixed-point and fixed-figure results using the Caristi-type inequality in $b_v(s)$ -metric spaces. The proved results generalize some known results in the literature and they are supported by necessary examples. Finally, we construct an application to *SReLU* activation functions.

2. Main Results

In this section, we first prove Caristi-type fixed-point theorems and investigate some solutions to the fixed-circle problem using the Caristi-type inequality in $b_v(s)$ -metric spaces. The next definition is the modification of Definition 1.6 to $b_v(s)$ -metric spaces in Ref. 53.

Definition 5. Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_v(s)$ -metric space and Γ a selfmapping on \mathfrak{E} . If the set

$$\left\{\mathfrak{a}\in\mathfrak{E}:\lim_{i}\Gamma^{m_{i}}\mathfrak{a}=u\Longrightarrow\lim_{i}\Gamma\Gamma^{m_{i}}\mathfrak{a}=\Gamma u\right\}$$

is not empty whenever $\{ \mathfrak{e} \in \mathfrak{E} : \lim_{i} \Gamma^{m_{i}} \mathfrak{e} = u \}$ is nonempty set, then Γ is called weakly orbitally continuous.
In the following four theorems, we suppose that $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a complete $b_v(s)$ -metric space and Γ a self-mapping on \mathfrak{E} .

Theorem 2. Assume that $\xi : \mathfrak{E} \to [0, \infty)$ is a function such that for all $\mathfrak{e}, \mathfrak{a} \in \mathfrak{E}$ we have

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \xi(\mathfrak{e}) - \xi(\Gamma\mathfrak{e}). \tag{1}$$

If Γ is weakly orbitally continuous, then Γ has a fixed point.

Proof. Let $\mathfrak{e}_0 \in \mathfrak{E}$. Consider a sequence $\{\mathfrak{e}_n\}$ by $\mathfrak{e}_1 = \Gamma \mathfrak{e}_0$, $\mathfrak{e}_2 = \Gamma \mathfrak{e}_1, \ldots, \mathfrak{e}_n = \Gamma \mathfrak{e}_{n-1}, \ldots$, that is, $\mathfrak{e}_n = \Gamma^n \mathfrak{e}_0$. By inequality (1), we have

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0,\mathfrak{e}_1) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0,\Gamma\mathfrak{e}_0) \leq \xi(\mathfrak{e}_0) - \xi(\Gamma\mathfrak{e}_0) = \xi(\mathfrak{e}_0) - \xi(\mathfrak{e}_1).$$

In the same manner, we obtain the following inequalities:

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(u_1, u_2) &\leq \xi(\mathfrak{e}_1) - \xi(\mathfrak{e}_2) \\ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_2, \mathfrak{e}_3) &\leq \xi(\mathfrak{e}_2) - \xi(\mathfrak{e}_3) \\ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_3, \mathfrak{e}_4) &\leq \xi(\mathfrak{e}_3) - \xi(\mathfrak{e}_4) \\ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_4, \mathfrak{e}_5) &\leq \xi(\mathfrak{e}_4) - \xi(\mathfrak{e}_5) \\ &\vdots \\ d(\mathfrak{e}_{n-1}, \mathfrak{e}_n) &\leq \xi(\mathfrak{e}_{n-1}) - \xi(\mathfrak{e}_n) \\ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+1}) &\leq \xi(\mathfrak{e}_n) - \xi(\mathfrak{e}_{n+1}). \end{split}$$

If we add the above inequalities, we obtain

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{0},\mathfrak{e}_{1}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{1},\mathfrak{e}_{2}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{2},\mathfrak{e}_{3}) + \dots + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_{n}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n},\mathfrak{e}_{n+1}) \\ &\leq \xi(\mathfrak{e}_{0}) - \xi(\mathfrak{e}_{n+1}) \leq \xi(\mathfrak{e}_{0}). \end{split}$$

Taking limit as $n \to \infty$, we find

$$\sum_{n=0}^{\infty}\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+1})\leq\xi(\mathfrak{e}_0).$$

From the above inequality, $\{\mathfrak{e}_n\}$ is a Cauchy sequence. From the completeness of \mathfrak{E} , there is a point $r \in \mathfrak{E}$ such that

$$\lim_{n \to \infty} \mathbf{e}_n = r \quad \text{and} \quad \lim_{n \to \infty} \Gamma^{\alpha} \mathbf{e}_n = r$$

for every $\alpha \geq 1$. Assume that Γ is weakly orbitally continuous mapping. Since $\{\Gamma^n \mathfrak{e}_0\}$ converges for every $\mathfrak{e}_0 \in \mathfrak{E}$, weak orbital

continuity implies that there exists $\mathfrak{w}_0 \in \mathfrak{E}$ such that $\Gamma^n \mathfrak{w}_0 \to \mathfrak{z}$ and $\Gamma^{n+1}\mathfrak{w}_0 \to \Gamma \mathfrak{z}$ for some $\mathfrak{z} \in \mathfrak{E}$. Taking limit as $n \to \infty$, we have $\mathfrak{z} = \Gamma \mathfrak{z}$, that is, Γ has the fixed point. Hence, the proof of this theorem is completed. \Box

Definition 6 (Ref. 54). Let Γ and Θ be self-mappings on a metric space (\mathfrak{E}, d) .

- (i) A point $\mathfrak{e}_0 \in \mathfrak{E}$ is called a common fixed point of Γ and Θ if $\mathfrak{e}_0 = \Gamma \mathfrak{e}_0 = \Theta \mathfrak{e}_0$.
- (ii) A point $\mathfrak{e}_0 \in \mathfrak{E}$ is said to be a coincidence point of Γ and Θ if $\mathfrak{e}_0 = \Gamma \mathfrak{e} = \Theta \mathfrak{e}$.
- (iii) The mappings $\Gamma, \Theta : \mathfrak{E} \to \mathfrak{E}$ are called weakly compatible if $\Gamma \Theta \mathfrak{e} = \Theta \Gamma \mathfrak{e}$ whenever $\Gamma \mathfrak{e} = \Theta \mathfrak{e}$.

Theorem 3. Let Γ, Θ be weakly compatible contractive self-mappings on \mathfrak{E} . Let $\xi : \mathfrak{E} \to \mathbb{R}^+$ be a lower semicontinuous function such that

$$\xi \mathfrak{e} \le \mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \Theta \mathfrak{e}) \tag{2}$$

for all $\mathfrak{e} \in \mathfrak{E}$ satisfying

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma\mathfrak{a}) \leq \xi(\Theta\mathfrak{e}) - \xi(\Gamma\mathfrak{e}) + \xi(\Theta\mathfrak{a}) - \xi(\Gamma\mathfrak{a})$$
(3)

for all $\mathfrak{e}, \mathfrak{a} \in \mathfrak{E}$ with $\mathfrak{e} \neq \mathfrak{a}$. Then Γ and Θ have a unique common fixed point.

Proof. The proof will be given in two parts:

(Existence) First, we show the existence of common fixed point. Suppose

$$\Gamma \mathfrak{e}_{n-1} = \Theta \mathfrak{e}_n = \mathfrak{e}_n$$

for each $n \in \mathbb{N}$. Then, by (3) and the above equations, we have

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n},\mathfrak{e}_{n+1}) &= \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}_{n-1},\Gamma\mathfrak{e}_{n}) \leq \xi(\Theta\mathfrak{e}_{n-1}) - \xi(\Gamma\mathfrak{e}_{n-1}) + \xi(\Theta\mathfrak{e}_{n}) - \xi(\Gamma\mathfrak{e}_{n}) \\ &= \xi(\mathfrak{e}_{n-1}) - \xi(\mathfrak{e}_{n}) + \xi(\mathfrak{e}_{n}) - \xi(\mathfrak{e}_{n+1}) \\ &= \xi(\mathfrak{e}_{n-1}) - \xi(\mathfrak{e}_{n+1}) \\ &\leq \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}_{n-1},\Theta\mathfrak{e}_{n-1}) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}_{n+1},\Theta\mathfrak{e}_{n+1}). \end{split}$$

From the last statements, we find

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+2}) \le \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_n) \tag{4}$$

 Γ and Θ are continuous because they are contractive mappings. From the continuity and boundedness from below of ξ , we conclude that \mathfrak{e}_n is decreasing to a point $\mathfrak{e}_0 \in \mathfrak{E}$ and for all $\mathfrak{e} \in \mathfrak{E}$, $\xi(\mathfrak{e}_0) \leq \xi(\mathfrak{e})$.

Since $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a complete $b_v(s)$ -metric space, then $\mathfrak{e}_n \to \mathfrak{e}_0$. On the other hand, there exists a subsequence of $\{\mathfrak{e}_n\}$ such that $\mathfrak{e}_{n_k} \to \mathfrak{e}_0$ for all $n, k \in \mathbb{N}$. Then we obtain

$$\lim_{n,k\to\infty} \mathfrak{e}_{n_k} = \Gamma\left(\lim_{n,k\to\infty} \mathfrak{e}_{n_k}\right) = \Gamma(\mathfrak{e}_0)$$

and

$$\lim_{n\to\infty}\mathfrak{e}_n=\Theta\left(\lim_{n\to\infty}\mathfrak{e}_n\right)=T(\mathfrak{e}_0).$$

Therefore, \mathfrak{e}_0 is a coincidence fixed point of Γ and Θ . Hence, \mathfrak{e}_0 is a common fixed point of Γ and Θ .

(Uniqueness) Assume that $\Gamma(\pi) = \Theta(\pi) = \pi$ for all $\pi \in \mathfrak{E}$ such that $\mathfrak{e}_0 \neq \pi$. From (3), we obtain

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\pi,\mathfrak{e}_{0}) &= d(\Gamma\pi,\Gamma\mathfrak{e}_{0}) \\ &\leq \xi(\Theta\pi) - \xi(\Gamma\pi) + \xi(\Theta\mathfrak{e}_{0}) - \xi(\Gamma\mathfrak{e}_{0}) \\ &\leq \xi(\pi) - \xi(\pi) + \xi(\mathfrak{e}_{0}) - \xi(\mathfrak{e}_{0}) \\ &\leq \mathfrak{b}_{\mathfrak{v}}(\Gamma\pi,\Theta\pi) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\pi,\Theta\pi) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}_{0},\Theta\mathfrak{e}_{0}) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}_{0},\Theta\mathfrak{e}_{0}) \\ &\leq 0. \end{split}$$

It means that $\pi = \mathfrak{e}_0$, that is, this fixed point is unique.

In the following theorem, inspired by Ćirić-type contraction³⁹ and the used technique in Ref. 55, we use the number $C(\mathfrak{e}, \mathfrak{a})$ defined as

$$C(\mathfrak{e},\mathfrak{a}) = \max \left\{ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{a}), \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}), \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a},\Gamma\mathfrak{a}), \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a},\Gamma\mathfrak{e}), \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{a}) \right\}.$$

Theorem 4. Let $\varpi : \mathfrak{E} \to [0,\infty)$ be a lower semicontinuous and bounded below function. If the following condition holds for all $\mathfrak{e}, \mathfrak{a} \in \mathfrak{E}$

 $(b_v(s) - \mathfrak{MRCC})$. Given $\varepsilon > 0$ there exists a $\delta > 0$ such that $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}) > 0$ implies

$$\varepsilon \leq \left[\varpi(\mathfrak{e}) - \varpi(\Gamma\mathfrak{e})\right] C(\mathfrak{e}, \mathfrak{a}) < \varepsilon + \delta \Longrightarrow \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}, \Gamma\mathfrak{a}) < \varepsilon,$$

then given $\mathfrak{e} \in \mathfrak{E}$, the sequence of iterates $\{\Gamma^n \mathfrak{e}\}$ is a Cauchy sequence and $\lim_{n \to \infty} \Gamma^n \mathfrak{e} = \mathfrak{z}$ for some $\mathfrak{z} \in \mathfrak{E}$.

Proof. Using $(b_v(s) - \mathfrak{MRCC})$, we get that if $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}) > 0$ then

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma\mathfrak{a}) < [\varpi(\mathfrak{e}) - \varpi(\Gamma\mathfrak{e})] C(\mathfrak{e},\mathfrak{a}).$$
(5)

Let $\mathfrak{e}_0 \in \mathfrak{E}$ and let us define $\{\mathfrak{e}_n\}$ in \mathfrak{E} by $\mathfrak{e}_n = \Gamma \mathfrak{e}_{n-1}$, that is,

$$\mathfrak{e}_n = \Gamma^n \mathfrak{e}_0.$$

If $\mathfrak{e}_n = \mathfrak{e}_{n+1}$ for some *n* then $\mathfrak{e}_n = \mathfrak{e}_{n+1} = \mathfrak{e}_{n+2} = \cdots$, that is, $\{\mathfrak{e}_n\} = \{\Gamma^n \mathfrak{e}\}$ is a Cauchy sequence and $\mathfrak{e}_n \in Fix(\Gamma)$. Suppose that $\mathfrak{e}_n \neq \mathfrak{e}_{n+1}$ for each *n* and $\mathfrak{e}_n = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1}, \mathfrak{e}_n)$. Using (5), we have

$$\begin{aligned} \mathbf{c}_{n+1} &= \mathbf{b}_{\mathbf{v}}(\mathbf{e}_n, \mathbf{e}_{n+1}) = \mathbf{b}_{\mathbf{v}}(\Gamma \mathbf{e}_{n-1}, \Gamma \mathbf{e}_n) < \left[\overline{\omega}(\mathbf{e}_{n-1}) - \overline{\omega}(\mathbf{e}_n) \right] C(\mathbf{e}_{n-1}, \mathbf{e}_n) \\ &= \left[\overline{\omega}(\mathbf{e}_{n-1}) - \overline{\omega}(\mathbf{e}_n) \right] \max \left\{ \mathbf{b}_{\mathbf{v}}(\mathbf{e}_{n-1}, \mathbf{e}_n), \mathbf{b}_{\mathbf{v}}(\mathbf{e}_n, \mathbf{e}_{n+1}), \mathbf{b}_{\mathbf{v}}(\mathbf{e}_{n-1}, \mathbf{e}_{n+1}) \right\} \\ &= \left[\overline{\omega}(\mathbf{e}_{n-1}) - \overline{\omega}(\mathbf{e}_n) \right] \beta_N. \end{aligned}$$
(6)

Case 1: If $\beta_N = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1}, \mathfrak{e}_n)$, then using (6), we obtain

$$\mathfrak{c}_{n+1} = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+1}) < [\varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_n)] \mathfrak{c}_n$$

and so

$$0 < \frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n} < \varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_n),$$

for each $n \in \mathbb{N}$. So, $\{\varpi(\mathfrak{e}_n)\}$ is nonincreasing and positive hence it converges to some $t \geq 0$. For all $n \in \mathbb{N}$, we obtain

$$\sum_{m=1}^{n} \frac{\mathfrak{c}_{m+1}}{\mathfrak{c}_{m}} < \sum_{m=1}^{n} \left[\varpi(\mathfrak{e}_{m-1}) - \varpi(\mathfrak{e}_{m}) \right]$$
$$= \varpi(\mathfrak{e}_{0}) - \varpi(\mathfrak{e}_{n}) \to \varpi(\mathfrak{e}_{0}) - t < \infty \text{ as } n \to \infty$$

and

$$\sum_{m=1}^{n} \frac{\mathfrak{c}_{m+1}}{\mathfrak{c}_m} < \infty \Longrightarrow \lim_{n \to \infty} \frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n} = 0.$$

Therefore, for $\beta \in (0, 1)$, there is $n_0 \in \mathbb{N}$ such that

$$\frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n} \leq \beta \text{ for all } n \geq n_0$$

and we get

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+1}) \leq \beta \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_n) \text{ for all } n \geq n_0.$$

Case 2: If $\beta_N = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+1})$, then using (6), we have

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+1}) < [\varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_n)] \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+1})$$

By a similar approach used in Case 1, $\{\varpi(\mathfrak{e}_n)\}\$ is a positive and nonincreasing sequence and so it converges to some $t \ge 0$. Since $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+1}) > 0$, we get

$$1 < \varpi(\mathbf{e}_{n-1}) - \varpi(\mathbf{e}_n) \to 0 \text{ as } n \to \infty,$$

a contradiction.

Case 3: If $\beta_N = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1}, \mathfrak{e}_{n+1})$, then using (6), we have

$$\begin{split} \mathfrak{c}_{n+1} &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n},\mathfrak{e}_{n+1}) < \left[\varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_{n})\right] \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_{n+1}) \\ &\leq \left[\varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_{n})\right] \left(\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_{n}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n},\mathfrak{e}_{n+1})\right) \\ &= \left[\varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_{n})\right] \left(\mathfrak{c}_{n} + \mathfrak{c}_{n+1}\right) \end{split}$$

and so

$$0 < \frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n + \mathfrak{c}_{n+1}} < \varpi(\mathfrak{e}_{n-1}) - \varpi(\mathfrak{e}_n).$$

 $\{\varpi(\mathfrak{e}_n)\}\$ converges to some $t \ge 0$ since it is a positive and nonincreasing sequence. For all $n \in \mathbb{N}$, we obtain

$$\sum_{m=1}^{n} \frac{\mathfrak{c}_{m+1}}{\mathfrak{c}_m + \mathfrak{c}_{m+1}} < \sum_{m=1}^{n} \left[\varpi(\mathfrak{e}_{m-1}) - \varpi(\mathfrak{e}_m) \right]$$
$$= \varpi(\mathfrak{e}_0) - \varpi(\mathfrak{e}_n) \to \varpi(\mathfrak{e}_0) - t < \infty \text{ as } n \to \infty$$

and

$$\sum_{m=1}^{n} \frac{\mathfrak{c}_{m+1}}{\mathfrak{c}_m + \mathfrak{c}_{m+1}} < \infty \Longrightarrow \lim_{n \to \infty} \frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n + \mathfrak{c}_{n+1}} = 0.$$

Hence, for $\beta' \in (0, \frac{1}{2})$, there is $n_0 \in \mathbb{N}$ such that

$$\frac{\mathfrak{c}_{n+1}}{\mathfrak{c}_n + \mathfrak{c}_{n+1}} \le \beta' \text{ for all } n \ge n_0$$

and we have

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n,\mathfrak{e}_{n+1}) \leq \beta \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n-1},\mathfrak{e}_n)$$
 for all $n \geq n_0$,

where $\beta = \frac{\beta'}{1-\beta'}$.

Now, we show that $\{\mathfrak{e}_n\}$ is a Cauchy sequence and $\{\mathfrak{e}_n\}$ converges to some $\mathfrak{z} \in \mathfrak{E}$. By the above cases, we say that the sequence $\{\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_n, \mathfrak{e}_{n+1})\}$ is bounded below and nonincreasing. Hence, it converges to some $\mathfrak{e} \geq 0$. Since $\beta < 1$, we easily prove $\mathfrak{e} = 0$. For each $n_1, n_2 \in \mathbb{N}$ $(n_1 > n_2)$, we get

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n_1},\mathfrak{e}_{n_2}) \leq \sum_{m=n_2}^{n_1-1} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_m,\mathfrak{e}_{m+1}) \leq \frac{\beta^{n_2}}{1-\beta} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0,\mathfrak{e}_1),$$

that is,

$$\lim_{n\to\infty}\sup\left\{\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_{n_1},\mathfrak{e}_{n_2}):n_1>n_2\right\}=0.$$

Consequently, $\{\mathfrak{e}_n\}$ is Cauchy and there is $\mathfrak{z} \in \mathfrak{E}$ such that $\{\mathfrak{e}_n\} \to \mathfrak{z}$ since $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a complete $b_v(s)$ -metric space.

Recently, some fixed-point results have been obtained using the notion of k-continuity with different approaches (for example, see Refs. 56, 57 and the references therein).

A self-mapping $\Gamma : \mathfrak{E} \to \mathfrak{E}$ is called $k_{b_v(s)}$ -continuous, $k = 1, 2, 3, \ldots$, if $\Gamma^k \mathfrak{e}_n \to \Gamma \mathfrak{t}$ whenever $\{\mathfrak{e}_n\}$ is a sequence in \mathfrak{E} such that $\Gamma^{k-1}\mathfrak{e}_n \to \mathfrak{t}$.

Theorem 5. Let $\Gamma : \mathfrak{E} \to \mathfrak{E}$ be a mapping satisfying $(b_v(s) - \mathfrak{MRCC})$ given in Theorem 4. If Γ is $k_{b_v(s)}$ -continuous, then Γ has a fixed point \mathfrak{z} in \mathfrak{E} , that is, $\mathfrak{z} \in Fix(\Gamma)$.

Proof. Let $\mathfrak{e}_0 \in \mathfrak{E}$ and let us define a sequence $\{\mathfrak{e}_n\}$ in \mathfrak{E} by $\mathfrak{e}_n = \Gamma \mathfrak{e}_{n-1}$, that is,

$$\mathfrak{e}_n = \Gamma^n \mathfrak{e}_0$$

Using Theorem 4, we say that $\{\mathfrak{e}_n\}$ is Cauchy. So, there is a point $\mathfrak{z} \in \mathfrak{E}$ such that $\{\mathfrak{e}_n\} \to \mathfrak{z}$ since $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a complete metric space. Also, we have $\Gamma^p \mathfrak{e}_n \to \mathfrak{z}$ for each $p \geq 1$.

Let Γ be a $k_{b_v(s)}$ -continuous self-mapping. $k_{b_v(s)}$ -continuity of Γ implies that $\Gamma^k \mathfrak{e}_n \to \Gamma \mathfrak{z}$ since $\Gamma^{k-1} \mathfrak{e}_n \to \mathfrak{z}$ and so we get $\Gamma \mathfrak{z} = \mathfrak{z}$ as $\Gamma^k \mathfrak{e}_n \to \mathfrak{z}$ from the uniqueness of the limit point. Therefore, $\mathfrak{z} \in Fix(\Gamma)$.

Now, we present new fixed-figure results on a $b_v(s)$ -metric space. To do this, we are inspired by the Caristi-type inequality.²⁰ Let us give the following notions such as a circle and a disc:

Definition 7. Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_v(s)$ -metric space, $r \geq 0$ and $\mathfrak{e}_0 \in \mathfrak{E}$. Then the circle $\mathfrak{C}_{\mathfrak{e}_0, r}^{b_v(s)}$ is defined by

$$\mathfrak{C}_{\mathfrak{e}_{0},r}^{b_{v}(s)} = \{ \mathfrak{e} \in \mathfrak{E} : \mathfrak{b}_{\mathfrak{v}}\left(\mathfrak{e},\mathfrak{e}_{0}\right) = r \}$$

and the disc $\mathfrak{D}_{\mathfrak{c}_0,r}^{b_v(s)}$ is defined by

$$\mathfrak{D}_{\mathfrak{e}_{0},r}^{b_{\mathfrak{v}}(s)} = \left\{ \mathfrak{e} \in \mathfrak{E} : \mathfrak{b}_{\mathfrak{v}}\left(\mathfrak{e},\mathfrak{e}_{0}\right) \leq r \right\}.$$

Example 2. Let $\mathfrak{E} = \mathbb{N}$ and the function $\mathfrak{b}_{\mathfrak{v}} : \mathbb{N} \times \mathbb{N} \to [0, \infty)$ be defined as

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{a}) = \begin{cases} 0; & \mathfrak{e} = \mathfrak{a} \\ 1; & \mathfrak{e} \text{ or } \mathfrak{a} \notin \{1,2\} \text{ and } \mathfrak{e} \neq \mathfrak{a}, \\ 10; & \mathfrak{e}, \mathfrak{a} \in \{1,2\} \text{ and } \mathfrak{e} \neq \mathfrak{a} \end{cases}$$

for all $\mathfrak{e}, \mathfrak{a} \in \mathbb{N}$. Then $(\mathbb{N}, \mathfrak{b}_{\mathfrak{v}})$ is a $b_v(s)$ -metric space with v = 8 and $s = \frac{10}{9}$ (see Ref. 58). Then we get

$$\begin{split} \mathfrak{C}_{1,1}^{b_v(s)} &= \{ \mathfrak{e} \in \mathbb{N} : \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, 1) = 1 \} = \mathbb{N} - \{ 1, 2 \}, \\ \mathfrak{D}_{1,1}^{b_v(s)} &= \{ \mathfrak{e} \in \mathbb{N} : \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, 1) \leq 1 \} = \mathbb{N} - \{ 2 \}, \\ \mathfrak{C}_{1,10}^{b_v(s)} &= \{ \mathfrak{e} \in \mathbb{N} : \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, 1) = 10 \} = \{ 2 \} \end{split}$$

and

$$\mathfrak{D}_{1,10}^{b_{v}(s)} = \{ \mathfrak{e} \in \mathbb{N} : \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, 1) \leq 10 \} = \mathbb{N}.$$

What is the notion of a fixed figure on a $b_v(s)$ -metric space?

Definition 8. Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_v(s)$ -metric space, \mathcal{F} a figure and $\Gamma : \mathfrak{E} \to \mathfrak{E}$ a self-mapping. If $\mathcal{F} \subseteq Fix(\Gamma)$ then \mathcal{F} is called as the fixed figure of Γ .

Especially, if $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$ then $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)}$ is a fixed circle of Γ and if $\mathfrak{D}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$ then $\mathfrak{D}_{\mathfrak{c}_0,r}^{b_v(s)}$ is a fixed disc of Γ .

Since we obtain some existence results for fixed circle of a selfmapping on $b_v(s)$ -metric spaces, we define the function $\varsigma : \mathfrak{E} \to [0, \infty)$ as

$$\varsigma(\mathfrak{e}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{e}_0), \tag{7}$$

for all $\mathfrak{e} \in \mathfrak{E}$.

From now on, we assume that $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ is a $b_v(s)$ -metric space, $\mathfrak{C}_{\mathfrak{e}_0, r}^{b_v(s)}$ is any circle on \mathfrak{E} and the function $\varsigma : \mathfrak{E} \to [0, \infty)$ is defined as in (7).

Theorem 6. If there is $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfying

$$(C_{b_v(s)}1)$$
 $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \varsigma(\mathfrak{e}) - \varsigma(\Gamma\mathfrak{e})$

and

$$(C_{b_v(s)}2)$$
 $\mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) \ge r,$

for each $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r}$, then $\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$. Then we prove $\mathfrak{e} \in Fix(\Gamma)$. By $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$, we get

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &\leq \varsigma(\mathfrak{e}) - \varsigma(\Gamma\mathfrak{e}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) \\ &= r - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) \leq r - r = 0 \end{split}$$

and so $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})=0$, that is, $\mathfrak{e}\in Fix(\Gamma)$. Consequently, we get

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma).$$

Example 3. Let (\mathfrak{E}, d) be a $b_v(s)$ -metric space and c a constant such that $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{c}, \mathfrak{e}_0) > r$ with any circle $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0, r}$. Let us identify $\Gamma : \mathfrak{E} \to \mathfrak{E}$ as

$$\Gamma \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0, r} \\ \mathfrak{c}, & \text{otherwise}, \end{cases}$$

for all $\mathfrak{e} \in \mathfrak{E}$. Then the conditions $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ are satisfied and so $\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma)$. Indeed, for $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r}$, we get

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}) = 0 \leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) \\ &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(T\mathfrak{e},\mathfrak{e}_0) = \varsigma(\mathfrak{e}) - \varsigma(\Gamma\mathfrak{e}) \end{split}$$

and

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \mathfrak{e}_0) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{e}_0) = r.$$

Theorem 7. If $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfies $(C_{b_v(s)}1)$ and

$$(C_{b_{v}(s)}3)$$
 $h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \geq r,$

for each $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{x_0,r}$ and some $h \in [0,1)$, then $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$. Then we prove $\mathfrak{e} \in Fix(\Gamma)$. On the contrary, suppose $\mathfrak{e} \notin Fix(\Gamma)$. By $(C_{b_v(s)}1)$ and $(C_{b_v(s)}3)$, we find

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &\leq \varsigma(\mathfrak{e}) - \varsigma(\Gamma\mathfrak{e}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_{0}) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \\ &= r - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \\ &\leq h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) - \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \\ &= h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) < \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}), \end{split}$$

a contradiction with $h \in [0, 1)$. Then we get $\mathfrak{e} \in Fix(\Gamma)$ and so

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma).$$

Example 4. Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_v(s)$ -metric space and c a constant such that $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{c}, \mathfrak{e}_0) < r$ with any circle $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0, r}$. Let us identify $\Gamma : \mathfrak{E} \to \mathfrak{E}$ as

$$\Gamma \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0, r} \\ \mathfrak{c}, & \text{otherwise}, \end{cases}$$

for all $\mathfrak{e} \in \mathfrak{E}$. Then the conditions $(C_{b_v(s)}1)$ and $(C_{b_v(s)}3)$ are satisfied and so $\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma)$. Indeed, for $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r}$, we get

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}) = 0 \leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) \\ &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(T\mathfrak{e},\mathfrak{e}_0) = \varsigma(\mathfrak{e}) - \varsigma(\Gamma\mathfrak{e}) \end{split}$$

and

$$h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) = h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) = r.$$

In the following fixed-circle theorems, we modify the Caristi inequality using the function $\varsigma : \mathfrak{E} \to [0, \infty)$ defined as in (7).

Theorem 8. If there is $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfying

$$(C_{b_{v}(s)}4) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \varsigma(\mathfrak{e}) + \varsigma(\Gamma\mathfrak{e}) - 2r$$

and

$$(C_{b_v(s)}5) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) \leq r,$$

for each $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r}$, then $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$. Then we prove $\mathfrak{e} \in Fix(\Gamma)$. On the contrary, suppose $\mathfrak{e} \notin Fix(\Gamma)$. By $(C_{b_v(s)}4)$ and $(C_{b_v(s)}5)$, we find

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &\leq \varsigma(\mathfrak{e}) + \varsigma(\Gamma\mathfrak{e}) - 2r \\ &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_{0}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) - 2r \\ &= \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) - r \\ &\leq \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) - [\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \\ &= -\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}), \end{split}$$

a contradiction. Then we get $\mathfrak{e} \in Fix(\Gamma)$ and so

$$\mathfrak{C}^{b_{v}(s)}_{\mathfrak{c}_{0},r} \subseteq Fix(\Gamma).$$

Example 5. Let us consider Example 3. Then the conditions $(C_{b_v(s)}4)$ and $(C_{b_v(s)}5)$ are satisfied and so $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$.

Theorem 9. If there is $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfying $(C_{b_n(s)}4)$ and

$$(C_{b_v(s)}6) \quad \mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \mathfrak{e}_0) \leq r,$$

for each $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{x_0,r}$, then $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)}$. Then we prove $\mathfrak{e} \in Fix(\Gamma)$. On the contrary, suppose $\mathfrak{e} \notin Fix(\Gamma)$. By $(C_{b_v(s)}4)$ and $(C_{b_v(s)}5)$, we find

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &\leq \varsigma(\mathfrak{e}) + \varsigma(\Gamma\mathfrak{e}) - 2r \\ &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) + \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) - 2r \\ &= \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) - r \\ &\leq r - r = 0, \end{split}$$

a contradiction. Then we get $\mathfrak{e} \in Fix(\Gamma)$ and so

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r} \subseteq Fix(\Gamma).$$

Example 6. Let us consider Example 4. Then the conditions $(C_{b_v(s)}4)$ and $(C_{b_v(s)}6)$ are satisfied and so $\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma)$.

Theorem 10. If there is $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfying

$$(C_{b_v(s)}7) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \varsigma(\mathfrak{e}) - r$$

or

$$(C_{b_v(s)}8) \ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \varsigma(\Gamma\mathfrak{e}) - r$$

and

$$(C_{b_{v}(s)}9) \ \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) \leq r + h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}),$$

for each $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r}$ and some $h \in [0,1)$, then $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)}$. Then we prove $\mathfrak{e} \in Fix(\Gamma)$. On the contrary, suppose $\mathfrak{e} \notin Fix(\Gamma)$. If $(C_{b_v(s)}7)$ holds, we get

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) \leq \varsigma(\mathfrak{e}) - r = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\mathfrak{e}_0) - r = r - r = 0,$$

a contradiction. Then we get $\mathfrak{e} \in Fix(\Gamma)$ and so

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma).$$

On the other hand, if $(C_{b_v(s)}7)$ holds, using $(C_{b_v(s)}9)$, we get

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) &\leq \varsigma(\Gamma\mathfrak{e}) - r = \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_0) - r \\ &\leq r + h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) - r \\ &= h\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}), \end{split}$$

a contradiction with $h \in [0, 1)$. Then we get $\mathfrak{e} \in Fix(\Gamma)$ and so

$$\mathfrak{C}^{b_{\upsilon}(s)}_{\mathfrak{e}_{0},r} \subseteq Fix(\Gamma).$$

Example 7. Let us consider Examples 3 and 4. Then the conditions $(C_{b_v(s)}7)$, $(C_{b_v(s)}8)$ and $(C_{b_v(s)}9)$ are satisfied and so $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq$ $Fix(\Gamma).$

To obtain new fixed-circle results, the used auxiliary functions can be changed. For example, let us define the mapping $\eta: [0,\infty) \to \mathbb{R}$ as

$$\eta(\mathbf{e}) = \begin{cases} \mathbf{e} - r; & \mathbf{e} > 0\\ 0; & \mathbf{e} = 0, \end{cases}$$
(8)

for all $\mathfrak{e} \in [0,\infty)$ and any $\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r}$.

Theorem 11. If there is $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfying

- $\begin{array}{ll} (\mathrm{i}) & \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\mathfrak{e}_{0}) = r \ for \ each \ \mathfrak{e} \in \mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{0},r}, \\ (\mathrm{ii}) & \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma\mathfrak{a}) > r \ for \ each \ \mathfrak{e}, \mathfrak{a} \in \mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{0},r} \ and \ \mathfrak{e} \neq \mathfrak{a}, \end{array}$
- (iii) $\mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \Gamma \mathfrak{a}) \leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{a}) \eta(\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}))$ for each $\mathfrak{e}, \mathfrak{a} \in \mathfrak{C}_{\mathfrak{e}_{0}, r}^{b_{v}(s)}$,

then $\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_0,r} \subseteq Fix(\Gamma).$

Proof. Let $\mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r}$. Using (i), we get $\Gamma \mathfrak{e} \in \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_0,r}$. Now, we prove $\mathfrak{e} \in Fix(\Gamma)$. On the contrary, let $\mathfrak{e} \notin Fix(\Gamma)$. Using (ii), we have

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma^{2}\mathfrak{e}) > r \tag{9}$$

and using (iii), we get

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma^{2}\mathfrak{e}) \leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) - \eta(\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})) = \Gamma\mathfrak{e} - \Gamma\mathfrak{e} + r = r,$$

a contradiction with (9). Then $\mathfrak{e} \in Fix(\Gamma)$ and so $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$. \Box

The identity mapping $I_{\mathfrak{E}} : \mathfrak{E} \to \mathfrak{E}$, defined as $I_{\mathfrak{E}}(\mathfrak{e}) = \mathfrak{e}, \mathfrak{e} \in \mathfrak{E}$, fixes every circle on a $b_v(s)$ -metric space. So, as a natural problem, is there a new theorem which excludes the possibility of identity mapping $I_{\mathfrak{G}}$?

Theorem 12. Γ satisfies the condition

$$(I_{b_{v}(s)}) \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}) + \sum_{i=1}^{v-1} \mathfrak{b}_{\mathfrak{v}}(u_{i}, u_{i+1}) < \varsigma(\mathfrak{e}) - \varsigma(u_{v}),$$

for all $\mathfrak{e} \in \mathfrak{E}$ and $u_i \in \mathfrak{E}$ such that all u_i are different for $i \in \{1, 2, \dots, v\}$ if and only if $\Gamma = I_{\mathfrak{E}}$.

Proof. Let $\mathfrak{e} \in \mathfrak{E}$ and $\mathfrak{e} \notin Fix(\Gamma)$. By $(I_{b_v(s)})$ and (B3), we obtain

$$\begin{split} \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \sum_{i=1}^{v-1} \mathfrak{b}_{\mathfrak{v}}(u_i, u_{i+1}) < \varsigma(\mathfrak{e}) - \varsigma(u_v) &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(u_v, \mathfrak{e}_0) \\ &\leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, u_1) + \mathfrak{b}_{\mathfrak{v}}(u_1, u_2) + \cdots \\ &+ \mathfrak{b}_{\mathfrak{v}}(u_v, \mathfrak{e}_0) - \mathfrak{b}_{\mathfrak{v}}(u_v, \mathfrak{e}_0) \\ &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, u_1) + \sum_{i=1}^{v-1} \mathfrak{b}_{\mathfrak{v}}(u_i, u_{i+1}) \end{split}$$

and so if we take $\Gamma \mathfrak{e} = u_1$, we have

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) < \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}),$$

a contradiction. Hence, we get $\mathfrak{e} \in Fix(\Gamma)$ for all $\mathfrak{e} \in \mathfrak{E}$, that is, $\Gamma = I_{\mathfrak{E}}$.

The converse statement is clear.

In the following proposition, we see that the number of fixed circles is not to be one.

Proposition 1. Let $\mathfrak{C}_{\mathfrak{c}_1,r_1}^{b_v(s)}, \ldots, \mathfrak{C}_{\mathfrak{c}_n,r_n}^{b_v(s)}$ any given circles on \mathfrak{E} . Then there is at least one $\Gamma : \mathfrak{E} \to \mathfrak{E}$ such that

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{c}_1,r_1} \subseteq Fix(\Gamma), \ldots, \mathfrak{C}^{b_v(s)}_{\mathfrak{c}_n,r_n} \subseteq Fix(\Gamma).$$

Proof. Assume that \mathfrak{c} is constant such that

$$\mathfrak{b}_{\mathfrak{v}}(\mathfrak{c},\mathfrak{e}_i) \neq r \quad (1 \leq i \leq n).$$

Let us identify $\Gamma : \mathfrak{E} \to \mathfrak{E}$ by

$$\Gamma \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \in \bigcup_{i=1}^{n} \mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{i}, r_{i}} \\ \mathfrak{c}, & \text{otherwise}, \end{cases}$$

for all $\mathfrak{e} \in \mathfrak{E}$ and the mappings $\xi_i : \mathfrak{E} \to [0, \infty)$ as

$$\xi_i(\mathfrak{e}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{e}_i) \ (1 \le i \le n).$$

The conditions $(C_{b_v(s)}1)$, $(C_{b_v(s)}2)$ and $(C_{b_v(s)}3)$ are satisfied by Γ . Consequently, we get

$$\mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{1},r_{1}} \subseteq Fix(\Gamma), \dots, \mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{n},r_{n}} \subseteq Fix(\Gamma).$$

By Proposition 1, searching for conditions of uniqueness makes sense for a fixed circle. To do this, we use the Banach-type⁵⁹ and \acute{C} irić-type³⁹ contractions.

Theorem 13. Let $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfy $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ (or $(C_{b_v(s)}3)$). If it satisfies the condition

 $(B) \quad \mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \Gamma \mathfrak{a}) \leq h \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{a}),$

for all $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$, $\mathfrak{a} \in \mathfrak{E} - \mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$ and some $h \in [0,1)$, then $\mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$ is unique such that $\mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{C}_{\mathfrak{c}_1,r_1}^{b_v(s)}$, $\mathfrak{C}_{\mathfrak{c}_2,r_2}^{b_v(s)}$ be any two circles and $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfies $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ (or $(C_{b_v(s)}3)$) for each $\mathfrak{C}_{\mathfrak{c}_1,r_1}^{b_v(s)}$ and $\mathfrak{C}_{\mathfrak{c}_2,r_2}^{b_v(s)}$. Hence, we have

$$\mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{1},r_{1}} \subseteq Fix(\Gamma) \text{ and } \mathfrak{C}^{b_{v}(s)}_{\mathfrak{e}_{2},r_{2}} \subseteq Fix(\Gamma).$$

Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_1,r_1}^{b_v(s)}$ and $\mathfrak{a} \in \mathfrak{C}_{\mathfrak{e}_2,r_2}^{b_v(s)}$ be any arbitrary points such that $\mathfrak{e} \neq \mathfrak{a}$. By (B), we get

$$\mathfrak{b}_{\mathfrak{v}}(\Gamma \mathfrak{e}, \Gamma \mathfrak{a}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{a}) \leq h \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{a}) < \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{a}),$$

a contradiction. Consequently, the fixed circle of Γ is unique. \Box

In the following theorem, we use the number $C(\mathfrak{e}, \mathfrak{a})$.

Theorem 14. Let $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfy $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ (or $(C_{b_v(s)}3)$). If it satisfies the condition

$$(C) \quad \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e}, \Gamma\mathfrak{a}) \leq hC(\mathfrak{e}, \mathfrak{a}),$$

for all $\mathfrak{e} \in \mathfrak{C}_{x_0,r}^{b_v(s)}$, $\mathfrak{a} \in \mathfrak{E} - \mathfrak{C}_{x_0,r}^{b_v(s)}$ and some $h \in [0,1)$, then $\mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)}$ is unique such that $\mathfrak{C}_{\mathfrak{e}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$.

Proof. Let $\mathfrak{C}_{\mathfrak{e}_1,r_1}^{b_v(s)}$, $\mathfrak{C}_{\mathfrak{e}_2,r_2}^{b_v(s)}$ be any two circles and $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfies $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ (or $(C_{b_v(s)}3)$) for each $\mathfrak{C}_{\mathfrak{e}_1,r_1}^{b_v(s)}$ and $\mathfrak{C}_{\mathfrak{e}_2,r_2}^{b_v(s)}$. Hence, we have

$$\mathfrak{C}^{b_v(s)}_{\mathfrak{e}_1,r_1} \subseteq Fix(\Gamma) \text{ and } \mathfrak{C}^{b_v(s)}_{\mathfrak{e}_2,r_2} \subseteq Fix(\Gamma).$$

Let $\mathfrak{e} \in \mathfrak{C}_{\mathfrak{e}_1,r_1}^{b_v(s)}$ and $\mathfrak{a} \in \mathfrak{C}_{\mathfrak{e}_2,r_2}^{b_v(s)}$ be any arbitrary points such that $\mathfrak{e} \neq \mathfrak{a}$. By (C), we get

$$\mathfrak{b}_\mathfrak{v}(\Gamma\mathfrak{e},\Gamma\mathfrak{a}) = \mathfrak{b}_\mathfrak{v}(\mathfrak{e},\mathfrak{a}) \leq hC(\mathfrak{e},\mathfrak{a}) < C(\mathfrak{e},\mathfrak{a}) = \mathfrak{b}_\mathfrak{v}(\mathfrak{e},\mathfrak{a}),$$

a contradiction. Consequently, the fixed circle of Γ is unique.

Now, we define the following number ρ as

$$\rho = \inf \left\{ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}) : \mathfrak{e} \notin Fix(\Gamma), \ \mathfrak{e} \in \mathfrak{E} \right\}$$
(10)

and the function $\zeta : \mathfrak{E} \to [0, \infty)$ as

$$\zeta(\mathfrak{e}) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}), \tag{11}$$

for all $\mathfrak{e} \in \mathfrak{E}$ since we acquire a new fixed-figure theorem. In the following theorems, we use the Meir–Keeler-type technique.

Theorem 15. If there is $\mathfrak{e}_0 \in \mathfrak{E}$ such that

(a) $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0, \Gamma \mathfrak{e}) \leq \rho$ and $0 \leq \zeta(\mathfrak{e}) \leq 1$ for all $\mathfrak{e} \in \mathfrak{D}_{\mathfrak{e}_0, \rho}^{b_v(s)}$,

(b) For all $\mathfrak{e} \in \mathfrak{E}$,

$$\zeta(\mathfrak{e}) > 0 \Longrightarrow \zeta(\mathfrak{e}) < [\zeta(\mathfrak{e}) - \zeta(\mathfrak{e}_0)] C(\mathfrak{e}, \mathfrak{e}_0),$$

then

(1) $\mathfrak{e}_0 \in Fix(\Gamma),$ (2) $\mathfrak{D}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma),$ (3) $\mathfrak{C}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma).$

Proof. (1) Let $\rho = 0$. Then we get $\mathfrak{D}_{\mathfrak{c}_0,\rho}^{b_v(s)} = \mathfrak{C}_{\mathfrak{c}_0,\rho}^{b_v(s)} = {\mathfrak{c}_0}$. If $\zeta(\mathfrak{e}_0) > 0$, by (b), we find

$$\zeta(\mathfrak{e}_0) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0, \Gamma \mathfrak{e}_0) < [\zeta(\mathfrak{e}_0) - \zeta(\mathfrak{e}_0)] C(\mathfrak{e}_0, \mathfrak{e}_0) = 0,$$

a contradiction. So, it should be $\zeta(\mathfrak{e}_0) = 0$, that is,

$$\mathfrak{e}_0 \in Fix(\Gamma).$$

(2) If $\rho = 0$, then using the condition (1), we have

$$\mathfrak{e}_0 \in Fix(\Gamma)$$
, that is, $\mathfrak{D}^{b_v(s)}_{\mathfrak{e}_0,\rho} \subseteq Fix(\Gamma)$.

Let $\rho > 0$ and $\mathfrak{e} \in D^{b_v(s)}_{\mathfrak{e}_0,\rho}$. If $\zeta(\mathfrak{e}) > 0$, by (a), (b), (1) and the definition of ρ , we find

$$\begin{split} \zeta(\mathfrak{e}) &= \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) < \left[\zeta(\mathfrak{e}) - \zeta(\mathfrak{e}_0)\right] C(\mathfrak{e},\mathfrak{e}_0) = \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})C(\mathfrak{e},\mathfrak{e}_0) \\ &= \left[\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})\right]^2, \end{split}$$

a contradiction. Hence, it should be $\zeta(\mathfrak{e}) = 0$ and so we have

$$\mathfrak{e} \in Fix(\Gamma)$$
, that is, $\mathfrak{D}^{b_v(s)}_{\mathfrak{e}_0,\rho} \subseteq Fix(\Gamma)$.

(3) This is a natural consequence of the condition (2).

Example 8. Let (\mathfrak{E}, d) be a $b_v(s)$ -metric space and $\Gamma : \mathfrak{E} \to \mathfrak{E}$ defined as

$$\Gamma \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \in \mathfrak{D}_{\mathfrak{e}_0,\rho}^{b_v(s)} \\ \mathfrak{e}_0, & \text{otherwise}, \end{cases}$$

for all $\mathfrak{e} \in \mathfrak{E}$ such that $0 \leq \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \mathfrak{e}_0) \leq 1$. Then (a) and (b) of Theorem 15 are satisfied and so $\mathfrak{D}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma)$. As a consequence, we have $\mathfrak{e}_0 \in Fix(\Gamma)$ and $\mathfrak{C}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma)$.

To obtain a new fixed-disc result, we need the following definition.

Definition 9 (Ref. 60). Let \mathbb{F} be the family of all functions $F : (0, \infty) \to \mathbb{R}$ such that

- (F_1) F is strictly increasing,
- (F_2) For every sequence $\{\alpha_n\}$ in $(0,\infty)$ the following holds:

$$\lim_{n \to \infty} \alpha_n = 0 \quad \Leftrightarrow \quad \lim_{n \to \infty} F(\alpha_n) = -\infty,$$

(F₃) There is $k \in (0, 1)$ such that $\lim_{\alpha \to 0^+} \alpha^k F(\alpha) = 0$.

Some examples of functions that satisfy the above conditions are $F(\mathfrak{e}) = \ln(\mathfrak{e}), F(\mathfrak{e}) = \ln(\mathfrak{e}) + \mathfrak{e}, F(\mathfrak{e}) = -\frac{1}{\sqrt{\mathfrak{e}}}$ and $F(\mathfrak{e}) = \ln(\mathfrak{e}^2 + \mathfrak{e})$ (see Ref. 60 for more details).

Theorem 16. If there are $\mathfrak{e}_0 \in \mathfrak{E}$, t > 0 and $F \in \mathbb{F}$ such that

- (a) $\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}_0, \Gamma \mathfrak{e}) \leq \rho$ and $0 \leq \zeta(\mathfrak{e}) \leq 1$ for all $\mathfrak{e} \in \mathfrak{D}_{\mathfrak{e}_0, \rho}^{b_v(s)}$,
- (b) For all $\mathfrak{e} \in \mathfrak{E}$,

 $\zeta(\mathfrak{e}) > 0 \Longrightarrow t + F\left(\zeta(\mathfrak{e})\right) \leq F\left(\left[\zeta(\mathfrak{e}) - \zeta(\mathfrak{e}_0)\right]C(\mathfrak{e},\mathfrak{e}_0)\right),$

then

(1) $\mathfrak{e}_0 \in Fix(\Gamma),$ (2) $\mathfrak{D}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma),$ (3) $\mathfrak{C}_{\mathfrak{e}_0,\rho}^{b_v(s)} \subseteq Fix(\Gamma).$

Proof. (1) Let $\rho = 0$. Then it is an immediate result of the condition (b).

(2) If $\rho = 0$, by (1), we have

$$\mathfrak{e}_0 \in Fix(\Gamma)$$
, that is, $\mathfrak{D}^{b_v(s)}_{\mathfrak{e}_0,\rho} \subseteq Fix(\Gamma)$.

Let $\rho > 0$ and $\mathfrak{e} \in D^{b_v(s)}_{\mathfrak{e}_0,\rho}$. If $\zeta(\mathfrak{e}) > 0$, by (a), (b), (1) and the definition of ρ , we find

$$\begin{split} t + F\left(\zeta(\mathfrak{e})\right) &= t + F\left(\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})\right) \\ &\leq F\left(\left[\zeta(\mathfrak{e}) - \zeta(\mathfrak{e}_{0})\right]C(\mathfrak{e},\mathfrak{e}_{0})\right) = F\left(\left[\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})\right]^{2}\right) \\ &\leq F\left(\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e})\right), \end{split}$$

a contradiction with t > 0. So, it should be $\zeta(\mathfrak{e}) = 0$ and so we have

$$\mathfrak{e} \in Fix(\Gamma)$$
, that is, $\mathfrak{D}^{b_{\mathfrak{v}}(s)}_{\mathfrak{e}_0,\rho} \subseteq Fix(\Gamma)$.

(3) This is obvious.

Example 9. Let us consider the $b_v(s)$ -metric space $(\mathbb{N}, \mathfrak{b}_{\mathfrak{v}})$ recalled in Example 2. Let us identify $\Gamma : \mathbb{N} \to \mathbb{N}$ as

$$\Gamma \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \in \mathbb{N} - \{2\} \\ 1, & \mathfrak{e} = 2, \end{cases}$$

for all $\mathfrak{e} \in \mathbb{N}$. Then we get

$$\rho = \inf \left\{ \mathfrak{b}_{\mathfrak{v}}(\mathfrak{e}, \Gamma \mathfrak{e}) : \mathfrak{e} \notin Fix(\Gamma), \, \mathfrak{e} \in \mathbb{N} \right\} = 1$$

and Γ satisfies the conditions (a) and (b) of Theorem 16 with $\mathfrak{e}_0 = 1$, $t = \ln 10$ and $F(\mathfrak{e}) = \ln \mathfrak{e}$. Then we get $1 \in Fix(\Gamma)$, $\mathfrak{D}_{1,1}^{b_v(s)} \subseteq Fix(\Gamma)$ and $\mathfrak{C}_{1,1}^{b_v(s)} \subseteq Fix(\Gamma)$.

Consequently, we attain the following outcomes:

- If we take v = s = 1, then Theorem 6 coincides with Theorem 2.1 given in Ref. 33 on a metric space.
- Taking v = s = 1, we conclude that Theorem 7 coincides with Theorem 2.3 given in Ref. 33 on a metric space.
- Theorems 15 and 16 generalize Theorem 5, Theorem 8, Corollary 2 and Corollary 4 given in Ref. 61.
- Since $b_v(s)$ -metric is the generalization of some metric spaces such as b, rectangular and v-generalized, then the obtained results can be considered in these spaces. Therefore, the obtained fixed-figure results are important in these spaces.
- Theorems 13 and 14 can be considered for Theorems 8–11.
- The contractive conditions used for uniqueness need not be unique. For example, Kannan-type contractions⁶² can be used as follows: Let $(\mathfrak{E}, \mathfrak{b}_{\mathfrak{v}})$ be a $b_{v}(s)$ -metric space, $\mathfrak{C}_{\mathfrak{c}_{0},r}^{b_{v}(s)}$, any circle on \mathfrak{E} and

 $\Gamma : \mathfrak{E} \to \mathfrak{E}$ satisfies $(C_{b_v(s)}1)$ and $(C_{b_v(s)}2)$ (or $(C_{b_v(s)}3)$). If it satisfies the condition

$$(K) \quad \mathfrak{b}_{\mathfrak{v}}(\Gamma\mathfrak{e},\Gamma\mathfrak{a}) \leq h \left[\mathfrak{b}_{\mathfrak{v}}(\mathfrak{e},\Gamma\mathfrak{e}) + \mathfrak{b}_{\mathfrak{v}}(\mathfrak{a},\Gamma\mathfrak{a}) \right],$$

for all $\mathfrak{e} \in \mathfrak{C}_{x_0,r}^{b_v(s)}$, $\mathfrak{a} \in \mathfrak{E} - \mathfrak{C}_{x_0,r}^{b_v(s)}$ and some $h \in [0,1)$, then $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)}$ is unique such that $\mathfrak{C}_{\mathfrak{c}_0,r}^{b_v(s)} \subseteq Fix(\Gamma)$.

• In Theorems 6 and 7, the center of a fixed circle does not have to be fixed. But in Theorems 15 and 16, the center of a fixed circle is a fixed point.

3. An Application to Activation Functions

Activation functions are used in neural network and they are also known transfer functions because they are used to get the output of the node. The main goal of these functions is to determine the output of neural network like yes or no. Activation functions can be divided as follows:

- Linear activation functions,
- Nonlinear activation functions.

There exist a lot of examples of activation functions in the literature (for some examples, see Ref. 63). One of them is Rectified Linear Unit (ReLU) activation function (see Ref. 64). ReLU has been widely used for deep learning applications. This function is defined by

$$ReLU(\mathfrak{e}) = \max\{0, \mathfrak{e}\} = \begin{cases} \mathfrak{e}, & \mathfrak{e} \ge 0\\ 0, & \mathfrak{e} < 0, \end{cases}$$

which performs a threshold operation to each input element. Some types of ReLU are listed below:

• Leakly ReLU (LReLU) is defined by

$$LReLU(\mathfrak{e}) = \alpha \mathfrak{e} + \mathfrak{e} = \begin{cases} \mathfrak{e}, & \mathfrak{e} > 0\\ \alpha \mathfrak{e}, & \mathfrak{e} \leq 0, \end{cases}$$

where $\alpha = 0.01.^{65}$

• Parametric ReLU (PReLU) is defined by

$$PReLU(\mathfrak{e}) = \max\{0, \mathfrak{e}\} + \alpha \min\{0, \mathfrak{e}\} = \begin{cases} \mathfrak{e}, & \mathfrak{e} > 0\\ \alpha \mathfrak{e}, & \mathfrak{e} \le 0, \end{cases}$$

where α is the negative slope controlling parameter.⁶⁶

• Randomized Leaky ReLU (*RReLU*) is defined by

$$RReLU(\mathfrak{e}) = \begin{cases} \mathfrak{e}, & \mathfrak{e} \ge 0\\ \alpha \mathfrak{e}, & \mathfrak{e} < 0, \end{cases}$$

where U(l, u) is a random number sampled from a uniform distribution, $\alpha \sim U(l, u)$, l < u and $l, u \in [0, 1]$.⁶⁷

• S-Shaped Rectified Linear Activation Unit (SReLU) is defined by

$$SReLU\left(\mathfrak{e}\right) = \begin{cases} \sqcup_{l} + \dashv_{l} \left(\mathfrak{e} - \sqcup_{l}\right), & \mathfrak{e} \leq \sqcup_{l} \\ \mathfrak{e}, & \sqcup_{l} < \mathfrak{e} < \sqcup_{r} \\ \sqcup_{r} + \dashv_{r} \left(\mathfrak{e} - \sqcup_{r}\right), & \mathfrak{e} \geq \sqcup_{r}, \end{cases}$$

where \sqcup_l , \dashv_l , \sqcup_r , \dashv_r are parameters.⁶⁸

These types of activation functions have been used in the application of fixed-circle problem with different aspects on metric and some generalized metric spaces. For example, to obtain an application, the real valued discontinuous activation functions and ReLU activation functions were used in metric spaces (see Ref. 47 and the references therein for more examples).

To obtain a new application for activation functions, we use SReLU activation functions. For this purpose, let us take $\mathfrak{E} = \mathbb{N}$, $\sqcup_l = 0$, $\sqcup_r = 2$, $\dashv_l = 3$ and $\dashv_r = 4$. Then we get

$$SReLU(x) = \begin{cases} 3x, & x = 0\\ x, & 0 < x < 2\\ 4x - 6, & x \ge 2 \end{cases}$$

for all $x \in \mathbb{N}$, as seen in Figure 1.



Fig. 1. The constructed activation function: SReLU(x).

If we consider the $b_v(s)$ -metric space $(\mathbb{N}, \mathfrak{b}_v)$ given in Example 2, then *SReLU* satisfies the conditions of Theorem 6 for each $\mathfrak{C}_{2,10}^{b_v(s)} = \{1\}$. Consequently, $\mathfrak{C}_{2,10}^{b_v(s)} \subseteq Fix(SReLU)$. Note that the center of a fixed circle $\mathfrak{C}_{2,10}^{b_v(s)}$ is not fixed by *SReLU*.

4. Conclusion

In this chapter, we prove Caristi-type fixed-point theorems on $b_v(s)$ metric spaces. On the other hand, we investigate some geometric properties of the fixed-point set $Fix(\Gamma)$ of a self-mapping Γ . In this context, we obtain new fixed-circle and fixed-disc results with auxiliary function families. The obtained theoretical results are supported by some illustrative examples. Finally, to show the importance of our results, we give an application to S-shaped rectified linear activation unit. It is hoped that this chapter will inform new nonunique fixedpoint theorems with geometric approaches and will shed light on new research areas on generalized metric spaces.

Acknowledgments

The authors express their gratitude to the anonymous referees for their helpful suggestions and corrections.

References

- 1. P. Debnath, B. S. Choudhury, and M. Neog, Fixed set of set valued mappings with set valued domain in terms of start set on a metric space with a graph, *Fixed Point Theory Appl.* **2017**, 5 (2016).
- 2. P. Debnath and M. de la Sen, Fixed points of eventually δ -restrictive and $\delta(\varepsilon)$ -restrictive set-valued maps in metric spaces, Symmetry 12(1), 127 (2020).
- P. Debnath and H. M. Srivastava, New extensions of Kannan's and Reich's fixed point theorems for multivalued maps using Wardowski's technique with application to integral equations, *Symmetry* 12(7), 1090 (2020).

- P. Debnath and H. M. Srivastava, Global optimization and common best proximity points for some multivalued contractive pairs of mappings, Axioms. 9(3), 102 (2020).
- P. Debnath, M. Neog, and S. Radenovic, Set valued Reich type g-contractions in a complete metric space with graph, *Rend. Circ. Mat. Palermo Ser. 2.* 69(3), 917–924 (2020).
- P. Debnath, Banach, Kannan, Chatterjea, and Reich-type contractive inequalities for multivalued mappings and their common fixed points, *Math. Methods Appl. Sci.* 45(3), 1587–1596 (2021).
- P. Debnath, Optimization through best proximity points for multivalued f-contractions, *Miskolc Math. Notes* 22(1), 143–151 (2021).
- P. Debnath, Set-valued Meir-keeler, Geraghty and Edelstein type fixed point results in b-metric spaces, *Rend. Circ. Mat. Palermo Ser. 2* 70(3), 1389–1398 (2021).
- P. Debnath, N. Konwar, and S. Radenovic, Metric Fixed Point Theory Applications in Science, Engineering and Behavioural Sciences. Springer, Singapore (2021).
- P. Debnath, H. M. Srivastava, P. Kumam, and B. Hazarika, Fixed Point Theory and Fractional Calculus Recent Advances and Applications. Springer, Singapore (2022).
- 11. M. Neog and P. Debnath, Fixed points of set valued mappings in terms of start point on a metric space endowed with a directed graph, *Mathematics*. **5**(2), 24 (2017).
- M. Neog, P. Debnath, and S. Radenovic, New extension of some common fixed point theorems in complete metric spaces, *Fixed Point Theory* 20(2), 567–580 (2019).
- I. A. Bakhtin, The contraction mapping principle in quasimetric spaces, Funct. Anal. Unianowsk Gos. Ped. Inst. 30, 26–37 (1989).
- H. Aydi, M. F. Bota, E. Karapinar, and Z. D. Mitrovic, A fixed point theorem for set-valued quasi-contractions in b-metric spaces, *Fixed Point Theory Appl.* 88, 1–8 (2012).
- S. Czerwik, Contraction mappings in b-metric spaces, Acta Math. Inform. Univ. Ostraviensis 1, 5–11 (1993).
- S. Czerwik, Nonlinear set-valued contraction mappings in b-metric spaces, Atti Sem. Math. Fis. Univ. Modena. 46, 263–276 (1998).
- A. Gholidahneh, S. Sedghi, O. Ege, Z. D. Mitrovic, and M. de la Sen, The Meir-Keeler type contractions in extended modular b-metric spaces with an application, *AIMS Math.* 6(2), 1781–1799 (2021).
- M. Iqbal, A. Batool, O. Ege, and M. de la Sen, Fixed point of almost contraction in b-metric spaces, J. Math. 3218134, 1–6 (2020).

- M. Iqbal, A. Batool, O. Ege, and M. de la Sen, Fixed point of generalized weak contraction in b-metric spaces, J. Function Spaces 2042162, 1–8 (2021).
- J. Caristi, Fixed point theorems for mappings satisfying inwardness conditions, *Trans. Am. Math. Soc.* 215, 241–251 (1976).
- 21. O. Ege, Complex valued rectangular b-metric spaces and an application to linear equations, J. Nonlinear Sci. Appl. 8(6), 1014–1021 (2015).
- 22. Z. D. Mitrovic and S. Radenovic, The banach and reich contractions in $b_v(s)$ -metric spaces, J. Fixed Point Theory Appl. **19**, 3087–3095 (2017).
- 23. Z. D. Mitrovic, L. K. Dey, and S. Radenovic, A fixed point theorem of Sehgal-Guseman in $b_v(s)$ -metric spaces, An. Univ. Craiova, Math. Comput. Sci. Ser. 47(2), 244–251 (2020).
- 24. M. S. Abdullahi and P. Kumam, Partial $b_v(s)$ -metric spaces and fixed point theorems, J. Fixed Point Theory Appl. **20**(113), 1–13 (2018).
- Z. D. Mitrovic, H. Aydi, Z. Kadelburg, and G. S. Rad, On some rational contractions in b_v(s)-metric spaces, *Rend. Circ. Mat. Palermo 2.* 69, 1193–1203 (2020).
- 26. H. Aydi, D. Bajovic, and Z. D. Mitrovic, Some common fixed point results in partial $b_v(s)$ -metric spaces, *Acta Math. Univ. Comenianae* **89**, 153–160 (2020).
- S. Aleksic, Z. D. Mitrovic, and S. Radenovic, A fixed point theorem of Jungck in b_v(s)-metric spaces, *Period. Math. Hung.* 77, 224–231 (2018).
- F. Dong, P. Ji, and X. Wang, Pata-type fixed point results in b_v(s)metric spaces, Int. J. Anal. Appl. 17, 342–360 (2019).
- 29. I. Karahan and I. Isik, Generalizations of Banach, Kannan and Ciric fixed point theorems in $b_v(s)$ -metric spaces, U.P.B. Sci. Bull., Ser. A **81**, 73–80 (2019).
- Z. D. Mitrovic and S. Radenovic, On Meir-Keeler contraction in Branciari b-metric spaces, Trans. A. Razmadze Math. Inst. 173, 83–90 (2019).
- 31. Z. D. Mitrovic, H. Aydi, and S. Radenovic, On Banach and Kannan type results in cone $b_v(s)$ -metric spaces over Banach algebra, *Acta Math. Univ. Comenianae* **89**, 143–152 (2020).
- 32. T. L. Shateri, O. Ege, and M. de la Sen, Common fixed point on the $b_v(s)$ -metric space of function-valued mappings, *AIMS Math.* **6**(1), 1065–1074 (2021).
- N. Y. Ozgur and N. Tas, Some fixed-circle theorems on metric spaces, Bull. Malaysian Math. Sci. Soc. 42(4), 1433–1449 (2019).
- N. Mlaiki, N. Tas, and N. Y. Ozgur, On the fixed-circle problem and Khan type contractions, Axioms 7(4), 80 (2018).
- M. S. Khan, A fixed point theorem for metric spaces, *Rend. Inst. Math.* Univ. Trieste 8, 69–72 (1976).

- 36. N. Tas, N. Y. Ozgur, and N. Mlaiki, New types of f_C -contractions and the fixed-circle problem, *Mathematics* **6**(10), 188 (2018).
- U. Celik and N. Ozgur, On the fixed-circle problem, *Facta Univ. Ser.* Math. Inf. 35(5), 1273–1290 (2020).
- S. K. Chatterjea, Fixed point theorems, C.R. Acad. Bulgare Sci. 25, 727-730 (1972).
- L. Ciric, A generalization of Banach's contraction principle, Proc. Am. Math. Soc. 45(2), 267–273 (1974).
- 40. G. E. Hardy and T. D. Rogers, A generalization of a fixed point theorem of Reich, *Can. Math. Bull.* **16**, 201–206 (1973).
- S. Reich, Some remarks concerning contraction mappings, Canad. Math. Bull. 14, 121–124 (1971).
- N. Mlaiki, U. Celik, N. Tas, N. Y. Ozgur, and A. Mukheimer, Wardowski type contractions and the fixed-circle problem on s-metric spaces, J. Math. 9127486, 1–9 (2018).
- N. Y. Ozgur, Fixed-disc results via simulation functions, *Turkish J. Math.* 43, 2794–2805 (2019).
- H. Aydi, N. Tas, N. Y. Ozgur, and N. Mlaiki, Fixed-discs in rectangular metric spaces, *Symmetry* 11(2), 294 (2019).
- 45. N. Y. Ozgur and N. Tas, Some fixed-circle theorems and discontinuity at fixed circle, *AIP Conf. Proc.* **1926**, 020048 (2018).
- N. Y. Ozgur and N. Tas, Fixed-circle problem on s-metric spaces with a geometric viewpoint, *Facta Uni. Ser. Math. Inform.* 34(3), 459–472 (2019).
- 47. N. Tas, Bilateral-type solutions to the fixed-circle problem with rectified linear units application, *Turkish J. Math.* 44, 1330–1344 (2020).
- N. Ozgur and N. Tas, Geometric properties of fixed points and simulation functions, arXiv:2102.05417, 21 (2021). https://doi.org/10.48550/ arXiv.2102.05417
- 49. G. Z. Ercinar, Some geometric properties of fixed points. PhD thesis, Eskisehir Osmangazi University, Turkey (2020).
- M. Joshi and A. T. S. K. Padaliya, Fixed point to fixed ellipse in metric spaces and discontinuous activation function, *Appl. Math. E-Notes.* 21, 225–237 (2021).
- N. Tas and N. Ozgur, New fixed-figure results on metric spaces. In eds. S. H. M. K. P. Debnath, P. and B. Hazarika, *Fixed Point Theory* and Fractional Calculus-Recent Advances and Applications, pp. 33–62. Springer, Singapore (2022).
- H. Aytimur, S. Guvenc, and N. Tas, New fixed figure results with the notion of k-ellipse, Math. Morav. 27(1), 37–52 (2023).
- A. Pant, R. P. Pant, and M. C. Joshi, Caristi type and Meir-Keeler type fixed point theorems, *Filomat* 33(12), 3711–3721 (2019).

- 54. M. Arshad, J. Ahmad, and E. Karapinar, Some common fixed point results in rectangular metric spaces, *Int. J. Anal.* **2013**, 1–7 (2013).
- 55. E. Karapinar, F. Khojasteh, and W. Shatanawi, Revisiting Ciric-type contraction with Caristi's approach, *Symmetry* **11**(6), 726 (2019).
- 56. L. V. Nguyen, On fixed points of asymptotically regular mappings, *Rend. Circ. Mat. Palermo 2* **70**, 709–719 (2021).
- R. P. Pant, V. Rakocevic, D. Gopal, A. Pant, and M. Ram, A general fixed point theorem, *Filomat* 35(12), 4061–4072 (2021).
- 58. I. Isik, Fixed point theorems in generalized metric spaces. Master's thesis, Erzurum Technical University, Turkey (2019).
- 59. S. Banach, Sur les operations dans les ensembles abstrait et leur application aux equations integrals, *Fund. Math.* **3**, 133–181 (1922).
- 60. D. Wardowski, Fixed points of a new type of contractive mappings in complete metric spaces, *Fixed Point Theory Appl.* **2012**, 94 (2012).
- N. Tas, New answers to the Rhoades' open problem and the fixed-circle problem, Conf. Proc. Sci. Technol. 3(1), 160–165 (2020).
- R. Kannan, Some results on fixed points, Bull. Calcutta Math. Soc. 60, 71–76 (1968).
- 63. C. Nwankpa, W. Ijomah, A. Gachagan, and S. Marshall, Activation functions: Comparison of trends in practice and research for deep learning. In 2nd International Conference on Computational Sciences and Technologies, pp. 124–133, MUET Jamshoro (December 2020).
- V. Nair and G. E. Hinton, Rectified linear units improve restricted Boltzmann machines. In *Proceedings of the 27th International Confer*ence on Machine Learning, pp. 807–814, Haifa (2010).
- A. Maas, A. Hannun, and A. Ng, Rectifier nonlinearities improve neural network acoustic models. In *Proceedings of the 30th International Conference on Machine Learning*, pp. 1–6, Atlanta, GA (2013).
- 66. K. He, X. Zhang, S. Ren, and J. Sun, Delving deep into rectifiers: Surpassing human-level performance on imagenet classification. In *IEEE International Conference on Computer Vision*, Santiago, Chile. pp. 1026–1034 (2015).
- B. Xu, N. Wang, H. Kong, T. Chen, and M. Li, Empirical evaluation of rectified activations in convolution network, arXiv:1505.00853, 5 pages (2015). https://doi.org/10.48550/arXiv.1505.00853.
- X. Jin, C. Xu, J. Feng, Y. Wei, J. Xiong, and S. Yan, Deep learning with s-shaped rectified linear activation units, AAAI'16: Proceedings of the Thirtieth AAAI Conference on Artificial Intelligence 30(1), 1737–1743 (2016).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0011

Chapter 11

Extended Interpolative Hardy–Rogers–Geraghty–Wardowski Contractions and an Application

Samira Hadi Bonab^{†,∥}, Vahid Parvaneh^{*,‡,**}, Zohreh Bagheri^{§,††}, and Roghayeh Jalal Shahkoohi^{¶,‡‡}
[†]Department of Mathematics, Ardabil Branch, Islamic Azad University, Ardabil, Iran
[‡]Department of Mathematics, Gilan-E-Gharb Branch, Islamic Azad University, Gilan-E-Gharb, Iran
[§]Department of Mathematics, Azadshahr Branch, Islamic Azad University, Azadshahr, Iran
[¶]Department of Mathematics, Aliabad Katoul Branch, Islamic Azad University, Aliabad Katoul, Iran
[¶]hadi.23bonab@gmail.com
^{**}zam.dalahoo@gmail.com
^{††}zohrehbagheri@yahoo.com
^{‡‡}rog.jalal@gmail.com

The goal of this research is to present the concept of extended interpolative Hardy–Rogers–Geraghty contractions using Wardowski contractions. That is, we combine four ideas: Hardy–Rogers, Geraghty, Wardoowski, and interpolative contractions. We also include an example and an application to confirm the results.

 $^{^{*}}$ Corresponding author.

1. Introduction

Banach's fixed-point theorem¹ is an essential technique in the theory of metric fixed points in mathematics, and it has been widely expanded. Hardy–Rogers contractions,² Geraghty contractions,³ and Wardowski contractions⁴ are three prominent Banach contraction extensions. The reader might refer to Refs. 5–11 for more information on the generalizations of this well-known principle.

In 2018, Karapinar¹² introduced the interpolative Kannan Contraction in metric space (MS) and proved its fixed, point theorem. Next, in Ref. 13, Karapinar *et al.* showed that the fixed point of interpolative Kannan contraction mapping may not be unique. To do this, Karapinar introduced the concept of interpolative Ćirić–Reich–Rus (CRR), type mappings to obtain fixed-point theorems in MS.

In 2019, Debonat *et al.*⁶ defined a new and modified CRR-type contraction in *b*-MS and created the corresponding fixed-point result. In the same year, Mohammadi *et al.*¹⁴ introduced an extended interpolative CRR-type *F*-contraction and produced some related fixed-point results. For more results on interpolation and *F*-contractions, see Refs. 15 and 16 and Refs. 17–22, respectively.

A new type of contraction mapping on a metric space termed F-contraction was developed in Ref. 4, and a fixed-point theorem for such a map in a complete metric space (CMS) was proved. Wardowski contraction is used to prove some results in fixed-point theory.

Definition 1 (Ref. 4). In an MS (Λ, \aleph) , a mapping $L : \Lambda \to \Lambda$ is called an Q-contraction if

$$\aleph(\mathrm{L}_{\mathcal{I}},\mathrm{L}_{\mathcal{I}}') > 0 \Rightarrow \tau + \mathcal{Q}(\aleph(\mathrm{L}_{\mathcal{I}},\mathrm{L}_{\mathcal{I}}')) \le \mathcal{Q}(\aleph(\mathfrak{I},\mathfrak{I}')),$$

for some $\tau > 0$ and for all $j, j' \in \Lambda$, where $\mathcal{Q} : (0, +\infty) \to (-\infty, +\infty)$ is a mapping so that:

- (F1) \mathcal{Q} is strictly increasing,
- (F2) for each sequence $\{\alpha_n\}$ of positive real numbers,

$$\lim_{n \to \infty} \alpha_n = 0 \Leftrightarrow \lim_{n \to \infty} \mathcal{Q}(\alpha_n) = -\infty$$

(F3)
$$\lim_{\alpha \to 0^+} \alpha^{\nu} \mathcal{Q}(\alpha) = 0$$
 for some $\nu \in (0, 1)$.

The Kannan contraction appears to be modified in an interpolative Kannan contraction. Karapinar *et al.*¹³ improved this notion. Many of the existing contraction-type inequalities have been adjusted to use the pattern of an interpolative Kannan contraction since Karapinar's invention of an interpolative Kannan contraction.¹²

Definition 2 (Ref. 23). A mapping L on an MS (Λ, \aleph) is called Kannan if there is $\mho \in [0, \frac{1}{2})$ so that

$$\aleph(\mathbf{L}_{\mathcal{I}},\mathbf{L}_{\mathcal{I}}') \leq \mho[\aleph(\jmath,\mathbf{L}_{\mathcal{I}}) + \aleph(\jmath',\mathbf{L}_{\mathcal{I}}')]$$

for all $j, j' \in \Lambda$.

Theorem 1 (Ref. 12). Let (Λ, \aleph) be a CMS. Let $L : \Lambda \to \Lambda$ be an interpolative Kannan-type contraction, that is,

$$\aleph(Lj, Lj') \le \mho(\aleph(j, Lj))^{\rho} (\aleph(j', Lj'))^{1-\rho}$$

for all $j, j' \in \Lambda$ with $j \neq Lj$, where $\mho \in [0, 1)$ and $\rho \in (0, 1)$. Then there is a unique fixed point in Λ for L.

In Ref. 13 a contradictory example for Theorem 1 was presented. A modified version of this theorem is the following theorem.

Theorem 2 (Ref. 13). In a CMS (Λ, \aleph) , let the mapping $L: \Lambda \to \Lambda$ be given so that

$$\aleph(L_{\mathcal{I}}, L_{\mathcal{I}}') \leq \mho(\aleph(j, L_{\mathcal{I}}))^{\rho} . (\aleph(j', L_{\mathcal{I}}'))^{1-\rho}$$

for all $j, j' \in \Lambda - Fix(L)$ where $Fix(L) = \{z \in \Lambda : Lz = z\}$. Then L has a unique fixed point in Λ .

Hardy and Rogers^2 generalized the principle of Banach contraction as follows:

Theorem 3 (Ref. 2). In a CMS (Λ, \aleph) , let the mapping $L : \Lambda \to \Lambda$ be given so that

$$\begin{split} \aleph(Lj, Lj') &\leq \rho \aleph(j, j') + \varrho \aleph(j, Lj) + \sigma \aleph(j', Lj') \\ &+ \eta \left[\frac{\aleph(j, Lj') + \aleph(j', Lj)}{2} \right] \end{split}$$

for all $j, j' \in \Lambda$, where coefficients are nonnegative reals so that their sum is less than one. Then L possesses a unique fixed point in Λ .

In Ref. 24, using the interpolation approach, the Hardy–Rogers contraction was generalized as follows:

Definition 3 (Ref. 24). In an MS (Λ, \aleph) , we say that the mapping $L : \Lambda \to \Lambda$ is an interpolative Hardy–Rogers-type contraction if there exists $\mho \in [0, 1)$ and $\rho, \varrho, \sigma \in (0, 1)$ with $\rho + \varrho + \sigma < 1$ so that

$$\begin{split} \aleph(\mathbf{L}j,\mathbf{L}j') &\leq \mho(\aleph(j,j'))^{\rho}.(\aleph(j,\mathbf{L}j))^{\varrho}.(\aleph(j',\mathbf{L}j'))^{\sigma} \\ &\cdot \left[\frac{\aleph(j,\mathbf{L}j') + \aleph(j',\mathbf{L}j)}{2}\right]^{(1-\rho-\varrho-\sigma)} \end{split}$$

for all $j, j' \in \Lambda - Fix(\mathbf{L})$.

Theorem 4 (Ref. 24). In a CMS (Λ, \aleph) , if L be an interpolative Hardy-Rogers-type contraction, then L possesses a fixed point in Λ .

In 1972, Reich²⁵ presented the following theorem as a generalization of the concepts of Kannan and Banach contraction:

Theorem 5 (Ref. 25). In a CMS (Λ, \aleph) , let $L : \Lambda \to \Lambda$ such that

 $\aleph(Lj, Lj') \le \rho \aleph(j, j') + \varrho \aleph(j, Lj) + \sigma \aleph(j', Lj'),$

for all $j, j' \in \Lambda$, where $\rho, \varrho, \sigma \geq 0$ so that $\rho + \varrho + \sigma < 1$. Then L admits a unique fixed point.

Ćirić proved the following theorem, which is one of the known results in fixed-point theory:

Theorem 6. In a CMS (Λ, \aleph) , if $L : \Lambda \to \Lambda$ be a self-mapping, such that the inequality

$$\aleph(Lj, Lj') \le \mho[\aleph(j, j') + \aleph(j, Lj) + (\aleph(j', Lj')],$$

for all $j, j' \in \Lambda$, where $\mho \in [0, \frac{1}{3})$, then L has a unique fixed point.

In 1973, Geraghty³ presented the following theorem as a generalization of the Banach contraction principle.

Theorem 7 (Ref. 3). In a CMS (Λ, \aleph) , let the mapping $L : \Lambda \to \Lambda$ be given so that

$$\aleph(Lj, Lj') \le \mho(\aleph(j, j'))\aleph(j, j')$$

where $\mathfrak{V}: [0,\infty) \to [0,1)$ is a function so that $\mathfrak{V}(t_n) \to 1$ yields that $t_n \to 0$ as $n \to \infty$. Then there is a unique fixed point $j^* \in \Lambda$ for L.

Definition 4 (Ref. 26). In a *b*-MS (Λ, b, s) , we call the mapping $L : \Lambda \to \Lambda$ is an interpolative CRR-type Geraghty contraction if there exists $\rho, \rho \in (0, 1)$ with $\rho + \rho < 1$, so that

$$\begin{split} \aleph(\mathbf{L}_{\mathcal{I}},\mathbf{L}_{\mathcal{J}}') &\leq \mho[(\aleph(\jmath,\jmath'))^{\rho}.(\aleph(\jmath,\mathbf{L}_{\mathcal{I}}))^{\varrho}.(\aleph(\jmath',\mathbf{L}_{\mathcal{J}}'))^{(1-\rho-\varrho)}]\\ .(\aleph(\jmath,\jmath'))^{\rho}.(\aleph(\jmath,\mathbf{L}_{\mathcal{I}}))^{\varrho}.(\aleph(\jmath',\mathbf{L}_{\mathcal{J}}'))^{(1-\rho-\varrho)} \end{split}$$

for all $j, j' \in \Lambda - Fix(\mathbf{L})$, where $\mathfrak{V} : [0, \infty) \to [0, \frac{1}{s})$ is a function satisfying $\mathfrak{V}(t_n) \to \frac{1}{s}$ which implies $t_n \to 0$ as $n \to \infty$.

In Ref. 14, a contraction of the extended interpolative CRR type was introduced with the Wardowski approach as follows:

Recall that \mathbb{Q} is the set of all functions $\mathcal{Q}: (0,\infty) \to \mathbb{R}$ such that:

- (F1) \mathcal{Q} is strictly increasing,
- (F2) $\lim_{n \to \infty} t_n = 0$ iff $\lim_{n \to \infty} \mathcal{Q}(t_n) = -\infty$ for each sequence $t_n \in (0, \infty)$,
- (F3) $\lim_{t \to 0^+} t^{\nu} \mathcal{Q}(t) = 0 \text{ for some } \nu \in (0,1).$

Definition 5 (Ref. 14). In an MS (Λ, \aleph) , we call the mapping $L : \Lambda \to \Lambda$ an extended interpolative CRR-type Q-contraction if there exist $\rho, \varrho \in (0, 1)$ with $\rho + \varrho < 1, \tau > 0$ and $Q \in \mathbb{Q}$ so that

$$\begin{aligned} \tau + \mathcal{Q}(\aleph(\mathrm{L}_{\mathcal{I}},\mathrm{L}_{\mathcal{I}}')) &\leq \rho \mathcal{Q}(\aleph(j,j')) + \varrho \mathcal{Q}(\aleph(j,\mathrm{L}_{\mathcal{I}})) \\ &+ (1 - \rho - \varrho) \mathcal{Q}(\aleph(j',\mathrm{L}_{\mathcal{I}}')), \end{aligned}$$

for all $j, j' \in \Lambda - Fix(\mathbb{L})$ with $\aleph(\mathbb{L}j, \mathbb{L}j') > 0$.

Theorem 8 (Ref. 14). Any extended interpolative CRR-type Q-contraction self-mapping on a CMS (Λ, \aleph) possesses a fixed point in Λ .

To derive novel fixed-point results, we combine and unify the notions of Hardy–Rogers, Geraghty, Wardowski, and interpolative contraction in this chapter. Many of the results in the literature are generalized by our findings.

2. Main Results

In this section, we will first introduce the following definitions and then present the main results.

Let \mathbb{Q} represent the set of all functions $\mathcal{Q}: (0,\infty) \to \mathbb{R}$ such that

(Q1) Q is strictly increasing and continuous.

 $(\mathcal{Q}2) \lim_{n \to \infty} \alpha_n = 1 \Leftrightarrow \lim_{n \to \infty} \mathcal{Q}(\alpha_n) = 0, \text{ for any sequence } \alpha_n \in (0, \infty).$

Definition 6. In an MS (Λ, \aleph) , we call the mapping $L : \Lambda \to \Lambda$ an **extended interpolative Hardy–Rogers-type Geraghty** Q**contraction** if there are $\rho, \varrho, \sigma \in (0, 1)$ with $\rho + \varrho + \sigma < 1, \varrho + \sigma < 1$, $\rho + \varrho < 1$ and $Q \in \mathbf{Q}$ such that

$$\begin{aligned} \mathcal{Q}(\aleph(\mathrm{L}_{j},\mathrm{L}_{j}')) &\leq \mathcal{Q}\left(\mho\left((\aleph(j,j'))^{\rho}.(\aleph(j,\mathrm{L}_{j}))^{\varrho}.(\aleph(j',\mathrm{L}_{j}'))^{\sigma}\right) \\ &\cdot \left(\frac{\aleph(j,\mathrm{L}_{j}')+\aleph(j',\mathrm{L}_{j})}{2}\right)^{(1-\rho-\varrho-\sigma)} \right) \\ &+ \rho \mathcal{Q}(\aleph(j,j')) + \varrho \mathcal{Q}(\aleph(j,\mathrm{L}_{j})) + \sigma \mathcal{Q}(\aleph(j',\mathrm{L}_{j}')) \\ &+ (1-\rho-\varrho-\sigma)\mathcal{Q}\left(\frac{\aleph(j,\mathrm{L}_{j}')+\aleph(j',\mathrm{L}_{j})}{2}\right) \end{aligned}$$
(1)

for all $j, j' \in \Lambda - Fix(L)$ with $\aleph(Lj, Lj') > 0$ and $\mho \in S$ where S is the class of functions $\mho : [0, \infty) \to [0, 1)$ so that $\mho(\alpha_n) \to 1$ implies $\alpha_n \to 0$ as $n \to \infty$.

Theorem 9. Any extended interpolative Hardy–Rogers-type Geraghty Q-contraction self-mapping L on a CMS (Λ, \aleph) possesses a fixed point in Λ .

Proof. Choose $j_0 \in \Lambda$ and set

$$j_n = \mathbf{L} j_{n-1},$$

for every n of positive integers. If there exists m such that $j_m = j_{m+1}$, then j_m is a fixed point of L. Suppose that $j_n \neq j_{n+1}$ for all $n \ge 0$. Then by (1) we have

$$\begin{aligned} \mathcal{Q}(\aleph(j_{n+1}, j_n)) &= \mathcal{Q}(\aleph(\mathrm{L}j_n, \mathrm{L}j_{n-1})) \\ &\leq \mathcal{Q}\bigg(\mho \bigg((\aleph(j_n, j_{n-1}))^{\rho} .(\aleph(j_n, \mathrm{L}j_n))^{\varrho} .(\aleph(j_{n-1}, \mathrm{L}j_{n-1})))^{\sigma} \\ &\cdot \bigg[\frac{\aleph(j_n, \mathrm{L}j_{n-1}) + \aleph(j_{n-1}, \mathrm{L}j_n)}{2} \bigg]^{(1-\rho-\varrho-\sigma)} \bigg) \bigg) \\ &+ \rho \mathcal{Q}(\aleph(j_n, j_{n-1})) + \varrho \mathcal{Q}(\aleph(j_n, \mathrm{L}j_n)) + \sigma \mathcal{Q}(\aleph(j_{n-1}, \mathrm{L}j_{n-1}))) \\ &+ (1-\rho-\varrho-\sigma) \mathcal{Q}\left(\frac{\aleph(j_n, \mathrm{L}j_{n-1}) + \aleph(j_{n-1}, \mathrm{L}j_n)}{2} \right) \\ &\leq \mathcal{Q}\bigg(\mho \bigg((\aleph(j_n, j_{n-1}))^{\rho} .(\aleph(j_n, j_{n+1}))^{\varrho} .(\aleph(j_{n-1}, j_n))^{\sigma} \\ &\cdot \bigg[\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} \bigg]^{(1-\rho-\varrho-\sigma)} \bigg) \bigg) \\ &+ \rho \mathcal{Q}(\aleph(j_n, j_{n-1})) + \varrho \mathcal{Q}(\aleph(j_n, j_{n+1})) + \sigma \mathcal{Q}(\aleph(j_{n-1}, j_n))) \end{aligned}$$

$$+ (1 - \rho - \varrho - \sigma) \mathcal{Q} \left(\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} \right)$$

$$\leq \rho \mathcal{Q}(\aleph(j_n, j_{n-1})) + \varrho \mathcal{Q}(\aleph(j_n, j_{n+1})) + \sigma \mathcal{Q}(\aleph(j_{n-1}, j_n)))$$

$$+ (1 - \rho - \varrho - \sigma) \mathcal{Q} \left(\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} \right).$$
(2)

Suppose that $\aleph(j_n, j_{n-1}) < \aleph(j_{n+1}, j_n)$ for some $n \ge 1$. Thus, $\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} < d(j_n, j_{n+1}).$

$$\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} < d(j_n, j_{n+1}).$$

Consequently, the inequality (2) yields

$$\begin{aligned} \mathcal{Q}(\aleph(j_{n+1},j_n)) &\leq \rho \mathcal{Q}(\aleph(j_n,j_{n-1})) + \varrho \mathcal{Q}(\aleph(j_n,j_{n+1})) \\ &+ \sigma \mathcal{Q}(\aleph(j_{n-1},j_n)) + (1-\rho-\varrho-\sigma)\mathcal{Q}(\aleph(j_n,j_{n+1})). \end{aligned}$$

So,

$$(\rho + \sigma)\mathcal{Q}(\aleph(j_{n+1}, j_n)) \le (\rho + \sigma)\mathcal{Q}(\aleph(j_n, j_{n-1})),$$

which is a contradiction. Thus, we have $\aleph(j_{n-1}, j_n) < \aleph(j_n, j_{n+1})$ for all $n \ge 1$. Hence, $\{\aleph(j_n, j_{n+1})\}$ is a decreasing sequence with positive terms. Therefore, there is $r \ge 0$ so that

$$\lim_{n \to \infty} \aleph(j_{n+1}, j_n) = r.$$

We show that r = 0.

If r > 0, from continuity of Q

$$\begin{aligned} \mathcal{Q}(r) &\leq \lim_{n \to \infty} \mathcal{Q} \left(\mathcal{O} \left((\aleph(j_n, j_{n-1}))^{\rho} \cdot (\aleph(j_n, j_{n+1}))^{\varrho} \cdot (\aleph(j_{n-1}, j_n))^{\sigma} \right) \right) \\ &\cdot \left[\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2} \right]^{(1-\rho-\varrho-\sigma)} \right) \\ &+ \rho \mathcal{Q}(r) + \varrho \mathcal{Q}(r) + \sigma \mathcal{Q}(r) + (1-\rho-\varrho-\sigma) \mathcal{Q}(r) \\ &\leq \alpha \mathcal{Q}(r) + \varrho \mathcal{Q}(r) + \sigma \mathcal{Q}(r) + (1-\rho-\varrho-\sigma) \mathcal{Q}(r). \end{aligned}$$

So,

$$\lim_{n \to \infty} \mathcal{Q}\left(\mathcal{O}\left((\aleph(j_n, j_{n-1}))^{\rho} \cdot (\aleph(j_n, j_{n+1}))^{\varrho} \cdot (\aleph(j_{n-1}, j_n))^{\sigma} \right) \right) = 0.$$

which yields that

$$\lim_{n \to \infty} \Im\left(\aleph(j_n, j_{n-1}))^{\rho} \cdot (\aleph(j_n, j_{n+1}))^{\varrho} \cdot (\aleph(j_{n-1}, j_n))^{\sigma} \cdot \left[\frac{\aleph(j_{n-1}, j_n) + \aleph(j_n, j_{n+1})}{2}\right]^{(1-\rho-\varrho-\sigma)}\right) = 1.$$

So,

$$\lim_{n \to \infty} (r)^{\rho} . (r)^{\varrho} . (r)^{\sigma} . [r]^{(1-\rho-\varrho-\sigma)} = 0,$$

that is,

$$r = 0. \tag{3}$$

Now, to show that $\{j_n\}$ is a Cauchy sequence, suppose that there is $\varepsilon > 0$ so that for all *i* there exist m_i and n_i with $i < m_i < n_i$ such that

$$\aleph(j_{m_i}, j_{n_i}) \ge \varepsilon,\tag{4}$$

and

$$\aleph(j_{m_i}, j_{n_i-1}) < \varepsilon. \tag{5}$$

By (4), we have

$$\aleph(j_{m_i-1}, j_{n_i-1}) \le \aleph(j_{m_i-1}, j_{m_i}) + \aleph(j_{m_i}, j_{n_i-1}).$$
(6)

From (3) and (5), we have

$$\limsup_{i \to \infty} \aleph(j_{m_i-1}, j_{n_i-1}) \le \varepsilon.$$
(7)

As a result,

$$\limsup_{i\to\infty}\aleph(\jmath_{m_i-1},\jmath_{n_i})\leq\varepsilon.$$

On the other hand, one has

$$\begin{aligned} \mathcal{Q}(\aleph(j_{m_i}, j_{n_i})) &= \mathcal{Q}(\aleph(\mathrm{L}\iota_{m_i-1}, \mathrm{L}\iota_{n_i-1})) \\ &\leq \mathcal{Q}\bigg(\mathcal{O}\bigg(\aleph(j_{m_i-1}, j_{n_i-1}))^{\rho} .(\aleph(j_{m_i-1}, \mathrm{L}j_{m_i-1}))^{\varrho} \\ .(\aleph(j_{n_i-1}, \mathrm{L}j_{n_i-1}))^{\sigma} \\ &\cdot \bigg[\frac{\aleph(j_{m_i-1}, \mathrm{L}j_{n_i-1}) + \aleph(j_{n_i-1}, \mathrm{L}j_{m_i-1})}{2} \bigg]^{(1-\rho-\varrho-\sigma)} \bigg) \bigg) \\ &+ \rho \mathcal{Q}(\aleph(j_{m_i-1}, j_{n_i-1})) + \varrho \mathcal{Q}(\aleph(j_{m_i-1}, \mathrm{L}j_{m_i-1})) \\ &+ \sigma \mathcal{Q}(\aleph(j_{n_i-1}, \mathrm{L}j_{n_i-1})) \\ &+ (1-\rho-\varrho-\sigma) \mathcal{Q}\bigg(\frac{\aleph(j_{m_i-1}, \mathrm{L}j_{n_i-1}) + \aleph(j_{n_i-1}, \mathrm{L}j_{m_i-1})}{2} \bigg) \end{aligned}$$

$$\leq \rho \mathcal{Q}(\aleph(j_{m_i-1}, j_{n_i-1})) + \varrho \mathcal{Q}(\aleph(j_{m_i-1}, j_{m_i})) + \sigma \mathcal{Q}(\aleph(j_{n_i-1}, j_{n_i})) + (1 - \rho - \varrho - \sigma) \mathcal{Q}\left(\frac{\aleph(j_{m_i-1}, j_{n_i}) + \aleph(j_{n_i-1}, j_{m_i})}{2}\right).$$
(8)

Now, according to (3)-(7), one has

$$\begin{aligned} \mathcal{Q}(\varepsilon) &\leq \mathcal{Q}\left(\limsup_{i \to \infty} \aleph(j_{m_i}, j_{n_i})\right) \\ &\leq \rho \mathcal{Q}\left(\limsup_{i \to \infty} \aleph(j_{m_i-1}, j_{n_i-1})\right) + \rho \mathcal{Q}\left(\limsup_{i \to \infty} \aleph(j_{m_i-1}, j_{m_i})\right) \\ &+ \sigma \mathcal{Q}\left(\limsup_{i \to \infty} \aleph(j_{n_i-1}, j_{n_i})\right) \\ &+ (1 - \rho - \rho - \sigma) \mathcal{Q}\left(\limsup_{i \to \infty} \frac{\aleph(j_{m_i-1}, j_{n_i}) + \aleph(j_{n_i-1}, j_{m_i})}{2}\right) \\ &\leq \rho \mathcal{Q}(\varepsilon) + (1 - \rho - \rho - \sigma) \mathcal{Q}(\varepsilon) \\ &= (1 - \rho - \sigma) \mathcal{Q}(\varepsilon), \end{aligned}$$
(9)

which is a contradiction. So, we have shown that $\{j_n\}$ is a Cauchy sequence in A. Since (Λ, \aleph) is complete, $j_n \to j$ as $n \to \infty$, for some $j \in \Lambda$, that is, $\lim_{n \to \infty} j_n = j$. Suppose to the contrary that $j \neq Lj$. We now consider the following two cases:

Case 1: There exists a subsequence $\{j_{n_k}\}$ so that $Lj_{n_k} = Lj$ for all $k \in \mathbb{N}$. Therefore,

$$\aleph(\jmath, \mathbf{L}\jmath) = \lim_{k \to \infty} \aleph(\jmath_{n_k+1}, \mathbf{L}\jmath) = \lim_{k \to \infty} \aleph(\mathbf{L}\jmath_{n_k}, \mathbf{L}\jmath) = 0.$$

Case 2: There exists an $N \in \mathbb{N}$ so that $L_{j_k} \neq L_j$ for all $k \geq N$. Therefore, applying (1), we have

$$\begin{aligned} \mathcal{Q}(\aleph(j_{k+1}, \mathbf{L}_{j})) &= \mathcal{Q}(\aleph(\mathbf{L}_{j_{k}}, \mathbf{L}_{j})) \\ &\leq \mathcal{Q}\bigg(\mathcal{O}\bigg(\aleph(j_{k}, j))^{\rho} .(\aleph(j_{k}, \mathbf{L}_{j_{k}}))^{\varrho} .(\aleph(j, \mathbf{L}_{j}))^{\sigma} \\ &\cdot \bigg[\frac{\aleph(j_{k}, \mathbf{L}_{j}) + \aleph(j, \mathbf{L}_{j_{k}})}{2} \bigg]^{(1-\rho-\varrho-\sigma)} \bigg) \bigg) \end{aligned}$$

$$+ \rho \mathcal{Q}(\aleph(j_{k}, j)) + \varrho \mathcal{Q}(\aleph(j_{k}, \mathbf{L}j_{k})) + \sigma \mathcal{Q}(\aleph(j, \mathbf{L}j)) + (1 - \rho - \varrho - \sigma) \mathcal{Q}\left(\frac{\aleph(j_{k}, \mathbf{L}j) + \aleph(j, \mathbf{L}j_{k})}{2}\right) \leq \rho \mathcal{Q}(\aleph(j_{k}, j)) + \varrho \mathcal{Q}(\aleph(j_{k}, j_{k+1})) + \sigma \mathcal{Q}(\aleph(j, \mathbf{L}j)) + (1 - \rho - \varrho - \sigma) \mathcal{Q}\left(\frac{\aleph(j_{k}, \mathbf{L}j) + \aleph(j, j_{k+1})}{2}\right).$$
(10)

Letting $k \to \infty$ in the inequality (10), one has

$$\begin{aligned} \mathcal{Q}(\aleph(j, \mathrm{L}j)) &\leq \sigma \mathcal{Q}(\aleph(j, \mathrm{L}j)) + (1 - \rho - \varrho - \sigma) \mathcal{Q}\left(\frac{\aleph(j, \mathrm{L}j)}{2}\right) \\ &\leq \sigma \mathcal{Q}(\aleph(j, \mathrm{L}j)) + (1 - \rho - \varrho - \sigma) \mathcal{Q}(\aleph(j, \mathrm{L}j)) \\ &= (1 - \rho - \varrho) \mathcal{Q}(\aleph(j, \mathrm{L}j)). \end{aligned}$$

We deduce that j = Lj, as $\rho + \varrho < 1$.

We obtain the following result if we take $Q(t) = -\frac{1}{t} + 1$ in Theorem 9.

Corollary 1. For an MS (Λ, \aleph) , let $L : \Lambda \to \Lambda$ be a mapping. If there exists $\rho, \varrho, \sigma \in (0, 1)$ with $\rho + \varrho + \sigma < 1$, such that

$$\begin{split} &\aleph(Lj, Lj') \leq \mho \left((\aleph(j, j'))^{\rho} . (\aleph(j, Lj))^{\varrho} . (\aleph(j', Lj'))^{\sigma} \\ &\cdot \left(\frac{\aleph(j, Lj') + \aleph(j', Lj)}{2} \right)^{(1-\rho-\varrho-\sigma)} \right) \\ &+ \frac{\aleph(j, j')}{\rho + \varrho \left(\frac{\aleph(j, j')}{\aleph(j, Lj)} \right) + \sigma \left(\frac{\aleph(j, j')}{\aleph(j', Lj')} \right) + (1-\rho-\varrho-\sigma) \left(\frac{2\aleph(j, j')}{\aleph(j, Lj') + \aleph(j', Lj)} \right)}, \end{split}$$
(11)

for all $j, j' \in \Lambda - Fix(L)$, where $\mho \in S$, then L possesses a fixed point in Λ .

Taking $\mathcal{O}(t) = k$, where $k \in (0, 1)$ in Corollary 1, we have the following corollary.
Corollary 2. In an MS (Λ, \aleph) , if $L : \Lambda \to \Lambda$ be a mapping and there is $\rho, \varrho, \sigma \in (0, 1)$ with $\rho + \varrho + \sigma < 1$ such that $\aleph(L_{I}, L_{I}')$

$$\leq k + \frac{\aleph(j,j')}{\rho + \varrho\left(\frac{\aleph(j,j')}{\aleph(j,L_j)}\right) + \sigma\left(\frac{\aleph(j,j')}{\aleph(j',L_{j'})}\right) + (1 - \rho - \varrho - \sigma)\left(\frac{2\aleph(j,j')}{\aleph(j,L_{j'}) + \aleph(j',L_{j})}\right)},\tag{12}$$

for all $j, j' \in \Lambda - Fix(L)$, then L possesses a fixed point in Λ .

Example 1. Let $\Lambda = [0, 1]$, and let $d : \Lambda^2 \to \mathbb{R}_+$ with

$$\aleph(j,j') = \begin{cases} \max\{j,j'\}, & \text{if } j \neq j'; \\ 0, & otherwise. \end{cases}$$

Then (Λ, \aleph) is a CMS. Let $L : \Lambda \to \Lambda$ with

$$\mathbf{L}\boldsymbol{\jmath} = \begin{cases} 0, & \text{if } \boldsymbol{\jmath} \in \left[0, \frac{1}{2}\right); \\ \\ \frac{1}{4}, & \text{if } \boldsymbol{\jmath} \in \left[\frac{1}{2}, 1\right]. \end{cases}$$

for all $j \in \Lambda$, $\mathcal{Q}(t) = \ln t$, $\mathfrak{V}(t) = e^{-t}$ and also $\rho = \varrho = \sigma = \frac{1}{4}$.

Let $j, j' \in \Lambda - Fix(\mathbb{L})$ so that $\aleph(\mathbb{L}j, \mathbb{L}j') > 0$. Without loss of generality, one has the following case:

$$(j, j') \in \left\{ \left[0, \frac{1}{2}\right) \times \left[\frac{1}{2}, 1\right] \right\}.$$

It is easy to conclude that all conditions of Theorem 9 are satisfied and 0 is a fixed point of L.

3. An Application to Integral Equations

Let $\Lambda = C([0,T],\mathbb{R})$ be the set of all real valued continuous functions with domain [0, L] which is endowed with the metric

$$\aleph(j,j') = \sup_{\iota \in [0,T]} |j(\iota) - j'(\iota)| = \parallel j - j' \parallel_{\infty}.$$

Consider the following Fredholm integral equation:

$$j(\iota) = \int_0^T \Upsilon(\iota, s, j(s)) ds + \Gamma(\iota),$$
(13)

for all $s, \iota \in [0, T]$, where $\Upsilon : [0, T]^2 \times \mathbb{R} \to \mathbb{R}$ and $\Gamma : [0, T] \to \mathbb{R}$ are continuous.

Now, we consider the following assumption:

for all $j, j' \in \Lambda$, assume that the following condition holds:

$$\begin{aligned} |\Upsilon(s,\iota,\jmath(s)) - \Upsilon(s,\iota,\jmath'(s))| &\leq k/T \\ &+ \frac{|\jmath(\iota) - \jmath'(\iota)|}{T\rho + T\varrho\left(\frac{\|\jmath-\jmath'\|_{\infty}}{\|\jmath-L_{\jmath}\|_{\infty}}\right) + T\sigma\left(\frac{\|\jmath-\jmath'\|_{\infty}}{\|\jmath'-L_{\jmath'}\|_{\infty}}\right) + T(1-\rho-\varrho-\sigma)\left(\frac{2\|\jmath-\jmath'\|_{\infty}}{\|\jmath-L_{\jmath'}\|_{\infty}+\|\jmath'-L_{\jmath}\|_{\infty}}\right)} \end{aligned}$$

Theorem 10. Suppose that the above assumptions are true. Then the integral equation (13) possesses a unique solution in Λ .

Proof. We define $L : \Lambda \to \Lambda$ by

$$\mathcal{L}_{\mathcal{J}}(\iota) = \int_{0}^{T} \Upsilon(\iota, \upsilon, \jmath(\upsilon)) d\upsilon + \Gamma(\iota), \quad \forall \upsilon, \iota \in [0, T].$$

Then, for every $j, j' \in \Lambda$, we have

$$\begin{split} |\mathcal{L}j(\iota) - \mathcal{L}j'(\iota)| \\ &= \left| \int_0^T \Upsilon(\iota, \upsilon, j(\upsilon)) - \Upsilon(\iota, \upsilon, j'(\upsilon)) d\upsilon \right| \\ &\leq \int_0^T |\Upsilon(\iota, \upsilon, j(\upsilon)) - \Upsilon(\iota, \upsilon, j'(\upsilon))| d\upsilon \\ &\leq k + \frac{|j(\iota) - j'(\iota)|}{\rho + \varrho \left(\frac{||j - j'||_{\infty}}{||j - \mathcal{L}_j||_{\infty}}\right) + \sigma \left(\frac{||j - j'||_{\infty}}{||j' - \mathcal{L}_j'||_{\infty}}\right) + (1 - \rho - \varrho - \sigma) \left(\frac{2||j - j'||_{\infty}}{||j - \mathcal{L}_j'||_{\infty} + ||j' - \mathcal{L}_j||_{\infty}}\right)} \\ &\leq k + \frac{||j - j'||_{\infty}}{\rho + \varrho \left(\frac{||j - j'||_{\infty}}{||j - \mathcal{L}_j ||_{\infty}}\right) + \sigma \left(\frac{||j - j'||_{\infty}}{||j' - \mathcal{L}_j'||_{\infty}}\right) + (1 - \rho - \varrho - \sigma) \left(\frac{2||j - j'||_{\infty}}{||j - \mathcal{L}_j'||_{\infty} + ||j' - \mathcal{L}_j ||_{\infty}}\right)}. \end{split}$$

Take the supremum on the left hand side to find that

$$\begin{split} \aleph(\mathbf{L}j,\mathbf{L}j') \\ &= \| \mathbf{L}j - \mathbf{L}j' \|_{\infty} \\ &\leq k + \frac{\| j - j' \|_{\infty}}{\rho + \varrho \left(\frac{\| j - j' \|_{\infty}}{\| j - \mathbf{L}_j \|_{\infty}} \right) + \sigma \left(\frac{\| j - j' \|_{\infty}}{\| j' - \mathbf{L}_j ' \|_{\infty}} \right) + (1 - \rho - \varrho - \sigma) \left(\frac{2 \| j - j' \|_{\infty}}{\| j - \mathbf{L}_j ' \|_{\infty} + \| j' - \mathbf{L}_j \|_{\infty}} \right) \\ &= k + \frac{\aleph(j, j')}{\rho + \varrho \left(\frac{\aleph(j, j')}{\aleph(j, \mathbf{L}_j)} \right) + \sigma \left(\frac{\aleph(j, j')}{\aleph(j', \mathbf{L}_j')} \right) + (1 - \rho - \varrho - \sigma) \left(\frac{2\aleph(j, j')}{\aleph(j, \mathbf{L}_j') + \aleph(j', \mathbf{L}_j)} \right)}. \end{split}$$

From the above inequality, we get

$$\frac{1}{\aleph(\mathbf{L}_{\mathcal{I}},\mathbf{L}_{\mathcal{I}}')} \ge \frac{1}{k} + \rho\left(\frac{1}{\aleph(j,j')}\right) + \rho\left(\frac{1}{\aleph(j,\mathbf{L}_{\mathcal{I}})}\right) + \sigma\left(\frac{1}{\aleph(j',\mathbf{L}_{\mathcal{I}}')}\right) + (1 - \rho - \rho - \sigma)\left(\frac{2}{\aleph(j,\mathbf{L}_{\mathcal{I}}') + \aleph(j',\mathbf{L}_{\mathcal{I}})}\right).$$

This is equivalent to

$$\frac{-1}{\aleph(\mathrm{L}j,\mathrm{L}j')} \leq \frac{-1}{k} + \rho\left(\frac{-1}{\aleph(j,j')}\right) + \rho\left(\frac{-1}{\aleph(j,\mathrm{L}j)}\right) + \sigma\left(\frac{-1}{\aleph(j',\mathrm{L}j')}\right) + (1 - \rho - \rho - \sigma)\left(\frac{-2}{\aleph(j,\mathrm{L}j') + \aleph(j',\mathrm{L}j)}\right).$$

Therefore,

$$\begin{pmatrix} \frac{-1}{\aleph(\mathrm{L}j,\mathrm{L}j')} + 1 \end{pmatrix} \leq \left(\frac{-1}{k} + 1\right) + \rho\left(\frac{-1}{\aleph(j,j')} + 1\right)$$
$$+ \rho\left(\frac{-1}{\aleph(j,\mathrm{L}j)} + 1\right) + \sigma\left(\frac{-1}{\aleph(j',\mathrm{L}j')} + 1\right)$$
$$+ (1 - \rho - \rho - \sigma)\left(\frac{-2}{\aleph(j,\mathrm{L}j') + \aleph(j',\mathrm{L}j)} + 1\right).$$

Taking $Q(t) = -\frac{1}{t} + 1$ and $\mho(t) = k$, we get

$$\begin{aligned} \mathcal{Q}(\aleph(\mathrm{L}j,\mathrm{L}j')) &\leq \mathcal{Q}\left(\mho\left((\aleph(j,j'))^{\rho}.(\aleph(j,\mathrm{L}j))^{\varrho}.(\aleph(j',\mathrm{L}j'))^{\sigma}\right.\\ &\left. \cdot \left(\frac{\aleph(j,\mathrm{L}j')+\aleph(j',\mathrm{L}j)}{2}\right)^{(1-\rho-\varrho-\sigma)}\right)\right) \\ &+ \rho\mathcal{Q}(\aleph(j,j')) + \varrho\mathcal{Q}(\aleph(j,\mathrm{L}j)) + \sigma\mathcal{Q}(\aleph(j',\mathrm{L}j')) \\ &+ (1-\rho-\varrho-\sigma)\mathcal{Q}\left(\frac{\aleph(j,\mathrm{L}j')+\aleph(j',\mathrm{L}j)}{2}\right). \end{aligned}$$

Therefore, it is obvious that the mapping L has all the conditions of Theorem 9. Hence, the Fredholm integral Equation (13) admits a unique solution because L possesses a unique fixed point. \Box

4. Conclusions

In this chapter, we stated the concept of extended interpolative Hardy–Rogers-Geraghty-Wardowski contractions. We have combined the four ideas of Hardy–Rogers, Geraghty, Wardoowski, and interpolative contractions. Using these ideas, we presented a new development of the Banach contraction principle. An example and an application are provided to support the results and distinguish them from other results. Our results extend and generalized the relevant results in Refs. 2, 3, 13, 14, 24, and 26.

Acknowledgments

We thank the editors for their kind support. We are also grateful to the learned referees for useful suggestions which helped us to improve manuscript.

References

- 1. S. Banach, Sur les opérations dans les ensembles abstraits et leur application aux équations intégrales, *Fund. Math.* **3**, 133–181 (1922).
- 2. G. E. Hardy and T. D. Rogers, A generalization of a fixed point theorem of Reich, *Canad. Math. Bull.* **16**, 201–206 (1973).
- M. A. Geraghty, On contractive mappings, Proc. Am. Math. Soc. 40, 604–608 (1973).
- 4. D. Wardowski, Fixed points of a new type of contractive mappings in complete metric spaces, *Fixed Point Theory Appl.* **94**, 6 (2012).
- L. Cirić, On contraction type mappings, Math. Balkanica. 1, 52–57 (1971).
- P. Debnath and M. de La Sen, Fixed-points of interpolative Círić-Reich-Rus-type contractions in b-metric spaces, Symmetry 12(1), 1–9 (2020). https://doi.org/10.3390/sym12010012
- P. Debnath, N. Konwar, and S. Radenović, eds., Metric Fixed Point Theory—Applications in Science, Engineering and Behavioural Sciences. Forum for Interdisciplinary Mathematics. Springer, Singapore (2021).
- P. Debnath, H. M. Srivastava, P. Kumam, and B. Hazarika, eds., Fixed point theory and fractional calculus—recent advances and applications. Forum for Interdisciplinary Mathematics. Springer, Singapore (2022).

- S. Hadi Bonab, R. Abazari, A. Bagheri Vakilabad, and H. Hosseinzadeh, Generalized metric spaces endowed with vector-valued metrics and matrix equations by tripled fixed point theorems, *J. Inequal. Appl.* Paper No. 204, 16 (2020).
- H. Hosseinzadeh, S. Hadi Bonab, and K. Amini Sefidab, Some common fixed point theorems for four mapping in generalized metric spaces, *Thai J. Math.* 20(1), 425–437 (2022).
- V. Parvaneh, S. Hadi Bonab, H. Hosseinzadeh, and H. Aydi, A tripled fixed point theorem in C*-algebra-valued metric spaces and application in integral equations, Adv. Math. Phys. Art. ID 5511746, 6 (2021).
- E. Karapinar, Revisiting the Kannan type contractions via interpolation, Adv. Theory Nonlinear Anal. Appl. 2, 85–87 (2018).
- E. Karapinar, R. Agarwal, and H. Aydi, Interpolative Reich-Rus-Círicítype contractions on partial metric spaces, *Mathematics*. 6,256 (2018).
- 15. H. Aydi, C.-M. Chen, and E. Karapinar, Interpolative Cirić-Reich-Rus type contractions via the Branciari distance, *Mathematics*. **784** (2019).
- P. Debnath and M. de La Sen, Set-valued interpolative Hardy-Rogers and set-valued Reich-Rus-Círić-type contractions in b-metric spaces, *Mathematics.* 7(9), 849 (2019).
- 17. M. Alansari and M. U. Ali, On interpolative F-contractions with shrink map, Adv. Diff. Equ. Paper No. 282, 13 (2021).
- 18. M. Mudesh, A. Hussian, M. Arshad, and E. Ameer, Recent thought of α_* -Geraghty *f*-contraction with application, *J. Math. Extension*. **16**(7), 1–28 (2022).
- P. Debnath, Optimization through best proximity points for multivalued F-contractions, Miskolc Math. Notes. 22(1), 143–151 (2021).
- N. Fabiano, V. Parvaneh, D. Mirković, L. Paunović, and S. Radenović, On W-contractions of Jungck-Ćirić-Wardowski-type in metric spaces, *Cogent Math. Stat.* 7(1), Art. ID 1792699, 12 (2020).
- P. V. M. B. Isik, H. and I. Altun, Common fixed point results for generalized Wardowski type contractive multi-valued mappings, *Mathematics*. 7(11) (2019).
- V. Parvaneh, F. Golkarmanesh, and R. George, Fixed points of Wardowski-Ćirić-Presić type contractive mappings in a partial rectangular b-metric space, J. Math. Anal. 8(1), 183–201 (2017).
- R. Kannan, Some results on fixed points, Bull. Calcutta Math. Soc. 60, 71–76 (1968).

- 24. E. Karapinar, O. Alqahtani, and H. Aydi, On interpolative Hardy-Rogers type contractions, *Symmetry.* **11**, 8 (2018).
- S. Reich, Fixed points of contractive functions, Boll. Un. Mat. Ital. (4).
 5, 26–42 (1972).
- 26. S. S. Alzaid, A. Fulga, and F. S. Alshammari, Discussion on Geraghty type hybrid contractions, *J. Funct. Spaces.* Art. ID 6614291, 8 (2020).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0012

Chapter 12

(η, ψ) -Rational F-Contractions and Weak-Wardowski Contractions in a Triple-Controlled Modular-Type Metric Space

Hemant Kumar Nashine^{\dagger,\ddagger,\parallel}, Samira Hadi Bonab^{\\$,**}, and Vahid Parvaneh^{*,\P,\dagger\dagger}

[†]Mathematics Division, School of Advanced Sciences and Languages, VIT Bhopal University, Bhopal, MP 466114, India
[‡]Department of Mathematics and Applied Mathematics, University of Johannesburg, Kingsway Campus, Auckland Park 2006, South Africa
[§]Department of Mathematics, Ardabil Branch, Islamic Azad University, Ardabil, Iran
[¶]Department of Mathematics, Gilan-E-Gharb Branch, Islamic Azad University, Gilan-E-Gharb, Iran
[∥]hemantkumar.nashine@vitbhopal.ac.in drhemantnashine@gmail.com
^{**}hadi.23bonab@gmail.com

In this chapter, we present the concept of triple-controlled modular \hbar -metric spaces and (η, ψ) -rational *F*-contractions and weak-Wardowski

^{*}Corresponding author.

contractions. We also use these concepts to survey some of the fixed point results in triple-controlled fuzzy \hbar -metric spaces. Some examples and an application are provided to confirm the results.

1. Introduction

There are dozens of generalizations of metric spaces, e.g. vector-valued metric spaces, 1,2 double-controlled metric spaces, 3 b-rectangular metric spaces, 4 generalized parametric metric spaces, 5 C*-algebra-valued metric spaces⁶ etc. In these settings, the Banach's fixed point theorem⁷ was developed by several authors. For a detailed relevant study on new fixed point results we refer the reader to Refs. 8–11.

In Refs. 5,12,13 the authors used the (α, ψ) -rational type contractive mappings to prove some fixed point theorems. In this chapter, we introduce the concept of triple-controlled modular \hbar -metric space and via these concepts obtain some new fixed point results.

Definition 1 (Ref. 14). Let \mathcal{Q} be a nonempty set and $s \geq 1$ be a real number. A mapping $d: \mathcal{Q}^2 \longrightarrow \mathbb{R}$ is called a *b*-metric on \mathcal{Q} , if

(1) $d(\ell, \ell') \ge 0$ for each $\ell, \ell' \in \mathcal{Q}$ and $d(\ell, \ell') = 0$ if and only if $\ell = \ell'$;

(2) $d(\ell, \ell') = d(\ell', \ell)$ for each $\ell, \ell' \in \mathcal{Q}$;

(3) $d(\ell, \ell') \leq s[d(\ell, \ell'') + d(\ell'', \ell')]$ for each $\ell, \ell', \ell'' \in \mathcal{Q}$.

Then (\mathcal{Q}, d) is called a *b*-metric space.

Definition 2 (Ref. 14). Let \mathcal{Q} be a nonempty set and $\wp : \mathcal{Q} \times \mathcal{Q} \rightarrow [1, \infty)$. A function $d_{\wp} : \mathcal{Q} \times \mathcal{Q} \rightarrow [0, \infty)$ is called an extended *b*-metric if for all $\ell, \ell', \ell'' \in \mathcal{Q}$ it satisfies

 $\begin{aligned} & (d_{\wp}1) \ d_{\wp}(\ell,\ell') = 0 \text{ if and only if } \ell = \ell'; \\ & (d_{\wp}2) \ d_{\wp}(\ell,\ell') = d_{\wp}(\ell',\ell); \\ & (d_{\wp}3) \ d_{\wp}(\ell,\ell'') \leq \wp(\ell,\ell'') [d_{\wp}(\ell,\ell') + d_{\wp}(\ell',\ell'')]. \end{aligned}$

The pair (\mathcal{Q}, d_{\wp}) is called an extended *b*-metric space.

Definition 3 (Ref. 3). Let \mathcal{Q} be a nonempty set and $\wp, \wp' : \mathcal{Q} \times \mathcal{Q} \to [1, \infty)$. A function $\rho : \mathcal{Q} \times \mathcal{Q} \to [0, \infty)$ is called a doublecontrolled metric if for all $\ell, \ell', \ell'' \in \mathcal{Q}$ it satisfies

- $(\rho 1) \ \rho(\ell, \ell') = 0$ if and only if $\ell = \ell'$;
- $(\rho 2) \ \rho(\ell, \ell') = \rho(\ell', \ell);$
- $(\rho3) \ \rho(\ell,\ell'') \le \wp(\ell,\ell')\rho(\ell,\ell') + \wp'(\ell',\ell'')\rho(\ell',\ell'').$

The pair (\mathcal{Q}, ρ) is called a double controlled metric space.

Definition 4 (Ref. 15). Let \mathcal{Q} be a nonempty set. The function $\sigma : (0, +\infty) \times \mathcal{Q} \times \mathcal{Q} \rightarrow [0, +\infty]$ is said to be a modular metric on \mathcal{Q} if

- 1. $\ell = \ell'$ if and only if $\sigma_{\lambda}(\ell, \ell') = 0$ for all $\lambda > 0$;
- 2. $\sigma_{\lambda}(\ell, \ell') = \sigma_{\lambda}(\ell', \ell);$
- 3. $\sigma_{\lambda+\mu}(\ell,\ell') \leq \sigma_{\lambda}(\ell,\ell'') + \sigma_{\mu}(\ell'',\ell')$ for all $\ell,\ell',\ell'' \in \mathcal{Q}$ and all $\lambda,\mu > 0$, for all $\lambda,\mu > 0$ and for all $\ell,\ell',\ell'' \in \mathcal{Q}$. The pair (\mathcal{Q},σ) is called a modular metric space.

Definition 5 (Ref. 16). Let \mathcal{Q} be a nonempty set and $s \geq 1$ be a real number. A mapping $\hat{\sigma} : (0, +\infty) \times \mathcal{Q} \times \mathcal{Q} \to [0, +\infty]$ is said to be a modular *b*-metric if the following statements for all $\ell, \ell', \ell'' \in \mathcal{Q}$ hold:

- (a) $\hat{\sigma}_{\lambda}(\ell, \ell') = 0$ for all $\lambda > 0$ if and only if $\ell = \ell'$;
- (b) $\widehat{\sigma}_{\lambda}(\ell, \ell') = \widehat{\sigma}_{\lambda}(\ell', \ell);$
- (c) $\widehat{\sigma}_{\lambda+\mu}(\ell,\ell') \leq s(\widehat{\sigma}_{\lambda}(\ell,\ell'') + \widehat{\sigma}_{\mu}(\ell'',\ell'))$. Then $(\mathcal{Q},\widehat{\sigma})$ is called a modular *b*-metric space.

Definition 6 (Ref. 17). Let (\mathcal{Q}, d) be a metric space. A mapping $L : \mathcal{Q} \to \mathcal{Q}$ is said to be an *F*-contraction if there is $\tau > 0$ such that for all $\ell, \ell' \in \mathcal{Q}$,

$$d(\mathrm{L}\ell, \mathrm{L}\ell') > 0 \Rightarrow \tau + F(d(\mathrm{L}\ell, \mathrm{L}\ell')) \le F(d(\ell, \ell')),$$

where $F: (0, +\infty) \to (-\infty, +\infty)$ is a mapping satisfying the following conditions:

(F1) F is strictly increasing, (F2)

$$\lim_{n \to \infty} a_n = 0 \Leftrightarrow \lim_{n \to \infty} F(a_n) = -\infty$$

for any sequence $\{a_n\}$ of positive real numbers,

(F3) there exists $k \in (0, 1)$ such that $\lim_{a \to 0^+} a^k F(a) = 0$.

Definition 7 (Ref. 18). Let \mathcal{Q} be a nonempty set and let $\alpha : \mathcal{Q}^2 \to [0, \infty)$. A mapping $\mathcal{L} : \mathcal{Q} \to \mathcal{Q}$ is called an α -admissible mapping if

$$\alpha(\ell, \ell') \ge 1 \Rightarrow \alpha(\mathrm{L}\ell, \mathrm{L}\ell') \ge 1 \quad \text{for all } \ell, \ell' \in \mathcal{Q}.$$

In this chapter, we study the new structure of generalized triplecontrolled modular metric spaces. In fact, we combine the ideas of controlled metric spaces, modular metric spaces and generalized metric spaces. We present some fixed point results in this new framework.

2. Main Results

In this section, we first introduce the following definitions and then present the main results. Now, we present the concept of a triplecontrolled modular \hbar -metric space (TCM \hbar M space).

Definition 8. Let \mathcal{Q} be a nonempty set and $\alpha_{\lambda}, \beta_{\lambda}, \gamma_{\lambda} : (0, \infty) \times \mathcal{Q}^3 \to [1, \infty)$. Suppose that the mapping $\hbar : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty)$ be a function satisfying the following properties:

 $\begin{array}{l} (\hbar 1) \quad \hbar_{\lambda}(\ell,\ell',\ell'') = 0 \ \text{iff} \ \ell = \ell' = \ell'', \\ (\hbar 2) \end{array}$

$$\begin{split} \hbar_{\lambda+\mu+\eta}(\ell,\ell',\ell'') &\leq \alpha_{\lambda}(\ell,a,\ell)\hbar_{\lambda}(\ell,a,\ell) + \beta_{\lambda}(\ell',a,\ell')\hbar_{\mu}(\ell',a,\ell') \\ &+ \gamma_{\lambda}(\ell'',a,\ell'')\hbar_{\eta}(\ell'',a,\ell'') \end{split}$$

for all $\ell, \ell', \ell'', a \in Q$ and for all $\lambda, \mu, \eta > 0$ (rectangle inequality).

Then, the function \hbar is called a TCM \hbar M on Q and (Q, \hbar) is called a TCM \hbar M space.

Remark 1. Note that in such extended spaces, we will not have the symmetry property. Otherwise, α_{λ} , β_{λ} , and γ_{λ} must be equal to 1.

Example 1. Let $\mathcal{Q} = C([a, b], (-\infty, +\infty))$ be the set of all continuous real-valued functions on [a, b]. Define $\hbar : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty)$ by

$$\hbar_{\lambda}(\ell, \ell', \ell'') = \sup_{t \in [a, b]} \frac{|\ell(t) - \ell'(t)| + |\ell'(t) - \ell''(t)|}{\lambda}, \quad \lambda > 0,$$

and

$$\alpha_{\lambda}(\ell(t), \ell'(t), \ell''(t)) = \beta_{\lambda}(\ell(t), \ell'(t), \ell''(t)) = \gamma_{\lambda}(\ell(t), \ell'(t), \ell''(t))$$

=
$$\sup_{t \in [a,b]} \max\{|\ell(t)|, |\ell'(t)|, |\ell''(t)|\} + \lambda + 1.$$

For all $\ell, \ell', \ell'', a \in \mathcal{Q}$, we have

$$\begin{split} \hbar_{\lambda}(\ell(t), \ell'(t), \ell''(t)) \\ &= \sup_{t \in [a,b]} \frac{|\ell(t) - \ell'(t)| + |\ell'(t) - \ell''(t)|}{\lambda} \\ &\leq \sup_{t \in [a,b]} \frac{|\ell(t) - a(t)| + |a(t) - \ell'(t)|}{\lambda} \\ &+ \sup_{t \in [a,b]} \frac{|\ell'(t) - a(t)| + |a(t) - \ell''(t)|}{\lambda} \\ &+ \sup_{t \in [a,b]} \frac{|\ell(t) - a(t)| + |a(t) - \ell''(t)|}{\lambda} \\ &\leq \hbar_{\lambda}(l(t), a(t), l(t)) + \hbar_{\lambda}(l'(t), a(t), l'(t)) + \hbar_{\lambda}(l''(t), a(t), l''(t)) \\ &\leq \alpha_{\lambda}(l(t), a(t), l(t)) \hbar_{\lambda}(l(t), a(t), l(t)) + \beta_{\lambda}(l'(t), a(t), l'(t)) \\ &\quad \hbar_{\lambda}(l'(t), a(t), l'(t)) + \gamma_{\lambda}(l''(t), a(t), l''(t)) \hbar_{\lambda}(l''(t), a(t), l''(t)). \end{split}$$

In the last part of the above inequality, we use the fact that for all $\ell, \ell', a \in \mathcal{Q}$, we have $\alpha_{\lambda}(\ell(t), a(t), \ell(t)) \geq 1$, $\beta_{\lambda}(\ell'(t), a(t), \ell'(t)) \geq 1$ and $\gamma_{\lambda}(\ell''(t), a(t), \ell''(t)) \geq 1$. Hence, \hbar is a TCM \hbar M space.

Definition 9. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space. Then

(i) A sequence $\{\ell_n\}$ is called convergent if and only if there exists $\ell \in \mathcal{Q}$ such that $\hbar_{\lambda}(\ell_n, \ell, \ell_n)$ or $\hbar_{\lambda}(\ell, \ell_n, \ell)$ go to 0 as $n \to \infty$. In this case, we write

$$\lim_{n \to +\infty} \ell_n = \ell.$$

(ii) A sequence $\{\ell_n\}$ is called \hbar -Cauchy if, for each $\epsilon > 0$, there exists a $N \in \mathbb{N}$ such that $\hbar_{\lambda}(\ell_n, \ell_m, \ell_n) < \epsilon$ for all $m, n \ge N$. **Definition 10.** The TCM \hbar M space (Q, \hbar) is called complete if, for each \hbar -Cauchy sequence $\{\ell_n\}$, there exists $\ell \in Q$ such that

$$\lim_{n \to +\infty} \hbar_{\lambda}(\ell_n, \ell, \ell_n) = 0.$$

Remark 2. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space. Then $\hbar_{\lambda}(\ell_n, \ell, \ell_n) \to 0$ $\Leftrightarrow \hbar_{\lambda}(\ell, \ell_n, \ell) \to 0.$

Lemma 1. Let (\mathcal{Q},\hbar) be a TCM $\hbar M$ space with continuous control function β_{λ} . If there exist sequences $\{\ell_n\}$ and $\{\ell'_n\}$ such that $\lim_{n\to\infty} \ell_n = \ell$ and $\lim_{n\to\infty} \ell'_n = \ell'$, then

$$\beta_{\lambda}^{-1}(\ell',\ell,\ell')\beta_{\lambda}^{-1}(\ell,\ell',\ell)\hbar_{\lambda}(\ell,\ell',\ell) \leq \lim \sup_{n \to \infty} \hbar_{\frac{\lambda}{9}}(\ell_n,\ell'_n,\ell_n)$$

and

$$\lim \sup_{n \to \infty} \hbar_{\lambda}(\ell_n, \ell'_n, \ell_n) \le \beta_{\lambda}(\ell', \ell, \ell') \beta_{\lambda}(\ell, \ell', \ell) \hbar_{\frac{\lambda}{9}}(\ell, \ell', \ell)$$

In particular, if $\ell = \ell'$, then we have $\limsup_{n\to\infty} \hbar_{\lambda}(\ell_n, \ell'_n, \ell_n) = 0$. Moreover, suppose that $\{\ell_n\}$ is convergent to ℓ and $\ell' \in \mathcal{Q}$ is arbitrary. Then, we have

$$\beta_{\lambda}^{-1}(\ell',\ell,\ell')\beta_{\lambda}^{-1}(\ell,\ell',\ell)\hbar_{\lambda}(\ell,\ell',\ell) \leq \lim\sup_{n\to\infty} \lim_{n\to\infty} \frac{\hbar_{\lambda}}{9}(\ell_n,\ell',\ell_n)$$

and

$$\lim \sup_{n \to \infty} \hbar_{\lambda}(\ell_n, \ell', \ell_n) \le \beta_{\lambda}(\ell', \ell, \ell') \beta_{\lambda}(\ell, \ell', \ell) \hbar_{\frac{\lambda}{9}}(\ell, \ell', \ell)$$

Proof. (a) Using the triangular inequality, one writes

$$\begin{split} \hbar_{\lambda}(\ell,\ell',\ell) &\leq \alpha_{\lambda}(\ell,\ell_{n},\ell)\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell) + \beta_{\lambda}(\ell',\ell_{n},\ell')\hbar_{\frac{\lambda}{3}}(\ell',\ell_{n},\ell') \\ &+ \gamma_{\lambda}(\ell,\ell_{n},\ell)\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell)) \\ &\leq (\alpha_{\lambda}(\ell,\ell_{n},\ell) + \gamma_{\lambda}(\ell,\ell_{n},\ell))\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell) + \beta_{\lambda}(\ell',\ell_{n},\ell') \\ &\times [(\alpha_{\lambda}(\ell',\ell'_{n},\ell') + \gamma_{\lambda}(\ell',\ell'_{n},\ell'))\hbar_{\frac{\lambda}{9}}(\ell',\ell'_{n},\ell') \\ &+ \beta_{\lambda}(\ell_{n},\ell'_{n},\ell_{n})\hbar_{\frac{\lambda}{9}}(\ell_{n},\ell'_{n},\ell_{n})] \end{split}$$

and

$$\begin{split} \hbar_{\lambda}(\ell_{n},\ell_{n}',\ell_{n}) &\leq \alpha_{\lambda}(\ell_{n},\ell,\ell_{n})\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n}) + \beta_{\lambda}(\ell_{n}',\ell,\ell_{n}')\hbar_{\frac{\lambda}{3}}(\ell_{n}',\ell,\ell_{n}') \\ &+ \gamma_{\lambda}(\ell_{n},\ell,\ell_{n})\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n})) \\ &\leq (\alpha_{\lambda}(\ell_{n},\ell,\ell_{n}) + \gamma_{\lambda}(\ell_{n},\ell,\ell_{n}))\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n}) \\ &+ \beta_{\lambda}(\ell_{n}',\ell,\ell_{n}')[(\alpha_{\lambda}(\ell_{n}',\ell',\ell_{n}') + \gamma_{\lambda}(\ell_{n}',\ell',\ell_{n}'))\hbar_{\frac{\lambda}{9}} \\ &\times (\ell_{n}',\ell',\ell_{n}') + \beta_{\lambda}(\ell,\ell',\ell)\hbar_{\frac{\lambda}{9}}(\ell,\ell',\ell)]. \end{split}$$

Taking the limit as $n \to \infty$, the result is obtained.

(b) Using the triangular inequality,

$$\begin{split} \hbar_{\lambda}(\ell,\ell',\ell) &\leq \alpha_{\lambda}(\ell,\ell_{n},\ell)\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell) + \beta_{\lambda}(\ell',\ell_{n},\ell')\hbar_{\frac{\lambda}{3}}(\ell',\ell_{n},\ell') \\ &+ \gamma_{\lambda}(\ell,\ell_{n},\ell)\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell)) \\ &\leq (\alpha_{\lambda}(\ell,\ell_{n},\ell) + \gamma_{\lambda}(\ell,\ell_{n},\ell))\hbar_{\frac{\lambda}{3}}(\ell,\ell_{n},\ell) \\ &+ \beta_{\lambda}(\ell',\ell_{n},\ell')[(\alpha_{\lambda}(\ell',\ell',\ell') + \gamma_{\lambda}(\ell',\ell',\ell'))\hbar_{\frac{\lambda}{9}}(\ell',\ell',\ell') \\ &+ \beta_{\lambda}(\ell_{n},\ell',\ell_{n})\hbar_{\frac{\lambda}{9}}(\ell_{n},\ell',\ell_{n})] \end{split}$$

and

$$\begin{split} \hbar_{\lambda}(\ell_{n},\ell',\ell_{n}) &\leq \alpha_{\lambda}(\ell_{n},\ell,\ell_{n})\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n}) + \beta_{\lambda}(\ell',\ell,\ell')\hbar_{\frac{\lambda}{3}}(\ell',\ell,\ell') \\ &+ \gamma_{\lambda}(\ell_{n},\ell,\ell_{n})\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n})) \\ &\leq (\alpha_{\lambda}(\ell_{n},\ell,\ell_{n}) + \gamma_{\lambda}(\ell_{n},\ell,\ell_{n}))\hbar_{\frac{\lambda}{3}}(\ell_{n},\ell,\ell_{n}) \\ &+ \beta_{\lambda}(\ell',\ell,\ell')[(\alpha_{\lambda}(\ell',\ell',\ell') + \gamma_{\lambda}(\ell',\ell',\ell'))\hbar_{\frac{\lambda}{9}}(\ell',\ell',\ell') \\ &+ \beta_{\lambda}(\ell,\ell',\ell)\hbar_{\frac{\lambda}{9}}(\ell,\ell',\ell)]. \end{split}$$

Let **F** be the set of all functions $F : \mathcal{R}^+ \to \mathcal{R}$ such that

- (F1) F is a continuous and strictly increasing mapping,
- (F2) $\lim \mu_n = 0$ if and only if $\lim F(\mu_n) = -\infty$ for each sequence $\{\mu_n\}$ in $(0, +\infty)$.

Now, let Ψ be the collection of all mappings $\psi : [0, +\infty) \to [0, +\infty)$ such that

(i) ψ is strictly increasing and upper semicontinuous; (ii) $\{\psi^n(t)\}_{n\in\mathbb{N}}$ converges to 0 as $n \to \infty$, for all t > 0; (iii) $\psi(t) < t$ for all t > 0.

Definition 11. Let \mathcal{Q} be a nonempty set and let $\eta : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$. A mapping $L : \mathcal{Q} \to \mathcal{Q}$ is called an η -admissible mapping if

$$\eta_{\lambda}(\ell, \ell', \ell) \ge 1 \Rightarrow \eta_{\lambda}(\mathrm{L}\ell, \mathrm{L}\ell', \mathrm{L}\ell) \ge 1 \quad \text{for all } \ell, \ell' \in \mathcal{Q} \text{ and } \lambda > 0.$$

Definition 12. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space and $\eta : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$. We call the mapping $L : \mathcal{Q} \to \mathcal{Q}$ an (η, ψ) -rational *F*-contractive mapping of type-I if

$$\begin{aligned} &\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell) > 0 \\ &\Rightarrow \tau + F(\beta_{\lambda}(\mathrm{L}\ell',\mathrm{L}\ell,\mathrm{L}\ell')\beta_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)\eta_{\lambda}(\ell,\ell',\ell)\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)) \\ &\leq F(\psi(\mathcal{M}(\ell,\ell',\ell))), \end{aligned}$$
(1)

for all $\ell, \ell' \in \mathcal{Q}$ where $\psi \in \Psi, \tau > 0$ $\lambda > 0$ and

$$\mathcal{M}(\ell,\ell',\ell) = \max\left\{ \hbar_{\lambda}(\ell,\ell',\ell), \hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell), \hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell'), \\ \frac{\hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell)\hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell')}{1+\hbar_{\lambda}(\ell,\ell',\ell)}, \frac{\hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell)\hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell')}{1+\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)} \right\}.$$

Note that if $F(t) = \ln(t)$ for all t > 0 and $\beta_{\lambda} = 1$, then we get

$$\eta_{\lambda}(\ell, \ell', \ell)\hbar_{\lambda}(\mathrm{L}\ell, \mathrm{L}\ell', \mathrm{L}\ell)) \leq e^{-\tau}(\psi(\hbar_{\lambda}(\ell, \ell', \ell)))$$
$$\leq \psi(\hbar_{\lambda}(\ell, \ell', \ell)), \quad \lambda > 0.$$

Therefore, L is an (η, ψ) -contraction, which will be a generalization of Samet's work.¹⁸

We now state the main theorem of this section as follows which says that any (η, ψ) -rational *F*-contractive mapping of type-I defined in a complete TCM \hbar M space has a unique fixed point: **Theorem 1.** Let (\mathcal{Q}, \hbar) be a complete $TCM\hbar M$ space, $\eta : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$, and L be an η -admissible self-mapping on \mathcal{Q} so that

- (i) there is $\ell_0 \in \mathcal{Q}$ such that $\eta_{\lambda}(\ell_0, L\ell_0, \ell_0) \geq 1$;
- (ii) L is an (η, ψ) -rational F-contractive mapping of type-I.

Then L has a unique fixed point in Q.

Proof. Choose an $\ell_0 \in \mathcal{Q}$ such that it satisfies $\eta_{\lambda}(\ell_0, L\ell_0, \ell_0) \geq 1$, and set

$$\ell_n = \mathcal{L}(\ell_{n-1}), \ n = 1, 2, \dots$$

As L is η -admissible,

$$\eta_{\lambda}(\ell_0, \mathrm{L}\ell_0, \ell_0) = \eta_{\lambda}(\ell_0, \ell_1, \ell_0) \ge 1 \Rightarrow \eta_{\lambda}(\mathrm{L}\ell_0, \mathrm{L}\ell_1, \mathrm{L}\ell_0)$$
$$= \eta_{\lambda}(\ell_1, \ell_2, \ell_1) \ge 1.$$

By induction, we have $\eta_{\lambda}(\ell_n, \ell_{n+1}, \ell_n) \ge 1$ for all $n \ge 0$. By condition (1), we get

$$F(\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1}))$$

$$= F(\hbar_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}\ell_{n+1}, \mathrm{L}\ell_{n}))$$

$$\leq F(\beta_{\lambda}(\mathrm{L}\ell_{n+1}, \mathrm{L}\ell_{n}, \mathrm{L}\ell_{n+1})\beta_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}\ell_{n+1}, \mathrm{L}\ell_{n})$$

$$\times \eta_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})\hbar_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}\ell_{n+1}, \mathrm{L}\ell_{n}))$$

$$\leq F(\psi(\mathcal{M}(\ell_{n}, \ell_{n+1}, \ell_{n}))) - \tau,$$

where

$$\mathcal{M}(\ell_n, \ell_{n+1}, \ell_n) = \max\left\{ \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n), \hbar_{\lambda}(\ell_n, \mathcal{L}\ell_n, \ell_n), \\ \frac{\hbar_{\lambda}(\ell_{n+1}, \mathcal{L}\ell_{n+1}, \ell_{n+1}),}{\frac{\hbar_{\lambda}(\ell_n, \mathcal{L}\ell_n, \ell_n)\hbar_{\lambda}(\ell_{n+1}, \mathcal{L}\ell_{n+1}, \ell_{n+1})}{1 + \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n)}, \\ \frac{\hbar_{\lambda}(\ell_n, \mathcal{L}\ell_n, \ell_n)\hbar_{\lambda}(\ell_{n+1}, \mathcal{L}\ell_{n+1}, \ell_{n+1})}{1 + \hbar_{\lambda}(\mathcal{L}\ell_n, \mathcal{L}\ell_{n+1}, \mathcal{L}\ell_n)} \right\}$$

H. K. Nashine et al.

$$= \max\left\{ \begin{split} &\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n}), \hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n}), \\ & \frac{\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1}), \\ & \frac{\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1})}{1 + \hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})}, \\ & \frac{\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1})}{1 + \hbar_{\lambda}(\ell_{n+1}, \mathbf{L}\ell_{n+2}, \ell_{n+1})} \right\} \\ &= \max\left\{\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n}), \hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1})\right\}$$

If $\mathcal{M}(\ell_n, \ell_{n+1}, \ell_n) = \hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1})$, then we simply see that it is impossible. Therefore, $\mathcal{M}(\ell_n, \ell_{n+1}, \ell_n) = \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n)$ for all $n \in \mathbb{N}$, and

$$F(\hbar_{\lambda}(\ell_{n+1},\ell_{n+2},\ell_{n+1})) \leq F(\psi(\mathcal{M}(\ell_{n},\ell_{n+1},\ell_{n}))) - \tau$$
$$= F(\psi(\hbar_{\lambda}(\ell_{n},\ell_{n+1},\ell_{n}))) - \tau$$
$$< F(\hbar_{\lambda}(\ell_{n},\ell_{n+1},\ell_{n})) - \tau.$$

This implies that

$$\hbar_{\lambda}(\ell_{n+1},\ell_{n+2},\ell_{n+1}) < \hbar_{\lambda}(\ell_n,\ell_{n+1},\ell_n).$$

On the other hand,

$$F(\hbar_{\lambda}(\ell_{n+1},\ell_{n+2},\ell_{n+1})) \leq F(\hbar_{\lambda}(\ell_{n},\ell_{n+1},\ell_{n})) - \tau$$
$$\leq F(\hbar_{\lambda}(\ell_{n-1},\ell_{n},\ell_{n-1})) - 2\tau$$
$$\vdots$$
$$\leq F(\hbar_{\lambda}(\ell_{0},\ell_{1},\ell_{0})) - n\tau.$$

Now, by (F2), if $n \to \infty$, then

$$\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1}) \to 0^+.$$
 (2)

Now, we show that $\ell_n \neq \ell_m$ for $n \neq m$. Let $\ell_n = \ell_m$ for some n > m. So, we have $\ell_{n+1} = L\ell_n = L\ell_m = \ell_{m+1}$. By continuing this

method, we conclude that $\ell_{n+k} = \ell_{m+k}$ for all $k \in \mathbb{N}$. Then inequality (1) yields

$$F(\hbar_{\lambda}(\ell_{m}, \ell_{m+1}, \ell_{m})) = F(\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n}))$$

$$= F(\hbar_{\lambda}(\mathrm{L}\ell_{n-1}, \mathrm{L}\ell_{n}, \mathrm{L}\ell_{n-1}))$$

$$\leq F(\beta_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}\ell_{n-1}, \mathrm{L}\ell_{n})\beta_{\lambda}(\mathrm{L}\ell_{n-1}, \mathrm{L}\ell_{n}, \mathrm{L}\ell_{n-1})$$

$$\times \eta_{\lambda}(\ell_{n-1}, \ell_{n}, \ell_{n-1})\hbar_{\lambda}(\mathrm{L}\ell_{n-1}, \mathrm{L}\ell_{n}, \mathrm{L}\ell_{n-1}))$$

$$\leq F(\psi(\mathcal{M}(\ell_{n-1}, \ell_{n}, \ell_{n-1}))) - \tau, \qquad (3)$$

where

$$\mathcal{M}(\ell_{n-1}, \ell_n, \ell_{n-1}) = \max \left\{ \begin{split} \hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1}), & \hbar_{\lambda}(\ell_{n-1}, \mathbf{L}\ell_{n-1}, \ell_{n-1}), \\ & \times \ \hbar_{\lambda}(\ell_n, \mathbf{L}\ell_n, \ell_n), \\ & \times \ \frac{\hbar_{\lambda}(\ell_{n-1}, \mathbf{L}\ell_{n-1}, \ell_{n-1})\hbar_{\lambda}(\ell_n, \mathbf{L}\ell_n, \ell_n)}{1 + \hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1})}, \\ & \times \ \frac{\hbar_{\lambda}(\ell_{n-1}, \mathbf{L}\ell_{n-1}, \ell_{n-1})\hbar_{\lambda}(\ell_n, \mathbf{L}\ell_n, \ell_n)}{1 + \hbar_{\lambda}(\mathbf{L}\ell_{n-1}, \mathbf{L}\ell_n, \mathbf{L}\ell_{n-1})} \right\} \\ & = \max \left\{ h_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1}), h_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1}), \\ & \times \ \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n), \\ & \times \ \frac{\hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1})\hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n)}{1 + \hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1})}, \\ & \times \ \frac{\hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1})\hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n)}{1 + \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n)} \right\} \\ & = \max \left\{ h_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1}), h_{\lambda}(\ell_n, \ell_{n+1}, \ell_n) \right\}. \\ \text{If } \mathcal{M}(\ell_{n-1}, \ell_n, \ell_{n-1}) = \hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n), \text{ then} \end{cases}$$

$$F(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m)) \leq F(\psi(\hbar_{\lambda}(\ell_n, \ell_{n+1}, \ell_n))) - \tau$$
$$\leq F(\psi^{n-m+1}(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m))) - \tau.$$

This implies that

$$\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m) \le \psi^{n-m+1}(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m)).$$
(4)

If
$$\mathcal{M}(\ell_{n-1}, \ell_n, \ell_{n-1}) = \hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1})$$
, then

$$F(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m)) \leq F(\psi(\hbar_{\lambda}(\ell_{n-1}, \ell_n, \ell_{n-1}))) - \tau$$

$$\leq F(\psi^{n-m}(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m))) - \tau.$$

This implies that

$$\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m) \le \psi^{n-m}(\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m)).$$
(5)

By (4) and (5), we have

$$\hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m) < \hbar_{\lambda}(\ell_m, \ell_{m+1}, \ell_m),$$

a contradiction. Hence, $\ell_n \neq \ell_m$ for all $n \neq m$.

Now, we show that $\{\ell_n\}$ is a Cauchy sequence in (\mathcal{Q}, \hbar) . Let there be $\varepsilon > 0$ so that for all $i \in \mathbb{N}$, there are m_i, n_i with $i < m_i < n_i$ so that

$$\hbar_{\lambda}(\ell_{m_i}, \ell_{n_i}, \ell_{m_i}) \ge \varepsilon, \text{ for all } \lambda > 0, \tag{6}$$

i.e.

$$\hbar_{\lambda}(\ell_{m_i}, \ell_{n_{i-1}}, \ell_{m_i}) < \varepsilon.$$
(7)

From (6), we have

$$\varepsilon \leq \hbar_{9\lambda}(\ell_{m_{i}}, \ell_{n_{i}}, \ell_{m_{i}}) \leq \alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})) + \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\hbar_{3\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})) + \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})] = \left[\alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\right] \times \hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\hbar_{3\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})] \leq \left[\alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\right] \times \hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}}) \times \hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}}).$$

Letting $i \to \infty$ and using (2), we get

$$\varepsilon \leq \limsup_{i \to \infty} \beta_{\lambda}(\ell_{n_i}, \ell_{m_{i+1}}, \ell_{n_i}) \beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_i}, \ell_{m_{i+1}}) \times \hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_i}, \ell_{m_{i+1}}).$$
(8)

On the other hand, we have

$$F(\beta_{\lambda}(\ell_{n_{i}},\ell_{m_{i+1}},\ell_{n_{i}})\beta_{\lambda}(\ell_{m_{i+1}},\ell_{n_{i}},\ell_{m_{i+1}})\hbar_{\lambda}(\ell_{m_{i+1}},\ell_{n_{i}},\ell_{m_{i+1}}))$$

$$=F(\beta_{\lambda}(\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}})\beta_{\lambda}(\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}}))$$

$$\leq F(\beta_{\lambda}(\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}})\beta_{\lambda}(\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}}))$$

$$\times \eta_{\lambda}(\ell_{m_{i}},\ell_{n_{i-1}},\ell_{m_{i}})\hbar_{\lambda}(\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}}))$$

$$\leq F(\psi(\mathcal{M}(\ell_{m_{i}},\ell_{n_{i-1}},\ell_{m_{i}}))-\tau, \qquad (9)$$

where

$$\mathcal{M}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}}) = \max\left\{ \hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}}), \hbar_{\lambda}(\ell_{m_{i}}, \mathrm{L}\ell_{m_{i}}, \ell_{m_{i}}), \\ \times \ \hbar_{\lambda}(\ell_{n_{i-1}}, \mathrm{L}\ell_{n_{i-1}}, \ell_{n_{i-1}}), \\ \times \ \frac{\hbar_{\lambda}((\ell_{m_{i}}, \mathrm{L}\ell_{m_{i}}, \ell_{m_{i}})\hbar_{\lambda}(\ell_{n_{i-1}}, \mathrm{L}\ell_{n_{i-1}}, \ell_{n_{i-1}}))}{1 + \hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \mathrm{L}\ell_{n_{i-1}}, \ell_{n_{i-1}})} \right\} \\ \times \ \frac{\hbar_{\lambda}((\ell_{m_{i}}, \mathrm{L}\ell_{m_{i}}, \ell_{m_{i}}, \ell_{m_{i-1}}, \mathrm{L}\ell_{m_{i}}), \hbar_{\lambda}(\ell_{m_{i}}, \ell_{m_{i-1}}, \ell_{m_{i}}))}{1 + \hbar_{\lambda}(\mathrm{L}\ell_{m_{i}}, \mathrm{L}\ell_{n_{i-1}}, \mathrm{L}\ell_{m_{i}})} \right\} \\ = \max\left\{ \hbar_{\lambda}(\ell_{n_{i-1}}, \ell_{n_{i}}, \ell_{n_{i-1}}), \\ \times \ \hbar_{\lambda}(\ell_{n_{i-1}}, \ell_{n_{i}}, \ell_{n_{i-1}}), \\ \times \ \frac{\hbar_{\lambda}((\ell_{m_{i}}, \ell_{m_{i-1}}, \ell_{m_{i}})\hbar_{\lambda}(\ell_{n_{i-1}}, \ell_{n_{i}}, \ell_{n_{i-1}}))}{1 + \hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}})} \right\}.$$

Taking the upper limit as $n \to \infty$ and by (2), we have

$$F(\varepsilon) \leq \limsup_{i \to \infty} F(\beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}})$$

$$\times \hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}}))$$

$$\leq \limsup_{i \to \infty} F(\psi(\mathcal{M}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}}))) - \tau$$

$$\leq F(\psi(\varepsilon)) - \tau$$

$$< F(\varepsilon), \qquad (10)$$

which is a contradiction. So, $\{\ell_n\}$ is a Cauchy sequence in \mathcal{Q} . Since \mathcal{Q} is complete, then there exists $\ell \in \mathcal{Q}$ such that $\ell_n \to \ell$, i.e.

$$\lim_{n \to \infty} \hbar_{\lambda}(\ell, \ell_n, \ell) = \lim_{n \to \infty} \hbar_{\lambda}(\ell_n, \ell, \ell_n) = 0.$$

Now, we show that ℓ is a fixed point of L.

First, let L be continuous. Then we have

$$u = \lim_{n \to \infty} \ell_{n+1} = \lim_{n \to \infty} \mathcal{L}\ell_n = \mathcal{L}u.$$

Let L be not continuous. Now, by Lemma 1, we have

$$F\left(\beta_{\lambda}(u, \mathrm{L}u, u)\beta_{\lambda}(\mathrm{L}u, u, \mathrm{L}u)\eta_{\lambda}(u, u, u)\beta_{\lambda}^{-1}(u, \mathrm{L}u, u)\right)$$

$$\times \beta_{\lambda}^{-1}(\mathrm{L}u, u, \mathrm{L}u)\hbar_{\lambda}(u, \mathrm{L}u, u)\right)$$

$$\leq F\left(\lim\sup_{n\to\infty}\beta_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n})\beta_{\lambda}(\mathrm{L}u, \mathrm{L}\ell_{n}, \mathrm{L}u)\eta_{\lambda}(\ell_{n}, u, \ell_{n})\right)$$

$$\times \limsup_{n\to\infty}\hbar_{\frac{\lambda}{9}}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n}))$$

$$\leq F\left(\psi\left(\limsup_{n\to\infty}\mathcal{M}(\ell_{n}, u, \ell_{n})\right)\right) - \tau,$$

where

$$\begin{split} \mathcal{M}(\ell_{n}, u, \ell_{n}) &= \max \left\{ \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n}), \hbar_{\frac{\lambda}{9}}(\ell_{n}, \mathrm{L}\ell_{n}, \ell_{n}), \hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u), \\ &\frac{\hbar_{\frac{\lambda}{9}}(\ell_{n}, \mathrm{L}\ell_{n}, \ell_{n})\hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u)}{1 + \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n})}, \frac{\hbar_{\frac{\lambda}{9}}(\ell_{n}, \mathrm{L}\ell_{n}, \ell_{n})\hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u)}{1 + \hbar_{\frac{\lambda}{9}}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n})} \right\} \\ &= \max \left\{ \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n}), \hbar_{\frac{\lambda}{9}}(\ell_{n}, \ell_{n+1}, \ell_{n}), \hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u), \\ &\frac{\hbar_{\frac{\lambda}{9}}(\ell_{n}, \ell_{n+1}, \ell_{n})\hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u)}{1 + \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n})}, \frac{\hbar_{\frac{\lambda}{9}}(\ell_{n}, \ell_{n+1}, \ell_{n})\hbar_{\frac{\lambda}{9}}(u, \mathrm{L}u, u)}{1 + \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n})} \right\}. \end{split}$$

This gives, as $n \to \infty$,

$$\begin{aligned} \beta_{\lambda}(u, \mathrm{L}u, u) \beta_{\lambda}(\mathrm{L}u, u, \mathrm{L}u) \eta_{\lambda}(u, u, u) \beta_{\lambda}^{-1} \\ &\times (\mathrm{L}u, u, \mathrm{L}u) \beta_{\lambda}^{-1}(u, \mathrm{L}u, u) \hbar_{\lambda}(u, \mathrm{L}u, u) \\ &\leq \psi \left(\limsup_{n \to \infty} \mathcal{M}(\ell_n, u, \ell_n) \right) \\ &= \psi(\hbar_{\lambda}(u, \mathrm{L}u, u)) \\ &< \hbar_{\lambda}(u, \mathrm{L}u, u). \end{aligned}$$

So, $\psi(\hbar_{\lambda}(u, Lu, u)) = \hbar_{\lambda}(u, Lu, u)$. Hence, u is a fixed point of L. To show the uniqueness, assume that there exists $v \neq u \in \mathcal{Q}$ such that Lu = u and Lv = v. Thus,

$$F(\hbar_{\lambda}(u, v, u)) = F(\hbar_{\lambda}(Lu, Lv, Lu))$$

$$\leq F(\beta_{\lambda}(Lv, Lu, Lv)\beta_{\lambda}(Lu, Lv, Lu))$$

$$\times \eta_{\lambda}(u, v, u)\hbar_{\lambda}(Lu, Lv, Lu))$$

$$\leq F(\psi(\mathcal{M}(u, v, u))) - \tau,$$

where

$$\mathcal{M}(u, v, u) = \max\left\{ \begin{split} &\hbar_{\lambda}(u, v, u), \hbar_{\lambda}(u, \mathrm{L}u, u), \hbar_{\lambda}(v, \mathrm{L}v, v), \\ &\frac{\hbar_{\lambda}(u, \mathrm{L}u, u)\hbar_{\lambda}(v, \mathrm{L}v, v)}{1 + \hbar_{\lambda}(u, v, u)}, \frac{\hbar_{\lambda}(u, \mathrm{L}u, u)\hbar_{\lambda}(v, \mathrm{L}v, v)}{1 + \hbar_{\lambda}(\mathrm{L}u, \mathrm{L}v, \mathrm{L}u)} \right\} \\ &= \max\left\{ \hbar_{\lambda}(u, v, u), \hbar_{\lambda}(u, u, u), \hbar_{\lambda}(v, v, v), \\ &\frac{\hbar_{\lambda}(u, u, u)\hbar_{\lambda}(v, v, v)}{1 + \hbar_{\lambda}(u, v, u)}, \frac{\hbar_{\lambda}(u, u, u)\hbar_{\lambda}(v, v, v)}{1 + \hbar_{\lambda}(u, v, u)} \right\} \\ &= \hbar_{\lambda}(u, v, u), \end{split}$$

which is a contradiction. Then u = v. Therefore, the fixed point of L is unique.

Definition 13. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space and $\eta : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$. We call the mapping $L : \mathcal{Q} \to \mathcal{Q}$ an (η, ψ) -rational *F*-contractive mapping of type-II if for all $\ell, \ell' \in \mathcal{Q}$

$$\begin{aligned} &\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell) > 0 \\ &\Rightarrow \tau + F(\beta_{\lambda}(\mathrm{L}\ell',\mathrm{L}\ell,\mathrm{L}\ell')\beta_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)\eta_{\lambda}(\ell,\ell',\ell)\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)) \\ &\leq F(\psi(\mathcal{M}(\ell,\ell',\ell))), \end{aligned} \tag{11}$$

where $\psi \in \Psi$, $\tau > 0$ and

$$\mathcal{M}(\ell,\ell',\ell) = \max\left\{ \begin{split} &\hbar_{\lambda}(\ell,\ell',\ell), \hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell), \hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell'), \\ &\frac{\hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell)\hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell')}{1+\hbar_{\lambda}(\ell,\ell',\ell)+\hbar_{\lambda}(\ell,\mathrm{L}\ell',\ell)+\hbar_{\lambda}(\ell',\mathrm{L}\ell,\ell')}, \\ &\frac{\hbar_{\lambda}(\ell,\mathrm{L}\ell',\ell)\hbar_{\lambda}(\ell,\ell',\ell)}{1+\hbar_{\lambda}(\ell,\mathrm{L}\ell,\ell)+\hbar_{\lambda}(\ell',\mathrm{L}\ell,\ell')+\hbar_{\lambda}(\ell',\mathrm{L}\ell',\ell')} \right\}. \end{split}$$

294

We now state another main theorem of this section, which says that any (η, ψ) -rational *F*-contractive mapping of type-II defined in a complete TCM \hbar M space has a unique fixed point.

Theorem 2. Let (\mathcal{Q}, \hbar) be a complete $LCM\hbar M$ space, $\eta : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$, and L be an η -admissible self-mapping on \mathcal{Q} so that

(i) there is $\ell_0 \in \mathcal{Q}$ such that $\eta_{\lambda}(\ell_0, L\ell_0, \ell_0) \geq 1$;

(ii) L is an (η, ψ) -rational F-contractive mapping of type-II.

Then L has a unique fixed point in Q.

Proof. Following the proof of Theorem 1, we can complete the proof. \Box

Example 2. Let $\mathcal{Q} = C([a, b], [0, 1])$ be the set of all continuous real-valued functions on [a, b]. Consider the TCM \hbar M space and the functions α_{λ} , β_{λ} and γ_{λ} presented in Example 1. Choose

$$\mathcal{L}\ell(t) = \begin{cases} \frac{\ell(t)}{6} & \text{if } \ell(t) \in \left[0, \frac{1}{2}\right);\\\\ \frac{\ell(t)}{12} & \text{if } \ell(t) \in \left[\frac{1}{2}, 1\right]. \end{cases}$$

Also, $\psi(t) = \frac{t}{3}$, $\tau = \frac{1}{4}$, $F(\ell) = \ln \ell$ and $\eta_{\lambda}(\ell(t), \ell'(t), \ell''(t)) = 1$. Consider the following cases:

(1) Let $\ell(t), \ell'(t) \in [0, \frac{1}{2})$. Then if $\ell' \leq \ell$, we have

$$\begin{aligned} \tau + F(\beta_{\lambda}(\mathrm{L}\ell'(t), \mathrm{L}\ell(t), \mathrm{L}\ell'(t))\beta_{\lambda}(\mathrm{L}\ell(t), \mathrm{L}\ell'(t), \mathrm{L}\ell(t)) \\ & \times \eta_{\lambda}(\ell(t), \ell'(t), \ell(t))\hbar_{\lambda}(\mathrm{L}\ell(t), \mathrm{L}\ell'(t), \mathrm{L}\ell(t))) \\ &= \tau + \ln\left(\sup_{t \in [a,b]} \max\left\{\left|\frac{\ell(t)}{6}\right|, \left|\frac{\ell'(t)}{6}\right|\right\} + 1\right) \\ & \times \left(\sup_{t \in [a,b]} \max\left\{\left|\frac{\ell(t)}{6}\right|, \left|\frac{\ell'(t)}{6}\right|\right\} + 1\right) \left(\sup_{t \in [a,b]} \frac{2|\ell(t) - \ell'(t)|}{6\lambda}\right) \end{aligned}$$

$$\leq \tau + \ln\left(\sup_{t\in[a,b]} \left|\frac{\ell(t)}{6}\right| + 1\right) \left(\sup_{t\in[a,b]} \left|\frac{\ell(t)}{6}\right| + 1\right)$$
$$\times \left(\sup_{t\in[a,b]} \frac{2|\ell(t) - \ell'(t)|}{6\lambda}\right) \leq \ln\left(\sup_{t\in[a,b]} \frac{|\ell(t) - \ell'(t)|}{3\lambda}\right)$$
$$= F(\psi(\hbar_{\lambda}(\ell(t),\ell'(t),\ell(t)))) \leq F(\psi(\mathcal{M}(\ell(t),\ell'(t),\ell(t)))).$$

(2) Let $\ell(t), \ell'(t) \in [\frac{1}{2}, 1]$. Then if $\ell' \leq \ell$, we have

$$\begin{aligned} \tau + F(\beta_{\lambda}(\mathrm{L}\ell'(t), \mathrm{L}\ell(t), \mathrm{L}\ell'(t))\beta_{\lambda}(\mathrm{L}\ell(t), \mathrm{L}\ell'(t), \mathrm{L}\ell(t)) \\ &\times \eta_{\lambda}(\ell(t), \ell'(t), \ell(t))\hbar_{\lambda}(\mathrm{L}\ell(t), \mathrm{L}\ell'(t), \mathrm{L}\ell(t))) \\ &= \tau + \ln\left(\sup_{t\in[a,b]} \max\left\{\left|\frac{\ell(t)}{12}\right|, \left|\frac{\ell'(t)}{12}\right|\right\} + 1\right) \\ &\times \left(\sup_{t\in[a,b]} \max\left\{\left|\frac{\ell(t)}{12}\right|, \left|\frac{\ell'(t)}{12}\right|\right\} + 1\right) \left(\sup_{t\in[a,b]} \frac{2|\ell(t) - \ell'(t)|}{12\lambda}\right) \\ &\leq \tau + \ln\left(\sup_{t\in[a,b]} \left|\frac{\ell(t)}{12}\right| + 1\right) \left(\sup_{t\in[a,b]} \left|\frac{\ell(t)}{12}\right| + 1\right) \\ &\times \left(\sup_{t\in[a,b]} \frac{2|\ell(t) - \ell'(t)|}{12\lambda}\right) \leq \ln\left(\sup_{t\in[a,b]} \frac{|\ell(t) - \ell'(t)|}{3\lambda}\right) \\ &= F(\psi(\hbar_{\lambda}(\ell(t), \ell'(t), \ell(t)))) \leq F(\psi(\mathcal{M}(\ell(t), \ell'(t), \ell(t)))). \end{aligned}$$

(3) Let $\ell'(t) \in [0, \frac{1}{2})$ and $\ell(t) \in [\frac{1}{2}, 1]$. Clearly, $\ell' \leq \ell$ and then we have

$$\begin{aligned} \tau + F(\beta_{\lambda}(\mathbb{L}\ell'(t),\mathbb{L}\ell(t),\mathbb{L}\ell(t))\beta_{\lambda}(\mathbb{L}\ell(t),\mathbb{L}\ell'(t),\mathbb{L}\ell(t)) \\ &\times \eta_{\lambda}(\ell(t),\ell'(t),\ell(t))\hbar_{\lambda}(\mathbb{L}\ell(t),\mathbb{L}\ell'(t),\mathbb{L}\ell(t))) \\ &= \tau + \ln\left(\sup_{t\in[a,b]} \max\left\{\left|\frac{\ell(t)}{12}\right|,\left|\frac{\ell'(t)}{6}\right|\right\} + 1\right) \\ &\times \left(\sup_{t\in[a,b]} \max\left\{\left|\frac{\ell(t)}{12}\right|,\left|\frac{\ell'(t)}{6}\right|\right\} + 1\right) \left(\sup_{t\in[a,b]} \frac{2|\ell(t) - 2\ell'(t)|}{12\lambda}\right) \end{aligned}$$

$$\leq \tau + \ln\left(\sup_{t\in[a,b]}\left|\frac{\ell(t)}{6}\right| + 1\right)\left(\sup_{t\in[a,b]}\left|\frac{\ell(t)}{6}\right| + 1\right)$$
$$\times\left(\sup_{t\in[a,b]}\frac{2|\ell(t)-\ell'(t)|}{6\lambda}\right) \leq \ln\left(\sup_{t\in[a,b]}\frac{|\ell(t)-\ell'(t)|}{3\lambda}\right)$$
$$= F(\psi(\hbar_{\lambda}(\ell(t),\ell'(t),\ell(t)))) \leq F(\psi(\mathcal{M}(\ell(t),\ell'(t),\ell(t)))).$$

Therefore, all hypotheses of Theorem 1 are satisfied and L has a unique fixed point.

3. Weak-Wardowski Contraction

Let **F** be the set of all functions $F : \mathbb{R}^+ \to \mathbb{R}$ such that

- (F1) F is a continuous and strictly increasing mapping,
- (F2) $\lim \mu_n = 0$ if and only if $\lim F(\mu_n) = -\infty$ for each sequence $\{\mu_n\}$ in $(0, +\infty)$.

Let Ψ' be the collection of all functions $\psi':\mathbb{R}\to (0,+\infty)$ such that

- (i) ψ' is continuous;
- (ii) $\sum_{n=1}^{\infty} \psi'(t_n) = \infty$ and $\liminf \psi'(t_n) > 0$ for all $\{t_n\} \subseteq \mathbb{R}$.

Definition 14. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space. We call the mapping $L : \mathcal{Q} \to \mathcal{Q}$ a weak-Wardowski contraction if there are $\psi' \in \Psi'$ and $F \in \mathbf{F}$ such that

$$F(\beta_{\lambda}(\mathrm{L}\ell, \mathrm{L}\ell', \mathrm{L}\ell)\beta_{\lambda}(\mathrm{L}\ell', \mathrm{L}\ell, \mathrm{L}\ell')\hbar_{\lambda}(\mathrm{L}\ell, \mathrm{L}\ell', \mathrm{L}\ell))$$

$$\leq F(\hbar_{\lambda}(\ell, \ell', \ell)) - \psi'(F(\hbar_{\lambda}(\ell, \ell', \ell))), \ \lambda > 0,$$
(12)

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$.

We now state another main theorem of this section, which says that any weak-Wardowski contraction defined in a complete $TCM\hbar M$ space has a unique fixed point.

Theorem 3. Let (\mathcal{Q}, \hbar) be a complete $TCM\hbar M$ space. Then, any weak-Wardowski contraction $L : \mathcal{Q} \to \mathcal{Q}$ has a unique fixed point.

Proof. We choose an $\ell_0 \in \mathcal{Q}$, and set

$$\ell_n = \mathcal{L}(\ell_{n-1}), \ n = 1, 2, \dots$$

By condition (12), we get

$$F(\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1}))$$

$$= F(\hbar_{\lambda}(L\ell_{n}, L\ell_{n+1}, L\ell_{n}))$$

$$\leq F(\beta_{\lambda}(L\ell_{n}, L\ell_{n+1}, L\ell_{n})\beta_{\lambda}(L\ell_{n+1}, L\ell_{n}, L\ell_{n+1}))$$

$$\times \hbar_{\lambda}(L\ell_{n}, L\ell_{n+1}, L\ell_{n}))$$

$$\leq F(\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})) - \psi'(F(\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n}))))$$

$$\leq F(\hbar_{\lambda}(\ell_{n-1}, \ell_{n}, \ell_{n-1})) - \psi'(F(\hbar_{\lambda}(\ell_{n-1}, \ell_{n}, \ell_{n-1}))))$$

$$- \psi'(F(\hbar_{\lambda}(\ell_{n}, \ell_{n+1}, \ell_{n})))$$

$$\vdots$$

$$\leq F(\hbar_{\lambda}(\ell_0, \ell_1, \ell_0)) - \sum_{i=0}^{n-1} \psi'(F(\hbar_{\lambda}(\ell_{i+1}, \ell_{i+2}, \ell_{i+1})))$$

$$\rightarrow -\infty, \ (as \ n \rightarrow \infty).$$

Then from (F2), we have

$$\hbar_{\lambda}(\ell_{n+1}, \ell_{n+2}, \ell_{n+1}) \to 0^+.$$
 (13)

Now, we show that $\{\ell_n\}$ is a Cauchy sequence in $(\mathcal{Q}, \hbar_{\lambda})$. Let there be $\varepsilon > 0$ so that for all $i \in \mathbb{N}$ there are m_i, n_i with $i < m_i < n_i$ such that

$$\hbar_{\lambda}(\ell_{m_i}, \ell_{n_i}, \ell_{m_i}) \ge \varepsilon, \text{ for all } \lambda > 0, \tag{14}$$

i.e.

$$\hbar_{\lambda}(\ell_{m_i}, \ell_{n_{i-1}}, \ell_{m_i}) < \varepsilon.$$
(15)

From (14), we have

$$\begin{split} \varepsilon &\leq \hbar_{9\lambda}(\ell_{m_{i}}, \ell_{n_{i}}, \ell_{m_{i}}) \\ &\leq \alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) \\ &+ \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\hbar_{3\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})) \\ &+ \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) \\ &= \left[\alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\right]\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) \\ &+ \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\hbar_{3\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}}) \\ &\leq \left[\alpha_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) + \gamma_{\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}})\right]\hbar_{3\lambda}(\ell_{m_{i}}, \ell_{m_{i+1}}, \ell_{m_{i}}) \\ &+ \beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}})\hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}}). \end{split}$$

Letting $i \to \infty$ and using (13), we get

$$\varepsilon \leq \limsup_{i \to \infty} \beta_{\lambda}(\ell_{n_i}, \ell_{m_{i+1}}, \ell_{n_i}) \beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_i}, \ell_{m_{i+1}}) \\ \times \hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_i}, \ell_{m_{i+1}}).$$
(16)

On the other hand, we have

$$F(\beta_{\lambda}(\ell_{n_{i}},\ell_{m_{i+1}},\ell_{n_{i}})\beta_{\lambda}(\ell_{m_{i+1}},\ell_{n_{i}},\ell_{m_{i+1}})\hbar_{\lambda}(\ell_{m_{i+1}},\ell_{n_{i}},\ell_{m_{i+1}}))$$

$$=F(\beta_{\lambda}(\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}})\beta_{\lambda}(\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}}))$$

$$\times\hbar_{\lambda}(\mathrm{L}\ell_{m_{i}},\mathrm{L}\ell_{n_{i-1}},\mathrm{L}\ell_{m_{i}}))$$

$$\leq F(\hbar_{\lambda}(\ell_{m_{i}},\ell_{n_{i-1}},\ell_{m_{i}}))-\psi'(F(\hbar_{\lambda}(\ell_{m_{i}},\ell_{n_{i-1}},\ell_{m_{i}})). (17))$$

Laking the upper limit as $n \to \infty$, we have

$$F(\varepsilon) \leq \limsup_{i \to \infty} F(\beta_{\lambda}(\ell_{n_{i}}, \ell_{m_{i+1}}, \ell_{n_{i}})\beta_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}})$$

$$\times \hbar_{\lambda}(\ell_{m_{i+1}}, \ell_{n_{i}}, \ell_{m_{i+1}}))$$

$$\leq \limsup_{i \to \infty} F(\hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}}))$$

$$-\liminf_{i \to \infty} \psi'(F(\hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}})))$$

$$< F(\varepsilon) - \liminf_{i \to \infty} \psi'(F(\hbar_{\lambda}(\ell_{m_{i}}, \ell_{n_{i-1}}, \ell_{m_{i}}))),$$

which is a contradiction. So, $\{\ell_n\}$ is a Cauchy sequence in \mathcal{Q} . Since \mathcal{Q} is complete, then there exists $\ell \in \mathcal{Q}$ such that $\ell_n \to \ell$, i.e.

$$\lim_{n \to \infty} \hbar_{\lambda}(\ell_n, \ell, \ell_n) = 0.$$

Now, we show that ℓ is a fixed point of L.

First, let L be continuous. Then we have

$$u = \lim_{n \to \infty} \ell_{n+1} = \lim_{n \to \infty} \mathcal{L}\ell_n = \mathcal{L}u.$$

Let L be not continuous. Now, using Lemma 1, we have

$$\begin{split} F(\beta_{\lambda}(\mathrm{L}u, u, \mathrm{L}u)\beta_{\lambda}(u, \mathrm{L}u, u)\beta_{\lambda}^{-1}(\mathrm{L}u, u, \mathrm{L}u)\beta_{\lambda}^{-1}(u, \mathrm{L}u, u)\hbar_{\lambda}(u, \mathrm{L}u, u)) \\ &\leq F\left(\limsup_{n \to \infty} \hbar_{\frac{\lambda}{9}}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n})\right) \\ &\leq \limsup_{n \to \infty} F(\beta_{\lambda}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n})\beta_{\lambda}(\mathrm{L}u, \mathrm{L}\ell_{n}, \mathrm{L}u)\hbar_{\frac{\lambda}{9}}(\mathrm{L}\ell_{n}, \mathrm{L}u, \mathrm{L}\ell_{n})) \\ &\leq \limsup_{n \to \infty} F(\hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n})) - \liminf_{n \to \infty} \psi'(F(\hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n}))) \\ &< F\left(\limsup_{n \to \infty} \hbar_{\frac{\lambda}{9}}(\ell_{n}, u, \ell_{n})\right) \to -\infty. \end{split}$$

So, this gives that

$$\hbar_{\lambda}(u, \mathbf{L}u, u) = 0.$$

Hence, u is a fixed point of L. To show the uniqueness, assume that there exists $v \neq u \in Q$ such that Lu = u and Lv = v. Thus,

$$\begin{aligned} F(\hbar_{\lambda}(u, v, u)) &= F(\hbar_{\lambda}(\mathrm{L}u, \mathrm{L}v, \mathrm{L}u)) \\ &\leq F(\beta_{\lambda}(\mathrm{L}v, \mathrm{L}u, \mathrm{L}v)\beta_{\lambda}(\mathrm{L}u, \mathrm{L}v, \mathrm{L}u)\hbar_{\lambda}(\mathrm{L}u, \mathrm{L}v, \mathrm{L}u)) \\ &\leq F(\hbar_{\lambda}(u, v, u)) - \psi'(F(\hbar_{\lambda}(u, v, u))) \\ &< F(\hbar_{\lambda}(u, v, u)), \end{aligned}$$

which is a contradiction. Then u = v. Therefore, the fixed point of L is unique.

Taking $\psi'(t) = \tau$, $\beta_{\lambda} = 1$, we have the following:

Corollary 1. Let (\mathcal{Q}, \hbar) be a TCM \hbar M space and $L : \mathcal{Q} \to \mathcal{Q}$ be a self-mapping such that

$$\tau + F(\hbar_{\lambda}(L\ell, L\ell', L\ell)) \le F(\hbar_{\lambda}(\ell, \ell', \ell)), \ \lambda > 0, \tag{18}$$

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} .

If in the above corollary, we define $F(t) = \ln t$, then we have the following.

Corollary 2. Let (Q, \hbar) be a TCM \hbar M space and $L : Q \to Q$ be a self-mapping such that

$$\hbar_{\lambda}(L\ell, L\ell', L\ell) \le k\hbar_{\lambda}(\ell, \ell', \ell), \ \lambda > 0, \tag{19}$$

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} .

Taking $\psi(t) = \tau + \sum_{i=1}^{\infty} t^{2i}$, $\beta_{\lambda} = 1$, we have the following:

Corollary 3. Let (Q, \hbar) be a TCM \hbar M space and $L : Q \to Q$ be a self-mapping such that

$$F(\hbar_{\lambda}(L\ell, L\ell', L\ell)) \leq F(\hbar_{\lambda}(\ell, \ell', \ell))$$
$$-\tau - \sum_{i=1}^{\infty} F(\hbar_{\lambda}(\ell, \ell', \ell))^{2i}, \ \lambda > 0, \qquad (20)$$

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} .

4. Triple-Controlled Fuzzy *h*-Metric Spaces

In this section, we introduce the concept of a triple-controlled fuzzy \hbar -metric (TCF \hbar M) space. We create a relationship between TCM \hbar M and TCF \hbar M, and present some new fixed point results in TCF \hbar M space.

Definition 15 (Schweizer and Sklar (Ref. 19)). A binary operation $\star : [0,1] \times [0,1] \rightarrow [0,1]$ is called a continuous *t*-norm if it satisfies the following assertions:

- (T1) \star is commutative and associative;
- (T2) \star is continuous;
- (T3) $a \star 1 = a$ for all $a \in [0, 1]$;
- (T4) $a \star b \leq c \star d$ when $a \leq c$ and $b \leq d$, with $a, b, c, d \in [0, 1]$.

Definition 16 (Ref.20). A three-tuple $(\mathcal{Q}, M, *)$ is said to be a fuzzy metric space if \mathcal{Q} is an arbitrary set, * is a continuous *t*-norm and *M* is a fuzzy set on $\mathcal{Q}^2 \times (0, \infty)$ satisfying the following conditions:

- (i) $M(\ell, \ell', t) > 0;$
- (ii) $M(\ell, \ell', t) = 1$ for all t > 0 if and only if $\ell = \ell'$;
- (iii) $M(\ell, \ell', t) = M(\ell', \ell, t);$
- (iv) $M(\ell, \ell', t) * M(\ell', \ell'', s) \le M(\ell, \ell'', t+s);$
- (v) $M(\ell, \ell', \cdot) : (0, \infty) \to [0, 1]$ is continuous; for all $\ell, \ell', \ell'' \in \mathcal{Q}$ and for all t, s > 0.

The function $M(\ell, \ell', t)$ denotes the degree of nearness between ℓ and ℓ' with respect to t.

Definition 17. A TCF \hbar M is an ordered triple $(\mathcal{Q}, \Lambda, \star)$ such that \mathcal{Q} is a nonempty set, \star is a continuous *t*-norm and Λ is a fuzzy set on $\mathcal{Q}^3 \times (0, \infty)$ satisfying the following conditions, for all $\ell, \ell', a \in \mathcal{Q}$ and t, s, r > 0,

- (F1) $\Lambda(\ell, \ell', \ell'', t) > 0;$
- (F2) $\Lambda(\ell, \ell', \ell'', t) = 1$ if and only if $\ell = \ell' = z$;
- (F3) $\Lambda(\ell, \ell', \ell'', \alpha_t(\ell, a, \ell)t + \beta_s(\ell', a, \ell')s + \gamma_r(\ell'', a, \ell'')r) \ge \Lambda(\ell, a, \ell, t) \star \Lambda(\ell', a, \ell', s) \star \Lambda(\ell'', a, \ell'', r);$
- (F5) $\Lambda(\ell, \ell', \ell, \cdot) : (0, +\infty) \to (0, 1]$ is left continuous.

Definition 18. Let $(\mathcal{Q}, \Lambda, \star)$ be a TCF \hbar M space and $\{a_n\}$ be a sequence in \mathcal{Q} and $a \in \mathcal{Q}$.

- (i) $\{a_n\}$ is said to be convergent and converges to a if $\lim \Lambda(a_n, a, a_n, t) = 1$ or $\lim \Lambda(a, a_n, a, t) = 1$ for all t > 0.
- (ii) $\{a_n\}$ is said to be a Cauchy sequence if and only if for all $\varepsilon \in (0,1)$ and t > 0, there exists n_0 such that $\Lambda(a_n, a_m, a_n, t) > 1 \varepsilon$ for all $m, n \ge n_0$.
- (iii) $(\mathcal{Q}, \Lambda, \star)$ is said to be complete if every Cauchy sequence is a convergent sequence.

Definition 19. Let $(\mathcal{Q}, \Lambda, \star)$ be a TCF \hbar M space. The TCF \hbar M Λ is called rectangular if

$$\frac{1}{\Lambda(\ell,\ell',\ell'',\lambda)} - 1 \le \alpha_{\lambda}(\ell,a,\ell) \left[\frac{1}{\Lambda(\ell,a,\ell,\lambda)} - 1 \right] + \beta_{\lambda}(\ell',a,\ell') \left[\frac{1}{\Lambda(\ell',a,\ell',\lambda)} - 1 \right] + \gamma_{\lambda}(\ell'',a,\ell'' \left[\frac{1}{\Lambda(\ell'',a,\ell'',\lambda)} - 1 \right]$$

for all $\ell, \ell', a \in \mathcal{Q}$ and $\lambda > 0$.

Remark 3. Note that $\hbar_{\lambda}(\ell, \ell', \ell'') = \frac{1}{\Lambda(\ell, \ell', \ell'', \lambda)} - 1$ is a TCM \hbar M space, whenever Λ is a rectangular TCF \hbar M.

Definition 20. Let $(\mathcal{Q}, \Lambda, \star)$ be a rectangular TCF \hbar M space and $\eta_{\lambda} : (0, \infty) \times \mathcal{Q}^3 \to [0, \infty]$. We call the mapping $L : \mathcal{Q} \to \mathcal{Q}$ a Fuzzy (η, ψ) -rational *F*-contractive mapping of type-I if for all $\ell, \ell' \in \mathcal{Q}$, there exists a function $\psi \in \Psi$ and $\tau > 0$ satisfying

$$\tau + F(\beta_{\lambda}(\mathcal{L}\ell', \mathcal{L}\ell, \mathcal{L}\ell')\beta_{\lambda}(\mathcal{L}\ell, \mathcal{L}\ell', \mathcal{L}\ell)\eta_{\lambda}(\ell, \ell', \ell))\frac{1}{\Lambda(\mathcal{L}\ell, \mathcal{L}\ell', \mathcal{L}\ell, \lambda)} - 1)$$

$$\leq F(\psi(\mathcal{M}_{\lambda}(\ell, \ell', \ell))), \ \lambda > 0,$$
(21)

where $\eta_{\lambda}(\ell, \ell', \ell) \geq 1$ and

$$\mathcal{M}_{\lambda}(\ell,\ell',\ell) = \max\left\{\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)} - 1, \frac{1}{\Lambda(\ell,\mathrm{L}\ell,\ell,\lambda)} - 1, \\ \times \frac{1}{\Lambda(\ell',\mathrm{L}\ell',\ell',\lambda)} - 1, \\ \times \frac{\left(\frac{1}{\Lambda(\ell,\mathrm{L}\ell,\ell,\lambda)} - 1\right)\left(\frac{1}{\Lambda(\ell',\mathrm{L}\ell',\ell',\lambda)} - 1\right)}{\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)}}, \\ \times \frac{\left(\frac{1}{\Lambda(\ell,\mathrm{L}\ell,\ell,\lambda)} - 1\right)\left(\frac{1}{\Lambda(\ell',\mathrm{L}\ell',\ell',\lambda)} - 1\right)}{\frac{1}{\Lambda(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell,\lambda)}}\right\}.$$

We now state another result of this section, which says that any fuzzy (η, ψ) -rational *F*-contractive mapping of type-I defined in a complete rectangular TCF \hbar M space has a unique fixed point.

Theorem 4. Let $(\mathcal{Q}, \Lambda, \star)$ be a complete rectangular TCFħM space, $\eta_{\lambda} : (0, \infty) \times \mathcal{Q}^3 \rightarrow [0, \infty]$ and let L be an η_{λ} -admissible self-mapping on \mathcal{Q} , so that

- (i) there is an $\ell_0 \in \mathcal{Q}$ such that $\eta_{\lambda}(\ell_0, L\ell_0, \ell_0) \geq 1$;
- (ii) L is a Fuzzy (η, ψ) -rational F-contractive mapping of type-I.

Then L has a unique fixed point in Q.

Proof. Using Remark 3 and Theorem 1, we can get the proof. \Box

Definition 21. Let $(\mathcal{Q}, \Lambda, \star)$ be a rectangular TCF \hbar M space. We call the mapping $L : \mathcal{Q} \to \mathcal{Q}$ a fuzzy weak-Wardowski contraction if there are $\psi' \in \Psi'$ and $F \in \mathbf{F}$ such that

$$F\left(\beta_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)\beta_{\lambda}(\mathrm{L}\ell',\mathrm{L}\ell,\mathrm{L}\ell')\frac{1}{\Lambda(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell,\lambda)}-1\right)$$

$$\leq F\left(\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)}-1\right)-\psi'\left(F\left(\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)}-1\right)\right),\ \lambda>0,$$

(22)

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$.

We now state another result of this section, which says that any weak-Wardowski contraction defined in a complete rectangular $TCF\hbar M$ space has a unique fixed point.

Theorem 5. Let $(\mathcal{Q}, \Lambda, \star)$ be a complete rectangular TCF $\hbar M$ space. Then, any fuzzy weak-Wardowski contraction $L : \mathcal{Q} \to \mathcal{Q}$ has a unique fixed point.

Proof. Using Remark 3 and Theorem 3, we can get the proof. \Box Corollary 4. Let $(\mathcal{Q}, \Lambda, \star)$ be a rectangular TCFħM space, so that

$$\tau + F\left(\frac{1}{\Lambda(\ell,\ell,\ell,\ell,\lambda)} - 1\right) \le F\left(\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)} - 1\right), \ \lambda > 0, \ (23)$$

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} .

Corollary 5. Let $(\mathcal{Q}, \Lambda, \star)$ be a rectangular TCFħM space so that

$$\frac{1}{\Lambda(\ell\ell,\ell\ell',\ell\ell,\lambda)} - 1 \le k \frac{1}{\Lambda(\ell,\ell',\ell,\lambda)} - 1, \ \lambda > 0, \tag{24}$$

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} . Corollary 6. Let $(\mathcal{Q}, \Lambda, \star)$ be a rectangular TCFħM space so that

$$F\left(\frac{1}{\Lambda(\ell\ell,\ell\ell',\ell\ell,\lambda)} - 1\right) \le F\left(\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)} - 1\right)$$
$$-\tau - \sum_{i=1}^{\infty} F\left(\frac{1}{\Lambda(\ell,\ell',\ell,\lambda)} - 1\right)^{2i}, \ \lambda > 0,$$
(25)

for all $\ell, \ell' \in \mathcal{Q}$ with $L\ell \neq L\ell'$. Then, L has a unique fixed point in \mathcal{Q} .

5. Application

Let $\mathcal{Q} = C([a, b], (-\infty, +\infty))$ be the set of real continuous functions defined on [a, b]. Consider the following Fredholm integral equation:

$$\ell(t) = \int_{a}^{b} f(t, s, \ell(s))ds + g(t)$$
(26)

for all $s, t \in [a, b]$, where $f : [a, b]^2 \times (-\infty, +\infty) \to (-\infty, +\infty)$ and $g : [a, b] \to (-\infty, +\infty)$. Define $\hbar : \mathcal{Q}^3 \times (0, +\infty) \to [0, +\infty)$ by

$$\hbar_{\lambda}(\ell,\ell',\ell'') = \frac{\mid \ell - \ell' \mid + \mid \ell' - z \mid}{\lambda}$$

for all $\ell, \ell', \ell'' \in \mathcal{Q}$ and $\lambda > 0$. Now, we define the mappings $\alpha_{\lambda}, \beta_{\lambda}, \gamma_{\lambda} : (0, \infty) \times \mathcal{Q}^3 \to [1, \infty)$ by

$$\alpha_{\lambda}(\ell, \ell', \ell'') = \beta_{\lambda}(\ell, \ell', \ell'') = \gamma_{\lambda}(\ell, \ell', \ell'') = |\ell| + |\ell'| + |\ell''| + 1, \ t > 0$$

Then (\mathcal{Q}, \hbar) is a complete TCMhM space. Now, we consider the following assumption:

for all $\ell, \ell' \in \mathcal{Q}$, assume that the following condition holds:

$$\begin{split} \frac{|f(s,t,\ell(s)) - f(s,t,\ell'(s))|}{\lambda} \\ & \frac{1}{\int_a^b f(t,s,\ell(s))ds + g(t) + 2\int_a^b f(t,s,\ell'(s))ds + 2g(t) + \lambda + 1}}{2\int_a^b f(t,s,\ell(s))ds + 2g(t) + \int_a^b f(t,s,\ell'(s))ds + g(t) + \lambda + 1}}{2(b-a)} \\ & \times F^{-1} \left[F\left(\psi\left(\frac{|\ell - \ell'|}{\lambda}\right)\right) - \tau \right]. \end{split}$$

where we have the following theorem.

Theorem 6. Suppose that above assumptions hold. Then the integral equation (26) has a unique solution in Q.

Proof. We define $L : \mathcal{Q} \to \mathcal{Q}$ by

$$\mathcal{L}(\ell)(t) = \int_{a}^{b} f(t, s, \ell(s)) ds + g(t), \quad \forall s, t \in [a, b].$$

Then, for every $\ell, \ell' \in \mathcal{Q}$ and $\lambda > 0$, we have

$$\begin{split} \hbar_{\lambda}(\mathbf{L}\ell,\mathbf{L}\ell',\mathbf{L}\ell) &= \frac{|\mathbf{L}\ell-\mathbf{L}\ell'|+|\mathbf{L}\ell-\mathbf{L}\ell'|}{\lambda} \\ &= \frac{2}{\lambda} \left| \int_{a}^{b} f(t,s,\ell(s)) - f(t,s,\ell'(s)) ds \right| \\ &\leq \frac{2}{\lambda} \int_{a}^{b} |f(t,s,\ell(s)) - f(t,s,\ell'(s))| ds \\ &\leq \beta_{\lambda}^{-1}(\mathbf{L}\ell',\mathbf{L}\ell,\mathbf{L}\ell')\beta_{\lambda}^{-1}(\mathbf{L}\ell,\mathbf{L}\ell',\mathbf{L}\ell)F^{-1} \\ &\times \left[F\left(\psi\left(\frac{2|\ell-\ell'|}{\lambda}\right)\right) - \tau \right] \right) \\ &= \beta_{\lambda}^{-1}(\mathbf{L}\ell',\mathbf{L}\ell,\mathbf{L}\ell')\beta_{\lambda}^{-1}(\mathbf{L}\ell,\mathbf{L}\ell',\mathbf{L}\ell)F^{-1} \\ &\times \left[F(\psi(\hbar_{\lambda}(\ell,\ell',\ell))) - \tau \right] \\ &\leq \beta_{\lambda}^{-1}(\mathbf{L}\ell',\mathbf{L}\ell,\mathbf{L}\ell')\beta_{\lambda}^{-1}(\mathbf{L}\ell,\mathbf{L}\ell',\mathbf{L}\ell)\eta_{\lambda}^{-1}(\ell,\ell',\ell)F^{-1} \\ &\times \left[F(\psi(\mathcal{M}(\ell,\ell',\ell))) - \tau \right] \end{split}$$

for $\lambda > 0$. This implies that

$$F(\beta_{\lambda}(\mathrm{L}\ell',\mathrm{L}\ell,\mathrm{L}\ell')\beta_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell)\eta_{\lambda}(\ell,\ell',\ell)\hbar_{\lambda}(\mathrm{L}\ell,\mathrm{L}\ell',\mathrm{L}\ell))$$

$$\leq F(\psi(\mathcal{M}(\ell,\ell',\ell))) - \tau,$$

Therefore, it is obvious that the mapping L estimates all the conditions of Theorem 1. Hence, L has a unique fixed point, i.e. the Fredholm integral equation (26) has a unique solution. \Box

Acknowledgments

We thank the editors for their kind support. We are also grateful to the learned referees for the useful suggestions which helped us in improving the manuscript.

References

- S. Hadi Bonab, R. Abazari, A. Bagheri Vakilabad, and H. Hosseinzadeh, Generalized metric spaces endowed with vector-valued metrics and matrix equations by tripled fixed point theorems, *J. Inequal. Appl.* Paper No. 204, 16 (2020).
- A.-D. Filip and A. Petruşel, Fixed point theorems on spaces endowed with vector-valued metrics, *Fixed Point Theory Appl.* Art. ID 281381, 15 (2010).
- M. N. A. H. Abdeljawad, T. and N. Souayah, Double controlled metric type spaces and some fixed point results, *Mathematics*. 6(12), 1–10 (2018).
- J. R. Roshan, V. Parvaneh, Z. Kadelburg, and N. Hussain, New fixed point results in b-rectangular metric spaces, Nonlinear Anal. Model. Control. 21(5), 614–634 (2016).
- T. Stephen, Y. Rohen, N. Mlaiki, M. Bina, N. Hussain, and D. Rizk, On fixed points of rational contractions in generalized parametric metric and fuzzy metric spaces, *J. Inequal. Appl.* Paper No. 125, 15 (2021).
- V. Parvaneh, S. Hadi Bonab, H. Hosseinzadeh, and H. Aydi, A tripled fixed point theorem in C*-algebra-valued metric spaces and application in integral equations, Adv. Math. Phys. Art. ID 5511746, 6 (2021).
- S. Banach, Sur les opérations dans les ensembles abstraits et leur application aux équations intégrales, *Fund. Math.* 3, 133–181 (1922).
- P. Debnath, M. Neog, and S. Radenović, Set valued Reich type G-contractions in a complete metric space with graph, *Rend. Circ. Mat. Palermo* (2). 69(3), 917–924 (2020).
- P. Debnath and H. M. Srivastava, New extensions of Kannan's and Reich's fixed point theorems for multivalued maps using Wardowski's technique with application to integral equations, *Symmetry.* 12(7), 1090 (2020).
- 10. P. Debnath and M. de La Sen, Fixed points of eventually δ -restrictive and $\delta(\varepsilon)$ -restrictive set-valued maps in metric spaces, *Symmetry.* **12**(1), 127 (2020).
- P. Debnath and H. M. Srivastava, Global optimization and common best proximity points for some multivalued contractive pairs of mappings, Axioms. 9(3), 102 (2020).
- 12. H. H. Alsulami, S. Chandok, M.-A. Taoudi, and I. M. Erhan, Some fixed point theorems for (α, ψ) -rational type contractive mappings, *Fixed Point Theory Appl.* **97**, 12 (2015).
- 13. P. Salimi, A. Latif, and N. Hussain, Modified α - ψ -contractive mappings with applications, *Fixed Point Theory Appl.* **151**, 19 (2013).
- 14. S. M. Kamran, T. and Q. U. Ain, A generalization of *b*-metric space and some fixed point theorems, *Mathematics*. **5**(19) (2017).
- N. Hussain, S. Khaleghizadeh, P. Salimi, and A. A. N. Abdou, A new approach to fixed point results in triangular intuitionistic fuzzy metric spaces, *Abstr. Appl. Anal.* Art. ID 690139, 16 (2014).
- M. E. Ege and C. Alaca, Some results for modular b-metric spaces and an application to system of linear equations, Azerb. J. Math. 8(1), 3–14 (2018).
- 17. D. Wardowski, Fixed points of a new type of contractive mappings in complete metric spaces, *Fixed Point Theory Appl.* **94**, 6 (2012).
- 18. B. Samet, C. Vetro, and P. Vetro, Fixed point theorems for α - ψ contractive type mappings, *Nonlinear Anal.* **75**(4), 2154–2165 (2012).
- B. Schweizer and A. Sklar, Statistical metric spaces, *Pacific J. Math.* 10, 313–334 (1960).
- V. Parvaneh, N. Hussain, M. A. Kutbi, and M. Khorshidi, Some fixed point results in extended parametric b-metric spaces with application to integral equations, *J. Math. Anal.* 10(5), 14–33 (2019).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0013

Chapter 13

Sufficient Conditions for Mittag-Leffler Function Associated with Conic Regions

Amit Soni^{*,‡} and Deepak Bansal^{†,§}

*Department of Mathematics, Govt. Engineering College, Bikaner 334004, Rajasthan, India [†]Department of Mathematics, University College of Engg. and Technology, Bikaner 334004, Rajasthan, India [‡]aamitt1981@gmail.com [§]deepakbansal_79@yahoo.com

The main purpose of this chapter is to find sufficient conditions under which the convolution of two functions

$$\mathbb{L}_{p,q}(z) * g(z) = \Gamma(q) \, z E_{p,q}(z) = \sum_{k=0}^{\infty} \frac{\Gamma(q)}{\Gamma(pk+q)} b_k z^{k+1},$$

(where $E_{p,q}(z)$ denotes the well-known Mittag-Leffler function with parameters p and q) belongs to μ -uniformly convex class of type δ , which is denoted by $UCV(\mu, \delta)$, μ -uniformly starlike function of type δ is denoted by $S_p(\mu, \delta)$ and classes S^*_{λ} and C_{λ} defined by Ponnusamy and Rønning.¹

1. Introduction

Let a class of normalized holomorphic function in the unit disc $\mathbb{U} = \{z \in \mathbb{C} : |z| < 1\}$ be denoted by \mathcal{A} and have Taylors series expansion of the form

$$g(z) = z + \sum_{k=2}^{\infty} b_k z^k.$$
(1)

The class S is the well-known subclass of A including those functions which are also univalent in \mathbb{U} . For each fixed $g \in S$ and $k \geq 2$, De-Branges² has shown that

$$|b_k| \le k. \tag{2}$$

A function $g \in \mathcal{A}$ is said to be starlike if $g(\mathbb{U})$ is starlike w.r.t. origin, and if $g(\mathbb{U})$ is convex domain, it is said to be convex. These classes are subsets of \mathcal{A} and are denoted by \mathcal{S}^* and \mathcal{C} , respectively. The classes $\mathcal{S}^*(\mu)$ and $\mathcal{C}(\mu)$ consist of those functions which are starlike and convex of order μ , $0 \leq \mu < 1$. These generalized classes are defined by holomorphic characterization $\Re(z\xi'(z)/\xi(z)) > \mu$ and $\Re(z\xi''(z)/\xi'(z)+1) > \mu$ for $z \in \mathbb{U}$ and $\xi(z) \in \mathcal{A}$. It can be easily seen that $\mathcal{S}^*(0) = \mathcal{S}^*$ and $\mathcal{C}^* = \mathcal{C}$. The detailed study of classes $\mathcal{C}(\mu)$ and $\mathcal{S}^*(\mu)$ are done by Robertson³ (see also Ref. 4).

A function $\zeta(z) \in \mathcal{A}$ is said to be uniformly convex in \mathbb{U} if it has the condition that the image arc $\zeta(\gamma)$ is a convex arc for every circular arc γ included in the unit disc \mathbb{U} , with center λ (also in \mathbb{U}). Goodman⁵ first introduced this class. He introduced the uniform starlike functions in another study.⁶

Bharti $et~al.^7$ introduced the subclasses $UCV(\mu,\delta)$ with $\mu\geq 0$ and $0\leq \delta<1$ as

$$UCV(\mu, \delta) := \left\{ \zeta \in \mathcal{A} : \Re \left\{ \frac{z\zeta''(z)}{\zeta'(z)} + 1 \right\} \\ \ge \mu \left| \frac{z\zeta''(z)}{\zeta'(z)} \right| + \delta, z \in \mathbb{U} \right\}.$$
(3)

The class $S_p(\mu, \delta)$ was also discussed in Ref. 7, as

$$S_{p}(\mu, \delta) := \left\{ \zeta \in \mathcal{A} : \Re \left\{ \frac{z\zeta'(z)}{\zeta(z)} + 1 \right\} \\ \geq \mu \left| \frac{z\zeta'(z)}{\zeta(z)} \right| + \delta, z \in \mathbb{U} \right\}.$$

$$(4)$$

It is clear to observe that

$$\zeta(z) \in UCV(\mu, \delta) \Leftrightarrow z\zeta'(z) \in \mathcal{S}_p(\mu, \delta).$$

Further, $UCV(\mu, 0) = \mu - UCV$ and $S_p(\mu, 0) = \mu - S_p$, where $\mu - UCV$ is called class of μ -uniformly convex function and $\mu - S_p$ is called class of μ -starlike function in U. The class $\mu - UCV$ was introduced by Kanas and Wisniowska⁸ and the class $\mu - S_p$ was investigated in Ref. 9. In particular, when $\mu = 1$, we obtain $1 - UCV \equiv UCV$ and $1 - S_p \equiv S_p$, where UCV and S_p are well-known classes of uniformly convex functions and parabolic starlike functions in the unit disc U, respectively (see works of Goodman,^{5,6} Ma and Minda,¹⁰ and Rønning,¹¹ and also Refs.¹²⁻¹⁹).

A function $g \in \mathcal{A}$ is said to be in the class $P_m^{\tau}(n)$ if it satisfies the inequality

$$\left|\frac{(1-m)\frac{g(z)}{z} + mg'(z) - 1}{2\tau(1-n) + (1-m)\frac{g(z)}{z} + mg'(z) - 1}\right| < 1,$$
(5)

where $0 \leq m \leq 1$, n < 1 and $\tau \in \mathbb{C} \setminus \{0\}$. The class $P_m^{\tau}(n)$ was introduced by Swaminathan.²⁰ The importance of this class lies in the fact that for m = 1, the class $P_1^{\tau}(n)$ reduces in the class studied by Gangadharan *et al.*²¹ Further, for $\tau = e^{i\xi} \cos(\xi)$ where $\xi \in (-\pi/2, \pi/2)$, class $P_m^{\tau}(n)$ reduces in the class studied earlier by Kim and Srivastava.²² Throughout this Chapter, we will take $\tau = e^{i\xi} \cos(\xi)$.

Next, we consider the classes

$$\mathcal{S}_{\lambda}^{*} = \left\{ g \in \mathcal{A} : \left| \frac{zg'(z)}{g(z)} - 1 \right| < \lambda, (z \in \mathbb{U}, \lambda > 0) \right\}$$
(6)

and

$$\mathcal{C}_{\lambda} = \left\{ g \in \mathcal{A} : \left| \frac{z g''(z)}{g'(z)} \right| < \lambda, (z \in \mathbb{U}, \lambda > 0) \right\},\tag{7}$$

obviously,

$$f(z) \in \mathcal{C}_{\lambda} \iff zf'(z) \in \mathcal{S}_{\lambda}^*, \quad \lambda > 0.$$

The classes \mathcal{S}^*_{λ} and C_{λ} were introduced by Ponnusamy and Rønning.¹

The function $E_p(z)$ defined by

$$E_p(z) = \sum_{k=0}^{\infty} \frac{z^k}{\Gamma(kp+1)} \quad (z \in \mathbb{C}, \, \Re(p) > 0)$$
(8)

was introduced by Mittag-Leffler.²³ A more general function $E_{p,q}(z)$ is defined by

$$E_{p,q}(z) = \sum_{k=0}^{\infty} \frac{z^k}{\Gamma(kp+q)} \quad (z, p, q \in \mathbb{C}; \, \Re(p) > 0). \tag{9}$$

This function, sometimes called a Mittag-Leffler type function, first appeared in a paper by Wiman. $^{24,\,25}$

The importance of the Mittag-Leffler function was rediscovered when its connection to fractional calculus was fully understood. The Mittag-Leffler function appears in the solution of fractional order integral and differential equations, especially in the study of fractional generalization of kinetic equation, modeling fractional order viscoelastic materials, study of certain deterministic models, stochastic models, super-diffusive transport and in the study of complex systems. The most essential properties of these entire functions can be found in Refs. 26–31.

Note that Mittag-Leffler function $E_{p,q}$ does not belong to the family \mathcal{A} . Thus, it is obvious to consider the following normalization of Mittag-Leffler function:

$$\mathbb{L}_{p,q}(z) = \Gamma(q) \, z E_{p,q}(z) \tag{10}$$
$$= \sum_{k=0}^{\infty} \frac{\Gamma(q)}{\Gamma(pk+q)} z^{k+1} \ (z,p,q \in \mathbb{C}; \, Re(p) > 0, q \neq 0, -1, \ldots).$$

Formula (10) holds for $p, q, z \in \mathbb{C}$, but in this chapter, we restrict ourselves to the case where $p, q \in \mathbb{R}$ and $z \in \mathbb{U}$. It is easy to note that $\mathbb{L}_{p,q}(z)$ contains the following well-known functions as its special case, e.g.

$$\begin{cases} \mathbb{L}_{0,1}(z) = \frac{z}{1-z}, \ \mathbb{L}_{1,1}(z) = z \ \exp(z), \ \mathbb{L}_{2,1}(z) = z \ \cosh(\sqrt{z}), \\ \mathbb{L}_{1,2}(z) = \exp(z) - 1, \ \mathbb{L}_{1,3}(z) = \frac{2(\exp(z) - z - 1)}{z}, \\ \mathbb{L}_{1,4}(z) = \frac{6(\exp(z) - 1 - z) - 3z^2}{z^2}, \ \mathbb{L}_{2,2}(z) = \sqrt{z} \ \sinh(\sqrt{z}). \end{cases}$$
(11)

If $g, h \in \mathcal{A}$, where g(z) is given by (1) and h(z) is defined by

$$h(z) = z + \sum_{k=2}^{\infty} c_k \ z^k,$$

then their convolution (Hadamard product) g * h is defined by

$$(g * h)(z) = z + \sum_{k=2}^{\infty} b_k c_k \ z^k \ (z \in \mathbb{U}).$$

In view of the above definition, we have

$$\mathbb{L}_{p,q}(z) * g(z) = z + \sum_{k=2}^{\infty} \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} b_k z^k$$
$$= z + \sum_{k=2}^{\infty} t_k b_k z^k \left(\text{where } t_k = \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right).$$
(12)

Recently, several mathematicians have studied some well-known special functions (from geometric function theory point of view) and obtained sufficient conditions such that these special functions have certain geometric properties such as close-to-convexity, convexity, starlikeness, etc. in \mathbb{U} . In this context, many results are available in the literature for different special functions. One can refer to works regarding the hypergeometric functions in Refs. 32–34, Bessel function in Refs. 35 and 36, Wright function in Ref. 37, Mittag-Leffler function in Ref. 38 and Mathieu-type power series in Ref. 39.

Finding the relation between various classes of holomorphic functions is an interesting research problem and has contributed many results in the past. We are particularly interested in the following problem.

Problem 1. For a class of holomorphic function $f \in P_m^{\tau}(n)$, find sufficient conditions such that the $\mathbb{L}_{p,q}(z) * g(z)$ belongs to the classes $UCV(\mu, \delta), S_p(\mu, \delta), S_{\lambda}^*$ and C_{λ} . Here, $\mathbb{L}_{p,q}(z)$ is the Mittag-Leffler operator of the form (10).

One can refer to the works of Dixit and Porwal,¹³ Gangadharan et al.,²¹ Din and Yalcin,⁴⁰ Porwal and Moin⁴¹ and references therein, which have earlier solved the similar Problem 1 for different linear operator (other than the Mittag-Leffler function).

2. Preliminaries

The following lemmas will be required to prove our key findings.

Lemma 1 (see Ref. 7). Let $g \in A$. If the inequality

$$\sum_{k=2}^{\infty} \left[(1+\mu)k - (\mu+\delta) \right] k |b_k| \le 1 - \delta$$
 (1)

holds, then $g \in UCV(\mu, \delta)$.

Lemma 2 (see Ref. 7). Let $g \in A$. If the inequality

$$\sum_{k=2}^{\infty} \left[k(1+\mu) - (\mu+\delta) \right] |b_k| \le 1-\delta$$
 (2)

holds, then $g \in S_p(\mu, \delta)$.

Lemma 3 (see Ref. 20). Let $g(z) \in S$ and it belongs to class given in (1). If $g \in P_m^{\tau}(n)$, then

$$|b_k| \le \frac{2|\tau|(1-n)}{1+m(k-1)}.$$
(3)

Lemma 4 (see Ref. 21). Let $g \in A$. If the inequality

$$\sum_{k=2}^{\infty} \left(\lambda + k - 1\right) |b_k| \le \lambda, \ \lambda > 0, \tag{4}$$

holds, then $g \in \mathcal{S}^*_{\lambda}$.

Lemma 5 (see Ref. 21). Let $g \in A$. If the inequality

$$\sum_{k=2}^{\infty} k \left(\lambda + k - 1\right) \left|b_k\right| \le \lambda, \ \lambda > 0, \tag{5}$$

holds, then $g \in C_{\lambda}$.

3. Main Results

The first theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in UCV(\mu, \delta)$, whenever g(z) of Eq. (1) belongs to the class $P_m^{\tau}(n)$.

Theorem 1. Let $g \in P_m^{\tau}(n) \cap S$ be of the form (1) with $\tau = e^{i\xi}\cos(\xi)$ $(\xi \in (-\pi/2, \pi/2)), 0 < m \leq 1, n < 1$ and p, q > 0. If

$$\frac{2(1-n)\cos\xi}{m} \left(\frac{(q+1)\left((1+\mu)(q+1)+(1-\delta)q\right)}{q^3}\right) \le (1-\delta),$$

then $\mathbb{L}_{p,q}(z) * g(z) \in UCV(\mu, \delta).$

Proof. Using Lemma 1 on (12), the convolution operator $\mathbb{L}_{p,q}(z) * g(z) \in UCV(\mu, \delta)$, provided that

$$\sum_{k=2}^{\infty} k \left[k(1+\mu) - (\mu+\delta) \right] |t_k| \le 1 - \delta.$$
 (6)

Using Lemma 3 with $\tau = e^{i\xi} \cos(\xi)$, we have

$$\begin{split} &\sum_{k=2}^{\infty} k \left[k(1+\mu) - (\mu+\delta) \right] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \left| b_k \right| \\ &\leq 2(1-n) \cos \xi \sum_{k=2}^{\infty} k \left[k(1+\mu) - (\mu+\delta) \right] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \\ &\cdot \frac{1}{1+m(k-1)} \\ &\leq \frac{2(1-n) \cos \xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \frac{k\Gamma(q)}{\Gamma(p(k-1)+q)} \right. \\ &\left. - (\mu+\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right\} \end{split}$$

$$\begin{split} &= \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{(k-1+1)\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right\} \\ &= \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{(k-1)\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \\ &+ (1-\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right\} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \frac{(k-1)}{(q)_{k-1}} + (1-\delta) \sum_{k=2}^{\infty} \frac{1}{(q)_{k-1}} \right\} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \frac{(1+\mu)}{q} \left(1 - \frac{1}{(q+1)} \right)^{-2} \\ &+ \frac{(1-\delta)}{q} \left(1 - \frac{1}{(q+1)} \right)^{-1} \right\} \\ &= \frac{2(1-n)\cos\xi}{m} \frac{(q+1)\left((1+\mu)(q+1) + (1-\delta)q\right)}{q^3} \\ &\leq (1-\delta), \end{split}$$

using the inequality

$$\frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \le \frac{1}{(q)_{k-1}} \le \frac{1}{q(q+1)^{k-2}} \ (\forall \ k \ge 2)$$
(7)

and by the given hypothesis. This completes the proof of Theorem 1. $\hfill \Box$

The second theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in S_p(\mu, \delta)$, whenever g(z) of the form (1) belongs to the class $P_m^{\tau}(n)$.

Theorem 2. Let $g \in P_m^{\tau}(n) \cap S$ be of the form (1) with $0 < m \le 1, n < 1$ and p, q > 0. If

$$\frac{2(1-n)\cos\xi}{m} \left\{ \frac{(q+1)}{2q^2} (3+2\mu-\delta) \right\} \le (1-\delta),$$

then $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{S}_p(\mu, \delta)$.

Proof. Using Lemma 2 on (12), the convolution operator $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{S}_p(\mu, \delta)$, provided that

$$\sum_{k=2}^{\infty} [k(1+\mu) - (\mu+\delta)] |t_k| \le 1 - \delta.$$

Using Lemma 3 with $\tau = e^{i\xi} \cos(\xi)$, we have

$$\begin{split} &\sum_{k=2}^{\infty} \left[k(1+\mu) - (\mu+\delta) \right] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \left| b_k \right| \\ &\leq 2(1-n)\cos\xi \sum_{k=2}^{\infty} \left[k(1+\mu) - (\mu+\delta) \right] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \\ &\cdot \frac{1}{1+m(k-1)} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \frac{\Gamma q}{k\Gamma(p(k-1)+q)} \right\} \\ &= \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{(k-1+1)\Gamma(q)}{k\Gamma(p(k-1)+q)} \right) \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{k\Gamma(p(k-1)+q)} \right\} \end{split}$$

$$= \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{(k-1)\Gamma(q)}{k\Gamma(p(k-1)+q)} \right) + (1-\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{k\Gamma(p(k-1)+q)} \right\}$$

$$\leq \frac{2(1-n)\cos\xi}{m} \left\{ (1+\mu) \sum_{k=2}^{\infty} \left(\frac{k-1}{k} \right) \frac{1}{(q)_{k-1}} + (1-\delta) \sum_{k=2}^{\infty} \left(\frac{1}{k} \right) \frac{1}{(q)_{k-1}} \right\}$$

$$\leq \frac{2(1-n)\cos\xi}{m} \left\{ \frac{1+\mu}{q} + \frac{(1-\delta)}{2q} \right\} \left(1 - \frac{1}{q+1} \right)^{-1}$$

$$= \frac{2(1-n)\cos\xi}{m} \left\{ \frac{(q+1)}{2q^2} (3+2\mu-\delta) \right\} \leq (1-\delta),$$

using the inequality (7) and by the given hypothesis. This completes the proof of Theorem 2. $\hfill \Box$

The third theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in S^*_{\lambda}$, whenever g(z) of the form (1) belongs to the class $P^{\tau}_m(n)$.

Theorem 3. Let $g \in P_m^{\tau}(n) \cap S$ be of the form (1) with $0 < m \leq 1$, n < 1 and p, q > 0. If

$$\frac{(1-n)\cos\xi}{m}\left\{\frac{(q+1)\left(\lambda+1\right)}{q^2}\right\} \le \lambda,$$

then $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{S}^*_{\lambda}$.

Proof. In view of Lemma 4 and (12), it is sufficient to show that

$$\sum_{k=2}^{\infty} \left(\lambda + k - 1\right) |t_k| \le \lambda.$$

Now, using Lemma 3 with $\tau = e^{i\xi} \cos(\xi)$, we have

$$\begin{split} &\sum_{k=2}^{\infty} \left(\lambda + k - 1\right) |t_k| \\ &= \sum_{k=2}^{\infty} \left(\lambda + k - 1\right) \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} |b_k| \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \sum_{k=2}^{\infty} \left(\frac{\Gamma(q)}{\Gamma(p(k-1)+q)}\right) \right\} \\ &+ (\lambda - 1) \sum_{k=2}^{\infty} \left(\frac{\Gamma(q)}{k\Gamma p(k-1)+q}\right) \right\} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \sum_{k=2}^{\infty} \frac{1}{(q)_{k-1}} + (\lambda - 1) \sum_{k=2}^{\infty} \frac{1}{k(q)_{k-1}} \right\} \\ &\leq \frac{(1-n)\cos\xi}{m} \left\{ \frac{(q+1)(\lambda+1)}{q^2} \right\} \leq \lambda, \end{split}$$

using the inequality (7) and by the given hypothesis. This completes the proof of Theorem 3. $\hfill \Box$

The fourth theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in C_{\lambda}$, whenever g(z) of the form (1) belongs to the class $P_m^{\tau}(n)$.

Theorem 4. Let $g \in P_m^{\tau}(n) \cap S$ be of the form (1) with $0 < m \leq 1, n < 1$ and p, q > 0. If

$$\frac{2(1-n)\cos\xi}{m}\left\{\frac{(q+1)(q(1+\lambda)+1)}{q^3}\right\} \le \lambda,$$

then $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{C}_{\lambda}$.

Proof. In view of Lemma 5 and (12), it is sufficient to show that

$$\sum_{k=2}^{\infty} k \left(\lambda + k - 1\right) |t_k| \le \lambda.$$

Now, using Lemma 3 with $\tau = e^{i\xi} \cos(\xi)$, we have

$$\begin{split} &\sum_{k=2}^{\infty} k\left(\lambda+k-1\right) |t_k| \\ &= \sum_{k=2}^{\infty} k\left(\lambda+k-1\right) \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} |b_k| \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \sum_{k=2}^{\infty} \left(\frac{k\Gamma(q)}{\Gamma(p(k-1)+q)}\right) \right\} \\ &+ (\lambda-1) \sum_{k=2}^{\infty} \left(\frac{\Gamma(q)}{\Gamma(p(k-1)+q)}\right) \right\} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \sum_{k=2}^{\infty} \frac{(k-1)}{(q)_{k-1}} + \lambda \sum_{k=2}^{\infty} \frac{1}{(q)_{k-1}} \right\} \\ &\leq \frac{2(1-n)\cos\xi}{m} \left\{ \frac{(q+1)(q(1+\lambda)+1)}{q^3} \right\} \leq \lambda \end{split}$$

by the given hypothesis. This completes proof of Theorem 4. \Box

The fifth theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{S}_p(\mu, \delta)$, whenever g(z) of the form (1) belongs to the class \mathcal{S} .

Theorem 5. Let p > 0, q > 1 satisfy the inequality

$$\left[\frac{4(1+\mu)(q+1)}{q(q-1)} + \frac{(2+\mu-\delta)(q+1)^2}{q^3} + \frac{(1-\mu)(q+1)}{q^2}\right] \le (1-\delta).$$

Then the convolution $\mathbb{L}_{p,q}(z) * g(z)$ maps $g(z) \in S$ into $S_p(\mu, \delta)$.

Proof. Let $g(z) \in S$. Using Lemma 1, it is sufficient to show that

$$\sum_{k=2}^{\infty} [k(1+\mu) - (\mu+\delta)] |t_k| \le 1 - \delta.$$

Using (2), we have

$$\begin{split} &= \sum_{k=2}^{\infty} \left[k(1+\mu) - (\mu+\delta) \right] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \left| b_k \right| \\ &= (1+\mu) \sum_{k=2}^{\infty} \left(\frac{k\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \left| b_k \right| \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \left(\frac{\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \left| b_k \right| \\ &\leq (1+\mu) \sum_{k=2}^{\infty} \left(\frac{k^2 \Gamma(q)}{\Gamma(p(k-1)+q)} \right) \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \left(\frac{k\Gamma(q)}{\Gamma(p(k-1)+q)} \right) \\ &= (1+\mu) \sum_{k=2}^{\infty} \frac{((k-1)^2 + 2(k-1) + 1)\Gamma(q)}{\Gamma(p(k-1)+q)} \\ &- (\mu+\delta) \sum_{k=2}^{\infty} \frac{(k-1+1)\Gamma(q)}{\Gamma(p(k-1)+q)} \\ &= (1+\mu) \sum_{k=2}^{\infty} \frac{(k-1)^2 \Gamma(q)}{\Gamma(p(k-1)+q)} + (2+\mu-\delta) \sum_{k=2}^{\infty} \frac{(k-1)\Gamma(q)}{\Gamma(p(k-1)+q)} \\ &+ (1-\delta) \sum_{k=2}^{\infty} \frac{\Gamma(q)}{\Gamma(p(k-1)+q)}. \end{split}$$

Now, using the inequality $(k-1)^2 \leq 2^k \ \forall k \geq 2$ and inequality (7), we have

$$\sum_{k=2}^{\infty} [k(1+\mu) - (\mu+\delta)] \frac{\Gamma(q)}{\Gamma(p(k-1)+q)} |b_k|$$

$$\leq \frac{(1+\mu)2^2}{q} \sum_{k=0}^{\infty} \left(\frac{2}{q+1}\right)^k + \frac{(2+\mu-\delta)}{q} \sum_{k=0}^{\infty} \left(\frac{k+1}{(q+1)^k}\right)$$

$$+ \frac{(1-\delta)}{q} \left(\sum_{k=0}^{\infty} \frac{1}{q+1}\right)^k$$

$$= \frac{4(1+\mu)(q+1)}{q(q-1)} + \frac{(2+\mu-\delta)(q+1)^2}{q^3} + \frac{(1-\mu)(q+1)}{q^2}$$

$$\leq 1-\delta$$

by the given hypothesis. This completes the proof of Theorem 5. \Box

The sixth theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{S}^*_{\lambda}$, whenever g(z) of the form (1) belongs to the class \mathcal{S} .

Theorem 6. Let p > 0, q > 1 satisfy the inequality

$$\frac{(q+1)}{q^3(q-1)} \left(q^2(2\lambda+5) - q\lambda - (\lambda+1) \right) \le \lambda.$$

Then $\mathbb{L}_{p,q}(z) * g(z)$ maps $g(z) \in \mathcal{S}$ into \mathcal{S}^*_{λ} .

Proof. The proof of Theorem 6 is similar to the proof of Theorem 5, therefore we omit the details. \Box

The seventh theorem provides sufficient condition, so that $\mathbb{L}_{p,q}(z) * g(z) \in \mathcal{C}_{\lambda}$, whenever g(z) of the form (1) belongs to the class \mathcal{S} .

Theorem 7. Let p > 0, q > 2 satisfy the inequality

$$\frac{(q+1)}{q^3(q-1)(q-2)} \left(q^3(7\lambda+2) - 3q^2(5\lambda-2) + 4(\lambda-1) \right) \le \lambda.$$

Then $\mathbb{L}_{p,q}(z) * g(z)$ maps $g(z) \in \mathcal{S}$ into \mathcal{C}_{λ} .

Proof. Using the inequality $k^3 \leq 3^k \ \forall k \geq 2$ and proceeding similar to the proof of Theorem 5, we obtain the required result. \Box

4. Conclusion

The importance of the results discussed in Section 3 lies in the fact that when we give particular values to various parameters in (12), we obtain the operators which transform functions belonging to class $P_m^{\tau}(n)$ into the functions belonging to class $UCV(\mu, \delta)$, $S_p(\mu, \delta)$, S_{λ} and C_{λ} .

Acknowledgments

The authors acknowledge the sincere efforts of the anonymous reviewers.

References

- S. Ponnusamy and F. Rønning, Starlikeness properties for convolutions involving hypergeometric series, Ann. Univ. Mariae Curie-Sklodowska Sect. A. 52(1), 141–155 (1998).
- L. De Branges, A proof of the bieberbach conjecture, Acta Math. 154 (1), 137–152 (1985).
- M. I. Robertson, On the theory of univalent functions, Ann. Math. pp. 374–408 (1936).
- H. Silverman, Univalent functions with negative coefficients, Proc. Am. Math. Soc. 51(1), 109–116 (1975).
- A. W. Goodman. On uniformly convex functions. Ann. Pol. Math., vol. 56, pp. 87–92 (1991).
- A. W. Goodman, On uniformly starlike functions, J. Math. Anal. Appl. 155(2), 364–370 (1991).
- R. Bharati, R. Parvatham, and A. Swaminathan, On subclasses of uniformly convex functions and corresponding class of starlike functions, *Tamkang J. Math.* 28(1), 17–32 (1997).
- S. Kanas and A. Wisniowska, Conic regions and k-uniform convexity, J. Comput. Appl. Math. 105(1-2), 327–336 (1999).
- S. Kanas and A. Wisniowska, Conic domains and starlike functions, *Rev. Roum. Math. Pures Appl.* 45(4), 647–658 (2000).
- W. Ma and D. Minda.Uniformly convex functions. Ann. Pol. Math. 57, 165–175 (1992).
- F. Rønning, Uniformly convex functions and a corresponding class of starlike functions, *Proc. Am. Math. Soc.* 118(1), 189–196 (1993).
- D. Bansal, M. K. Soni, and A. Soni, Certain geometric properties of the modified dini function, Anal. Math. Phys. 9(3), 1383–1392 (2019).
- K. Dixit and S. Porwal, On a certain class of k-uniformly convex functions with negative coefficients, *Bull. Cal. Math. Soc.* 100(6), 639–652 (2008).
- B. Frasin, Comprehensive family of uniformly analytic functions, Tamkang J. Math. 36(3), 243–254 (2005).
- S. Hussain, A. Rasheed, M. A. Zaighum, and M. Darus, A subclass of analytic functions related to-uniformly convex and starlike functions, *J. Funct. Spaces.* **2017** (2017).

- S. Kanas, Differential subordination related to conic sections, J. Math. Anal. Appl. 317(2), 650–658 (2006).
- S. Kanas and T. Sugawa. On conformal representations of the interior of an ellipse. Ann. Acad. Sci. Fenn. Math. 31, 329 (2006).
- M. K. Soni, A. Soni, and D. Bansal, Some geometric properties of analytic series whose coefficients are reciprocal of Fuss-Catalan numbers, *Electron. J. Math. Anal. Appl.* 6(2), 246–254 (2018).
- S. Kanas and H. M. Srivastava, Linear operators associated with kuniformly convex functions, *Integral Transforms Special Funct.* 9(2), 121–132 (2000).
- A. Swaminathan, Certain sufficiency conditions on gaussian hypergeometric functions, J. Inequal. Pure Appl. Math. 5(4), 1–10 (2004).
- A. Gangadharan, T. N. Shanmugam, and H. M. Srivastava, Generalized hypergeometric functions associated with k-uniformly convex functions, *Comput. & Math. Appl.* 44(12), 1515–1526 (2002).
- 22. Y. C. Kim and H. Srivastava, Fractional integral and other linear operators associated with the gaussian hypergeometric function, *Complex Variables Ellipt. Equat.* **34**(3), 293–312 (1997).
- G. M. Mittag-Leffler, Sur la nouvelle fonction eα (x), CR Acad. Sci. Paris. 137(2), 554–558 (1903).
- 24. A. Wiman, Über den fundamentalsatz in der theorie der funcktionen $e_a(x)$, Acta Math. **29**, 191–201 (1905).
- 25. A. Wiman, Über die nullstellen der funktionen $e_a(x)$, Acta Math. **29**(1), 217–234 (1905).
- A. A. Attiya, Some applications of Mittag-Leffler function in the unit disk, *Filomat.* **30**(7), 2075–2081 (2016).
- 27. M. Dzhrbashyan. Integral transforms and representations of functions in the complex domain (1966).
- A. Erdėlyi, W. Magnus, F. Oberhettinger, and F. Tricomi (Eds.). *Higher Transcendental Functions*, Vol. 1, McGraw-Hill, New York (1953), p. 162.
- B. A. Frasin, T. Al-Hawary, and F. Yousef, Some properties of a linear operator involving generalized mittag-leffler function, *Stud. Univ. Babes-Bolyai Math.* 65(1), 67–75 (2020).
- R. Hilfer, Applications of Fractional Calculus in Physics. World Scientific Publishing Company, Singapore, pp. 87–130 (2000).
- K. Lang, Astrophysical Formulae: Vol. 1: Radiation, Gas Processes and High Energy Astrophysics. Vol. 2: Space, *Time, Matter and Cosmology*. Springer, Berlin (1999).
- S. S. Miller and P. T. Mocanu, Univalence of Gaussian and confluent hypergeometric functions, *Proc. Am. Math. Soc.* 110(2), 333–342 (1990).

- S. Ponnusamy and M. Vuorinen, Univalence and convexity properties for gaussian hypergeometric functions, *Rocky Mountain J. Math.* **31**(1), 327–353 (2001).
- S. Ruscheweyh and V. Singh, On the order of starlikeness of hypergeometric functions, J. Math. Anal. Appl. 113(1), 1–11 (1986).
- 35. Á. Baricz, *Generalized Bessel Functions of the First Kind*. Springer, Heidelberg; New York (2010).
- Á. Baricz and S. Ponnusamy, Starlikeness and convexity of generalized Bessel functions, *Integral Transforms and Special Funct.* 21(9), 641–653 (2010).
- J. Prajapat, Certain geometric properties of the wright function, Integral Transforms and Special Funct. 26(3), 203-212 (2015).
- D. Bansal and J. Prajapat, Certain geometric properties of the Mittag-Leffler functions, Complex Variables Ellipt. Equations. 61(3), 338–350 (2016).
- D. Bansal and J. Sokol, Geometric properties of Mathieu-type power series inside unit disk, J. Math. Ineq. 13, 911–918 (2019).
- 40. M. U. Din and S. Yalçin, Mapping properties for conic regions associated with Wright functions, *Honam Math. J.* **42**(2), 377–389 (2020).
- S. Porwal and M. Ahmad, Some sufficient conditions for generalized Bessel functions associated with conic regions, *Vietnam J. Math.* 43(1), 163–172 (2015).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0014

Chapter 14

Results on the Existence of Solutions for Some Controlled Optimization Problems

Savin Treanță

Department of Applied Mathematics, University Politehnica of Bucharest, Academy of Romanian Scientists, 54 Splaiul Independentei, 050094 Bucharest, Romania savin.treanta@upb.ro

In this chapter, by using some generalized convexity and Fréchet differentiability hypotheses of the involved functionals, we establish several connections between the solutions of some new (weak) vector-controlled variational inequalities and (proper, weak) efficient solutions associated with certain multiobjective-controlled variational problems. Also, the notion of invex set with respect to some given functions has an important role for proving the main results derived in the chapter.

1. Introduction

The importance of multiobjective/vector optimization problems in the mathematical modelling of natural phenomena or engineering is well known. For the study of these types of problems, it was necessary to define some concepts of *efficient solutions*. Geoffrion¹ introduced a pretty narrow definition of efficiency called *proper efficient solu*tions. On the other side, Klinger² analyzed *improper solutions* for the vector maximum problem. Later, Kazmi³ investigated, by using vector variational-like inequalities, the existence of a *weak minimum* for constrained optimization problems. Moreover, Ghaznavighosoni and Khorram⁴ stated conditions of efficiency for approximating (weakly, properly) efficient points in general multiobjective optimization problems.

The mathematical concept of *convexity* is almost inevitable in optimization theory to formulate the optimality or efficiency conditions. Moreover, its generalization was necessary in the study of some concrete problems in applied sciences, engineering, or natural phenomena. In consequence, Hanson⁵ defined the notion of *invexity* and, over time, many other extensions have been introduced (see, for instance, Antczak,⁶ Arana-Jiménez *et al.*,⁷ Mishra *et al.*,⁸ Ahmad⁹), such as pseudoinvexity, univexity, quasiinvexity, preinvexity, approximate convexity, and so on. Also, some of these generalized convexities have been transposed into the multidimensional context determined by multiple or curvilinear integrals (see, for instance, Mititelu and Treanță,¹⁰ Treanță¹¹).

On the other hand, variational inequalities have been defined and studied to model concrete problems in mechanics, physics, engineering, traffic analysis, or natural phenomena. In this regard, Giannessi¹² analyzed, obtaining some remarkable results, the vector variational inequalities. As we all know, an application of scalar/vector variational inequalities is the study of existence of solutions for scalar/multiobjective optimization problems under some suitable hypotheses. Thus, many papers investigated the connections between the solutions of these inequalities and optimal or efficient solutions associated to the corresponding optimization problems (see, for instance, Ruiz-Garzón *et al.*,¹³ Marinoschi,¹⁴ Jayswal *et al.*,¹⁵).

We divide variational problems into (1) classical variational problems and (2) continuous-time variational problems. Some interesting results on multiobjective continuous-time program and vector variational inequality have been formulated by Kim.¹⁶ Also, since optimal control problems can be regarded as continuous-time variational problems, their analysis from this point of view has generated various important results in engineering problems, economics, or processes coming from game theory, operations research. Thus, quite recently, Treanță¹⁷ and Jha *et al.*¹⁸ have established some necessary and sufficient optimality or efficiency conditions, saddle-point criterion, and well-posedness results for several multidimensional control problems driven by functionals of curvilinear or multiple integral type.

In this chapter, we introduce a class of (weak) vector-controlled variational inequalities and multiobjective variational control problems defined by functionals of multiple integral type. We state certain connections between the solutions of the multidimensional variational problems under examination. The main novelty element in this mathematical framework is the presence of the control variables. Also, the notion of invex set with respect to some given functions has a decisive role in proving the main results included in the chapter.

The chapter is continued with some preliminaries. In Section 3, we state some characterization results for the solutions associated with the considered variational problems. Finally, Section 4 includes the conclusions of this chapter and further research development.

2. Problem Description

In this chapter, we begin with P as a set in \mathbb{R}^m , that is supposed to be compact, and $P \ni \varsigma = (\varsigma^{\gamma}), \gamma = \overline{1, m}$, as a multi-variable of evolution. Also, we introduce \mathbb{Q} as the space of all piecewise differentiable *state* functions $x : P \to \mathbb{R}^n$ and \mathbb{T} as the space of all *control* functions $\eta : P \to \mathbb{R}^k$, which are supposed to be piecewise continuous. In addition, on $\mathbb{Q} \times \mathbb{T}$, we define the scalar product

$$\begin{split} \langle (x,\eta),(\pi,v)\rangle &= \int_{P} [x(\varsigma) \cdot \pi(\varsigma) + \eta(\varsigma) \cdot v(\varsigma)] d\varsigma \\ &= \int_{P} \left[\sum_{i=1}^{n} x^{i}(\varsigma) \pi^{i}(\varsigma) + \sum_{j=1}^{k} \eta^{j}(\varsigma) v^{j}(\varsigma) \right] d\varsigma \\ &\quad (\forall)(x,\eta), (\pi,v) \in \mathsf{Q} \times \mathsf{T}, \end{split}$$

together with the norm induced by it.

By using the vector-valued functions $\phi = (\phi^l) : P \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^p$, $l = \overline{1, p}$, we introduce the following vector functional defined by

multiple integrals:

$$\Phi: \mathsf{Q} \times \mathsf{T} \to \mathbb{R}^p, \quad \Phi(x, \eta) = \int_P \phi\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma$$
$$= \left(\int_P \phi^1\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma, \dots, \int_P \phi^p\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma\right).$$

In the following, D_{α} , $\alpha \in \{1, \ldots, m\}$, denotes the operator of total derivative, and we assume that the Lagrange-type densities

$$\phi = (\phi^1, \dots, \phi^p) : P \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^p$$

are of C^1 -class. Also, throughout the chapter, we will use the following rules associated with equalities and inequalities:

$$\begin{split} a &= b \Leftrightarrow a^l = b^l, \; a \leq b \Leftrightarrow a^l \leq b^l, \; a < b \Leftrightarrow a^l < b^l, \\ a &\preceq b \Leftrightarrow a \leq b, \; a \neq b, \; l = \overline{1, p} \end{split}$$

for any *p*-tuples $a = (a^1, \ldots, a^p)$, $b = (b^1, \ldots, b^p)$ in \mathbb{R}^p .

Further, we introduce the following partial differential equation constrained *vector variational control problem*:

$$(P) \quad \min_{(x,\eta)} \left\{ \Phi(x,\eta) = \int_P \phi\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma \right\} \text{ subject to } (x,\eta) \in S,$$

where

$$\begin{split} \Phi(x,\eta) &= \int_{P} \phi\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma \\ &= \left(\int_{P} \phi^{1}\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma, \dots, \int_{P} \phi^{p}\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma\right) \\ &= \left(\Phi^{1}(x,\eta), \dots, \Phi^{p}(x,\eta)\right) \end{split}$$

and

$$S = \left\{ (x,\eta) \in \mathbf{Q} \times \mathbf{T} \mid x_{\alpha}^{i}(\varsigma) = \frac{\partial x^{i}}{\partial \varsigma^{\alpha}}(\varsigma) = H_{\alpha}^{i}(\varsigma, x(\varsigma), \eta(\varsigma)), \\ Y(\varsigma, x(\varsigma), \eta(\varsigma)) \leq 0, \ (x,\eta)|_{\partial P} = \text{given} \right\}.$$

In the definition of S, we have considered that the C^1 -class functions $H_{\alpha} = (H_{\alpha}^i) : P \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^n, \ i = \overline{1, n}, \ \alpha = \overline{1, m},$ define the

following partial differential equations of evolution:

$$x^i_\alpha(\varsigma) = H^i_\alpha\left(\varsigma, x(\varsigma), \eta(\varsigma)\right), \quad i = \overline{1, n}, \ \alpha = \overline{1, m}$$

and verify the closeness relations $D_{\gamma}H^{i}_{\alpha} = D_{\alpha}H^{i}_{\gamma}$, $\alpha, \gamma = \overline{1, m}$, $\alpha \neq \gamma$, $i = \overline{1, n}$. Also, $Y = (Y^{r}) : P \times \mathbb{R}^{n} \times \mathbb{R}^{k} \to \mathbb{R}^{q}$, $r = \overline{1, q}$, are assumed to be C^{1} -class functions.

Definition 1 (Mititelu and Treanță¹⁰). A point $(x^0, \eta^0) \in S$ is called an *efficient solution* in (P) if there exists no other $(x, \eta) \in S$ such that $\Phi(x, \eta) \preceq \Phi(x^0, \eta^0)$ or, equivalently, $\Phi^l(x, \eta) - \Phi^l(x^0, \eta^0) \leq 0$, $(\forall) l = \overline{1, p}$, with strict inequality for at least one l.

Definition 2 (Geoffrion¹). A point $(x^0, \eta^0) \in S$ is called a *proper* efficient solution in (P) if $(x^0, \eta^0) \in S$ is an efficient solution in (P) and there exists a positive real number M such that, for all $l = \overline{1, p}$, we have

$$\Phi^{l}(x^{0},\eta^{0}) - \Phi^{l}(x,\eta) \leq M \left(\Phi^{s}(x,\eta) - \Phi^{s}(x^{0},\eta^{0})\right)$$

for some $s \in \{1, \ldots, p\}$ such that

$$\Phi^s(x,\eta) > \Phi^s(x^0,\eta^0),$$

whenever $(x, \eta) \in S$ and

$$\Phi^l(x,\eta) < \Phi^l(x^0,\eta^0).$$

Definition 3. A point $(x^0, \eta^0) \in S$ is called a *weak efficient solution* in (P) if there exists no other $(x, \eta) \in S$ such that $\Phi(x, \eta) < \Phi(x^0, \eta^0)$ or, equivalently, $\Phi^l(x, \eta) - \Phi^l(x^0, \eta^0) < 0$, $(\forall)l = \overline{1, p}$.

According to Treanță,¹¹ for $x \in \mathbb{Q}$ and $\eta \in \mathsf{T}$, we consider the vector functional of multiple integral type

$$L: \mathbf{Q} \times \mathbf{T} \to \mathbb{R}^p, \quad L(x,\eta) = \int_P \kappa(\varsigma, x(\varsigma), x_\alpha(\varsigma), \eta(\varsigma)) \, d\varsigma$$

and introduce the concepts of invexity and pseudoinvexity associated with L.

Definition 4. If there exist

$$\rho: P \times \mathbb{R}^n \times \mathbb{R}^k \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^n,$$

$$\rho = \rho\left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right) = \left(\rho^i\left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right)\right),$$

$$i = \overline{1, n},$$

of C^1 -class with $\rho\left(\varsigma, x^0(\varsigma), \eta^0(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right) = 0$, $(\forall)\varsigma \in P$, $\rho|_{\partial P} = 0$, and

$$\begin{aligned} \sigma : P \times \mathbb{R}^n \times \mathbb{R}^k \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^k, \\ \sigma &= \sigma \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right) \\ &= \left(\sigma^j \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right)\right), \ j = \overline{1, k}, \end{aligned}$$

of C^0 -class with $\sigma(\varsigma, x^0(\varsigma), \eta^0(\varsigma), x^0(\varsigma), \eta^0(\varsigma)) = 0$, $(\forall)\varsigma \in P$, $\sigma|_{\partial P} = 0$, such that

$$L(x,\eta) - L(x^{0},\eta^{0}) \geq \int_{P} \left[\frac{\partial \kappa}{\partial x} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \rho + \frac{\partial \kappa}{\partial x_{\alpha}} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) D_{\alpha} \rho \right] d\varsigma + \int_{P} \left[\frac{\partial \kappa}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma,$$

for any $(x,\eta) \in \mathbb{Q} \times \mathsf{T}$, then L is said to be *invex at* $(x^0,\eta^0) \in \mathbb{Q} \times \mathsf{T}$ with respect to ρ and σ .

Definition 5. In the above definition, with $(x, \eta) \neq (x^0, \eta^0)$, if we replace \geq with >, we say that L is strictly invex at $(x^0, \eta^0) \in \mathbb{Q} \times \mathbb{T}$ with respect to ρ and σ .

Definition 6. If there exist

$$\begin{split} \rho : P \times \mathbb{R}^n \times \mathbb{R}^k \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^n, \\ \rho &= \rho \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma) \right) \\ &= \left(\rho^i \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma) \right) \right), \ i = \overline{1, n}, \end{split}$$

of C^1 -class with $\rho\left(\varsigma, x^0(\varsigma), \eta^0(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right) = 0$, $(\forall)\varsigma \in P$, $\rho|_{\partial P} = 0$, and

$$\sigma: P \times \mathbb{R}^n \times \mathbb{R}^k \times \mathbb{R}^n \times \mathbb{R}^k \to \mathbb{R}^k,$$

$$\sigma = \sigma \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right)$$

$$= \left(\sigma^j \left(\varsigma, x(\varsigma), \eta(\varsigma), x^0(\varsigma), \eta^0(\varsigma)\right)\right), \ j = \overline{1, k},$$

of C^0 -class with $\sigma(\varsigma, x^0(\varsigma), \eta^0(\varsigma), x^0(\varsigma), \eta^0(\varsigma)) = 0$, $(\forall)\varsigma \in P$, $\sigma|_{\partial P} = 0$, such that

$$\begin{split} L(x,\eta) &- L\left(x^{0},\eta^{0}\right) < 0\\ \Rightarrow \int_{P} \left[\frac{\partial \kappa}{\partial x} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \rho \right. \\ &\left. + \frac{\partial \kappa}{\partial x_{\alpha}} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) D_{\alpha} \rho \right] d\varsigma \\ &\left. + \int_{P} \left[\frac{\partial \kappa}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma < 0, \end{split}$$

or, equivalently,

$$\begin{split} &\int_{P} \left[\frac{\partial \kappa}{\partial x} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \rho \right. \\ &\left. + \frac{\partial \kappa}{\partial x_{\alpha}} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) D_{\alpha} \rho \right] d\varsigma \\ &\left. + \int_{P} \left[\frac{\partial \kappa}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), x^{0}_{\alpha}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma \ge 0 \Rightarrow L\left(x, \eta\right) \\ &\left. - L\left(x^{0}, \eta^{0}\right) \ge 0, \end{split}$$

for any $(x,\eta) \in \mathbb{Q} \times \mathbb{T}$, then L is said to be *pseudoinvex at* $(x^0,\eta^0) \in \mathbb{Q} \times \mathbb{T}$ with respect to ρ and σ .

Some examples of invex and pseudoinvex multiple integral functionals can be consulted in Treanță¹¹ (see Examples 1 and 2).

Definition 7. The nonempty subset $X \times U \subset Q \times T$ is said to be *invex with respect to* ρ *and* σ if

$$(x^{0}, \eta^{0}) + \lambda \left(\rho \left(\varsigma, x, \eta, x^{0}, \eta^{0} \right), \sigma \left(\varsigma, x, \eta, x^{0}, \eta^{0} \right) \right) \in \mathsf{X} \times \mathsf{U},$$
for all $(x, \eta), \ (x^{0}, \eta^{0}) \in \mathsf{X} \times \mathsf{U}$ and $\lambda \in [0, 1].$

333

S. Treanță

Now, in order to formulate and prove some results on the existence of solutions for problem (P), we introduce the following *(weak)* vector-controlled variational inequalities:

I. Find $(x^0, \eta^0) \in S$ such that there exists no $(x, \eta) \in S$ satisfying

$$(\text{VI}) \quad \left(\int_{P} \left[\frac{\partial \phi^{1}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{1}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma, \dots, \\ \int_{P} \left[\frac{\partial \phi^{p}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{p}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma \right) \leq 0.$$

II. Find $(x^0, \eta^0) \in S$ such that there exists no $(x, \eta) \in S$ satisfying

$$(WVI) \quad \left(\int_{P} \left[\frac{\partial \phi^{1}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{1}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma, \dots, \\ \int_{P} \left[\frac{\partial \phi^{p}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{p}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma) \right) < 0.$$

Next, we present an illustrative example to verify that the abovementioned class of vector-controlled variational inequalities is solvable at a given point.

Example 1. Let us consider $m = p = 2, n = k = 1, P = [0, 1] \times [0, 1]$. Also, we assume that $x, \eta : P \to \mathbb{R}$ are piecewise differentiable functions and $\rho, \sigma : P \times \mathbb{R}^4 \to \mathbb{R}$ are given by: $\rho = 0$, and $\sigma = e^{x^0(\varsigma)} - e^{x(\varsigma)}$, $(\forall)_{\varsigma} \in P \setminus \partial P$ and $\sigma = 0$ for $\varsigma \in \partial P$. Define the following Lagrange-type densities

$$\phi = \left(\phi^1, \phi^2\right) : P \times \mathbb{R}^2 \to \mathbb{R}^2,$$

as

$$\phi^{1}(\varsigma, x(\varsigma), \eta(\varsigma)) = -\eta(\varsigma) - \frac{1}{2} - x(\varsigma),$$

$$\phi^{2}(\varsigma, x(\varsigma), \eta(\varsigma)) = e^{\eta(\varsigma)} + \frac{1}{2}.$$

Further, we can easily note that $(x^0, \eta^0) = (0, 0)$ is a solution for the associated vector-controlled variational inequality (VCVI). Indeed, we have

$$\left(\int_{P} \left[\frac{\partial \phi^{1}}{\partial x}\left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right)\rho + \frac{\partial \phi^{1}}{\partial \eta}\left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right)\sigma\right]d\varsigma, \\ \int_{P} \left[\frac{\partial \phi^{2}}{\partial x}\left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right)\rho + \frac{\partial \phi^{2}}{\partial \eta}\left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right)\sigma\right]d\varsigma\right) \\ = \left(\int_{P} \left(e^{x(\varsigma)} - 1\right)d\varsigma, \int_{P} e^{\eta(\varsigma)}\left(1 - e^{x(\varsigma)}\right)d\varsigma\right) \nleq (0, 0)$$

for all piecewise differentiable functions $x, \eta : P \to \mathbb{R}$.

3. Main Results

In this section, we will state some characterization results and connections between the solutions of the considered (weak) vectorcontrolled variational inequalities and (proper, weak) efficient solutions of the introduced vector variational control problem (P).

Theorem 1. Let $S \subset \mathbb{Q} \times \mathbb{T}$ be an invex set with respect to ρ and σ , and let $(x^0, \eta^0) \in S$ be a proper efficient solution of (P). If each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$, is Fréchet differentiable at $(x^0, \eta^0) \in S$, then the pair (x^0, η^0) solves (VI).

Proof. By reductio ad absurdum, consider that $(x^0, \eta^0) \in S$ is a proper efficient solution of (P), but it does not satisfy (VI). In consequence, there exists $(x, \eta) \in S$ such that, for all $l = \overline{1, p}$, we have

$$\int_{P} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma < 0 \qquad (1)$$

and, for $s \neq l$,

$$\int_{P} \left[\frac{\partial \phi^{s}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{s}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma \leq 0.$$
 (2)

By hypothesis, we have that $S \subset Q \times T$ is an invex set with respect to ρ and σ . Thus, we can consider the pair

$$(z,w) = (x^0,\eta^0) + \lambda_n \left(\rho\left(\varsigma, x, \eta, x^0, \eta^0\right), \sigma\left(\varsigma, x, \eta, x^0, \eta^0\right) \right) \in S, \ (\forall)n,$$

for some sequence $\{\lambda_n\}$ of positive real numbers, satisfying $\lambda_n \to 0$ as $n \to \infty$.

Further, we apply that each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$ is Fréchet differentiable at $(x^0, \eta^0) \in S$ and obtain the following equality:

$$\Phi^{l}(z,w) - \Phi^{l}(x^{0},\eta^{0})$$

$$= \int_{P} \lambda_{n} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma$$

$$+ \parallel \lambda_{n} \left(\rho \left(\varsigma, x, \eta, x^{0}, \eta^{0}\right), \sigma \left(\varsigma, x, \eta, x^{0}, \eta^{0}\right) \right) \parallel \cdot G^{l}(z,w), \quad (3)$$

where $G^l: V_{(x^0,\eta^0)} \to \mathbb{R}$ is a continuous function defined on a neighborhood of (x^0,η^0) , denoted by $V_{(x^0,\eta^0)}$, with $\lim_{n\to\infty} G^l(z,w) = 0$. By dividing (3) with λ_n and taking the limit, we get

$$\lim_{n \to \infty} \frac{1}{\lambda_n} \left[\Phi^l(z, w) - \Phi^l(x^0, \eta^0) \right]$$
$$= \int_P \left[\frac{\partial \phi^l}{\partial x} \left(\varsigma, x^0(\varsigma), \eta^0(\varsigma) \right) \rho + \frac{\partial \phi^l}{\partial \eta} \left(\varsigma, x^0(\varsigma), \eta^0(\varsigma) \right) \sigma \right] d\varsigma.$$
(4)

Combining relations (1) and (4), it results that

$$\Phi^l(z,w) - \Phi^l(x^0,\eta^0) < 0$$

for some $n \geq N$, with N a natural number.

Next, since $(x^0, \eta^0) \in S$ is a proper efficient solution of (P), we consider the nonempty set

$$\mathcal{M} = \left\{ s \in \{1, \dots, p\} \mid \Phi^s(x^0, \eta^0) - \Phi^s(z, w) \le 0, \ (\forall) n \ge N \right\}.$$

For $s \in \mathcal{M}$, by considering the Fréchet differentiability of $\int_P \phi^s(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$ at $(x^0, \eta^0) \in S$, we obtain

$$\Phi^{s}(z,w) - \Phi^{s}(x^{0},\eta^{0})$$

$$= \int_{P} \lambda_{n} \left[\frac{\partial \phi^{s}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \rho + \frac{\partial \phi^{s}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma$$

$$+ \parallel \lambda_{n} \left(\rho \left(\varsigma, x, \eta, x^{0}, \eta^{0}\right), \sigma \left(\varsigma, x, \eta, x^{0}, \eta^{0}\right)\right) \parallel \cdot G^{s}(z,w), \quad (5)$$

where $G^s: V_{(x^0,\eta^0)} \to \mathbb{R}$ is a continuous function defined on a neighborhood of (x^0,η^0) , denoted by $V_{(x^0,\eta^0)}$, with $\lim_{n\to\infty} G^s(z,w) = 0$. By dividing (5) with λ_n and taking the limit, we get

$$\lim_{n \to \infty} \frac{1}{\lambda_n} \left[\Phi^s(z, w) - \Phi^s(x^0, \eta^0) \right] \\= \int_P \left[\frac{\partial \phi^s}{\partial x} \left(\varsigma, x^0(\varsigma), \eta^0(\varsigma) \right) \rho + \frac{\partial \phi^s}{\partial \eta} \left(\varsigma, x^0(\varsigma), \eta^0(\varsigma) \right) \sigma \right] d\varsigma.$$

By using the property of the set \mathcal{M} , for $n \geq N$, we get

$$\int_{P} \left[\frac{\partial \phi^{s}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{s}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma \ge 0. \quad (6)$$

Combining relations (2) and (6), we get

$$\int_{P} \left[\frac{\partial \phi^{s}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{s}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma = 0$$

for some $n \ge N$, with N a natural number, and $s \ne l$, $s \in \mathcal{M}$. Finally, for $s \ne l$, $s \in \mathcal{M}$, by computing the limit

$$\frac{\frac{1}{\lambda_n} \left[\Phi^l(x^0, \eta^0) - \Phi^l(z, w) \right]}{\frac{1}{\lambda_n} \left[\Phi^s(z, w) - \Phi^s(x^0, \eta^0) \right]},$$

we find that it is ∞ as $n \to \infty$, which contradicts the proper efficiency of (x^0, η^0) for (P). The proof is now complete. \Box

The next theorem provides a characterization of the efficient solutions for (P) by using the vector-controlled variational inequality (VI).

Theorem 2. Let $(x^0, \eta^0) \in S$ be a solution of (VI). If each multiple integral

$$\int_{P} \phi^{l}\left(\varsigma, x(\varsigma), \eta(\varsigma)\right) d\varsigma, \quad l = \overline{1, p}$$

is Fréchet differentiable and invex at $(x^0, \eta^0) \in S$ with respect to ρ and σ , then the pair (x^0, η^0) is an efficient solution of (P).

Proof. By reductio ad absurdum, consider that $(x^0, \eta^0) \in S$ is a solution of (VI), but it is not an efficient solution of (P). In consequence, there exists $(x, \eta) \in S$ such that, for all $l = \overline{1, p}$,

$$\Phi^{l}(x,\eta) - \Phi^{l}(x^{0},\eta^{0}) \le 0,$$
(7)

with strict inequality for at least one l.

By hypothesis, each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$ is Fréchet differentiable and invex at $(x^0, \eta^0) \in S$ with respect to ρ and σ . In consequence, we have

$$\Phi^{l}(x,\eta) - \Phi^{l}(x^{0},\eta^{0})$$

$$\geq \int_{P} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma)\right) \sigma \right] d\varsigma \quad (8)$$

for any $(x, \eta) \in S$ and $l = \overline{1, p}$.

On combining inequalities (7) and (8), we find that, for all $l = \overline{1, p}$, there exists $(x, \eta) \in S$ such that

$$\int_{P} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma \leq 0,$$

with strict inequality for at least one l, which contradicts that $(x^0, \eta^0) \in S$ is solution of (VI). The proof is now complete. \Box

The next result formulates a sufficient condition for a pair $(x^0, \eta^0) \in S$ to be a solution of (WVI).

Theorem 3. Let $S \subset \mathbb{Q} \times \mathbb{T}$ be an invex set with respect to ρ and σ , and let $(x^0, \eta^0) \in S$ be a weak efficient solution of (P). If each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$, is Fréchet differentiable at $(x^0, \eta^0) \in S$, then the pair (x^0, η^0) solves (WVI).

Proof. Since $(x^0, \eta^0) \in S$ is a weak efficient solution of (P), it results that there exists no other feasible solution $(x, \eta) \in S$ such that $\Phi(x, \eta) < \Phi(x^0, \eta^0)$, or, equivalently,

$$\Phi^{l}(x,\eta) - \Phi^{l}(x^{0},\eta^{0}) < 0, \quad (\forall)l = \overline{1,p}.$$
(9)

By hypothesis, we have that $S \subset \mathbb{Q} \times \mathsf{T}$ is an invex set with respect to ρ and σ . Thus, for all $\lambda \in [0, 1]$, we have

$$(z,w) = (x^0,\eta^0) + \lambda \left(\rho\left(\varsigma, x, \eta, x^0, \eta^0\right), \sigma\left(\varsigma, x, \eta, x^0, \eta^0\right)\right) \in S.$$

Thus, by using (9), we can see that there exists no other feasible solution $(x, \eta) \in S$ such that $\Phi(z, w) < \Phi(x^0, \eta^0)$, or, equivalently,

$$\Phi^{l}(z,w) - \Phi^{l}(x^{0},\eta^{0}) < 0, \quad (\forall)l = \overline{1,p}.$$
 (10)

Further, we apply that each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$, is Fréchet differentiable at $(x^0, \eta^0) \in S$ and, proceeding as in the proof of Theorem 1, by (10), we obtain that there exists no other feasible solution $(x, \eta) \in S$ such that

$$\int_{P} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma < 0$$

for all $l = \overline{1, p}$. This ends the proof.

The following theorem provides a characterization of weak efficient solutions for (P) by using the weak vector-controlled variational inequality (WVI).

Theorem 4. Let $(x^0, \eta^0) \in S$ be a solution of (WVI). If each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$, is Fréchet differentiable and pseudoinvex at $(x^0, \eta^0) \in S$ with respect to ρ and σ , then the pair (x^0, η^0) is a weak efficient solution of (P).

 \square

Proof. By reductio ad absurdum, consider that $(x^0, \eta^0) \in S$ is a solution of (WVI) but it is not a weak efficient solution of (P). In consequence, there exists $(x, \eta) \in S$ such that, for all $l = \overline{1, p}$,

$$\Phi^{l}(x,\eta) - \Phi^{l}(x^{0},\eta^{0}) < 0.$$

By hypothesis, each multiple integral $\int_P \phi^l(\varsigma, x(\varsigma), \eta(\varsigma)) d\varsigma$, $l = \overline{1, p}$, is Fréchet differentiable and pseudoinvex at $(x^0, \eta^0) \in S$ with respect to ρ and σ . In consequence, we have

$$\int_{P} \left[\frac{\partial \phi^{l}}{\partial x} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \rho + \frac{\partial \phi^{l}}{\partial \eta} \left(\varsigma, x^{0}(\varsigma), \eta^{0}(\varsigma) \right) \sigma \right] d\varsigma < 0$$

for any $(x, \eta) \in S$ and $l = \overline{1, p}$. This contradicts that $(x^0, \eta^0) \in S$ is a solution of (WVI). The proof is complete. \Box

4. Conclusions

In this chapter, we have established certain connections between the solutions of the considered (weak) vector-controlled-variational inequalities and (proper, weak) efficient solutions associated with a class of multiobjective controlled variational problems defined by functionals of multiple integral type. A crucial role for proving the main results formulated in this chapter was played by the concept of invex set with respect to some given functions, the generalized convexity and Fréchet differentiability of the functionals which were involved. As further development, taking into account the concept of *variational derivative* (see, for instance, Treanță¹⁷) for functionals of multiple integral type, new aspects can be formulated based on the results presented in this chapter.

Acknowledgments

The author thankfully acknowledges the contribution of the reviewers toward improvement of the manuscript.

References

- A. M. Geoffrion, Proper efficiency and the theory of vector maximization, J. Math. Anal. Appl. 22, 618–630 (1968).
- A. Klinger, Improper solutions of the vector maximum problem, Oper. Res. 15, 570–572 (1967).
- K. R. Kazmi, Existence of solutions for vector optimization, Appl. Math. Lett. 9(6), 19–22 (1996).
- B. A. Ghaznavi-ghosoni and E. Khorram, On approximating weakly/properly efficient solutions in multi-objective programming, *Math. Comput. Modelling.* 54(11–12), 3172–3181 (2011).
- M. A. Hanson, On sufficiency of the Kuhn-Tucker conditions, J. Math. Anal. Appl. 80(2), 545–550 (1981).
- T. Antczak, Exact penalty functions method for mathematical programming problems involving invex functions, *Eur. J. Oper. Res.* 198(1), 29–36 (2009).
- M. Arana-Jiménez, V. Blanco, and E. Fernández, On the fuzzy maximal covering location problem, *Eur. J. Oper. Res.* 283(2), 692–705 (2020).
- S. K. Mishra, S. Y. Wang, and K. K. Lai, Nondifferentiable multiobjective programming under generalized *d*-univexity, *European J. Oper. Res.* 160(1), 218–226 (2005).
- I. Ahmad, On second-order duality for minimax fractional programming problems with generalized convexity, *Abstr. Appl. Anal.* pp. Art. ID 563924, 15 (2011).
- C. Mititelu and S. Treanță, Efficiency conditions in vector control problems governed by multiple integrals, J. Appl. Math. Comput. 57(1–2), 647–665 (2018).
- S. Treanță, Saddle-point optimality criteria in modified variational control problems with partial differential equation constraints, *Optimal Control Appl. Methods.* 41(4), 1160–1175 (2020).
- F. Giannessi. Theorems of alternative, quadratic programs and complementarity problems. In Variational Inequalities and Complementarity Problems (Proc. Internat. School, Erice, 1978), pp. 151–186. Wiley, Chichester (1980).
- G. Ruiz-Garzón, R. Osuna-Gómez, and A. Rufián-Lizana, Relationships between vector variational-like inequality and optimization problems, *Eur. J. Oper. Res.* 157(1), 113–119 (2004).
- G. Marinoschi, Control approach to an ill-posed variational inequality, Math. Model. Nat. Phenom. 9(4), 153–170 (2014).

- A. Jayswal, S. Choudhury, and R. U. Verma, Exponential type vector variational-like inequalities and vector optimization problems with exponential type invexities, *J. Appl. Math. Comput.* 45(1–2), 87–97 (2014).
- M. H. Kim, Relations between vector continuous-time program and vector variational-type inequality, J. Appl. Math. Comput. 16(1-2), 279–287 (2004).
- S. Treanță, On well-posed isoperimetric-type constrained variational control problems, J. Differ. Equations. 298, 480–499 (2021).
- S. Jha, P. Das, S. Bandhyopadhyay, and S. Treanță, Well-posedness for multi-time variational inequality problems via generalized monotonicity and for variational problems with multi-time variational inequality constraints, J. Comput. Appl. Math. 407, Paper No. 114033, 13 (2022).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0015

Chapter 15

The Theory of Nonabsolute Integrable Function Spaces over R^{∞} and Its Various Applications

Hemanta Kalita *,‡ and Bipan Hazarika $^{\dagger,\$}$

*Department of Mathematics, Assam Don Bosco University, Tepesia, Sonapur 782402, Assam, India †Department of Mathematics, Gauhati University, Guwahati 781014, Assam, India [‡]hemanta30kalita@gmail.com hemanta.kalita@dbuniversity.ac.in [§]bh_rgu@yahoo.co.in bh_gu@gauhati.ac.in

We develop a Banach space $K^{p}(\mathbf{R}^{\infty})$, $1 \leq p \leq \infty$. We discuss the Lebesgue spaces L^{p} over \mathbf{R}^{∞} are subsets of $K^{p}(\mathbf{R}^{\infty})$ as dense continuous compact embedding. The spaces of finitely additive measure \mathcal{M} over \mathbf{R}^{∞} are subsets of $K^{p}(\mathbf{R}^{\infty})$ as dense continuous compact embedding. Also, we discuss that $K^{p}(\mathbf{R}^{\infty})$ contains Henstock–Kurzweil integrable functions over \mathbf{R}^{∞} . Further, we obtain Hilbert space $GD^{2}(\mathbf{R}_{\mathbf{I}}^{\infty})$. Finally, we demonstrate that Fourier transformation and convolution operators can be extended on the Hilbert spaces $K^{2}(\mathbf{R}^{\infty})$ as linear operators.

1. Introduction

The formation of Banach spaces with Luxemburg norm of Henstock– Kurzweil-integrable function is not possible (see Refs. 1–9). Yeong¹⁰
addressed this drawback, which can be solved if a canonical approach is developed. In Kuelbs–Steadman spaces denoted by $KS^p(S), \forall 1 \leq$ $r < \infty$, where $S \subset \mathbf{R}^{\mathbf{m}}$ was developed by Gill and Zachary (see Ref. 11). An interesting fact about Kuelbs–Steadman spaces is that they are canonical in nature (see Refs. 11, 12). These spaces are directly related to the Feynman operator calculus and path integral. Kuelbs–Steadman spaces are separable, also Kuelbs–Steadman spaces containing the Lebesgue space L^p are continuously dense compact embedding. These spaces contains L^p as dense, continuous compact embedding and they are also separable. Gill and Zachary mainly focused in a Banach space which contains Henstock-Kurzweil integrable functions. Henstock–Kurzweil integrable functions are equivalent to the Denjoy integrable function in a restricted sense (see Refs. 3, 5, 13, 14). The motivation behind the construction of KS^p spaces was to find a Banach space with additive measures as well as containing Henstock-Kurzweil integrable functions. Simultaneously, Gill and Myers introduced an equivalent approach of Lebesgue measure theory over \mathbf{R}^{∞} (see Ref. 15). These ideas seem useful to construct separable Banach spaces containing Henstock-Kurzweil integrable functions. Gill et al.¹⁶ discussed the approach of the Kuelbs-Steadman spaces over \mathbf{R}^{∞} . Kalita *et al.*,¹⁷ expanded the theory of the spaces KS^p on separable Banach spaces. In Ref. 16, we indicated that all theorems of $K^p(\mathbf{R}^{\mathbf{m}}_{\mathbf{I}})$ are extendable to $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. In this chapter, we discuss the construction of $K^p(\mathbf{R}^{\infty}_{\mathbf{L}})$ and we will discuss all results that we mention in Ref. 16.

2. Preliminaries

In this section, we recall few definitions, which are important in our main section. We denote the set of real numbers by \mathbf{R} .

Definition 1. A σ -algebra on a set X is a nonempty collection Σ of subsets of X closed under complement and closed under countable unions and countable intersections. The pair (X, Σ) is called a measurable space.

Definition 2. Let (X, Σ_1, μ_1) and (Y, Σ_1, μ_2) be two measurable spaces. A product measure $\mu_1 \times \mu_2$ is defined to be a measure on

the measurable space $(X \times Y, \Sigma_1 \times \Sigma_2)$, if it satisfies the property

$$(\mu_1 \times \mu_2)(A \times B) = \mu_1(A)\mu_2(B),$$

for $A \in \Sigma_1, B \in \Sigma_2$.

Definition 3. The Borel measures on the Euclidean space \mathbb{R}^n can be obtained as the product of n copies of Borel measures on the real line \mathbb{R} .

Definition 4. Let μ be a measure on \mathbb{R}^n equipped with the Borel σ algebra $B(\mathbb{R}^n)$. Then μ is said to be translation-invariant or invariant under translations if $\mu(x+A) = \mu(A)$ for all $x \in \mathbb{R}^n$ and $A \in B(\mathbb{R}^n)$.

Definition 5. Let (X, Σ, μ) be a measurable space. A transformation $T: X \to X$ is said to be measurable if $T^{-1}(A) \in \Sigma$ for all $A \in \Sigma$. We say that T is a measure-preserving transformation or equivalently μ is said to be T-invariant measure if $\mu(T^{-1}(A)) = \mu(A)$ for all $A \in \Sigma$.

Definition 6 ([3, Definition 9.3]). A function $\mathbf{f} : [a, b] \to \mathbf{R}$ is Henstock–Kurzweil integrable on [a, b] if for each $\epsilon > 0$, there exists a positive function δ on [a, b] and a real number A such that

$$\left|\sum_{i=1}^{n} \mathbf{f}(x_i)(d_i - c_i) - A\right| < \epsilon.$$

Let us denote $HK(\mathbf{R})$ as the space of Henstock–Kurzweil integrable functions. Recall the norm for $\mathbf{f} \in HK(\mathbf{R})$ as

$$||\mathbf{f}||_{HK(\mathbf{R})} = \sup_{s} \left| \int_{-\infty}^{s} \mathbf{f}(r) dx(r) \right|.$$
(1)

The norm defined by Eq. (1) was introduced by Alexiewicz (see Ref. 9). One can follow Refs. 3,9 for the details of Henstock–Kurzweil integrals. Throughout the chapter, we consider $\mathbf{I} = [-\frac{1}{2}, \frac{1}{2}]$.

Definition 7 ([11, Definition 2.5]). If $\mathbf{A}_m = A \times \mathbf{I}_m$; $\mathbf{B}_m = B \times \mathbf{I}_m$ (*mth* order box sets in \mathbf{R}^{∞}). We recall

- (i) $\mathbf{A}_m \cup \mathbf{B}_m = (A \cup B) \times \mathbf{I}_m;$
- (ii) $\mathbf{A}_m \cap \mathbf{B}_m = (A \cap B) \times \mathbf{I}_m;$
- (iii) $\mathbf{B}_m^c = B^c \times \mathbf{I}_m$.

Definition 8 ([15, Definition 1.11]). Let us consider $\mathbf{R}_{\mathbf{I}}^{m} = \mathbf{R}^{m} \times \mathbf{I}_{m}$. Also, let **T** be a linear transformation on \mathbf{R}^{m} . The linear transformation on $\mathbf{R}_{\mathbf{I}}^{m}$ can be considered as $\mathbf{T}_{I}(\mathbf{A}_{m}) = \mathbf{T}(A)$, where $\mathbf{A}_{m} = A \times \mathbf{I}_{m}$.

Let us denote the Borel σ -algebra for $\mathbf{R}_{\mathbf{I}}^m$ by $B(\mathbf{R}_{\mathbf{I}}^m)$. We can define the topology on $\mathbf{R}_{\mathbf{I}}^m$ via the class of open sets $\mathbf{D}_m = \{U \times \mathbf{I}_m\}$, where $U \subset \mathbf{R}^m$ is an open set.

For given two measurable spaces, one can obtain a product measurable space and a product measure on that space with the similar concept of the Cartesian product of sets and the product of sets and the product topology of two topological spaces. When Ain $B(\mathbf{R}^m)$, and $\nu_{\infty}(\mathbf{A}_m)$ on $\mathbf{R}_{\mathbf{I}}^m$, the properties of product measure gives $\nu_{\infty}(\mathbf{A}_m) = \nu_m(A) \times \pi_{i=m+1}^{\infty} \nu_{\mathbf{I}}(\mathbf{I}) = \nu_m(A)$, where $\nu_m(A)$ is Lebesgue measure of A. Recalling (see Ref. 11 Theorem 2.7), if $\nu_{\infty}(.)$ is a measure on $B(\mathbf{R}_{\mathbf{I}}^m)$ then $\nu_{\infty}(.)$ coincides with the *m*-dimensional Lebesgue measure on \mathbf{R}^m .

Corollary 1 ([11, Corollary 2.8]). For each $m \in \mathbf{N}$, the measure $\nu_{\infty}(.)$ is translation as well as rotational invariant on $(\mathbf{R}_{\mathbf{I}}^{m}, B(\mathbf{R}_{\mathbf{I}}^{m}))$.

The theory on $\mathbf{R}_{\mathbf{I}}^m$ is completely parallel to that on \mathbf{R}^m . Since $\mathbf{R}_{\mathbf{I}}^m \subset \mathbf{R}_{\mathbf{I}}^{m+1}$, one can define $\widehat{\mathbf{R}}_{\mathbf{I}}^\infty = \lim_{n \to \infty} \mathbf{R}_{\mathbf{I}}^m = \bigcup_{k=1}^{\infty} \mathbf{R}_{\mathbf{I}}^k$. Let $\mathbf{X}_1 = \widehat{\mathbf{R}}_{\mathbf{I}}^\infty$. We consider the topology τ_1 induced by the class of open sets $D \subset \mathbf{X}_1$ so that $D = \bigcup_{n=1}^{\infty} \mathbf{D}_m = \bigcup_{n=1}^{\infty} \{U \times \mathbf{I}_m : U \text{ is open in } \mathbf{R}^m\}$. Let $\mathbf{X}_2 = \mathbf{R}^\infty \setminus \widehat{\mathbf{R}}_{\mathbf{I}}^\infty$, and we assume a discrete topology τ_2 on \mathbf{X}_2 induced by the discrete metric so that for $x, y \in \mathbf{X}_2, x \neq y$, $\mathbf{d}_2(x, y) = 1$ and for x = y, $\mathbf{d}_2(x, y) = 0$. We denote the co-product of (\mathbf{X}_1, τ_1) and (\mathbf{X}_2, τ_2) on $(\mathbf{R}_{\mathbf{I}}^\infty, \tau)$ by $(\mathbf{X}_1, \tau_1) \otimes (\mathbf{X}_2, \tau_2)$. In this case, every open set in $(\mathbf{R}_{\mathbf{I}}^\infty, \tau)$ is the disjoint union of two open sets $\mathbf{G}_1 \cup \mathbf{G}_2$ with \mathbf{G}_1 in (\mathbf{X}_1, τ_1) and \mathbf{G}_2 in (\mathbf{X}_2, τ_2) .

Being sets, $\mathbf{R}_{\mathbf{I}}^{\infty} = \mathbf{R}^{\infty}$. But, since each point in \mathbf{X}_2 is open and closed in $\mathbf{R}_{\mathbf{I}}^{\infty}$ and points in \mathbf{R}^{∞} are neither open nor closed, hence, as a topological space, $\mathbf{R}_{\mathbf{I}}^{\infty} \neq \mathbf{R}^{\infty}$.

Likewise, if $B(\mathbf{R}_{\mathbf{I}}^{m})$ is the Borel σ -algebra of $\mathbf{R}_{\mathbf{I}}^{m}$, at that time, $B(\mathbf{R}_{\mathbf{I}}^{m}) \subset B(\mathbf{R}_{\mathbf{I}}^{m+1})$ can be defined as

$$\widehat{B}(\mathbf{R}_{\mathbf{I}}^{\infty}) = \lim_{m \to \infty} B(\mathbf{R}_{\mathbf{I}}^{m}) = \bigcup_{k=1}^{\infty} B(\mathbf{R}_{\mathbf{I}}^{k}).$$

If $B(\mathbf{R}_{\mathbf{I}}^{\infty})$ is the tiniest σ -algebra carrying $\widehat{B}(\mathbf{R}_{\mathbf{I}}^{\infty}) \cup P\left(\mathbf{R}^{\infty} \setminus \bigcup_{k=1}^{\infty} (\mathbf{R}_{\mathbf{I}}^{k})\right)$, where P(.) is the power set, then clearly the class $B(\mathbf{R}_{\mathbf{I}}^{\infty})$ coincides with the Borel σ -algebra produced by the τ -topology on $\mathbf{R}_{\mathbf{I}}^{\infty}$. We can find $\widehat{B}(\mathbf{R}_{\mathbf{I}}^{\infty}) \subset B(\mathbf{R}_{\mathbf{I}}^{\infty})$ (see Ref. 15 Lemma 1.15).

3. Measurable Functions

In Ref. 15, Gill and Myres have shown that the Lebesgue measure ν_m of \mathbf{R}^m can be extended to the measure $\nu_{\infty}(.)$ of \mathbf{R}^{∞} . The theory of measurable functions on $\mathbf{R}_{\mathbf{I}}^{\infty}$ was discussed in Refs. 11, 16, 18. We recall the measurable function on $\mathbf{R}_{\mathbf{I}}^{\infty}$. Assume $\mathbf{x} = (x_1, x_2, ...) \in \mathbf{R}_{\mathbf{I}}^{\infty}$, $\mathbf{I}_m = \prod_{k=m+1}^{\infty} [\frac{-1}{2}, \frac{1}{2}]$ and let $\mathbf{h}_m(\hat{x}) = \chi_{\mathbf{I}_m}(\hat{\mathbf{x}})$, where $\hat{\mathbf{x}} = (x_i)_{i=m+1}^{\infty}$. Assume the class of measurable functions on \mathbf{R}^m is represented by M^m . Let $\mathbf{x} \in \mathbf{R}_{\mathbf{I}}^{\infty}$ and $f^m \in M^m$. Let an essentially tame measurable function of order m (or e_m -tame) on $\mathbf{R}_{\mathbf{I}}^{\infty}$ be

$$\mathbf{f}(\mathbf{x}) = \mathbf{f}^m(\overline{\mathbf{x}}) \otimes \mathbf{h}_m(\widehat{\mathbf{x}}),$$

where $\overline{\mathbf{x}} = (x_i)_{i=1}^m$. Then $M_{\mathbf{I}}^m = \{\mathbf{f}(x) : \mathbf{f}(x) = \mathbf{f}^m(\overline{\mathbf{x}}) \otimes \mathbf{h}_m(\widehat{\mathbf{x}}), \mathbf{x} \in \mathbf{R}_{\mathbf{I}}^\infty\}$ is the class of all e_m -tame functions.

Definition 9 ([11, **Definition 2.47**]). We call a mapping \mathbf{f} : $\mathbf{R}_{\mathbf{I}}^{\infty} \to \mathbf{R}$ to be measurable with $\mathbf{f} \in M_{\mathbf{I}}$, if there exists a sequence $\{\mathbf{f}_m \in M_{\mathbf{I}}^m\}$ of e_m -tame functions, so that

$$\lim_{m \to \infty} \mathbf{f}_m(x) \to \mathbf{f}(x) \quad with \ \nu_{\infty} - (a.e.).$$

Recalling the existence of functions fulfilling Definition 9 is not obvious.

Theorem 1 ([11, Theorem 2.48] (Existence)). Let us consider $\mathbf{f} : \mathbf{R}_{\mathbf{I}}^{\infty} \to (-\infty, \infty)$ in addition to $\mathbf{f}^{-1}(A) \in B(\mathbf{R}_{\mathbf{I}}^{\infty})$ for all $A \in B(\mathbf{R})$, then for a family of functions $\{\mathbf{f}_m\}$, such that $\mathbf{f}_m(x) \to \mathbf{f}(x)$, ν_{∞} (-a.e.), where $\mathbf{f}_m \in M_{\mathbf{I}}^m$.

Remark 1. Recognizing that any measurable set A, with nonzero measure, is concentrated in \mathbf{X}_1 that is to say, $\nu_{\infty}(A) = \nu_{\infty}(A \cap \mathbf{X}_1)$. In addition it follows that the essential support of the limit function

 $\mathbf{f}(x)$ in Definition 9, i.e. $\{x : \mathbf{f}(x) \neq 0\}$ is concentrated in $\mathbf{R}_{\mathbf{I}}^{N}$, for some N.

4. R_{I}^{∞} and Its Integration Theory

The construction of the integration theory on $\mathbf{R}_{\mathbf{I}}^{\infty}$ is similar to the integration theory of \mathbf{R}^m . All theorems for Lebesgue measure are also applicable in the theory of $\mathbf{R}_{\mathbf{I}}^m$. Let us consider the class of integrable functions on $\mathbf{R}_{\mathbf{I}}^m$ be $L^1(\mathbf{R}_{\mathbf{I}}^m)$. As $L^1(\mathbf{R}_{\mathbf{I}}^m) \subset L^1(\mathbf{R}_{\mathbf{I}}^{m+1})$, recall $L^1(\widehat{\mathbf{R}}_{\mathbf{I}}^{\infty}) = \bigcup_{m=1}^{\infty} L^1(\mathbf{R}_{\mathbf{I}}^m)$.

A measurable function \mathbf{f} is in $L^1(\mathbf{R}_{\mathbf{I}}^\infty)$ if the following condition satisfies $\mathbf{f} \in L^1(\mathbf{R}_{\mathbf{I}}^\infty)$, if for a Cauchy-sequence $\{\mathbf{f}_m\} \subset L^1(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty)$ in the company of $\mathbf{f}_m \in L^1(\mathbf{R}_{\mathbf{I}}^m)$, along with

$$\lim_{m \to \infty} \mathbf{f}_m(x) = \mathbf{f}(x), \ \nu_{\infty} - (a.e.).$$

Theorem 2 ([16, Theorem 1.8]). $L^1(\mathbf{R}^{\infty}_{\mathbf{I}}) = L^1(\widehat{\mathbf{R}}^{\infty}_{\mathbf{I}}).$

For $f \in L^1(\mathbf{R}^{\infty}_{\mathbf{I}})$, the integral of **f** is defined as follows:

$$\int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{f}(x) d\nu_{\infty}(x) = \lim_{m \to \infty} \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{f}_{m}(x) d\nu_{\infty}(x), \tag{2}$$

for any Cauchy-sequence $\{f_m\} \subset L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ converges to $\mathbf{f}(x)$ -a.e. (see Ref. 16 Definition 1.9). We can find from Ref. 15, Theorem 3.15, that when $\mathbf{f} \in L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$, then the integral (2) exists and all theorems that are true for $\mathbf{f} \in L^1(\mathbf{R}_{\mathbf{I}}^m)$, also hold for $\mathbf{f} \in L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$.

5. Kuelbs–Steadman Spaces $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ $1 \leq p \leq \infty$

Recalling Henstock–Kurzweil integrals generalize the Riemann and Lebesgue integrals. In the theory of Henstock⁶ and Kurzweil,¹⁴ finitely additive measure can be generated by the Henstock–Kurzweil integral (HK integral). In the restricted sense, Denjoy integrals are equivalent to the Henstock–Kurzweil integrals and also the Perron integrals (see Ref. 3 and references therein). The theory of Henstock–Kurzweil integral is easier in comparison with the theory

of Lebesgue integral. Henstock–Kurzweil integrals give useful variants of the theorems of Lebesgue integral. The widespread use of the Henstock–Kurzweil integral in mathematics, engineering, physics and chemistry is unexplored due to the lack of a Banach space structure of Henstock–Kurzweil integral. The possibility of development of this condition began indirectly in 1965, when Gross^{19} proved that every separable Banach space contains a separable Hilbert space as a continuous dense embedding. This was the extension work of Wieners theory, which used the (densely embedded Hilbert) Sobolev space $\mathcal{H}^1_0[0,1] \subset C_0[0,1]$. Kuelbs generalized Gross's theorem (see Ref. 12) in the year 1970.

Theorem 3 ([16, Theorem 2.1] Gross–Kuelbs). For a separable Banach space \overline{B} there exist two other Banach spaces, H_1, H_2 , such that $H_1 \subset \overline{B} \subset H_2$ all as continuous dense embedding with the followings

$$\left\langle T_{12}^{\frac{1}{2}}u, T_{12}^{\frac{1}{2}}v\right\rangle_{1} = \left\langle u, v\right\rangle_{2} and$$
$$\left\langle T_{12}^{\frac{-1}{2}}u, T_{12}^{\frac{-1}{2}}v\right\rangle_{2} = \left\langle u, v\right\rangle_{1},$$

where T_{12} is a positive trace class operator defined on H_2 .

This work appeared for the first time in Ref. 20 in association to the Henstock–Kurzweil integral.

Lemma 1 ([12, Kuelbs Lemma]). Let \overline{B} be a separable Banach space then $H \supset \overline{B}$ is a continuous dense embedding, where H is a separable Hilbert space.

It is very straight forward to see that some Henstock-Kurzweil integrable-type functions are in H. It is very complicated to predict whether these nonabsolute integrals are Henstock-Kurzweil or not. Steadman gave an effective hint for this (see Ref. 20). Before the construction of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$, we need to understand the construction of $K^p(\mathbf{R}^m_{\mathbf{I}})$.

Formation of $K^p(\mathbf{R}_{\mathbf{I}}^m)$: Let us consider $m \in \mathbf{N}$ and let $\mathbf{Q}_{\mathbf{I}}^m$ be the set $\{\mathbf{x} \in \mathbf{R}_{\mathbf{I}}^m\}$, so that $(\mathbf{x}_1, \mathbf{x}_2, ..., \mathbf{x}_m)$ are initial co-ordinates. These co-ordinates are rational. Clearly, this is a countable dense

set of $\mathbf{R}_{\mathbf{I}}^{m}$. For every k and i, we can arrange $\mathbf{Q}_{\mathbf{I}}^{\mathbf{m}} = \{\mathbf{x}_{1}, \mathbf{x}_{2}, ...\}$. Let us assume a closed cube $\mathcal{B}_{k}(x_{i})$ centered at x_{i} along with edge $e_{k} = \frac{1}{2^{l-1}\sqrt{n}}, \ l \in \mathbf{N}$. We can choose a map $\mathbf{N} \times \mathbf{N}$ bijectively to \mathbf{N} of natural order and suppose the resulting set of all closed cubes

$$\{\mathcal{B}_k(x_i) | (l,i) \in \mathbf{N} \times \mathbf{N}\}\$$

centered at a point in $\mathbf{Q}_{\mathbf{I}}^{\mathbf{m}}$ be $\{\mathcal{B}_k : k \in \mathbf{N}\}$. Let the characteristic function of \mathcal{B}_k be $\mathbf{E}_{\mathbf{k}}(\mathbf{x})$ such that for $1 \leq p < \infty$, $\mathbf{E}_{\mathbf{k}}(\mathbf{x}) \in \mathbf{L}^{\mathbf{p}}(\mathbf{R}_{\mathbf{I}}^{\mathbf{m}}) \cap \mathbf{L}^{\infty}(\mathbf{R}_{\mathbf{I}}^{\mathbf{m}})$. We can define $F_k(.)$ on $L^1(\mathbf{R}_{\mathbf{I}}^m)$ by

$$F_k(\mathbf{f}) = \int_{\mathbf{R}_{\mathbf{I}}^m} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d}\nu_{\infty}(\mathbf{x}).$$

Since each \mathcal{B}_k is a cube with sides parallel to the co-ordinate axes, $F_k(.)$ is well defined for all Henstock–Kurzweil integrable functions and is a bounded linear functional on $L^p(\mathbf{R}^m_{\mathbf{I}})$ for $1 \leq p \leq \infty$. Let $\mathbf{t}_k > 0$ so that $\sum_{k=1}^{\infty} \mathbf{t}_k = 1$. We can define an inner product $\langle . \rangle$ on $L^1(\mathbf{R}^m_{\mathbf{I}})$ as follows:

$$\begin{split} \left\langle \mathbf{f},\mathbf{g} \right\rangle &= \sum_{k=1}^{\infty} \mathbf{t}_k \bigg(\int_{\mathbf{R}_{\mathbf{I}}^m} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d} \nu_{\infty}(\mathbf{x}) \bigg) \\ &\times \bigg(\int_{\mathbf{R}_{\mathbf{I}}^n} \mathbf{E}_{\mathbf{k}}(\mathbf{y}) \mathbf{g}(\mathbf{y}) \mathbf{d} \nu_{\infty}(\mathbf{y}) \bigg)^{\mathbf{c}}. \end{split}$$

With this inner product, the completion of $L^1(\mathbf{R}_{\mathbf{I}}^m)$ is the *p*-summable space $K^2(\mathbf{R}_{\mathbf{I}}^m)$. This space contains Henstock–Kurzweil integrable functions. We can find this as follows:

$$||\mathbf{f}||_{K^2}^2 = \sum_{k=1}^{\infty} \mathbf{t}_k \bigg| \int_{\mathbf{R}_{\mathbf{I}}^n} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) d\nu_{\infty}(\mathbf{x}) \bigg|^2.$$

If we replace **R** by $\mathbf{R}_{\mathbf{I}}^{m}$, then the Alexiewicz norm defined in Eq. (1) becomes

$$||\mathbf{f}||_{HK(\mathbf{R}_{\mathbf{I}}^{m})} = \sup_{r>0} \left| \int_{\mathcal{B}_{r}} \mathbf{f}(x) d\nu_{\infty}(x) \right| < \infty,$$
(3)

and $||\mathbf{f}||_{K^2}^2 \leq \sup_k |\int_{\mathcal{B}_k} \mathbf{f}(x) d\nu_{\infty}(x)|^2 < \infty$. So, $\mathbf{f} \in K^2(\mathbf{R}_{\mathbf{I}}^m)$.

Theorem 4 ([16, Theorem 2.3]). For every $p, 1 \leq p \leq \infty$, $L^p(\mathbf{R}^m_{\mathbf{I}})$ is a densely continuous subspace of $K^2(\mathbf{R}^m_{\mathbf{I}})$.

For $1 \leq p \leq \infty$, the norm of $K^p(\mathbf{R}^m_{\mathbf{I}})$ can be defined as

$$||\mathbf{f}||_{K^{p}(\mathbf{R}_{\mathbf{I}}^{m})} = \begin{cases} \left(\sum_{k=1}^{\infty} \mathbf{t}_{k} \left| \int_{\mathbf{R}_{\mathbf{I}}^{m}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) d\nu_{\infty}(\mathbf{x}) \right|^{p} \right)^{\frac{1}{p}}, \text{ for } 1 \leq p < \infty; \\ \sup_{k \geq 1} \left| \int_{\mathbf{R}_{\mathbf{I}}^{m}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) d\nu_{\infty}(\mathbf{x}) \right|, \text{ for } p = \infty \end{cases}$$

with $\mathbf{x} = (\mathbf{x}_1, \mathbf{x}_2, \mathbf{x}_3, \dots) \in \mathbf{R}_{\mathbf{I}}^{\infty}$ and $\mathbf{R}_{\mathbf{I}}^{\infty} = \lim_{m \to \infty} \mathbf{R}_{\mathbf{I}}^m = \bigcup_{m=1}^{\infty} \mathbf{R}_{\mathbf{I}}^k$ and $\mathbf{R}_{\mathbf{I}}^m = \mathbf{R}^m \times \mathbf{I}_m$, where $\mathbf{I} = [\frac{-1}{2}, \frac{1}{2}]$.

Obviously, $||f||_{K^p}$ defines on \tilde{L}^p and with respect to this norm L^p is the completion of K^p . As L^p is the completion of K^p , then we have the following theorem:

Theorem 5 ([16, Theorem 2.4]). For every q, $1 \leq q \leq \infty$, $L^q(\mathbf{R}^m_{\mathbf{I}})$ is a densely continuous subspace of $K^p(\mathbf{R}^m_{\mathbf{I}})$. Also, the inclusion is continuous embedding.

Remark 2. For each $p, 1 \leq p < \infty$, the Hölder and Minkowski inequalities hold for $L^p(\mathbf{R}^m_{\mathbf{I}})$. The norm of $K^p(\mathbf{R}^m_{\mathbf{I}})$ is derived from $L^p(\mathbf{R}^m_{\mathbf{I}})$ and the completion of $L^p(\mathbf{R}^m_{\mathbf{I}})$ is $K^p(\mathbf{R}^m_{\mathbf{I}})$. So, for $1 \leq p < \infty$, one can easily show that Hölder and Minkowski inequalities hold in $K^p(\mathbf{R}^m_{\mathbf{I}})$.

Theorem 6 ([16, Theorem 2.5]). For $p, 1 \le p \le \infty$, the following are true:

- (1) When $\mathbf{f}_m \to \mathbf{f}$ weakly in $L^p(\mathbf{R}^m_{\mathbf{I}})$, then $\mathbf{f}_m \to \mathbf{f}$ strongly in $K^p(\mathbf{R}^m_{\mathbf{I}})$.
- (2) $K^p(\mathbf{R}^m_{\mathbf{I}})$ is uniformly convex for 1 .
- (3) $K^q(\mathbf{R}^m_{\mathbf{I}})$ the dual space of $K^p(\mathbf{R}^m_{\mathbf{I}})$ for $1 and <math>\frac{1}{n} + \frac{1}{q} = 1$.
- (4) $K^{\infty}(\mathbf{R}^m_{\mathbf{I}}) \subset K^p(\mathbf{R}^m_{\mathbf{I}})$ for $1 \leq p < \infty$.
- (5) $C_c(\mathbf{R}^m_{\mathbf{I}})$ is dense subset of $K^p(\mathbf{R}^m_{\mathbf{I}})$.

Theorem 7 ([16, Theorem 2.6]). For every $p, 1 \leq p \leq \infty$, then the test function spaces $D(\mathbf{R}_{\mathbf{I}}^m) \subset K^p(\mathbf{R}_{\mathbf{I}}^m)$ are continuously embedding.

6. GD^2 Gross Steadman–Hilbert Spaces

Hilbert spaces play an important role in the extension of the Feynman operator Calculus to the nonreflexive Banach spaces. In this section, we will construct a Hilbert space on $\mathbf{R}_{\mathbf{I}}^{m}$, which is a natural compliment to $K^{2}(\mathbf{R}_{\mathbf{I}}^{m})$ in a certain general sense. For the construction of this Hilbert space, we fix \overline{B} and define $GD_{\overline{B}}^{2}(\mathbf{R}_{\mathbf{I}}^{m})$ by

$$GD_{\overline{B}}^{2}(\mathbf{R}_{\mathbf{I}}^{m}) = \left\{ \mathbf{f} \in \overline{B} : \sum_{n=1}^{\infty} \mathbf{t}_{m}^{-1} | < \mathbf{f}, \ \mathbf{E}_{\mathbf{m}} >_{\mathbf{2}} |^{\mathbf{2}} < \infty \right\}$$

and $\langle \mathbf{f}, \mathbf{g} \rangle_1 = \sum_{m=1}^{\infty} \mathbf{t}_m^{-1} \langle \mathbf{f}, \mathbf{E}_m \rangle_2 \langle \mathbf{E}_m, \mathbf{g} \rangle_2$ where $\mathbf{g} \in \overline{B}$. We can extend the Gross–Kuelbs theorem in this sense as follows:

Theorem 8. The operator T_{12} is a positive trace class operator on \overline{B} with a bounded extension to $K^2(\mathbf{R}^m_{\mathbf{I}})$. Additionally, $GD^2_{\overline{B}}(\mathbf{R}^m_{\mathbf{I}}) \subset \overline{B} \subset K^2(\mathbf{R}^m_{\mathbf{I}})$ (as continuous dense embedding), $\left\langle T_{12}^{\frac{1}{2}}u, T_{12}^{\frac{1}{2}}v \right\rangle_1 = \left\langle u, v \right\rangle_2$ and $\left\langle T_{12}^{\frac{-1}{2}}u, T_{12}^{\frac{-1}{2}}v \right\rangle_2 = \left\langle u, v \right\rangle_1$.

Proof. We can prove the theorem with the similar technique of Ref. 11 Theorem 3.30. \Box

Changing \overline{B} by $K^p(\mathbf{R}^m_{\mathbf{I}})$, one can find the following relation:

$$GS^2_{\overline{B}}(\mathbf{R}^m_{\mathbf{I}}) \to K^p(\mathbf{R}^m_{\mathbf{I}}) \to K^2(\mathbf{R}^m_{\mathbf{I}}).$$

7. The Family $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$

In this section, we will construct the spaces $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$, $1 \leq p \leq \infty$. We are motivated by the approach of the construction of $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$. Using the same approach that led to $L^1[\mathbf{R}_{\mathbf{I}}^{\infty}]$, since $\mathbf{R}_{\mathbf{I}}^m \subset \mathbf{R}_{\mathbf{I}}^{m+1}$, then $K^p(\mathbf{R}_{\mathbf{I}}^m) \subset K^p(\mathbf{R}_{\mathbf{I}}^{m+1})$. We can have an increasing sequence so that $K^p(\mathbf{R}_{\mathbf{I}}^m) = \bigcup_{m=1}^{\infty} K^p(\mathbf{R}_{\mathbf{I}}^m)$.

Definition 10 ([16, Definition 2.7]). For $1 \le p \le \infty$, we say that a measurable function $\mathbf{f} \in K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ if we can find a Cauchy sequence

 $\{\mathbf{f}_m\} \subset K^p(\widehat{\mathbf{R}}_{\mathbf{I}}^{\infty}]$ along with $\mathbf{f}_m \in K^p(\mathbf{R}_{\mathbf{I}}^m)$ such that $\lim_{m \to \infty} \mathbf{f}_m(x) = \mathbf{f}(x) \ \nu_{\infty}$ -a.e.

The functions of $K^p(\widehat{\mathbf{R}}_{\mathbf{I}}^{\infty})$ are different from the functions in its closure of $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ by those sets of measure zero.

Theorem 9 ([16, Theorem 2.8]). $K^p(\widehat{\mathbf{R}}^{\infty}_{\mathbf{I}}) = K^p(\mathbf{R}^{\infty}_{\mathbf{I}}).$

Definition 11. For $f \in K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$, the integral of **f** is defined as follows:

$$\int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{f}(x) d\nu_{\infty}(x) = \lim_{m \to \infty} \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{f}_m(x) d\nu_{\infty}(x), \tag{4}$$

where $\{f_m\} \subset K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is any Cauchy sequence converging to $\mathbf{f}(x)$ -a.e.

Theorem 10. Suppose $\mathbf{f} \in K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. The integral of \mathbf{f} defined in Definition 11 exists and its uniqueness holds for every $\mathbf{f} \in K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$.

Proof. Since the family of functions $\{\mathbf{f_m}\}$ is Cauchy, the uniqueness holds if the integral exists. We can prove the existence with standard argument by assuming $\mathbf{f}(x) \geq 0$. In addition, we consider the sequence to to be increasing such that the integral should exist. Finally, for the general case, we follow the standard decomposition.

To form $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$, let us consider a countable dense set of functions $\{\mathbf{E}_{\mathbf{m}}(\mathbf{x})\}_{\mathbf{m}=\mathbf{1}}^{\infty}$ on the unit ball of $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ and considering $\{\mathbf{E}_{\mathbf{m}}^*\}_{\mathbf{m}=\mathbf{1}}^{\infty}$ be any analogous set of duality mapping in $L^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty})$ along with if \overline{B} is $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$. We can find from the Kuelbs lemma that a certain class of Henstock–Kurzweil-type integrable functions are in the Hilbert space H. We are not confident enough that the Henstock–Kurzweil-type integrable functions in H are Henstock–Kurzweil integrable. The known fact is that $\widehat{\mathbf{Q}}_{\mathbf{I}}^{\infty} = \lim_{m \to \infty} \mathbf{Q}_{\mathbf{I}}^{\mathbf{m}} = \bigcup_{\mathbf{k}=\mathbf{1}}^{\infty} \mathbf{Q}_{\mathbf{I}}^{\mathbf{k}}$, where $\mathbf{Q}_{\mathbf{I}}^{\mathbf{n}}$ is the set $\{\mathbf{x} \in \mathbf{R}_{\mathbf{I}}^n\}$ so that the initial (first) co-ordinates $(\mathbf{x}_1, \mathbf{x}_2, \dots, \mathbf{x}_n)$ are rational. Clearly, this is a countable dense set of $\mathbf{R}_{\mathbf{I}}^{\infty}$. For every k and i, we can arrange $\mathbf{Q}_{\mathbf{I}}^{\infty} = \{\mathbf{x}_{\mathbf{1}}, \mathbf{x}_{2}, \dots\}$. Let us assume a closed cube $\mathcal{B}_k(x_i)$ centered at x_i along with edge $e_k = \frac{1}{2^{l-1}\sqrt{m}}, \ l \in \mathbf{N}$.

We can choose a map $\mathbf{N} \times \mathbf{N}$ bijectively to \mathbf{N} of natural order and suppose the resulting set of all closed cubes

$$\{\mathcal{B}_k(x_i)|\ (l,i)\in\mathbf{N}\times\mathbf{N}\}\$$

centered at a point in $\mathbf{Q}_{\mathbf{I}}^{\infty}$ be $\{\mathcal{B}_k : k \in \mathbf{N}\}$. Let the characteristic function of \mathcal{B}_k be $\mathbf{E}_{\mathbf{k}}(\mathbf{x})$ such that for $1 \leq p < \infty$, $\mathbf{E}_{\mathbf{k}}(\mathbf{x}) \in \mathbf{L}^{\mathbf{p}}(\mathbf{R}_{\mathbf{I}}^{\infty}) \cap \mathbf{L}^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty})$. We can define $F_k(.)$ on $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ by

$$F_k(\mathbf{f}) = \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d}\nu_{\infty}(\mathbf{x}).$$

Now, \mathcal{B}_k are cubes with sides parallel to the co-ordinate axes. So, $F_k(.)$ is well defined for all Henstock–Kurzweil integrable functions and is a bounded linear functional on $L^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ for $1 \leq p \leq \infty$. Let us fix $\mathbf{t}_k > 0$ so that $\sum_{k=1}^{\infty} \mathbf{t}_k = 1$. We can define an inner product $\langle . \rangle$ on $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ as follows:

$$egin{aligned} &\left\langle \mathbf{f},\mathbf{g}
ight
angle &=\sum_{k=1}^{\infty}\mathbf{t}_{k}igg(\int_{\mathbf{R}_{\mathbf{I}}^{\infty}}\mathbf{E}_{\mathbf{k}}(\mathbf{x})\mathbf{f}(\mathbf{x})\mathbf{d}
u_{\infty}(\mathbf{x})igg) \ & imesigg(\int_{\mathbf{R}_{\mathbf{I}}^{m}}\mathbf{E}_{\mathbf{k}}(\mathbf{y})\mathbf{g}(\mathbf{y})\mathbf{d}
u_{\infty}(\mathbf{y})igg)^{\mathbf{c}}. \end{aligned}$$

With this inner product, the completion of $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ is the *p*-summable space $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. This space contains Henstock–Kurzweil integrable functions. We can find this as follows:

$$||\mathbf{f}||_{K^2}^2 = \sum_{k=1}^{\infty} \mathbf{t}_k \bigg| \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d} \nu_{\infty}(\mathbf{x}) \bigg|^2.$$

If we replace $\mathbf{R}_{\mathbf{I}}^{n}$ by $\mathbf{R}_{\mathbf{I}}^{\infty}$, then the Alexiewicz norm defined in Eq. (3) becomes

$$\begin{aligned} ||\mathbf{f}||_{HK(\mathbf{R}_{\mathbf{I}}^{\infty})} &= \sup_{r>0} \left| \int_{\mathcal{B}_{r}} \mathbf{f}(x) d\nu_{\infty}(x) \right| < \infty, \end{aligned}$$

and $||\mathbf{f}||_{K^{2}}^{2} \leq \sup_{k} \left| \int_{\mathcal{B}_{k}} \mathbf{f}(x) d\nu_{\infty}(x) \right|^{2} < \infty.$ So, $\mathbf{f} \in K^{2}(\mathbf{R}_{\mathbf{I}}^{\infty}).$

Theorem 11. For every p, $1 \le p \le \infty$, $L^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a densely continuous subspace of $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$.

Proof. Since for every p, $1 \le p \le \infty$, $L^p(\mathbf{R}^m_{\mathbf{I}})$ is a densely continuous subspace of $K^2(\mathbf{R}^n_{\mathbf{I}})$. Moreover, the closure of $\bigcup_{n=1}^{\infty}, K^2(\mathbf{R}^n_{\mathbf{I}})$ is $K^2(\mathbf{R}^\infty_{\mathbf{I}})$. In this way, $K^2(\mathbf{R}^\infty_{\mathbf{I}})$ contains $\bigcup_{n=1}^{\infty} L^p(\mathbf{R}^m_{\mathbf{I}})$ which is dense in $L^p(\mathbf{R}^\infty_{\mathbf{I}})$, as it is the closure.

For $1 \leq p \leq \infty$, the norm of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ spaces can be defined as

$$||\mathbf{f}||_{K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} = \begin{pmatrix} \left(\sum_{k=1}^{\infty} \mathbf{t}_{k} \left| \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) d\nu_{\infty}(\mathbf{x}) \right|^{p} \right)^{\frac{1}{p}}, \text{ for } 1 \leq p < \infty; \\ \sup_{k \geq 1} \left| \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) d\nu_{\infty}(\mathbf{x}) \right|, \text{ for } p = \infty \end{cases}$$

with $\mathbf{x} = (\mathbf{x}_1, \mathbf{x}_2, \mathbf{x}_3, \dots) \in \mathbf{R}_{\mathbf{I}}^{\infty}$. and $\mathbf{R}_{\mathbf{I}}^{\infty} = \lim_{m \to \infty} \mathbf{R}_{\mathbf{I}}^m = \bigcup_{m=1}^{\infty} \mathbf{R}_{\mathbf{I}}^k$ and $\mathbf{R}_{\mathbf{I}}^m = \mathbf{R}^m \times \mathbf{I}_m$, where $\mathbf{I} = [\frac{-1}{2}, \frac{1}{2}]$.

Obviously, $||f||_{K^p}$ defines on \overline{L}^p and with respect to this norm L^p is the completion of K^p . As L^p is the completion of K^p , then we have the following theorem:

Theorem 12. For every q, $1 \leq q \leq \infty$, $L^q(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a densely continuous subspace of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. Also, the inclusion is continuous embedding.

Proof. Proof is similar to Ref. 16 Theorem 2.3. \Box

Remark 3. For each $p, 1 \leq p < \infty$, the Hölder and Minkowski inequalities hold for $L^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. The norm of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is derived from $L^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ and the completion of $L^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. So, for $1 \leq p < \infty$, one can easily show that Hölder and Minkowski inequalities hold in $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$.

Theorem 13. For each $p, 1 \leq p \leq \infty$, the following are true for $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$.

- (a) If $\mathbf{f}_m \to \mathbf{f}$ weakly in $L^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. Then $\mathbf{f}_m \to \mathbf{f}$ strongly in $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$.
- (b) If $1 , then <math>K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is uniformly convex.
- (c) If $1 and <math>\frac{1}{p} + \frac{1}{q} = 1$, then the dual space of $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ is $K^q(\mathbf{R}_{\mathbf{I}}^{\infty})$.
- (d) $K^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty}) \subset K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})$ for $1 \leq p < \infty$.
- (e) $C_c(\mathbf{R}^{\infty}_{\mathbf{I}}) \subset K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ as dense.

Proof. (a) Let $\{\mathbf{f}_m\}$ weakly converge with its limit in $L^p(\mathbf{R}_{\mathbf{I}}^{\infty})$. At this point, for each k, we have

$$\int_{\mathbf{R}_{\mathbf{I}}^m} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) |\mathbf{f}_{\mathbf{m}}(\mathbf{x}) - \mathbf{f}(\mathbf{x})| \mathbf{d}
u_\infty(\mathbf{x}) o \mathbf{0}.$$

Now, for every $\mathbf{f}_m \in K^p(\mathbf{R}^m_{\mathbf{I}})$ for every m, we can have

$$\lim_{m o \infty} \int_{\mathbf{R}_{\mathbf{I}}^m} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) |\mathbf{f}_{\mathbf{m}}(\mathbf{x}) - \mathbf{f}(\mathbf{x})| \mathbf{d}
u_{\infty}(\mathbf{x}) o \mathbf{0}.$$

(b) For each m, $L^p(\mathbf{R}_{\mathbf{I}}^m)$ is uniformly convex. We know for $1 \leq q \leq \infty$, $L^p(\mathbf{R}_{\mathbf{I}}^m)$ is dense in $K^q(\mathbf{R}_{\mathbf{I}}^m)$ with compact embedding. So, for each n, and for $1 \leq q \leq \infty$, $\bigcup_{m=1}^{\infty} L^p(\mathbf{R}_{\mathbf{I}}^m)$ is uniformly convex in $\bigcup_{m=1}^{\infty} K^q(\mathbf{R}_{\mathbf{I}}^m)$ as a dense and compact embedding. But $L^p(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty) = \bigcup_{m=1}^{\infty} L^p(\mathbf{R}_{\mathbf{I}}^m)$. So, for $1 \leq q \leq \infty$, $L^p(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty)$ is uniformly convex in $K^q(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty)$ as dense and compact embedding.

Since the closure of $K^{q}(\widehat{\mathbf{R}}_{\mathbf{I}}^{\infty})$ is $K^{q}(\mathbf{R}_{\mathbf{I}}^{\infty})$, $K^{q}(\mathbf{R}_{\mathbf{I}}^{\infty})$ is uniformly convex.

(c) Using (2), we have for $1 , <math>K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ is reflexive. At the same time, for all k and $\frac{1}{p} + \frac{1}{q} = 1$,

$$\{K^{p}(\mathbf{R}_{\mathbf{I}}^{k})\}^{*} = K^{q}(\mathbf{R}_{\mathbf{I}}^{k})$$
$$K^{p}(\mathbf{R}_{\mathbf{I}}^{k}) \subset K^{p}(\mathbf{R}_{\mathbf{I}}^{k+1}), \Rightarrow \bigcup_{k=1}^{\infty} \left\{K^{p}(\mathbf{R}_{\mathbf{I}}^{k})\right\}^{*}$$
$$= \bigcup_{k=1}^{\infty} K^{q}(\mathbf{R}_{I}^{k}).$$

As each $\mathbf{f} \in K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is the limit of a sequence $\{\mathbf{f}_m\} \subset \bigcup_{k=1}^{\infty} K^p(\mathbf{R}^k_{\mathbf{I}})$, hence for $1 and <math>\frac{1}{p} + \frac{1}{q} = 1$, the dual space of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is $K^q(\mathbf{R}^{\infty}_{\mathbf{I}})$.

(d) Suppose $\mathbf{f} \in K^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty})$. Then for all k, $\left|\int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x})\mathbf{f}(\mathbf{x}) \right|$ $\mathbf{d}\nu_{\infty}(\mathbf{x})$ is uniformly bounded. Consequently, for $1 \leq p < \infty$, $\left|\int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d} \nu_{\infty}(\mathbf{x})\right|^{p}$ is uniformly bounded. Finally, the definition of $K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})$ gives

$$\left[\sum \left|\int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{E}_{\mathbf{k}}(\mathbf{x}) \mathbf{f}(\mathbf{x}) \mathbf{d} \nu_{\infty}(\mathbf{x})\right|^{p}\right]^{\frac{1}{p}} \leq M ||f||_{K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} < \infty$$

Hence, $\mathbf{f} \in K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$.

(e) Since $C_c(\mathbf{R}_{\mathbf{I}}^n) \subset K^p(\mathbf{R}_{\mathbf{I}}^n)$ is dense, so easily we can find $C_c^{\infty}(\mathbf{R}_{\mathbf{I}}^\infty)$ is dense in $K^p(\mathbf{R}_{\mathbf{I}}^\infty)$, $\forall p$.

Corollary 2. $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is strictly convex.

Proposition 1. $C_c^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty})$ is subset of $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ as densely continuous.

Proof. Since $\forall p, C_c^{\infty}(\mathbf{R}_{\mathbf{I}}^m)$ is in $L^p(\mathbf{R}_{\mathbf{I}}^m)$ as dense. Also, $L^p(\mathbf{R}_{\mathbf{I}}^\infty) \subset K^2(\mathbf{R}_{\mathbf{I}}^\infty)$. Hence, $C_c^{\infty}(\mathbf{R}_{\mathbf{I}}^\infty)$ is a subset of $K^2(\mathbf{R}_{\mathbf{I}}^\infty)$ as densely continuous.

Remark 4. $C_0^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty})$ is a dense subset of $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$.

Theorem 14. $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a subset of $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ as dense and continuous embedding.

Proof. Since $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is separable, assume any fixed set of corresponding duality mappings $\{\mathbf{u}_m\} \mathbf{f}_m \in K^q(\mathbf{R}^{\infty}_{\mathbf{I}})$. Then,

$$egin{aligned} \mathbf{f}_m(\mathbf{u}_m) &= \langle \mathbf{u}_m, \mathbf{f}_m
angle \ &= ||\mathbf{u}_m||^2_{K^p(\mathbf{R}^\infty_{\mathbf{I}})} \ &= ||\mathbf{f}_m||^2_{K^q(\mathbf{R}^\infty_{\mathbf{I}})}. \end{aligned}$$

Assume a set of positive numbers $\{r_m\}_{m=1}^{\infty}$ such that $\sum_{m=1}^{\infty} r_m = 1$, also assume

$$\langle \mathbf{u}, \mathbf{v} \rangle = \sum_{m=1}^{\infty} r_m \mathbf{f}_m(\mathbf{u}) \overline{\mathbf{f}_m}(\mathbf{v}).$$

Since $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ is Hilbert space, the completion of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ generated with respect to this inner product is

$$\begin{split} ||\mathbf{u}||^2 &= \sum_{n=1}^{\infty} r_m |\mathbf{f}_m(\mathbf{u})|^2 \\ &\leq \sup_m |\mathbf{f}_m(\mathbf{u})|^2 \\ &= ||\mathbf{u}||_{K^p(\mathbf{R}_m^\infty)}^2. \end{split}$$

Clearly, it is an inner product on $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. Hence, $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is dense in $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ and it is continuous embedding. \Box

Remark 5. We know that $L^1(\mathbf{R}_{\mathbf{I}}^{\infty}) \subset K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$. For $1 , <math>K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ is reflexive. Let the space of bounded finitely additive set functions be $\mathcal{M}(\mathbf{R}_{\mathbf{I}}^{\infty})$ which is defined on the Borel sets $B(\mathbf{R}_{\mathbf{I}}^{\infty})$. Then, $\{L^1(\mathbf{R}_{\mathbf{I}}^{\infty})\}^{**} = \mathcal{M}(\mathbf{R}_{\mathbf{I}}^{\infty}) \subset K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$. In the area of Feynman integral, the Dirac delta measure and the free-particle Green's functions are contained by these spaces.

8. The Space $\mathcal{M}[\mathbf{R}_{\mathrm{I}}^{\infty}]$

We recall that a uniformly bounded sequence $\{\nu_n\} \subset \mathcal{M}(\mathbb{R}^m)$ is weakly convergent to $\nu(\nu_n \to \nu)$ if for each bounded and uniformly continuous function $\mathbf{h}(x)$, the following holds:

$$\int_{\mathbf{R}^m} \mathbf{h}(x) d\nu_m \to \int_{\mathbf{R}^m} \mathbf{h}(x) d\nu_m$$

Definition 12. We called a uniformly bounded sequence $\{\nu_m\} \subset \mathcal{M}(\mathbf{R}_{\mathbf{I}}^{\infty})$ as weakly convergent to ν ($\nu_m \rightarrow \nu$) if the following holds: if for each bounded uniformly continuous function $\mathbf{h}(x)$,

$$\lim_{m \to \infty} \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{h}(x) d\nu_m \to \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{h}(x) d\nu_m$$

Proposition 2. $\nu_m \to \nu$ strongly in $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ when $\nu_m \to \nu$ weakly in $\mathcal{M}(\mathbf{R}^{\infty}_{\mathbf{I}})$.

Proof. As $\nu_n \rightarrow^w \nu$ weakly in $\mathcal{M}(\mathbf{R}_{\mathbf{I}}^{\infty}]$, it gives

$$\lim_{m \to \infty} \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{h}(x) d\nu_m \to \int_{\mathbf{R}_{\mathbf{I}}^{\infty}} \mathbf{h}(x) d\nu \text{ for every } n.$$

Now,

$$\lim_{m \to \infty} ||\nu_m - \nu||_{K^p(\mathbf{R}^\infty_{\mathbf{I}})} = 0.$$

Hence, $\nu_m \to \nu$ in $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ strongly.

9. $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ and Fourier and Convolution Operators

We will extend the theory of operators in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ from $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$. To execute this, we need $GD^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. In the similar approach of the construction of $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ and $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$, we can construct $GD^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. We will mention the construction in brief as follows:

Since $GD^2(\mathbf{R}_{\mathbf{I}}^m) \subset GD^2(\mathbf{R}_{\mathbf{I}}^{m+1}), \forall m$, therefore, $GD^2(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty) = \bigcup_{m=1}^{\infty} GD^2(\mathbf{R}_{\mathbf{I}}^m)$. Like in other spaces in the setting above, we can say $\mathbf{f} \in GD^2(\mathbf{R}_{\mathbf{I}}^\infty)$ if we can find a Cauchy sequence $\{\mathbf{f}_m\} \subset GD^2(\widehat{\mathbf{R}}_{\mathbf{I}}^\infty)$ along with $\mathbf{f}_m \in GD^2(\mathbf{R}_{\mathbf{I}}^\infty)$ and $\lim_{m\to\infty} \mathbf{f}_m(x) = \mathbf{f}(x), \nu_\infty$ -a.e.

Theorem 15. $GD^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a dense subset of $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ with continuous embedding.

Proof. The closure of $\bigcup_{m=1}^{\infty} K^2(\mathbf{R}_{\mathbf{I}}^m)$ is $K^2(\mathbf{R}_{\mathbf{I}}^\infty)$. Moreover, $\bigcup_{m=1}^{\infty} K^2(\mathbf{R}_{\mathbf{I}}^m)$ is in $K^2(\mathbf{R}_{\mathbf{I}}^\infty)$, a dense subset as its closure is $GD^2(\mathbf{R}_{\mathbf{I}}^\infty)$. $GD^2(\mathbf{R}_{\mathbf{I}}^\infty)$ is a dense subset of $K^2(\mathbf{R}_{\mathbf{I}}^\infty)$ with continuous embedding.

Theorem 16. $GD^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a dense subset of $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ with continuous embedding

Proof. Proof is similar to the proof of Theorem 15. \Box

We can find $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is in $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ as dense. We can find the inclusion relation from Theorems 13 and 16 as follows:

$$GD^2(\mathbf{R}^\infty_{\mathbf{I}}) \subset K^p(\mathbf{R}^\infty_{\mathbf{I}}) \subset K^2(\mathbf{R}^\infty_{\mathbf{I}}).$$

 \Box

The separable Banach space $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ is in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ as a dense subset. Moreover, $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ is in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ as a continuous dense embedding. We find that the densely closed linear operator of $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ has a densely closed linear operator in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. Let $L[K^p(\mathbf{R}_{\mathbf{I}}^{\infty})], L[K^2(\mathbf{R}_{\mathbf{I}}^{\infty})]$ be the class of linear operators of the spaces $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ and $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$, respectively.

Theorem 17. Let the separable Banach spaces $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$ be continuous dense embedding in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ with a bounded linear operator A_L on $K^p(\mathbf{R}_{\mathbf{I}}^{\infty})$. Let A_L be self-adjoint in nature on $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ implying A_L is bounded in nature on $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ with $||A_L||_{K^2(\mathbf{R}_{\mathbf{I}}^{\infty})} \leq K_s||A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^{\infty})}$, for a positive constant K_s .

Proof. Let $\mathbf{u} \in \mathbf{K}^{\mathbf{p}}(\mathbf{R}_{\mathbf{I}}^{\infty})$. Let us assume k = 1 and $||\mathbf{u}||_{\mathbf{K}^{2}(\mathbf{R}_{\mathbf{I}}^{\infty})} = \mathbf{1}$. Since A_{L} is self-adjoint, we get

$$\begin{aligned} ||A_L \mathbf{u}||^2_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)} &= \langle A_L \mathbf{u}, \mathbf{A}_L \mathbf{u} \rangle \\ &= \langle \mathbf{u}, \mathbf{A}_L^2 \mathbf{u} \rangle \\ &\leq ||\mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)} ||\mathbf{A}_L^2 \mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)} \\ &= ||A_L^2 \mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)}. \end{aligned}$$

Thus, $||A_L \mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)}^{\mathbf{2m}} \leq ||\mathbf{A}_{\mathbf{L}}^{\mathbf{2m}}\mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)}$ for all *m*. Therefore,

$$\begin{aligned} ||A_{L}\mathbf{u}||_{\mathbf{K}^{2}(\mathbf{R}_{\mathbf{I}}^{\infty})} &\leq \left(||A_{L}^{2m}\mathbf{u}||_{\mathbf{K}^{2}(\mathbf{R}_{\mathbf{I}}^{\infty})} \right)^{\frac{1}{2m}} \\ &\leq \left(||A_{L}^{2m}\mathbf{u}||_{\mathbf{K}^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} \right)^{\frac{1}{2m}} \\ &\leq \left(||A_{L}^{2m}||_{K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} \right)^{\frac{1}{2m}} \left(||\mathbf{u}||_{\mathbf{K}^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} \right)^{\frac{1}{2m}} \\ &\leq ||A_{L}||_{K^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} \left(||\mathbf{u}||_{\mathbf{K}^{p}(\mathbf{R}_{\mathbf{I}}^{\infty})} \right)^{\frac{1}{2m}}. \end{aligned}$$

Since $m \to \infty$, $||A_L \mathbf{u}||_{\mathbf{K}^2(\mathbf{R}_{\mathbf{I}}^\infty)} \le ||\mathbf{A}_{\mathbf{L}}||_{\mathbf{K}^p(\mathbf{R}_{\mathbf{I}}^\infty)}$ for \mathbf{u} in a dense set of the unit ball of $K^2(\mathbf{R}_{\mathbf{I}}^\infty)$. \Box

Theorem 18. Suppose $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$ is a separable Banach space. If $K^p(\mathbf{R}^{\infty}_{\mathbf{I}}) \subset K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$, then a bounded linear operator A_L on $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$

can be extended to $Li[K^2(\mathbf{R}^{\infty}_{\mathbf{I}})]$ with the following:

$$||A_L||_{K^2(\mathbf{R}_{\mathbf{I}}^\infty)} \le k||A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^\infty)},$$

with $k = \inf \left\{ M ||A_L^* A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^\infty)} \le M ||A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^\infty)}^2 \right\}$ for M > 0.

Proof. Let us assume a bounded self-adjoint linear operator A_L on $K^p(\mathbf{R}^{\infty}_{\mathbf{I}})$. Using Lax's theorem (see Ref. 11 Theorem 3.29), A_L is bounded on $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$. Also,

$$\begin{aligned} ||A_L^*A_L||_{K^2(\mathbf{R}_{\mathbf{I}}^\infty)} &\leq ||A_L||_{K^2(\mathbf{R}_{\mathbf{I}}^\infty)}^2 \\ &\leq ||A_L^*A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^\infty)} \\ &\leq k||A_L||_{K^p(\mathbf{R}_{\mathbf{I}}^\infty)}^2. \end{aligned}$$

Recalling that $K^2(\mathbf{R}^m)$ is an effective alternate to formulate the extension of the Feynman operator, it is found that the Fourier and convolution have an extension to the Hilbert space $K^2(\mathbf{R}^m)$. The Schrödinger and Heisenberg theories have faithful representations on $K^2(\mathbf{R}^m)$. We demonstrate that the Fourier and convolution operators can be extended on $K^2(\mathbf{R}^m_{\mathbf{I}})$. Let us consider F_r as the Fourier transform on $K^2(\mathbf{R}^m_{\mathbf{I}})$. Let C_r be the convolution operator on $K^2(\mathbf{R}^\infty_{\mathbf{I}})$.

Theorem 19. The Fourier transform F_r can be extended on $K^2(\mathbf{R}^{\infty}_{\mathbf{I}})$ as a bounded linear operator.

Proof. We know that $C_0(\mathbf{R}_{\mathbf{I}}^{\infty})$ is in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. At the same time $F_r: L^1(\mathbf{R}_{\mathbf{I}}^{\infty}) \to C_0(\mathbf{R}_{\mathbf{I}}^{\infty})$ is a bounded linear operator. Also, we know that $L^{\infty}(\mathbf{R}_{\mathbf{I}}^{\infty}) \subset K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. Again, $L^1(\mathbf{R}_{\mathbf{I}}^{\infty})$ is dense in $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$. Hence, we can claim with Theorem 18 that F_r can be extended on $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ to a bounded linear operator.

Theorem 20. The convolution operator (transformation) C_r can be extended on $K^2(\mathbf{R}_{\mathbf{I}}^{\infty})$ as a bounded linear operator.

Proof. Proof is similar to Theorem 19.

Acknowledgments

We thank the reviewers for careful reading and comments for betterment of the chapter.

References

- A. Alexiewicz, Linear functionals on Denjoy-integrable functions, Collog. Math. 1, 289–293 (1948).
- T. L. Gill and W. W. Zachary, A new class of Banach spaces, J. Phys. A. 41(49), 495206, 15 (2008).
- R. A. Gordon, The integrals of Lebesgue, Denjoy, Perron, and Henstock. Vol. 4, Graduate Studies in Mathematics. American Mathematical Society, Providence, RI (1994).
- R. Henstock, *The General Theory of Integration*. Oxford Mathematical Monographs. The Clarendon Press, Oxford University Press, New York (1991). Oxford Science Publications.
- R. Henstock, Lectures on the Theory of Integration. Vol. 1, Series in Real Analysis. World Scientific Publishing Co., Singapore (1988).
- R. Henstock, A Riemann-type integral of Lebesgue power, Canadian J. Math. 20, 79–87 (1968).
- J. Kurzweil, Henstock-Kurzweil Integration: Its Relation to Topological Vector Spaces. Vol. 7, Series in Real Analysis. World Scientific Publishing Co., Inc., River Edge, NJ (2000).
- 8. J. Kurzweil, Nichtabsolut konvergente Integrale. vol. 26, Teubner-Texte zur Mathematik [Teubner Texts in Mathematics]. BSB B. G. Teubner Verlagsgesellschaft, Leipzig (1980). With English, French and Russian summaries.
- 9. C. Swartz, *Introduction to Gauge Integrals*. World Scientific Publishing Co., Singapore (2001).
- T. Y. Lee, Henstock-Kurzweil Integration on Euclidean Spaces. Vol. 12, Series in Real Analysis. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ (2011).
- T. L. Gill and W. W. Zachary, Functional Analysis and the Feynman Operator Calculus. Springer, Cham (2016).
- J. Kuelbs, Gaussian measures on a Banach space, J. Funct. Anal. 5, 354–367 (1970).
- R. Henstock, *The General Theory of Integration*. Oxford Mathematical Monographs, The Clarendon Press, Oxford University Press, New York (1991). Oxford Science Publications.
- T. H. Hildebrandt, On bounded linear functional operations, Trans. Am. Math. Soc. 36(4), 868–875 (1934).
- T. L. Gill and T. Myers, Constructive analysis on Banach spaces, *Real Anal. Exchange.* 44(1), 1–36 (2019).
- T. L. Gill, H. Kalita, and B. Hazarika, A family of banach spaces over r[∞], Proc. Singapore Nat. Acad. Sci. 14 (2), 1–7 (2020).

- H. Kalita, B. Hazarika, and T. Myers, Kuelbs-Steadman spaces on separable Banach spaces, *Facta Univ. Ser. Math. Inform.* 36(5), 1065–1077 (2021).
- Y. Yamasaki, Measures on Infinite-Dimensional Spaces. Vol. 5, Series in Pure Mathematics. World Scientific Publishing Co., Singapore (1985).
- L. Gross. Abstract Wiener spaces. In Proc. Fifth Berkeley Sympos. Math. Statist. and Probability (Berkeley, CA, 1965/66), Vol. II: Contributions to Probability Theory, Part 1, pp. 31–42. Univ. California Press, Berkeley, CA (1967).
- 20. V. Y. Steadman, A theory of operators on Banach spaces. ProQuest LLC, Ann Arbor, MI (1988). Thesis (Ph.D.)–Howard University.

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0016

Chapter 16

Subclass of Analytic Functions Involving Mittag-Leffler Type Borel Distribution Series

P. Thirupathi Reddy^{*,†,§} and B. Venkateswarlu^{‡,††}

[†]Department of Mathematics, D.R.K. Institute of Science & Technology, Bowrampet 500 043, Hyderabad, India [‡]Department of Mathematics, GSS, GITAM University, Doddaballapur - 562 163, Bengaluru Rural, Karnataka, India [§]reddypt2@gmail.com ^{††}bvlmaths@gmail.com

In this chapter, we discuss a subclass $TS_{\varphi,\varsigma}^{\ \xi}(\vartheta,\hbar,\ell)$ of univalent mappings with negative coefficients related to Borel distribution series in the unit disk $U = \{\omega \in C : |\omega| < 1\}$. For mappings in our class, we obtain fundamental properties, such as coefficient inequality, distortion and covering theorems, radii of starlikeness, convexity and close-to-convexity, extreme points, Hadamard product, and closure theorems.

1. Introduction

Let \mathcal{A} indicate the class of all mappings $\aleph(\omega)$ of the type

$$\aleph(\omega) = \omega + \sum_{\kappa=2}^{\infty} a_{\kappa} \omega^{\kappa} \tag{1}$$

^{*}Corresponding author.

in the open, the unit disk $\mathbb{U} = \{\omega : |\omega| < 1\}$. Let S be the subclass of \mathcal{A} consisting of univalent mappings satisfying the following usual normalization condition $\aleph(0) = 0$ and $\aleph'(0) = 1$. We denote by S the subclass of \mathcal{A} consisting of $\aleph(\omega)$ which are all univalent in \mathbb{U} . A mapping $\aleph \in \mathcal{A}$ is a starlike mapping of the order $v, v(0 \le v < 1)$ if it satisfies

$$\Re\left\{\frac{\omega\aleph'(\omega)}{\aleph(\omega)}\right\} > \upsilon, \quad (\omega \in \mathbb{U}),$$
(2)

we denote this class by $S^*(v)$.

A mapping $\aleph \in \mathcal{A}$ is a convex mapping of the order $v, v(0 \le v < 1)$ if it satisfies

$$\Re\left\{1+\frac{\omega\aleph''(\omega)}{\aleph'(\omega)}\right\} > \upsilon, \quad (\omega \in \mathbb{U}),$$
(3)

we denote this class with K(v).

Denote by T the subclass of $\mathcal A$ consisting of mappings \aleph of the form

$$\aleph(\omega) = \omega - \sum_{\kappa=2}^{\infty} a_{\kappa} \omega^{\kappa}, \qquad (a_{\kappa} \ge 0).$$
(4)

Silverman pioneered and extensively researched this subclass.¹

For $\aleph \in \mathcal{A}$ given by (1) and $g(\omega)$ given by

$$g(\omega) = \omega + \sum_{\kappa=2}^{\infty} b_{\kappa} \omega^{\kappa}$$
(5)

their convolution (or Hadamard product), indicated by $(\aleph \ast g),$ is defined as

$$(\aleph * g)(\omega) = \omega + \sum_{\kappa=2}^{\infty} a_{\kappa} b_{\kappa} \omega^{\kappa} = (g * \aleph)(\omega), \quad (\omega \in \mathbb{U}).$$
(6)

Note that $\aleph * g \in \mathcal{A}$.

1.1. Mittag-Leffler type Borel distribution

Geometric mapping theory, complex analysis, and other related fields rely heavily on operator theory. Convolution of certain analytic mappings can express a wide range of derivative and integral operators. Let $E_{\varsigma}(\omega)$ and $E_{\varsigma,\xi}(\omega)$ be mappings defined by

$$E_{\varsigma}(\omega) = \sum_{\kappa=0}^{\infty} \frac{\omega^{\kappa}}{\Gamma(\varsigma\kappa+1)}, \quad (\omega \in \mathbb{C}, \ \Re(\varsigma) > 0)$$

and

$$E_{\varsigma,\xi}(\omega) = \frac{1}{\Gamma(\xi)} + \sum_{\kappa=1}^{\infty} \frac{\omega^{\kappa}}{\Gamma(\varsigma\kappa + \xi)} \quad (\varsigma, \xi \in \mathbb{C}, \Re(\varsigma) > 0, \Re(\xi) > 0).$$

It can also be written in another way:

$$\begin{split} E_{\varsigma,\xi}(\omega) &= \frac{1}{\Gamma(\xi)} + \sum_{\kappa=2}^{\infty} \frac{\omega^{\kappa-1}}{\Gamma(\varsigma(\kappa-1)+\xi)}, \\ &\times (\varsigma,\xi \in \mathbb{C}, \Re(\varsigma) > 0, \Re(\xi) > 0). \end{split}$$

Mittag-Leffler² acquainted the mapping $E_{\varsigma}(\omega)$ and it is thus renowned as the Mittag-Leffler mapping.

A more general mapping $E_{\zeta,\xi}$ generalizing $E_{\zeta}(\omega)$ was acquainted by Wiman³ and specified by

$$E_{\varsigma,\xi}(\omega) = \sum_{\kappa=0}^{\infty} \frac{\omega^{\kappa}}{\Gamma(\varsigma\kappa+\xi)}, \quad (\omega,\varsigma,\xi\in\mathcal{C}, \Re(\varsigma) > 0, \Re(\xi) > 0).$$

As a special case, the mapping $E_{\varsigma,\xi}$ possesses many well-known mappings, for example,

$$E_{1,1}(\omega) = e^{\omega}, \quad E_{1,2}(\omega) = \frac{e^{\omega} - 1}{\omega},$$

$$E_{2,1}(\omega^2) = \cosh \omega, \quad E_{2,1}(-\omega^2) = \cos \omega, \quad E_{2,2}(\omega^2) = \frac{\sinh \omega}{\omega},$$

$$E_{2,2}(-\omega^2) = \frac{\sin \omega}{\omega}c,$$

$$E_{3}(\omega) = \frac{1}{2} \left[e^{\omega^{1/3}} + 2e^{-\frac{1}{2}\omega^{1/3}} \cos\left(\frac{\sqrt{3}}{2}\omega^{1/3}\right) \right],$$

$$E_{4}(\omega) = \frac{1}{2} \left[\cos \omega^{1/4} + \cosh \omega^{1/4} \right].$$

The Mittag-Leffler mapping appears naturally when solving fractional order differential and integral equations. There are many other properties of the Mittag-Leffler mapping and the generalized Mittag-Leffler mapping that can be realized, e.g. in Refs. 4–10. Observe that Mittag-Leffler mapping, $E_{\zeta,\xi}(\omega)$, does not belong to the family \mathcal{A} . Thus, it is natural to consider the following normalization of Mittag-Leffler function as below:

$$E_{\varsigma,\xi}(\omega) = \omega \Gamma(\xi) E_{\varsigma,\xi}(\omega) = \omega + \sum_{\kappa=2}^{\infty} \frac{\Gamma(\xi)}{\Gamma(\varsigma(\kappa-1)+\xi)} \omega^{\kappa}, \qquad (7)$$

is holds for complex parameters ς, ξ and $\omega \in \mathbb{C}$. In this chapter, we focus on the case of real-valued mappings ς, ξ and $\omega \in \mathbb{U}$.

Elementary distributions such as Pascal, Poisson, logarithmic, binomial, and beta negative binomial have been examined theoretically in geometric function theory. We suggest readers to see Refs.11 and 12 for a more complete research.

Wanas and Khuttar¹³ recently added the Borel distribution (BD), whose probability mass mapping is

$$P(x = \rho) = \frac{(\rho \wp)^{\rho - 1} e^{-\wp \rho}}{\rho!}, \quad \rho = 1, 2, 3, \dots$$

and they added a series $\mathcal{M}_{\wp}(\omega)$ whose coefficients are probabilities of the Borel distribution (BD):

$$\mathcal{M}_{\wp}(\omega) = \omega + \sum_{\kappa=2}^{\infty} \frac{[\wp(\kappa-1)]^{\kappa-2} e^{-\wp(\kappa-1)}}{(\kappa-1)!} \omega^{\kappa}, (0 < \wp \le 1).$$
(8)

The above series is convergent with the domain of convergence of the entire complex plane, according to a well-known ratio test.

The Mittag-Leffler-type Borel distribution was described as follows:

$$\mathcal{P}_{\wp}(\varsigma,\xi;\rho) = \frac{(\wp\rho)^{\rho-1}}{E_{\varsigma,\xi}(\wp\rho)\Gamma(\varsigma\rho+\xi)}, \quad \rho = 0, 1, 2, \dots,$$

where

$$E_{\varsigma,\xi}(\omega) = \sum_{\kappa=0}^{\infty} \frac{\omega^{\kappa}}{\Gamma(\varsigma\kappa+\xi)} \quad (\varsigma, \ \xi \in \mathbb{C}, \ \Re(\varsigma) > 0, \ \Re(\xi) > 0).$$

Thus, by using (7) and (8) and the convolution operator, the Mittag-Leffler-type Borel distribution series is specified as

$$\begin{aligned} \mathscr{B}_{\wp}(\varsigma,\xi)(\omega) &= \omega + \sum_{\kappa=2}^{\infty} \frac{[\wp(\kappa-1)]! [\wp(\kappa-1)]^{\kappa-2} e^{-\wp(\kappa-1)}}{(\kappa-1)! E_{\varsigma,\xi}(\wp(\kappa-1)) \Gamma(\varsigma(\kappa-1)+\xi)} \\ &\times \omega^{\kappa} (0 < \wp \le 1). \end{aligned}$$

Further, by the convolution operator, we specify

$$\mathcal{B}_{\wp}(\varsigma,\xi)\aleph(\omega) = \mathcal{B}_{\wp}(\varsigma,\xi)(\omega) * \aleph(\omega)$$

$$= \omega + \sum_{\kappa=2}^{\infty} \frac{[\wp(\kappa-1)]![\wp(\kappa-1)]^{\kappa-2}e^{-\wp(\kappa-1)}}{(\kappa-1)!E_{\varsigma,\xi}(\wp(\kappa-1))\Gamma(\varsigma(\kappa-1)+\xi)} a_{\kappa}\omega^{\kappa}$$

$$= \omega + \sum_{\kappa=2}^{\infty} \Lambda(\kappa)a_{\kappa}\omega^{\kappa}, \quad (\varsigma,\xi\in\mathbb{C},\Re(\varsigma) > 0,$$

$$\Re(\xi) > 0, 0 < \wp \le 1), \tag{9}$$

where

$$\Lambda(\kappa) = \frac{[\wp(\kappa-1)]![\wp(\kappa-1)]^{\kappa-2} e^{-\wp(\kappa-1)}}{(\kappa-1)! E_{\varsigma,\xi}(\wp(\kappa-1)) \Gamma(\varsigma(\kappa-1)+\xi)}.$$
 (10)

If $\aleph \in T$ is indicated by (1), then we have

$$\mathscr{B}_{\wp}(\varsigma,\xi)\aleph(\omega) = \omega - \sum_{\kappa=2}^{\infty} \Lambda(\kappa) a_{\kappa} \omega^{\kappa}, \qquad (11)$$

where $\Lambda(\kappa)$ is indicated by (9).

The study of operators is essential in geometric function theory. Many differential and integral operators can be expressed as convolutions of analytic functions. This approach is found to facilitate further mathematical research and to increase comprehension of the geometric and symmetric features of such operators. The significance of convolution in operator theory may be clearly appreciated from the work in Ref. 14. Furthermore, probability is applied in a wide range of real-life applications from insurance to forecasting and politics to economic forecasting. We recommend Ref. 15 to the reader for other applications. For more details, see Refs. 16–21. Now, we establish a new subclass of mappings in the class \mathcal{A} by using the Mittag-Leffler-type Borel distribution series $\mathscr{B}\wp(\varsigma,\xi)$,

Definition 1. For $0 \leq \vartheta < 1, 0 \leq \hbar < 1, 0 < \ell < 1, \delta \geq 0, \sigma \in [0,1]$ and $\varrho \in \mathbb{N}$, let $TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell)$ be the subclass of \aleph consisting of mappings of the form (4) and its geometrical condition satisfies

$$\left| \frac{\vartheta \left((\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega))' - \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega} \right)}{\hbar \left(\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega) \right)' + (1 - \vartheta) \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega}}{\omega} \right| < \ell, \quad \omega \in \mathbb{U},$$

where $\mathscr{B}^{\xi}_{\wp,\varsigma} \aleph(\omega)$ is given by (9).

2. Coefficient Inequality

Theorem 1. Let $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$ if and only if

$$\sum_{\kappa=2}^{\infty} [\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)a_{\kappa} \le \ell(\hbar + (1 - \vartheta)), \qquad (12)$$

where $0 < \ell < 1, 0 \leq \vartheta < 1, 0 \leq \hbar < 1, \delta \geq 0, \sigma \in [0, 1]$ and $\varrho \in \mathbb{N}$. The result (12) is sharp for the mapping

$$\aleph(\omega) = \omega - \frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}\omega^{\kappa}, \quad \kappa \ge 2.$$

Proof. Suppose (12) holds true and $|\omega| = 1$. Then we obtain

$$\begin{aligned} \left| \vartheta \left((\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega))' - \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega} \right) \right| \\ &- \ell \left| \hbar \left(\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega))' + (1 - \vartheta) \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega} \right) \right| \\ &= \left| -\vartheta \sum_{\kappa=2}^{\infty} (\kappa - 1) \Lambda(\kappa) a_{\kappa} \omega^{\kappa - 1} \right| \\ &- \ell \left| \hbar + (1 - \vartheta) - \sum_{\kappa=2}^{\infty} (\kappa \hbar + 1 - \vartheta) \Lambda(\kappa) a_{\kappa} \omega^{\kappa - 1} \right| \end{aligned}$$

$$\leq \sum_{\kappa=2}^{\infty} [\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)a_{\kappa} - \ell(\hbar + (1 - \vartheta))$$

$$\leq 0.$$

Hence, by maximum modulus principle, $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Now, assume that $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$ so that

$$\left|\frac{\vartheta\left((\mathscr{B}_{\wp,\varsigma}^{\xi}\aleph(\omega))'-\frac{\mathscr{B}_{\wp,\varsigma}^{\xi}\aleph(\omega)}{\omega}\right)}{\hbar(\mathscr{B}_{\wp,\varsigma}^{\xi}\aleph(\omega))'+(1-\vartheta)\frac{\mathscr{B}_{\wp,\varsigma}^{\xi}\aleph(\omega)}{\omega}}\right| < \ell, \quad \omega \in \mathbb{U}.$$

Hence,

$$\left| \vartheta \left((\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega))' - \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega} \right) \right| \\ < \ell \left| \hbar \left(\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega))' + (1 - \vartheta) \frac{\mathscr{B}_{\wp,\varsigma}^{\xi} \aleph(\omega)}{\omega} \right) \right|.$$

Therefore, we get

$$\left| -\sum_{\kappa=2}^{\infty} \vartheta(\kappa-1)\Lambda(\kappa)a_{\kappa}\omega^{\kappa-1} \right|$$

< $\ell \left| \hbar + (1-\vartheta) - \sum_{\kappa=2}^{\infty} (\kappa\hbar + 1 - \vartheta)\Lambda(\kappa)a_{\kappa}\omega^{\kappa-1} \right|.$

Thus,

$$\sum_{\kappa=2}^{\infty} [\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)a_{\kappa} \le \ell(\hbar + (1 - \vartheta)),$$

and this completes the proof.

Corollary 1. Let $\aleph \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell)$. Then

$$a_{\kappa} \leq \frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)} \omega^{\kappa}, \quad \kappa \geq 2.$$

3. Distortion and Covering Theorem

Theorem 2. Let the mapping $\aleph \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell)$. Then

$$\begin{split} |\omega| &- \frac{\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega|^2 \\ &\leq |\aleph(\omega)| \\ &\leq |\omega| + \frac{\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega|^2 \end{split}$$

The result is sharp and attained as

$$\aleph(\omega) = \omega - \frac{\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]}\omega^2.$$

Proof.

$$|\aleph(\omega)| = \left| \omega - \sum_{\kappa=2}^{\infty} a_{\kappa} \omega^{\kappa} \right| \le |\omega| + \sum_{\kappa=2}^{\infty} a_{\kappa} |\omega|^{\kappa}$$
$$\le |\omega| + |\omega|^2 \sum_{\kappa=2}^{\infty} a_{\kappa}.$$

By Theorem 1, we get

$$\sum_{\kappa=2}^{\infty} a_{\kappa} \le \frac{\ell(\hbar + (1-\vartheta))}{[\vartheta + \ell(2\hbar + 1 - \vartheta)]\Lambda(\kappa)}.$$
(13)

Thus,

$$|\aleph(\omega)| \le |\omega| + \frac{\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega|^2.$$

Also,

$$\begin{split} \aleph(\omega)| &\geq |\omega| - \sum_{\kappa=2}^{\infty} a_{\kappa} |\omega|^{\kappa} \\ &\geq |\omega| - |\omega|^2 \sum_{\kappa=2}^{\infty} a_{\kappa} \\ &\geq |\omega| - \frac{\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega|^2. \end{split}$$

Then the proof of the theorem follows.

Theorem 3. Let $\aleph \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell)$. Then

$$\begin{aligned} 1 - \frac{2\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega| &\leq |\aleph'(\omega)| \\ &\leq 1 + \frac{2\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]} |\omega| \end{aligned}$$

with equality for

$$\aleph(\omega) = \omega - \frac{2\ell(\hbar + (1 - \vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]}\omega^2.$$

Proof. Note that

$$\Lambda(2,\varrho,\sigma,\delta)[\vartheta + \ell(2\hbar + 1 - \vartheta)] \sum_{\kappa=2}^{\infty} \kappa a_{\kappa}$$

$$\leq \sum_{\kappa=2}^{\infty} \kappa[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)a_{\kappa}$$

$$\leq \ell(\hbar + (1 - \vartheta)) \tag{14}$$

from Theorem 1. Thus,

$$|\aleph'(\omega)| = \left| 1 - \sum_{\kappa=2}^{\infty} \kappa a_{\kappa} \omega^{\kappa-1} \right|$$

$$\leq 1 + \sum_{\kappa=2}^{\infty} \kappa a_{\kappa} |\omega|^{\kappa-1}$$

$$\leq 1 + |\omega| \sum_{\kappa=2}^{\infty} \kappa a_{\kappa}$$

$$\leq 1 + |\omega| \frac{2\ell(\hbar + (1-\vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]}.$$
 (15)

On the other hand,

$$|\aleph'(\omega)| = \left| 1 - \sum_{\kappa=2}^{\infty} \kappa a_{\kappa} \omega^{\kappa-1} \right|$$

$$\geq 1 - \sum_{\kappa=2}^{\infty} \kappa a_{\kappa} |\omega|^{\kappa-1}$$

$$\geq 1 - |\omega| \sum_{\kappa=2}^{\infty} \kappa a_{\kappa}$$

$$\geq 1 - |\omega| \frac{2\ell(\hbar + (1-\vartheta))}{\Lambda(2)[\vartheta + \ell(2\hbar + 1 - \vartheta)]}.$$
(16)

Combining (15) and (16), we get the result.

4. Radii of Starlikeness, Convexity and Close-to-Convexity

Theorem 4. Let $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then \aleph is starlike in $|\omega| < R_1$ of order ϑ , $0 \leq \vartheta < 1$, where

$$R_{1} = \inf_{\kappa} \left\{ \frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa, \wp, \vartheta)}{(\kappa-\vartheta)\ell(\hbar + (1-\vartheta))} \right\}^{\frac{1}{\kappa-1}}, \quad \kappa \ge 2.$$
(17)

Proof. \aleph is starlike of order $\vartheta, 0 \leq \vartheta < 1$ if

$$\Re\left\{\frac{\omega\aleph'(\omega)}{\aleph(\omega)}\right\} > \vartheta.$$

Thus, it is enough to show that

$$\left|\frac{\omega\aleph'(\omega)}{\aleph(\omega)} - 1\right| = \left|\frac{-\sum_{\kappa=2}^{\infty} (\kappa - 1)a_{\kappa}\omega^{\kappa - 1}}{1 - \sum_{\kappa=2}^{\infty} a_{\kappa}\omega^{\kappa - 1}}\right| \le \frac{\sum_{\kappa=2}^{\infty} (\kappa - 1)a_{\kappa}|\omega|^{\kappa - 1}}{1 - \sum_{\kappa=2}^{\infty} a_{\kappa}|\omega|^{\kappa - 1}}.$$

Thus,

$$\left|\frac{\omega \aleph'(\omega)}{\aleph(\omega)} - 1\right| \le 1 - \vartheta \quad \text{if} \quad \sum_{\kappa=2}^{\infty} \frac{(\kappa - \vartheta)}{(1 - \vartheta)} a_{\kappa} |\omega|^{\kappa - 1} \le 1.$$
(18)

Hence, by Theorem 1, (18) will be true if

$$\frac{\kappa - \vartheta}{1 - \vartheta} |\omega|^{\kappa - 1} \le \frac{(\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))}$$

or if

$$|\omega| \le \left[\frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{(\kappa-\vartheta)\ell(\hbar + (1-\vartheta))}\right]^{\frac{1}{\kappa-1}}, \kappa \ge 2.$$
(19)

The theorem follows easily from (19).

Theorem 5. Let $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then \aleph is convex in $|\omega| < R_2$ of order $\vartheta, 0 \leq \vartheta < 1$, where

$$R_{2} = \inf_{\kappa} \left\{ \frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\kappa(\kappa-\vartheta)\ell(\hbar + (1-\vartheta))} \right\}^{\frac{1}{\kappa-1}}, \ \kappa \ge 2.$$
(20)

Proof. \aleph is convex of order $\vartheta, 0 \le \vartheta < 1$ if

$$\Re\left\{1+\frac{\omega\aleph''(\omega)}{\aleph'(\omega)}\right\} > \vartheta.$$

As a result, demonstrating that

$$\left|\frac{\omega\aleph''(\omega)}{\aleph'(\omega)}\right| = \left|\frac{-\sum_{\kappa=2}^{\infty}\kappa(\kappa-1)a_{\kappa}\omega^{\kappa-1}}{1-\sum_{\kappa=2}^{\infty}\kappa a_{\kappa}\omega^{\kappa-1}}\right| \le \frac{\sum_{\kappa=2}^{\infty}\kappa(\kappa-1)a_{\kappa}|\omega|^{\kappa-1}}{1-\sum_{\kappa=2}^{\infty}\kappa a_{\kappa}|\omega|^{\kappa-1}}.$$

Thus,

$$\left|\frac{\omega\aleph''(\omega)}{\aleph'(\omega)}\right| \le 1 - \vartheta \quad \text{if} \quad \sum_{\kappa=2}^{\infty} \frac{\kappa(\kappa - \vartheta)}{(1 - \vartheta)} a_{\kappa} |\omega|^{\kappa - 1} \le 1.$$
(21)

Hence, by Theorem 1, (21) will be true if

$$\frac{\kappa(\kappa-\vartheta)}{1-\vartheta}|\omega|^{\kappa-1} \leq \frac{(\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta))\Lambda(\kappa)}{\ell(\hbar+(1-\vartheta)}$$

or if

$$|\omega| \le \left[\frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\kappa(\kappa - \vartheta)\ell(\hbar + (1 - \vartheta))}\right]^{\frac{1}{\kappa-1}}, \kappa \ge 2.$$
 (22)

The theorem follows easily from (22).

Theorem 6. Let $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then \aleph is close-to-convex in $|\omega| < R_3$ of order ϑ , $0 \le \vartheta < 1$, where

$$R_{3} = \inf_{\kappa} \left\{ \frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa, \wp, \vartheta)}{\kappa\ell(\hbar + (1 - \vartheta))} \right\}^{\frac{1}{\kappa-1}}, \ \kappa \ge 2.$$
(23)

Proof. \aleph is close-to-convex of order $\vartheta, 0 \leq \vartheta < 1$ if

$$\Re\left\{\aleph'(\omega)\right\} > \vartheta.$$

Thus, it is enough to show that

$$|\aleph'(\omega) - 1| = \left| -\sum_{\kappa=2}^{\infty} \kappa a_{\kappa} \omega^{\kappa-1} \right| \le \sum_{\kappa=2}^{\infty} \kappa a_{\kappa} |\omega|^{\kappa-1}$$

Thus,

$$|\aleph'(\omega) - 1| \le 1 - \vartheta \quad \text{if} \quad \sum_{\kappa=2}^{\infty} \frac{\kappa}{(1-\vartheta)} a_{\kappa} |\omega|^{\kappa-1} \le 1.$$
 (24)

Hence, by Theorem 1, (24) will be true if

$$\frac{\kappa}{1-\vartheta}|\omega|^{\kappa-1} \le \frac{(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\ell(\hbar + (1-\vartheta))}$$

or if

$$|\omega| \le \left[\frac{(1-\vartheta)(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\kappa\ell(\hbar + (1-\vartheta))}\right]^{\frac{1}{\kappa-1}}, \kappa \ge 2.$$
 (25)

The theorem follows easily from (25).

5. Extreme Points

Theorem 7. Let $\aleph_1(\omega) = \omega$ and

$$\aleph_{\kappa}(\omega) = \omega - \frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)} \omega^{\kappa}, \quad for \quad \kappa = 2, 3, \dots$$

Then $\aleph \in TS^{\xi}_{\wp,\varsigma}(\vartheta,\hbar,\ell)$ if and only if it can be expressed in the form

$$\aleph(\omega) = \sum_{\kappa=1}^{\infty} \theta_{\kappa} \aleph_{\kappa}(\omega), \quad where \quad \theta_{\kappa} \ge 0 \quad and \quad \sum_{\kappa=1}^{\infty} \theta_{\kappa} = 1.$$

Proof. Assume that $\aleph(\omega) = \sum_{\kappa=1}^{\infty} \theta_{\kappa} \aleph_{\kappa}(\omega)$, hence we get

$$\aleph(\omega) = \omega - \sum_{\kappa=2}^{\infty} \frac{\ell(\hbar + (1 - \vartheta))\theta_{\kappa}}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)} \omega^{\kappa}.$$

Now, $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$, since

$$\sum_{\kappa=2}^{\infty} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} \times \frac{\ell(\hbar + (1 - \vartheta))\theta_{\kappa}}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}$$
$$= \sum_{\kappa=2}^{\infty} \theta_{\kappa} = 1 - \theta_{1} \le 1.$$

Conversely, suppose $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then we show that \aleph can be written in the form $\sum_{\kappa=1}^{\infty} \theta_{\kappa} \aleph_{\kappa}(\omega)$. Now, $\aleph \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$ implies, from Theorem 1,

$$a_{\kappa} \leq \frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}.$$

Setting $\theta_{\kappa} = \frac{[\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta)]\Lambda(\kappa)}{\ell(\hbar+(1-\vartheta))}a_{\kappa}, \kappa = 2, 3, \dots$ and $\theta_{1} = 1 - \sum_{\kappa=2}^{\infty}\theta_{\kappa}$, we obtain $\aleph(\omega) = \sum_{\kappa=1}^{\infty}\theta_{\kappa}\aleph_{\kappa}(\omega)$.

6. Hadamard Product

Theorem 8. Let $\aleph, g \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then $\aleph * g \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$ for

$$\begin{split} \aleph(\omega) &= \omega - \sum_{\kappa=2}^{\infty} a_{\kappa} \omega^{\kappa}, g(\omega) = \omega - \sum_{\kappa=2}^{\infty} b_{\kappa} \omega^{\kappa} \quad and \ (\aleph * g)(\omega) \\ &= \omega - \sum_{\kappa=2}^{\infty} a_{\kappa} b_{\kappa} \omega^{\kappa}, \end{split}$$

where

$$\zeta \ge \frac{\ell^2(\hbar + (1 - \vartheta))\vartheta(\kappa - 1)}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]^2\Lambda(\kappa) - \ell^2(\hbar + (1 - \vartheta))(\kappa\hbar + 1 - \vartheta)}$$

Proof. $\aleph \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell)$ and so

$$\sum_{\kappa=2}^{\infty} \frac{\left[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)\right]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} a_{\kappa} \le 1$$
(26)

and

$$\sum_{\kappa=2}^{\infty} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} b_{\kappa} \le 1.$$
(27)

We have to find the smallest number ζ such that

$$\sum_{\kappa=2}^{\infty} \frac{\left[\vartheta(\kappa-1) + \zeta(\kappa\hbar + 1 - \vartheta)\right]\Lambda(\kappa)}{\zeta(\hbar + (1 - \vartheta))} a_{\kappa} b_{\kappa} \le 1.$$
(28)

By Cauchy–Schwarz inequality

$$\sum_{\kappa=2}^{\infty} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} \sqrt{a_{\kappa}b_{\kappa}} \le 1.$$
(29)

Therefore, it is enough to show that

$$\frac{[\vartheta(\kappa-1)+\zeta(\kappa\hbar+1-\vartheta)]\Lambda(\kappa)}{\zeta(\hbar+(1-\vartheta))}a_{\kappa}b_{\kappa}$$
$$\leq \frac{[\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta)]\Lambda(\kappa)}{\ell(\hbar+(1-\vartheta))}\sqrt{a_{\kappa}b_{\kappa}},$$

i.e.

$$\sqrt{a_{\kappa}b_{\kappa}} \le \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\zeta}{[\vartheta(\kappa-1) + \zeta(\kappa\hbar + 1 - \vartheta)]\ell}.$$
(30)

From (29),

$$\sqrt{a_{\kappa}b_{\kappa}} \leq \frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}.$$

Thus, it is enough to show that

$$\frac{\ell(\hbar + (1 - \vartheta))}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)} \le \frac{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]\zeta}{[\vartheta(\kappa - 1) + \zeta(\kappa\hbar + 1 - \vartheta)]\ell}$$

which simplifies to

$$\zeta \ge \frac{\ell^2(\hbar + (1 - \vartheta))\vartheta(\kappa - 1)}{[\vartheta(\kappa - 1) + \ell(\kappa\hbar + 1 - \vartheta)]^2\Lambda(\kappa) - \ell^2(\hbar + (1 - \vartheta))(\kappa\hbar + 1 - \vartheta)}.$$

7. Closure Theorems

Theorem 9. Let $\aleph_j \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell), j = 1, 2, \dots, s$. Then

$$\aleph_g(\omega) = \sum_{j=1}^s c_j \aleph_j(\omega) \in T \ S^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell).$$

For
$$\aleph_j(\omega) = \omega - \sum_{\kappa=2}^{\infty} a_{\kappa,j} \omega^{\kappa}$$
, where $\sum_{j=1}^{s} c_j = 1$.

Proof. Now,

$$g(\omega) = \sum_{j=1}^{s} c_j \aleph_j(\omega)$$
$$= \omega - \sum_{\kappa=2}^{\infty} \sum_{j=1}^{s} c_j a_{\kappa,j} \omega^{\kappa}$$
$$= \omega - \sum_{\kappa=2}^{\infty} e_{\kappa} \omega^{\kappa},$$
where
$$e_{\kappa} = \sum_{j=1}^{s} c_j a_{\kappa,j}$$
. Thus, $g(\omega) \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$ if
$$\sum_{\kappa=2}^{\infty} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} e_{\kappa} \leq 1,$$

i.e. if

$$\sum_{\kappa=2}^{\infty} \sum_{j=1}^{s} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} c_j a_{\kappa,j}$$
$$= \sum_{j=1}^{s} c_j \sum_{\kappa=2}^{\infty} \frac{[\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta)]\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} a_{\kappa,j}$$
$$\leq \sum_{j=1}^{s} c_j = 1.$$

Theorem 10. Let $\aleph, g \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$. Then

$$h(\omega) = \omega - \sum_{\kappa=2}^{\infty} (a_{\kappa}^2 + b_{\kappa}^2) \omega^{\kappa} \in TS^{\xi}_{\wp,\varsigma}(\vartheta, \hbar, \ell), where$$

$$\zeta \geq \frac{2\vartheta(\kappa-1)\ell^2(\hbar+(1-\vartheta))}{[\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta)]^2\Lambda(\kappa)-2\ell^2(\hbar+(1-\vartheta))(\kappa\hbar+1-\vartheta)}.$$

Proof. Since $\aleph, g \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell)$, Theorem 1 yields

$$\sum_{\kappa=2}^{\infty} \left[\frac{(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} a_{\kappa} \right]^2 \le 1$$

and

$$\sum_{\kappa=2}^{\infty} \left[\frac{(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} b_{\kappa} \right]^2 \le 1.$$

We obtain from the last two inequalities

$$\sum_{\kappa=2}^{\infty} \frac{1}{2} \left[\frac{(\vartheta(\kappa-1) + \ell(\kappa\hbar + 1 - \vartheta))\Lambda(\kappa)}{\ell(\hbar + (1 - \vartheta))} \right]^2 (a_{\kappa}^2 + b_{\kappa}^2) \le 1.$$
(31)

But $h(\omega) \in TS_{\wp,\varsigma}^{\xi}(\vartheta, \hbar, \ell, \zeta)$ if and only if

$$\sum_{\kappa=2}^{\infty} \frac{\left[\vartheta(\kappa-1) + \zeta(\kappa\hbar + 1 - \vartheta)\right]\Lambda(\kappa)}{\zeta(\hbar + (1 - \vartheta))} (a_{\kappa}^2 + b_{\kappa}^2) \le 1, \qquad (32)$$

where $0 < \zeta < 1$, however, (31) implies (32) if

$$\frac{[\vartheta(\kappa-1)+\zeta(\kappa\hbar+1-\vartheta)]\Lambda(\kappa)}{\zeta(\hbar+(1-\vartheta))} \le \frac{1}{2} \left[\frac{(\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta))\Lambda(\kappa)}{\ell(\hbar+(1-\vartheta))}\right]^2.$$

Simplifying, we get

$$\zeta \geq \frac{2\vartheta(\kappa-1)\ell^2(\hbar+(1-\vartheta))}{[\vartheta(\kappa-1)+\ell(\kappa\hbar+1-\vartheta)]^2\Lambda(\kappa)-2\ell^2(\hbar+(1-\vartheta))(\kappa\hbar+1-\vartheta)}.$$

8. Concluding Remarks

This research has introduced the study of the Mittag-Leffler-type Borel distribution series related to analytic function and some basic properties of geometric function theory. Accordingly, some results to coefficient estimates, growth and distortion theorem, radii of starlikeness, convexity, close-to-convexity and convolution properties have also been considered, inviting future research for this field of study. We hope that this distribution series play a significant role in several branches of mathematics, science and technology.

Acknowledgments

The authors are very much grateful to the anonymous reviewers for suggesting some key points.

References

- H. Silverman, Univalent functions with negative coefficients, Proc. Am. Math. Soc. 51, 109–116 (1975).
- G. M. Mittag-Leffler, Sur la nouvelle mapping e_a(x), C. R. Acad. Sci. Paris. 137, 554–558 (1903).
- 3. A. Wiman, Uber die Nullstellen der Funktionen $E_{\alpha}(x)$, Acta Math. **29**(1), 217–234 (1905).
- N. Alessa, B. Venkateswarlu, P. Thirupathi Reddy, K. Loganathan, and K. Tamilvanan, A new subclass of analytic functions related to Mittag-Leffler type Poisson distribution series, *J. Funct. Spaces.* Art. ID 6618163, 7 (2021).
- A. A. Attiya, Some applications of Mittag-Leffler function in the unit disk, *Filomat.* **30**(7), 2075–2081 (2016).
- D. Bansal and J. K. Prajapat, Certain geometric properties of the Mittag-Leffler functions, *Complex Var. Elliptic Equations* 61(3), 338– 350 (2016).
- B. A. Frasin, An application of an operator associated with generalized Mittag-Leffler function, *Konuralp J. Math.* 7(1), 199–202 (2019).
- B. A. Frasin, T. Al-Hawary, and F. Yousef, Some properties of a linear operator involving generalized Mittag-Leffler function, *Stud. Univ. Babeş-Bolyai Math.* 65(1), 67–75 (2020).
- M. Garg, P. Manohar, and S. L. Kalla, A Mittag-Leffler-type function of two variables, *Integral Transforms Spec. Funct.* 24(11), 934–944 (2013).
- K. A. Selvakumaran, H. A. Al-Kharsani, D. Baleanu, S. D. Purohit, and K. S. Nisar, Inclusion relationships for some subclasses of analytic functions associated with generalized Bessel functions, *J. Comput. Anal. Appl.* 24(1), 81–90 (2018).
- S. M. El-Deeb, T. Bulboacă, and J. Dziok, Pascal distribution series connected with certain subclasses of univalent functions, *Kyungpook Math. J.* 59(2), 301–314 (2019).
- W. Nazeer, Q. Mehmood, S. M. Kang, and A. Ul Haq, An application of binomial distribution series on certain analytic functions, *J. Comput. Anal. Appl.* 26(1), 11–17 (2019).
- A. K. Wanas and J. A. Khuttar, Applications of Borel distribution series on analytic mappings, *Earthline J. Math. Sci.* 4, 71–82 (2020).
- B. Khan, H. M. Srivastava, S. Arjika, S. Khan, N. Khan, and Q. Z. Ahmad, A certain q-Ruscheweyh type derivative operator and its applications involving multivalent functions, *Adv. Difference Equations* pp. Paper No. 279, 14 (2021).
- W. Yin, W. Yang, and H. Liu, A neural network scheme for recovering scattering obstacles with limited phaseless far-field data, *J. Comput. Phys.* 417, 109594, 18 (2020).

- M. G. Khan, B. Ahmad, N. Khan, W. K. Mashwani, S. Arjika, B. Khan, and R. Chinram, Applications of Mittag-Leffer type Poisson distribution to a subclass of analytic functions involving conic-type regions, *J. Funct. Spaces.* Art. ID 4343163, 9 (2021).
- H. Silverman, A survey with open problems on univalent functions whose coefficients are negative, *Rocky Mountain J. Math.* 21(3), 1099– 1125 (1991).
- H. Silverman, Integral means for univalent functions with negative coefficients, Houston J. Math. 23(1), 169–174 (1997).
- J. E. Littlewood, On inequalities in the theory of functions, *Proc. Lond. Math. Soc.* (2). 23(7), 481–519 (1925).
- S. M. Popade, R. N. Ingle, and P. T. Reddy, On a certain subclass of analytic functions defined by a differential operator, *Malaya J. Mat.* 8(2), 576–580 (2020).
- S. Porwal and M. Kumar, A unified study on starlike and convex functions associated with Poisson distribution series, *Afr. Mat.* 27(5-6), 1021–1027 (2016).

This page intentionally left blank

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0017

Chapter 17

Hybrid Accelerated Conjugate Gradient Method for Solving Nonlinear System of Equations

Abubakar Sani Halilu^{*,†,‡,**}, Arunava Majumder^{§,††}, Mohammed Yusuf Waziri^{¶,‡‡}, and Idris Ahmed^{||,§§} [†]Department of Mathematics, Sule Lamido University, Kafin Hausa, Nigeria [‡]Department of Mathematics, Lovely Professional University, Phagwara-144411, India [§]Department of Mathematics, Lovely Professional University, Phagwara-144411, India [¶]Department of Mathematical Sciences, Bayero University, Kano, Nigeria Department of Mathematics, Sule Lamido University, Kafin Hausa, Nigeria **abubakars.halilu@slu.edu.ng ^{††}am.arunavamajumder@amail.com ^{‡‡}muwaziri@buk.edu.na §§idrisahamedaml1988@amail.com

Conjugate gradient methods have proved to be the most appropriate iterative techniques for managing large-scale nonlinear equation systems due to their low storage requirement and global convergence properties.

^{*}Corresponding author.

This chapter presents a hybrid method by incorporating the approach proposed by Halilu and Waziri¹⁴ with the conjugate gradient method to obtain a new conjugate gradient parameter. Under suitable conditions, the global convergence of our method is achieved through a derivative-free line search. The numerical results shown in this paper illustrate the effectiveness of the proposed method.

1. Introduction

The typical nonlinear system of equations is defined by

$$\Phi(z) = 0,\tag{1}$$

where $\Phi : \mathbb{R}^n \to \mathbb{R}^n$ is a nonlinear map. Throughout this article, the space \mathbb{R}^n denotes the *n*-dimensional real space and $\Phi_k = \Phi(z_k)$.

Problem (1) can be applied in control theory and compressive sensing problems.^{1–3} Some iterative methods used to solve these problems include Newton and quasi-Newton methods,^{4–7} Gauss–Newton methods,^{8,9} Levenberg–Marquardt methods^{10–12} and derivative-free methods.^{13–16}

Moreover, problem (1) is analogous to the following problem of unconstrained optimization⁸:

$$\min \phi(z), \quad z \in \mathbb{R}^n. \tag{2}$$

Let ϕ be a merit function defined as

$$\phi(z) = \frac{1}{2} \|\Phi(z)\|^2, \tag{3}$$

where $\phi : \mathbb{R}^n \to \mathbb{R}$ and $\| \cdot \|$ is the Euclidean norm.

The Newton method, which can obtain its search direction based on solving the following linear equation, is the most common scheme for solving (1):

$$\Phi_k + \Phi'_k d_k = 0, \tag{4}$$

where Φ'_k is the Jacobian matrix of Φ_k at x_k or its approximation and d_k is the search direction. Although the Newton method can easily be implemented with quadratic rate of convergence,⁴ it is however not applicable for solving large-scale problem because it requires the Jacobian matrix computation at each iteration.^{4,5}

Due to these shortcomings of the Newton method, conjugate gradient (CG) approaches are appropriate.^{3,16–18,20} The methods are very promising for solving the large-scale problems due to their simplicity and low storage requirements. A typical CG method is implemented using the iterative procedure

$$z_{k+1} = z_k + s_k, \quad s_k = \alpha_k d_k, \quad k = 0, 1, \dots,$$
 (5)

where α_k , $s_k = x_{k+1} - x_k$ is a step length and d_k is calculated by

$$d_k = \begin{cases} -\Phi_k, & \text{if } k = 0, \\ -\Phi_k + \beta_k d_{k-1} & \text{if } k \ge 1, \end{cases}$$
(6)

where β_k is a called CG parameter. The motivation after any CG method is to have a helpful CG parameter. As a result, each CG approach corresponds to a specific CG parameter. Among the well-known CG parameters are

$$\beta_k^{FR} = \frac{\|\Phi_k\|^2}{\|\Phi_{k-1}\|^2}, \quad \beta_k^{PRP} = \frac{\Phi_k^T y_{k-1}}{\|\Phi_{k-1}\|^2},$$
$$\beta_k^{HS} = \frac{\Phi_k^T y_{k-1}}{d_{k-1}^T y_{k-1}}, \quad \text{and} \quad \beta_k^{DY} = \frac{\|\Phi_k\|^2}{d_{k-1}^T y_{k-1}},$$

which are called Fletcher–Reeves (FR),¹⁷ Polak–Ribire–Polyak (PRP),¹⁸ Hestenes–Stiefel (HS),¹⁹ and Dai–Yuan (DY),²⁰ respectively. Note that $y_{k-1} = \Phi_k - \Phi_{k-1}$.

Numerous scientists have built various formulas for the CG parameters to make it easier to apply these methods to multiple areas that produce global convergence properties with good numerical outcomes. As a result, hybrid methods have been developed, which combine various approaches to present a practical approach with improved numerical performance and convergence properties.

In recent decades, adaptive modifications of the classical CG methods discussed above have been incorporated into the solution of problem (1). Cheng in Ref. 21 demonstrated the global convergence of a nonlinear monotone PRP-type algorithm by employing the approach of PRP for unconstrained optimization (2). In addition, Ming *et al.*²² introduced a family of CG methods in Ref. 23 for finding (2). The numerical results demonstrated the efficacy of the

technique in Ref. 22 when compared to the existing method. Motivated by Ref. 23, Waziri *et al.*²⁴ introduced a method that is globally converged. Subsequently, a Hager–Zhang family of CG methods for monotone nonlinear equations has been presented by Waziri *et al.*,⁹ who were motivated by the global optimization problems in Ref. 25 and the hyperplane approach in Ref. 26.

Hybrid methods and parametric families have been introduced to address the shortcomings of some existing methods. As the name implies, they are based on combining different techniques to present a better approach with improved numerical performance and convergence properties. Ahmad and Storey²⁷ proposed the following hybrid approach for unconstrained optimization problems:

$$\beta_k = \begin{cases} \beta_k^{PRP} & \text{if } 0 \le \beta_k^{PRP} \le \beta_k^{FR}, \\ \beta_k^{FR} & \text{otherwise.} \end{cases}$$
(7)

When an iteration jam occurs again, the PRP update parameter is used. Consequently, Hu and Storey²⁸ proposed that β_k be defined as

$$\beta_k = \max\{0, \min\{\beta_k^{PRP}, \beta_k^{FR}\}\}.$$
(8)

According to Nocedal and Gilbert,²⁹ β_k^{PRP} can be negative. Consequently, they proposed the following CG parameter:

$$\beta_k = \max\{-\beta_k^{FR}, \min\{\beta_k^{PRP}, \beta_k^{FR}\}\}$$
(9)

in an attempt to broaden the available options for the PRP update parameter. In Ref. 30, Dai and Yuan used the DY parameter in conjunction with HS parameter to present the following CG parameter:

$$\beta_k = \max\{0, \min\{\beta_k^{DY}, \beta_k^{HS}\}\}$$
(10)

and

$$\beta_k = \max\{-c\beta_k^{DY}, \min\{\beta_k^{DY}, \beta_k^{HS}\}\}, \quad c = \frac{1-\sigma}{1+\sigma} > 0,$$
(11)

where $0 < \sigma < 1$.

Dai and Ni investigated various CG methods for large-scale unconstrained optimization problems in Ref. 31 and found that the hybrid CG approach (10) produced the best results. By combining the LSand DY parameters as a convex combination, Liu and Li introduced a new CG parameter³² in the following way:

$$\beta_k = (1 - \theta_k)\beta_k^{LS} + \theta_k \beta_k^{DY}, \qquad (12)$$

where $\theta_k \in [0, 1]$ is the hybridization parameter. It can be seen that when $\theta_k = 0$, $\beta_k = \beta_k^{LS}$ and when $\theta_k = 1$, $\beta_k = \beta_k^{DY}$.

Another type of hybrid conjugate gradient methods is developed by combining the gradient methods with the Picard–Mann hybrid iterative process³³ defined by

$$\begin{cases} x_1 = x \in \Omega, \\ q_k = (1 - \varrho_k) x_k + \varrho_k T x_k, \\ x_{k+1} = T q_k, \quad k \in \mathbb{N}, \end{cases}$$
(13)

where $\{\varrho_k\}$ is in (0,1). Let Ω be a nonempty convex subset of a normed space **E**, and let $T : \Omega \longrightarrow \Omega$ be a mapping. In Ref. 34, Petrovic *et al.* combined the gradient descent (SM) method³⁵ with the iterative process in Ref. 33. Numerical results reveal that the hybrid method³⁴ converges faster than SM method in Ref. 35. Furthermore, Petrovic *et al.*³⁶ presented a modified hybrid method to improve the numerical version of the double direction method in Ref. 37. As a result, Halilu *et al.*³⁸ combined the hybrid process in Ref. 33 with the method presented in Ref. 14 and proposed the CG method for solving convex constrained nonlinear monotone equations. Recently, a hybrid double direction method for solving nonlinear monotone equations with convex constraint has been introduced by Halilu *et al.*³⁹ via the hybrid iterative process presented in Ref. 33.

The above literature clearly shows that the hybridization process enhances the effectiveness of some current methods. The appealing properties of hybridized approaches and the scarcity of hybridization methods for solving (1) in the literature motivated this study.

Following is how this chapter is organized. The proposed method's algorithm is presented in Section 2. Section 3 demonstrates the proposed algorithm's convergence analysis. Section 4 contains a report on some numerical experiments. Section 5 concludes the chapter.

2. Hybrid Accelerated Conjugate Gradient Algorithm

This section contains details of the proposed method. Let us begin with the technique proposed in Ref. 14. The method approximated the Jacobian via

$$\Phi'_k \approx \gamma_k I,\tag{14}$$

where $\gamma_k > 0$ is a nonnegative parameter and I is an identity matrix. The technique in Ref. 14 develops a sequence of iterates $\{z_k\}$ with $z_{k+1} = z_k + (\alpha_k + \alpha_k^2 \gamma_k) d_k$ and the direction d_k is given as

$$d_k = -\gamma_k^{-1} \Phi_k. \tag{15}$$

The parameter $\gamma_k > 0$ is obtained as

$$\gamma_{k+1} = \frac{y_k^T y_k}{y_k^T s_k},\tag{16}$$

where $y_k = \Phi_{k+1} - \Phi_k$ and $s_k = (\alpha_k + \alpha_k^2 \gamma_k) d_k$. Even though the convergence properties of the method¹⁴ are strong, its numerical performance is defined as weaker. Consequently, we are inspired to enhance its numerical convergence by introducing a hybrid method with efficacious numerical results. This is made possible by combining (6) and (15) as follows:

$$\gamma_k^{-1} \Phi_k = (\Phi_k - \beta_k s_{k-1}).$$
(17)

By substituting (16) into (17), we have

$$(y_{k-1}^T s_{k-1})\Phi_k = y_{k-1}^T y_{k-1}(\Phi_k - \beta_k s_{k-1}).$$
 (18)

By multiplying (18) by y_{k-1}^T , it can be written as

$$\beta_k(y_{k-1}^T y_{k-1})(y_{k-1}^T s_{k-1}) = (y_{k-1}^T y_{k-1})(y_{k-1}^T \Phi_k) - (y_{k-1}^T \Phi_k)(y_{k-1}^T s_{k-1}).$$
(19)

Now, we write the CG parameter as follows:

$$\beta_k = \frac{(y_{k-1} - s_{k-1})^T y_{k-1} (y_{k-1}^T \Phi_k)}{(y_{k-1}^T y_{k-1}) (y_{k-1}^T s_{k-1})},$$
(20)

and the search direction is given by

$$d_{k} = \begin{cases} -\Phi_{k} & \text{if } k = 0, \\ -\Phi_{k} + \frac{(y_{k-1} - s_{k-1})^{T} y_{k-1} (y_{k-1}^{T} \Phi_{k})}{(y_{k-1}^{T} y_{k-1}) (y_{k-1}^{T} s_{k-1})} s_{k-1} & \text{if } k \ge 1. \end{cases}$$

$$(21)$$

To achieve a better direction toward the solution, we proposed the following CG search direction:

$$d_k = -\Phi + \beta_k^{HACG} s_{k-1}, \tag{22}$$

with

$$\beta_k = \frac{(y_{k-1} - s_{k-1})^T y_{k-1}(y_{k-1}^T \Phi_k)}{\psi_k},\tag{23}$$

where

$$\psi_k = \max\{(y_{k-1}^T y_{k-1})(y_{k-1}^T s_{k-1}), \rho \| s_{k-1} \|^3\}.$$

The following line search proposed in Ref. 8 can be used to compute the step length α_k .

Let $\Omega_1 > 0$, $\Omega_2 > 0$ and $r \in (0, 1)$ be a constant, and let $\{\sigma_k\}$ be a positive sequence such that

$$\sum_{k=0}^{\infty} \sigma_k < \sigma < \infty, \tag{24}$$

with

$$\phi(z_k + \alpha_k d_k) - \phi(z_k) \le -\Omega_1 \|\alpha_k \Phi_k\|^2 - \Omega_2 \|\alpha_k d_k\|^2 + \sigma_k \phi(z_k).$$
(25)

Let a_k be the smallest nonnegative integer a such that (25) holds for $\alpha = r^a$. Let $\alpha_k = r^{a_k}$.

Remark 1. It is clear that the line search (25) is well defined. Otherwise, for any integer i > 0,

$$\phi(z_k + r^a d_k) - \phi(z_k) > -\Omega_1 \|r^a \Phi_k\|^2 - \Omega_2 \|r^a d_k\|^2 + \sigma_k \phi(z_k).$$

Let $a \to \infty$, then $0 \ge \sigma_k f(z_k)$. This leads to a contradiction since $\sigma_k \phi(z_k)$ is positive.

Algorithm 1 (HACG). Step 1: Given x_0 , $d_0 = -\Phi_0$, $\epsilon = 10^{-4}$, set k = 0. Step 1: Compute Φ_k . Step 2: If $||\Phi_k|| \le \epsilon$, then stop, else go to Step 2. Step 3: Compute α_k (using (25)). Step 4: Set $z_{k+1} = z_k + \alpha_k d_k$. Step 5: Compute d_{k+1} using (21). Step 6: Set k = k + 1 and go to Step 2.

3. Convergence Analysis

In this section, the convergence result of Algorithm 1 (HACG) is presented. Now, let us defined the level set

$$\S = \{ z | \| \Phi(z) \| \le \| \Phi_0 \| \}.$$
(26)

Assumption 1. To analyze the convergence properties of Algorithm 1, the following assumptions are required:

- Assumption 1: There exists $z^* \in \mathbb{R}^n$ such that $\Phi(z^*) = 0$.
- Assumption 2: Φ is continuously differentiable in some neighborhood, say N of z^* containing §.
- Assumption 3: The Jacobian of Φ is bounded and positive definite on Q, i.e. there exists a positive constants H > h > 0 such that

$$\|\Phi'(z)\| \le H \quad \forall z \in Q \tag{27}$$

and

$$h||d||^2 \le d^T \Phi'(z) d \quad \forall z \in Q, d \in \mathbb{R}^n.$$
(28)

Remark 2. Assumption 1 implies that there exists a constant H > h > 0 such that

$$h\|d\| \le \|\Phi'(z)d\| \le H\|d\| \quad \forall z \in Q, d \in \mathbb{R}^n,$$
(29)

$$h\|z - y\| \le \|\Phi(z) - \Phi(y)\| \le H\|z - y\| \quad \forall y, z \in N.$$
 (30)

Lemma 1. Suppose that Assumptions 1–3 hold and Algorithm 1 generates $\{z_k\}$. Then there exists a constant h > 0 such that for all k,

$$y_k^T s_k \ge h \|s_k\|^2.$$
(31)

Proof. By mean-value theorem, we have $y_k^T s_k = s_k^T (\Phi_{k+1} - \Phi_k) = s_k^T \Phi'(\xi) s_k \ge h ||s_k||^2$, where $\xi = z_k + \zeta(z_{k+1} - z_k), \zeta \in (0, 1)$; the last inequality follows from (28).

From Lemma 1 and (30), we have

$$\frac{y_k^T s_k}{\|s_k\|} \ge h, \qquad \frac{\|y_k\|^2}{y_k^T s_k} \le \frac{H^2}{h}.$$
(32)

Lemma 2. Suppose that Assumptions 1–3 hold and Algorithm 1 generates $\{z_k\}$. Then we have

$$\lim_{k \to \infty} \|\alpha_k d_k\| = \lim_{k \to \infty} \|s_k\| = 0$$
(33)

and

$$\lim_{k \to \infty} \|\alpha_k \Phi_k\| = 0.$$
(34)

Proof. By (25), we have, for all k > 0,

$$\Omega_2 \|\alpha_k d_k\|^2 \le \Omega_1 \|\alpha_k \Phi_k\|^2 + \Omega_2 \|\alpha_k d_k\|^2 \le \|\Phi_k\|^2 - \|\Phi_{k+1}\|^2 + \sigma_k \|\Phi_k\|^2.$$
(35)

By summing the above inequality, we have

$$\Omega_{2} \sum_{i=0}^{k} \|\alpha_{i}d_{i}\|^{2} \leq \sum_{i=0}^{k} \left(\|\Phi_{i}\|^{2} - \|\Phi_{i+1}\|^{2}\right) + \sum_{i=0}^{k} \sigma_{i} \|\Phi_{i}\|^{2}$$

$$= \|\Phi_{0}\|^{2} - \|\Phi_{k+1}\|^{2} + \sum_{i=0}^{k} \sigma_{i} \|\Phi_{i}\|^{2}$$

$$\leq \|\Phi_{0}\|^{2} + \|\Phi_{0}\|^{2} \sum_{i=0}^{k} \sigma_{i}$$

$$\leq \|\Phi_{0}\|^{2} + \|\Phi_{0}\|^{2} \sum_{i=0}^{\infty} \sigma_{i}.$$
(36)

From (26) and (24), the series $\sum_{i=0}^{\infty} \|\alpha_i d_i\|^2$ is convergent. This implies (33). Using similar arguments as above, (34) holds. \Box

Lemma 3. Suppose that Assumptions 1–3 hold and Algorithm 1 generates $\{z_k\}$. Then there exist a constant $m_1 > 0$ such that for all k > 0,

$$\|d_k\| \le m_1. \tag{37}$$

Proof. From the level set, (23) and (30), we have

$$\begin{aligned} |\beta_{k}^{HACG}| &= \left| \frac{(y_{k-1} - s_{k-1})^{T} y_{k-1}(y_{k-1}^{T} \Phi_{k})}{\psi_{k}} \right| \\ &= \left| \frac{\|y_{k-1}\|^{2} y_{k-1}^{T} \Phi_{k} - (y_{k-1}^{T} s_{k-1})(y_{k-1}^{T} \Phi_{k})}{\psi_{k}} \right| \\ &\leq \frac{\|\|y_{k-1}\|^{2} y_{k-1}^{T} \Phi_{k} + (y_{k-1}^{T} s_{k-1})(y_{k-1}^{T} \Phi_{k})\|}{\psi_{k}} \\ &\leq \frac{\|\|y_{k-1}\|^{2} y_{k-1}^{T} \Phi_{k} + (y_{k-1}^{T} s_{k-1})(y_{k-1}^{T} \Phi_{k})\|}{\rho \|s_{k-1}\|^{3}} \\ &\leq \frac{\|y_{k-1}\|^{3} \|\Phi_{k}\|}{\rho \|s_{k-1}\|^{3}} + \frac{\|y_{k-1}\|^{2} \|s_{k-1}\| \|\Phi_{k}\|}{\rho \|s_{k-1}\|^{3}} \\ &\leq \frac{H^{3} \|s_{k-1}\|^{3} \|\Phi_{k}\|}{\rho \|s_{k-1}\|^{3}} + \frac{H^{2} \|s_{k-1}\|^{3} \Phi_{k}\|}{\rho \|s_{k-1}\|^{3}} \\ &\leq \frac{H^{3} \Phi_{0}\|}{\rho} + \frac{H^{2} \Phi_{0}\|}{\rho} \\ &= \frac{(H+1)H^{2} \|\Phi_{0}\|}{\rho}. \end{aligned}$$

Taking $\lambda = \frac{(H+1)H^2 \|\Phi_0\|}{\rho}$, we have

$$|\beta_k^{HACG}| \le \lambda. \tag{39}$$

Therefore, from (22) and (39), we have

$$\begin{aligned} \|d_k\| &= \| -\Phi_k + \beta_k^{HACG} s_{k-1} \| \\ &\leq \|\Phi_k\| ||\beta_k^{HACG}| \|s_{k-1}\| \\ &\leq \|\Phi_k\| \lambda \|s_{k-1}\| \\ &\leq \|\Phi_0\| \lambda P, \end{aligned}$$

$$\tag{40}$$

Since $\lim_{k \to \infty} ||s_k|| = 0$, then there exists a positive constant P such that $||s_{k-1}|| < P$. Taking $m_1 = ||\Phi_0||\lambda P$, we have (37). We can deduce that for all k, (37) holds.

Lemma 4. Suppose that Assumptions 1–3 hold and Algorithm 1 generates $\{z_k\}$. Then $\{z_k\} \subset \S$.

Proof. From the norm decent line search in (25), we have $\|\Phi_{k+1}\| \leq \|\Phi_k\|$. Moreover, we have, for all k,

$$\|\Phi_{k+1}\| \le \|\Phi_k\| \le \|\Phi_{k-1}\| \le \dots \le \|\Phi_0\|$$

This implies that $\{z_k\} \subset \S$.

Theorem 1. Suppose that Assumptions 1–3 hold and Algorithm 1 generates $\{z_k\}$. Assume further for all k > 0,

$$\alpha_k \ge c \frac{|\Phi_k d_k|}{\|d_k\|^2},\tag{41}$$

where c > 0. Then

$$\lim_{k \to \infty} \|\Phi_k\| = 0. \tag{42}$$

Proof. From (37), (33) and the boundedness of $\{||d_k||\}$, we have

$$\lim_{k \to \infty} \alpha_k \|d_k\|^2 = 0.$$
(43)

From (41) and (43), it follows that

$$\lim_{k \to \infty} |\Phi_k^T d_k| = 0.$$
(44)

On the other hand, (22) and (39) lead to

$$\begin{split} \|\Phi_k\|^2 &= -\Phi_k^T d_k + \beta_k^{HACG} \|\Phi_k\| \|s_{k-1}\|, \\ &\leq |\Phi_k^T d_k| + |\beta_k^{HACG}| \|\Phi_k\| \|s_{k-1}\|, \\ &\leq |\Phi_k^T d_k| + \lambda \|\Phi_0\| \|s_{k-1}\|. \end{split}$$

Thus,

$$0 \le \|\Phi_k\|^2 \le |\Phi_k^T d_k| + \lambda \|\Phi_0\| \|s_{k-1}\| \longrightarrow 0.$$
(45)

Therefore,

$$\lim_{k \to \infty} \|\Phi_k\| = 0. \tag{46}$$

The proof is completed.

4. Numerical Results

This section compares the proposed method (HACG) to the following existing methods to demonstrate its efficiency and robustness:

- (1) IDFDD algorithm proposed in Ref. 14,
- (2) ADLCG algorithm proposed in Ref. 24.

All the three algorithms have been implemented with the same line search technique and the following parameters are set: $\Omega_1 = \Omega_2 = 10^{-4}$, r = 0.2 and $\sigma_k = \frac{1}{(k+1)^2}$. The computational codes used in this study were written in Matlab 9.4.0 (R2018a) and operated on a computer with a 1.80 GHz CPU processor and 4 GB RAM. The program is terminated when the total number of iterations overreaches 1000 or when $\|\Phi_k\| \leq 10^{-4}$. We employ the symbol "–" to represent a failure due to the following conditions:

- (i) memory requirements;
- (ii) when the iterations reach 1000; however, no z_k satisfies the stopping criterion.

Using some Benchmark test problems with various initial points, the methods were tested. Problems 1 and 2 are from Ref. 9 and Problems 3 is from Ref. 14. We experimented with the three methods with the dimension (n values), 1000, 10,000, and 100,000 with different initial points:

$$z1 = \left(1, \frac{1}{2}, \frac{1}{3}, \dots, \frac{1}{n}\right)^{T},$$

$$z2 = \left(1 - 1, 1 - \frac{1}{2}, 1 - \frac{1}{3}, \dots, 1 - \frac{1}{n}\right)^{T},$$

$$z3 = (2, 4, 6, \dots, 2n)^{T},$$

$$z4 = (1, 3, 5, \dots, 2n - 1)^T,$$

 $z5 = (0.5, 0.5, \dots, 0.5)^T.$

Problem 1:

$$\Phi_i(z) = 2\left(n + i(1 - \cos z_i) - \sin z_i - \sum_{j=1}^n \cos z_j\right) (2\sin z_i - \cos z_i)$$
$$i = 1, 2, 3, \dots, n.$$

Problem 2:

$$\Phi_i(z) = z_i - \left(1 - \frac{c}{2n} \sum_{j=1}^n \frac{\mu_i z_j}{\mu_i + \mu_j}\right)^{-1}$$

$$i = 1, 2, 3, \dots, n, \quad j = 1, 2, 3, \dots, n,$$

with $c \in [0, 1)$ and $\mu = \frac{i-0.5}{n}$ (we take c = 0.1 in this experiment). **Problem 3:**

$$\Phi(z) = \begin{pmatrix} 2 & -1 \\ 0 & 2 & -1 \\ & \ddots & \ddots & \ddots \\ & & \ddots & \ddots & -1 \\ & & & -1 & 2 \end{pmatrix} z + (\sin z_1 - 1, \sin z_2 - 1, \dots, \sin z_n - 1)^T.$$

Table 1. Numerical results for Problem 2

			HA	ACG		AD	\mathbf{LCG}		IDF	DD
Dimension	Guess	ITRN	TM	$\ \Phi_k\ $	ITRN	TM	$\ \Phi_k\ $	ITRN	TM	$\ \Phi_k\ $
1000	z1	6	0.00146	2.30E-05	8	0.00176	2.82E-05	95	0.01652	8.94E-05
	z2	5	0.00138	7.03E-07				49	0.01145	8.82E-05
	z3	8	0.00241	7.04E-07	9	0.00226	4.30E-05	88	0.01505	8.94E-05
	z4	6	0.00133	1.71E-06				17	0.00403	7.38E-05
10,000	z1	4	0.00153	6.30E-11				37	0.00906	9.71E-05
	z2	5	0.00123	1.73E-06	5	0.00131	6.11E-05	36	0.00492	9.63E-05
	z3	6	0.00164	1.81E-06	9	0.00159	6.51 E-05	20	4.37E-03	9.52E-05
	z4	6	0.00147	2.47E-06	8	0.00196	2.83E-05	39	0.00584	8.86E-05
100,000	z1	4	0.00137	2.90E-05	4	0.00174	2.49E-05	25	0.00639	7.36E-05
	z2	3	0.00104	7.72E-08				25	0.00403	8.29E-05
	z3	7	0.00155	3.34E-07	11	0.00274	3.26E-05	38	0.00924	8.74E-05
	z4	5	0.00174	4.81E-06	6	0.00185	3.15 E-05	24	0.00451	7.80E-05

			HA	HACG ADLCG		LCG	IDFDD		DD	
Dimension	Guess	ITRN	ТМ	$\ \Phi_k\ $	ITRN	ТМ	$\ \Phi_k\ $	ITRN	TM	$\ \Phi_k\ $
1000	z1	5	0.01489	2.56E-06	26	0.04494	6.62E-05	31	0.05437	8.90E-05
	z2	4	0.01312	5.68E-05	29	0.06025	9.51E-05	45	0.08011	9.58E-05
	z3	5	0.01882	6.38E-07	32	0.04791	8.32E-05	67	0.11789	7.21E-05
	z4	4	0.01257	1.52E-06	32	0.04575	8.28E-05	67	0.11571	7.17E-05
10,000	z1	5	0.01592	9.76E-05	25	0.03704	8.80E-05	31	0.05514	9.48E-05
	z2	4	0.01512	1.59E-04	31	0.04863	9.48E-05	53	7.99E-02	6.89E-05
	z3	5	0.01331	2.70E-05	32	0.05302	6.54E-05	81	0.14985	7.41E-05
	z4	5	0.01312	9.52E-06	38	0.0515	6.53E-05	81	0.13035	7.41E-05
100,000	z1	4	0.01638	1.02E-04	26	0.03682	7.46E-05	32	0.0603	8.72E-05
	z2	5	0.01756	1.81E-04	33	0.04575	7.71E-05	60	0.09579	7.04E-05
	z3	5	0.01299	2.87E-05	41	0.05545	9.57E-05	95	0.14284	7.60E-05
	z4	5	0.01167	4.09E-07	41	0.06811	9.57 E- 05	95	0.14711	7.60E-05

Table 2. Numerical results for Problem 2

Table 3. Numerical results for Problem 3

			HA	ACG		AD	\mathbf{LCG}		IDI	FDD
Dimension	Guess	ITRN	ТМ	$\ \Phi_k\ $	ITRN	TM	$\ \Phi_k\ $	ITRN	TM	$\ \Phi_k\ $
1000	z1	8	0.01852	1.27E-05	11	0.0029	5.01E-05	27	0.01349	9.24E-05
	z2	18	0.00692	3.78E-05	20	0.00478	5.25E-05			
	z3	33	0.00854	5.68 E-07	32	0.00843	3.95E-05			
	z4	33	0.00767	5.59E-07	32	0.00753	3.98E-05			
	z5	23	0.01325	5.23E-05	21	0.03219	2.57 E-05			
10,000	z1	9	0.00251	6.33E-07	11	0.00293	6.19E-05	30	0.01161	7.94E-05
	z2	23	0.02048	1.13E-05	23	0.0056	9.75 E-05			
	z3	43	0.00591	1.63E-07	41	0.06081	5.25E-05			
	z4	43	0.00815	1.62E-07	41	0.00892	3.72E-05			
	z5	28	0.09247	7.64 E-05	25	0.09236	9.00E-05			
100,000	z1	9	0.00246	1.38E-05	11	0.00317	3.19E-05	22	0.00694	7.85E-05
	z2	28	0.00502	2.72E-06	29	0.00699	3.10E-05			
	z3	61	0.00958	4.76E-07	56	0.01182	5.99E-05			
	z4	61	0.03234	4.76E-07	59	0.0136	9.18E-05			
	z5	32	0.80454	6.91 E- 05	28	0.76903	8.54E-01			

Table 4. Overview of the results recorded in the Tables 1–3

Methods	ITRN	Percentage	\mathbf{TM}	Percentage
HACG	27	69.23	29	74.36
ADLCG	9	23.08	10	25.64
IDFDD	0	0	0	0
Undecided	3	7.69	0	0

The implemented numerical results of the HACG, ADLCG, and IDFDD methods are reported in Tables 1–3, where the following can be noted:

- "Guess" denotes the initial point,
- "ITRN" denotes the total number of iterations,
- "TM' denotes the CPU time (seconds),
- " $\|\Phi_k\|$ " is the norm of residual at the termination point.

Although all three methods solved the numerical experiment's test problems, the HACG method converges to the solution of (1) more quickly because it has fewer iterations than the ADLCG and IDFDD methods, except for Problem 3, where the ADLCG beats the HACG in some iterations. Moreover, the ADLCG and IDFDD methods failed during the iteration process by clear indication from Tables 1 and 3. This shows that the proposed method is very effective solving problem (1). Moreover, as shown in Tables 1 and 3, the HACG method solved the three problems with less CPU time. Our method beats the ADLCG and IDFDD methods tested for nearly all problems because it has the fewest iterations and processor time of the methods compared.

To illustrate which approach is the best, Tables 4 recapitulates the results recorded in Tables 1–3. From Table 4, the HACG approach solved 69.23% (27 out of 39) of iterations. At the same time, the ADLCG and IDFDD methods have solved 23.08% (9 out of 39) and 0% (0 out of 39), respectively. The summary table indicates that seven problems have the same number of iterations in the HACG and ADLCG methods, which is 7.69% and is undecided. For the processor time, the methods ADLCG and IDFDD are respectively resolved 25.64% (10 out of 39) and 0 percent (0 out of 39), and the HACG method solved 74.36% (29 out of 39) of the problems with less CPU time.

Dolan and Moré⁴⁰ evaluation tool displayed figures of each of the three methods used in the experimentations to analyze the results described in Tables 1–3. Figures 1 and 2 depict the performance profiles of the HACG, ADLCG, and IDFDD methods in terms of iteration numbers and CPU time, using data from Tables 1–3 and the strategy proposed in Ref. 40. A fraction $p(\tau)$ of the problems is plotted from two graphs where a method is within τ of the best time. The top curve in both figures corresponds to the HACG scheme.



Fig. 1. Performance profile for the function evaluation.



Fig. 2. Performance profile for the CPU time (seconds).

As a result, it outperforms the other approaches in terms of minimum iteration numbers and CPU time, hence we deduce that HACG method is effective for solving problem (1).

5. Conclusion

In this work, a hybrid accelerated conjugate gradient method for solving systems of nonlinear equations is presented. The proposed

method (HACG) is a derivative-free, it has no computation and storage of the Jacobian matrix at each iteration, therefore it has low memory requirements and computational cost. In addition, it has improved the convergence of the double direction method.¹⁴ As a result, it was compared with the $ADLCG^{24}$ and IDFDD methods.¹⁴ In addition, the numerical results of the three (3) methods are reported in Tables 1–3. It can be noted that the HACG method is very effective because it has the smallest number of iterations and CPU time compared to the ADLCG and IDFDD methods. Table 4 summarized Tables 1–3, highlighting that the HACG method is a winner in terms of CPU time and the number of iterations. In Figures 1 and 2, the performance profile of Dolan and $Moré^{40}$ was used to interpret the results in Tables 1–3. The figures depict the effectiveness of the HACG method. Therefore, we deduce that the proposed method is promising and appropriate for solving problem (1). Future work includes modifying the proposed method using the inertial-based approach for nonlinear system equations with a motion control application.

Acknowledgments

We thank the reviewers and the editors for their support.

References

- M. Sun, M. Y. Tian, and Y. J. Wang, Multi-step discrete-time Zhang neural networks with application to time-varying nonlinear optimization, *Discrete Dyn. Nat. Soc.* 2019, 1–4 (2019).
- 2. K. Meintjes and A. P. Morgan, A methodology for solving chemical equilibrium systems, *Appl. Math. Comput.* **22**, 333–361 (1987).
- Y. H. Xiao and H. Zhu, A conjugate gradient method to solve convex constrained monotone equations with applications in compressing sensing, J. Math. Anal. Appl. 405, 310–319 (2013).
- J. E. Dennis and R. B. Schnabel, Numerical Methods for Unconstrained Optimization and Non-linear Equations, Prentice Hall, Englewood Cliffs, NJ (1983).
- W. M. Yusuf, W. J. Leong, and M. A. Hassan, Jacobian-free diagonal newton method for solving nonlinear systems with singular Jacobian, *Malay J. Math. Sci.* 5, 241–255 (2011).

- G. Yuan and X. Lu, A new backtracking in exact BFGS method for symmetric nonlinear equations, *Comput. Math. App.* 55, 116–129 (2008).
- Q. R. Yana, X. Z. Penga, and D. H. Li, A globally convergent derivativefree method for solving large-scale nonlinear monotone equations, *J. Comp. App. Math.* 234, 649–657 (2010).
- D. Li and M. Fukushima, A global and superlinear convergent Gauss-Newton based BFGS method for symmetric nonlinear equation, *SIAM J Numer. Anal.* 37, 152–172 (1999).
- M. Y. Waziri, K. Ahmad, and J. Sabiu, A family of Hager-Zhang conjugate gradient methods for system of monotone nonlinear equations, *Appl. Math. and Comput.* 361, 645–60 (2019).
- D. W. Marquardt, An algorithm for least-squares estimation of nonlinear parameters, SIAM J. Appl. Math. 11, 431–441 (1963).
- 11. K. Levenberg, A method for the solution of certain non-linear problems in least squares, *Q. Appl. Math.* **2**, 164–166 (1944).
- C. Kanzow, N. Yamashita, and M. Fukushima, Levenberg-Marquardt methods for constrained nonlinear equations with strong local convergence properties, J. Comput. Appl. Math. 172, 375–397 (2004).
- A. S. Halilu and M. Y. Waziri, A transformed double step length method for solving large-scale systems of nonlinear equations, *J Numer. Math. Stoch.* 9, 20–32 (2017).
- A. S. Halilu and M. Y. Waziri, An improved derivative-free method via double direction approach for solving systems of nonlinear equations, *J Ramanujan Math. Soc.* 33(1), 75–89 (2018).
- A. S. Halilu and M. Y. Waziri, Enhanced matrix-free method via double step length approach for solving systems of nonlinear equations, *Int.* J. App. Math Res. 6, 147–156 (2017).
- H. Abdullahi, A. S. Halilu, and M. Y. Waziri, A modified conjugate gradient method via a double direction approach for solving large-scale symmetric nonlinear systems, J. Numer. Math. Stoch. 10(1), 32–44 (2013).
- R. Fletcher and C. M. Reeves, Function minimization by conjugate gradient methods, J. Comput. 7(1), 149–154 (1964).
- B. T. Polyak, The conjugate gradient method in extreme problems, USSR Comput. Math. Math. Phys. 9, 94–112 (1962).
- M. R. Hestenes and E. Stiefel, Method of conjugate gradient for solving linear systems, J. Res. Nat. Bur. Stand. 49, 409–436 (1952).
- Y. H. Dai and Y. X. Yuan, A nonlinear conjugate gradient method with a strong global convergence property, SIAM. J. Optim. 10(1), 177–182 (1999).

- W. Cheng, A PRP type method for systems of monotone equations, Math. Comput. Model. 50, 15–20 (2009).
- M. Li, H. Liu, and Z. Liu, A new family of conjugate gradient methods for unconstrained optimization, J. Appl. Math. Comput. 58, 219–234 (2018).
- 23. Y. H. Dai and L. Z. Liao, New conjugacy conditions and related nonlinear conjugate gradient methods, *Appl. Math. Optim.* **3**, 87–101 (2001).
- M. Y. Waziri, K. Ahmed, and J. Sabiu, A Dai-Liao conjugate gradient method via modified secant equation for system of nonlinear equations, *Arab. J. Math.* 3, https://doi.org/10.1007/s40065-019-0264-6 (2019).
- W. W. Hager and H. Zhang, A survey of nonlinear conjugate gradient methods, *Pac. J. Optim.* 2(1), 35–58 (2006).
- 26. M. V. Solodov and B. F. Svaiter, A globally convergent inexact newton method for systems of monotone equations. In M. Fukushima and L. Qi (Eds.). *Reformulation: Nonsmooth, Piecewise Smooth, Semismooth* and Smoothing Methods, pp. 355–369, Springer, Kluwer Academic Publishers (1999).
- D. Touati-Ahmed and C. Storey, Efficient hybrid conjugate gradient techniques, J. Opt. Theo. Appl. 64, 379–397 (1990).
- Y. F. Hu and C. Storey, Global convergence result for conjugate gradient methods, J. Opt. Theor. Appl. 71, 399–405 (1991).
- J. C. Gilbert and J. Nocedal, Global convergence properties of conjugate gradient methods for optimization, SIAM J. Opt. 2, 21–42 (1992).
- Y. H. Dai and Y. Yuan, An efficient hybrid conjugate gradient method for unconstrained, Ann. Oper. Res. 103, 33–47 (2001).
- Y. H. Dai and Q. Ni, Testing different conjugate gradient methods for large-scale unconstrained optimization, J. Comput. Math. 21, 311–320 (2003).
- 32. J. K. Li and S. J. Liu, New hybrid conjugate gradient method for unconstrained optimization, *Appl. Math. Comput.* **245**, 36–43 (2014).
- H. K. Safeer, A Picard-Mann hybrid iterative process, *Fixed Point Theor. Appl.* 69, 2–10 (2013).
- M. Petrovic, V. Racocevic, N. Kontrec, S. Panic, and D. Ilic, Hybridization of accelerated gradient descent method, *Numer. Algorithm.* 79, 769–786 (2018).
- M. J. Petrovic and P. S. Stanimirovic, Accelerated gradient descent methods with line search, Numer. Algorithm. 51, 503–520 (2010).
- M. J. Petrovic, P. S. Stanimirovic, N. Kontrec, and J. Mladenovic, Hybrid modification of accelerated double direction method, *Math. Prob. Eng.* 2018, 8 (2018).
- M. J. Petrovic and P. S. Stanimirovic, Accelerated double direction method for solving unconstrained optimization problems, *Mathl. Prob. Eng.* (2014) https://doi.org/10.1155/2014/965104.

- A. S. Halilu, A. Majumder, M. Y. Waziri, and K. Ahmed, Signal recovery with convex constrained nonlinear monotone equations through conjugate gradient hybrid approach, *Math. Comput. Simul.* 187, 520–539 (2021) http://doi.org/10.1016/j.matcom.2021.03.020.
- A. S. Halilu, A. Majumder, M. Y. Waziri, A. M. Awwal, and K. Ahmed, On solving double direction methods for convex constrained monotone nonlinear equations with image restoration, *Comput. Appl. Math.* 40(239), 1–27 (2021) https://doi.org/10.1007/s40314-021-01624-1.
- 40. E. Dolan and J. Moré, Benchmarking optimization software with performance profiles, J. Math. Prog. **91**, 201–213 (2002).

 \odot 2023 World Scientific Publishing Company https://doi.org/10.1142/9789811272608_0018

Chapter 18

A Closed Form of Integral Transforms in terms of Lauricella Function $F_A^{(n)}$ and Their Numerical Simulations

Abdelmajid Belafhal^{\dagger , §}, Halima Benzehoua^{\dagger , ¶}, and Talha Usman^{$*, \ddagger, \parallel$}

[†]LPNAMME, Laser Physics Group, Department of Physics, Faculty of Sciences, Chouaïb Doukkali University, P. B 20, 24000 El Jadida, Morocco [‡]Department of General Requirements, University of Technology and Applied Sciences, Sur-411, Sultanate of Oman [§]belafhal@gmail.com [¶]benzahouahalima1@gmail.com

The main object of this chapter is to investigate integral transforms involving the product of two Bessel functions and the Whittaker function. These integral transforms are given in terms of the Lauricella function $F_A^{(3)}$ of three variables and $F_A^{(4)}$ of four variables. Interesting special cases of our main results are deduced by taking suitable values of the index of the Whittaker function. We also perform some numerical simulations using the Laguerre–Gauss quadrature method and it is found that there is a good agreement between the numerical and theoretical

^{*}Corresponding author.

evaluations. These results can be used in the Fresnel diffraction by a helical axicon of many laser fields, such as Laguerre–Bessel–Gaussian beams.

1. Introduction and Preliminaries

In this chapter, we evaluate integral transforms using special functions, with a focus on Bessel functions. The use of Bessel functions is motivated by their ability to handle complex and abstract situations (see, Ref. 13). These functions have been widely applied in mathematical physics and have been used to solve problems in areas such as mathematics, physics, optimization, cybernetic technology, and biology by reducing the complexity of the problem to the evaluation of integral transforms involving special functions such as Bessel, Whittaker, and Kummer functions (see Refs. 2–5). Earlier, we have published a series of papers in laser physics (see Refs. 6–9) and we have elaborated some applications of these theories in this field. Our results will be useful in a variety of domains of physics, such as plasma, radio and laser physics.

The following definitions are essential for the present investigation: We recall some definitions as the hypergeometric ${}_{p}F_{q}$ given by (see Refs. 10 and 11)

$${}_{p}F_{q}(\alpha_{1},\ldots,\alpha_{p};\beta_{1},\ldots,\beta_{q};z) = \sum_{n=0}^{\infty} \frac{(\alpha_{1})_{n}\ldots(\alpha_{p})_{n}}{(\beta_{1})_{n}\ldots(\beta_{q})_{n}} \frac{z^{n}}{n!}.$$
 (1)

In this definition, we assume that $\beta_j \neq 0, -1, -2, \ldots; j = 1, \ldots, q$.

The shifted factorial (the Pochhammer symbol) is defined in terms of the familiar Gamma function (see Ref. 11)

$$(\lambda)_{\nu} = \frac{\Gamma\left(\lambda + \nu\right)}{\Gamma\left(\lambda\right)}.$$
(2)

We give the definition of Humbert's confluent hypergeometric series of two variables (see Ref. 11)

$$\Psi_2\left[\alpha;\gamma,\gamma';x,y\right] = \sum_{m,n=0}^{\infty} \frac{(\alpha)_{m+n}}{(\gamma)_m(\gamma')_n} \frac{x^m}{m!} \frac{y^n}{n!},\tag{3}$$

with $|x| < \infty$ and $|y| < \infty$, and the series representation for the Bessel function of the first kind of order ν for each $x \in C$ $(-\infty, 0)$

is given by (see Ref. 12)

$$J_{\nu}(x) = \sum_{k=0}^{\infty} \frac{(-1)^{k} \left(\frac{x}{2}\right)^{\nu+2k}}{k! \Gamma(\nu+k+1)}.$$
(4)

The denominator parameters of the above functions are neither zero nor negative integers.

Indeed, recall the triple hypergeometric function ${\cal F}^{(3)}_A$ given by

$$F_A^{(3)}[a, b_1, b_2, b_3; c_1, c_2, c_3; x, y, z] = \sum_{l,m,n=0}^{\infty} \frac{(a)_{l+m+n}(b_1)_l(b_2)_m(b_3)_n}{(c_1)_l(c_2)_m(c_3)_n} \frac{x^l}{l!} \frac{y^m}{m!} \frac{z^n}{n!},$$
(5)

with |x| + |y| + |z| < 1, and the hypergeometric function $F_A^{(4)}$ of four variables defined in the form

$$F_A^{(4)}[a, b_1, b_2, b_3, b_4; c_1, c_2, c_3, c_4; x, y, z, t] = \sum_{j,l,m,n=0}^{\infty} \frac{(a)_{j+l+m+n}(b_1)_j(b_2)_l(b_3)_m(b_4)_n}{(c_1)_j(c_2)_l(c_3)_m(c_4)_n} \frac{x^j}{j!} \frac{y^l}{l!} \frac{z^m}{m!} \frac{t^n}{n!}, \quad (6)$$

with |x| + |y| + |z| + |t| < 1.

The Whittaker function $M_{s,\xi}$ is defined by (see Ref. 11)

$$M_{s,\xi}(z) = z^{\xi + \frac{1}{2}} \exp\left(-\frac{z}{2}\right) {}_{1}F_{1}\left(\frac{1}{2} + \xi - s; 2\xi + 1; z\right).$$
(7)

We also recall some simple special cases of the Whittaker function by following the notation of Buchholz,¹⁰ which are deduced from (7)as follows:

$$M_{\mp r, -r - \frac{1}{2}}(z) = z^{-r} \exp\left(\mp \frac{z}{2}\right),\tag{8}$$

$$M_{0,\frac{1}{2}}(z) = 2\sinh\left(\frac{z}{2}\right),\tag{9}$$

$$M_{a+\frac{1}{4},-\frac{1}{4}}\left(z^{2}\right) = (-1)^{a} \frac{a!}{(2a)!} \exp\left(-\frac{z^{2}}{2}\right) \sqrt{z} H_{2a}\left(z\right), \tag{10}$$

$$M_{a+\frac{3}{4},\frac{1}{4}}\left(z^{2}\right) = \frac{\left(-1\right)^{a}}{2} \frac{a!}{(2a+1)!} \exp\left(-\frac{z^{2}}{2}\right) \sqrt{z} H_{2a+1}\left(z\right), \quad (11)$$

$$M_{\pm\left(b+\frac{1+a}{2}\right)\frac{a}{2}}(z) = \frac{b!}{(a+1)_b} z^{\frac{1+a}{2}} \exp\left(\pm\frac{z}{2}\right) L_b^{(a)}(\mp z),\tag{12}$$

$$M_{0,r}(z) = 2^{2r} \Gamma(r+1) \sqrt{z} I_r\left(\frac{z}{2}\right),$$
(13)

$$M_{r-\frac{1}{2},r}(z) = 2r \exp\left(\frac{z}{2}\right) z^{\frac{1}{2}-r} \gamma(2r,z),$$
(14)

$$M_{-\frac{1}{4},\frac{1}{4}}\left(z^{2}\right) = \frac{\exp\left(z^{2}/2\right)}{2}\sqrt{\pi z} erf(z),$$
(15)

and

$$M_{\pm\left(\frac{1}{2}-r\right)r}\left(z\right) = z^{r+\frac{1}{2}}\exp\left(\pm\frac{z}{2}\right){}_{1}F_{1}\left(2r;2r+1;\mp z\right),\qquad(16)$$

with $L_b^{(a)}$ and H_a as the generalized Laguerre polynomial and the Hermite polynomial, respectively. The functions γ and erf are known as incomplete Gamma and error functions.

2. Main Results

In this section, we evaluate closed forms of two integral transforms involving the product of Bessel functions and Whittaker functions. The used weight in the integrand of our transformation is $e^{-p\rho^2+2q\rho}$ with $\Re(p) > 0$. The considered integrals will be expressed in terms of Lauricella functions $F_A^{(3)}$ of three variables and $F_A^{(4)}$ of four variables.

Theorem 1. Let the conditions $\Re(p) > 0, \Re(\mu) > -1$ and $\frac{|\chi|^2}{16} + |\lambda|^2 + |q|^2 < |p|$ be verified, then the undermentioned transformation holds true:

$$\int_{0}^{\infty} \rho^{\mu} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= R_{0} \left\{ \frac{\Gamma(\alpha_{J})}{2\sqrt{p}} F_{A}^{(3)} \left[\alpha_{J, -, -, -}; \varepsilon + 1, \nu + 1, \frac{1}{2}; x, y, z \right] \right.$$

$$\left. + \frac{q}{p} \Gamma(\beta_{J}) F_{A}^{(3)} \left[\beta_{J, -, -, -}; \varepsilon + 1, \nu + 1, \frac{3}{2}; x, y, z \right] \right\},$$

where

$$R_{0} = \frac{\left(\frac{\lambda}{2\sqrt{p}}\right)^{\varepsilon}}{\left(\sqrt{p}\right)^{\mu}\varepsilon!} \frac{\left(\frac{\chi}{2\sqrt{p}}\right)^{\nu}}{\nu!},$$

$$x = -\frac{\lambda^{2}}{4p}, \quad y = -\frac{\chi^{2}}{4p}, \quad z = \frac{q^{2}}{p},$$
 (17)

$$\alpha_J = \frac{\mu + \nu + \varepsilon + 1}{2}, \quad \beta_J = \frac{\mu + \nu + \varepsilon}{2} + 1, \tag{18}$$

and $F_A^{(3)}$ is the Lauricella series in three variables given by (5).

Proof. Substituting the expansion of Bessel function J_{ε} , we use Lemma 1 (see Appendix) and denoting the left-hand side of (17) by R, we get, after some simplifications, the following expression:

$$R = \frac{\left(\frac{\lambda}{2\sqrt{p}}\right)^{\varepsilon}}{2p^{\frac{\mu+1}{2}}} \frac{\left(\frac{\chi}{2\sqrt{p}}\right)^{\nu}}{\nu!} \left[\sum_{l,m,n=0}^{\infty} A_{l\ m\ n} \frac{x^{l}}{n!} \frac{y^{m}}{m!} \frac{z^{k}}{k!} + \frac{2q}{\sqrt{p}} \sum_{l,m,n=0}^{\infty} B_{l\ m\ n} \frac{x^{l}}{n!} \frac{y^{m}}{m!} \frac{z^{k}}{k!}\right],\tag{19}$$

where

$$A_{l\ m\ n} = \frac{\Gamma\left(\frac{\mu+\nu+\varepsilon+1}{2}+l\right)\left(\frac{\mu+\nu+\varepsilon+1}{2}+l\right)_{m+n}}{\Gamma\left(\varepsilon+l+1\right)\left(\nu+1\right)_{m}\left(\frac{1}{2}\right)_{n}}$$
(20)

and

$$B_{l\ m\ n} = \frac{\Gamma\left(\frac{\mu+\nu+\varepsilon}{2}+1+l\right)\left(\frac{\mu+\nu+\varepsilon}{2}+1+l\right)_{m+n}}{\Gamma\left(\varepsilon+l+1\right)\left(\nu+1\right)_m\left(\frac{3}{2}\right)_n},\tag{21}$$

with x, y and z given by (17).

By using the identity (see Ref. 11),

$$(\lambda + m)_n = \frac{(\lambda)_{m+n}}{(\lambda)_m} \tag{22}$$

the expressions of $A_{l \ m \ n}$ and $B_{l \ m \ n}$ become

$$A_{l\ m\ n} = \frac{1}{\varepsilon!} \Gamma\left(\alpha_J\right) \frac{(\alpha_J)_{l+m+n}}{(\varepsilon+1)_l (\nu+1)_m \left(\frac{1}{2}\right)_n} \tag{23}$$

and

$$B_{l\ m\ n} = \frac{1}{\varepsilon!} \Gamma\left(\beta_J\right) \frac{(\beta_J)_{l+m+n}}{(\varepsilon+1)_l (\nu+1)_m \left(\frac{3}{2}\right)_n},\tag{24}$$

where α_J and β_J are expressed in (18).

Finally, with the help of these last equations, (23) and (24), (19) yields to (17) by noting that the triple summation of (19) is equal to the Lauricella series $F_A^{(3)}$. This completes the proof.

Theorem 2. Let the conditions $\Re(p) > 0, \Re(\mu) > -1$ and $\frac{|\chi|^2}{16} + |\lambda|^2 + |q|^2 < |p|^2$ be verified, then the undermentioned integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) M_{s,\zeta}(2\gamma \rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho
= L_{0} \Biggl\{ \frac{\Gamma(\alpha_{L})}{2\sqrt{p+\gamma}} F_{A}^{(4)} \Biggl[\alpha_{L}, \frac{1}{2} + \zeta - s, ..., ...; 2\zeta
+1, \varepsilon + 1, \nu + 1, \frac{1}{2}; x, y, z, t \Biggr]
+ \frac{q}{p+\gamma} \Gamma(\beta_{L}) F_{A}^{(4)} \Biggl[\beta_{L}, \frac{1}{2} + \zeta - s, ..., ...; 2\zeta + 1, \varepsilon
+1, \nu + 1, \frac{3}{2}; x, y, z, t \Biggr] \Biggr\},$$
(25)

where

$$L_0 = \frac{(2\gamma)^{\zeta + \frac{1}{2}}}{(p+\gamma)^{\frac{\mu+\nu+\varepsilon+1}{2} + \zeta}} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!} \frac{\left(\frac{\chi}{2}\right)^{\varepsilon}}{\varepsilon!},\tag{26}$$

$$x = \frac{2\gamma}{(p+\gamma)}, \ y = -\frac{\lambda^2}{(p+\gamma)}, \ z = -\frac{\chi^2}{4(p+\gamma)}, \ t = \frac{q^2}{(p+\gamma)},$$
(27)

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \zeta + 1 \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} + \zeta + \frac{3}{2}.$$
 (28)

Proof. The expression of the Whittaker function is given by (see Ref. 10)

$$M_{s,\zeta}(2\gamma\rho^2) = (2\gamma)^{\zeta+\frac{1}{2}} \sum_{l=0}^{\infty} \frac{\left(\frac{1}{2} + \zeta - s\right)_l}{(2\zeta+1)_l} \frac{(2\gamma)^l}{l!} e^{-\gamma\rho^2} \rho^{2l+2\zeta+1}.$$
 (29)

Substituting (29) in (32) and denoting its left-hand side by L, (32) becomes

$$L = (2\gamma)^{\zeta + \frac{1}{2}} \sum_{l=0}^{\infty} \frac{\left(\frac{1}{2} + \zeta - s\right)_l}{(2\zeta + 1)_l} \frac{(2\gamma)^l}{l!} L_l,$$
(30)

where

$$\begin{split} L_l &= \frac{\left(\frac{\lambda}{2\sqrt{p+\gamma}}\right)^{\varepsilon}}{\varepsilon!(p+\gamma)^{\frac{\mu+1}{2}+l+\zeta}} \frac{\left(\frac{\chi}{2\sqrt{p+\gamma}}\right)^{\nu}}{\nu!} \\ &\times \left\{ \frac{\Gamma\left(\alpha_L^l\right)}{2\sqrt{p+\gamma}} \sum_{m,\,n,\,k=0}^{\infty} \frac{\left(\alpha_L^l\right)_{m+n+k}}{(\varepsilon+1)_m(\nu+1)_n\left(\frac{1}{2}\right)_k} \frac{y^m}{m!} \frac{z^n}{n!} \frac{t^k}{k!} \right. \\ &+ \frac{q}{(p+\gamma)} \Gamma\left(\beta_L^l\right) \sum_{m,\,n,\,k=0}^{\infty} \frac{\left(\beta_L^l\right)_{m+n+k}}{(\varepsilon+1)_m(\nu+1)_n\left(\frac{3}{2}\right)_k} \frac{y^m}{m!} \frac{z^n}{n!} \frac{t^k}{k!} \right\}, \end{split}$$

with

$$\alpha_L^l = \alpha_L + l \ and \ \beta_L^l = \beta_L + l. \tag{31}$$

Making the use of the identity,

$$\Gamma(\lambda+l)(\lambda+l)_{m+n+k} = \Gamma(\lambda)(\lambda)_{l+m+n+k},$$
(32)

(30) can be rearranged as

$$L = (2\gamma)^{\zeta + \frac{1}{2}} \frac{\left(\frac{\lambda}{2\sqrt{p+\gamma}}\right)^{\varepsilon}}{\varepsilon!(p+\gamma)^{\frac{\mu+1}{2}+\zeta}} \frac{\left(\frac{\chi}{2\sqrt{p+\gamma}}\right)^{\nu}}{\nu!} \left\{ \frac{\Gamma(\alpha_L)}{2\sqrt{p+\gamma}} \times \sum_{l,m,n,k=0}^{\infty} \frac{(\alpha_L)_{l+m+n+k} \left(\frac{1}{2}+\zeta-s\right)_l}{(2\zeta+1)_l (\varepsilon+1)_m (\nu+1)_n \left(\frac{1}{2}\right)_k} \frac{x^l}{n!} \frac{y^m}{m!} \frac{z^n}{n!} \frac{t^k}{k!}$$
(33)

$$+\frac{q}{(p+\gamma)}\Gamma(\beta_{L})\sum_{l,m,n,k=0}^{\infty} \times \frac{(\beta_{L})_{l+m+n+k}(\frac{1}{2}+\zeta-s)_{l}}{(2\zeta+1)_{l}(\varepsilon+1)_{m}(\nu+1)_{n}(\frac{3}{2})_{k}}\frac{x^{l}}{l!}\frac{y^{m}}{m!}\frac{z^{n}}{n!}\frac{t^{k}}{k!}\bigg\}.$$
(34)

Finally, with the resulting expression deduced from (34) in terms of the Lauricella function of four variables $F_A^{(4)}$, we get the required result. This completes the proof of Theorem 2.

3. Special Cases

In this section, we recall the identities (8)-(18) of the Whittaker function given in terms of some other special functions. By taking some particular values of the index of the Whittaker function, we establish the following corollaries.

3.1. Case of $\mu = -\nu = \frac{1}{2}$

In this case, we use the well-known identity (see Ref. 12):

$$J_{-\frac{1}{2}}(z) = \sqrt{\frac{2}{\pi z}} \cos{(z)}.$$
 (35)

Corollary 1. The following integral transform holds true:

$$\int_{0}^{\infty} \cos(\chi\rho) J_{\varepsilon}(\lambda\rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= \frac{\left(\frac{\lambda}{2\sqrt{p}}\right)^{\varepsilon}}{\varepsilon!} \left\{ \frac{\Gamma\left(\frac{\varepsilon+1}{2}\right)}{2\sqrt{p}} F_{A}^{(3)}\left[\frac{\varepsilon+1}{2}, -, -, -; \varepsilon+1, \frac{1}{2}, \frac{1}{2}; x, y, z\right] \right.$$

$$\left. + \frac{q}{p} \Gamma\left(\frac{\varepsilon}{2}+1\right) F_{A}^{(3)}\left[\frac{\varepsilon}{2}+1, -, -, -; \varepsilon+1, \frac{1}{2}, \frac{3}{2}; x, y, z\right] \right\}, \quad (36)$$

where x, y, and z are given by (17), with the conditions, $\Re(p > 0)$ and $\frac{|\chi|^2}{16} + |\lambda|^2 + |q|^2 < |p|$. **Proof.** Taking $\mu = -\nu = \frac{1}{2}$ and applying Theorem 1, (48) is proved.

Corollary 2. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \cos(\chi\rho) J_{\varepsilon}(\lambda\rho) M_{s,\zeta}(2\gamma\rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho \\ &= \frac{(2\gamma)^{\zeta+\frac{1}{2}}}{(p+\gamma)^{\frac{\varepsilon+1}{2}+\zeta}} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\varepsilon!} \\ &\times \left\{ \frac{\Gamma\left(\frac{\varepsilon}{2}+\zeta+1\right)}{2\sqrt{p+\gamma}} F_{A}^{(4)} \left[\frac{\varepsilon}{2}+\zeta+1, \frac{1}{2}+\zeta-s, ..., ...; 2\zeta+1, \varepsilon\right. \\ &+ 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{q}{p+\gamma} \Gamma\left(\frac{\varepsilon+3}{2}+\zeta\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon+3}{2}+\zeta, \frac{1}{2}+\zeta-s, ..., ...; 2\zeta+1, \varepsilon\right. \\ &+ 1, \frac{1}{2}, \frac{3}{2}; x, y, z, t \right] \bigg\}, \end{split}$$
(37)

where x, y, z and t are given by (2.13) with the condition \Re (p) > 0, \Re $(\mu) > -1$ and $\frac{|\chi|^2}{16} + |\lambda|^2 + |q|^2 < |p|^2$.

Proof. Let $\mu = -\nu = \frac{1}{2}$ and with the help of Theorem 2, one obtains (51).

3.2. Case of $\mu = \nu = \frac{1}{2}$

The Bessel function is expressed as (see Ref. 12):

$$J_{\frac{1}{2}}(z) = \sqrt{\frac{2}{\pi z}} \sin(z).$$
(38)

Corollary 3. Let \Re (p) > 0, \Re $(\mu) > -1$ and $|x|^2 + |y|^2 + |z|^2 < 1$. Then

$$\int_{0}^{\infty} \sin(\chi\rho) J_{\varepsilon}(\lambda\rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= \frac{\left(\frac{\lambda}{2\sqrt{p}}\right)^{\varepsilon}}{\varepsilon!} \frac{\chi}{\sqrt{p}} \left\{ \frac{\Gamma\left(\frac{\varepsilon}{2}+1\right)}{2\sqrt{p}} F_{A}^{(3)}\left[\frac{\varepsilon}{2}+1, \dots, \dots, \frac{\varepsilon}{2}; \varepsilon+1, \frac{3}{2}, \frac{1}{2}; x, y, z\right] \right.$$

$$\left. + \frac{q}{p} \Gamma\left(\frac{\varepsilon+3}{2}\right) F_{A}^{(3)}\left[\frac{\varepsilon+3}{2}, \dots, \dots, \frac{\varepsilon}{2}; \varepsilon+1, \frac{3}{2}, \frac{3}{2}; x, y, z\right] \right\}, \quad (39)$$

where x, y and z are given in (17).

Proof. Making use of $\mu = \nu = \frac{1}{2}$ and (56) and with the aid of Theorem 1, (57) is proved.

Corollary 4. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \sin(\chi\rho) J_{\varepsilon}(\lambda\rho) M_{s,\zeta}(2\gamma\rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho \\ &= \frac{(2\gamma)^{\zeta+\frac{1}{2}} \chi}{(p+\gamma)^{\frac{\varepsilon}{2}+\zeta+1}} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\varepsilon!} \Biggl\{ \frac{\Gamma\left(\frac{\varepsilon+3}{2}+\zeta\right)}{2\sqrt{p+\gamma}} \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon+3}{2}+\zeta, \frac{1}{2}+\zeta-s, ..., ...; 2\zeta+1, \varepsilon+1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{q}{p+\gamma} \Gamma\left(\frac{\varepsilon}{2}+\zeta+2\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon}{2}+\zeta+2, \frac{1}{2}+\zeta-s, ..., ...; 2\zeta+1, \varepsilon+1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right] \Biggr\}, \end{split}$$

$$(40)$$

where x, y, z and t are given by (27).

Proof. Taking $\mu = \nu = \frac{1}{2}$ and applying Theorem 2, one obtains (60).

3.3. Case of $s = \mp r$ and $\xi = -r - \frac{1}{2}$

The Whittaker function is given as (see Ref. 12)

$$M_{\mp r, -r - \frac{1}{2}}(z) = z^{-r} \exp\left(\mp \frac{z}{2}\right).$$
 (41)

Corollary 5. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{\mu-2r} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) e^{-(p \pm \gamma)\rho^{2}+2q\rho} d\rho \\ &= L_{5} \Biggl\{ \frac{\Gamma\left(\alpha_{K}\right)}{2\sqrt{p+\gamma}} F_{A}^{(4)} \Biggl[\alpha_{L}, -r \pm r, ..., ..; -2r, \varepsilon + 1, \nu \\ &+ 1, \frac{1}{2}; x, y, z, t \Biggr] \\ &+ \frac{q}{(p+\gamma)} \Gamma\left(\beta_{L}\right) F_{A}^{(4)} \Biggl[\beta_{L}, -r \pm r, ..., ..; -2r, \varepsilon + 1, \nu \\ &+ 1, \frac{3}{2}; x, y, z, t \Biggr] \Biggr\}, \end{split}$$

where

$$L_5 = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\mu+\nu+\varepsilon}{2}-r}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{42}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{1}{2} - r \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} - r + 1.$$
 (43)

x, y, z and t are given by (27).

Proof. With the help of Theorem 2 and the use of (41), we arrive at (42). \Box
Corollary 6. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{-2r} \cos(\chi \rho) J_{\varepsilon}(\lambda \rho) e^{-(p \pm \gamma)\rho^{2} + 2q\rho} d\rho \\ &= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p + \gamma)^{\frac{\varepsilon}{2} - r} \varepsilon!} \Biggl\{ \frac{\Gamma\left(\frac{\varepsilon + 1}{2} - r\right)}{2\sqrt{p + \gamma}} \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon + 1}{2} - r, -r \pm r, ..., ..; -2r, \varepsilon + 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{q}{(p + \gamma)} \Gamma\left(\frac{\varepsilon}{2} - r + 1\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon}{2} - r + 1, -r \pm r, ..., ..; -2r, \varepsilon + 1, \frac{1}{2}, \frac{3}{2}; x, y, z, t \right] \Biggr\}, \end{split}$$

where x, y, z and t are given by (27).

Proof. Using (41) and Theorem 2 with $\mu = -\nu = \frac{1}{2}$, (71) is proved.

Corollary 7. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{-2r} \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) e^{-(p \pm \gamma)\rho^{2} + 2q\rho} d\rho \\ &= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{\varepsilon! (p+\gamma)^{\frac{\varepsilon+1}{2} - r}} \Biggl\{ \frac{\Gamma\left(\frac{\varepsilon}{2} - r + 1\right)}{2\sqrt{p+\gamma}} \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon}{2} - r + 1, -r \pm r, _, _, _; -2r, \varepsilon + 1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{q}{(p+\gamma)} \Gamma\left(\frac{\varepsilon+3}{2} - r\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon+3}{2} - r, -r \pm r, _, _; -2r, \varepsilon + 1, \frac{3}{2}, \frac{3}{2}; x, y, z, t \right] \Biggr\}, \end{split}$$
(45)

where x, y, z and t are given by (27).

Proof. Using Theorem 2 and (41) with $\mu = \nu = \frac{1}{2}$, one easily proves (76).

3.4. Case of $s = \pm \left(\frac{1}{2} - r\right)$ and $\xi = r$

The Whittaker function becomes (see Ref. 12)

$$M_{\pm\left(\frac{1}{2}-r\right)r}\left(z\right) = z^{r+\frac{1}{2}}\exp\left(\pm\frac{z}{2}\right){}_{1}F_{1}\left(2r;2r+1;\pm z\right).$$
(46)

Corollary 8. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu+2r+1} J_{\nu}(\chi\rho) J_{\varepsilon}(\lambda\rho)_{1} F_{1}(2r; 2r+1; \pm 2\gamma\rho^{2}) e^{-(\pm\gamma+p)\rho^{2}+2q\rho} d\rho
= L_{8} \left\{ \frac{\Gamma(\alpha_{L})}{2\sqrt{p+\gamma}} \right.
\times F_{A}^{(4)} \left[\alpha_{L}, \frac{1}{2} + r \pm \left(\frac{1}{2} - r \right), ..., ..; 2r+1, \varepsilon+1, \nu
+1, \frac{1}{2}; x, y, z, t \right]
+ \frac{q}{(p+\gamma)} \Gamma(\beta_{L})
\times F_{A}^{(4)} \left[\beta_{L}, \frac{1}{2} + r \pm \left(\frac{1}{2} - r \right), ..., ..; 2r+1, \varepsilon+1, \nu
+1, \frac{3}{2}; x, y, z, t \right] \right\},$$
(47)

where

$$L_8 = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\mu+\nu+\varepsilon+1}{2}+r}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{48}$$

$$\alpha_L = \frac{\mu + \varepsilon + \nu}{2} + r + 1 \quad and \quad \beta_L = \frac{\mu + \varepsilon + \nu}{2} + r + \frac{3}{2}.$$
 (49)

x, y, z and t are given by (27).

Proof. Taking $s = \pm (\frac{1}{2} - r)$ and $\xi = r$, and applying Theorem 2 and (46), we arrive at (47).

Corollary 9. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{2r+1} \cos(\chi \rho) J_{\varepsilon}(\lambda \rho)_{1} F_{1}(2r; 2r+1; \pm 2\gamma \rho^{2}) e^{-(\pm \gamma + p)\rho^{2} + 2q\rho} d\rho \\ &= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\varepsilon+1}{2} + r} \varepsilon!} \Biggl\{ \frac{\Gamma\left(\frac{\varepsilon}{2} + r+1\right)}{2\sqrt{p+\gamma}} \\ &\times F_{A}^{(4)}\left[\frac{\varepsilon}{2} + r+1, \frac{1}{2} + r \pm \left(\frac{1}{2} - r\right), -, -, -; 2r+1, \varepsilon \right. \\ &\left. +1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{q}{\left(p+\gamma\right)} \Gamma\left(\frac{\varepsilon+3}{2} + r\right) \\ &\times F_{A}^{(4)}\left[\frac{\varepsilon+3}{2} + r, \frac{1}{2} + r \pm \left(\frac{1}{2} - r\right), -, -, -; 2r+1, \varepsilon \right. \\ &\left. +1, \frac{1}{2}, \frac{3}{2}; x, y, z, t \right] \Biggr\}, \end{split}$$
(50)

where x, y, z and t are given by (27).

Proof. Using $\mu = -\nu = \frac{1}{2}$ and applying (46) and Theorem 2, we prove (89).

Corollary 10. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{2r+1} \sin(\chi\rho) J_{\varepsilon}(\lambda\rho)_{1} F_{1}(2r;2r+1;\pm 2\gamma\rho^{2}) e^{-(\pm\gamma+p)\rho^{2}+2q\rho} d\rho \\ &= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{\left(p+\gamma\right)^{\frac{\varepsilon}{2}+r+1} \varepsilon!} \Biggl\{ \frac{\Gamma\left(\frac{\varepsilon+3}{2}+r\right)}{2\sqrt{p+\gamma}} \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon+3}{2}+r,\frac{1}{2}+r\pm\left(\frac{1}{2}-r\right), -, -, -; 2r+1, \varepsilon \right. \\ &\left. +1,\frac{3}{2},\frac{1}{2};x,y,z,t \right] \\ &\left. +\frac{q}{\left(p+\gamma\right)} \Gamma\left(\frac{\varepsilon}{2}+r+2\right) \end{split}$$

$$\times F_A^{(4)} \left[\frac{\varepsilon}{2} + r + 2, \frac{1}{2} + r \pm \left(\frac{1}{2} - r \right), _, _, _; 2r + 1, \varepsilon \\ + 1, \frac{3}{2}, \frac{3}{2}; x, y, z, t \right] \right\},$$

where x, y, z and t are given by (27).

Proof. Taking $\mu = \nu = \frac{1}{2}$ and the identity (46), and applying Theorem 2, we prove (94).

3.5. Case of $s = \pm \left(\frac{a+1}{2} + b\right)$ and $\xi = \frac{a}{2}$

The Whittaker function becomes (see Ref. 12)

$$M_{\pm\left(b+\frac{1+a}{2}\right),\frac{a}{2}}(z) = \frac{L_b^{(a)}(\pm z)}{L} z^{\frac{a+1}{2}} \exp\left(\mp\frac{z}{2}\right),\tag{51}$$

where

$$L = \frac{(a+1)_b}{b!}.\tag{52}$$

Corollary 11. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu+a+1} J_{\nu}(\chi\rho) J_{\varepsilon}(\lambda\rho) L_{b}^{(a)} (\pm 2\gamma\rho^{2}) e^{-(\pm\gamma+p)\rho^{2}+2q\rho} d\rho$$

$$= \frac{(1+a)_{b}}{b!} L_{11} \left\{ \Gamma(\alpha_{L}) \times F_{A}^{(4)} \left[\alpha_{L}, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a+1}{2} + b \right), -, -, -; a+1, \varepsilon+1, \nu + 1, \frac{1}{2}; x, y, z, t \right] + \frac{q}{(p+\gamma)} \Gamma(\beta_{L}) \times F_{A}^{(4)} \left[\beta_{L}, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a+1}{2} + b \right), -, -, -; a+1, \varepsilon+1, \nu + 1, \frac{3}{2}; x, y, z, t \right] \right\},$$
(53)

where

$$L_{11} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon+a+1}{2}}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{54}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon + a}{2} + 1 \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon + a + 3}{2}.$$
 (55)

x, y, z and t are given by (27).

Proof. Leting the condition $s = \pm \left(\frac{a+1}{2} + b\right)$ and $\xi = \frac{a}{2}$, and applying Theorem 2, one easily finds (101).

Corollary 12. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{a+1} \cos(\chi \rho) J_{\varepsilon}(\lambda \rho) L_{b}^{(a)} (\pm 2\gamma \rho^{2}) e^{-(\pm \gamma + p)\rho^{2} + 2q\rho} d\rho \\ &= L_{12} \Biggl\{ \Gamma\left(\frac{\varepsilon + a}{2} + 1\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon + a}{2} + 1, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a + 1}{2} + b\right), -, -, -; a + 1, \varepsilon \\ &+ 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \right] \\ &+ \frac{2q}{\sqrt{p + \gamma}} \Gamma\left(\frac{\varepsilon + a + 3}{2}\right) \\ &\times F_{A}^{(4)} \left[\frac{\varepsilon + a + 3}{2}, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a + 1}{2} + b\right), -, -, -; a + 1, \varepsilon \\ &+ 1, \frac{1}{2}, \frac{3}{2}; x, y, z, t \right] \Biggr\}, \end{split}$$
(56)

where

$$L_{12} = \frac{(1+a)_b}{b!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{a+\varepsilon+1}{2}} \varepsilon!}.$$
(57)

x, y, z and t are given by (27).

Proof. Using $\mu = -\nu = \frac{1}{2}$ and under the conditions of Theorem 2 and the use of (51), one obtains (108).

Corollary 13. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{a+1} \sin(\chi\rho) J_{\varepsilon}(\lambda\rho) L_{b}^{(a)} (\pm 2\gamma\rho^{2}) e^{-(\pm\gamma+p)\rho^{2}+2q\rho} d\rho
= L_{13} \left\{ \Gamma\left(\frac{\varepsilon+a+3}{2}\right)
\times F_{A}^{(4)} \left[\frac{\varepsilon+a+3}{2}, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a+1}{2} + b\right), ..., ...; a+1, \varepsilon
+ 1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right]
+ \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon+a}{2} + 2\right)
\times F_{A}^{(4)} \left[\frac{\varepsilon+a}{2} + 2, \frac{1}{2} + \frac{a}{2} \pm \left(\frac{a+1}{2} + b\right), ..., ...; a+1, \varepsilon
+ 1, \frac{3}{2}, \frac{3}{2}; x, y, z, t \right] \right\},$$
(58)

where

$$L_{13} = \frac{(1+a)_b}{b!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{(p+\gamma)^{\frac{a+\varepsilon}{2}+1} \varepsilon!}.$$
(59)

x, y, z and t are given by (27).

Proof. Let $\mu = \nu = \frac{1}{2}$ and with (51), and by using Theorem 2, we find (114).

3.6. Case of $s = 0, \xi = r$

The Whittaker function is given as (see Ref. 12)

$$M_{0,r}\left(z\right) = \frac{1}{C\sqrt{2\gamma}}\sqrt{z}I_r\left(\frac{z}{2}\right),\tag{60}$$

where

$$C = \frac{1}{2^{2r}\sqrt{2\gamma}r!} \tag{61}$$

and I_r is the modified Bessel function of order r.

Corollary 14. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu+1} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) I_{r}(\gamma \rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho
= \frac{(2\gamma)^{r}}{2^{2r} r!} L_{14} \left\{ \Gamma(\alpha_{L}) F_{A}^{(4)} \left[\alpha_{L}, \frac{1}{2} + r, ..., .; 2r + 1, \varepsilon + 1, \nu \right.
\left. + 1, \frac{1}{2}; x, y, z, t \right] \right.
\left. + \frac{q}{(p+\gamma)} \Gamma(\beta_{L}) F_{A}^{(4)} \left[\beta_{L}, \frac{1}{2} + r, ..., .; 2r + 1, \varepsilon + 1, \nu \right.
\left. + 1, \frac{3}{2}; x, y, z, t \right] \right\},$$
(62)

where

$$L_{14} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon+1}{2}+r}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{63}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + r + 1 \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} + r + \frac{3}{2}.$$
 (64)

x, y, z and t are given by (27).

Proof. By taking s = 0, $\xi = r$ and using Theorem 2, one obtains (122).

Corollary 15. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho \cos(\chi \rho) J_{\varepsilon}(\lambda \rho) I_{r}\left(\gamma \rho^{2}\right) e^{-p\rho^{2}+2q\rho} d\rho \\ &= L_{15} \Biggl\{ \Gamma\left(\frac{\varepsilon}{2}+r+1\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2}+r+1,\frac{1}{2}+r,...,2;2r+1,\varepsilon\right. \\ &+ 1,\frac{1}{2},\frac{1}{2};x,y,z,t \right] + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2}+r+\frac{3}{2}\right) \\ &\times F_{A}^{(4)}\left[\frac{\varepsilon}{2}+r+\frac{3}{2},\frac{1}{2}+r,...,2;2r+1,\varepsilon+1,\frac{1}{2},\frac{3}{2};x,y,z,t\right] \Biggr\}, (65) \end{split}$$

where

$$L_{15} = \frac{(2\gamma)^r}{2^{2r}r!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\varepsilon+1}{2}+r}\varepsilon!}.$$
(66)

x, y, z and t are given by (27).

Proof. With $\mu = -\nu = \frac{1}{2}$ and the identity (60), we arrive at (127) by applying Theorem 2.

Corollary 16. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) I_{r}\left(\gamma \rho^{2}\right) e^{-p\rho^{2}+2q\rho} d\rho \\ &= L_{16} \Biggl\{ \Gamma\left(\frac{\varepsilon}{2}+r+\frac{3}{2}\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2}+r+\frac{3}{2},\frac{1}{2}+r,\ldots,2;2r+1,\varepsilon \right. \\ &\left. +1,\frac{3}{2},\frac{1}{2};x,y,z,t\right] + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2}+r+2\right) \\ &\left. \times F_{A}^{(4)} \left[\frac{\varepsilon}{2}+r+2,\frac{1}{2}+r,\ldots,2;2r+1,\varepsilon+1,\frac{3}{2},\frac{3}{2};x,y,z,t\right] \Biggr\}, \end{split}$$

where

$$L_{16} = \frac{(2\gamma)^r}{2^{2r}r!} \frac{\chi(\frac{\lambda}{2})^{\varepsilon}}{(p+\gamma)^{\frac{\varepsilon}{2}+r+1}\varepsilon!}.$$
(67)

x, y, z and t are given by (27).

Proof. As the above corollary, we take $\mu = \nu = \frac{1}{2}$, (60) and under the conditions of Theorem 2, one proves (132).

3.7. Case of s = 0 and $\xi = \frac{1}{2}$

In this case, the Whittaker function is expressed as (see Ref. 12)

$$M_{0,\frac{1}{2}}(z) = 2\sinh\left(\frac{z}{2}\right).$$
 (68)

Corollary 17. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu+1} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) \sinh(\gamma \rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= \gamma L_{17} \left\{ \Gamma(\alpha_{L}) F_{A}^{(4)} \left[\alpha_{L}, 1, ..., .; 2, \varepsilon + 1, \nu + 1, \frac{1}{2}; x, y, z, t \right] + \frac{q}{(p+\gamma)} \Gamma(\beta_{L}) F_{A}^{(4)} \left[\beta_{L}, 1, ..., .; 2, \varepsilon + 1, \nu + 1, \frac{3}{2}; x, y, z, t \right] \right\},$$
(69)

where

$$L_{17} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon}{2}+1}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{70}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{3}{2} \quad and \quad \beta_L = \frac{\mu + \nu + \varepsilon}{2} + 2. \tag{71}$$

x, y, z and t are given by (27).

Proof. With s = 0, $\xi = \frac{1}{2}$ and (137), one proves (138) by the use of Theorem 2.

Corollary 18. The following integral transform holds true:

$$\int_{0}^{\infty} \rho \cos(\chi \rho) J_{\varepsilon}(\lambda \rho) \sinh(\gamma \rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\varepsilon}{2}+1} \varepsilon!} \gamma \left\{ \Gamma\left(\frac{\varepsilon+3}{2}\right) F_{A}^{(4)}\left[\frac{\varepsilon+3}{2}, 1, ..., ...; 2, \varepsilon\right.$$

$$+ 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \right] + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2}+2\right) F_{A}^{(4)}$$

$$\times \left[\frac{\varepsilon}{2}+2, 1, ..., ...; 2, \varepsilon+1, \frac{1}{2}, \frac{3}{2}; x, y, z, t\right] \right\},$$
(72)

where x, y, z and t are given by (27).

Proof. Let $\mu = -\nu = \frac{1}{2}$ and (137), one can deduce (143) with the aid of Theorem 2.

Corollary 19. The following integral transform holds true:

$$\int_{0}^{\infty} \rho \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) \sinh(\gamma \rho^{2}) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{\left(p+\gamma\right)^{\frac{\varepsilon+3}{2}} \varepsilon!} \gamma \left\{ \Gamma\left(\frac{\varepsilon}{2}+2\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2}+2, 1, \ldots, \ldots; 2, \varepsilon\right] + 1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right] + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon+5}{2}\right) F_{A}^{(4)}$$

$$\times \left[\frac{\varepsilon+5}{2}, 1, \ldots, \ldots; 2, \varepsilon+1, \frac{3}{2}, \frac{3}{2}; x, y, z, t\right] \right\},$$
(73)

where x, y, and z are given by (27).

Proof. With $\mu = \nu = \frac{1}{2}$, s = 0 and $\xi = \frac{1}{2}$, (146) can be proved easily by using Theorem 2.

3.8. Case of $s = a + \frac{1}{4}$ and $\xi = -\frac{1}{4}$

In this case, the Whittaker function is given by (see Ref. 12)

$$M_{a+\frac{1}{4},-\frac{1}{4}}\left(z^{2}\right) = \frac{S}{\left(2\gamma\right)^{1/4}} \exp\left(-\frac{z^{2}}{2}\right) \sqrt{z} H_{2a}\left(z\right), \tag{74}$$

where

$$S = (-1)^{a} \frac{a!}{(2a)!} (2\gamma)^{1/4}, \qquad (75)$$

with $\gamma \neq 0$.

Corollary 20. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu + \frac{1}{2}} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) H_{2a}(\sqrt{2\gamma}\rho) e^{-p\rho^{2} + 2q\rho} d\rho
= L_{20} \frac{(2a)!}{(-1)^{a} a!} \left\{ \Gamma(\alpha_{L}) F_{A}^{(4)} \left[\alpha_{L}, -a, -, -, -; \frac{1}{2}, \varepsilon \right] + 1, \nu + 1, \frac{1}{2}; x, y, z, t \right]
+ \frac{q}{(p+\gamma)} \Gamma(\beta_{L}) F_{A}^{(4)} \left[\beta_{L}, 1, -, -, -; \frac{1}{2}, \varepsilon + 1, \nu + 1, \frac{3}{2}; x, y, z, t \right] ,$$
(76)

where

$$L_{20} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon}{2}+\frac{1}{4}}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{77}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{3}{4} \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{5}{4}.$$
 (78)

x, y, z and t are given by (27).

Proof. Setting $s = a + \frac{1}{4}$ and $\xi = -\frac{1}{4}$ and using (74), it's easy to deduce (76) with the help of Theorem 2.

Corollary 21. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\frac{1}{2}} \cos(\chi\rho) J_{\varepsilon}(\lambda\rho) H_{2a}(\sqrt{2\gamma}\rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= L_{21} \left\{ \Gamma\left(\frac{\varepsilon}{2} + \frac{3}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{3}{4}, -a, -, -, -; \frac{1}{2}, \varepsilon + 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t\right] \right.$$

$$\left. + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2} + \frac{5}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{5}{4}, 1, -, -, -; \frac{1}{2}, \varepsilon + 1, \frac{1}{2}, \frac{3}{2}; x, y, z, t\right] \right\},$$
(79)

where

$$L_{21} = \frac{(2a)!}{(-1)^a a!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\varepsilon}{2}+\frac{1}{4}}\varepsilon!}.$$
 (80)

x, y, z and t are given by (27).

Proof. With $\mu = -\nu = \frac{1}{2}$ and under the conditions of the above corollary, one obtains (156).

Corollary 22. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\frac{1}{2}} \sin(\chi\rho) J_{\varepsilon}(\lambda\rho) H_{2a}(\sqrt{2\gamma}\rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= L_{22} \left\{ \Gamma\left(\frac{\varepsilon}{2} + \frac{5}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{5}{4}, -a, -, -, -; \frac{1}{2}, \varepsilon\right]$$

$$+ 1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right]$$

$$+ \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2} + \frac{7}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{7}{4}, 1, -, -, -; \frac{1}{2}, \varepsilon\right]$$

$$+ 1, \frac{3}{2}, \frac{3}{2}; x, y, z, t \right]$$
(81)

where

$$L_{22} = \frac{(2a)!}{(-1)^a a!} \frac{\chi \left(\frac{\lambda}{2\sqrt{p+\gamma}}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\varepsilon}{2}+\frac{3}{4}}\varepsilon!}.$$
(82)

x, y, z and t are given by (27).

Proof. If we take $\mu = \nu = \frac{1}{2}$, and under the conditions of Corollary 20, we arrive at (160).

3.9. Case of $s = a + \frac{3}{4}$ and $\xi = \frac{1}{4}$

The Whittaker function is expressed as (see Ref. 12)

$$M_{a+\frac{3}{4},\frac{1}{4}}\left(z^{2}\right) = \frac{S'}{\left(2\gamma\right)^{1/4}} \exp\left(-\frac{z^{2}}{2}\right) \sqrt{z} H_{2a+1}\left(z\right), \tag{83}$$

where

$$S' = \frac{(-1)^a}{2} \frac{a!(2\gamma)^{1/4}}{(2a+1)!},\tag{84}$$

with $\gamma \neq 0$.

Corollary 23. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\mu + \frac{1}{2}} J_{\nu}(\chi \rho) J_{\varepsilon}(\lambda \rho) H_{2a+1}(\sqrt{2\gamma}\rho) e^{-p\rho^{2} + 2q\rho} d\rho
= L_{23} \frac{2(a+1)!}{(-1)^{a}a!} \sqrt{2\gamma} \Biggl\{ \Gamma(\alpha_{L}) F_{A}^{(4)} \Biggl[\alpha_{L}, -a, ..., ...; \frac{3}{2}, \varepsilon
+ 1, \nu + 1, \frac{1}{2}; x, y, z, t \Biggr]
+ \frac{q}{(p+\gamma)} \Gamma(\beta_{L}) F_{A}^{(4)} \Biggl[\beta_{L}, -a, ..., ...; \frac{3}{2}, \varepsilon + 1, \nu
+ 1, \frac{3}{2}; x, y, z, t \Biggr] \Biggr\},$$
(85)

where

$$L_{23} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon}{2}+\frac{3}{4}}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{86}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{5}{4} \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{7}{4}.$$
 (87)

x, y, z and t are given by (27).

Proof. This transformation is obtained if one uses s and ξ of the present case and if we apply Theorem 2.

428

Corollary 24. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\frac{1}{2}} \cos(\chi\rho) J_{\varepsilon}(\lambda\rho) H_{2a+1}(\sqrt{2\gamma}\rho) e^{-p\rho^{2}+2q\rho} d\rho$$

$$= L_{24} \left\{ \Gamma\left(\frac{\varepsilon}{2} + \frac{5}{4}\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2} + \frac{5}{4}, -a, -, -, -; \frac{3}{2}, \varepsilon + 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t\right] + \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2} + \frac{7}{4}\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2} + \frac{7}{4}, -a, -, -, -; \frac{3}{2}, \varepsilon + 1, \frac{1}{2}, \frac{3}{2}; x, y, z, t\right] \right\}, \quad (88)$$

where

$$L_{24} = \frac{2(a+1)!}{(-1)^a a!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{(p+\gamma)^{\frac{\varepsilon}{2}+\frac{3}{4}}\varepsilon!} \sqrt{2\gamma}.$$
(89)

x, y, z and t are given by (27).

Proof. Under the conditions $\mu = -\nu = \frac{1}{2}$ and the conditions of Corollary 23, we arrive at (171).

Corollary 25. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{\frac{1}{2}} \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) H_{2a+1}(\sqrt{2\gamma}\rho) e^{-p\rho^{2}+2q\rho} d\rho
= L_{25} \left\{ \Gamma\left(\frac{\varepsilon}{2} + \frac{7}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{7}{4}, -a, -, -, -; \frac{3}{2}, \varepsilon +1, \frac{3}{2}, \frac{1}{2}; x, y, z, t\right]
+ \frac{2q}{\sqrt{p+\gamma}} \Gamma\left(\frac{\varepsilon}{2} + \frac{9}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2} + \frac{9}{4}, -a, -, -, -; \frac{3}{2}, \varepsilon +1, \frac{3}{2}, \frac{3}{2}; x, y, z, t\right] \right\},$$
(90)

where

$$L_{25} = \frac{2(a+1)!}{(-1)^a a!} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{(p+\gamma)^{\frac{\varepsilon}{2} + \frac{5}{4}} \varepsilon!} \sqrt{2\gamma}.$$
(91)

x, y, z and t are given by (27).

Proof. Let the condition $\mu = \nu = \frac{1}{2}$, and under the conditions of Corollary 23, one obtains (175).

3.10. Case of $s = r - \frac{1}{2}$ and $\xi = r$

The Whittaker function becomes (see Ref. 12)

$$M_{r-\frac{1}{2},r}(z) = 2r \exp\left(\frac{z}{2}\right) z^{\frac{1}{2}-r} \gamma(2r,z),$$
(92)

with $\gamma(a, x)$ as the incomplete gamma function.

Corollary 26. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{-2r+1} \cos(\chi\rho) J_{\varepsilon}(\lambda\rho) \gamma(2r, 2\gamma'\rho^{2}) e^{-(p-\gamma')\rho^{2}+2q\rho} d\rho
= L_{26} \left\{ \Gamma\left(\frac{\varepsilon}{2}+r+1\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2}+r+1, 1, ..., ..., ...; 2r+1, \varepsilon\right.
\left.+1, \frac{1}{2}, \frac{1}{2}; x, y, z, t\right] \right.
\left.+\frac{2q}{\sqrt{p+\gamma'}} \Gamma\left(\frac{\varepsilon}{2}r+\frac{3}{2}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2}+r+\frac{3}{2}, 1, ..., ...; 2r+1, \varepsilon\right.
\left.+1, \frac{1}{2}, \frac{3}{2}; x, y, z, t\right] \right\},$$
(93)

where

$$L_{26} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\varepsilon+1}{2}+r}\varepsilon!}.$$
(94)

x, y, z and t are given by (27).

430

Proof. With the help of $\mu = -\nu = \frac{1}{2}$ and (92), and applying Theorem 2, one obtains (180).

Corollary 27. The following integral transform holds true:

$$\int_{0}^{\infty} \rho^{-2r+1} \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) \gamma(2r, 2\gamma' \rho^{2}) e^{-(p-\gamma')\rho^{2}+2q\rho} d\rho
= L_{27} \left\{ \Gamma\left(\frac{\varepsilon}{2} + r + \frac{3}{2}\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2} + r + \frac{3}{2}, 1, ..., ...; 2r + 1, \varepsilon \right.
\left. + 1, \frac{3}{2}, \frac{1}{2}; x, y, z, t \right] \right.
\left. + \frac{2q}{\sqrt{p+\gamma'}} \Gamma\left(\frac{\varepsilon}{2} + r + 2\right) F_{A}^{(4)} \left[\frac{\varepsilon}{2} + r + 2, 1, ..., ...; 2r + 1, \varepsilon \right.
\left. + 1, \frac{3}{2}, \frac{3}{2}; x, y, z, t \right] \right\},$$
(95)

where

$$L_{27} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\varepsilon!(p+\gamma)^{\frac{\varepsilon}{2}+r+1}}.$$
(96)

x, y, z and t are given by (27).

Proof. By the use of $\mu = \nu = \frac{1}{2}$, and (92), and with the help of Theorem 2, it is easy to deduce (184).

3.11. Case of $s = -\frac{1}{4}$ and $\xi = \frac{1}{4}$

In this case, the Whittaker function can be expressed in terms of the error function erf as (see Ref. 12)

$$M_{-\frac{1}{4},\frac{1}{4}}(z^2) = \frac{\exp(z^2/2)}{2}\sqrt{\pi z} erf(z).$$
(97)

Corollary 28. The following integral transforms holds true:

$$\int_{0}^{\infty} \rho^{\mu-2r+1} J_{\nu}(\chi\rho) J_{\varepsilon}(\lambda\rho) \operatorname{erf}(\sqrt{2\gamma}\rho) \operatorname{e}^{-(p-\gamma)\rho^{2}+2q\rho} d\rho
= 2\sqrt{\frac{2\gamma}{\pi}} L_{28} \left\{ \Gamma(\alpha_{L}) F_{A}^{(4)} \left[\alpha_{L}, \frac{1}{2}, ..., ...; \frac{3}{2}, \varepsilon+1, \nu \right. \\ \left. + 1, \frac{1}{2}; x, y, z, t \right] \right. \\ \left. + \frac{q}{(p+\gamma')} \Gamma(\beta_{L}) F_{A}^{(4)} \left[\beta_{L}, \frac{1}{2}, ..., ...; \frac{3}{2}, \varepsilon+1, \nu \right. \\ \left. + 1, \frac{3}{2}; x, y, z, t \right] \right\},$$
(98)

where

$$L_{28} = \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon}}{\left(p+\gamma\right)^{\frac{\mu+\nu+\varepsilon}{2}+\frac{3}{4}}\varepsilon!} \frac{\left(\frac{\chi}{2}\right)^{\nu}}{\nu!},\tag{99}$$

$$\alpha_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{5}{4} \text{ and } \beta_L = \frac{\mu + \nu + \varepsilon}{2} + \frac{7}{4}.$$
(100)

x, y, z and t are given by (27).

Proof. With $s = -\frac{1}{4}$, $\xi = \frac{1}{4}$ and the use of (97), Theorem 2 yields (188).

Corollary 29. The following integral transform holds true:

$$\begin{split} \int_0^\infty \rho^{-2r+1} \cos(\chi\rho) J_\varepsilon(\lambda\rho) \, erf(\sqrt{2\gamma}\rho) e^{-(p-\gamma)\rho^2 + 2q\rho} d\rho \\ &= 2\sqrt{\frac{2\gamma}{\pi}} \frac{\left(\frac{\lambda}{2}\right)^\varepsilon}{(p+\gamma)^{\frac{\varepsilon}{2} + \frac{3}{4}}\varepsilon!} \Biggl\{ \Gamma\left(\frac{\varepsilon}{2} + \frac{5}{4}\right) F_A^{(4)}\left[\frac{\varepsilon}{2} + \frac{5}{4}, \frac{1}{2}, -, -, -; \frac{3}{2}, \varepsilon\right] \\ &+ 1, \frac{1}{2}, \frac{1}{2}; x, y, z, t \Biggr] \end{split}$$

$$+\frac{2q}{\sqrt{p+\gamma'}}\Gamma\left(\frac{\varepsilon}{2}+\frac{7}{4}\right)F_{A}^{(4)}\left[\frac{\varepsilon}{2}+\frac{7}{4},\frac{1}{2},-,-,-;\frac{3}{2},\varepsilon\right] +1,\frac{1}{2},\frac{3}{2};x,y,z,t\right],$$
(101)

where x, y, z and t are given by (27).

Proof. Taking the conditions $\mu = -\nu = \frac{1}{2}$ and $s = -\xi = -\frac{1}{4}$, and applying (97) and Theorem 2, one arrives at (194).

Corollary 30. The following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{-2r+1} \sin(\chi \rho) J_{\varepsilon}(\lambda \rho) \, erf(\sqrt{2\gamma}\rho) e^{-(p-\gamma)\rho^{2}+2q\rho} d\rho \\ &= 2\sqrt{\frac{2\gamma}{\pi}} \frac{\left(\frac{\lambda}{2}\right)^{\varepsilon} \chi}{\varepsilon!(p+\gamma)^{\frac{\varepsilon}{2}+\frac{5}{4}}} \left\{ \Gamma\left(\frac{\varepsilon}{2}+\frac{7}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2}+\frac{7}{4},\frac{1}{2},...,.;\frac{3}{2},\varepsilon\right] \\ &+ 1,\frac{3}{2},\frac{1}{2};x,y,z,t \right] \\ &+ \frac{2q}{\sqrt{p+\gamma'}} \Gamma\left(\frac{\varepsilon}{2}+\frac{9}{4}\right) F_{A}^{(4)}\left[\frac{\varepsilon}{2}+\frac{9}{4},\frac{1}{2},..,.;\frac{3}{2},\varepsilon\right] \\ &+ 1,\frac{3}{2},\frac{3}{2};x,y,z,t \right] \bigg\}, \end{split}$$
(102)

where x, y, z and t are given by (27).

Proof. As the above corollary, we arrive at (197) if we take $\mu = \nu = \frac{1}{2}$.

4. Numerical Simulations

In this section, we compare our theoretical results of (17) and (32) given by its second number and its first number, which is the integral evaluated by using the Laguerre–Gauss quadrature. This integral,

i	x_i	$w_i e^{x_i}$
1	0.093307812017	0.239578170311
2	0.492691740302	0.560100842793
3	1.215595412071	0.887008262919
4	2.269949526204	1.22366440215
5	3.667622721751	1.57444872163
6	5.425336627414	1.94475197653
$\overline{7}$	7.565916226613	2.34150205664
8	10.120228568019	2.77404192683
9	13.130282482176	3.25564334640
10	16.654407708330	3.80631171423
11	20.776478899449	4.45847775384
12	25.623894226729	5.27001778443
13	31.407519169754	6.35956346973
14	38.530683306486	8.03178763212
15	48.026085572686	11.5277721009

Table 1.Abscissas and weight factors forLaguerre integration.

denoted by R, can be expressed as

$$R = \sum_{i=1}^{15} \omega_i e^{x_i} g(x_i), \tag{103}$$

where x_i and $\omega_i e^{x_i}$ are given in Table 1 (see Ref. 13) and $g(x_i) = x_i^{\mu} J_{\nu}(\chi x_i) J_{\varepsilon}(\lambda x_i) e^{-px_i^2 + 2qx_i}$, which relates to (17) and $g(x_i) = x_i^{\mu} J_{\nu}(\chi x_i) J_{\varepsilon}(\lambda x_i) M_{s,\zeta} (2\gamma \rho^2) e^{-px_i^2 + 2qx_i}$, which corresponds to (32).

To compare our numerical simulations and the second member of (17), we illustrate in Figure 1 both the expressions with the following calculation parameters: $\nu = 1, \mu = 1, \epsilon = 1$, where the condition of Theorem 1 is verified. One can deduce from this figure that there is an excellent agreement between the two considered evaluations.

Also, we show in Figure 2, a comparison between our theoretical and numerical results with the following set of parameters: $\nu = 1, \mu = 1, \epsilon = 1, \chi = 2$ and $\lambda = 2$. It is clear from this illustration that there is an excellent agreement between (103) and the second member of (17).

To show the equivalence between our numerical simulations and the second member of (36), we display the two formulas in Figure 1 with the following calculation parameters: $\nu = 1, \mu = 1, \zeta = 1$,



Fig. 1. Illustration of (17) as a function of: (a) λ with $\chi = 2, p = 14$ and q = 2 and (b) χ with $\lambda = 6, p = 14$ and q = 5.



Fig. 2. Illustration of (17) as a function of: (a) p with $q = 3, \chi = 2$ and $\lambda = 2$ and (b) q with $p = 20, \chi = 2$ and $\lambda = 8$.

 $s = 1, \epsilon = 1$, and $\gamma = 1$, where the condition of Theorem 2 is verified. The obtained results show good compatibility of the numerical solution obtained using the Laguerre–Gauss quadrature and the closed-form solution.

Moreover, we illustrate a comparison of our theoretical and numerical results in Figure 3 and the other calculation parameters are the same as those taken in Figure 4. From these figures, we clearly observe that our result obtained theoretically is identical to that obtained numerically.



Fig. 3. Illustration of (36) as a function of: (a) p with $q = 1, \chi = 2$ and $\lambda = 3$ and (b) q with $p = 14, \chi = 3$ and $\lambda = 3$.



Fig. 4. Illustration of (36) as a function of: (a) χ with $\lambda = 6$, p = 12 and q = 7 and (b) λ with $\chi = 8$, p = 20 and q = 1.

5. Conclusion

In this investigation, we have derived two integral transforms in terms of the Lauricella function of three and four variables, involving the product of two Bessel and Whittaker functions. Some corollaries are derived from the main theorems as particular cases. To compare our theoretical and numerical results, some numerical simulations have been done. The obtained results show that there is an excellent agreement between the numerical solution obtained using the Laguerre– Gauss quadrature and our theoretical results. These results can be used in the Fresnel diffraction by a helical axicon of many laser fields, such as Laguerre–Bessel–Gaussian beams.

6. Appendix

We use the following lemma in this study.

Lemma 1. For $\Re(p) > 0$, $\Re(\mu) > -1$, and $|x|^2 + |y|^2 < 1$, the following integral transform holds true:

$$\begin{split} &\int_{0}^{\infty} \rho^{\mu} J_{\nu}(\chi \rho) e^{-p\rho^{2}+2q\rho} d\rho \\ &= K_{0} \left\{ \Gamma\left(\alpha\right) \Psi_{2}\left[\alpha;\nu+1,\frac{1}{2};x,y\right] + \frac{2q}{\sqrt{p}} \Gamma\left(\beta\right) \right. \\ &\times \Psi_{2}\left[\beta;\nu+1,\frac{3}{2};x,y\right] \right\}, \end{split}$$

where

$$K_0 = \frac{1}{2p^{\frac{\mu+1}{2}}} \frac{\left(\frac{\chi}{2\sqrt{p}}\right)^{\nu}}{\nu!},$$
(104)

$$x = -\frac{\chi^2}{4p}, \quad y = \frac{q^2}{p} \tag{105}$$

$$\alpha = \frac{\mu + \nu + 1}{2} \text{ and } \beta = \frac{\mu + \nu}{2} + 1.$$
 (106)

Proof. By denoting the left-hand side of (104) by K and by using the identity $e^x = chx + shx$ and the series representation of the Bessel function J_{ν} of the first kind of order ν given by (4), we get

$$K = (\chi/2)^{\nu} \sum_{k=0}^{\infty} \frac{(-\chi^2/4)^k}{k! \Gamma(\nu+k+1)} K_k,$$
(107)

where

$$K_k = \int_0^\infty \rho^{\mu+\nu+2k} e^{-p\rho^2} ch(2q\rho) d\rho + \int_0^\infty \rho^{\mu+\nu+2k} e^{-p\rho^2} sh(2q\rho) d\rho.$$
(108)

Making use of the following identities (Ref. 12)

$$\int_{0}^{\infty} x^{2\alpha - 1} e^{-\beta x^{2}} sh(\gamma x) dx$$
$$= \frac{\Gamma(2\alpha)}{2(2\beta)^{\alpha}} e^{\gamma^{2}/8\beta} \left[D_{-2\alpha} \left(-\frac{\gamma}{\sqrt{2\beta}} \right) - D_{-2\alpha} \left(\frac{\gamma}{\sqrt{2\beta}} \right) \right], \quad (109)$$

with $\Re(\alpha) > -\frac{1}{2}$, $\Re(\beta) > 0$,

and

$$\int_{0}^{\infty} x^{2\alpha-1} e^{-\beta x^{2}} ch(\gamma x) dx$$
$$= \frac{\Gamma(2\alpha)}{2(2\beta)^{\alpha}} e^{\gamma^{2}/8\beta} \left[D_{-2\alpha} \left(-\frac{\gamma}{\sqrt{2\beta}} \right) + D_{-2\alpha} \left(\frac{\gamma}{\sqrt{2\beta}} \right) \right], \quad (110)$$

with $\Re(\alpha) > 0, \ \Re(\beta) > 0,$

(107) becomes

$$K = \frac{(\chi/2)^{\nu}}{(2p)^{\frac{\mu+\nu+1}{2}}} e^{\frac{q^2}{2p}} \sum_{k=0}^{\infty} \frac{\left(-\chi^2/8p\right)^k}{k!} \frac{\Gamma\left(\mu+\nu+1+2k\right)}{\Gamma\left(\nu+1+k\right)} \times D_{-(\mu+\nu+1+2k)} \left(-\sqrt{\frac{2}{p}}q\right).$$
(111)

In these last equations, $D_{-2\delta}$ is the parabolic cylinder function given in terms of the Kummer's function as

$$D_{-2\delta}(z) = \frac{\sqrt{\pi}}{2^{\delta}} e^{-\frac{z^2}{4}} \left[\frac{1}{\Gamma\left(\frac{1}{2} + \delta\right)} {}_1F_1\left(\delta; \frac{1}{2}; \frac{z^2}{2}\right) -\frac{\sqrt{2}z}{\Gamma\left(\delta\right)} {}_1F_1\left(\frac{1}{2} + \delta; \frac{3}{2}; \frac{z^2}{2}\right) \right].$$
(112)

The use of this last identity yields the following expression:

$$K = \frac{\sqrt{\pi}}{(4p)^{\frac{\mu+1}{2}}} \left(\frac{x}{4\sqrt{p}} \right)^{\nu} \left\{ \sum_{k=0}^{\infty} \frac{\left(-\frac{x^2}{16p} \right)^k}{k! \Gamma \left(\nu + 1 + k \right)} \varepsilon_{11}^k \right. \\ \left. \times F_1 \left(\frac{\mu + \nu + 1}{2} + k; \frac{1}{2}; \frac{q^2}{2} \right) \right. \\ \left. + \frac{2q}{\sqrt{p}} \sum_{k=0}^{\infty} \frac{\left(-\frac{x^2}{16p} \right)^k}{k! \Gamma \left(\nu + 1 + k \right)} \varepsilon_{21}^k F_1 \left(\frac{\mu + \nu}{2} + 1 + k; \frac{3}{2}; \frac{q^2}{2} \right) \right\},$$
(113)

where

$$\varepsilon_1^k = \frac{\Gamma\left(\mu + \nu + 1 + 2k\right)}{\Gamma\left(\frac{\mu + \nu}{2} + 1 + k\right)} \tag{114}$$

,

and

$$\varepsilon_2^k = \frac{\Gamma\left(\mu + \nu + 1 + 2k\right)}{\Gamma\left(\frac{\mu + \nu + 1}{2} + k\right)}.$$
(115)

To evaluate the expression of ε_1^k and ε_2^k , we use the identities (see Ref. 11)

$$(\lambda + m)_n = \frac{(\lambda)_{m+n}}{(\lambda)_m} \tag{116}$$

and

$$\Gamma(2z) = \frac{2^{2z-1}}{\sqrt{\pi}} \Gamma(z) \Gamma\left(z + \frac{1}{2}\right), \quad z \neq 0, \ -\frac{1}{2}, \ -1, \ -\frac{3}{2}, \dots$$
(117)

Now, in view of (114), (115), (116), and (117), K can be expressed in terms of the Humbert function of two variables ψ_2 and the required result (104) easily follows. This completes the proof.

Acknowledgments

The authors express their gratitude to the reviewers for checking the manuscript minutely.

References

- Y. I. Salamin, Fields of a Bessel-Bessel light bullet of arbitrary order in an under-dense plasma, *Sci. Rep.* 8, 1–8 (2018).
- P. A. Becker, Infinite integrals of Whittaker and Bessel functions with respect to their indices, J. Math. Phys. 50(12), 123515, 21 (2009).
- J. Choi and P. Agarwal, Certain integral transform and fractional integral formulas for the generalized Gauss hypergeometric functions, *Abstr. Appl. Anal.* Art. ID 735946, 7 (2014).
- J. Choi, P. Agarwal, S. Mathur, and S. D. Purohit, Certain new integral formulas involving the generalized Bessel functions, *Bull. Korean Math.* Soc. 51(4), 995–1003 (2014).
- J. E. Littlewood, A note on integral transforms associated with Humbert's confluent hypergeometric function, *Electron. J. Math. Anal. Appl.* 4, 259–265 (2016).
- A. Belafhal, E. M. El Halba, and T. Usman, An integral transform involving the product of Bessel functions and Whittaker function and its application, *Int. J. Appl. Comput. Math.* 6(6), Paper No. 177, 11 (2020).
- A. Belafhal, N. Nossir, and T. Usman, Integral transforms involving orthogonal polynomials and its application in diffraction of cylindrical waves, *Comput. Appl. Math.* 41(3), Paper No. 100, 21 (2022).
- A. Belafhal, E. M. El Halba, and T. Usman, An integral transform and its application in the propagation of Lorentz-Gaussian beams, *Commun. Math.* 29(3), 483–491 (2021).
- A. Belafhal, S. Chib, F. Khannous, and T. Usman, Evaluation of integral transforms using special functions with applications to biological tissues, *Comput. Appl. Math.* 40(4), Paper No. 156, 23 (2021).
- I. S. Gradshteyn and I. M. Ryzhik, *The Confluent Hypergeometric Function*. Springer, Berlin-Heidelberg (1969).
- H. M. Srivastava and P. W. Karlsson, *Multiple Gaussian Hypergeomet*ric Series. Ellis Horwood Series: Mathematics and Its Applications, Ellis Horwood Ltd., Chichester; Halsted Press [John Wiley & Sons, Inc.], New York (1985).
- I. S. Gradshteyn and I. M. Ryzhik, *Table of Integrals, Series, and Prod*ucts, 5th edn. Academic Press, Inc., San Diego, CA (1996). CD-ROM version 1.0 for PC, MAC, and UNIX computers.
- A. Belafhal, E. M. El Halba, and T. Usman, Handbook of mathematical functions with formulas, graphs, and mathematical tables, *Am. J. Phys.* 56, 958 (1988).

Index

Α

 σ -algebra, 344 absolute ideal class group, 79 additive measure, 343 additive scator coefficients, 207 Alexiewicz norm, 350 algorithm H, 173, 180, 182 algorithm *M*, 169–176, 179–180, 182-186, 188, 190-191 angular momentum, 226 Apéry set, 101, 103 Apéry constant, 10 Apostol–Bernoulli numbers, 102 Argand diagram, 200 arithmetic functions, 137, 139-140 arithmetic progression, 2, 99, 153 arithmetic sequence, 102 asymptotic notation, 3, 7 Atiyah, Michael, 8

В

b-metric space, 232–233 Banach space, 343 Banach, Stefan, 262 Bernoulli lemniscata, 228 Bernoulli number, 7, 10 Bessel function, 406, 421 beta-transform, 81 blackbody cavity, 33 Borel σ -algebra, 346 Borel distribution (BD), 368 Borel distribution series, 365 Borel measures, 345 bounded and uniformly continuous, 358 bounded finitely additive set, 358 bounded linear functional, 350 bounded linear operator, 360–361 Brun's constant, 4 Brun's theorem, 4

С

Q-contraction, 262 Caristi's fixed-point theorem, 233 Caristi-type fixed-point theorems, 235 Caristi-type inequality, 242 Cauchy sequence, 233, 239, 241, 348 Cauchy's identity, 17 Cauchy-Schwarz inequality, 378 characteristic function, 69, 350, 354 choreography, 228 Chowla–Selberg integral formula, 82, 87 Ciric, Lj. B., 264 Clausen function, 56, 61 clay mathematics institute, 6 Clements. 3 closed cubes, 350

Index

Collatz conjecture, 10, 31-32 Collatz function, 32, 38-39 color partitions of positive integer, 140common fixed point, 237 compact embedding, 356 complete metric space, 233 completion, 358 complex numbers, 152 components exponential function, 197, 201, 205 congruence for $\sigma_{Moex}(n)$, 147 congruences, 138 congruences for $\sigma_{Mex}(n)$, 142 congruent modulo n, 20conjugate gradient methods, 385, 389 conjugate theta function, 67 conjugation, 199 continuous compact, 343 continuous dense, 360 continuous embedding, 351 control functions, 329 convolution operator, 359, 361 countable dense set, 349, 353 critical line, 7 cusphere, 201, 227

D

de Polignac's conjecture, 3 Debonat, Pradip, 262 decomposition theorem, 88 Dedekind zeta function, 79 Dedekind eta function, 77–78 defined minimal odd exludant, 139 Denjoy integrable function, 344 dense and continuous embedding, 357 dense subset, 351 densely closed linear operator, 360 densely continuous subspace, 351 determinant condition, 177–178 DFT, 68 Diophantine equation, 32, 42 Dirichlet class number formula, 89 Dirichlet's theorem, 2 discrete topology, 346

distribution of primes, 2 dual space, 351 duplication formula, 59

E

efficient solutions, 327 elliptic scators, 196 elliptical trajectory, 196 energy, 226 Euclid's result, 2 Euclidean metric, 204 Euclidean rotations, 205 Euclidean space, 227, 345 Euler constant, 10, 61 Euler criterion, 175 Euler digamma function, 61 Euler product, 91 Euler's criterion, 178 Euler's formula, 198 Euler's identity, 16 Euler's recursion formula, 138 Euler's theorem, 138 Eulerian numbers, 102 even parts, 147 extended interpolative Hardy-Rogers-Geraghty contractions, 261 extended interpolative Hardy–Rogers-type Geraghty Q-contraction, 266

F

Fermat number, 9
Fermat prime, 9
Fermat's two-square theorem, 169, 171, 175
Feynman operator, 352
Fibonacci numbers, 9, 173
Fibonacci primes, 10
fixed circle, 243, 254
fixed figure, 242
fixed point, 263–266, 271–272, 274, 280, 282, 286–287, 292–295, 297–298, 300–301, 304, 307

fixed-circle, 247 fixed-circle theorems, 245 fixed-figure problem, 235 fixed-figure theorem, 250 fixed-point, 262 fixed-point set, 234, 256 fixed-point theory, 232 Fourier, 359 Fourier coefficient, 68 Fourier transformation, 343, 361 Fourier-Bessel expansion, 92 Frobenius number, 99–100, 102 Frobenius problem, 100 fuzzy metric space, 302

G

gamma function, 430 Gauss sum, 73 general Gauss sum, 73 generalized (inverse) Eisenstein formula, 63 generalized Bernoulli number, 74generalized convexity, 327 generalized Fibonacci numbers, 186 - 188generalized Fibonacci sequence, 169, 174, 187 generalized metric spaces, 256 generating function, 18-19, 138, 145, 154genus, 100 Geraghty contractions, 262 Geraghty, Michael A., 264 Goldbach, Christian, 5 Goldbach conjecture, 5 Gross-Kuelbs, 349 Gross–Kuelbs theorem, 352 group structure, 177

Η

Hölder and Minkowski inequalities, 351

Hadamard product, 313, 365-366, 378 Hardy, Godfrey Harold, 263 Hardy-Ramanujan, 21 Hardy-Rogers contraction, 262, 264 Hecke group, 173-174, 182, 191 Heegner numbers, 10 Helfgott, Harald, 5 Henstock-Kurzweil integrable, 343, 345, 349 Hilbert space, 343, 358 holomorphic Eisenstein series, 81 holomorphic function, 314 Humbert function, 439 Humbert's confluent hypergeometric series, 406 Hurwitz formula, 55 Hurwitz zeta function, 55 hypercomplex space, 195

Ι

identity matrix, 390 infinite product for theta, 91 initial gaps, 99 inner product, 350 integral transform, 405, 408, 412, 417-418, 422-424, 426, 429-430, 432-433, 436 interpolative contraction, 261, 265, 275invariant measure, 345 invex set, 327 irregular primes, 10

J

Jacobi theta functions, 64 Jacobi's identity, 17 Jacobian matrix, 386 Jing-Run Chen, 4

K

k-rowed, 15 Karapinar, Erdal, 262 Kronecker character, 73 Kuelbs lemma, 349 Kuelbs–Steadman spaces, 344 Kummer's Fourier series, 60 Kurepa's hypothesis, 31, 42–43, 46–47

\mathbf{L}

l-regular, 16 Lagrange's four-square theorem, 11 Lagrange-type densities, 330 Laguerre-Gauss quadrature method, 405, 433 Laurent expansion, 88 Lauricella series, 410 Lebesgue measure, 346-347 Lebesgue spaces, 343 Lerch formula, 60 Lerch zeta function, 55 Lerch–Chowla–Selberg formula, 77, 89 Lerch–Chowla–Selberg ingredients, 90 limit of a sequence, 356 Lorentz metric, 204 Luxemburg norm, 343

M

Maynard, James, 4 Möbius transformation, 170–171 matrix multiplication, 172 maximum modulus principle, 371 measurable functions, 347 measurable space, 345–346 Mellin-Barnes integral, 81 Mersenne number, 8 Mersenne prime, 8 mex function, 139 millennium prize problems, 6 minimal excludant of the partition, 137, 139 minimal odd exludant, 137 Mittag-Leffler function, 312, 314 Mittag-Leffler mapping, 367, 368 Mittag-Leffler-type Borel distribution series, 370, 381 modular group, 169-172, 174-175, 190 - 191modular metric space, 281–282

Mohammadi, Babak, 262 moment of inertia, 223 multiple integral functionals, 333 multiplicative representation, 208

Ν

natural compliment, 352 natural order, 354 nearest integer algorithm, 181 neural network, 254 Newton method, 387 nonabsolute integrals, 349 nonlinear equations, 388 nonlinear map, 386 nonreflexive Banach spaces, 352 number theorem, 2 numerical analysis, 31

0

odd parts, 147 odd prime, 156 open problem, 1, 3, 11 open question, 174, 186 open sets, 346 overpartition, 16, 154

Р

p-summable space, 350 parabolic starlike function, 311 parity symbol, 76 partial zeta function, 70 partition, 14, 151 partition of a positive integer, 138 Pedaprolu Murty, 5 perfect, 15 periodic Bernoulli polynomial, 61 Planck radiation law, 35 Planck's black body radiation density, 32, 47 Planck's radiation density function, 35Planck's radiation law, 36-37 plane partition, 15 Pochhammer symbol, 406

Polignac's conjecture, 47 Polymath8 project, 4 positive integers, 1 power set, 347 prime number, 2, 7, 9, 42 product measure, 344, 346 product topology, 346 projections, 203

Q

q-series, 137
quadratic reciprocity, 175
quadratic residue, 169, 171–173, 175–178, 180, 183, 188

R

 ℓ -regular overpartition, 154 ℓ -regular partition, 153 Rademacher series, 21 Ramanujan, 138 Ramanujan's general theta function, 143Ramanujan's identities, 17 Ramanujan's theta functions, 161 reciprocal relation, 59 recursion formula, 138 regular primes, 10 Reich, Simeon, 264 relatively prime, 3 representation theory, 153 restricted partition functions, 138 restricted partitions, 14 Riemann-Hecke-Bochner (RHB) correspondence, 59, 67 Riemann hypothesis, 5–6 Riemann zeta function, 5–7, 10, 54 Rogers's identities, 18 Rogers, Tiffany D., 263 rotational invariant, 346

\mathbf{S}

scator, 209, 222 scator algebra, 195, 199 scator elements, 197 scator function, 197 scator space, 221 Schweizer, Berthold, 301 self-adjoint, 360 separable Banach space, 349, 360 separable Hilbert space, 349 Sergusov, 3 singular overpartition, 162 Sklar, Abe, 301 Smith-Cornacchia algorithm, 173 starlike function, 311 Stefan-Boltzmann law, 35, 37 Stirling's formula, 40 strongly, 358 sum-of-divisors function, 82 Sylvester number, 100 Sylvester sums, 100

\mathbf{T}

tame functions, 347 tame measurable function, 347 Tao, Terence, 4 theta function, 59 theta-transformation formula, 64 three-body choreographies, 215 three-body problem, 226 topology, 346 translation-invariant, 345 triple hypergeometric function, 407triple-controlled fuzzy \hbar -metric (TCFħM) spaces, 280, 301–304 triple-controlled modular \hbar -metric space, 280 triple-controlled modular \hbar -metric (TCMħM) spaces, 279–280, 282-284, 286-287, 294-295, 297-298, 301, 305 twin prime conjecture, 3

U

uniformly bounded sequence, 356, 358 uniformly convex function, 311, 351, 355 univalent mappings, 366 unrestricted partition, 14, 151–152

\mathbf{V}

vector-controlled variational inequalities, 334

W

Wardowski, Dariusz, 265 Waring's problem, 11 weak Goldbach conjecture, 5
weakly converge, 356, 358
weakly orbitally continuous, 235–236
Whittaker function, 407, 412, 417, 425, 431, 436

Y

Yitan Zhang, 4